There is no tracing the connection of ancient nations, but by language; and therefore I am always sorry when any language is lost, because languages are the pedigree of nations.

Samuel Johnson

Index of 7	Fables	XV
Acknowle	edgments	XIX
Preface	X	XV
A Note or	1 Transcription X	XIX
Interlinea	r Morpheme-by-Morpheme Glosses X	XX
Abbreviat	ions and SymbolsXX	XIII
Grammati	ical Category LabelsXX	XIV
1.	INTRODUCTION	1
1.1	The Name of the Language	1
1.2 1.2.1 1.2.2 1.2.3 1.2.4	Ethnology Religion Society Family Life Population Distribution	3 3 4 5 7
1.3	Demography	8
1.4	Genetic Affiliation	11
1.5 1.5.1 1.5.2 1.5.3 1.5.4	Previous Research The Early Period The Nineteenth Century Turn of the Century The Twentieth Century	13 14 16 20 23
1.6 1.6.1 1.6.2 1.6.3 1.6.4	Sociolinguistic Situation Multilingualism and Language Attitudes Contexts of Use and Language Choice Language Viability Loan Words	29 30 31 32 33
1.7	Subdialects	36
2.	PHONOLOGY	45

2.1	Preliminary Considerations	45
2.2	Phonological Units	47
2.3	Phonetic Characteristics of Distinctive Segments	49
2.3.1	Consonants	49
2.3.1.1	Labials	50
2.3.1.2	Labiodentals	51
2.3.1.3	Interdental	52
2.3.1.4	Alveolars	52
2.3.1.5	Post-Alveolars	55
2.3.1.6	Palatal	56
2.3.1.7	Velars	56
2.3.1.7	Uvulars	57
2.3.1.9	Pharyngeals	58
2.3.1.9	Glottal	58 59
2.3.2	Vowels	59
2.3.2.1	Close front unrounded vowel	60
2.3.2.2	Close back rounded vowel	61
2.3.2.3	Close-mid front unrounded vowel	62
2.3.2.4	Mid central unrounded vowel	63
2.3.2.5	Close-mid back rounded vowel	63
2.3.2.6	Open front unrounded vowel	64
2.3.2.7	Open back rounded vowel	64
2.4	Phonotactics	65
2.4.1	Syllable Structure	65
2.4.2	Distribution of Non-Syllabic Segments	66
2.4.2.1	Approximants /w/ and /j/	66
2.4.2.2	Fricative /v/ and Approximant /w/	67
2.4.2.3	Voiced Pharyngeal Fricative /s/	67
2.4.2.4	Voiceless Glottal Fricative /h/	67
2.4.2.5	Consonant Clusters	68
2.4.2.5	Admissible Vowels	72
2.4.3		72
	Word Final Vowels	
2.4.3.2	Word-Initial Vowels	72
2.4.3.3	Diphthongs	73
2.4.3.4	Hiatus	74
2.4.4	Suprasegmental Features	75
2.4.4.1	Vowel Length	75
2.4.4.2	Consonant Gemination	75
2.4.4.3	Word Stress	77
2.4.4.4	Sentence Intonation	81
2.5	Segmental Morphophonology	81
2.5.1	Assimilatory Processes	81

VIII

2.5.1.1	Assimilation	82
2.5.1.2	Umlaut	83
2.5.1.3	Vowel Coloring	84
2.5.2	Dissimilatory Processes	84
2.5.3	Insertion	85
2.5.3.1	Excrescence	86
2.5.3.2	Anaptyxis	86
2.5.3.3	Prothesis and Triradicality	86
2.5.3.4	/h/-Epenthesis	87
2.5.3.5	Breaking	87
2.5.4	Lenition	88
2.5.4.1	Fricativization	89
2.5.4.2	Oral Depletion	89
2.5.5	Loss	90
2.5.5.1	Voiceless Glottal Fricative /h/	90
2.5.5.2	Аросоре	91
2.5.5.3	Elision	92
2.5.7	Other Alternations between Segments	92
2.4	-	0.0
2.6	Suprasegmental Morphophonology	93
2.6.1	Stress and Morphological Processes	93
2.6.1.1	The Abstraction and Indefinite Morphemes	94
2.6.1.2	The Foreign Plural Morpheme and the Nominal Augment	94
2.6.1.3	The Contextual Forms of the Verb	94
2.6.2	Stress and Phonological Structure	95
2.6.2.1	Geminate Simplification	95
2.6.2.2	Pretonic Reduction	95
2.6.2.3	Aphaeresis	96
2.6.2.4	Syncope	96
2.7	Allegro Rules of Neo-Mandaic	98
2.7.1	Consonants in Allegro Speech	98
2.7.1.1	Partial Anticipatory Contact Assimilation	99
2.7.1.2	Total Anticipatory Contact Assimilation	100
2.7.1.3	Partial Anticipatory Distant Assimilation	
2.7.1.4	Partial Perseverant Contact Assimilation	
2.7.1.5	Paragoge	
2.7.1.6	Nasal Lenition	
2.7.1.7	Fricativization	
2.7.1.8	Devoicing of Final Stops	
2.7.1.9	Syllable-Closing and Intervocalic /h/	
2.7.2	Vowels in Allegro Speech	
2.7.2.1	Centralization	
2.7.2.2	Umlaut	
2.7.2.3	Anaptyxis	
	r	100

2.7.2.4	Word-Final Reduction	
3.	THE NOUN	109
3.1	Structure of the Noun	110
3.1.1	Unaugmented Patterns	111
3.1.1.1	The CVC- Template	111
3.1.1.2	The CVCC- Template	113
3.1.1.3	The CVCVC- Template	116
3.1.1.4	The CCVC- Template	
3.1.1.5	The CVCCVC- Template	122
3.1.1.6	Quadriradical Patterns	
3.1.2	Augmented Patterns	
3.1.2.1	Preformative <i>m</i>	124
3.1.2.2	Preformative <i>t</i>	
3.1.2.3	Sufformative -t	
3.1.2.4	Sufformative - <i>ānā</i>	
3.1.2.5	Sufformative -āyā	
3.1.2.6	Sufformative - <i>i</i>	
3.1.2.7	Sufformative - utā / -uxtā	127
3.2	Nouns and Gender	
3.3	Nouns and Number	
3.3.1	Inflectional Plurals	
3.3.2	Lexicalized Plurals	
3.3.3	Number Marking of Loan Words	
3.3.4	Noun Compounds and Number	131
3.4	Nouns and State	131
3.4.1	The Emphatic State	
3.4.2	The Construct State and the Ezāfe	
3.4.3	The Absolute State and the L2are	
3.5	Nouns and Case-Marking	
3.5.1	Subject of a Transitive or Intransitive Verb	
3.5.2	The Object of a Transitive Verb	
3.5.3	Nouns and Phrasal Verbs	
3.5.4	Nouns as Complements and Adjuncts	137
3.6	Identifiability and Referentiality	
3.6.1	Nonmorphological Marking of Referentiality	
3.6.2	Morphological Marking of Referentiality	
3.6.3	Marking of Identifiability	
3.6.4	Genericness	
3.7	Modifiers	146
5.1	MOUTHORS	140

Х

3.7.1	Appositive Nouns	146
3.7.2	Adjectives	
3.7.2.1	Attributive Adjectives	146
3.7.2.2	Predicative Adjectives	
3.7.2.3	Absolute and Relative Comparison	
3.7.3	Quantifiers	
3.7.3.1	Numerals	
3.7.3.2	Non-Numeral Quantifiers	
3.7.4	Nouns and Possession	
3.7.4.1	Pronominal Clitics	
3.7.4.2	The Construct Chain	
3.7.4.3	Possession Indicated By Preposition	
3.7.4.4	The Relative Pronoun <i>d</i>	
3.7.4.5	<i>Māl</i> Construction	
4.	PRONOUNS	155
4.1	Personal Pronouns	
4.1.1	Independent Personal Pronouns	
4.1.2	Enclitic Personal Pronouns	
4.1.2.1	Pronominal Object Suffixes	
4.1.2.2	Pronominal Adjuncts and Complements	
4.1.2.3	Pronouns and Possession	
4.1.2.4	Personal Pronouns on Quantifiers	
4.1.3	Reflexive Pronouns	
4.1.3.1	Reflexive Pronouns as Anaphoric Objects	
4.1.3.2	Emphatic Uses of the Reflexive Pronoun	
4.1.3.3	Reflexive Pronouns and Possession	
4.1.4	Reciprocal Pronouns	
4.2	Demonstrative Pronouns	
4.2.1	Locative Demonstrative Pronouns	
4.2.2	Anomalous Forms of the Demonstrative Pronoun	163
4.3	Subordinating Conjunctions	
4.3.1	Indefinite Antecedents	
4.3.2	Definite Antecedents	
4.3.2.1	Non-Restrictive Relative Clauses	
4.3.2.2	Restrictive Relative Clauses	
4.3.3	The Relativizer	
4.3.4	Headless Relative Clauses	
4.4	Indefinite Pronouns	167
4.4.1	Indefinite Pronouns Derived from Nouns	
4.4.2	Indefinite Pronouns Derived from Quantifiers	168

Table of	Contents
----------	----------

4.4.3	Compound Indefinite Pronouns	169
4.5	Interrogative Clauses	
4.5.1	Interrogative Pronouns	170
4.5.1.1	Interrogative Pronoun as Subject	171
4.5.1.2	Interrogative Pronoun as Object	171
4.5.2	Other Question Words	
5.	THE VERB	174
5.1	The Structure of the Verb	
5.1.1	Verb Classes	
5.1.1.1	Thematic Vowel Classes	
5.1.1.2	Root Consonant Classes	
5.1.1.3	Verbal Stem Classes	
5.1.1.4	Verb Category Classes	
5.2	Tense and Aspect	239
5.2.1	The Perfective	239
5.2.1.1	The Simple Past	239
5.2.1.2	The Past Perfect	240
5.2.1.3	The Resultative-Stative	240
5.2.2	The Imperfective	241
5.2.2.1	The Simple Present	241
5.2.2.2	The Progressive Present	241
5.2.2.3	The Habitual Present	242
5.2.2.4	The Habitual Past	242
5.2.2.5	The Narrative Present	243
5.2.2.6	The Future	243
5.3	Moods and Modalities	
5.3.1	The Indicative Mood	
5.3.1.1	Declarative Statements	
5.3.1.2	Interrogative Statements	
5.3.2	The Subjunctive Mood	
5.3.2.1	The Deontic Modality	
5.3.2.2	The Optative Modality	
5.3.2.3	The Potential Modality	
5.3.3	The Imperative	
5.3.3.1	Commands	
5.3.3.2	Negative Commands and Prohibitions	247
5.3.4	Moods and Adverbial Clauses	
5.3.4.1	Time Clauses	
5.3.4.2	Circumstantial Clauses	
5.3.4.3	Purpose Clauses	

XII

5.3.4.4	Reason Clauses	
5.3.4.5	Simultaneous Clauses	
5.3.4.6	Conditional Clauses	
5.4	Voice	251
5.4.1	The Active Voice	
5.4.2	The Middle Voice	
5.4.3	The Passive Voice	
5.4.3.1	The Analytic Passive	
5.4.3.2	The Impersonal Passive	
	-	
6.	CONCLUSION	256
6.1	Classical Mandaic and the Modern Dialects	256
6.1.1	Phonology	
6.1.2	Morphology	
6.1.2.1	The Noun	
6.1.2.2	The Verb	
6.1.3	Syntax	
6.1.4	Lexicon	
6.2	Conclusions and Desiderata	
7.	TEXT COLLECTION	
Text I:	Speech, June 13, 1999	273
Text II:	Meeting Lady Drower	276
Text III:	Two Trips to Shushtar	278
Text IV:	The Bridge of Shushtar	
Text V:	The City of Mice who Eat Iron	
Texts VI-	X: Short Texts	
8.	LEXICON	
BIBLIOGR	АРНҮ	
INDEX		373

Index of Tables

Table 1:	Kinship Terms	6
Table 2:	Census of Mandaean Families, 1877	9
Table 3:	Loan Words in Neo-Mandaic	34
Table 4:	Dialect Differences	37
Table 5:	Swadesh List of 207 Common Terms in Neo-Mandaic	39
Table 6:	Phoneme Inventory (Consonants)	48
Table 7:	Transcription Scheme (Consonants)	48
Table 8:	Phoneme Inventory (Vowels)	49
Table 9:	Phonemic Transcription Scheme (Vowels)	49
Table 10:	Some Minimal and Near-Minimal Pairs (Vowels)	60
Table 11:	Independent Pronouns and Stress	79
Table 12:	Allomorphs of the Proclitic Morphemes	93
Table 13:	Cardinal Numbers	149
Table 14:	Independent Personal Pronouns	156
Table 15:	Objective / Possessive Pronominal Clitics	157
Table 16:	Quantifiers with Enclitic Pronominal Suffixes	159
Table 17:	Inflected Forms of the Reflexive Pronoun	159
Table 18:	Forms of the Demonstrative Pronouns	162
Table 19:	$(a \sim o)$ Thematic Vowel Class	176
Table 20:	$(e \sim o)$ Thematic Vowel Class	176
Table 21:	$(o \sim o)$ Thematic Vowel Class	177
Table 22:	Personal Suffixes on the Strong Verb	180
Table 23:	The Strong Perfective with Object Suffixes	181
Table 24:	The Strong Imperative with Object Suffixes	182
	The Strong Imperfective with Object Suffixes	
Table 26:	Personal Suffixes on the I-Weak Verb	184
Table 27:	The I-Weak Perfective with Object Suffixes	184
Table 28:	The Imperative of the I-Weak with Object Suffixes	185
Table 29:	Irregular I-Weak Perfectives with Object Suffixes	185
	Irregular I-Weak Imperatives with Object Suffixes	
Table 31:	The I-Weak Imperfective with Object Suffixes	186
	Personal Suffixes on II-Weak and Geminate Verbs	
Table 33:	II-Weak & Geminate Perfectives with Object Suffixes	188
	II-Weak & Geminate Imperatives with Object Suffixes	
Table 35:	II-Weak & Geminate Imperfectives with Object Suffixes	189
Table 36:	The Perfective and Imperative of <i>atā</i> ~ <i>ati</i> (<i>āti</i>)	191

Table 37:	Personal Suffixes on III-Weak Verbs	192
	The III-Weak Perfective with Object Suffixes	
Table 39:	The III-Weak Imperative with Object Suffixes	194
Table 40:	The III-Weak Imperfective with Object Suffixes	194
Table 41:	The Perfective of <i>abā</i> (<i>bāyi</i>)	195
Table 42:	The Imperfective of <i>abā</i> (<i>bāyi</i>)	195
Table 43:	Perfective and Imperative of <i>ezgā</i> ~ <i>ezgi (āzi)</i>	196
Table 44:	Imperfective of <i>ezgā</i> ~ <i>ezgi</i> (āzi)	196
	Perfective of <i>tammā</i>	
Table 46:	Imperfective of <i>yād</i>	198
	Perfective of <i>yād</i>	
	D-Stem Verbs	
Table 49:	G-Stem Roots with D-Stem Equivalents	201
	Personal Suffixes on the Strong <i>D</i> -Stem Verb	
Table 51:	The Strong <i>D</i> -Stem Perfective with Object Suffixes	203
Table 52:	The Strong <i>D</i> -Stem Imperative with Object Suffixes	204
Table 53:	The Strong <i>D</i> -Stem Imperfective with Object Suffixes	204
	Personal Suffixes on the I-Weak D-Stem Verb	
Table 55:	The I-Weak D-Stem Perfective with Object Suffixes	205
Table 56:	The I-Weak D-Stem Imperative with Object Suffixes	206
	The I-Weak <i>D</i> -Stem Imperfective with Object Suffixes	
	The Paradigm of <i>šayyel</i> ~ <i>šayyel</i> (<i>amšiyel</i>)	
	Personal Suffixes on the III-Weak D-Stem Verb	
Table 60:	The III-Weak <i>D</i> -Stem Perfective with Object Suffixes	208
	The III-Weak <i>D</i> -Stem Imperative with Object Suffixes	
Table 62:	The III-Weak <i>D</i> -Stem Imperfective with Object Suffixes	209
	Personal Suffixes on Quadriradical Verbs	
	C-Stem Verbs	
Table 65:	G-Stem Roots with C-Stem Equivalents	213
	Personal Suffixes on the Strong <i>C</i> -Stem Verb	
	The Strong C-Stem Perfective with Object Suffixes	
	The Strong C-Stem Imperative with Object Suffixes	
	The Strong <i>C</i> -Stem Imperfective with Object Suffixes	
Table 70:	Personal Suffixes on the III-Weak C-Stem Verb	218
	The III-Weak C-Stem Perfective with Object Suffixes	
	Perfective of <i>atti</i> ~ <i>atti</i> (<i>matti</i>) with Object Suffixes	
	The III-Weak C-Stem Imperative with Object Suffixes	
	Imperative of <i>atti</i> ~ <i>atti</i> (<i>matti</i>) with Object Suffixes	
	The III-Weak C-Stem Imperfective with Object Suffixes	
Table 76:	Imperfective of <u>atti</u> ~ <u>atti</u> (matti) with Object Suffixes	221
Table 77:	<i>tG</i> -Stem Verbs	221
	<i>G</i> -Stem Roots with <i>tG</i> -Stem Equivalents	
	Personal Suffixes on the Strong <i>tG</i> -Stem Verb	
Table 80	Personal Suffixes on the Verb "To Baptize"	225

Index of Tables

Table 81:	Some Examples of Phrasal Verbs	227
Table 82:	The Short Copula	230
Table 83:	The Long Copula (and its Negation)	231
Table 84:	The Long Copula with Proclitic <i>l</i> - (and its Negation)	232
Table 85:	The Verb "To Be" with Proclitic <i>l</i> - (and its Negation)	232
Table 86:	The Copula in the Analytic passive	233
Table 87:	The Modal Verb "To Want" with the Verb "To Say"	235
Table 88:	The Modal Verb 'To Permit' with the Verb 'To Speak'	236
Table 89:	The Modal Verb 'To Be Able' with the Verb 'To Speak'	237
Table 90:	The Modal Particle "Must" with the Verb "To Speak"	238
Table 91:	The G-Stem Passive Participles	253

Acknowledgments

I first learned about the Mandaeans of Iraq and Iran through my mother, Janet Elizabeth Haberl, who cultivated my interests and inspired me to always pursue them, no matter where they might lead me. Not unlike many other scholars of the ancient Near East, I developed an interest in the region, including its history, religions, and languages, at a very early age. These interests-particularly those related to its languages and religions—were not entirely satisfied by the sort of literature that was directed towards children, and so I also developed an interest in ancient magical and ritual practices, in the hope of acquiring a fuller picture of the world at which the literature only hinted. Many concerned parents would have attempted to stifle such an interest; my mother, who was a public school teacher and knew well the importance of stimulating intellectual curiosity, attempted to cultivate these interests and channel them into more scholarly venues. In addition to introducing me to the world of museums and libraries, she bought me a reprint of E.A. Wallis Budge's Amulets and Superstitions, which contains a brief discussion of the Mandaeans and the British Museum's collection of Mandaic amulets.¹ At the time, I was intrigued by this survival of this ancient faith, but had no idea that the political situation in the Middle East had brought a small community of its adherents to within thirty miles of my hometown.

This same book (despite all of its flaws, which were myriad) also inspired me to study epigraphy with William A. Ward, the former Dean of Faculty at the American University of Beirut and Director of the Program in Ancient Studies at Brown University. At one point during the course of my studies, Ward reminisced about the students to whom he had previously taught epigraphy in Beirut, some of whom

¹ E.A. Wallis Budge, *Amulets and Superstitions*, 1st edition reprint (New York: Dover, 1978), 239–49.

were descendents of the Assyrian and Chaldean Christians that had settled in Lebanon in the early part of the century and who spoke Neo-Aramaic natively. These students had a unique perspective on the texts we studied in class; in his words, "they were able to read these inscriptions as if they were reading the Sunday *Times*." It was then that the germ of this project was first sown in my mind. Unfortunately, Ward passed away in 1996, and my mother passed away four years later. I profoundly regret that the two people who were most influential in my decision to pursue these studies are not alive to share in the progress of the work they inspired.

It was not until after I had begun my studies at Harvard University that I had the opportunity to meet speakers of Neo-Aramaic, in the form of the very same community that had attracted my attention so many years before and which had come to settle in my own backyard. In 1999, I was approached by James F. Coakley, who was organizing the Thirteenth International Conference of the ARAM Society on the Mandaeans in concert with Jorunn J. Buckley, the foremost scholar in North America on the Mandaean religion and the modern community. He enlisted me to acquire the necessary permits for the conference and to aid the participants (many of whom were Mandaeans from Iraq, Iran, and the diaspora community) in finding a place to stay during the conference. In the course of our search to find appropriate places for the priests to stay and perform various rituals, including a baptism in the Charles River, we sought the help of two Mandaeans: Nasser Sobbi, to inspect these locations to ensure that they were suitable for the priests, and Wisam Breegi, to test the waters of the river to ensure that they were potable and met exacting standards of ritual purity. It was in this capacity that I was first introduced to the modern Mandaean community and my chief informant.

While it need hardly be mentioned that this grammar would not have been possible without Nasser Sobbi's participation, I feel compelled to do proper justice to his role in its development. With limitless hospitality, he opened his home to me and welcomed me into the fold of his family, and with the patience of a saint he tolerated the barrage of questions to which I subjected him, many of which must have seemed peculiarly trivial (if not downright peculiar). Without his patient support and that of his family, including his wife Shukrieh, his son Isa, and his daughters Freshteh and Juliette, I would have neither been able to collect this data nor process it in such a short period of time.

I feel uniquely honored to have worked with an informant who was so knowledgeable, not only in matters related to his own language and culture, but in practically every sphere of life. In subsequent encounters with members of the Mandaean community, including Sinan Abdullah, Lamea Abbas Amara, Mamoon and Shafia Aldolaimi, Abdolkarim and Saeed Moradi, Suhaib Nashi, Qais Saidi, and many others who have taken the time to interact with me and encourage me in my studies, I discovered that although Sobbi's hospitality and generosity were far from uncommon among the Mandaeans, he was cherished and esteemed by all who knew him. It was not long before I came to realize exactly why Nasser Sobbi is held in such high esteem by the members of his community—my own life has been immeasurably enriched by knowing him, far beyond the subject matter of this grammar.

Just as this project would not have been possible without the contributions of Sobbi and the Mandaean community, it would also not have been possible without the knowledge base imparted to me by my professors. I owe John Huehnergard for the present status of my knowledge about comparative Semitics and the historical grammar of the various Semitic languages. He supported my decision to pursue this extremely demanding project, and his support carried me to its successful completion. I was first introduced to Neo-Aramaic dialectology by Wolfhart P. Heinrichs, who taught me Turoyo and inspired me to examine about the other Neo-Aramaic dialects in a comparative light. I was first introduced to the Iranian languages and cultures by Antonio Panaino, who taught me Old Persian through the grammar written by Prods Oktor Skjærvø and inspired me to pursue the study of the Iranian languages in addition to the Semitic languages. Consequently, when I came to Harvard I welcomed the opportunity to learn much more from Skiærvø himself. In addition to extending my studies in Iranian languages, Skjærvø introduced me to Iranian religious traditions such as Zoroastrianism and Manichaeism. I also consider myself privileged to have had the ability to study with other scholars such as Jo Ann Hackett, who endowed me with a rigorous methodology for the study of individual classical and epigraphic Semitic languages, Wheeler M. Thackston, who tutored me in the languages of the modern Middle East such as Persian and colloquial Levantine Arabic, and Lisa Lavoie, who taught me the linguistic field methods that I would use in the preparation of this grammar. That they succeeded in imparting anything at all to me is a testament to their skill in pedagogy.

My own efforts at scholarship eventually drew the attention of other scholars both Mandaean and non-Mandaean, including Sabah Aldihisi, John Bolender, Julie Ebadirad, and John Harnsberry, who contributed so much to my understanding of the modern Mandaean community and its precarious position within the Middle East today. I have also had the opportunity to present the results of my research at the meetings of the Harvard Semitic Seminar, the American Oriental Society, and the North American Conference on Afroasiatic Linguistics, and benefited immensely from the discussions that ensued on each occasion. In the course of writing this grammar, both my colleagues and the members of my committee reviewed my drafts and offered much advice and insightful criticism. In this regard I would particularly like to thank my good friends Yuhan S.D. Vevaina and Melissa Haynes, who also performed an invaluable service by putting me up (and putting up with me) on the numerous occasions when I returned to Boston from "the field" to meet with my advisors and consult the libraries at Harvard. As I prepared the manuscript for publication, I also came to appreciate the support of the series editor, Otto Jastrow, and the tireless attention of the staff at Harrassowitz, particularly Reinhard Friedrich and Miriam Würfel. Obviously, despite the help I received from these and other individuals, all mistakes and omissions to be found within it are mine alone.

I would also not have been able to finish this grammar without the generous support of my employers over the years. Throughout the eight years that I was engrossed in my graduate studies, I worked in a variety of capacities for the Semitic Museum at Harvard University, and would like to thank its Director, Lawrence E. Stager, Assistant

Acknowledgments

Director Joseph A. Greene, Assistant Curator James A. Armstrong, and Administrative Assistant Dena Davis. I was also employed by the Shelby White–Leon Levy Program for Archaeological Publications and would like to thank Shelby White, Philip J. King, and Kimberley Connors-Hughes for all of the help they have given me over the years. During this time I also worked for Vieco, LLC, and would like to thank its proprietors, May and Greg Guitteau. In my final year, I received a dissertation completion fellowship from the Harvard Graduate Society, which permitted me to focus exclusively on writing. This fellowship was made possible through a generous grant from Sidney R. Knafel.

I would also like to thank my colleagues among the faculty and staff at the Center for Middle Eastern Studies (CMES) at Rutgers, the State University of New Jersey. I am much obliged to my colleagues Peter Golden, Jawid Mojaddedi, and our Administrative Assistant Aretha Crayton for their help and support, without which I would not have been able to raise the requisite subvention for the publisher. I am also very much indebted to the faculty of the newly established Department of African, Middle Eastern, and South Asian Languages and Literature (AMESALL), my colleagues Richard Serrano, Ousseina Alidou, Alamin Mazroui, Paul Sprachman, our Administrative Assistant Stephanie Gutierrez, and many other present and former colleagues at Rutgers.

I cannot conclude these remarks without acknowledging the emotional (and at times financial) support that I have received over the years from the administrative staff at the Department of Near Eastern Languages and Civilizations, including Jennifer Petrallia, Maggie Stanley, Kathleen Cloutier, Kim Sue De Wall, Mary Medlin, and Brenna Wells. I owe a debt of gratitude to all of these individuals.

> C.G. Häberl Astoria, NY, 2008

Preface

This work is a description of a previously undocumented dialect of Neo-Mandaic, a Semitic language belonging to the Eastern subgroup of Aramaic dialects. This description is based upon newly collected texts, and constitutes the most thorough description of any dialect of Neo-Mandaic.

The data contained within are unique for numerous reasons. Neo-Mandaic, which represents the latest stage of the phonological and morphological development of Classical Mandaic, is the only surviving dialect of Aramaic directly descended from any of the attested dialects of Late Antiquity. While vernacular dialects of Aramaic do survive elsewhere in the Middle East, the other dialects thus far documented do not appear to be descended from any of the attested literary dialects. Furthermore, in most other dialects, the verbal system has been completely restructured; the synthetic forms characteristic of earlier dialects of Aramaic such as the suffix conjugation, the prefix conjugation, the imperative, and the entire system of derived stems have been replaced by new analytic forms. Neo-Mandaic preserves most of the original forms with the exception of the prefix-conjugation. Finally, as the modern reflex of Classical Mandaic, the liturgical language of the Mandaean religion, Neo-Mandaic deserves to be considered as both a living language of the modern Middle East and also the vehicle of one its greatest religious traditions, much like Hebrew, Arabic, and Persian. It is my belief that an increase in our understanding of the living language can only lead to a renewed appreciation of Mandaic as a classical language.

Unfortunately, contemporary attitudes towards Neo-Mandaic, both within the Mandaean community and outside it, range from the blithely indifferent to the hostile. The Mandaeans' Arab neighbors describe it as a kind of 'mixed-up' or 'backwards' Arabic, and allophone Mandaeans describe it as a kind of Persian or Arabic jargon, overlain with a few Mandaic words. The scholar most responsible for our knowledge of the modern Mandaean community, Ethel S. Drower, described the vernacular language in pejorative terms throughout her career, and even Rudolf Macuch, the first scholar to document a Neo-Mandaic dialect, called its value into question, except in regard to its potential to illustrate aspects of the written language. Although I too embarked upon my study of the modern language with more than a few concerns, engendered by the educated opinions of scholars such as these, I have since observed the value of documenting Neo-Mandaic for its own right, not merely for the study of the classical language and other related Aramaic dialects, and it is my hope that my work will restore a sense of perspective regarding the language and demonstrate its value to both the scholarly and the Mandaean communities.

Mandaic is severely endangered today, and all signs indicate that it will become extinct with the current generation of speakers. As an account of an endangered language, this work also addresses one of the chief concerns of linguistics in the 21st century, namely the impending loss of the majority of the world's languages and the concomitant blow to both linguistic and cultural diversity that it represents. While it will not be possible to save all of the world's languages from extinction, we may still have an opportunity to document many of them before they disappear without a trace. This work is my own modest contribution to this task.

Like other first attempts at description, my efforts to write this grammar were confounded by the lack of an appropriate model. Despite the two collections of texts published by Macuch, which were furnished with sketch grammars, and the information on Neo-Mandaic contained within his *Handbook of Classical and Modern Mandaic*, many features of the grammar of Neo-Mandaic are documented for the first time herein. I have opted to follow the most current standards for language documentation, even though these standards have not been extensively applied to other languages of the Middle East. As the methodology of language documentation evolves, it is inevitable that field linguists will discover some practices that apply well to one language or language family do not necessarily apply as well to another language or language family. It is my hope that I have successfully balanced these standards with the demands and reality of documenting an endangered Middle Eastern language like Neo-Mandaic.

One minor deviation from these standards is the unfortunate fact that nearly all of the information contained in this grammar has been collected from one individual, Nasser Sobbi of Flushing, New York. This was necessitated by the small size of the Mandaean community in the United States and the even smaller number of Mandaeans who speak Neo-Mandaic in their daily life. While his wife Shukrieh, who is also a fluent speaker of Mandaic, was willing to communicate in Mandaic with me, and often contributed vocabulary during the rare moments when Sobbi himself was at a loss for words, she was not comfortable with being taped. Sobbi's brother Dakhil Shooshtary also speaks Mandaic fluently, but his work schedule prevented him from meeting with us during the times that I was visiting with Sobbi and his family. I did speak with him over the phone on several occasions during this period, and also communicated with Abdolkarim Moradi of Syosset, NY, who also speaks Neo-Mandaic fluently. Moradi was thrilled that I had taken an interest in his language and encouraged me (in Mandaic) to continue my studies, but was also unable to meet with me. In the future I hope to collect information from these two individuals to complement that which I have already collected from Sobbi. Fortunately, for the time being, scholars interested in Neo-Mandaic dialectology may avail themselves of the two text collections published by Macuch from his two informants.

During my early recording sessions in 2003, I made use of a Sony MZ-R30 Portable MiniDisc Recorder equipped with a noise-canceling microphone to record my sessions with Sobbi. I was unhappy with the quality of the recordings that were thus produced and found it difficult to reproduce them or convert them to other formats. I decided to adopt the Moving Pictures Experts Group Audio Layer 3 (MP3) format, as recordings in this format preserve most of their original fidelity at one tenth of their original size, making samples of the texts I had collected more portable and more easily reproduced. I purchased a Macintosh iBook laptop computer and recorded subsequent sessions with

SoundStudio 2.2.4, a freeware audio recording and editing tool developed by Felt Tip software. I then converted the resulting (uncompressed) data in Audio Interchange File Format (AIFF) generated by SoundStudio into MP3 format with iTunes, a proprietary digital media player application developed by Apple Computer. Although the MP3 files are much smaller in size, making them more easy to store and distribute, the compression involved in their conversion compromises the data, rendering them useless for phonological analysis. For this reason, I have retained the original uncompressed files and have arranged to have them archived at the Semitisches Tonarchiv of Ruprecht-Karls-Universität Heidelberg (http://www.semarch.uni-hd.de/index.php4).

In addition to the Swadesh list (Table 5) and the ten texts which can be found in the text collection, I have also collected a number of other texts which I hope to publish separately. With the aid of these texts, and the data gathered by Drower and Macuch, in addition to the samples of Neo-Mandaic found in Nicholas Siouffi and Jacques de Morgan (see §1.5), I intend to produce a complete reference grammar of Neo-Mandaic. Such a reference grammar would necessarily require a more comparative and diachronic approach than the present work, the focus of which is primarily descriptive and synchronic, and thus restricted primarily to the phonology, inflectional paradigms, and morphosyntax of one particular dialect as derived from the texts in the accompanying text collection.

After a general introduction to the language, its speakers, and the history of scholarship, the phonology is described in the second section. The third section is dedicated to the noun phrase, the fourth to the pronouns, and the fifth chapter to the verb phrase. Section six concludes the discussion of the grammar some general observations about the language. Most of the examples in these chapters are culled from a collection of ten texts (five long texts, and five shorter ones), which are included in the second volume; the text of origin for examples drawn from the text collection is cited after each example.²

² Some of the data, such as paradigms, isolated vocabulary items, and a few examples of morphosyntactic constructions in context, were drawn from other sources such as the ones mentioned above.

The final portion of the grammar is a lexicon, which is followed by a brief bibliography.

A Note on Transcription

For ease of reference and comparison, most of the words given in this grammar have been transcribed according to the conventions generally used for their language of origin. All Neo-Mandaic words, including loan words, follow the phonemic transcription system outlined in Tables 7 and 9. Classical Mandaic proper nouns that are still in common use (such as the titles of religious texts) are likewise transcribed according to this system. Other Classical Mandaic words are printed in **bold type**, according to the conventions adopted by Drower and Macuch in their dictionary, to distinguish them from Neo-Mandaic.

In several places throughout the grammar (but primarily in section 2), the International Phonetic Alphabet is employed to illustrate the phonology of Neo-Mandaic. Phonemic transcriptions are enclosed within /slashes/; phonetic transcriptions are enclosed within [brackets]. On rare occasions (e.g. in the Persian words derived from Arabic), a transcription reflecting the orthographic representation of a given phoneme or word will be indicated within <a href="mailto: angle brackets>.

The transcription system adopted for examples from Arabic and Persian is also outlined in Tables 1 and 2. Transcriptions of Arabic vowels reflect the phonemic distinction between quantitatively long and short vowels, $a : \bar{a}$, $i : \bar{i}$, and $u : \bar{u}$. This distinction is not characteristic of Persian, which distinguishes between lax and tense vowels, $a : \bar{a}$, e : i, and o : u. The transcription of vowels in Persian words (including Persian words of Arabic origin) reflects this distinction rather than the quantitative distinction of Arabic. Likewise, as the transcription of Persian words (including those derived from Arabic) represents phonemic distinctions rather than orthographic ones, the consonants $\langle h \rangle$ and $\langle h \rangle$ will always be represented as $\langle h \rangle$, $\langle s \rangle$, $\langle s \rangle$ and $\langle t \rangle$ will always be represented as $\langle s \rangle$, and $\langle z \rangle$, $\langle z \rangle$, $\langle d \rangle$, and $\langle d \rangle$ will always be represented as $\langle z \rangle$, regardless of their orthographic representation. The citation form for verbs in Neo-Mandaic within the text consists of the three principal parts (i.e. the perfective stem, the imperative stem, and the active participle), e.g. $\partial hab \sim \partial hob (\bar{a}heb)$ 'to give.' Within tables and the lexicon, verbs are cited and arranged according to their consonantal root, e.g. '-h-b 'to give.' The citation form for Persian verbs is the infinitive, e.g. $d\bar{a}dan$ 'to give.'

Whenever the original form of a loan word is offered, the transcription generally represents the standard form of the language of origin (e.g. Modern Standard Arabic) rather than the specific dialect that yielded the word (e.g. the colloquial dialect of Khorramshahr). Exceptions are made for purely colloquial words with no equivalents in the standard language. While this has the effect of obscuring the precise origin of these loan words, it reflects the current state of knowledge on the Arabic of Khorramshahr, which is limited.

Commonly known proper names and other familiar words are cited according to their conventional English-language form without any diacritics. Hence, the name of the capital of the Ninawa governorate in northern Iraq is given as 'Mosul,' not *al-Mawşil*. The names of contemporary Mandaeans are also reproduced in their conventional English-language forms without diacritics, e.g. 'Nasser Sobbi' instead of 'Nāṣir Ṣubbī.'

Interlinear Morpheme-by-Morpheme Glosses

Samples of Neo-Mandaic are given a line-by-line analysis with interlinear glosses, utilizing a slightly modified version of the Leipzig Rules,³ followed by a free translation. I opted to use the Leipzig Rules because they reflect current notational conventions in glossing morphosyntax and because they have the most currency among linguists. Each example is left-aligned, vertically, word for word, with the gloss beneath it.

³ C. Lehmann, "Directions for interlinear morphemic translations." *Folia Linguistica* 16 (1983): 193–224; more recently W. Croft, *Typology and Universals* (Cambridge: Cambridge University Press, 2003), pp. xix–xxv.

Preface

(0.1) anā agar gən-it āt guš əbud-ø I if sleep.PFV-1SG you ear do.IMP.F-SG When I fall asleep, keep a lookout. (IV.46)

Segmentable morphemes are separated by a hyphen. On occasion, an expected morpheme may not be overt (e.g. the number of *abud* above), in which case it is indicated by \emptyset . Obviously, one morpheme in the object language may correspond to several in the gloss language, in which case they are separated in the gloss language by periods, with the exception of person and number. Clitic boundaries are indicated by an equal sign, in both the example and the gloss.

Example:

(0.2)	šerš=an	m=awwál	ālm	həw-at
	religion=1PL	from=first	world	be.PFV-3SG.F
	Our religion has existed from the first epoch of time. (I.8)			

In the examples above, each distinct morpheme is indicated by an abbreviated grammatical category label. In addition to the standard ones most commonly used among linguists, I have also adopted a few extra category labels to represent those features of Neo-Mandaic not generally found in other languages (a full list of category labels follows). Wherever possible, I have attempted to keep the gloss as simple as possible, by restricting the category labels to reflect only the the language itself. Furthermore, distinctions present in the morphosyntactic glosses should not be mistaken for a direct translation, which is given in the line that follows them. It is often the case that more than one English word is required to express a single Neo-Mandaic word in different contexts; this is particularly true with regard to the prepositions. In such instances, I have assigned the most common English gloss to the Neo-Mandaic word without regard to context.

Abbreviations and Symbols

adj.	adjective
adv.	adverb
C	consonant
CM	Classical Mandaic
coll.	collective noun
conj.	conjunction
dem. pron.	demonstrative pronoun
f.	feminine
indef. pron.	indefinite pronoun
indep. pron.	independent pronoun
interject.	interjection
imperf.	imperfective aspect
impv.	imperative
m.	masculine
n.	noun
NM	Neo-Mandaic
PCM	Post-Classical Mandaic
S.O.	someone (animate argument of verb)
s.t.	something (inanimate argument of verb)
subj.	subject
subjv.	subjunctive mood
pass. ptc.	passive participle
pl.	plural
perf.	perfective aspect
prep.	preposition
rel. pron.	relative pronoun
V	vowel
v.i.	intransitive verb
v.t.	transitive verb

Grammatical Category Labels

1	First Person
2	Second Person
3	Third Person
AUG	Nominal Augment
С	C-Stem, Causative
COMP	Comparative Suffix
COP	Copula
D	D-Stem, Transitivizing/Denominal
EZ	Persian <i>ezāfe</i>
F	Feminine
G	G-Stem, Basic Stem
IMP	Imperative
IPFV	Imperfective Aspect
IMPF	Imperfect (a.k.a. Prefix Conjugation)
IND	Indicative Morpheme
INDF	Indefinite / Non-Specific Morpheme
Μ	Masculine
NEG	Negative Morpheme
OBJ	Enclitic Object Marker
PASS	Passive Participle
PFV	Perfective Aspect (a.k.a. Suffix Conjugation)
PL	Plural
PTC	Active Participle
Q	Q-Stem, Quadriradical Verbs
REF	Referential Object Marker
REL	Relativizer
RES	Restrictive Clause Antecedent Marker
SBJV	Subjunctive Mood
SG	Singular
tC	Middle of <i>C</i> -Stem (originally with prefixed <i>t</i> -)
tD	Middle of <i>D</i> -Stem (originally with prefixed <i>t</i> -)
tG	Middle of <i>G</i> -Stem (originally with prefixed <i>t</i> -)

1. Introduction

1.1 The Name of the Language

The term that the Mandaeans most often employ to describe themselves is *Mandā'í* or *Mandāyí* 'Mandaean.'⁴ To their non-Mandaean neighbors in Iraq and Iran, they are more commonly known as the *şābi'ūn*, i.e. 'the Sabians,' or colloquially as the *şubba*. The singular of this term, *şubbi*, is not an uncommon surname for Mandaeans from Iran, including the family of my chief informant, Nasser Sobbi of Flushing, NY. Unlike most other terms by which ethnic groups are known to outsiders, the term 'Sabian' is in no way pejorative, but instead refers to one of the four 'peoples of the book' (*ahl al-kitāb*) with whom the Mandaeans have traditionally been identified by their Muslim neighbors. According to one hypothesis, this term derives from the Aramaic root related to baptism, in reference to the most characteristic Mandaean sacrament; the cognate in Neo-Mandaic is *şəbā ~ ṣəbi* (*ṣābi*) 'to baptize.'

The language which is the subject of this grammar is also known as *mandāyí*. The Mandaeans further distinguish between a written form, *Genzā* 'treasure' after the title of their chief scripture, the *Genzā Rabbā* 'Great Treasure,' and a spoken form, *rațnā*, a word which comes from the root meaning 'to whisper or mutter.' This is also the term used for the language of profane or magical texts, which are generally written in a later stage of the language than that found in the sacred texts. Interestingly, this same root has a cognate in Arabic, *rațana* ~ *yarțunu* (*rațāna*), which means 'to speak gibberish.'

⁴ In classical Mandaic, **mandaiia**, the plural of which is *Mandayānā* 'Mandaeans' in Neo-Mandaic.

An account of numerous origins proposed for the term $mand\bar{a}yi$ would compose a study of no small scale in itself. Drower cites in passing no fewer than three possible etymologies for this word:⁵

• manda 'knowledge' (cf. Biblical Aramaic מָּהָדָע in Dan. 2:21, 4:31, 33, and 5:12), which would make the Mandaeans the only known group from Late Antiquity to identify *themselves* as Gnostics (as opposed to being identified as such by other groups). This is the conventional etymology, and the one most often cited.

• **mandia** 'a cultic hut,' a word apparently derived from Pahlavi **m**'nd *m*and 'dwelling place.' According to this theory, the term **mandaiia** was coined on the analogy of **mašknaiia**, those of the **maškna**-sanctuary, which was another ancient name for members of the sect.

• **madai** 'Media,' elsewhere described in unflattering terms;⁶ in the *Harran Gāwetā*, the **nașuraiia** or Mandaean priesthood is described as having originated in the **tura <u>d</u>-madai** 'Mountain of Media.' Intriguingly, nowhere in this document is the term Mandaean ever employed.⁷

Presiding over this discussion is the Mandaean savior spirit, **manda** <u>d</u>**hiia** Mandā d-Heyyi, whose name can be translated as either 'Knowledge of Life' or 'Abode of Life,' given the aforementioned possible etymologies for **manda**. While I prefer the conventional etymology, I feel obliged to note that the identification of **manda** with 'Gnosis' was a product of western scholarship and was not current among the Mandaeans themselves until relatively recently. Edmondo Lupieri voiced another reservation against identifying **mandaiia** with the term 'Gnostic,' namely that this term generally designates the lay people, whereas the initiated priesthood—the true Gnostics—are known explicitly as the **naşuraiia**, i.e. those who possess **naşiruta** 'esoteric knowledge.'⁸

- 5 Ethel S. Drower, *The Mandaeans of Iraq and Iran*, 2nd edition reprint (Piscataway, NJ: Gorgias Press, 2002), 5–13.
- 6 Macuch noted that Sheikh Abdullah Khaffagi of Ahvāz, his primary source for the traditional pronunciation of the written language, pronounced this word as *manday* rather than the expected *madday*; Rudolf Macuch, *Handbook of Classical and Modern Mandaic* (Berlin: De Gruyter, 1965), li.
- 7 Ethel S. Drower, *The Haran Gawaita and the Baptism of Hibil-*Ziwa (Vatican City: Biblioteca Apostolica Vaticana, 1953), ix.
- 8 Edmondo Lupieri, *The Mandaeans: The Last Gnostics*. Translated by Charles Hindley (Grand Rapids, MI: Eerdmans, 2002), 9.

1.2 Ethnology

Given that it is entirely possible that the name of the Mandaeans may derive from a savior figure (like the Christians), their land of origin (like the Jews), their house of worship (like the Templars), or a defining characteristic of their religion (like the Muslims), the issue of the origins of their name will probably remain insoluble until further information becomes available.

1.2 Ethnology

While a complete ethnography of the Mandaeans would merit a monograph of its own, it is nonetheless important to acquaint oneself with the people who speak a language in order to fully understand that language. Even though such an ethnography of the Mandaeans has yet to be written, we are fortunate to have detailed descriptions of Mandaean life during the 1930s, courtesy of Drower, and of contemporary Mandaeans, courtesy of Jorunn J. Buckley.⁹ For this reason, a few remarks on the ethnology of the Mandaeans will suffice.

1.2.1 Religion

Just as Mandaic represents the only classical Aramaic dialect to survive to the present day, the Mandaean religion is the only surviving member of the pre-Islamic Gnostic sects from late antiquity. While the relationship of the Mandaeans to these other Gnostic sects has long been a matter of debate,¹⁰ and the validity of traditional scholarly conceptions about what constitutes Gnosticism has lately come into question,¹¹ Mandaeism shares numerous important features with the other sects that have been called Gnostic. According to Drower, these features include a supreme being which delegates the creation of the material and spiritual worlds to beings of its own creation; dualism,

- 9 See Drower, *The Mandaeans*, for the former, and Jorunn J. Buckley, *The Mandaeans: Ancient Texts and Modern People* (Oxford: Oxford University Press, 2002) for the latter.
- 10 See Edwin Yamauchi, *Gnostic Ethics and Mandaean Origins* (Cambridge: Harvard University Press, 1970) on the relation of Mandaeism to other Gnostic sects.
- 11 See Karen L. King, *What is Gnosticism?* (Cambridge: Belknap, 2003) for the complete history of the debate.

1. Introduction

which manifests itself in binary oppositions such as good and evil, light and darkness, the material and the spiritual, and so on; the concept of the soul as an exile and a prisoner; a belief in the influence of planets and stars, which are places of detention after death; a savior spirit which assists the soul in its journey to the afterlife; 'mysteries' or sacraments to aid and purify the soul, often based upon preexisting seasonal or traditional rites; an esoteric cult language of metaphors and personified ideas; and the great secrecy enjoined upon the initiates.¹²

In terms of their religious practices in daily life, the Mandaeans show some similarities with other Middle Eastern religions (such as Zoroastrianism, Judaism, Christianity, Islam), but do not agree with any of these traditions according to all of their practices. They are bound by strong dietary taboos, like Jews and Muslims but unlike Christians and Zoroastrians. In addition to foods that are also forbidden to the first two groups, such as camels, horses, pigs, dogs, cats, rats, rabbits, and hares, Mandaeans refrain from eating beef, birds of prey, seafood without scales, and even mushrooms. Strong drink is also prohibited. Much like Jews, Zoroastrians, and Muslims, but unlike Manichaeans, Mandaeans view celibacy as a sin, and consider marriage and the procreation of children to be religious obligations.¹³ Unlike Jews and Muslims, but like Christians and Zoroastrians, they do not practice circumcision; in fact, circumcision renders male Mandaeans ineligible for the priesthood.¹⁴

1.2.2 Society

In their literature, Mandaeans are often described as having led a communal lifestyle. A prophecy related in the *Genzā Yaminā* tells of "a conference to ensure that the great shall not receive more honor than the slave," and the Mandaean folktales frequently make mention

¹² Ethel S. Drower, *The Secret Adam: A Study of Nasorean Gnosis* (Oxford: Clarendon, 1960), xvi.

¹³ Drower, The Mandaeans, 41–48.

¹⁴ Ibid., 147.

1.2 Ethnology

of a Golden Age of equality for all, without wealth or poverty.¹⁵ Unlike Persian, which distinguishes between different registers of familiar and polite speech, Neo-Mandaic does not feature any formal forms of address; in fact, the Mandaeans recognize very few titles at all. In Drower's time, the secular leaders of the individual Mandaean communities, designated by the austere title Rab Mandayānā, 'Leader of the Mandaeans,' were elected by their peers, although this practice has fallen into obsolescence. According to Drower, good health is the only sign of nobility recognized by Mandaean society, and the major divisions in Mandaean society are between the secular and the sacred rather than between economic classes. The priesthood is theoretically open to all but in practice tends to be hereditary, due to the rather strict standards of purity enjoined upon priests, which must be maintained over multiple generations. The clergy is divided into three classes: the esganda, or acolyte; the tarmida, or priest, and the ganzibra, or high priest. Once the aspiring priest has learned to read and write Mandaic, he is known as a *valupā* 'disciple;'¹⁶ once he has been initiated as a *tarmidā*, he is called by the title *šiex* 'sheikh,' a loan word from Arabic. The office of the ganzibra, who presides over weddings and death masses, is the highest position which most Mandaean priests can hope to attain. According to Drower, a ganzibrā automatically becomes a *rišāmmā* or "head of the people" after consecrating five priests, but the consecration of priests is a rare event today.¹⁷

1.2.3 Family Life

Closely-knit families are a hallmark of Mandaean communities, and the extended family is the basis for a wide-ranging social network. In

¹⁵ Ibid., 385: "There was such justice that there was no wrongdoing. There was no strife and no war. None were rich and none were poor, for all were equal."

¹⁶ In the modern dialect, this term has come to be applied to all literate Mandaeans, both priestly and lay.

¹⁷ Drower, *The Mandaeans*, 173. Sheikh Abdullah (currently resident in London), who has been responsible for the consecration of twenty priests, is considered to be a *rišāmmā* by his Iraqi compatriots, but most Iranian Mandaeans reject this identification. Jorunn Buckley, personal communication, March 19, 2006.

Drower's day, marriages were often arranged, albeit with lineage as the primary concern rather than wealth, and a first cousin was considered to be an ideal match (preferably between paternal cousins rather than maternal cousins). Polygamy was permitted, albeit uncommon, and most Mandaeans maintained (and still maintain) that monogamy is superior. Divorce was not permitted, although a husband could separate from his wife if he found her morally lacking or careless about following the standards for ritual purity. Women were not allowed to remarry so long as their husbands were still alive and rarely remarried after their husbands had died.¹⁸ Today, many of these traditions are no longer followed, and Mandaeans generally observe the laws and the customs of the people among whom they live.

Most of the Mandaean kinship terms are unsurprisingly derived directly from the classical language; only the colloquial names for the grandparents, aunts, and uncles been borrowed from the neighboring languages, such as Arabic. Several of the kinship terms have irregular plurals, which are indicated below.

paternal grandparents maternal grandparents parents paternal uncles / aunts maternal uncles / aunts siblings cousins spouse parents in-law siblings in-law children	Male jeddu babbā bā <u>b</u> ā 'ammu xālu ahhā ebbər 'ammu/e / xālu/e bellā hām ah bellā / e <u>tt</u> ā ebrā	Female jedde bibi; māmā emmā 'amme xāle hāṯā (pl. ahwāṯā) bərat 'ammu/e / xālu/e eṯtā (pl. enšā) hamāṯā hāṯ bellā / eṯtā bərattā (pl. bənātā)
•	1100111	

Table 1: Kinship Terms

18 Ibid., 59.
1.2 Ethnology

The kinship terms for elder members of the family are bipolar; when addressing their juniors, elders will often refer to them using their own kinship term, and be addressed in kind. Consequently, Nasser Sobbi might refer to his grandson as *jedduye* 'my grandfather' and likewise be addressed as *jedduye*.¹⁹

1.2.4 Population Distribution

Regarding the Mandaean community in her day, Drower wrote,

To-day, the principal centers of the Subba are in Southern Iraq, in the marsh districts and on the lower reaches of the Euphrates and Tigris; in the towns of Amarah, Nasoriyah, Basrah, at the junction of the two rivers at Qurnah, at Qal'at Salih, Halfayah, and Suq-ash-Shuyukh. Groups of them are to be found in the more northerly towns of Iraq: Kut, Baghdad, Diwaniyah, Kirkuk, and Mosul all have Subbi communities of varying size. The skill of the Subbi as craftsmen takes them far afield, and Subbi silvershops exist in Beyrut, Damascus, and Alexandria. In Persia the Mandaeans were once numerous in the province of Khuzistan, but their numbers have diminished, and the settlements in Muhammerah and Ahwaz along the banks of the Karun river are not so prosperous or so healthy as those in Iraq.²⁰

After the First Gulf War, Saddam Hussein embarked upon an aggressive program to drain the marshes of southern Iraq, in retaliation against the *Ma'dān* or "Marsh Arabs," who had risen up in rebellion against his regime at the conclusion of that war. As a result, the inhabitants of these marshes—Arab and Mandaean alike—were displaced, bringing an end to their traditional way of life. Today, most of the Mandaeans who live in Iraq and Iran reside in urban centers, although many continue to work as jewelers, as they did in Drower's time. Outside of Iraq and Iran, the largest communities of Mandaeans are found in Sweden, Australia, and the United States, although many Mandaeans fled to Jordan and Syria following the occupation of Iraq

¹⁹ Bipolarity in kinship terms is a typical feature of most colloquial Arabic dialects; for more information on this phenomenon, see Mahmoud Aziz F. Yassin, "Bi-polar Terms of Address in Kuwaiti Arabic," *Bulletin of the School of Oriental and African Studies* 40.2 (1977): 297–301.

²⁰ Ibid., 1–2.

in 2003, and small groups of Mandaeans can be found in most large cities throughout the world, from Madrid to Tokyo.

The dialect described here is that of Khorramshahr (formerly Muhammara), an Iranian river port city located at the confluence of the Kārūn River and the Shatt el-Arab in the province of Khuzestān. This province, which is bounded on the south by the Persian Gulf and on the west by Iraq, stands geographically, historically, and culturally between Mesopotamia and Iran. As late as the 19th century, Neo-Mandaic dialects were spoken in the streets of several cities in northern Khuzestān, including Shushtar, Dezful, and Shāh Wāli. At some point, the Mandaeans of these communities were forced to leave, as a result of persecution or disease, and took up residence in Khorramshahr and Ahvāz.²¹

1.3 Demography

No consensus exists regarding the total number of Mandaeans in Iraq or Iran, and the figures for the number of Neo-Mandaic speakers remaining are almost entirely speculative. Conservative estimates for the former range from 40,000 to 60,000²² and perhaps 100-200 speakers for the latter, almost all of whom are located in Iran and nearly all of whom are fifty years of age or older. There are no speakers of Neo-Mandaic under thirty years of age.²³

A rough census of all the Mandaean communities in what is today Iraq and Iran was last made in 1877 by the orientalist A. Houtum-

- 21 According to tradition, the local governor of Shūshtar instigated a massacre against Mandaeans in the region, drowning them in wells. The ruler of Persia at the time, Nāşeroddin Shāh (1848–1896), intervened on behalf of the Mandaeans and stopped the massacres; Buckley, *The Mandaeans*, 6. One of his chief architects, Abdolġaffār Najmolmolk, subsequently visited the region and reported that two hundred households of Mandaeans still resided there, but that there were many others in the Ottoman Empire who had come from Khuzestān. See Abdolġaffār Najmolmolk, *Safarnāme-ye Khuzestān* (Tehran: Elmi, 1962).
- 22 Julie A.J. Ebadirad, "Information on Mandaeans in Iran in Regards to Human Rights," *ASUTA* (Jan. 2003), 3.
- 23 Jorunn J. Buckley, personal communication. Buckley last visited Iran in 1996.

Schindler.²⁴ Houtum-Schindler arrived at the following numbers for the major Mandaean communities:

Table 2: Census of Mandaean Families, 1877

ʿAmāra	(Iraq)	130 families
Sūq ash-Shuyūkh	(Iraq)	120 families
Basra	(Iraq)	50 families
Muḥammara	(Iran)	30 families
Huwayza	(Iran)	10 families
Dezful	(Iran)	seven families
Shushtar	(Iran)	two families

Houtum-Schindler also estimated that about 200 families lived in various Arab camps along the Tigris and the Shatt el-Arab, for a total of roughly 550 families altogether. When he returned in 1886, he arrived at the same numbers. Curzon quoted these same figures in his *Persia and the Persian Question*, but noted that the Mandaeans were in numerical decline.²⁵

According to Nasser Sobbi, nearly all of the Mandaeans in Muhammara (today Khorramshahr) spoke Neo-Mandaic during his youth (the 1920s and 1930s), with the exception of two families who had originally come from Iraq. Indeed, the situation of Neo-Mandaic in Iraq a century ago must have been equivalent to the situation of Neo-Mandaic in Iran today; older members of the community, predominantly of Persian origin,²⁶ continued to speak Neo-Mandaic in their homes and even compose letters in it, as the texts collected by Jacques de Morgan attest, but the majority of the Iraqi Mandaeans, including the priests, did not speak the language natively.

Since the time of Curzon and Houtum-Schindler, widely divergent figures have been given for the population of Mandaeans in Iran and the number of Mandaic speakers among them. In his article on the

- 24 A. Houtum-Schindler, "Notes on the Sabians," *Proceedings of the Royal Geographic Society* (Nov. 1891): 665.
- 25 Curzon of Kedleston, George Nathaniel Gordon, Marquess, *Persia and the Persian Question*, vol. II (London: Longman and Co., 1892), 306.
- 26 Such as Drower's informant Hirmiz bar Anhar, whose family ultimately originated in Shushtar, according to Sobbi.

status of Mandaean studies, published in 1966, Edwin Yamauchi noted that there were likely only a few thousand Mandaeans left.²⁷ One of the most frequently quoted figures for the total Mandaean population (in Iraq and Iran) is that of Kurt Rudolph, who estimated a number of 15,000 Mandaeans total, without citing any source for this number.²⁸ Yamauchi, among many others, adopted this figure, and added on his own authority that there were perhaps only 200 speakers of Neo-Mandaic remaining.²⁹ In 1989, the total population of Mandaeans in Iran was reported to be 6,200,³⁰ a figure which subsequently came to be adopted by the United Nations High Commission in its reports.³¹

More recent figures come from Hezy Mutzafi of the Dept. of Hebrew and Semitic Languages at Tel Aviv University. In the article on Mandaic in the fourteenth edition of *Ethnologue*, Mutzafi is cited as providing an estimate of 23,000 Mandaeans in Iraq, of whom none speak Neo-Mandaic natively. In Iran, out of a total population of 5,000 Mandaeans, only a small number (ca. 800-1000) speak Neo-Mandaic by his reckoning. In the latest edition of *Ethnologue*, the number of native speakers has dropped to 500, but Mutzafi's other figures have been retained.³²

Ṣābe'in-e Irān Zamin, an album of photographic images of Mandaean life in Iran, placed the number of Iranian Mandaeans at

- 27 Edwin Yamauchi, "The Present Status of Mandaean Studies," *Journal of Near Eastern Studies* 25.2 (1966): 88.
- 28 Kurt Rudolph, *Mandaeism* (Leiden: E.J. Brill, 1978), 1.
- 29 Edwin Yamauchi, review of Zur Sprache und Literatur der Mandäer by Rudolf Macuch, Journal of the American Oriental Society 100.1 (1980): 80.
- 30 Michel Clevenot (ed.), *L'état des religions dans la monde* (Paris: Editions La Découverte, 1987), 242.
- 31 See, for example, the United Nations High Commissioner for Refugees *Report: Iran: Information on the Mandean Religion and Whether or not the Children of Mandeans are taken from their Homes and Forcibly Converted to Islam by Iranian Authorities*, August 26, 1997.
- 32 Raymond G. Gordon, Jr. (ed.), *Ethnologue: Languages of the World*, fifteenth edition (Dallas: SIL International, 2005). Online version: http://www.ethnologue.com/

1.4 Genetic Affiliation

25,000 and that of the Iraqi Mandaeans at 70,000.³³ In the foreword that she composed for the 2002 reprint of Ethel S. Drower's *The Mandaeans of Iraq and Iran*, Buckley cited figures of 100,000 to 140,000 Mandaeans worldwide, with the majority residing in Iraq.³⁴

1.4 Genetic Affiliation

Neo-Mandaic represents the latest stage of the phonological and morphological development of Classical Mandaic, a Northwest Semitic language of the Eastern Aramaic sub-family. Along with the other surviving dialects of Aramaic, it is classified as Neo-Aramaic; these form a constellation of dialects ranging from Lake Van and Lake Urmia in the north to Damascus and Ahvāz in the south, clustered in small groups. Having developed in isolation from one another, most Neo-Aramaic dialects are mutually unintelligible and should therefore be considered separate languages; however, determining the exact relationship between the various Neo-Aramaic dialects is a difficult task, fraught with many problems, which arise from our incomplete knowledge of these dialects and their relation to the Aramaic dialects of antiquity.³⁵

Aramaic became widespread throughout the Fertile Crescent largely as a result of the policies of the Neo-Assyrian (ca. 934-609 B.C.E.) and Neo-Babylonian (ca. 627-520) empires. The Achaemenids (576-330 B.C.E.) adopted it as an auxiliary language both for international communication and internal administrative use. It gradually came to supplant the native languages of the region, but due to its wide geographic distribution and political circumstances following the collapse of the empire, it soon evolved into two major sub-families—the Western sub-family, comprising Jewish Palestinian, Christian Palestinian, and Samaritan, and the Eastern sub-family,

³³ Abbas Tahvildar, *Ṣābe'in-e Irān Zamin* (English title: *Baptists of Iran*), photos by Abbas Tahvildar, text by Massoud Fourouzandeh and Alain Brunet (Tehran: Key Press, 2001).

³⁴ Jorunn J. Buckley, "Foreword," in Drower, The Mandaeans, i*.

³⁵ Wolfhart Heinrichs, "Introduction," in *Studies in Neo Aramaic*, ed. by Wolfhart Heinrichs, x–xv (Atlanta, GA: Scholars, 1990).

comprising Late Babylonian,³⁶ Syriac, and Mandaic. Some scholars (e.g. Stephen A. Kaufman and Daniel Boyarin) place Syriac in a third group, separate from Western and Eastern Aramaic.

Mandaic was first identified as an Aramaic dialect in 1604, the year in which Gerolamo Vecchietti, an Italian traveler, encountered a small village of Mandaeans *en route* to Basra from Baghdad:

I wanted to write a reminder of some of the words and, since the man was courteous, I sent for writing materials from the boat. Once I had written down many words I realized that the language was Chaldean [i.e. Aramaic].³⁷

This identification was subsequently corroborated by the Carmelite mission in Basra; Basil of St. Francis, a Portuguese Carmelite who founded the mission, reported to the Sacra Congregatio de Propaganda Fide that, while their language was similar to Syriac, Chaldean priests and Mandaeans "understood each other very little" when attempting to communicate in their respective idioms.³⁸

In his grammar of Mandaic, Theodor Nöldeke recognized the close affinity that Mandaic had with Syriac, but identified it more closely with the language of the Aramaic portions of the Babylonian Talmud:

Mandaic is closely related to the ordinary dialect of the Babylonian Talmud. Both the dialects are neighbors, geographically speaking [...] actually, we may assume that the language of the Babylonian Talmud was that used in Upper, and Mandaic that used in Lower Babylonia.³⁹

He attributed their close relationship to a common descent from the language of the *Nabaț al-Irāq*, i.e. the language of the pre-Islamic Aramaean inhabitants of Iraq:

A close relationship between Mandaic and the Talmudic language is apparent throughout the grammar: Mandaic, however, appears to be a later form than Talmudic, but not throughout, for the Mandaic texts are

37 Lupieri, The Mandaeans, 73.

³⁶ I.e. the dialects attested in the Aramaic portions of the Babylonian Talmud and those of the incantation bowls, according to Wolfhart Heinrichs.

³⁸ Ibid., 88.

³⁹ Theodor Nöldeke, *Mandäische Grammatik* (Halle: Waisenhaus, 1875), xxv. Translated by the present author.

linguistically purer and not as mixed with foreign elements, and represent the Aramaic speech of Babylon better than the Talmud. Had the Arabs preserved for us something more than a few accidentally introduced words of the dialect of the Iraqi Nabataeans, (i.e. the Aramaic-speaking inhabitants of Babylonia), we should again find the main features of Mandaic and Talmudic, and far more clearly than is now possible.⁴⁰

Most of the Neo-Aramaic dialects which survived to the present day belong to the Eastern sub-family; these include Central Neo-Aramaic (Țūrōyo and Mlaḥsô), Northeastern Neo-Aramaic (the largest Neo-Aramaic group, which includes various Jewish Neo-Aramaic dialects, and the dialects of the Assyrian and Chaldean Christians), and Neo-Mandaic. The only surviving remnant of the Western sub-family is Western Neo-Aramaic, spoken in the villages of Ma'lūla, Bakh'a, and Jubb 'Adīn to the northeast of Damascus. Of all of these dialects, Eastern or Western, only Neo-Mandaic can be described with any certainty as the direct descendent of one of the Aramaic dialects attested in Late Antiquity.⁴¹ For this reason, it is potentially of great value in reconstructing the history of this sub-family and the precise genetic relationship of its members to one another.

1.5 Previous Research

Neo-Mandaic was quite possibly the first of the Neo-Aramaic dialects to attract scholarly attention, and more so than any other Neo-Aramaic dialect it has captured the interest of some of the greatest Semitists of each age. Unfortunately, scholarship on Neo-Mandaic was fraught with numerous frustrations, false starts, and near-misses. While it would fall outside the scope of this grammar to describe the entire three hundred and fifty years of scholarship on Mandaic in detail,⁴² four periods of scholarly activity stand out. During the first period, the study of Mandaic was essentially the study of the spoken language, as the classical dialect remained virtually unknown. After a hiatus of over a century, the broad outlines of the classical language became

⁴⁰ Ibid., xxvi.

⁴¹ Macuch, Handbook, lv.

⁴² For further details on the history of western scholarship on the Mandaeans, consult Lupieri, *The Mandaeans*, 61–126.

1. Introduction

known, beginning with the publication of an edition of a classical text (the *Genzā Rabbā*) and culminating in a grammar of the written language. Within a fairly short time, numerous editions of classical texts began to appear, as well as incantation texts and manuscripts in a postclassical form of the language and even five written documents in Neo-Mandaic. Finally, the twentieth century saw a marked increase in the number of Mandaic texts available to scholarship and the first attempts at a descriptive grammar of a Neo-Mandaic dialect.

1.5.1 The Early Period

European scholars first became aware of the existence of the Mandaeans, or "St. John Christians," as they were initially known, in 1555.⁴³ The first encounters between the two occurred in the kingdom of Hormuz, which was then under Portuguese rule. Members of the Mandaean community in Basra traveled to Hormuz in the hopes of convincing the Portuguese to liberate them from their Muslim suzerains. Initially, the Portuguese assumed that they were a Christian community within the Ottoman Empire, occupying territory coveted by the Portuguese, persecuted by their Muslim neighbors, and willing to accept papal authority. In the following decades, several abortive attempts were made to establish a base in Basra; it was not until 1623, however, after the Persians had captured Hormuz with the help of the British and came to menace Basra as well, that the Augustinians and the Discalced Carmelites were permitted to establish missions there. At least three of the Carmelites attached to the mission in Basra devoted their attention to the Mandaic language, as it was spoken in that day.⁴⁴

The first of the Carmelite scholars of Mandaic, an Italian by the name of Ignazio di Gesù, who lived in Basra between 1641 and 1652, is credited with a work entitled *Scrinium quattuor linguarum orientalium, scilicet Persicae, Arabicae, Turcicae, et Mendaicae, in*

- 43 It should be noted that an itinerant monk by the name of Ricoldo di Montecroce encountered them as early as the 13th century, but his account was not publicized until the 1940s; Lupieri, *The Mandaeans*, 69.
- 44 See Roberta Borghero, "A 17th Century Glossary of Mandaic," *ARAM* 11–12 (1999–2000): 311–319.

14

1.5 Previous Research

quo continentur ea, quae pro incipientibus necessaria sunt ad recte, et cito loquendum his quattuor idiomatibus, which was apparently finished some time before 1656. This work has been lost, although a copy of his *Scrinium duarum*, limited to Persian and Arabic, is preserved at the National Library of Rome.⁴⁵ Ignazio de Gesù was also the author of the *Narratio originis, rituum et errorum Christianorum S. Joannis*, which remained the standard reference on the religion of the Mandaeans for over two centuries.⁴⁶

Matteo di San Giuseppe, another Italian Carmelite who took part in missions to Syria, Persia, Mesopotamia, and India, announced his intention to compose a "dictionary of the Mandaean language [...], which is almost unknown, and of Arabic, Turkish, Persian, and Latin," in 1651. This dictionary was presumably finished by 1667, the year in which he next mentions it in his correspondence. Roberta Borghero has suggested convincingly that this document is to be identified with the *Glossarium Sabico-Arabicum-Latinum-Turcum-Persicum* at the University Library in Leiden.⁴⁷ A French Carmelite, Ange de Saint Joseph, also made mention of Matteo di San Giuseppe's dictionary, which he described as an "*alphabetum, nec non onomasticon huiusce linguae*," in the introduction to his 1681 *Pharmacopoea Persica*. In the same work, he also disclosed his intention to write a "*clavis linguae Sabaitarum*," which has unfortunately not survived, if it was ever completed.⁴⁸

These same missionaries also assiduously collected Mandaic manuscripts, which eventually found their way to libraries in Rome

- 45 Borghero, *Glossary*, 316–17. Ignazio di Gesù's *Scrinia* were intended to be handbooks of oriental languages, containing the basic information necessary for beginners who want to speak these languages correctly.
- 46 Ignazio di Gesù, Narratio originis, rituum et errorum Christianorum S. Joannis, cui adiungitur discursus per modum dialogi in quo confutantur XXXIIII errores eiusdem nationis (Rome: Sacra Congregatio de Propaganda Fide, 1652).
- 47 Borghero, *Glossary*, 318. This *Glossarium* was to have a perennial influence upon future Mandaeologists; it was consulted by Theodor Nöldeke and Rudolf Macuch in the preparation of their grammars, and the contents of its Mandaic column were incorporated into the dictionary that Ethel S. Drower and Rudolf Macuch co-authored.
- 48 Borghero, Glossary, 313–14.

1. Introduction

and other cities of Europe.⁴⁹ The manuscripts collected by the missionaries were the basis for most of the editions published in the 19th century; western scholars would not acquire new manuscripts, for the most part, until the 20th century.

1.5.2 The Nineteenth Century

After a long hiatus, a new impetus was given to the study of Mandaic by the Swedish orientalist Matthias Norberg, who transcribed a manuscript of the *Genzā Rabbā* into Syriac characters and produced a translation of the text into Latin.⁵⁰ This was followed by a lexicon in 1816^{51} and an onomasticon in $1817.^{52}$ Unfortunately, neither his transcription nor his translation withstood academic scrutiny. Scholarship would have to wait for half a century before an acceptable edition of the *Genzā Rabbā* became available and well over a century before a full translation appeared.⁵³

This period also coincides with renewed efforts at field work among the Mandaeans. The first European to do so was the Assyriologist J.E. Taylor, who served for twelve years as the British vice-consul at Basra. In 1825, during the time that he served at Basra, he made the acquaintance of the priest Adam Yuhana. Over the next few years, Adam Yuhana instructed Taylor in Mandaic language and literature, and Taylor taught French, English, and Armenian (or perhaps German) to Adam's son, Yahia Bihram. According to Yahia

- 51 Matthias Norberg, *Lexidion Codicis Nasaraei: cui Liber Adami nomen / edidit Matth. Norberg* (Lund: Berlingianus, 1816).
- 52 Matthias Norberg, Onomasticon Codicis Nasaraei: cui Liber adami nomen / edidit Matth. Norberg (Lund: Berlingianus, 1817).
- 53 Lupieri, The Mandaeans, 114–15.

16

⁴⁹ The oldest Mandaic manuscripts in Rome were acquired by Ignazio di Gesù, and the oldest ones in Britain were acquired by Ange de Saint Joseph for Robert Huntingdon; Lupieri, *The Mandaeans*, 106, fn. 98.

⁵⁰ Matthias Norberg, Codex Nasaraeus: Liber Adami appellatus, Syriace transscriptus, loco vocalium, ubi vicem literarum gutturalium praestiterint, his substitutis Latineque redditus a Matth. Norberg (Lund: Berlingianus, 1815– 16).

Bihram, Taylor was so moved by his education in the faith that he resolved to adopt it in secret.⁵⁴

In 1854, the year in which Taylor conducted the British Museum excavations at Ur (Tall al-Muqayyar), he visited the German orientalist J. Heinrich Petermann at nearby Sūq ash-Shuyūkh. Petermann was traveling throughout the Middle East in the hopes of studying the Samaritans, the Druze, and the Mandaeans, and had settled in Sūq ash-Shuyūkh in the hopes of learning something about the Mandaean religion. During the three months that he lived in Sūq ash-Shuyūkh, he studied Mandaic language and literature with the very same Yahia Bihram whom Taylor had tutored; Taylor was, in fact, the only European whom he saw during his stay in Sūq ash-Shuyūkh.

Upon returning to Europe, Petermann published the account of his travels in the Middle East, including his stay in Sūq ash-Shuyūkh, in his *Reisen im Orient*.⁵⁵ In addition to providing western scholars with the first concrete accounts of the Mandaean community in nearly two hundred years, Petermann collected a number of Mandaean folktales, including the legend of the "Bridge of Shushtar," which is the most famous of Mandaean legends.⁵⁶ Several years later, Petermann published a critical edition of the *Genzā Rabbā* on the basis of four 16th and 17th century manuscripts.⁵⁷ This edition, the first Mandaic manuscript to be published in facsimile, included a brief introduction in Latin, but appeared without any commentary or translation. Despite the fact that a translation of this text would not appear until 1925,⁵⁸ Petermann's edition quickly became the basis for all studies of

- 54 Jorunn J. Buckley, "Glimpses of a Life: Yahia Bihram, Mandaean Priest," *History of Religions* 39.1 (1999): 38.
- 55 J. Heinrich Petermann, *Reisen im Orient*, vol. I–II (Leipzig: Von Veit, 1860 and 1861). The Mandaeans are discussed in volume II.
- 56 This legend is the subject of Text IV in the text collection.
- 57 J. Heinrich Petermann, *Thesaurus sive Liber Magnus vulgo Liber Adami* appellatus, opus Mandaeorum summi ponderis, vol. I–II (Leipzig: J.O. Weigel, 1867), now available in a reprint edition through Gorgias Press (Piscataway, NJ: 2007).
- 58 Mark Lidzbarski, *Ginza: Der Schatz; oder, Das grosse Buch der Mandäer* (Göttingen: Vandenhoeck and Ruprecht, 1925).

Mandaic language and literature and remained so until Jacques de Morgan and Ethel S. Drower brought new manuscripts to scholarly attention.

Petermann's efforts spurred a new growth of interest in the Mandaeans in Europe. Shortly after the publication of *Reisen im Orient*, the German philologist Theodor Nöldeke published an article on "the dialect of the Mandaeans,"⁵⁹ which relied primarily on Norberg's edition and the few other Mandaic texts which had been published since it. This article became the germ of the grammar he was to produce thirteen years later.⁶⁰ Over the next ten years, the Swiss orientalist Albert Socin traveled to the Middle East several times in the hopes of duplicating Petermann's successes, but was unable to enlist the aid of Yahia Bihram or, indeed, any other Mandaean. In the same year in which Petermann published his edition of the *Genzā Rabbā*, Julius Euting published a transcription of one third of the *Qolastā*, the Mandaean prayer book, based upon manuscripts in Paris and London.⁶¹

In 1873, the French vice-consul in Mosul, a Syrian Christian by the name of Nicholas Siouffi, sought Mandaean informants in Baghdad without success. Two years later, he was introduced to a young Mandaean priest from Shushtar named Adam, who had recently converted to Catholicism. This Adam was likely the grandson of Yahia Bihram, who had taught Mandaic to Petermann, and the greatgrandson of the Adam who had taught Mandaic to Taylor.⁶²

- 59 Theodor Nöldeke, "Ueber die Mundart der Mandäer," Abhandlungen der Historisch-Philologischen Classe der königlichen Gesellschaft der Wissenschaften zu Göttingen 10 (1862): 81–160.
- 60 I.e. his Mandäische Grammatik.
- 61 Julius Euting, Qolasta oder Gesänge und Lehren von der Taufe und dem Ausgang der Seele als mandäischer Text mit sämtlichen Varianten, nach Pariser und Londoner Manuscripten (Stuttgart: Karl J. Trübner ,1867). This edition was the basis for Mark Lidzbarski's Mandäische Liturgien, mitgeteilt, übersetzt und erklärt (Berlin: Weidmann, 1920). Ethel S. Drower translated the full text in her Canonical Prayerbook of the Mandaeans (Leiden: E.J. Brill, 1959).
- 62 This is perhaps not so great a coincidence as it might otherwise seem. The Mandaeans were, at this time, extremely secretive about their religion and were not well-disposed towards revealing either their texts or aspects of their dogma

According to Siouffi, Mandaic was the only language that Adam was able to read and write fluently, although he attended services in the East Syrian Rite and claimed to be able to understand "almost everything."⁶³ Over the next few years, he taught Siouffi to read Mandaic, and Siouffi transcribed the names and the texts which Adam gave him into an *ad hoc* system based upon French orthography. The samples which Siouffi published in his book on the Mandaeans betray numerous features which are typical of Neo-Mandaic rather than Classical Mandaic, such as the transcriptions *mambouha* and *masbetta* for Classical Mandaic **mambuga** and **masbuta**, or the centralization of the vowel /u/ suggested by the transcription *chichtroio*.⁶⁴ Siouffi's work represents the first attempt, however unsystematic, to represent the current pronunciation of Mandaic.

Nöldeke's grammar, which was published the year in which Siouffi first met Adam, was produced too soon to take advantage of Siouffi's attempts at transcribing Mandaic; Nöldeke himself considered the absence of such data to be the primary weakness of his work:

I regard the fact that I know the language exclusively through its literature, without almost any idea of the traditional pronunciation, to be the greatest deficiency of my book. In particular, the section on phonology has suffered as a result. Hopefully, this will soon be supplemented by the research of Prof. Petermann, the sole scholar (as far as we know) who has precise

to outsiders. Although Taylor never published any information about his encounters with Adam Yuhana and Yahia Bihram, he must have made a positive impression upon the two, thus ensuring that they and future generations of their family would continue to aid westerners when other Mandaeans were unwilling. Furthermore, Yahia Bihram was not just any priest; he and his cousin Ram Zihrun were the only two members of the clergy to survive the calamitous 1831 cholera epidemic which decimated the Mandaean community; see Buckley, "Glimpses," 34.

⁶³ The East Syrian (i.e. Chaldean) Rite to which Siouffi subscribed is delivered in Syriac, thus explaining the facility with which Adam followed the Mass. Lupieri, *The Mandaeans*, 121.

⁶⁴ Nicholas Siouffi, Études sur la religion des Soubbas ou Sabéens: leurs dogmes, leurs moeurs (Paris: Impremerie Nationale, 1880). The examples are taken from pp. 4, 162, and 23, respectively. They correspond to Neo-Mandaic mambuhā [mam'bu:hɔ] 'ritual drink,' mașbettā [mas'b3tta] 'baptism,' and šuštrāyā [ʃʉʃ'trɔ:jɔ] 'resident of Shushtar,' respectively.

1. Introduction

knowledge of that pronunciation. It is even more urgently wished that the highly esteemed scholar would publish his research on the traditional interpretation of the Mandaean books; if these were to be lost, we would be robbed of a possibly irreplaceable aid for the understanding of the writings. My friend Socin has still not managed to learn anything from the Mandaeans, among whom there are very few cognoscenti, about their language or literature. With some reliable knowledge about the exceptical tradition, we will undoubtedly arrive at a rather precise understanding and a dependable dictionary; without such knowledge, however, the interpretation [of the texts] will remain extremely spotty and uncertain.⁶⁵

J. Heinrich Petermann died the following year without publishing his notes. Albert Socin never did find a Mandaean informant in Iraq; he and his colleague Eugen Prym eventually turned their attention to the Neo-Aramaic dialect of Tūr ⁶Abdīn, from which they published some folktales with translations in 1881.⁶⁶ In the following year he published an edition of Jewish and Christian literary texts in Neo-Aramaic he had collected in the region between Urūmīya and Mosul;⁶⁷ the fact that Socin was unable to collect similar texts from the Iraqi dialect (or dialects) of Neo-Mandaic, which would disappear virtually without a trace within the next few generations, is one of the greatest tragedies to befall the study of Semitic languages in the modern era.

1.5.3 Turn of the Century

Over the next fifty years, the most significant developments in the study of Mandaic were related to the publication of new data on all stages of the Mandaic language—Classical, Postclassical, and Neo-Mandaic—in the form of new manuscripts and archaeological finds such as incantation bowls and lead amulets. The pioneer translator of Mandaic literature was the Dutch scholar A.J.H. Wilhelm Brandt, who

⁶⁵ Nöldeke, *Mandäische Grammatik*, vii, translated by the present author. Nöldeke's grammar was published in 1875, five years before Siouffi published the data he had collected from Adam.

⁶⁶ Eugen Prym and Albert Socin, Der neuaramäische Dialekt des Tur Abdin (Göttingen:Vanderhoech, 1881).

⁶⁷ Albert Socin, *Die neuaramäischen Dialekte von Urmia bis Mosul*, vols. I-II (Tübingen: Laupp, 1882).

used Petermann's edition and the grammar published by Nöldeke to produce an edition of a few tractates from the *Genzā Rabbā* in 1893.⁶⁸

To this period also belong Jacques de Morgan's discovery and subsequent publication of facsimiles of portions from the *Qolasta*, as well as three legends and two letters in Neo-Mandaic, the first such texts to be published and the only samples of Neo-Mandaic writing ever published.⁶⁹ In the same year (1904), a photographic reproduction of a Mandaic divan, the Diwan Abatar, originally brought to Rome by Ignazio di Gesù, was published by Julius Euting and B. Poertner.⁷⁰ These publications dramatically increased the number of Mandaic texts available to western scholars, including the German scholar Mark Lidzbarski, who published a series of new editions of the Drāši d-Yahva, the Qolastā, and the Genzā Rabbā from the manuscripts that had become available in the previous fifty years.⁷¹ In order to do so, he gathered lexical material from all of the manuscripts in European libraries, which he collected on index cards. This material was subsequently used by Drower and Macuch in their A Mandaic Dictionary.⁷² Lidzbarski's editions of the Drāši d-Yahvā (his Johannesbuch) and the Genzā Rabbā remain the standard editions to the present day.

- 68 A.J.H. Wilhelm Brandt, *Mandäische Schriften. Aus der grossen Sammlung heiliger Bücher Genza, oder Sidra Rabba, übersetzt und erläutert* (Göttingen: Vanderhoeck and Ruprecht, 1893).
- 69 Jaques de Morgan, Mission scientifique en Perse, tome V (études linguistiques), deuxième partie: textes mandaites (Paris: Imprimerie Nationale, 1904). The portions from the Qolastā are inaccurately identified as the Iniani ('niania Enyāni "antiphonal responses") and the Sidra D'Nichmata (sidra d-nišmata Sidrā d-Nešmātā "Book of Souls"); see Jorunn J. Buckley, "A study of the two liturgical collections in J. de Morgan's Textes Mandaïtes," Le Muséon 104 (1991): 191–203.
- 70 Julius Euting and B. Poertner, *Mandaeischer Diwan nach photographischer Aufname* (Strassburg: Karl J. Trübner, 1904).
- 71 For the *Drāši d-Yahya*, see Mark Lidzbarski, *Das Johannesbuch der Mandäer* (Giessen: Töpelmann, 1915); for the *Qolastā*, see Lidzbarski, *Mandäische Liturgien*, and for the *Genzā Rabbā*, see Lidzbarski, *Ginza*.
- 72 Ethel S. Drower and Rudolf Macuch, *A Mandaic Dictionary* (Oxford: Clarendon, 1963), v.

1. Introduction

The first Mandaic incantation bowls were collected and published by Henri Pognon, the French consul at Aleppo.⁷³ While incantation bowls in other scripts and dialects of Aramaic had been published, Pognon's commentary on the proper names found within the bowls and the glossary that accompanied the publication became an indispensable tool for all future publications of these texts and drew scholarly attention to a formerly unknown phase of the language.⁷⁴ Pognon dated the texts to the period immediately following the Islamic conquest on the basis of an Arab name in one of the bowls, but taking into consideration the the lack of Arabic influences elsewhere within the corpus. In his review of Pognon's work, Nöldeke noted that these incantation texts, as well as many other profane texts, were written in a later stage of the language than the classical texts.⁷⁵ Chronologically more proximate to the language of the classical texts (but still belonging to the postclassical stage of the language) are the lead amulets, the first of which was published by Lidzbarski, who dated it to the 5th century.⁷⁶

- 73 Henri Pognon, "Une incantation contre les genies malfaisants en mandaite," *Mémoires de la Société linguistique de Paris* 8 (1894): 193–234 and his *Inscriptions mandaites des coupes de Khouabir* (Paris: Imprimerie nationale, 1898).
- 74 The first Mandaic incantation text was published by Pognon himself five years prior to his *Inscriptions mandaites*. Other Aramaic incantation texts, in the Estrangela and "Jewish" square scripts rather than the Mandaic script, had been published piecemeal since Thomas Ellis' publication of seven bowls in 1853. See J.B. Segal and E.C.D. Hunter, *Catalogue of the Aramaic and Mandaic Incantation Bowls in the British Museum* (London: British Museum Press, 2000), 21 for the most recent word on the history of the scholarship of the incantation bowls. I have reviewed this catalogue in a forthcoming issue of *Orientalistische Literaturzeitung*.
- 75 Theodor Nöldeke, review of *Inscriptions mandaites des coupes de Khouabir* by Henri Pognon, *Wiener Zeitschrift für die Kunde des Morgenlands* 12 (1899): 143.
- 76 Mark Lidzbarski, "Ein mandäisches Amulett," in Florilegium ou recueil de travaux d'érudition dédiés à M. le Marquis Melchior de Vogüé à l'occasion du quatre-vingtième anniversaire de sa naissance, ed. G.C.C. Maspero (Paris: Geuthner, 1909), 349–73. Lidzbarski attributed the amulet to the 5th century on the basis of the palaeography; for more information on the script of the lead amulets, see my "Iranian Scripts for Aramaic Languages: The Origin of the

Within fifteen years of Pognon's publication, another corpus of incantation bowls discovered in an archaeological context, namely those from Nippur excavated by the University Museum in James A. Montgomerv.⁷⁷ by Philadelphia. was published Montgomery's publication was the most exhaustive vet attempted and established the standard for the publication of these texts even to this day. One of the many important contributions that Montgomery made to the debate was the assignation of a seventh-century C.E. date for the bowls on the basis of their archaeological provenance.⁷⁸ This confirmed both Pognon's suspicions about the age of the bowls and Nöldeke's suspicions about the relation of their language to that of the classical texts. Montgomery's dating has been upheld by subsequent archaeological excavations.⁷⁹

1.5.4 The Twentieth Century

The most recent phase of scholarship on the Mandaic language begins in 1930, with the death of Theodor Nöldeke and the beginning of Ethel S. Drower's research on the Mandaeans.⁸⁰ In 1932, when Drower first visited Ahvāz and Khorramshahr (then Muḥammara) and met a young Nasser Sobbi, she mentioned to him that the Mandaeans in Iraq and even their priests were unable to understand Mandaic, but in Iran even the children were able to speak it.⁸¹ We may assume that she was engaging in a bit of hyperbole; even though she later wrote

Mandaic Script," in the *Bulletin of the American Schools of Oriental Research* 341 (2006): 53–62.

⁷⁷ James A. Montgomery, *Aramaic Incantation Texts from Nippur* (Philadelphia: The University Museum Press, 1913).

⁷⁸ Montgomery, *Aramaic Incantation Texts*, 104: "The archaeological evidence then for the *terminus ad quem* of our texts is the seventh century (probably its beginning), with a fair leeway back into the preceding century."

⁷⁹ One bowl, excavated at Ana, has been securely dated within the 8th century; see Erica C.D. Hunter, "Two Mandaic Incantation Bowls from Nippur," *Baghdader Mitteilungen* 25 (1994): 607.

⁸⁰ According to her obituary, Drower began to study the literature, history, and religion of the Mandaeans in 1931; see J.B. Segal, "Obituary: Ethel Stefana, Lady Drower (1 December 1879–27 January 1972)," Bulletin of the School of Oriental and African Studies 35.3 (1972): 621.

⁸¹ See Text II, line 21.

that the colloquial dialect was falling into disuse all over the country in her day, she did not experience Socin's difficulty in enlisting the aid of native speakers, such as Hirmiz bar Anhar, who later became a friend and colleague to Sobbi. She even recorded (in an *ad hoc* transcription system of her own devising) a version of the "Bridge of Shushtar" legend that Petermann had collected eight decades earlier,⁸² but unfortunately never published it.⁸³ Rudolf Macuch was the first to publish this legend in the original language, although he never published it in its entirety, and the portion that he did publish differs in some respects from those of Petermann and Drower.⁸⁴

Five years later, however, Drower did publish an abridged translation of the legend in *The Mandaeans of Iraq and Iran*⁸⁵ along with some samples of what she called "bastard colloquial Mandaean [sic]."⁸⁶ Like Nöldeke before her, she attributed the irregularities of post-classical texts to the influence of the vernacular, which she dubbed a "debased jargon intermixed with foreign words."⁸⁷ Although Drower's ethnographic work among the Mandaeans is surprisingly modern in its sensitivity and objectivity, traits which were by no means common among her contemporaries, her wholly unjustified attitude towards the vernacular language belongs to an earlier era. This attitude may have contributed in part towards the mixed and often negative perceptions of the spoken language today within the Mandaean community itself (see §1.6 below). Nevertheless, the dozens of manuscripts that she brought to scholarly attention and furnished with exacting translations (which remain the standard to this day) have made it possible for scholars to apprehend the grammar of

- 82 Petermann, Reisen im Orient, 100.
- 83 The text is published for the first time here as Text IV in the text collection, courtesy of Jorunn J. Buckley, who possesses the original manuscript.
- 84 Rudolf Macuch, "The bridge of Shushtar. A legend in Vernacular Mandaic with Introduction, Translation and Notes," in *Studia semitica philologica necnon philosophica Ioanni Bakoš dedicata*, ed. Stanislav Segert, 153–172 (Bratislava: Slovenskej Akademie Ved, 1965).
- 85 Drower, The Mandaeans, 289–92.
- 86 Ibid., 214.
- 87 Ethel S. Drower, "A Mandaean Book of Black Magic," *Journal of the Royal Asiatic Society* (1943): 150.

the written language to a degree that would not have been possible otherwise.⁸⁸

Nearly three decades after Drower visited Ahvāz and Khorramshahr, Rudolf Macuch visited the community in Ahvāz. His encounter with the vernacular there was to have a profound influence upon his research. Although his *Handbook* treated the vernacular dialect as an auxiliary to the grammar of the Classical language, which was his primary focus, Macuch emphasized the importance of the vernacular in the first few pages of the book:

Nöldeke as well as a whole generation of mandaeologists [sic] after him knew nothing of the existence of a living colloquial Mandaic in Khuzistan, the language of the Mandaean quarters of Ahwāz and Khorramshahr, in which a part of classical forms and vocabulary continues its life until our day. Although this modern idiom with simplified morphology, overlaid [sic] with Arabic and Persian expressions and subdued to manifold foreign influences can furnish us only a very imperfect picture of original Mandaic as it was spoken sixteen or seventeen centuries ago, it is not void of interest. It may prove to be of similar importance as the study of modern Persian for Pahlavi.⁸⁹

He made no mention of Drower's trips to Ahvāz and Khorramshahr in 1932 nor of the references to the colloquial dialect in her 1937 and 1943 publications, which is surprising in light of Macuch and Drower's collaboration on *A Mandaic Dictionary*. This dictionary was published two years previously and made numerous references to colloquial forms, including those which Macuch had culled from the *Glossarium*.⁹⁰ Drower, who was the preeminent Mandaeologist of her

88 The work of producing translations and text editions of Mandaic manuscripts continues until the present day. One of the more recent editions published is that of the *Diwān Malkutā Əletā*; Jorunn J. Buckley, *The Scroll of Exalted Kingship* (**Diwan malkuta 'laita**), *Mandaean Manuscript No. 34 in the Drower Collection, Bodleian Library, Oxford* (New Haven, CN: American Oriental Society, 1993). The publisher of this volume has only just recently released several new text editions in its series *Mandäistische Forschungen*.

90 Macuch notes the fundamentally colloquial nature of the forms contained in the *Glossarium* at the very beginning of his *Handbook*; Macuch, *Handbook*, 1–2. He also deduces from the defective orthography employed by the author of the

⁸⁹ Rudolf Macuch, Handbook, liv.

time and by all accounts not only aware of the vernacular but also fairly fluent in it, surely must have informed Macuch about its existence, despite his claims to the contrary.⁹¹

Unfortunately, his departure from Iran in 1965 and the premature death of his informant Nasser Saburi in the same year prevented him from collecting any new data about Neo-Mandaic until 1990. The *Handbook* was released in 1965 to a mixed reception, which was somewhat mitigated by the fact that the data contained in it were available nowhere else and that it addressed an egregious lacuna in our understanding of Late Aramaic and Neo-Aramaic languages. Nevertheless, it suffered from some serious flaws:

The decision to combine the description of two widely divergent historical stages of the same language into a single handbook was perhaps not entirely felicitous and has raised some scholarly criticism. As a matter of fact, in the *Handbook* the modern vernacular is seen not so much as a language in its own right but as a continuation of a classical language; diachronic and synchronic treatments are merged. The picture of Neo-Mandaic which emerges from this treatment is somewhat blurred. There is no phonological analysis, the transcription is inconsistent and leaves the reader to guess what the underlying phonemic system might be. Similarly the morphological structure remains vague since the morphological categories and their functions are described only as reflexes of pre-existing older forms.⁹²

Although Macuch intended his handbook to encompass all stages of the language, he obstinately refused to incorporate any discussion of the incantation bowls into the dictionary or the *Handbook*, which was widely viewed as a serious omission.⁹³

Glossarium (which uses the Arabic *harakāt* to indicate vowels) that he was most likely unacquainted with the classical form of the language.

⁹¹ Macuch, *Handbook*, xlvi: "The existence of a vernacular dialect ... remained completely unknown until my personal discovery at the occasion of my visit to the Mandaean community of Ahwāz in 1953."

⁹² Otto Jastrow, review of Neumandaische Chrestomathie mit grammatischer Skizze, kommentierter Ubersetzung und Glossar. Unter Mitwirkung von Klaus Boekels by Rudolf Macuch, Bulletin of the School of Oriental and African Studies 55.3 (1992): 544–546.

⁹³ See, for example, Cyrus H. Gordon, review of *Handbook of Classical and* Modern Mandaic by Rudolf Macuch, Journal of Near Eastern Studies 26.2

Among the many scholars who tempered their praise for Macuch's contribution with a few sober notes of criticism was Joseph L. Malone, who composed a review of the *Handbook* targeted towards the general linguist.⁹⁴ Shortly after the publication of this review, Macuch contacted Malone directly and provoked him into publishing an apology in a subsequent issue of the same journal.⁹⁵ Several years later, Macuch savaged Malone (along with most of his other critics) in his *Zur Sprache und Literature der Mandäer*, protesting that he did not write his grammar for linguists, labeling Malone as "a complete beginner" and attributing to him an ignorance of "the elementary concepts of Mandaic verbal morphology."⁹⁶ It is perhaps not surprising, under the circumstances, that few other scholars took an interest in Neo-Mandaic during this time. To his credit, Malone continued to contribute to the study of Classical and Neo-Mandaic over the following decades, working primarily from Macuch's data.⁹⁷

Macuch's grandstanding and the shortcomings of his *Handbook* should not detract from the considerable contributions he made to the study of Neo-Mandaic, without which any future work would have been at a severe disadvantage. Macuch continued to work on Mandaic until the end of his life, finally publishing the texts he had gathered in

95 Joseph L. Malone, "Notes," Language 45.4 (1969): 951.

^{(1967):133} and Jonas C. Greenfield, review of *Handbook of Classical and Modern Mandaic* by Rudolf Macuch, *Oriens* 21 (1968): 394.

⁹⁴ Joseph L. Malone, review of *Handbook of Classical and Modern Mandaic* by Rudolf Macuch, *Language* 45.1 (1968): 191–203, 951.

⁹⁶ Rudolf Macuch, Zur Sprache und Literatur der Mandäer (Berlin: De Gruyter, 1976), 98. It should be noted that the subject of Malone's doctoral dissertation, completed two years before his review of the Handbook, was "A Morphological Grammar of the Classical Mandaic Verb."

⁹⁷ See, for example, his "Rules of synchronic analogy: A proposal based on evidence from three Semitic languages," *Foundations of Language* 5 (1969): 534–559, "Systematic Metathesis in Mandaic," *Language* 47.2 (1971): 394–415, "Classical mandaic radical metathesis, radical assimilation and the devil's advocate," *General Linguistics* 25 (1981): 92–121, "Modern and Classical Mandaic Phonology," in *Phonologies of Asia and Africa (Including the Caucasus)*, ed. Alan S. Kaye, 141 (Winona Lake, IN: Eisenbrauns, 1997), "Suffix metastasis and phonological change in classical Mandaic," *General Linguistics* 36.1–2 (1998): 1–16, and most recently "Metathesis and antiwedging in classical Mandaic," *General Linguistics* 36.4 (1999): 227–255.

Ahvāz in 1953 (after a hiatus of thirty-six years), along with his transcriptions of the texts published by de Morgan in 1904 and the portion of the "Bridge of Shushtar" legend that he had first published in 1965.⁹⁸ These were accompanied by a sketch grammar, the first attempt to describe the grammar of any Neo-Mandiac dialect as distinct from that of the classical language. Unfortunately, the flaws of the *Handbook* identified by Otto Jastrow above—namely his idiosyncratic transcription system and his failure to distinguish between diachronic and synchronic concerns—were retained in this new *Chrestomathie*.

The following year Macuch invited another Ahvāzi informant, Sheikh Salem Choheili, to Berlin. Macuch hosted Choheili for a full month and recorded thirteen cassette tapes of new material, the first Neo-Mandaic texts collected since 1953. He published these texts in transcription, accompanied by a second sketch grammar, only four months before his death in 1993.⁹⁹ In the introduction to his final work, Macuch wrote,

Under the current circumstances, the most urgent task facing Neo-Mandaic dialect research (which is at any rate rather restricted) remains the collection of samples from the dialect of Khorramshahr—formerly the strongest and most pure of the dialects, but today threatened by extinction. The dialect of Ahvāz will be preserved for future generations in its standardized form, but the dialect of Khorramshahr, which has never been standardized, will be lost without a trace. In a few years, it may already be too late to gather reliable traces of it.¹⁰⁰

I have outlined the history of Mandaic scholarship in order to illustrate a point. Although the Neo-Mandaic dialect of Khorramshahr has never been documented, and like many of the world's languages it is primarily unwritten, the business of preparing a descriptive grammar of Neo-Mandaic is unlike that of almost any other undocumented and unwritten language. Directly or indirectly, it has been the subject of

⁹⁸ Rudolf Macuch, Neumandäische Chrestomathie mit grammatischer Skizze, kommentierte Übersetzung und Glossar (Wiesbaden: Harrassowitz, 1989).

⁹⁹ Rudolf Macuch, Neumandäische Texte im Dialekt von Ahwāz (Wiesbaden: Harrassowitz, 1993).

¹⁰⁰ Ibid., xvii. Translated by the present author.

western scholarship for three and a half centuries. Throughout most of this time little or no distinction was made between the modern form of the language and the classical form, for which two excellent grammars and a dictionary exist. Most scholars were first introduced to the colloquial form of the language well over forty years ago. Consequently, a work such as this cannot be undertaken without constant reference to the scholarly tradition that precedes it. Many of these scholars may wish that I had employed a format analogous to that of Macuch for my description; others will perhaps feel that I have permitted his and Nöldeke's descriptions of the classical language to influence my own description of the modern one. In any case, an understanding of the history of scholarship surrounding this language is not only relevant but also crucial to any attempt at describing it.

1.6 Sociolinguistic Situation

My chief informant, Nasser Sobbi, was born in Khorramshahr on March 13, 1924. At the time, Khorramshahr, which was then known as Muḥammara, was the capital of the virtually autonomous sheikhdom of the same name, which controlled the province of 'Arabistān (which corresponded roughly to contemporary Khuzestān). The following year, Reza Pahlavi invaded Muḥammara and brought its ruler, Sheikh Khaz'al Khan, to Tehran, where he was placed under house arrest. For all official purposes, the province ceased to be known as 'Arabistān, and it was placed under the direct control of the central government in Tehran. In 1937, after the death of Khaz'al Khan, the Iranian cabinet voted to change the name of the city to Khorramshahr.

Among the many other changes brought by Reza Pahlavi to Khorramshahr was the introduction of universal and compulsory education in the language of the state, i.e. Persian, even among the largely Arabic-speaking inhabitants of the province. Not coincidentally, Sobbi's generation will almost certainly be among the last to speak Neo-Mandaic.

Until he was 18, Sobbi spent his life between Khorramshahr and Abadan, where he had worked as an apprentice jeweler to his uncle, Abdolkarim Moradi (currently of Syosset, NY), from the age of 14.

1. Introduction

After the outbreak of the Second World War, he traveled to Basra and Baghdad in search of work. In 1945, he returned to Khorramshahr and Abadan, as the oil industry had brought numerous foreigners (particularly British and Americans) to Khorramshahr and an attendant rise in the demand for jewelry. His family left Iran in 1952 to emigrate to Kuwait, where they resided until 1970. In that year, his family immigrated to the United States, where they have since resided.

1.6.1 Multilingualism and Language Attitudes

There are no monolingual speakers of Neo-Mandaic, and it is likely that there have not been any monolingual speakers for centuries, as they have always lived in close proximity to Arabs and have frequently been the minority in the places where they have dwelled. The remaining speakers of Neo-Mandaic are trilingual, also speaking the local dialects of Arabic (which are similar to that of Basra) and Persian, the language in which they were educated. Sobbi is probably unique in that he can read and write all three of these languages, as literacy in Mandaic is extremely uncommon.

The few remaining speakers of Neo-Mandaic are proud of their native tongue and would very much like to see it preserved. Members of the community in New York occasionally seek Sobbi out for lessons in the language, and the Mandaean community in Ahvāz has recently established a Mandaic-language school for children. Unfortunately, most Mandaeans (particularly those from Iraq) are completely unaware of the continued existence of their language, thinking it is only a part of the distant past. Other Mandaeans, who are aware of the existence of the modern dialect but do not speak it themselves, consider it to be a patois or even a kind of Arabic or Persian jargon.¹⁰¹ As mentioned previously, such attitudes towards

101 That is, Arabic (or Persian) with a few Mandaic words added. The Arabs of Khuzestān who have come into contact with Neo-Mandaic recognize that it has some similarities to Arabic, but is a completely different language; in the words of one Ahvāzi, *hum 'idhum fad luģa yaḥčūn-ha, mitl 'arabiyya bas maqlūb* 'they have a language that they speak, which is like Arabic but backwards.' John Harnsberry, personal communication.

undocumented languages are not uncommon, but, as in the case of Neo-Mandaic, they are often unjustified.

1.6.2 Contexts of Use and Language Choice

Sobbi speaks Neo-Mandaic natively and continues to use it on a regular basis with his wife Shukrieh and his brother Dakhil Shooshtary, as well as his uncle Abdolkarim Moradi, who has also immigrated to the United States. He has five children: four daughters, Freshteh, Juliette, Labiba, and Nabila, and one son, Isa. None of his children speaks Neo-Mandaic, although his son-in-law, Labiba's husband Tony Ebadi, grew up in a Neo-Mandaic-speaking household and can understand it. Consequently, the primary languages of his household are Arabic and English; his daughter Freshteh informed me that her parents speak in Mandaic when they do not wish to be understood by their children. Sobbi was educated in the standard dialect of Persian and can read and write all three of these languages. He has also become conversant in Urdu from his years of working in Kuwait with guest workers from Pakistan.

Sobbi describes his three primary languages thus:

When I was born, I spoke Mandaic. That was my mother and my father's language, and it was the language I grew up speaking. Our neighbors were Arabs, and so we talked Arabic with them. Later, when I went to school, I learned how to speak Persian, because Persian was the language of the schools.

Much has changed since his childhood, and today few of his relatives speak Neo-Mandaic. As noted above, he uses Neo-Mandaic primarily with his wife, his brother, and his uncle Moradi. Neo-Mandaic's sphere of usage is undeniably shrinking, as it has been shrinking for much of the last century. In the introduction to his *Handbook*, Macuch predicted:

As the knowledge of the literary language is dispensable to a Mandaean layman, since the sacramental rites produce their effect *ex opere operato*, so is also the knowledge of the vernacular, since living in an Arabic milieu all Mandaeans speak Arabic and in Iran mostly also Persian. Vernacular Mandaic as an exclusive language of Mandaean quarters and silversmith-workshops is of no great use to them. It survives only on account of its

simplicity in domestic and inter-Mandaean use being daily more and more invaded by the foreign elements by which it is surrounded.¹⁰²

1.6.3 Language Viability

The prognosis for the continued survival of Neo-Mandaic is grim. At this point in history, even the survival of the Mandaeans as a religious community is uncertain.

The dialect of Khorramshahr will almost certainly become extinct with the death of its few remaining speakers. Since Nasser Sobbi and his family left Khorramshahr, circumstances have resulted in the complete dispersion of the Khorramshahr community. During the Iranian Revolution in 1979, Khorramshahr was the scene of heavy fighting between the revolutionary guards and Arab nationalists. The following year, Saddam Hussein invaded and occupied Khorramshahr, which was subsequently recaptured by the Iranians in 1982. In the process, most of the city was destroyed and abandoned by its residents. Although the city has since been reconstructed, I am informed that the Mandaean community has not returned.¹⁰³

The Mandaeans elsewhere in Iran have not fared much better. Traditionally, the Mandaeans enjoyed a protected status as one of the peoples of the book. As a monotheistic religious community with a divinely-revealed scripture (the *Genzā Rabbā*) and a prophet (John the Baptist), they were free to practice their religion and could not be coerced to convert to Islam; prior to the Revolution, their status was codified in the laws of Iran. After the Iranian Revolution, the constitution of the Islamic Republic of Iran ceased to recognize them:

Zoroastrian, Jewish, and Christian Iranians are the only recognized religious minorities, who, within the limits of the law, are free to perform their religious rites and ceremonies, and to act according to their own canon in matters of personal affairs and religious education.¹⁰⁴

- 102 Macuch, Handbook, lxiv-lxv.
- 103 Nasser Sobbi, personal communication, June 13, 2003. Some Mandaeans have rebuilt their houses, but prefer to rent these out to non-Mandaean tenants rather than return; Jorunn Buckley, personal communication, March 19, 2006.
- 104 Article 13 of the Constitution of the Islamic Republic of Iran.

32

As a result, the Mandaeans have been plunged into a legislative limbo. In theory, they do not enjoy the freedom to practice their religion, nor do they have any representation in the Consultative Assembly, whereas Jews and Zoroastrians each elect one representative, Assyrian and Chaldean Christians jointly elect a single representative, and the Armenians elect two representatives.¹⁰⁵ To this day, their status under the laws of Iran has been a frequent subject of debate in the Consultative Assembly, but it has not yet been clarified.

In 1989, the government confiscated the *mandi* of Ahvāz, the chief Mandaean house of worship, and converted it into the headquarters for the Islamic religious police, the *Komiteh*. After waiting six years for the other shoe to drop, the Mandaeans were relieved when the Supreme Jurisprudent of Iran, Ali Khamenei, issued a fatwa stating that the Mandaeans possess the requisite characteristics to be considered a people of the book.¹⁰⁶ Although the Consultative Assembly has not yet amended the law to recognize the Mandaeans, they were subsequently permitted to open a private school in which their children could be schooled in their language. Much like other private schools in Iran, however, the headmaster must be a Muslim appointed by the Supreme ministry. Only time will tell if this will prevent the seemingly inevitable extinction of the Mandaic language.

1.6.4 Loan Words

Neo-Mandaic is unusual in that words borrowed from other languages are marked both phonologically¹⁰⁷ and morphologically; foreign nouns will only take possessive suffixes via an intermediary morpheme, *-d-*, and are pluralized using a different morpheme than that employed for native or nativized vocabulary.¹⁰⁸ Considering that the remaining speakers of Neo-Mandaic are fluent in Arabic and

- 105 Ibid., Article 64.
- 106 Buckley, The Mandaeans, 60.
- 107 Arabic loan words are commonly identified as such by the presence of phonemes not found elsewhere in the lexicon, such as \underline{d} , $\underline{\dot{g}}$, $\underline{\dot{h}}$, j, and $\overset{\circ}{.}$
- 108 For more information on the morpheme *-d-*, which appears on loan words before possessive suffixes, see Häberl, "Relative Pronoun."

Persian as well, most can identify loan words from these languages with near complete certainty. For these reasons, I have adopted a modified transcription system for words of foreign origin which have not yet been nativized to the Neo-Mandaic lexicon.¹⁰⁹

The loan words are not as pervasive as Drower and Macuch believed. Out of a list of two hundred and seven basic terms found in Table 5 at the end of this chapter,¹¹⁰ there are only fourteen loan words originally from Arabic, eleven loan words originally from Persian, and one possibly of Romance origin. There are also some common phrasal verbs that are calqued upon Persian verbs or incorporate Arabic and Persian elements.

Mandaic Form <i>wa<u>h</u>ši</i>	English Gloss animal	Language of Origin Arabic	Original Form <i>wa<u>h</u>šī</i>	English Gloss wild (adj.)
jihel	child	Arabic	jāhil	ignorant (adj.)
ġobār	dust	Arabic	ġubār	id.
baʿid	far	Arabic	baʿīd	id.
<i>ʿajāj</i>	fog	Arabic	'ajāj	dust cloud
j-m-d	to freeze	Arabic	j-m-d	id.
șun'a	good	Arabic	sun ^c	benefit
<u> </u> hašiš	grass	Arabic	<u> ḥašīš</u>	id.
şayd	hunt	Arabic	<i>sayd</i>	id.
qarn	horn	Arabic	qarn	id.
jadrā	root	Arabic	<i>ja<u>d</u>r</i>	id.
ġoșnā	stick	Arabic	ġuṣn	id.
hafif	thin	Arabic	xafīf	id.
<i>ț-w-</i> <u></u>	to throw	Arabic	<u></u> t-w- <u>h</u>	id.
dəros	correct	Persian	dorost	id.
perzi	a few	Persian	porze	a scrap
jangal	forest	Persian	jangal	id.
miwā	fruit	Persian	mive	id.

Table 3: Loan Words in Neo-Mandaic

- 109 This convention is comparable to the English orthographic convention of using Roman type for native and nativized words and italics for words still considered to be foreign.
- 110 This is the Swadesh list, which is commonly used for basic lexicography. The full 207 word list can be found in a number of sources, including Bert Vaux and Justin Cooper, *Introduction to Linguistic Field Methods* (Munich: LINCOM Europa, 1999), 44–45.

Mandaic Form	English Gloss	Language of Origin	Original Form	English Gloss
nazdik	near	Persian	nazdik	id.
gird	round	Persian	gird	id.
tiz	sharp	Persian	tiz	id.
narmā	smooth	Persian	narm	id.
rās	straight	Persian	rāst	id.
dəraxt	tree	Persian	deraxt	id.
par	leaf	Persian	par	id.
bābā	father	Unknown		
šerrā '-b-d	to fight	Persian	jang kardan	id.
šənā '-b-d	to swim	Persian	šenā kardan	id.
hol '-b-d	to push	Persian	hol dādan	id.
talg n-h- <u>t</u>	to snow	Persian	barf āmadan	id.
fekər '-b-d	to think	Persian	fekr kardan	id.
ohrā '-z-g- ^w /y	to walk	Persian	rāh raftan	id.

Table 3 (Cont.):

Note that some of these loan words differ substantially in meaning from the original form in the donor language, and a few have become partially nativized through the addition of the augment $-\bar{a}$. Even if we include words of foreign origin that have been completely adapted to Neo-Mandaic phonology, thus obscuring their origins, and phrasal verbs calqued upon other languages but composed entirely out of native lexical items, only 15% of the basic vocabulary in Table 5 can be attributed to foreign influence (primarily Arabic and Persian); the remaining 85% consists entirely of reflexes of Classical Mandaic vocabulary. When one considers the total lexicon of Neo-Mandaic, the proportion of loan words to native vocabulary rises; in this regard, however, Neo-Mandaic is comparable to most other languages, including Persian and even English. Certain semantic categories are dominated by loan words, while others remain virtually free of any foreign influence. The numbers employed, for example, are almost always the Persian ones, not the Mandaic ones; while speakers are generally familiar with both, they will only use the latter when they are making a conscious effort.

1. Introduction

1.7 Subdialects

After comparing his data from Ahvāz with recordings from Iraq made by Drower in 1953 and by Rudolph in 1969, Macuch declared in his contribution to *Zur Sprache und Literatur der Mändaer* that there were no differences in dialect between the Mandaean communities in Iraq and Iran.¹¹¹ While his conclusions undoubtedly hold true for the traditional pronunciation of Classical Mandaic, the recordings which he analyzed do not offer any idea of the pronunciation, grammar, or lexicon of the Iraqi vernacular, which was already moribund in 1937 and had most likely become entirely obsolete in 1969.

A better picture of the Iraqi vernacular is offered by the letters published by de Morgan in 1904, which originated in Baghdad.¹¹² While it is difficult to draw conclusions about the phonology of Iraqi Neo-Mandaic on the basis of these texts, the grammar and the vocabulary of the texts seem to accord closely with what we know about the two Iranian Neo-Mandaic dialects—including a liberal use of Persian loan words. Both Macuch's informant, Nasser Saburi, and my informant, Nasser Sobbi, were easily able to read these texts and even declared them to be highly entertaining. This is also true for the text sample which Drower had collected in Baghdad, which presented Sobbi with few difficulties in reading.

According to the testimony of Abdolġaffār Najmolmolk, many of the Mandaeans who were living in the Ottoman Empire toward the end of the 19th century were originally from Khuzestān.¹¹³ Drower and Sobbi have also confirmed that the knowledge of the vernacular was not widespread in Iraq in the early part of the last century, and at least one of the Iraqi Mandaeans who spoke Neo-Mandaic, Hirmiz bar Anhar, was descended from Mandaeans from Shushtar. On the basis of this anecdotal evidence and the indisputable conformity between the Iranian dialects and the few samples that we possess of the Iraqi dialects, it might be posited that the original Iraqi vernacular first encountered by Gerolamo Vecchietti in 1604 had died out at some

¹¹¹ Macuch, Sprache und Literatur, 75.

¹¹² De Morgan, Mission scientifique, 278-282.

¹¹³ See fn. 21 above.

point before the 19th century and that the Neo-Mandaic speakers encountered by Taylor, Petermann, and Drower in Iraq were the descendents of Mandaean refugees fleeing persecution in Shushtar, Dezful, and Shāh Wāli.

The Mandaeans of Shāh Wāli settled predominantly in Ahvāz, just as the Mandaeans of Shushtar and Dezful settled in Khorramshahr. As the Mandaeans rarely settled down in one place for very long, but were constantly traveling from community to community, these dialects were never in isolation from one another and remained mutually intelligible. Nonetheless, slight differences do exist, as illustrated by the samples in Table 4:

Nasser Sobbi	Salem Choheili	Nasser Saburi	English
(New York, 2003)	(Ahvāz, 1989)	(Ahvāz, 1953)	Gloss
biet	biet	bet	house
mienā	mienā	menā	water
waxt	vaxt	waqt	time
do <u>t</u> on	diyö <u>t</u> on	do <u>t</u> ōn	come (pl.)!
klātā	tlātā	klātā	three
orke	mork	orke	with
illi	ke	ke	which

Table 4: Dialect Differences

Salem Choheili, Macuch's informant in 1989, is a *yalupā* from Ahvāz, and a member of the chief priestly family in that city. Nasser Saburi, his informant in 1953, was also from Ahvāz, but not from a priestly family; like Nasser Sobbi, however, he was literate in Classical Mandaic. The most obvious distinction between the three columns is the distinction in vocabulary used by Sobbi and Saburi against Choheili. Both Sobbi and Saburi use the word *klātā* for three instead of *tlātā*, and the preposition *orke* instead of *mork*. Macuch mentions in his *Handbook* that the members of priestly families employ a different vocabulary than lay Mandaeans, which might explain the phenomenon observed here.¹¹⁴ Also note that Sobbi employs the colloquial Arabic

¹¹⁴ Macuch, *Handbook*, lviii–lix. Macuch, *Neumandäische Texte*, identifies Choheili as a candidate for the priesthood, but Choheili was not subsequently consecrated.

relative pronoun *illi* to introduce non-restrictive relative clauses and the Persian relative pronoun *ke* to introduce restrictive relative clauses, whereas the two Mandaeans from Ahvāz both use the Persian *ke* for both restrictive and non-restrictive clauses.

With regard to the phonology of the two subdialects, note that the vowel in the accented syllable of the words for 'house' and 'water' has broken in the speech of Sobbi and Choheili, but not in that of Saburi. Breaking is a typical feature of the Arabic dialects of southern Iraq and Khuzestān¹¹⁵ and may have spread into Neo-Mandaic within the last fifty years, which would explain why Saburi's speech does not share this feature.¹¹⁶ Also note that the voiceless uvular stop /q/ has been fricativized in the Arabic word *waqt* 'time' as pronounced by Sobbi and Choheili, but not Saburi; while fricativization does occasionally occur in Saburi's speech,¹¹⁷ it does not appear to be as pervasive as it is in the speech of the others. The fricativization of this word can be directly attributed to the influence of Persian, in which the word is commonly pronounced /væxt/. All occurrences of the voiced bilabial approximant /w/ in Sobbi and Saburi appear to correspond to the voiced labiodental fricative /v/ in Choheili, although, as Jastrow noted, it is difficult to ascertain the phonetic realization of this phoneme from Macuch's transcriptions.¹¹⁸

A further feature of Choheili's speech, not found in either Sobbi or Saburi's speech, is the glide that often appears in accented syllables, e.g. *diyötön* '(all of you) come!' The centralization of lax vowels in accented syllables is typical of both Choheili and Sobbi's speech; this feature was also typical of the speech of Siouffi's informant, but it has spread much further in Choheili's speech than it has in any of the others.

- 115 Bruce Ingham, "Languages of the Persian Gulf," in *The Persian Gulf States: A General Survey*, ed. by Alvin J. Cottrell, 323–24 (Baltimore: Johns Hopkins, 1980).
- 116 No examples of breaking appear in the "Bridge of Shushtar" text collected by Drower, ca. 1932.
- 117 Macuch, Neumandäische Chrestomathie, 42.
- 118 Jastrow, review of Neumandäische Chrestomathie, 545.

1.7 Subdialects

One of the most prominent differences between the subdialect of Sobbi and his wife and that of Macuch's informants is the complete absence of feminine plural forms in the paradigm of the verb. Both Saburi and Choheili gave second and third feminine plural forms of the verbal paradigm when they were elicited, and presumably used them in their daily speech, but these forms are not present at all in Sobbi's speech. Whether these represent a conscious archaism on the part of Macuch's informants or a genuine survival of the Classical Mandaic feminine plural forms is difficult to say.

The question of the differences between the various subdialects of Neo-Mandaic and their relationship to the classical language will be reexamined in Chapter 6. Further speculation on the dialect geography of Neo-Mandaic is hampered by the lack of informants from other dialect groups. Although more informants from Khorramshahr and Ahvāz are necessary to complete the picture, much more information could be derived from Macuch's texts if they were re-transcribed with a phonemic transcription system such as the one employed in this grammar. This is a necessary first step to writing the grammar of the dialect of Ahvāz, which Macuch never completed during his lifetime. Additionally, the contribution of a Neo-Mandaic speaker descended from the community of Dezful, such as Asad Askari of Ahvāz, would go very far towards completing the picture of the dialect geography of Neo-Mandaic.

No.	English	Neo-Mandaic	IPA	Transcription
17	all	gaš	[gɛ∫]	geš
204	and	u	[wə]	WƏ-
44	animal	uahši	[ˈwæħ∫i]	waḥši
168	ashes	tauša / țauša	['t ^s aʊ∫o]	țawšā
201	at	qar	[tcp]	qār
88	back	butar	['bɔ:θa.ı]	bātar
186	bad	bțala	[bəˈt ^s ɔːlɔ]	bəțālā
58	bark	mašk šagra	[ˌmɛ∫kˈʃɪʤrɔ]	mešk šejrā
206	because	ka	[kɛ]	ke
85	belly	karsa	['ka.isə]	karsā
27	big	rabu	[ˈræbbɔ]	rabbā
46	bird	supra	['s ^s oprɔ]	șoprā
94	bite	nakti	[nəˈxot]	<u>n-k</u> -t

Table 5: Swadesh List of 207 Common Terms in Neo-Mandaic

Table 5 (Cont.):

No.	English	Neo-Mandaic	IPA	Transcription
176	black	akumu	[oˈkuːmɔ]	okumā
64	blood	dma / zama	[dəˈmɒː]	dəmā
98	(it) blows	rahu ata	[rehhu ɛ'θɔ:]	rehhu '- $t^{-w}/_y$
65	bone	garma	[ˈɡɛɹmɔ]	germā
89	breast	8	[8]	8
99	breathe			
169	burn	qla	[qəˈliː]	q-1-w/v
39	child	gihal	[ˈdʒiːhɛl]	jihel
160	cloud	aiua	['i:və]	ibā
181	cold	msi	[məˈsiː]	məsi
122	come	duti	['do:θi]	$\frac{2-t}{y}$
196	correct	durust	[do'ros]	dorós
139	count	mni	[məˈniː]	m-n-w/v
114	cut	pşqi	['posqi]	p-s-q
178	day	iuma	['ju:mɔ]	yumā
109	die	mt	[me0]	m^{-w}/v^{-t}
118	dig	hpr	[həˈfo.]	h-p-r
188	dirty	taušana	[t ^s au'[o:no]	tawšānā
47	dog	kalba	[ˈkælbɔ]	kalbā
92	drink	št	[ʃot]	$\check{S}^{-W}/_{V}-t$
195	dry	ibuša	[jæˈbu:ʃɔ]	vabušā
192	dull	10 010 01	[]]0]	<i>j</i>
158	dust	gbar	[ку,рэт]	ğubār
73	ear	adna / ana	['onno]	onnā
159	earth	dašta	[ˈdɛʃtɔ]	deštā
93	eat	akal	[xol]	² -k-1
67	egg	baia	['bɛɪjɔ]	beyyā
74	eye	aina	['i:nɔ]	inā
127	fall	npal	[nəˈfol]	n-p-1
198	far	baiid	[ba'sid]	baʿid
66	fat (n.)	šamna	[ˈʃæmnɔ]	šamnā
43	father	baua	[ˈbɔːwɔ]	bābā
106	fear	dhalta	[dəˈholtə]	dəholtā
70	feather	gapu / gapa	[ˈgæppɔ]	gappā
20	few	parzi	['pɛɹzi]	perzi
111	fight	šr	[bov'e 13]	šer '-b-d
79	fingernail			-
167	fire	nura	['nu:rɔ]	nurā
45	fish	panida	[fo'ni:do]	fonidā
26	five	hamša	['hæm(ɔ]	hamšā
143	float	šanu	[sino:]	šenā
144	flow	npaq	[nə ^l foq]	n-p-q
59	flower	gal / gul	[gol]	gol
127	fly	phar	[tod'eq]	p-h-r
161	fog	•	[sa'dsods]	^r ajāj
80	foot	kraia	[kəˈrɔːjɔ]	kərāyā
52	forest		[dænˈgal]	jangal
25	four	arbu	['a.bɔ]	arbā

Table 5 (Cont.):

No.	English	Neo-Mandaic	IPA	Transcription
145	freeze	gmad	[ʤəˈmod]	j-m-d
54	fruit	miua	['mi:wo]	miwā
182	full	mli	[məˈliː]	m-1-w/v
128	give	ahau	[əˈhov]	<u>²-h-b</u>
185	good	şanii / şanua	[s [°] ʌnaˤa]	şun'a
60	grass	hašiš	[ħaˈʃiʃ]	hašiš
173	green	iaraqa	[jæˈroːqɔ]	yaroqā
86	guts	karsa	['kaisə]	karsā
71	hair	manza	['mænzɔ]	manzā
83	hand	aida	['i:dɔ]	idā
3	he	hui	['huwi]	huwi
72	head	riša	[ˈri∫ɔ]	rišā
102	hear	şat	[s ^s ot]	$S^{-W/y}-t$
90	heart	libu	['lɛbbɔ]	lebbā
31	heavy	kadira	[kæˈdiɪrɔ]	kadirā
9	here	hna	[ɛhˈnɔː]	ehnā
113	hit	mhi	[məˈhiː]	$m-h^{-w}/_{v}$
129	hold	lkuț	$[la'\chi ot^{S}]$	1-k-ț
68	horn	qarn	[qæ.m]	qarn
15	how	kmu	['kæmmɔ]	kammā
112	hunt	şaid	[s ^s aɪd]	sayd
41	husband	bla	[ˈbɛllɔ]	bellā
1	Ι	ana	[ˈæːnɔ]	anā
165	ice	talga	['tælgɔ]	talgā
205	if	agr	['ægæı]	agar
202	in	gu	[gou]	gāw
110	kill	gtal	[gəˈt ^s ol]	g-t-1
82	knee	burka	['bʌ.ɪkɔ]	burkā
103	know	bašqir	['baʃqɛı]	b-š-q-r
153	lake	iama	[ˈjæmmɔ]	yammā
100	laugh	ghak	[gəˈhox]	g-h-k
56	leaf	pr	[pæı]	par
200	left	asmala	[ˈsmɔːlɔ]	smālā
81	leg	kraia / lagra	[ˌlors.əl]	logrā
123	lie	gni	[gəˈniː]	g-n-w/y
108	live	hian hauin	[həˈjɔːnɔ]	həyānā
91	liver	kabda	[ˈkæbdɔ]	kabdā
28	long	arik	[əˈriːɣə]	ərikā
48	louse	qmulta	[qəˈmolθɔ]	qəmoltā
37	man	gabra	[ˈgævrɔ]	gabrā
18	many	ginza	['genzo]	genzā
63	meat	basra	['bɛsrɔ]	besrā
148	moon	sara	[ˈsɛrrɔ]	serrā
42	mother	ama	[ˈɛmmɔ]	emmā
171	mountain	ţur	[ˈtˤuːrə]	țurā
76	mouth	pama	['pommo]	pommā
207	name	ašma	[ˈɛ∫mɔ]	ešmā
34	narrow	zalil	[zæ'li:lɔ]	zalilā

Table 5 (Cont.):

No.	English	Neo-Mandaic	IPA	Transcription
197	near	nazdik	[næz ['] dik]	nazdik
87	neck	paqata	[fo'qAtt ^s o]	poquttā
183	new	hata	['hæθθɔ]	hattā
177	night	lilia	['lɪljɔ]	lilyā
75	nose	nhira	[nəˈhiːrə]	nəhirā
16	not	lu	[lu]	lu
184	old (an.)	rba	['ræbbɔ]	rabbā
10.	old (inan.)	hatiga	[hæˈtiĕɡɔ]	hatieqā
22	one	ahda	['ehdo]	ehdā
21	other	hurina	[ho'ri:no]	horinā
38	person	barnaša	[baj'no:[o]	barnāšā
142	play	talal	['t ^s allel]	D <i>t-1-1</i>
134	pull	tli	[təˈliː]	$t-l^{-w}/y$
135	push	hul aud	[hol ə'vod]	hol '-b-d
155	rain	matra	['mat ^s ro]	mațrā
172	red	şmaqa	[səˈmɔːqɔ]	səmāqā
199	right	iamin	[jɛˈmiːnɔ]	yaminā
152	river	nahra	['næhrɔ]	nahrā
170	road	ahra	['ohro]	ohrā
57	root	gazra	[ˈdzæðrɔ]	jadrā
61	rope	mašua	['mæ(wɔ]	jauta mašwā
187	rotten	sari	[sə'ri:]	səri
190	round			
190	rub	gird narki	[gi.id]	gird n n h
		parki	[ˈpoɪɣi]	p-r- <u>k</u>
155	salt	mahla	[ˈmɛhlɔ]	mehlā
157	sand	hala	['hɔ:lɔ]	hālā
140	say	amar	[əˈmoɪ]	² -m-r
117	scratch	huk	['hoːχɔ]	hokā
154	sea	iama	[ˈjæmmɔ]	yammā
101	see	hzi	[həˈziː]	h-z-w/y
55	seed	bazra	[ˈbæzrɔ]	bazrā
138	sew	huț	[hot [§]]	$h^{-w}/_{y}$ -t
191	sharp	tiz	[tiz]	tiz
33	short	gțan	[gəˈt ^s ɔːnɔ]	gəțānā
141	sing	zmr	[zəˈmo.]	z-m-r
124	sit	ihm	[jəˈhom]	<i>y-h-m</i>
62	skin	maška	[ˈmɛ∫kɔ]	meškā
162	sky	arqiha	[r'qi:hə]	ərqihā
107	sleep	gni	[gəˈniː]	g-n-w/y
32	small	hanin	[hoˈniːnɔ]	honinā
105	smell (v.)	arah	[ˈærrɛ]	D '- <i>r-h</i>
	smell (n.)	riha	[ˈriːhɔ]	rihā
166	smoke	gatra	[ˈgʌtˤrɔ]	guțrā
193	smooth	nrm	[ˈnaɹmɔ]	narmā
49	snake	hauia	[ˈhɛvjɔ]	hewyā
164	(it) snows	talg nihat	[talg nəˈhaθ]	talg nəha <u>t</u>
19	some	parz	[ˈpɛɹzi]	perzi
96	spit	araq	['arreq]	D '- <i>r-q</i>
1.7 Subdialects

Table 5 (Cont.):

No.	English	Neo-Mandaic	IPA	Transcription
115	split	palga	[palrb]	palgā
130	squeeze	asra	['as ^s ra]	C s-r-w/y
116	stab	mhi	[məˈhiː]	$m-h^{-w}/v$
125	stand	bgaş	[pəˌRo2 ₈]	b-g-ș
149	star	kakua	['koywo]	kokbā
53	stick	gasna	['RV2 _s u2]	ğuşnā
156	stone	glal	[gəˈlæltɔ]	gəlaltā
189	straight	raș	[ros]	rās
95	suck	šat	[∫ot]	\check{S} - $W/_{y}$ -t
147	sun	šamaš	[ˈ∫ɔ:mɛʃ]	šāmeš
146	swell	smar	[s ^s əmo]]	s-m-r
119	swim	šnu	[∫əˌnɔ ə'vod]	šenā '-b-d
69	tail	dama	[ˈdɛmmɔ]	demmā
8	that	aku	[ˈaχu]	a <u>k</u> u
10	there	akak	[ɛkˈkaɣ]	ekka <u>k</u>
6	they	huni	['honni]	honni
30	thick	smika	[səˈmiːɣɔ]	səmi <u>k</u> ā
	thick (fat)	alima	[əˈliːmɔ]	əlimā
35	thin	hapip	[hæˈfif]	hafif
104	think	pakr auad	[ˌfɛkrɛˈwod]	fekər '- <u>b</u> -d
7	this	auha	[ˈɔːhɔ]	āhā
2	thou	at	[ɔt]	āt
24	three	klata	[ˈklɔ:θɔ]	klātā
136	throw	țuhi	[ˈt ^s aʊħi]	<i>ṭ-[₩]/_y-ḥ</i>
137	tie	uaiaq	[ˈvaɪjɛq]	W - W/y - q
78	tongue	lšan	[lɛˈ∫ɔːnɔ]	lešānā
77	tooth	šana	[ˈʃɛnnɔ]	šennā
51	tree	drakt / šagra	[dəˈraxt]	dəraxt
126	turn	hadar	[həˈdo.ɪ]	h-d-r
23	two	trin	[trɛn]	tren
97	vomit	gha	[gəˈhɔː]	$g-h-w/_y$
121	walk	ahr azgu	[ˌohhr ɛzˈgiː]	ohhər '-z-g-"/y
180	warm	hamim	[hæˈmim]	hamim
132	wash	hauar	[ˈhawwɛɹ]	h - w/y - r
150	water	mia / mina	[ˈmiĕnɔ]	mienā
4	we	ani	[ˈæni]	ani
194	wet	msamar	[məˈsˤammaɪ]	məşammar
12	what	mia	[mu]	mu
14	when	hamdu	[ˈhɛmdɔ]	hemdā
13	where	alia	[ˈɛljɔ]	elyā
175	white	hbara	[həˈwɔːrɔ]	həwārā
11	who	man	[mæn]	man
29	wide	patia	[ˈpæθjɔ]	patyā
40	wife	ata	[ˈɛθθɔ]	e <u>ttā</u>
163	wind	rahua	[ˈrɛhwɔ]	rehwā
84	wing	gapa	[ˈgæppɔ]	gappā
133	wipe			
203	with	ark	[o.k]	ork

Table 5 (Cont.):

No.	English	Neo-Mandaic	IPA	Transcription
36	woman	ata (ainša)	[ˈɛθθɔ]	e <u>ttā</u>
50	worm	tla	[ˈtʌljɔ]	tulyā
179	year	šta	[ˈʃɛttɔ]	šettā
174	yellow	sapra	['sofrɔ]	soprā
5	you	atan	[ˈætton]	atton

2. Phonology

2.1 Preliminary Considerations

A number of factors complicate the study of Neo-Mandaic phonology. The first and most obvious complication is language contact. Extensive borrowing from languages such as Arabic and Persian has expanded the phonemic inventory of Neo-Mandaic. Four loan-phonemes, the postalveolar affricates \check{c} /tʃ/ and j /dʒ/ and the pharyngeal fricatives $(\Lambda)^{\Gamma}$ and \hbar /h/, characteristic of Arabic and Persian but not originally of Mandaic, ¹¹⁹ have been introduced into the phonology. Three other segments, a voiced interdental fricative $d/\delta/$, its pharyngealized equivalent $d/\delta^{\Gamma}/$, and a voiced alveolar fricative $z/z^{\Gamma}/$, are found in a few loan words of Arabic origin. They have been excluded from the phonemic inventory of Neo-Mandaic due to their marginal status. Furthermore, the fricatives /v/, /B/, $/\partial/$, $/\chi/$, and $/\theta/$ have a much different distribution in loan words; typically, their occurrence in word-initial position identifies a word as being of foreign origin.

Loan words which are identifiable as such (due to the presence of loan-phonemes among other factors) take a separate set of plural and possessive morphemes indicating their foreign origins. For these reasons, the phonemes f/, v/, $\chi/$, or $\kappa/$ are assigned the values f, v, x, and \dot{g} whenever they appear in words of foreign origin (rather than \tilde{p} , b/, k/, and \bar{g} respectively). The orthography of a given word should therefore indicate whether the word is a loan word or not and,

119 It should be noted that $\check{c}/tf/$ does appear in a small number of Mandaic words in which the phonemes t/t/ and $\check{s}/f/$ are articulated together. The absence of these phonemes in earlier stages of the language is clear from the fact that they appear only in loan words and are not consistently represented in the traditional writing system, which lacks characters for these phonemes and employs a variety of *ad hoc* orthographic strategies to represent them. consequently, whether it is likely to be marked as such in the morphology.

A second and more challenging obstacle (if less immediately obvious) is posed by the relatively large number of allophones attested for each phoneme. Among the thirty-five distinct consonantal and vocalic segments, at least two allophones are attested for all but nine segments, and thirteen segments have three or more regular realizations. Furthermore, a variety of sound changes have obscured the original phonemes of the language and given rise to new phonemes (such as /ə/, §2.3.2.4). Their allophones are generally conditioned by syllable structure, neighboring segments, and the position of the accent in relation to the syllable in which the segment occurs (§2.4); /r/, for example, is realized as a voiced alveolar trill [r] in syllable-initial position and an approximant [J] in post-vocalic position (see §2.3.1.4 below). The transcription system, which is phonemic, does not reflect these allophones; nor does it reflect sporadic assimilations, deletions, and other features of allegro speech.

Whenever I have established the phonemic status of a given segment, I have attempted to isolate the precise phonological environment that determines each of its allophones. The success of these attempts has been mitigated by a third important complication: unexplained phonetic variation. I hesitate to attribute all examples of unexplained phonetic variation to free variation, as it is clear that some of them (such as the alternation between Persian and Arabic pronunciations of the same Arabic word) have sociolinguistic or dialectal explanations.¹²⁰ Another important source of variation is the analogical and paradigmatic leveling of phonemes within a word. Even where a particular phoneme has multiple allophones determined by its environment, phonemes tend to be leveled towards one realization across inflectional paradigms no matter what the phonetic environment may be. This is the case with the phoneme /v/ in the

¹²⁰ Unfortunately, as noted in §1.7, I have only just begun to collect the evidence for Neo-Mandaic dialect variation, which falls beyond the scope of this grammar.

verbal root $\partial hab \sim \partial hob (\bar{a}heb)$ 'to give,' which is most often realized as a [v], regardless of its environment.

2.2 Phonological Units

This inventory of Neo-Mandaic phonemes begins with the consonants (see Table 6 below), and is organized primarily according to point of articulation and secondarily by presence or absence of voice and pharyngealization. Each consonant phoneme is introduced by its value in the transliteration scheme adopted specifically for Neo-Mandaic (details are found in Table 7 below) and is followed by its value in the International Phonetic Alphabet (given between /slashes/) and its chief allophones (given within [brackets]). Minimal pairs are supplied for consonants only where the phonemic independence of a particular phoneme might be considered controversial, because its allophones merge with those of other phonemes. The consonants are followed by the vowels, which are also organized by articulation and furnished with minimal pairs to demonstrate their phonemicity.

Wherever possible, I have attempted to supply examples from the text collection that accompanies this grammar in the second volume. Each such example is followed by a series of numbers indicating the number of the text and the specific line from which it has been collected. The Roman numeral (I-X) corresponds to the relevant text from among the ten, and the Arabic numeral (1-150) that follows it corresponds to the line within the text. Letters (a-z) indicate the arbitrary divisions in each line that correspond to the lines of the interlinear morphosyntactic glosses that follow each text. In some cases, it was not possible to supply examples directly from the text collected (including the Swadesh list, Table 5). These examples are not followed by any cipher.

Table 6: Phoneme Inventory (Consonants)

Stops and Affricates

	Bilabial	Labio-	Inter-	Alveolar	Post-	Palatal	Velar	Uvular	Pharyn	Glottal
		dental	dental		Alveolar				-geal	
Voiceless	р			t	ţſ		k	q		
Voiced	b			d	գ		g			
Emphatic				ť						
Fricatives										
	Bilabial	Labio-	Inter-	Alveolar	Post-	Palatal	Velar	Uvular	Pharyn	Glottal
		dental	dental		Alveolar				-geal	
Voiceless		f	θ	s	ſ			χ	ħ	h
Voiced		v		z				R	Ŷ	
Emphatic				s [°]						
Sonorants										
	Bilabial	Labio-	Inter-	Alveolar	Post-	Palatal	Velar	Uvular	Pharyn	Glottal
		dental	dental		Alveolar				-geal	
Nasal	m			n						
Lateral				1						
Apical				r						
Approximant	w					j				

Table 7: Transcription Scheme (Consonants)

Stops and Affricates

	Bilabial	Labio-			Post-	Palatal	Velar	Uvular	Pharyn	Glottal
		dental			Alveolar				-geal	
Voiceless	р			t	č		k	q		
Voiced	b			d	*j		g			
Emphatic				ţ						
Fricatives										
	Bilabial	Labio-	Inter-	Alveolar	Post-	Palatal	Velar	Uvular	Pharyn	Glottal
		dental	dental		Alveolar				-geal	
Voiceless		\bar{p}/f	<u>1</u>	5	š			<u>k</u> /*x	* <i>ḥ</i>	h
Voiced		<u>b</u> /*v		Z				$ar{g}/{}^{*}\dot{g}$	¢	
Emphatic				*ș						
Sonorants										
	Bilabial	Labio-	Inter-	Alveolar	Post-	Palatal	Velar	Uvular	Pharyn	Glottal
		dental	dental		Alveolar				-geal	
Nasal	m			п						
Lateral				1						
Apical				Г						
Approximant	W					У				

* The use of f, v, j, x, \dot{g} , or \dot{h} in a word indicates that it is of Arabic or Persian origin.

		-) (- /	
	Fron	t	Central	Back
Close	/i/			/u/
Mid		/e/	/ə/	/0/
Open		/a/		/p/

Table 8: Phoneme Inventory (Vowels)

Table 9: Phonemic Transcription Scheme (Vowels)

	Front		Central	Back
Close	i			U
Mid		е	ð	0
Open		а		ā

2.3 Phonetic Characteristics of Distinctive Segments

There are 35 distinctive segments in Neo-Mandaic: 28 consonants (§2.3.1) and 7 vowels (§2.3.2). The consonants are organized according to their points and manner of articulation as well as any coarticulations; each section includes a description of the most common allophones and the environments in which they occur. The vowels are also organized according to articulation and include descriptions of their most common allophones.

2.3.1 Consonants

There are 28 distinctive consonants in Neo-Mandaic, including two pharyngeal consonants (§2.3.1.9) that are found only in loan words borrowed from Arabic (see §1.6.4), and two post-alveolar affricates (§2.3.1.5) that are found primarily in loan words borrowed from that language and Persian (but see §2.1 above).

2. Phonology

2.3.1.1 Labials

р	/p/	[p ^h], [p]
b	/b/	[b]
т	/m/	[m], [ṃ]
W	/w/	[w], [ʊ], [v], [u]

Voiceless bilabial stop (aspirated): /p/ is lightly aspirated in most positions, except when it is the first element in a consonantal cluster, as in (2.2), or when it follows a fricative, as in (2.3). Nevertheless, aspiration is not generally indicated in the transcriptions, apart from those given below.

(2.1)	pol	[p ^h ol]	bridge (IV.19)
(2.2)	epseqni	[ɛp.ˈsɜq.ni]	we died out (I.14)
(2.3)	kaspā	[ˈkæs.pɔ]	silver (III.5)

Voiced bilabial stop: This phoneme is sporadically weakened to the voiced labiodental fricative /v/ in post-vocalic position (see 2.4.1.6). Nonetheless, the phonemic opposition between /b/ and the voiced labiodental fricative /v/ is demonstrated by the minimal pair (2.4) and (2.5).

(2.4)	zabnu	['zab.nu]	he sold them (V.5)
(2.5)	za <u>b</u> nu	['zav.nu]	he bought them

Bilabial nasal sonorant: Much like the other sonorants (q.v.), this segment becomes syllabic in word-initial position when followed by a consonant, particularly in the participial forms of the derived stems; in the phonemic transcriptions, [m] is represented by the digraph ∂m , as in (2.6); the syllabic status of the phoneme is confirmed by the absence of the usual anaptyctic / ∂ / to break up the consonant cluster, as in (2.7), or the prothetic / ∂ / before the initial consonant, as in (2.8); see §2.3.2.4 for the normal outcome of / ∂ / in these environments.

(2.6)	əmbašqer	[m.′ba∫.qɛı]	knowing (IV.35)
(2.7)	**məbašqer	[mə.′ba∫.qɛı]	id.
(2.8)	**əmbašqer	[ɛm.ˈba∫.qɛɹ]	id.

2.3 Phonetic Characteristics of Distinctive Segments

Labial-velar approximant: In the environment of the back vowels /u/, /o/, or /ɔ/, as in examples (2.9) - (2.11), /w/ is most often realized as a voiced labial-velar approximant [w], even in word-initial position. When following a back vowel, particularly /p/, /w/ is generally pronounced as [υ], forming a diphthong with the preceding vowel, e.g. (2.12) and (2.13).

(2.9)	gāwwu	[ˈɡɔʊ.wu]	in them (I.11)
(2.10)	wușle	[was ^s .'le]	piece (IX.12)
(2.11)	ziwā	[¹ zi:.wɔ]	splendor (I.9)
(2.12)	əlāw	[ə.ˈlɔʊ]	to (III.13)
(2.13)	gāw	[gou]	in (I.4)

/w/ is frequently realized as a voiced labiodental fricative [v] in intervocalic position whenever it follows or preceding a closed front vowel, either /e/, as in (2.14), or /i/, as in (2.15) and (2.16), but note (2.17).

(2.14)	hawén	[hæ.'vɛn]	may we be (I.14)
(2.15)	gāwwi	['gou.vi]	in him (X.1)
(2.16)	diwā	['di:.vɔ]	demon (IV.45)
(2.17)	huwi	['hu:.wi]	he (IV.35)

In certain environments, /w/ may become syllabic, in which case it merges with the close back vowel /u/. For examples and further details, see §2.4.2.5 below.

Previously this phoneme was distinct from the voiced labiodental fricative /v/, but the two have merged in many environments. There is also a considerable degree of variation in the realizations of both these phonemes. Nevertheless, these two phonemes remain distinct in the verbal root, as /v/ is retained throughout the inflectional paradigm of the verb whereas /w/ generally succumbs to deletion whenever it appears as the second or third radical consonant of the verbal root.

2.3.1.2 Labiodentals

$$ar{p}/f$$
 /f/ [f]
 b /v/ [v], [w], [v]

2. Phonology

Voiceless labiodental fricative: The independence of this phoneme from /p/ is demonstrated by the minimal pair (2.18) and (2.19).

(2.18)	șoprā	['s [°] oprɔ]	bird
(2.19)	șoprā	['s [°] ofrɔ]	yellow

Voiced labiodental fricative: /v/ is generally realized as a voiced labiodental fricative in intervocalic and post-vocalic position in the environment of a close front unrounded vowel, either /e/, as in (2.20) and (2.21), or /i/, as in (2.22).

(2.20)	<u>şəbyi</u>	['s ^s ev.ji]	he baptized him (I.9)
(2.21)	qā <u>b</u> ed	['qp:.ved]	he will do (IV.22)
(2.22)	láhriþi	['læh.rɪ. _l vi]	he hasn't destroyed it (IV.71)

When occurring in the environment of an open back rounded vowel /u/, as in (2.23), /o/, as in (2.24), or /o/, as in (2.25), /v/ is frequently realized as the labial-velar approximant [w], even in word-initial position, as in (2.26).

(2.23)	ə <u>b</u> ud	[ə.ˈwud]	do (f.sg.)! (IV.46)
(2.24)	ə <u>b</u> od	[ə.ˈwod]	do (m.sg.)! (IV.62)
(2.25)	qəmahre <u>b</u> āt	[qə. ₁ mah.rɛ.'wət]	you will destroy (IV.58)
(2.26)	<u>bābā</u>	['wɔ:.wɔ]	door (IX.5)

2.3.1.3 Interdental

t / θ /

Voiceless interdental fricative: The minimal pair (2.27) and (2.28) illustrates the contrast between this phoneme and the voiceless alveolar stop /t/.

[0]

(2.27)	a <u>tt</u> on	['aθ.θon]	they brought (IV.14)
(2.28)	atton	['at.ton]	you (pl.) (IV.110)

2.3.1.4 Alveolars

t	/t/	[t ^h], [t]
d	/d/	[d]
ţ	/t [°] /	[t [°]], [t]

\boldsymbol{S}	/s/	[s]
Ζ	/z/	[z]
Ş	/s [°] /	[s [°]], [s]
п	/n/	[n], [ŋ]
1	/1/	[1], [1], [1], [1], [1]
r	/r/	[r], [J]

Voiceless alveolar stop: /t/ is generally aspirated, as in (2.28), except when it occurs as the initial segment in a consonant cluster or following a fricative, as in (2.29) and (2.30). Apart from these examples, aspiration is not noted in the transcription. This phoneme is occasionally weakened to the voiceless interdental fricative / θ / in intervocalic position or in word-final position after a vowel; see 2.7.1.7.

(2.28)	āt	[ot ^h]	you (sg.) (V.23)
(2.29)	etber	[ˈɜt.fɛɹ]	it broke (IX.12)
(2.30)	əštā	[ə∫.'tɔ:]	now (I.10)

Voiced alveolar stop: This phoneme does not have any significant allophonic variation. In one instance, this phoneme is weakened to the voiced interdental fricative $/\delta$; see §2.7.1.7.

(2.31) $ehd\bar{a}$ [' ϵ h.dɔ] one	ne (IV.43)
---	------------

Voiceless pharyngealized alveolar stop: The pharyngeal articulation of the voiceless alveolar stop is very weak, at times barely audible, which is typical of secondary articulations in Neo-Mandaic such as aspiration. It is often lost when it is the first segment in a consonantal cluster, as in (2.33). When $/t^{\circ}/$ is weakened to [t] in other environments, it remains distinct from /t/, which is generally pronounced with aspiration.

(2.32)	<u>țā</u> b	[t ^s ou]	good (IV.38)
(2.33)	bațlu <u>k</u> tā	[bat.'lux.tə]	misfortune (I.14)

Voiceless alveolar sibilant: This phoneme has no significant allophonic variation.

2. H	Phonology

(2.34)	sāgat	['sɔː.qat]	she went up (IV.145)
(2.34)	sayai	[Soliqui	

Voiced alveolar sibilant: This phoneme has no significant allophonic variation. In fast speech, it is occasionally devoiced before voiceless stops; see §2.7.1.1.

	(2.35)	qāzi	['qɒ:.zi]	he goes (VIII.1)
--	--------	------	-----------	------------------

Voiceless pharyngealized alveolar sibilant: Much like other secondary articulations in Mandaic, such as the aspiration on the voiceless stops and the pharyngealization of the voiceless alveolar stop /t^c/, the pharyngeal articulation of this phoneme is very weak, even barely audible at times. Additionally, it is lost when followed by a second consonant in a cluster, as in examples (2.37) and (2.38).

(2.36)	səbyi	['s ^s ɛv.jı]	he baptized him (I.9)
(2.37)	eșțəbyon	[ɛs.ˈt ^s ɜb.jon]	they were baptized (I.10)
(2.38)	mișrā	[ˈmɪs.rɒ]	boundary (IV.20)

Alveolar nasal sonorant: Before a velar stop, /n/ is realized as [ŋ]. This allophone is characteristic of loan words of English and Persian origin, e.g. (2.40) and (2.41). Vowels in syllables closed by /n/ are regularly nasalized; this is generally not indicated in the transcription unless the following nasal is lost (see §2.7.1.6).

(2.39)	nodā	['noː.dɔ]	earthquake (IV.115)
(2.40)	Inglizí	[ıŋ.glı.ˈzi]	English (II.14)
(2.41)	čangā́	[tʃæŋ.ˈɡɒː]	harps (IV.81a; Persian čang-hā id.)

In certain environments, /n/ can be syllabified to [n]; in the phonemic transcription system, syllabic [n] is represented by the digraph *an*. For more information and further examples, see §2.4.2.5.

Alveolar lateral approximant: The Neo-Mandaic /l/ generally has a "clear" (i.e. alveolar) pronunciation, except in two environments: as the second segment in a consonant cluster, following a voiceless stop, it is devoiced, as in (2.43), and in the Arabic word *Allāh* (and consequently all names containing it) it is velarized to [1], as in (2.44).

(2.42)	elliye	[ˈɛl.li.je]	where is it (X.14)
(2.43)	klātā	[ˈkl̥ɔː.θɔ]	three (III.1)
(2.44)	Abdalla	[Sab.'dat.ta]	Abdallah (II.8)

In certain environments, /l/ can be syllabified to [1]; in the phonemic transcription system, syllabic [1] is represented by the digraph ∂l . For examples and more information, see §2.4.2.5.

Voiced alveolar trill: The Neo-Mandaic alveolar trill /r/ has two different realizations, depending upon its position within the syllable. In word-initial position, e.g. (2.45), syllable-initial position after another consonant, e.g. (2.46), when doubled, e.g. (2.47), and in intervocalic position, e.g. (2.48), /r/ is realized as a voiced alveolar trill [r]. In word-internal, syllable-final, and word-final positions, e.g. (2.49) and (2.50), it is realized as a voiced alveolar approximant [I].

(2.45)	rabbā	[ˈræb.bɔ]	great (X.11)
(2.46)	gabrā	[ˈgæv.rɔ]	man (V.18)
(2.47)	barra	['bar.ra]	outside (II.13)
(2.48)	qāri	['qɔ:.ri]	by him (IV.12)
(2.49)	karsi	[ˈkaɪ.si]	his stomach (X.2)
(2.50)	əmar	[ə.ˈma』]	he said (V.14)

In certain environments, /r/ can also appear as a syllabic consonant [r]; in the phonemic transcription system, it is represented by the digraph σr . See §2.4.2.5 for details.

2.3.1.5 Post-Alveolars

Š	/ʃ/	[∫]
č	/tʃ/	[tʃ]
j	/œ/	[ʤ]

Voiceless post-alveolar sibilant: This phoneme has no significant allophonic variation.

(2.51) $e \check{s} m i$ [' $\epsilon \int m i$] h	nis name	(11.7)
---	----------	--------

2. Phonology

Voiceless post-alveolar affricate: the voiceless post-alveolar affricate /tf/ is primarily, but not exclusively, limited to loan words from the languages with which speakers of Neo-Mandaic have come into contact; e.g. (2.52) and (2.53). It is also found in the number nine, (2.54), and all numerals derived from it, e.g. (2.55) and (2.59).

(2.52)	čāre	[ˈtʃɒː.rɛ]	remedy (IV.40; Persian čāra id.)
(2.53)	gač	[gæt∫]	gypsum (IV.4; Persian gač id.)
(2.54)	eččā	[ˈɛtʃ.tʃɔ]	nine (II.22; CM tša id.)
(2.55)	čin	[tʃin]	ninety (III.1; CM tšin id.)
(2.56)	čemmā	[['] t∫3m.mວ]	nine hundred (III.1; CM tšima id.)

Voiced post-alveolar affricate: Like its voiceless counterpart /tʃ/, the few examples of /dʒ/ collected in the texts derive predominantly from Arabic and Persian; see (2.57), (2.58), and (2.59).

(2.57)	jihel	[ˈʤiː.hɛl]	child (II.16; Arabic <i>jāhil</i> 'ingénu')
(2.58)	minjo-di	[mɪn.ˈʤoː.di]	out of (X.3b; Arabic <i>min juwwa</i> id.)
(2.59)	mujur	[mʌ.ˈʤuɹ]	how (V.24; Persian če-jur 'how')

2.3.1.6 Palatal

V	/i/	[j], [ɪ], [i]
2	' J '	LJJ7 L-J7 L-J

Palatal approximant: /j/ has a tendency to form a diphthong with the preceding vowel when closing a syllable, as in (2.61).

(2.60)	yəhom	[jə.ˈhom]	sit down! (IV.140)
(2.61)	ayhimi	[aɪ.ˈhiː.mi]	he made him sit down (IV.77)

In certain environments, /j/ may become syllabic, in which case it merges with the close front vowel /i/. For examples and further details, see 2.4.2.5 below.

2.3.1.7 Velars

k	/k/	[k ^h], [k]
g	/g/	[g]

Voiceless velar stop: /k/ is lightly aspirated in most positions, except when the first element in a consonantal cluster, as in (2.62) or

following a fricative or sibilant (2.63). Nevertheless, aspiration is not generally indicated in the transcriptions, apart from those given below.

(2.61)	kandi	[ˈkʰæn.di]	yet, still (IV.78)
(2.62)	kla <u>t</u> āyā	[kl̥ɛ.ˈθɔː.jɔ]	third (VIII.9)
(2.63)	askar	[ˈæs.kaɪ]	army (IV.103)

Voiced velar stop. This phoneme has no significant allophonic variation.

	(2.64)	gomlānķon	[gom.'lon.xon]	your camels (IV.124)
--	--------	-----------	----------------	----------------------

2.3.1.8 Uvulars

q	/q/	[d]' [R]' [B]
\underline{k} / x	$\chi/$	[χ]
\bar{g}/\dot{g}	$\langle \mathbf{R} \rangle$	[R]

Voiceless uvular stop: When this phoneme closes an accented syllable, it is often realized as a voiced uvular fricative / μ /, as in (2.66); this appears to be the result of Persian influence and not to be related to the other examples of fricativization listed in §2.7.1.7. /q/ also occasionally alternates with the voiced velar stop /g/, particularly in word-initial position, e.g. (2.67), possibly as a result of colloquial Arabic influence. Neighboring vocalic segments are regularly colored by the constriction of the uvula; this is not indicated in the transcriptions.

(2.65)	qazģān	[qaz.'ʁən]	cooking pot (X.2)
(2.66)	láqbèn	['laĸ.bɛn]	we don't want (IV.109)
(2.67)	qəhazelli	[ga. ['] zɛl.li]	they see him (IV.7)

Voiceless uvular fricative: The near-minimal pair (2.68) and (2.69) indicates the phonemic independence of χ from /k/. Neighboring vocalic segments are regularly colored by the constriction of the uvula; this is not indicated in the transcriptions.

(2.68)	yek	[jɛk]	one (IV.13)
(2.69)	šiex	[∫iĕχ]	sheikh (II.7)

2. Phonology

Voiced uvular fricative: /B/ is a comparatively rare phoneme that is more characteristic of Arabic and Persian loan words than reflexes of the Classical Mandaic lexicon. Neighboring vocalic segments are regularly colored by the constriction of the uvula, although this is not indicated in the transcriptions.

(2.70) $b = \bar{g} a \bar{s}$ [bə.'kus[§]] he stopped, stood still (IV.50)

2.3.1.9 Pharyngeals

ķ	/ħ/	[ħ], [h]
C	///	[ʕ], [ʔ], [Ø]

Voiceless pharyngeal fricative: The voiceless pharyngeal fricative /ħ/ is characteristic of loan words from Arabic. Neighboring vocalic segments are regularly colored by the constriction of the pharynx, although this is not indicated in the transcriptions. On occasion, this phoneme is pronounced as a voiceless glottal fricative /h/, possibly as a result of influence from Persian, e.g. (2.72). When asked about the different outcomes of this phoneme, the consultant responded that sometimes Mandaeans pronounce words containing it "the Arab way," and other times they pronounce them "the Persian way."

(2.71)	<u> ḥā</u> kem	[ˈħɒː.kɛm]	governor (IV.30; Arabic <i>hākim</i> 'judge')
(2.72)	<u> </u> hašiš	[hæ.ˈ∫i∫]	hashish (VII.6; Arabic <i>hašīš</i> 'grass')

Voiced pharyngeal fricative: Like /ħ/, the voiced pharyngeal fricative /f/ is also typically found in loan words of Arabic origin. Neighboring vocalic segments are regularly colored by the constriction of the pharynx, and Arabic loan words containing this phoneme are likely to have a "Persian" pronunciation as well as an "Arabic" one, in which case the voiced pharyngeal fricative is realized as a glottal stop /?/, as in (2.74), or even deleted, as in (2.75).¹²¹ The consultant opted not to exclude either pronunciation and insisted that the alternation between the two allophones was in no way an impediment to comprehension.

¹²¹ Note that the glottal stop /?/ appears only in Arabic loan words containing /?/ or as an allophone of /?/.

(2.73)	șun'a	['s [°] An. [°] a]	good (VI.2; Arabic <i>sun</i> "benefit")
(2.74)	jami'a	[&e.'mi:.?a]	all (IV.15; Arabic <i>jamī</i> ^c id.)
(2.75)	šam'	[∫am]	candle (IV.20; Arabic šam 'id.)

2.3.1.10 Glottal

h /h/ [h], [x], [Ø]

Voiceless glottal fricative: In word-final position, /h/ is regularly deleted; cf. (2.76) and (2.77), as well as (2.78) and (2.79). When closing a syllable, however, it is often strengthened to [x] before a consonant; e.g. (2.80). See §2.4.1.3 for more information on the treatment of /h/.

(2.76)	ərqihā	[ŗ.ˈqiː.hɔ]	sky (augmented) (IV.145)
(2.77)	ərqi	[ŗ.'qi:]	sky (contextual) (IX.12)
(2.78)	pərāhi	[pɛ.ˈrɔː.hi]	some money (VI.1; cf. Turkish para id.)
(2.79)	pərā	[pɛ.ˈrɔː]	money (V.V)
(2.80)	qəhazin <u>k</u> on	[qax.'zin.xon]	I see you (pl.) $(I.4)^{122}$

2.3.2 Vowels

The vowel system in Neo-Mandaic is composed of seven distinct vowels, of which six (*i*/i/, *u*/u/, *e*/e/, *o*/o/, and \bar{a} /a/, and /ɔ/) are principal phonemes, and one (∂ / ∂ /) is marginal. There are also five diphthongs, *ey*/εI/, *ay*/aI/, *aw*/au/, $\bar{a}y$ /oI/, and $\bar{a}w$ /oU/; these will be discussed in §2.4.2.3 below. The vowels are distinguished by quality rather than quantity.

As with the consonants, there is a considerable amount of allophonic variation among the vowels. In most cases, the allophonic realization of any given vowel can be determined by the syllable structure and the position of the stress. Three of the principle vowels, the "tense" vowels *i*, *u*, and \bar{a} , are frequently lengthened in open accented syllables, to [i:], [u:], and [o:] or [p:]. *i* and *u* are generally realized as [I] and [Λ] whenever they occur in closed syllables, either accented or unaccented. The other three principle vowels, the "lax"

vowels *o*, *e*, and *a* appear only exceptionally in open accented syllables. *e* is realized as [e] in open syllables and $[\varepsilon]$ in closed syllables. *a* is realized as [a] in closed accented syllables, and [a] or [æ] elsewhere.

Schwa (∂) has the widest allophonic variation of all the vowels. It is regularly fronted, backed, raised, or lowered in harmony with the vowel of the following syllable. Whenever it is followed by *w*, it is regularly raised and backed to [Λ]. When the accent falls on a syllable containing schwa, it becomes fronted and raised to [ϵ].

<i>i</i> [i] : <i>e</i> [e]	bie <u>t</u> i	[ˈbiĕ.θi]	his house
	bie <u>t</u> e	[ˈbiĕ.θe]	my house
<i>i</i> [ι]: <i>e</i> [ε]	a <u>k</u> lit	['ax.lɪt]	I ate
	a <u>k</u> let	['ax.lɛt]	you ate
и [ʌ] : о [о]	guš əbud	[ˌguʃə. ['] wud]	watch out (f.)!
	guš əbod	[ˌguʃə. ['] wod]	watch out (m.)!
<i>o</i> [o]:ə[ə]	horini	[hoˈriːni]	another
	hərini	[həˈriːni]	we defecated
<i>o</i> [o] : <i>ā</i> [ɔ]	o <u>b</u> di	['ov.di]	do it!
	ā <u>b</u> di	['ɒv.di]	he does it
a [a] : ā [ɔ]	bie <u>t</u> a	[ⁱ biĕ.θa]	her house
	bie <u>t</u> ā	[ⁱ biĕ.θɔ]	house

Table 10: Some Minimal and Near-Minimal Pairs (Vowels)

2.3.2.1 Close front unrounded vowel: /i/ is generally realized as a close front unrounded vowel [i] in open unaccented syllables, e.g. (2.81). In open accented syllables, /i/ is lengthened, e.g. (2.82); In closed syllables, /i/ is realized as [I]; in open pretonic syllables, /i/ is occasionally reduced to [I]; see §2.6.2.2.

(2.81)	bəratti	[bə.ˈrat.ti]	his daughter (IX.7)
(2.82)	diwā	['di:.vɔ]	demon (IV.45)
(2.83)	mimrā	['mɪmrə]	literature (I.5)

/i/ often breaks in open accented syllables, particularly in the environment of labial consonants (including the bilabial and labiodental consonants) and those articulated on either side of the palate (such as the postalveolar and velar consonants). See §2.5.3.5 for examples and further details.

/i/ and the Diphthong /aɪ/: The diphthong /aɪ/ appears in very few words, most of which are derived from foreign languages; see 2.3.1.6.1. It was originally much more common; already in the classical language, however, this diphthong had collapsed to a monophthong in closed syllables, cf. (2.84) and (2.85). The resulting vowel */e/ subsequently merged with /i/. The remaining examples of */aɪ/ in open accented syllables also collapsed and merged with /i/,¹²³ except in a few forms borrowed from the classical language such as (2.86). In unaccented open syllables, /e/ was preserved, e.g. (2.87) and (2.88). The modern reflex of this diphthong contrasts with /i/ in (2.88) and (2.89).

(2.84)	baita	*['ba1.tɔ]	house (emphatic state) ¹²⁴
(2.85)	bit	*[be0]	house (construct state)
(2.86)	ešmehon	[ɛʃ.ˈmeː.hon]	their names (I.1)
(2.87)	hemanu <u>t</u> an	[he.ma.'nu:.0æn]	our faith (I.8)
(2.88)	bie <u>t</u> e	[ˈbiĕ.θe]	my house (V.16)
(2.89)	bie <u>t</u> i	[ˈbiĕ.θi]	his house (IX.4)

2.3.2.2 Close back rounded vowel: /u/ is generally realized as a close back unrounded vowel [u] in open unaccented syllables, e.g. (2.90). In open accented syllables, /u/ is lengthened, e.g. (2.91). In accented and

- 123 There is some evidence that this is a relatively recent change; it does not appear to have occurred in the attested samples of the dialect of Iraq.
- 124 See §3.4 for more information on the states and their reflexes in Neo-Mandaic. Each noun has only one basic form, and those derived from the classical *status emphaticus* (q.v., §3.4.1) lose the augment $-\bar{a}$ whenever they are followed by another word in juncture.

unaccented closed syllables, /u/ is generally realized as [Λ], as in (2.92) and (2.93).

(2.90)	ehlu	['ɛh.lu]	they have (IV.45)
(2.91)	nurā	['nuː.rɔ]	fire (IV.114)
(2.92)	dukkā	[ˈdʌk.kə]	place (I.V)
(2.93)	guțluktā	[gat ^s .'lax.tə]	massacre (I.14)

/u/ and the Diphthong /au/: Both the labial-velar approximant /w/ (see \$2.3.1.1) and the voiced labiodental fricative /v/ (see \$2.3.1.2) form diphthongs with /a/ when they close a syllable containing this vowel. The diphthong /au/ is also found in words of foreign origin. Much like the Classical diphthong */aɪ/, this diphthong was formerly more common, but had already collapsed in closed accented syllables in the classical language; compare (2.94) with (2.95). The outcome merges with /u/, e.g. (2.95); subsequently the remaining examples of */au/ in open accented syllables collapsed and merged with /u/ or /o/, e.g. (2.96) and (2.97).

(2.94)	taura	*['tau.rɔ]	bull
(2.95)	turtā	['tʌtə]	cow (CM turta id.)
(2.96)	yumā	[ˈjuː.mɔ]	day (IV.123; CM iuma id.)
(2.97)	nodā	['noː.dɔ]	quake (IV.115; CM nauda id.)

2.3.2.3 Close-mid front unrounded vowel: /e/ is generally realized in the region of the open-mid front unrounded vowel [ε] in closed syllables, e.g. (2.98), and the close-mid front unrounded vowel [e] in open syllables, e.g. (2.99). /e/ generally does not appear in open accented syllables, except in loan words and certain archaic forms; see (2.86) above. In closed syllables, [ε] tends to freely vary with the fairly close, fairly front unrounded vowel [1], e.g. (2.100) and (2.101).

(2.98)	geš	[gɛʃ]	all (I.3)
(2.99)	luhnāye	[luh.ˈnɔː.je]	he isn't here (IX.10)
(2.100)	šehyānā	[∫ıh.′jɔː.nɔ]	ruler (IV.9)
(2.101)	šehyānā	[∫ɛh.ˈjɔː.nɔ]	ruler (IV.42)

2.3.2.4 Mid central unrounded vowel: $|\vartheta|$ is generally realized as a mid-central unrounded vowel $[\vartheta]$ in open syllables (2.102), and as an open-mid front unrounded vowel $[\varepsilon]$ in closed syllables (2.103). At first glance, $|\vartheta|$ is an unlikely candidate for an independent phoneme, as it is regularly inserted as an anaptyctic vowel to break up initial consonant clusters (see §2.5.3.2) or derives from a reduced lax vowel in a pretonic syllable (see §2.6.2.2). Nonetheless, its appearance is synchronically unpredictable, as demonstrated by the near minimal pair (2.104) and (2.105).

(2.102)	həzinu	[hə.ˈziː.nu]	we saw them (V.9)
(2.103)	nəp̄qat	['nɛf.qat]	she exited (II.15)
(2.104)	smālā	['smɔ:.lɔ]	left
(2.105)	səmāqā	[sə.ˈmɔː.qɔ]	red

/ə/ has a wide variety of allophones, varying with [1] and [ϵ] in a closed syllable (as noted above) or even in a pretonic open syllable, e.g. (2.106).¹²⁵ It is often deleted whenever it appears as the first element in an open, pretonic, word-initial syllable, e.g. (2.107), (2.108), and (2.109) or when it appears in an open syllable immediately preceding an accented syllable, the first element of which is a fricative or approximant, as in (2.110) and (2.111). See §2.4.7 below for other possible outcomes of /ə/.

(2.106)	əmalli	[ɛ.ˈmal.li]	he said to him (X.15)
(2.107)	əmalli	['mal.li]	id. (VI.5)
(2.108)	ənāšā	[ˈnɔ:.∫ɔ]	people (VII.5)
(2.109)	ə <u>b</u> ādā	['wɔ:.dɔ]	deed (IV.137)
(2.110)	həwā	[h ^w ɔː]	he was (III.20)
(2.111)	pəradyon	['prad.jon]	they fled (III.23)

2.3.2.5 Close-mid back rounded vowel: /o/ is generally realized as the close-mid back rounded vowel [o] in all closed syllables, e.g. (2.112) and (2.113), as well as open unaccented syllables, e.g. (2.114). /o/ is

¹²⁵ Malone describes this process as 'promotion without schwa coloring,' ibid., 145.

rare in open accented syllables; when it occurs, it is generally lengthened to [o:]; see (2.115) below.

(2.112)	abdonni	[av. ['] don.ni]	they did it (III.20)
(2.113)	əbadyon	[ə.'vad.jon]	they did (III.21)
(2.114)	do <u>t</u> on	[do.'θon]	come (pl.)! (VII.8)
(2.115)	do <u>t</u> i	[ˈdoː.θi]	come (m.sg.)! (I.5)

2.3.2.6 Open front unrounded vowel: /a/ is normally realized as the open front unrounded vowel [a] in open and closed unaccented syllables, e.g. (2.116) and (2.117). /a/ generally does not occur in open accented syllables, except in loan words and the first person singular and plural independent pronouns, where it is realized as [a:], e.g. (2.118). In closed accented syllables, /a/ is often realized as the open back unrounded vowel [a], e.g. (2.119), unless it is followed by the vowel /ɔ/, in which case it is raised to /æ/; see §2.5.2.2 below. [a] is also the regular allophone in the environment of an uvular consonant, e.g. (2.120), with few exceptions, e.g. (2.121).

(2.116)	le <u>k</u> ta	['lɜχ.ta]	she is not (I.13)
(2.117)	bandir	[bæn.ˈdi.ɪ]	bound (IV.146)
(2.118)	anā	[ˈaː.nɔ]	I (I.4)
(2.119)	əmallu	[ə.ˈmal.lu]	he said to them (X.5)
(2.120)	qalin	[qa.'lin]	I go (IV.64)
(2.121)	qəla <u>k</u> ti	[qɛ.ˈlæɣ.t ^s i]	he seizes him (V.18)

2.3.2.7 Open back rounded vowel: /5/ is regularly realized as [5] in open syllables, e.g. (2.122), although it may be realized as [b] in the same environment, e.g. (2.123). In open accented syllables, the phoneme is lengthened, e.g. (2.122) and (2.123). This phoneme is generally realized as [b] in closed accented syllables, e.g. (2.124), but is often realized as [5] in contextual forms, even when the loss of the augment results in the closing of the preceding syllable, e.g. (2.125). In open pretonic syllables, /5/ is reduced to [a]; see §2.4.4.1.

(2.122)	qəbānen	[qə.ˈbɔː.nɛn]	they build (IV.17)
(2.123)	qənāpeq	[qə.ˈnɒː.fɛq]	he exits (V.8)

2.4 Phonotactics

(2.124)	emrān <u>k</u> on	[ˌɛm.bə.ˈrɒn.χon]	your sheep (IV.124)
(2.125)	barnāš	[bæı.ˈnɔʃ]	person (contextual) (I.11)

2.4 Phonotactics

The list of allophones associated with each of the segments described above is by no means exhaustive. Excluding examples of free variation, the allophones of a given segment in Neo-Mandaic are determined by phonotactic considerations-the structure of the syllable in which they are found $(\S2.4.1)$, the position they occupy within this syllable and within the word of which it is a part (§2.4.2 the segments neighboring syllables. and \$2.4.3). in and suprasegmental features such as the position of the accent within the word and the intonation of the sentence $(\S2.4.4)$.

In addition to the regular allophonic processes described in this section, the phonemes of Neo-Mandaic are often affected by sporadic changes in fast or "Allegro" speech; see §2.7 for a partial inventory of these changes.

2.4.1 Syllable Structure

The lexicon of Neo-Mandaic is composed of words of one, two, three, four, and five syllables.

(2.126)	min	[mɪn]	from (I.13)
(2.127)	mendi	[ˈmɛn.di]	something (I.7)
(2.128)	qanāyā	[qaˈnɔːjɔ]	smith (III.5)
(2.129)	Mušmanānā	[mʌ∫.mɛ.ˈnɔ:.nɔ]	Muslims (IV.150)
(2.130)	əmal p enanni	[ɛ.ˌmal.fɛ.ˈnɑn.ni]	we will teach him (IV.92)

Neo-Mandaic syllables consist of an onset (which is optional in wordinitial syllables) and a rime. The rime consists of a nucleus (usually a vowel or a syllabic consonant) with or without a coda. The onset and the coda which frame the nucleus consist of consonants; the onset is mandatory for all word-internal syllables, but the coda is optional in all environments. The syllable patterns V, VC, CV, and CVC are the most common.

2.	Phonol	ogy

(2.131) ā	[0]	this (contextual) (I.5)
(2.132) <i>a<u>k</u></i>	[αχ]	that (contextual) (II.2)
(2.133) <i>mu</i>	[mu]	what (IV.33)
(2.134) <i>tum</i>	[tum]	then (II.12)

Considerably less common are syllables containing clusters of consonantal or vocalic segments, such as VCC, CCV, CCVC, CVCC, CVVC, and even CVVCC. For information on the consonant or vowel clusters tolerated in syllable-initial or syllable-final environment, consult §2.4.2.5 below.

(2.135)	ahl	[ahl]	family (IV.105)
(2.136)	klātā	[ˈkl̥ɔː.θə]	three (III.1)
(2.137)	șțānye	[ˈstən.je]	he is a boy
(2.138)	waxt	[væχt]	time (IV.18)
(2.139)	bie <u>t</u>	[biĕθ]	house (contextual) (III.11)
(2.140)	šielt <u>k</u> on	[ˈ∫iĕlt.χon]	I asked you (pl.)

2.4.2 Distribution of Non-Syllabic Segments

All consonants can open or close a syllable, with the exception of the labial-velar and palatal approximants /w/ and /j/ (\$2.4.2.1), the voiced labiodental fricative /v/, the labial-velar approximant /w/ (\$2.4.2.2), the voiced pharyngeal fricative /\$/ (\$2.4.2.3), and the voiceless glottal fricative /h/ (\$2.4.2.4).

Certain consonants, including the voiceless labiodental fricative /f/, the voiceless interdental fricative / θ /, the voiceless uvular fricative / χ /, and the voiced uvular fricative / κ / are not generally found in word-initial position in Mandaic words. Words beginning with these segments are typically loan words.

2.4.2.1 Approximants /w/ and /j/ – Both approximants may open a syllable, e.g. (2.141) and (2.142), but tend to form a diphthong with the preceding vowel when closing a syllable, e.g. (2.143) and (2.144).

(2.141) <i>yəmutā</i>	[jə.ˈmuː.θɔ]	oath (IV.120)
(2.142) <i>qamāy</i>	[qa.ˈmɔɪ]	before (IV.41)

(2.143) wuş	le [was ^s .'le]	piece (IX.12)
(2.144) gāw	[gou]	in (I.4)

2.4.2.2 Fricative /v/ and Approximant /w/: The distributions of these two phonemes (and the environments in which they merge) are discussed in sections 2.3.1.1 and 2.3.2.1 above. The enclitic conjunction *w*- is most often realized as a syllabic [u]. For more details on this conjunction, see 2.5.7 below.

2.4.2.3 Voiced Pharyngeal Fricative $/\Omega/$: The closure of a syllable by this phoneme is avoided through the insertion of an epenthetic vowel [a], e.g. (2.145). In word final environment, this epenthetic vowel is lowered and fronted to [a], as in (2.146).

(2.145) ya'ani	[ˈja.ʕa.ni]	meaning (II.2)
(2.146) <i>jamiʿa</i>	[&e.'mi:.?a]	entirety, all (IV.15)

2.4.2.4 Voiceless Glottal Fricative /h/: Neo-Mandaic does not generally tolerate /h/ in syllable-closing or word-final position, and generally deletes it in the latter environment and inserts an anaptyctic / ϑ / in the former environment (see §2.7.1.9 below).

Word-Final /h/: /h/ is regularly deleted in word-final environment; see §2.3.1.10.1 for examples. Word-final /h/ is occasionally preserved in words of foreign origin, e.g. (2.147).

(2.147) *farwāh* [faɪ'wəh] thanks (I.5; origin uncertain)

Syllable-Closing /h/: /h/ is very unstable in this environment. Whenever it closes a word-internal syllable, it may be preserved through the insertion of an anaptyctic /ə/ (see 2.7.1.9) or simply deleted; if the syllable is accented and contains a lax vowel, the onset of the following syllable is geminated to maintain the syllable structure, as in (2.148) and (2.149).

(2.148)	ahni	[ˈan.ni]	they (IV.17)
(2.149)	ehli	[ˈɛl.li]	he has (V.26)

2.4.2.5 Consonant Clusters

Permissible consonant clusters in Neo-Mandaic fall into two categories: clusters that form at the beginning or the end of a syllable, and clusters that span syllable boundaries. The former are strictly limited to certain combination of segments. The latter are less restricted; with few exceptions, Neo-Mandaic tolerates most clusters of two or occasionally even three consonants across a syllable boundary.

Word-Initial Consonant Clusters: Consonant clusters consisting of a stop followed by a sonorant, such as (2.150), a sibilant followed by a sonorant, as in (2.151), or a sibilant followed by a stop, such as (2.152), are tolerated in both syllable-final and syllable-initial environments.

(2.150) klātā	[ˈkl̥ɔː.θɔ]	three (III.1)
(2.151) <i>šlihā</i>	[ˈʃliː.hɔ]	apostle (I.9)
(2.152) <i>stānā</i>	[ˈst ^s ɔː.nɔ]	boy (X.17)

Note that sonorants are not tolerated as the initial segment of a syllable- or word-initial consonant cluster. In this environment, they generally become syllabified, as in (2.153) and (2.154). The approximants [w] and [j] are generally resolved through the insertion of an anaptyctic /ə/ between the first and the second segment of the cluster, but see (2.155) and (2.156).

(2.153)	əmḥade <u>t</u>	[m.ħa.ˈdɛθ]	I speak (I.7)
(2.154)	ərbiāhā	[ŗ.bɪ.ˈjɔː.hɔ]	fourth
(2.155)	u genz ənāš	[u ˈgɛnz ε.ˈnɔʃ]	and many people (I.14)
(2.156)	imā	[ˈiː.mɔ]	he swore (IV.121)

 $\langle \vartheta \rangle$ is regularly inserted as an anaptyctic vowel to break up initial consonant clusters, as in (2.157) and (2.158).

(2.157)	təmānā	[tə.ˈmɔː.nɔ]	eight (II.4)
(2.158)	rəwāhā	[rə.ˈwɔ:.hɒ]	relief (II.34)

Words built upon biradical roots tend to resolve initial consonant clusters through prothesis rather than anaptyxis (see §2.5.3.3). A few

doublets manifest the effects of both strategies, e.g. (2.159) and (2.160), as well as (2.161) and (2.162). Both members of such doublets are typically accented on the final syllable rather than the initial syllable, unlike most other bisyllabic nouns with initial *e*-; cf. (2.163).

(2.159)	hənā	[hə.ˈnɔː]	here (II.21)
(2.160)	əhnā	[ɛh.ˈnɔː]	here (IV.70)
(2.161)	šətā	[∫ə.'tɔː]	now (II.29)
(2.162)	əštā	[ε∫.'tɔː]	now (I.10)
(2.163)	ehdā	['ɛh.dɔ]	one (IV.43; never ** hədā)

Whenever the voiceless uvular stop q- appears after the phoneme /i/, the anaptyctic /ə/ disappears and the cluster is broken with a prothetic /ə/. A glide is then inserted to eliminate the hiatus created by the two vowels in contact, e.g. (2.164). This sound change also affects the perfective of the verb $q \partial r \bar{a} \sim q \partial r i (q \bar{a} r i)$ 'to read,' and even the initial syllable of nouns beginning with the syllable $q \partial$ -, e.g. (2.165) and (2.166).

(2.164)	bāzi əqdar' əlāw ṣṭāni	
	['bɒ:.zi jɛq.dɛ.'rɛ.ləʊ 'st ^s ə:.ni]	A hawk seizes a boy (V.25)
(2.165)	bu <u>t</u> i əqrā	
	['bu:.0i 'jɛq.rɒ]	he read a prayer (IV.19)
(2.166)	ləbašni əqmāšānan	
	[ˈlʷaʃ.nɪ ˌjɛq.mə.ˈʃəː.næn]	we put on our clothes (III.10)

Word-Final Consonant Clusters: Consonant clusters consisting of a sonorant and a stop or a sonorant and a fricative are tolerated in word final environment alone, e.g. (2.153) and (2.154).

(2.167) <i>kalb</i>	[kælb]	dog (contextual)
(2.168) <i>perz</i>	[pɛ.ɪz]	few (contextual) (V.1)

All other word-final clusters are not tolerated; note, however, that clusters in loan words borrowed from Arabic and Persian such as (2.169) and (2.170) are generally not subject to these restraints.

(2.169)	ahl	[ahl]	family (IV.105)
(2.170)	waxt	[væχt]	time (IV.18)

Whenever a sonorant forms the second element of a word-final consonant cluster, the cluster is eliminated by syllabifying the sonorant, as demonstrated in examples (2.171) to (2.176). Note that the coda of the accented syllable is geminated; see §2.4.4.2 for more details about this phenomenon.

(2.171)	eššəm	[ˈεʃ.∫m]	name (contextual) (I.9)
(2.172)	qəmabgeşşən	[qə.mab.ˈʁɜs.sˤņ]	I will stop (contextual) (IV.108)
(2.173)	šekəl-da	[ˈ∫ɛk.kļ.da]	her appearance (II.33b; Arabic šik1 id.)
(2.174)	ohhər	[ˈoh.hṛ]	road (contextual) (III.3)
(2.175)	elli	['ɛl.li]	where (contextual) (IX.11)
(2.176)	rehhu	[ˈrɛh.hu]	wind (contextual)

In the environment of the pharyngeal fricatives /f/ and / \hbar /, anaptyctic / ϑ / is realized as [a], e.g. (2.177) and (2.178). Note syllable-closing /f/ is not tolerated in Neo-Mandaic (see §2.4.2.3 above); as a result, anaptyctic / ϑ / will always intervene between /f/ and a following consonant, as in (2.179), but never between /f/ and a preceding consonant; e.g. (2.180) and (2.181). As noted in §2.4.2.3 above, Neo-Mandaic does not tolerate / ϑ / in word-final position; in this position, / ϑ / is fronted and lowered to /a/.

(2.177)	ba'ad	[ˈba.ʕad]	after (X.2; Arabic <i>ba'd</i> id.)
(2.178)	<u>şāya</u> ḥ	[ˈsˤɒ.jaħ]	scream (V.22; Arabic <i>sayh</i> id.)
(2.179)	baʿaden	[ba.sa.'den]	afterwards (III.10; Arabic ba dayn id.)
(2.180)	sun'a	[ˈsˤʌn.ʕa]	well (VI.2; Arabic <i>sun</i> ' 'benefit')
(2.181)	sun'aye	[s [°] ʌn.ˈŶa.jɛ]	he is well (II.4)

Clusters of Three or More Consonants: No more than two consonants may appear in a word-initial or word-final cluster. Whenever three consonants come into contact in either of these environments, an anaptyctic vowel, /9/, intervenes between the first and the second of the three segments, as in example (2.182), in which the conjunction *w*-and the indicative morpheme *q*- are appended to a verb in the imperfective, or example (2.183), in which the conjunction *w*- and the negative morpheme *lá*- are appended to a noun which begins with a consonant.

2.4 Phonotactics	2.4	Phonotactics
------------------	-----	--------------

(2.182)	w-əqda <u>k</u> arnā	[ˌwɛq.da.ˈɣaɪ.nɔ]	and I remember (II.33)
(2.183)	w-əl-Masihí	[ˌwɛl.mɑ.si.ˈħi]	and not Christian (I.12)

Word-internal clusters of as many as three consonants are permitted, provided that they span a syllable boundary, e.g. (2.184) and (2.185).

(2.184) šieltkon	[ˈ∫iĕlt.χon]	I asked you (pl.)
(2.185) genztar	['genz.tær]	utmost ¹²⁶

The Cluster /m/ + /r/: Neo-Mandaic does not tolerate clusters of the bilabial nasal /m/ and the alveolar trill /r/ in any environment. The voiced bilabial stop /b/ regularly intervenes between these two segments, e.g. (2.186) and (2.187), even across word boundaries, e.g. (2.188) and (2.189).

(2.186)	Moḥamrā	[mo. ^l ħam.bro]	Mohammerah (II.6)
(2.187)	lákamri	[ˈla.kam.bri]	he didn't return it (X.12)
(2.188)	əm-riš yum	[m.ˈbri∫ jum]	from the first day (I.7)
(2.189)	əm-riš šettā	[m.ˈbri∫ ˈʃɛt.tɔ]	from new year's day (IV.3)

Consonant Clusters with /h/: Cluster of the voiceless glottal fricative /h/ with another consonant are also not tolerated, even across a syllable boundary. If /h/ is the first item in the cluster, it is often deleted as mentioned above, in §2.4.2.4; if it is the second element in the cluster, it will also be deleted, and the first element will replace the /h/ as the onset of the following syllable, e.g. the plural morpheme $-h\bar{a}$ becomes $-\bar{a}$ after a consonant.

(2.190) <i>jihel</i>	[ˈʤiː.hɛl]	child (III.14)
(2.191) <i>jihelā</i>	[ˌʤi.hɛ.'lɔː]	children (II.4)

The voiced stops /b/, /g/, and /d/ are fricativized whenever they appear before voiceless stops in syllable-final or word-final clusters. Additionally, the voiceless interdental θ becomes /h/ whenever it is followed by the alveolar lateral approximant /l/. See §2.5.4.1 and §2.5.4.2 below for more details on these two sound changes.

2.4.3 Admissible Vowels

The Neo-Mandaic vowel system consists of six principal vowels and the reduced vowel /ə/; see §2.3.2 for more details. Under the right circumstances, such as those described in §2.3.1.1 and §2.4.2.5, the sonorants /m/, /n/, /r/, /w/, and /j/ can become syllabic as well. All of these may form the nucleus of a syllable.

2.4.3.1 Word Final Vowels

Of the six primary vowels, five may appear in word-final position; e.g. (2.192) to (2.196). The close-mid vowel /o/ never appears in word-final position. /ə/ may not appear in word-final environment; in the event that an anaptyctic vowel is required in word-final position, e.g. to prevent a pharyngeal from closing a final syllable, anaptyctic /ə/ is fronted and lowered to /a/; see §2.4.2.3 and §2.4.2.5 above for examples.

(2.192)	ehli	[ˈɛh.li]	he has (V.26)
(2.193)	ehlu	[ˈɛh.lu]	they have (IV.45)
(2.194)	napša	['næf.∫a]	her personality (II.33)
(2.195)	bie <u>tā</u>	[ˈbiĕ.θɔ]	the house (X.8)
(2.196)	bie <u>t</u> ā	[ˈbiĕ.θe]	my house (V.16)

The syllabic allophones of the sonorants may also appear in word-final position; see §2.4.2.5 above for examples.

2.4.3.2 Word-Initial Vowels

With the exception of /u/, all vowels appear in word-initial position, as in examples (2.197) to (2.202); note that all internal syllables must begin with a consonant, i.e. the onset of the syllable is optional in the initial syllable of the word but not word-internally.

(2.197)	inā	['iː.nɔ]	eye (VI.2)
(2.198)	ellinon	['ɛl.li.non]	where they are (III.22)
(2.199)	əmalli	[ə.ˈmal.li]	he said to him (VI.5)
(2.200)	obrā	['ob.rɔ]	mouse (V.10)

(2.201)	alpā	['æl.fɔ]	one thousand (III.1)
(2.202)	āhā	['ɔ:.hɔ]	this (X.4)

The syllabic allophones of the sonorants may also appear in wordinitial position; see §2.4.2.5 above for examples.

/ə/ is particularly unstable in this position; see \$2.3.2.4 above for further information on its allophones in this environment. It is regularly deleted in propretonic position, and raised and fronted to /e/ in closed initial syllables; see \$2.4.2.5 above and \$2.5.3.3 below for details on prothesis.

2.4.3.3 Diphthongs

Neo-Mandaic has a complement of five diphthongs: $/\epsilon_I/$, $/a_I/$, $/a_U/$, $/\sigma_I/$, and $/\sigma_U/$. Words which contain the diphthong $/\epsilon_I/$ are most often loan words, but this diphthong does form whenever /e/ and the palatal approximant /j/ come into contact, e.g. (2.203) and (2.204).

(2.203)	qabeyyen	[qa.ˈbɛɪ.jɛn]	they want (II.6)
(2.204)	heyyi	[ˈhɛɪ.ji]	life (I.9)

The diphthong /ai/ forms whenever the palatal approximant /j/ closes a syllable containing /a/; see \$2.3.1.6 above. Likewise, the CM diphthong */au/ became /u/ in closed syllables and /o/ or /u/ in open syllables (see \$2.3.2.2 above). The same diphthong regularly forms from the combination of the vowel /a/ and the labial-velar approximant /w/ or voiced labiodental fricative /v/ in syllable-closing position; see \$2.3.1.1 and \$2.3.1.2 above, respectively.

The collapse of these diphthongs is demonstrated synchronically by the paradigms of the verbs in which the third radical consists of a labial-dental or palatal approximant. Following the example of the third masculine singular imperfective, the first common singular imperfective, and the masculine singular imperative forms of the strong verb, which are vocalized CāCeC, CaCeCnā, and CəCoC, respectively, e.g. (2.205), (2.207), and (2.209), one would expect the diphthongs [ɛ1] and [o1] in the second syllable of the analogous forms in verbs with final weak radicals. Instead, the expected diphthongs have collapsed to [i:], e.g. (2.206), (2.208), and (2.210). 2. Phonology

(2.205)	qəgāțel	[qə.ˈgɒː.t [°] ɛl]	he will kill
(2.206)	qədāri	[qə.ˈdɔː.ri]	he will take
(2.207)	qəgațelnā	[qə.ga.ˈt [°] ɛl.nɔ]	I will kill
(2.208)	qədarinā	[qə.dɛ.ˈriː.nɔ]	I will take ¹²⁷
(2.209)	gəțol	[gə.ˈt [°] ol]	kill (m. sg.)!
(2.210)	dəri	[də.ˈriː]	take (m. sg.)!

The final two diphthongs, /3i/ and /3u/, are retained in open accented syllables, as examples (2.211) and (2.212) attest:

(2.211) qamāy	[qa.ˈmɔɪ]	before (IV.93)
(2.212) gāw	[ˈɡɔʊ]	in (I.10)

2.4.3.4 Hiatus

Hiatus (the clustering of syllabic segments) is not uncommon in Neo-Mandaic, particularly as the result of breaking (see §2.5.3.5 for a description of this phenomenon). Generally, hiatus is eliminated through two strategies in Neo-Mandaic: either gliding the initial vocalic segment, as in (2.213), or inserting a glide between the two segments in hiatus, as in (2.214).

(2.213)	bie <u>twātk</u> on	[b ^j εθ'wɔθχon]	your houses (IV.123)
(2.214)	bie <u>t</u>	[bɪ.ˈjɛθ]	house (contextual) (III.12)

Hiatus is also occasionally the result of the loss of intervening nonsyllabic segments, in words such as $\bar{s} \partial p \bar{i} \bar{a} h \bar{a}$ 'evil spirit,' a back formation from CM **šibiahia** 'planets;' cf. Syriac 's $\partial p \bar{i} a p \bar{a} p \bar{$

¹²⁷ The vowel ϵ / in the first syllable is the result of umlaut; see §2.4.3.5 below.

¹²⁸ The Mandaeans consider this word to be a portmanteau composed of *šubbā* 'seven' and *āhā* 'brother;' Macuch, *Handbook*, 98 considers the /h/ to be the reflex of the original */j/ attested in the Syriac cognate (possibly arising through dissimilation with the preceding /i/), and compares it to modern *arbiāyā* [rbi¹2:jo] 'fourth' (cf. Syriac *rəbī'āyā*), which can also be pronounced *arbiāhā* [rbi¹2:ho].

2.4 Phonotactics

(2.215)	šə <u>b</u> iāhā	[∫ı.ˈvʲɔː.hɔ]	evil spirit (IV.13)
(2.216)	šəbiāhā	[∫1.vi.'jɔ:.hɔ]	id. (IV.45)

A homorganic glide regularly develops after the high vowels /i/ and /u/ in hiatus, even across a word boundary; see §2.4.2.5 above for examples. This also occurs whenever the phonemes /f/ and /h/ (q.v.) are deleted in an intervocalic position, e.g. (2.217), (2.218), and (2.219).

(2.217) <i>səbuʿa</i>	[sə.ˈbuː.wa]	week (VIII.9)
(2.218) jamiʿa-dəkon	[&æ.mi.'ja.də.xon]	all of you (IV.124)
(2.219) <i>jihel</i>	[ˈʤiː.jɛl]	child (II.1)

2.4.4 Suprasegmental Features

Suprasegmental features in Neo-Mandaic include vowel length (§2.4.4.1), consonant gemination (§2.4.4.2), word stress (§2.4.4.3), and sentence intonation (§2.4.4.4). With the exception of word stress, none of these suprasegmental features are phonemic.

2.4.4.1 Vowel Length

Vowel length is entirely predictable in Neo-Mandaic, and depends entirely upon the placement of the accent and the syllable structure. As noted above ($\S2.3.2$), vowels occurring in open accented syllables are invariably long, i.e. [i:], [e:], [u:], [o:], and [o:], and those in open pretonic syllables are invariably short, i.e. [1], [ε], [0], [υ], [a], and [ϑ]. See $\S2.6.2.2$ for further details on pretonic reduction.

2.4.4.2 Consonant Gemination

Consonant gemination in Neo-Mandaic occurs in two forms. The first type of gemination, "phonological" gemination, affects consonants following an accented lax vowel. It may occur only in this environment, and the consonants thus affected become simplified whenever the accent moves from the lax vowel (see §2.5.6 below). The gemination of the second radical in the *D*-stem of the verb belongs to this category (see §5.1.1.3). It does not typically occur in loan words. The other type, "lexical" gemination, comprises a large

2. Phonology

number of geminated consonants inherited from earlier stages of the language, borrowed from other languages, or resulting from the total assimilation of one consonant to another. Lexical geminates are much more durable than phonological ones (i.e. less subject to simplification) and may occur in any word-internal environment.

As noted above in §2.4.1, all word-internal syllables must have an onset. Furthermore, the vowels /e/ and /o/ are rare in open accented syllables (save where they are the outcome of a collapsed diphthong), as is the vowel /a/ (which often merges with /ɔ/ in this environment); consult §2.3.2 for further information. Whenever a vowel is added to a closed accented syllable containing one of these three vowels, the coda of the accented syllable is geminated to provide the following syllable with an onset. The geminated consonant thus straddles the syllable boundary, e.g. (2.220), (2.221) and (2.222). If the second syllable is deleted, or the accent shifts from the first syllable, the gemination will be lost; see §2.6.2 for further details.

(2.220)	čemmā	[ˈtʃɜm.mɔ]	nine hundred (III.1; CM tšima id.)
(2.221)	zammā	[ˈzæm.mɔ]	blood ¹²⁹
(2.222)	bellā	[ˈbɛl.lɔ]	husband

Phonological gemination most often occurs whenever an inflectional morpheme (e.g. a pronominal suffix) is added to a closed accented syllable containing a lax vowel. The coda of the closed accented syllable is geminated, leaving the syllable structure intact, e.g. (2.223) becomes (2.224).

(2.223) <i>həzon</i>	[hə.ˈzon]	they saw (X.3)
(2.224) həzonna	[hə.ˈzon.na]	they saw her (IV.142)

With one exception, all Neo-Mandaic consonants can be geminated in this manner. The exception is the voiceless interdental fricative $/\theta/$. Whenever $/\theta/$ closes an accented syllable to which a vowel is added, the outcome is the cluster [χ t] rather than the expected [$\theta\theta$]. For example, when the pronominal suffixes are appended directly to the

¹²⁹ Cf. the more colloquial *dəmā* [də.¹mp:] 'blood;' both forms appear in CM: **dma** and **zma**.

copula *et [$\epsilon\theta$] (CM 'it), it regularly takes the form ekt- [$\epsilon\chi$ t], e.g. (2.225).

(2.225) *ektak* ['ɛɣ.taɣ] you (m.sg.) are (IV.51)

This sound change affects the conjugation of the verb $met \sim mot$ (*māyet*) to die; cf. (2.226) and (2.227) with (2.228). It is also responsible for the modern form of the abstract ending *-uktā* (CM - **uta**), e.g. (2.229). Additionally, it is attested in a rare variant of the word for cow, (2.230).

(2.226) <i>met</i>	$[m\epsilon\theta]$	he is dead (X.15)
(2.227) <i>mie<u>t</u></i>	[miĕθ]	I will die (contextual) (IV.84)
(2.228) <i>me<u>k</u>tat</i>	[ˈmɛɣ.tat]	she is dead (II.34)
(2.229) guțluktā	[gat.'lax.tə]	massacre (I.14)
(2.230) <i>ture<u>k</u>tā</i>	[tu.ˈrɛɣ.tɔ]	\cos^{130}

Examples of the sequence $[\theta\theta]$ do occur in Neo-Mandaic. Such examples are either lexical or result from the assimilation of certain consonants (such as dentals and, in an earlier stage of the language, nasals) to a following interdental fricative / θ /, e.g. (2.231) from CM '**nta** (cf. Syriac $\leftarrow attatat$ 'woman') and (2.232) from the verb *haddet* \sim *haddet* (*amhaddet*) 'to speak.'

(2.231)	e <u>ttā</u>	['ɛθ.θɔ]	woman (II.24)
(2.232)	qəm <u>hatt</u> en	[qm. haθ.θεn]	they speak (II.21)

2.4.4.3 Word Stress

As noted above, the lax vowels /e/ and /o/ only occur in open accented syllables as the reflex of a Classical Mandaic diphthong, the lax vowel /a/ generally merges with /o/ in this environment, and the semi-vowel / ϑ / is never found in this position.¹³¹ The accent generally falls upon a

¹³⁰ This form is derived from CM **turita**; cf. DC 46 **turikta**. The more usual form of this word is *turtā* ['tʌi.tɔ]. E.S. Drower and R. Macuch, A Mandaic Dictionary (Oxford: Clarendon, 1963), s.v. **tura**. For a translation of DC 46, see E.S. Drower, "A Mandaean Book of Black Magic." Journal of the Royal Asiatic Society (1943): 149–218.

¹³¹ An apparent exception to this rule is found at X.7, *qamdəkon* [qam.'do.Xõ] 'for you (pl.),' in which the word takes primary sentence stress, but in other

tense vowel, either /i/, /u/ or /ɔ/, and preferably upon a closed syllable. The placement of the accent is determined from the final syllable; if this syllable is not suitable, the accent will move towards the initial syllable until it comes to rest upon a suitable syllable. Effectively, nearly all words in Neo-Mandaic are accented upon the ultimate, penultimate, or antepenultimate syllable.

Ultimate Stress: Any final syllable (or ultima) that is closed and contains a long vowel automatically receives the accent, e.g. (2.233) and (2.234).

(2.233)	farwāh	[fa'woh]	thanks (I.5)
(2.234)	qəmahrə <u>b</u> āt	[qə.ˌmæh.rɛ.ˈwət]	you destroy (IV.58)

Likewise, whenever a word contains no tense vowels, the stress falls on the final syllable if there are no suitable preceding syllables, e.g. (2.235) and (2.236).

(2.235)	ə <u>k</u> al	[a.'xal]	he ate (V.11)
(2.236)	həzā	[hə.ˈzɔː]	he saw (VIII.3)

Penultimate Stress: In words of two syllables, if the final syllable is open or contains a lax vowel, then the accent will fall upon the penultimate syllable, if it contains a tense vowel or is closed, e.g. (2.237), (2.238), and (2.239).

(2.237)	gabrā	[ˈgæv.rɔ]	man (V.18)
(2.238)	bā <u>b</u> ā	['bɔː.wɔ]	father (I.8)
(2.239)	qallā	['qal.lɔ]	she goes (IV.102)

Antepenultimate Stress: In words of three or more syllables, if neither the ultima nor the penultima is suitable for receiving the accent, then it recedes to the antepenultimate syllable, if that syllable is closed or contains a tense vowel.

(2.240)	gațelnā <u>k</u> on	[ga. ^{'t^sɛl.nɒ._'χon]}	I will kill you (IV.93)
---------	---------------------	--	-------------------------

contexts this vowel does not take the word stress, e.g. *gešdəkon* ['gεʃ.dŏ.,χon] 'all of you.'
Under certain conditions and within certain environments, however, the stress is not predictable from the syllable structure:

Independent Pronouns: Pronouns comprising more than one syllable always take stress on the initial syllable.

1 sg.	anā	['a:.nɔ]	Ι
2 nd sg.	āt	[ɔt]	you (sg.)
3 rd m.sg.	huwi	['huː.wi]	he
3 rd f. sg.	hidā	['hi:.dɔ]	she
1 st pl.	ani	[ˈaː.ni]	we
2 nd pl.	atton	['at.ton]	you (pl.)
3 rd pl.	honni	['hon.ni]	they

Table 11: Independent Pronouns and Stress

Note that the first singular and plural personal pronouns receive stress on the penultima, despite the presence of a tense vowel in the ultima, giving rise to the attested, anomalous forms, instead of the expected forms (2.241) and (2.242).

(2.241) ** <i>ənā</i>	[ə.ˈnɔː]	Ι
(2.242) ** <i>əni</i>	[ə.ˈniː]	we

Contextual Forms: Apocopated or "contextual" forms preserve the stress in the same position as their augmented or "lexical" counterparts. In the paradigm of the verb in the derived stems, this has given rise to a phonemic contrast between the third plural form of the imperfective (2.243) and the apocopated form of the first plural imperfective (2.244), from (2.245).

(2.243)	qəmțallen	[qm.'t ^s al.len]	they play
(2.244)	qəmțallén	[qm.t ^s al.'lɛn]	we play (contextual) (II.4)
(2.245)	qəmtallenni	[qm.t ^s al.'lɛn.ni]	we play (lexical)

Loan Words: Loan words from other languages follow the rules of their language of origin governing stress. For this reason, non-nativized Persian loan words are generally stressed on the final syllable, e.g. (2.246). Likewise, in the phrasal verb construction, the

non-verbal element of the construction takes primary stress, and the verb retains secondary stress, e.g. (2.247).

(2.246)	komák	[ko.'mak]	help (III.2)
(2.247)	komák ə <u>b</u> adyon		
	[ko.'mak ə. ₁ vad.je	on]	they helped (III.2)

Enclitic Suffixes: Enclitic suffixes, including those introduced by the enclitic morpheme *-d-*, do not cause the accent to shift, e.g. (2.248) and (2.249).

(2.248)	bienā-dan	[ˈbiĕ.nɔ.dan]	among us (IV.81)
(2.249)	gešdə <u>k</u> on	[ˈgɛ∫.də.χon]	all of you (IV.124)

Accented Morphemes: Several morphemes take primary stress whenever they are affixed to a word. These include the derivational suffix -i, e.g. (2.250). In Mandaic, this morpheme is identical with the Persian abstraction morpheme,¹³² which makes abstractions out of substantives. This morpheme appears primarily on loan words from Persian, e.g. (2.251), which contrasts with (2.252). Another morpheme which takes primary stress is the plural morpheme $-h\bar{a}$, which appears on loan words, e.g. (2.253) and (2.254).

(2.250) <i>mandāyí</i>	[ˌmæn.dɔ.ˈjiː]	Mandaic (II.15)
(2.251) <i>nəjārí</i>	[nə.ˌʤɒ.ˈriː]	carpentry
(2.252) <i>nəjāri</i>	[nə.ˈdʒɒː.ri]	a carpenter ¹³³
(2.253) <i>jihelā</i>	[₁]dzi.hɛ. ¹ lɔ:]	children (II.4)
(2.254) xiyabānhā́	[<code>xi.ja.,bon.'hp:]</code>	avenues (III.22)

Negation: Much like the negative morpheme *na*- in Persian, the Neo-Mandaic negative morpheme $l\dot{a}$ - is prefixed to the verb that it modifies, and takes the primary stress of the word, e.g. (2.255), which becomes (2.256) when negated.

- 132 Wolfhart Heinrichs suggests another intriguing possibility for *Mandāyí*. If we assume that the final -*i* in this word is, in fact, the abstraction suffix rather than the adjectival suffix, then the term would mean 'Mandaism,' rather than 'Mandaic,' in parallel with the term *Hulawlá* (< **Hūdāyūțā* 'Judaism'), the name of the Jewish Neo-Aramaic dialect of Sanandaj (and other local dialects).
- 133 Note that the enclitic indefinite morpheme -i does not take the accent.

2.5 Segmentable Morphophonology

(2.255)	qəm <u>hatt</u> en	[qm.ˈħaθ.θɛn]	they talk (II.21)
(2.256)	láqəm <u>hatt</u> en	[ˈla.qm.ˌħaθ.θɛn]	they don't talk (II.21)

2.4.4.4 Sentence Intonation

Within the course of a sentence, the stress on individual words may become so weak as to be barely audible. The primary sentence stress, which takes the form of a high rising pitch, generally falls upon the last stressed syllable of the last word of the sentence (the word which takes primary sentence stress is indicated by an asterisk (*):

(2.257) Qabeyyen ābden qābin gāw Moḥamrā.
[qa.'bɛi.jɛn 'bv.dɛn qp.'vin gou *mo.'ħam.bro] They want to hold a wedding in Khorramshahr. (II.6)

When a relative clause precedes the main verb of a phrase, it is delivered in a monotone, with a high rising pitch on the final syllable of the clause:

(2.258) Gobrā ke gāw metelā Mandāyí abdonni.
['gov.rɔ ke gou mɛ.tɛ.lɔ mæn.dɔ.'jí: *av.'don.ni] In the stories, the men who made it were Mandaeans. [III.20]

Relative clauses following the main verb receive normal intonation.

2.5 Segmental Morphophonology

As Neo-Mandaic morphemes are combined, allomorphs arise as they are modified by neighboring segments. This influence of these neighboring segments may be divided into several broad categories, including assimilatory processes (§2.5.1), dissimilatory processes (§2.5.2), insertion (§2.5.3), lenition (§2.5.4), loss (§2.5.5), and geminate simplification (§2.5.6). In each category, sound changes affecting consonants will be listed first, followed by sound changes affecting vowels.

2.5.1 Assimilatory Processes

Both consonantal segments and vocalic segments may assimilate to neighboring segments. The assimilation of the former is illustrated in

2. Phonology

§2.5.1.1. Vocalic segments may either assimilate to other vocalic segments in neighboring syllables (§2.5.1.2) or become "colored" by consonantal segments in their environment (§2.5.1.3).

2.5.1.1 Assimilation

All of the examples of assimilation between consonantal segments involve the complete assimilation of one segment to the following segment (total anticipatory contact assimilation).

 $d > \theta / _\theta$: /d/ regularly assimilates whenever it comes into contact with / θ / in the paradigm of the verb *haddet* ~ *haddet* (*amhaddet*) 'to speak;' see §2.4.4.2 above. An apparent exception to this otherwise regular sound rule is (2.261), which occurred only once, in the context of the speech that Nasser Sobbi delivered at Harvard University; the regular form is given in (2.260). This may well be an example of metanalysis on the model of the forms in which the /d/ is preserved.

(2.259)	əm <u>h</u> ade <u>t</u>	[m.ħa. 'dεθ]	I speak (subjunctive) (I.7)
(2.260)	əmḥa <u>tt</u> etton	[m.ħaθ.'θɛt.ton]	you speak (subjunctive)
(2.261)	əmḥadṯetton	[m.ħat.'θet.ton]	you speak (subjunctive) (I.4)

 $n > 1 / _l: /n/$ regularly assimilates to /l/ in the third plural form of the paradigm of the imperfective with object suffixes; e.g. (2.263), from (2.262) with the third person object suffix.

(2.262)	qāmren	['qpm.ren]	they say (V.1)
(2.263)	qamrelli	[qam.ˈrɛl.li]	they will tell him (II.11)

In other environments, the regular assimilation of /n/ to a following consonant is extremely rare, though not unattested; most examples are inherited from earlier stages of Aramaic, e.g. (2.265), the feminine of (2.264).

(2.264)	horinā	[hoˈ.ri.nɔ]	other (m.) (VII.8)
(2.265)	horettā	[ho.ˈrɛt.tɔ]	other (f.) (III.2)

 $r > C / _C_{[+alveolar][+sonorant]}$: /r/ regularly assimilates to the following alveolar sonorant in the paradigm of the verb *smar* ~ *smor* (*āmer*) 'to say', e.g. (2.267), particularly before the object marker *l*-, e.g. (2.268),

but note (2.270) in place of the expected (2.269), which is likely a metanalysis of the sort seen in example (2.261) above, which also occurred in the same text (a speech delivered at Harvard University).

(2.267)	rāst qamennā	[rɒst qa.ˈmɛn.nɔ]	I speak the truth (V.15)
(2.268)	əmalli	[ə.ˈmal.li]	he said to him (X.15)
(2.269)	mallonni	[mal.'lon.ni]	they said to him (IV.16)
(2.270)	əmarlonnan	[ə.mai.'lon.næn]	they said to us (I.5)

2.5.1.2 Umlaut

The term "umlaut" here refers to a number of different sound changes which result in the assimilation of a class of vowels to vocalic segments in a neighboring syllable.

Whenever an open syllable containing the lax vowel /a/ is followed by an accented syllable containing the tense vowel /i/, the first vowel is raised and fronted to [ϵ]. This sound change is regular throughout the Mandaic lexicon, even among loan words, e.g. (2.271), but is most commonly encountered in the paradigm of the verbs with third weak radicals; e.g. (2.272), (2.273), and (2.274). When the accent falls upon the second syllable of the stem, the underlying diphthong collapses to /i/. This then causes the vowel of the initial syllable to raise and front to [α] in a number of forms, including the first common singular and the third masculine singular forms of the imperfective with object suffixes.

(2.271)	wazirā́	[vɛ.zi.ˈrɔː]	advisor (IV.11; Persian vazir-hā id.)
(2.272)	a <u>titon</u>	[ɛ.ˈθiː.ton]	you came (I.3)
(2.273)	hazin	[hɛ. ['] zin]	I see (IX.3)
(2.274)	láqdari	['laq.dɛ.ri]	he doesn't take it (V.26)

Likewise, the vowel of the proclitic negative morpheme, $l\dot{a}$, is regularly raised to [æ] before an accented syllable containing /i/, as in (2.275).

(2.275)	lášiyelni	[ˈlɛ.∫i. jɛl.ni]	we didn't ask

Umlaut does not occur when the initial syllable begins with an uvular or pharyngeal consonant, e.g. (2.276) and (2.277).

2. Phonology

(2.276)	qatinā	[qa.'θi:.nɔ]	I will come (IV.41)
(2.277)	qabin	[qa.'bin]	I want (I.3)

A similar change affects /a/ in pretonic syllables; see §2.7.2.2 in the allegro rules below. Umlaut is not indicated in the transcription.

2.5.1.3 Vowel Coloring

In accented syllables closed by the voiced alveolar approximate $[], [\varepsilon]$ is lowered to the open central unrounded vowel [a], e.g. (2.278) and (2.279).

(2.278)	qəmbašqart	[qm.ba∫.'qa.t]	you know (I.6)
(2.279)	əmzaharla	[m.za.ˈhaɪ.la]	may he protect her (II.35)

2.5.2 Dissimilatory Processes

Dissimilatory processes affect both consonantal segments (§2.5.2.1) and vocalic segments (§2.5.2.2 and §2.5.2.3).

j > h / 'jɔ:__ɔ: Whenever /jɔ/ follows a stressed syllable /jɔ/, it becomes /hɔ/, thus differentiating it from the preceding syllable. This sound change is apparent in word *šəbiāhā* 'evil spirit,' a back formation from CM **šibiahia** 'planets;' cf. Syriac $\langle x = x , \bar{x} , \bar{y} , \bar{y} \rangle$ 'the seven planets;' it also affects the ordinal morpheme $-y\bar{a}$, e.g. (2.281), which occasionally becomes $-h\bar{a}$ in (2.282) alongside the more common $-y\bar{a}$ (2.283).¹³⁴

(2.280)	šə <u>biāhā</u>	[ʃı.'vʲɔ:.hɔ]	evil spirit (IV.13)
(2.281)	kla <u>t</u> āyā	[klɛ.ˈθɔː.jɔ]	third (VIII.9)
(2.282)	ərbiāhā	[ŗ.bɪ.ˈjɔː.hɔ]	fourth (cf. Syriac ، ۲ <i>әḇīʿāyā</i> id.)
(2.283)	ərbiāyā	[ŗ.bɪ.ˈjɔː.jɔ]	id.

134 Macuch, *Handbook*, 98. As the original numbers have largely dropped out of use in Neo-Mandaic, to be replaced by Arabic and Persian numbers, it should not be surprising that the Mandaeans would employ a learned (classical) form like (2.283) rather than a colloquial form, like (2.282).

 $a > a / [C.C_3]$: The phoneme /a/ in closed accented syllables is regularly raised and fronted to [a] rather than the expected outcome [a].¹³⁵

(2.284) nahrā	[ˈnæh.rɔ]	river (III.13)
(2.285) <i>rabbā</i>	[ˈræb.bɔ]	great (I.14)

a > æ / _'Cɔ:.Cɔ: Whenever /a/ appears in pretonic position followed by two open syllables containing /ɔ/, it is raised to /æ/ or even /e/, thereby differentiating it from the following vowels. This most commonly happens when the syllable in question is followed by the augmented form of the plural morpheme, $-\bar{a}n\bar{a}$, e.g. (2.286) becomes (2.287) in the plural. In this example, the vowel /ɔ/ in the tonic syllable is reduced to /a/ when the accent moves to the following syllable (see §2.6.2.2), and it then undergoes dissimilation. Other examples include (2.289), which is the plural of (2.288), and (2.291), the plural of (2.290).

(2.286) <i>barnāšā</i>	[bæı.'nɔ:.∫ɔ]	person (IV.16)
(2.287) barnašānā	[ˌbæı.nɛ.ˈ∫ɔː.nɔ]	people (I.8)
(2.288) <i>həwārā</i>	[ˈhʷəː.rə]	white
(2.289) <i>həwarānā</i>	[hu.we.ˈrɔː.nɔ]	whites (IV.31)
(2.290) <i>Mandāyí</i>	[ˌmæn.dɔ.ˈjiː]	Mandaean (III.16)
(2.291) Mandayānā	[mæn.de.'jɔ:.nɔ]	Mandaeans (I.3)
2.5.3 Insertion		

Epenthetic processes in Neo-Mandaic generally involve consonant clusters (§2.4.2.5 above) and include the excrescence of a consonant in a consonant cluster (§2.5.3.1), the insertion of an anaptyctic /9/ between the first and the second element of a consonant cluster (§2.5.3.2), and the insertion of a prothetic vowel in word-initial position before consonant clusters (§2.5.3.3). A forth form of epenthesis involves the insertion of the voiceless glottal fricative /h/ between two morphemes (§2.5.3.4).

¹³⁵ Otto Jastrow noted the former outcome (but not the latter) in his review of Macuch's *Neumandäische Chrestomathie*; 546.

2.5.3.1 Excrescence

Excrescence involves the insertion of the voiced bilabial stop /b/ between two sonorants, the bilabial nasal /m/ and the alveolar trill /r/. This most commonly occurs when the proclitic preposition ∂m - 'from' is attached to a word beginning with /r/. See §2.4.2.5 above for examples.

2.5.3.2 Anaptyxis

Most Neo-Mandaic consonant clusters are regularly divided through the insertion of an anaptyctic /2/; see §2.4.2.5 for examples of this process.

2.5.3.3 Prothesis and Triradicality

Prothesis is often employed as one of a number of strategies to resolve initial consonant clusters, particularly among words formed from biradical root patterns. The majority of words of Mandaic origin are derived from triradical roots, and consequently those which preserve only two root consonants (or are derived from biradical roots) are susceptible to analogical leveling towards the model provided by these roots. In Neo-Mandaic, this leveling is most frequently accomplished through the addition of a prothetic vowel, most commonly /e/, as in example (2.292).

(2.292) $ehd\bar{a}$ [' ϵ h.də] one (IV.43; cf. CM had id.).

In many cases, the analogical change results in the coexistence of two forms—the new form taking its place beside the old. Generally, the new form assumes the original meaning of the word, while the original form survives in a more restricted sense.¹³⁶ For example, (2.293) becomes (2.294) when deprived of its augment (i.e. when followed by a noun in a construct relation, when followed by a vowel in juncture, or when appearing at the end of an utterance). In the

¹³⁶ This is consistent with Kuryłowicz's fourth law of analogy; Hans Henrich Hock, *Principles of Historical Linguistics* (Berlin: Mouton de Gruyter, 1991), 223–27.

patronymic, however, the word appears as (2.295), reflecting the original form.

(2.293)	ebrā	[ˈɛb.rɔ]	son (lexical) (V.29)
(2.294)	ebbər	[ˈɛb.brֽ]	son (contextual) (I.7)
(2.295)	bar	[ba.ı]	son of PN (IV.53; CM bar id.)

This is also the case with the noun (2.296), which regularly becomes (2.297) when deprived of its augment. The original form, (2.298), is preserved in the name of the gold ring which is part of the priest's ritual clothing, (2.299), as well as certain divine epithets, e.g. (2.300).

(2.296)	ešmā	[ˈɛʃ.mɔ]	name (lexical)
(2.297)	eššəm	[['] ε∫.∫m]	name (contextual) (I.9)
(2.298)	šom	[∫om]	name (CM šum id.)
(2.299)	šom yāwar	[∫om ˈjɒ.wæɪ]	Name of Yāwar
(2.300)	Šom Heyyi	[∫om 'hɛı.ji]	Name of Life (PN)

2.5.3.4 /h/-Epenthesis

Whenever the enclitic form of the copula (q.v., 5.1.1.4) is attached to a noun bearing a possessive suffix ending in a vowel, the voiceless glottal fricative /h/ is inserted between the vowel of the suffix and the copula, e.g. (2.301).¹³⁷

(2.301) *bābahye* [bɔ.'wah.je] he is her father (IX.14)

2.5.3.5 Breaking

The close front unrounded vowel /i/ often breaks in open accented syllables, particularly in the environment of labial consonants (including the bilabial and labiodental consonants) and those articulated on either side of the palate (such as the postalveolar and velar consonants). These include the bilabial stops /p/ and /b/, the bilabial nasal /m/, the voiceless post-alveolar sibilant /ʃ/, and the voiceless velar stop /k/, e.g. (2.302), (2.303), (2.304), (2.305), and (2.306) - but note (2.307). As the singular is the stem of the other

¹³⁷ See Macuch, *Neumandäische Texte*, for further examples from the dialect of Ahwāz.

inflected forms of the noun, the broken vowel is retained even when the syllable is closed, e.g. (2.308).

(2.302)	pienā	[ˈpiĕ.nɔ]	evening (V.16)
(2.303)	bie <u>tā</u>	[ˈbiĕ.θɔ]	house (IX.6)
(2.304)	mienā	[ˈmiĕ.nɔ]	water (IV.7)
(2.305)	šielni	[ˈ∫iĕl.ni]	we asked (III.10)
(2.306)	kief	[kiĕf]	humor
(2.307)	kef ā <u>b</u> den	['kef pv.den]	they made merry (IV.72)
(2.308)	bie <u>t</u> wā <u>t</u>	[biĕθ.ˈwɔθ]	houses (contextual) (III.13)

The breaking of vowels in this environment is an areal feature affecting Neo-Mandaic as well as the local dialects of Arabic, including the Muslim dialect of Baghdad and the dialects of Southern Iraq and Khuzestān.¹³⁸ This phoneme also breaks in accented syllables within Arabic loan words, such as (2.309) and (2.310).¹³⁹ Even the occasional Persian loan word is susceptible to this sound change, e.g. (2.311).

(2.309)	rafieqā́-di	[ra.fiĕ.ˈqɔː.di]	his friends (IV.98)
(2.310)	šiex	[ˈ∫iĕχ]	sheikh (II.7)
(2.311)	miez	[['] miĕz]	table (Persian <i>miz</i> id.)

2.5.4 Lenition

Two forms of lenition occur regularly in Neo-Mandaic, both of which involve consonants. Stops may be fricativized (§2.5.4.1) and fricatives may lose oral contact and become glottals (§2.5.4.2).

Bruce Ingham, Languages of the Persian Gulf, in Alvin J. Cottrell (ed.), The Persian Gulf States: A General Survey (Baltimore: Johns Hopkins, 1980), 323–24 gives a number of examples, including biet (house), shiex (sheikh), and bien (between), which are similar to the forms found in Mandaic.

¹³⁹ Werner Arnold, review of *Neumandäische Texte im Dialekt von Ahwāz*, by Rudolf Macuch, *Zeitschrift der deutschen morgenländischen Gesellschaft 146* (1996): 201 compares this feature to the breaking found in the Western Neo-Aramaic dialect of Jubb 'Adīn and the neighboring Arabic dialects of Qalamūn.

2.5.4.1 Fricativization

Fricativization or spirantization, i.e. the change of a stop to a fricative, is a general Aramaic phenomenon. In earlier stages of the language, the voiced stops /b/, /g/, and /d/, and their voiceless counterparts /p/, /k/, and /t/ (the so-called *beğadkepat* series) were regularly fricativized to [v], [κ], [δ], [χ], [f], and [θ] after vowels. This specific process is no longer productive in Neo-Mandaic or any other Neo-Aramaic dialect.

Fricativization does, however, regularly occur in Neo-Mandaic. The voiced stops /b/, /g/, and /d/ become the voiceless fricatives /f/, / χ /, and / θ / before voiceless stops. This typically occurs when the comparative suffix *-tar* is added to an adjective ending in /b/, e.g. (2.313), which is the comparative of (2.312).¹⁴⁰

Note also that the voiced velar stop /g/ has become / χ / throughout the conjugation of the verb *ləkat* ~ *ləkot* (*lāket*) 'to seize' (CM **lgat** ~ **lgut** (**lagit**) id.), the dental stop /d/ has become / θ / before the voiceless uvular fricative / χ / in (2.314), and the voiceless velar stop /k/ in the modern reflex of Classical Mandaic kd kma (2.315).

(2.312)	rabbā	[ˈræb.bɔ]	great (I.14)
(2.313)	raptar	[ˈræf.tæɪ]	greater
(2.314)	do <u>ta-t-k</u> on	[do.'θaθ.χon]	both of you
(2.315)	ka <u>t</u> kammā	['kæ0.,kæm.mɔ]	how much (I.14)

2.5.4.2 Oral Depletion

The loss of oral contact, but retention of glottal friction, results in the voiceless interdental fricative $/\theta$ / becoming /h/ before a syllable beginning with /l/. This sound change, and the inability of $/\theta$ / to be lengthened (see §2.4.4.2) are responsible for the allomorphs of the copula *[$\epsilon\theta$] (CM 'it). When followed by the preposition *l*- for, the copula takes the form *eh*-[ϵ h], e.g. (2.316). See §5.1.1.4 for further details on the copula.

140 For more examples of /b/ becoming /f/ before /t/ in the classical language, consult Nöldeke, *Mandäische Grammatik*, 48 and Macuch, *Handbook*, 56–57. Malone ("Mandaic Phonology," 148) suggests a rule: "a heterophonous cluster XY, X a nonrilled obstruent (especially b, β , q) and Y a stop (especially t) may dissimilate X in manner to become a spirant but assimilate X in voice."

(2.316) <i>ehle</i> [' ϵ h.le] I have (lit. then	re is for me) (III.4)
---	-----------------------

In addition to the copula, this sound change also affects the paradigm of the object suffixes. When the third feminine singular preterite form of a verb takes a pronominal object with *I*-, the $/\theta/$ is invariably weakened to an /h/, which then disappears, causing the following /l/ to become geminated, e.g. (2.317); for another example of [h] lost in this position and the concomitant gemination of the following consonant, see examples (2.148) and (2.149) under 2.4.2.4.2 above.

(2.317) $m\bar{a}relli$ [mɔ.'rɛl.li] she told him (IX.10)¹⁴¹

The voiced uvular fricative /B/ is depleted to /h/ in two words, both in the environment of the closed back rounded vowel /u/, i.e. the example given in (2.319), from (2.318), and the learned form found in (2.320), which is pronounced as in (2.321) colloquially.¹⁴²

(2.318)	zāġ	[ZDR]	chicken (Arabic zāġ 'crow')
(2.319)	zəhunān <u>k</u> on	[zɛ.hu.ˈnɔn.χon]	your (pl.) chickens (IV.124) ¹⁴³
(2.320)	mambugā	[mam.'bu:.sp]	sacramental water
(2.321)	mambuhā	[mam.ˈbuː.hɔ]	id.

2.5.5 Loss

In addition to the voiceless glottal fricative /h/, which is lost in a number of environments (\$2.5.5.1), vowels in word-final position are susceptible to loss (\$2.5.5.2 and (\$2.5.5.3).

2.5.5.1 Voiceless Glottal Fricative /h/

The voiceless glottal fricative /h/ is lost in several environments. When it is the second element in a consonant cluster, it is deleted (see §2.4.2.5 above). This deletion manifests itself in the allomorphs of the

¹⁴¹ This is presumably derived from a form like *amrahli < *amratli < *amrat + li.

¹⁴² Macuch observed this alternation in the dialect of Ahvāz as well; Macuch, *Neumandäische Texte*, 17.

¹⁴³ Note the appearance of the morpheme -un- before the plural morpheme $-\bar{a}n$ -. This may represent the 'Tehruni' pronunciation of Persian or the Luri plural morpheme /un/ which is used for animate beings.

plural morpheme $-h\bar{a}$, which appears regularly as $-h\bar{a}$ after vowels, e.g. (2.322) and $-\bar{a}$ after consonants, e.g. (2.323), on loan words in Neo-Mandaic.

(2.322) <i>șubbihā</i>	[ˌsˤʌb.bi.ˈhɔː]	Sabians (III.12; Persian <i>sobbi-hā</i> id.)
(2.323) <i>jihelā</i>	[ˌʤi.hɛ.ˈlɔː]	children (II.15; Arabic jāhil 'ingénu')

/h/ is also regularly lost in word-final environment; see (2.77) and (2.79) in section §2.3.1.10 for two examples of words in which this phoneme is deleted whenever it appears in word-final environment, after the loss of the augment (for which, see §2.5.5.2 below). Other environments in which /h/ is weakened or deleted include those described in §2.7.1.9 under the allegro rules below.

2.5.5.2 Apocope

Apocope is extremely common in Neo-Mandaic. Whenever a word is immediately followed by a word with which it is in juncture, i.e. semantically linked, the nominal augment, an unaccented word-final /ɔ/ found on most Neo-Mandaic nouns, is regularly apocopated. The form which preserves the augment is known as the lexical form; the apocopated form is known as the contextual form. The forms most likely to be in contextual form are pronouns followed by verbs, e.g. (2.325), nouns in construct with another noun, e.g. (2.327), nouns modified by adjectives, e.g. (2.329), generic nouns incorporated into the meaning of a verb, e.g. (2.331), quantifiers modifying nouns, e.g. (2.333), and adverbs modifying verbs, e.g. (2.335).

(2.324)	anā	[ˈaː.nɔ]	I (I.4)
(2.325)	an həwit	[an hɪ.'vit]	I was (II.16, III.4)
(2.326)	bie <u>tā</u>	[ˈbiĕ.θɔ]	house (X.8)
(2.327)	bie <u>t</u> dusti	[biĕθ 'dus.ti]	the house of a friend (V.4)
(2.328)	šerrā	[ˈ∫ɛr.rɔ]	war
(2.329)	šer horettā	[∫ɛr hu.ˈrɛt.tɔ]	the Second (World) War (II.2)
(2.330)	zipā	[ˈziː.fɔ]	lie
(2.331)	zip̄ qahbet	[zif 'qah.vet]	you lie (V.23)
(2.332)	perzā	['pɛr.zɔ]	little
(2.333)	perz perāhi	[pɛ.s pɛ.ˈrɔː.hi]	a little money (V.1)

2. Phonology

(2.334) <i>peršā</i>	['pɛı.∫ɔ]	tomorrow (V.19)
(2.335) perš qa <u>t</u> inā	['pɛı∫ qa'θiː.nɔ]	I'll come tomorrow (V.17)

It is also susceptible to loss at the end of an utterance, e.g. (2.336) and (2.337).

(2.336)	Mandayān	[ˌmæn.dɛ.ˈjɔn]	Mandaeans (I.5, III.12)
(2.337)	ərqi	[ŗ.'qi:]	heaven (X.12)

2.5.5.3 Elision

Much like apocope, elision involves the loss of the final vowel of a word when immediately followed by a second word. Elision differs from apocope in that any vowel or diphthong (not only $/ \circ /$) can be elided, and elision occurs only when the second word begins with one of the six primary vowels (excluding $/ \circ /$) or syllabic segments. In the transcriptions, elision is indicated with an apostrophe.

(2.338) qəmšielā	[qm.'∫iĕ.lɔ]	she asks
(2.339) qəmšiel' əmhattā	[qṃ.ˈ∫ ^j εl ṃˈħaθ.θɔ]	she asks to speak (II.31)
(2.340) aklelli	[ax.'lɛl.li]	they eat it
(2.341) aklell'əl-hadid	[aχ. ₁ lɛl.lɪl.ħaˈdid]	they eat iron (V.12)
(2.342) <i>jawāb-de</i>	[ʤa.ˈwɒb.de]	my answer
(2.343) jawāb-d'ahbita	[ʤa.ˈwɒb.ˌdæh.vi.tə]	I answered her (II.19, II.20)
(2.344) <i>ke</i>	[ke]	which (III.19)
(2.345) k'ehli	[ˈkɪl.li]	which has (V.26)

Elision is regular before the independent object marker *əl*- and verbal complements in the subjunctive. Elsewhere, elision is more characteristic of allegro speech (see §2.7 below).

2.5.7 Other Alternations between Segments

The monosyllabic proclitic morphemes, i.e. the preposition b- 'in,' the object marker ∂l -, the negative morpheme $l\dot{a}$ - 'not,' the preposition ∂m - 'from,' the indicative morpheme q-, and the conjunction w- 'and' have different allomorphs in response to the segments which immediately follow them, such as a consonant, a vowel, or a consonant cluster, as illustrated in Table 12. These allomorphs

represent a combination of the morphophonemic changes described above, such as anaptyxis and elision, as well as regular allophones of the segments involved, such as the syllabification of [w].

Table 12: Allomorphs of the Proclitic Morphemes

Independent	+ CV	+ V	+ C(ə)C	Gloss
-	<i>bə-</i> [bə]	<i>b-</i> [b]	<i>be-</i> [bɛ]	in, at, with
<i>əlāw</i> [ə.ˈlɔʊ]	<i>əl-</i> [ɛl]	<i>əl-</i> [ɛl]	<i>əl-</i> [εl]	REF
<i>lā</i> [lɔ]	<i>lá-</i> [la]	<i>I</i> -[1]	<i>lá-</i> [læ]	NEG
<i>min</i> [mɪn]	<i>əm-</i> [m]	<i>m-</i> [m]	<i>mə</i> - [mε]	with
<i>qā</i> [qɔ]	<i>qə-</i> [qə]	<i>q-</i> [q]	-	IND
<i>u</i> [u]	<i>u-</i> [u]	<i>w</i> -[w]	<i>wə</i> - [wε]	and

Note that anaptyctic $|\partial|$ is deleted after a proclitic morpheme, bringing the two consonantal segments formerly divided by the $|\partial|$ back into contact.

2.6 Suprasegmental Morphophonology

The position of the accent, while largely predictable, plays a minor role in distinguishing between several morphemes (§2.6.1), and a much more considerable role in determining the phonological structure of the Neo-Mandaic word (§2.6.2).

2.6.1 Stress and Morphological Processes

Stress plays a role in three morphological processes: distinguishing the abstraction morpheme from the indefinite morpheme (§2.6.1.1), distinguishing the plural morpheme found on loan words from the nominal augment found on original Mandaic vocabulary, (§2.6.1.2), and distinguishing the third person singular and plural of the derived stem imperfectives from the contextual forms of the first person singular and plural of these imperfectives (§2.6.1.2). 2.6.1.1 The Abstraction and Indefinite Morphemes

The abstraction morpheme -i (see §2.4.4.3) resembles the indefinite morpheme -i; these two morphemes are distinguished by stress alone. See §2.4.4.3 for examples.

2.6.1.2 The Foreign Plural Morpheme and the Nominal Augment

The plural morpheme $-h\dot{a}$ (originally derived from Persian; see §2.4.4.3) is used to pluralize loan words exclusively, including those from languages other than Persian. Whenever it follows a consonant, it assumes the form $-\dot{a}$, which closely resembles the nominal augment $-\ddot{a}$, e.g. (2.346) and (2.347). These two morphemes appear in complementary distribution, i.e. the former exclusively on non-nativized loan words and the latter exclusively on native vocabulary and nativized loan words, e.g. (2.348). Nonetheless, it was deemed necessary to indicate the stress on the plural morpheme.

(2.346)	darwišā	[ˌdæı.vi.ˈ∫ɔː]	dervishes (IV.20; Persian darwiš id.)
(2.347)	mimrā	[ˈmɪm.rɔ]	literature (I.5; CM mimra id.)
(2.348)	jisrā	[ˈʤɪs.rɔ]	bridge (IV.12; Arabic jisr id.)

2.6.1.3 The Contextual Forms of the Verb

The third person plural forms of the imperfective, e.g. (2.350), are identical to the contextual forms of the first person plural, e.g. (2.349), and the third person singular forms of the imperfective, e.g. (2.352), are nearly identical to those of the third person singular, e.g. (2.351) throughout most of the verbal paradigm,¹⁴⁴ save for the location of the accent and the gemination of the middle radical.

(2.349)	qəm <u>ḥatt</u> én	[qm.ħaθ.'θεn]	we speak (contextual) (II.22)
(2.350)	qəm <u>hatt</u> en	[qmˈ.ħaθ.θɛn]	they speak (II.21)
(2.351)	əmḥade <u>t</u>	[m.ħa.ˈdεθ]	I speak (subjunctive) (I.7)
(2.352)	qəmḥadde <u>t</u>	[qm.ˈħad.dεθ]	he speaks

144 The exception is the paradigm of the G-stem, in which the presence of a tense vowel in the accented syllable also distinguishes between the two forms; for more details, see §5.1.1.2.

2.6.2 Stress and Phonological Structure

A shift in the position of the accent provoked by a morphological process has profound implications for the phonological structure of a word in Neo-Mandaic; it can result in the simplification of a geminate consonant ($\S2.6.2.1$), the reduction of a vowel in the preceding syllable ($\S2.6.2.2$), the loss of /ə/ in word-initial position ($\S2.6.2.3$), and the complete syncope of a word-internal vowel ($\S2.6.2.4$).

2.6.2.1 Geminate Simplification

As described above in §2.4.4.2, there are two distinct types of geminated consonants in Neo-Mandaic, lexical and phonological geminates. The latter arise regularly whenever the accent falls upon a lax vowel in an open syllable; if the accent shifts, phonological gemination will be lost, as illustrated in examples (2.353) and (2.354).

(2.353)	qəmḥadde <u>t</u>	[qm.ˈħad.dεθ]	he speaks
(2.354)	qəmḥadeṯnā	[qm.ħa.ˈdɛθnɔ]	I speak

Geminates (both lexical and phonological) are also regularly simplified in word-final position, e.g. (2.355), which is the contextual form of (2.356).

(2.355)	rab	[ræb]	leader (II.2)
(2.356)	rabbā	[ˈræb.bɔ]	leader (II.3)

2.6.2.2 Pretonic Reduction

Vowels in open pretonic syllables in Neo-Mandaic are regularly subjected to reduction. The tense vowel /5/ is regularly reduced to its lax equivalent /a/ in an open syllables, whenever the accent is drawn away from the syllable, e.g. (2.358) from (2.357).

(2.357)	qəgāțel	[qə.ˈgɒː.t ^s ɛl]	he will kill
(2.358)	qəgatelnā	[qə.ga.ˈt ^s ɛl.nɔ]	I will kill

Its lax equivalent /a/ becomes /ə/ whenever the accent moves to the following syllable; cf. the lexical form (2.360) with its contextual form (2.359). Note that the other tense vowels /i/ and /u/ are also

reduced to their lax allophones [I] and $[\Lambda]$ in pretonic position, as in (2.362).

(2.359)	abd	[avd]	he did (contextual) (I.11)
(2.360)	əbad	[ə.'vad]	he did (V.27)
(2.361)	mu	[mu]	what (VI.6)
(2.362)	mujur	[mʌ.ˈ�ʒuɪ.nə]	how I am (IX.3; Persian če-jurid.)

2.6.2.3 Aphaeresis

Word-initial $\langle \vartheta \rangle$ may only occur immediately before the accent, i.e. in pretonic position, as the outcome of a reduced tense vowel. If the accent shifts from the syllable immediately following word-initial $\langle \vartheta \rangle$, it will be deleted.

(2.363)	əha <u>b</u> yon	[ə.ˈhav.jon]	they gave
(2.364)	hablonni	[hav.'lon.ni]	they gave it (X.10)

2.6.2.4 Syncope

Neo-Mandaic does not tolerate lax vowels in open pretonic syllables. For this reason, closed unstressed syllables which become open are subjected to one of two treatments if they contain a lax vowel; the most common treatment involves the syncope of the lax vowel. The vowel in the newly open consonant is deleted and the initial segment of the syllable closes the preceding syllable, provided that the syllable preceding it is open as well. Much like the accent, the vowel to be syncopated is always calculated retrogressively; i.e. from the final syllable back toward the initial one.

This sound change has a respectable pedigree within the language, being shared by other varieties of Aramaic as well. For example, compare the third masculine singular perfective form (2.365) with the third feminine singular preterite form (2.366).¹⁴⁵ The first form derives from **napáq* after pretonic reduction of the vowel in the first syllable (see §2.6.2.2 above) while the second is realized after syncope of the vowel in the second syllable of the underlying form,

¹⁴⁵ These are the forms attested in other dialects of Aramaic as well; cf. Syriac *nəpaq* and *nepqat*.

**napaqát.* This sound change is still productive; elsewhere in the conjugation of the verb, the /e/ found in the second syllable of the root of the imperfective is syncopated whenever the following syllable begins with a vowel, e.g. (2.367) and (2.368).

(2.365)	nəpaq	[nə.ˈfaq]	he came out (VIII.7)
(2.366)	nəpqat	['nɜf.qat]	she came out (II.15)
(2.367)	qā <u>b</u> ed	['qp:.ved]	he does (IV.22)
(2.368)	qābden	['qɒv.dɛn]	they do (IV.20)

The verb $ab\bar{a}$ ($b\bar{a}yi$) 'to want,' which possesses a lax vowel in the initial syllable of most of its forms, frequently succumbs to syncope whenever it is negated with the proclitic negative morpheme $l\dot{a}$ -, cf. (2.369) and (2.370). Other verbs also undergo syncope after the negative $l\dot{a}$ -, whenever it is prefixed to a form containing a /ə/ in an open initial syllable, e.g. (2.371). Likewise, after (2.372) is negated with lu [lu] 'non-,' the /ə/ is syncopated, as demonstrated in (2.373).

(2.369)	qabén	[qa.ˈbɛn]	we want (contextual) (IV.132)
(2.370)	láqbèn	['laĸ.bɛn]	we don't want (IV.109)
(2.371)	lábgesyon	['lab. ₁ ĸas [°] .jon]	they did not stay (IV.146)
(2.372)	hənāye	[hə.ˈnɔː.je]	it is here
(2.373)	luhnāye	[luh.'nɔː.je]	it is not here (IX.10)

Note that syncope cannot occur if the preceding syllable is closed. Consequently the lax vowel in the pretonic syllable of (2.374) has not succumbed to syncope, but is merely reduced to /ə/. In the event that the preceding syllable becomes open by any means (e.g. through the deletion of /h/), the vowel will immediately undergo syncope, as in (2.375).

(2.374)	qəmahrəbāt	[qə.mah.rɛ.ˈwət]	you destroy (IV.58)
(2.375)	qəmahrəbāti	[qə.mæı.ˈwɔː.ti]	you destroy it (IV.59)

Alternately, when preceded by a closed syllable, the lax vowel may be "lengthened" to its tense equivalent; this is most frequently the case in the imperfective paradigm of quadriradical verbs, e.g. (2.376), which becomes (2.377) with the addition of the second person singular

morpheme rather than ***qəmbašqret*. This also appears to be the case for loan words that have not yet been accommodated to Neo-Mandaic phonology; e.g. (2.378), which becomes (2.379) when augmented, instead of ***hākmā*. In either case, this has the effect of preserving the syllable structure of the word.

(2.376)	qəmbašqer	[qṁ.'ba∫.qɛı]	he knows (IV.35)
(2.377)	qəmbašqiret	[qm.ba∫.'qi:.rıt]	you know (IV.106)
(2.378)	<u> ḥākem</u>	[ˈħɒː.kɛm]	governor (IV.30; Arabic <i>hākim</i> 'judge')
(2.379)	<u> ḥākimā</u>	[ˌħɒ.ki.ˈmɔː]	governor (augmented) (IV.65)

2.7 Allegro Rules of Neo-Mandaic

With the exception of Text I, a formal speech which was delivered slowly and deliberately, the elicited texts were delivered in a normal tone of voice, and at times tended towards fast or "allegro" speech. The following should be considered a partial inventory of the features of allegro speech. Unfortunately, without further data, it is impossible to outline the full inventory of sound changes attributable to allegro speech, and some of the examples cited here may well represent allophonic variation or the results of regular sound changes for which we simply have insufficient data. Most of the examples are culled from the text collection; as a general rule of thumb, the more examples that follow a feature of allegro speech, the more common the phenomenon is within the text collection.

2.7.1 Consonants in Allegro Speech

Consonantal segments may be subject to both contact assimilation (between immediately neighboring segments) and distant assimilation (across one or more intervening segments). Either form of assimilation may be anticipatory (in which case the segment assimilates to a following segment) or perseverant (in which case the segment assimilates to a preceding segment). Furthermore, the assimilation may be total (in which case one segment completely assimilates to another) or partial (in which case one feature of a segment, such as its point or manner of articulation, assimilates to another). Other phenomena sporadically affecting consonants in Neo-Mandaic include

nasal lenition (\$2.7.1.5), fricativization (\$2.7.1.6), paragoge (\$2.7.1.7) and the devoicing of final stops (\$2.7.1.8). The various allophones of the voiceless glottal fricative /h/ in allegro speech are treated in a separate section (\$2.7.1.9).

2.7.1.1 Partial Anticipatory Contact Assimilation

Partial anticipatory contact assimilation involves the assimilation of a segment in voice, place of articulation, or manner of articulation to the segment immediately following it.

 $v > m / _n: /v/$ assimilates in its manner of articulation (but not in place of articulation) to /n/ in examples (2.380) and (2.381), the latter of which is the contextual form of *laqmahrebnā*.

(2.380)	qāhebnak	[qɔˈhɛm.nax]	I will give you (V.29)
(2.381)	láqmahrem	['laq. ₁ mah.rem]	I won't destroy (IV.61)

t > d / _C_[+voice]: /t/ assimilates in voice to /b/ across a word boundary in example (2.382).

(2.382) *nəpqat barra* ['nsf.qad 'bar.ra] she came outside (II.15)

 $d > t / _C_{[-voice]}$: /d/ assimilates in voice to /q/ across a word boundary in example (2.383).

 (2.383) ana qabed qam-dak ['a:.na qa'vɛt 'qam.daχ] I will do it for you. (IV.63)

 $z > s / _# #C_{[-voice]}$: /z/ assimilates in voice to /p/ across a word boundary in example (2.384).

(2.384) *perz perāhi* [pɛ.is pɛ.'rɔ:.hi] a little money (V.1)

 $z > s / _C_{[-voice]}$: /z/ also occasionally assimilates in voice when it comes into contact with a voiceless stop or fricative; e.g. (2.385), (2.386), and (2.387). Often, however, /z/ does not assimilate, e.g. (2.388).

(2.385)	rezqā́-dan	[rɛs.ˈqɔ:.dæn]	our livelihoods (IV.132)
(2.386)	Dezful	[dɛs.ˈful]	Dezful (III.6)
(2.387)	həztonne	[hɛs. 'ton.ne]	you saw me

2. Phonology

(2.388) <i>Dezful</i>	[dɛz.ˈful]	Dezful (III.7)
-----------------------	------------	----------------

 $\int s / u_t$: In examples (2.389) and (2.390), the phoneme $/\int/$ frequently alternates with the voiceless alveolar sibilant /s/. This outcome is presumably conditioned by the closed back vowel as well as the following alveolar consonant /t/, to which it assimilates in point of articulation; but notice that this alternation does not occur in either (2.391) or (2.392).

(2.389)	Šuštar	[ˈ∫us.tæ.ɪ]	Shushtar (IV.65)
(2.390)	kušțā	[kʌs.ˈt ^s əː]	truth (I.9)
(2.391)	deštā	[ˈdɛ∫.tɔ]	land (IV.98)
(2.392)	əštā	[ɛʃ.ˈtɔː]	now (I.10)

2.7.1.2 Total Anticipatory Contact Assimilation

Total anticipatory contact assimilation involves the complete assimilation of a segment to the segment immediately following it.

 $v > n / _n$: /v/ assimilates completely to /n/ in (2.393). More frequently, it assimilates in manner of articulation but not place of articulation in this environment (see 2.7.1 above).

(2.393) *ahebnak* [a.¹hɛn.nɑk] I give you (VIII.10)

1 > C / C: The /l/ of the proclitic preposition *əl*- assimilates to the following segment in (2.394) and (2.395); the same phoneme also assimilates across a word boundary in example (2.396).

(2.394)	bašqir əl-na <u>b</u> ša <u>k</u>		
	[bæ∫.'qi.1 ɛn.'næv	.∫αχ]	know yourself (m.sg.)!
(2.395)	qəmabğeşşən əl-	deštā	
	[qə.mab.'кзs.s [°] ər	n εd.ˈdɛ∫.tɔ]	I will stop the ground (IV.108)
(2.396)	čehél ruz	[ˈtʃɛr.ruz]	forty days (IV.137)

 $n > m / _m: /n/$ assimilates in place of articulation to /m/ across a word boundary in (2.397).

(2.397) *šeršan m-awwál ālm həwat* ['ʃɛɪ.ʃam mæv.'væl ɔlm hɛ.'vɑt] Our religion has existed from the first epoch of time. (I.8)

2.7.1.3 Partial Anticipatory Distant Assimilation

Partial anticipatory distant assimilation involves the assimilation of a segment to another segment in voice, manner of articulation, or place of articulation, over one or more intervening segments.

 $b > p / VC_{[-voice]}$: /b/ assimilates in voice to the following voiceless stops in example (2.398).

(2.398) $b \partial dqu$ ['pet.qu] he put them (V.4)

2.7.1.4 Partial Perseverant Contact Assimilation

Partial perseverant contact assimilation involves the assimilation of a segment in voice, manner of articulation, or place of articulation to that of a segment which immediately follows it.

 $v > f \ / \ C_{[\text{-voice}]}$: /v/ assimilates in voice to a preceding /t/ in example (2.399).

(2.399) *etber* ['st.fɛ1] it broke off (IX.12)

 $m > n / n_{:} /m/$ assimilates in place of articulation to a preceding [n] across a word boundary in (2.400).

(2.400) məḥatten Mandāyi [mə.'ħaθ.θεn ˌnæn.də.'ji] they speak Mandaic (II.32).

2.7.1.5 Paragoge

Paragoge is the word-final insertion of a segment, and is famously sporadic cross-linguistically. In Neo-Mandaic, it occurs in example (2.401), in which a voiced alveolar approximate [1] is inserted in word-final position. On one occasion, (2.402), the original Persian form, *bandi* 'prisoner,' was employed.

(2.401)	bandir	[bæn.ˈdi.ɪ]	prisoner (IV.129)
(2.402)	bandi	[bæn.'di:]	id. (IV.133)

2.7.1.6 Nasal Lenition

In syllable-final and particularly word-final position, /n/ is susceptible to weakening or loss after non-high vowels such as:

102	2. Phonology	
/e/		
(2.403) qəmṭallén (2.404) qəmḥaṯten (2.405) qāṯen m-Erāq /a/	[qṃ.t [°] al.'lɛ̃] [qṃ.'ħaθ.θɛ̃] ['qɒː.θɛ̃' mɛ.rɒq]	we play (II.4) they speak (II.21) they came from Iraq (II.23)
(2.406) <i>geš-dan</i> (2.407) <i>qəmwadyātan</i>	[ˈgɛʃ.dã] [qṃ.ˈvæ.di.jɔ.ˌtã]	all of us (II.4) you will take us (IV.122)
/ɔ/ (2.408) dukkān (2.409) zəģunān (2.410) pərāhān	[dʌk.'kõ] [zɛ.ʁu.'nõ] [pɛ.rɔ.'hõ]	places (I.4) chickens (IV.142) money (VIII.8)
/0/		
 (2.411) pardason (2.412) genz hilānnon (2.413) qamdəkon 	[paɪˈdɑː.sõ] [gɛnz hi.ˈlɒn.nõ] [qam.ˈdo.χõ]	they left (II.12) they are very strong (V.4) for you (pl.) (X.7)

As demonstrated by the examples, the preceding vowel continues to remain nasalized even after the weakening or loss of the nasal.

2.7.1.7 Fricativization

'Fricativization' or 'spirantization,' the change of a stop to a fricative, is a general Aramaic phenomenon, although the process is no longer productive in Neo-Mandaic (with few exceptions; see 2.5.4.1 above) or indeed any other Neo-Aramaic dialect. The phonemes /t/, /b/, and /q/ are sporadically fricativized in post-vocalic position:

 $t > \theta \ / \ V_:$ The voiceless alveolar stop /t/ is sporadically fricativized in post-vocalic position.

(2.414)	həwit	[hı.ˈviθ]	I was (II.16, III.5)
(2.415)	a <u>t</u> it	[ε. ['] θiθ]	I came (II.17)
(2.416)	amrate	[ˈam.rɑ.θə]	she told me (II.26)
(2.417)	metelā́	[mɛ.θɛ.'lɔː]	stories (III.20)
(2.418)	azimat	[a.zi.ˈmɑθ]	invocation (IV.20)

$b > v / V_:$ The voiced l	bilabial stop /b/	' is sporadically	fricativized in
post-vocalic position.			

(2.419)	qabin	[qa. ['] vin]	I want (II.17)
(2.420)	Hibel	['hiː.vɛl]	Hibel (p.n.) (IV.1)
(2.421)	mu qəbāt	[mʌ.ˈqʷət]	what do you want (IV.63)
(2.422)	gāw welāt	[ˌgəu vɛ.ˈlət]	in the city (V.20, V.26)

 $q > \chi / V_{-}$: The voiceless uvular stop /q/ is sporadically fricativized in post-vocalic position. This rule is also typical of Persian, in which it is regular.

(2.423)	nəpaq	[nə.ˈfaʁ]	he exited
(2.424)	láqbèn	['laĸ.bɛn]	I don't want (IV.109)
(2.425)	láqder	['lar.qe1]	he didn't take (V.26)

d > δ / V_: In one example, (2.426), the voiced alveolar stop /d/ was weakened to the voiced interdental fricative [δ] in intervocalic position. While this segment is otherwise rare in Neo-Mandaic, it must have enjoyed a much broader distribution in an earlier period of the language, as attested by a number of doublets in which the expected realization *[δ] has merged with the voiced alveolar sibilant /z/, cf. (2.427) with the literary expression (2.428); see also the rhyming pair (2.429) and (2.430), of which both mean 'gold,' but only the former is in current use. ¹⁴⁶

(2.426)	šadder	['∫ɑ:.ðɛ.]	send (m.sg.)! (IV.11)
(2.427)	dəmā	[də.ˈmɒː]	blood
(2.428)	besrā u zammā		
	[bes.rou.'zæm.m	ວ]	flesh and blood
(2.429)	-1	o] ['dæh.wo]	flesh and blood gold

146 Rudolf Macuch, "Some Orthographic-Phonetic Problems of Ancient Aramaic and the Living Aramaic Pronunciations," *MAARAV* 5–6 (1990): 226 considers the occurrence of this doublet and others where /z/ alternates with /d/ at the beginning of a word either to be relics of the historical orthography or "hyperarchaizing pseudo-corrections," but also notes the fact that the Mandaic orthography is otherwise phonetic, lacking in historical spellings, which militates against his argument. 2. Phonology

Defricativization: Fricativization was not observed in any other stops, and in one instance, example (2.431), a voiceless uvular fricative was realized as a voiceless velar stop, although this may be attributable to Arabic influence (as the second person pronominal suffix here mirrors the form of the Arabic suffix).

(2.431) <i>ahebnak</i> [a.'hen.nak] I give you (VIII.10)
--

2.7.1.8 Devoicing of Final Stops

Voiced stops in word-final position are occasionally devoiced to voiceless stops. This is best illustrated by example (2.432), which is ultimately derived from the Arabic *bilād*, pl. of *balad* 'place.' In this word, the final stop is always devoiced, and occasionally even fricativized, as in (2.433). Other Arabic loan words, such as (2.434), on the other hand, are often devoiced, but not always, as in (2.435).

(2.432)	welāt	[vɛ.ˈlət]	city or region (IV.20)
(2.433)	welāt	[νε'1οθ]	id. (III.11)
(2.434)	ʿād	[ʕət]	yet (III.13, III.14, III.15)
(2.435)	ʿād	[sod]	yet (X.17)

2.7.1.9 Syllable-Closing and Intervocalic /h/

Neo-Mandaic does not generally tolerate /h/ in syllable-closing or intervocalic environments, and employs a wide variety of strategies to prevent it from appearing in those environments, including deletion, anaptyxis, and, in one instance, voicing.

Cə.hVC > CVC: /h/ is frequently deleted in intervocalic position, particularly when following /ə/, in which case /ə/ is deleted as well, e.g. examples (2.436) - (2.440).

(2.436) q	ahazin <u>k</u> on	[qa.ˈzin.χon]	I see you (pl.) (I.4)
(2.437) q	əhazén	[qa.'zɛn]	we saw (II.14)
(2.438) q	əhāper	['qp:.fɛ.]	he dug (IV.125)
(2.439) č	ehel ruz	[tʃɛr.ruz]	40 days (IV.137)
(2.440) <i>y</i>	rəhem	[jɛm]	settled (IV.144)

 $h > ? / V_V$: When /h/ is deleted in intervocalic position, a glottal stop /?/ can be substituted in its place.

(2.441) *waxti həwit* ['væx.ti ?r'vit'] a time when I was (II.1)

CVh.CV > CV.ha.CV: An anaptyctic /a/ is occasionally inserted between syllable-closing /h/ and the initial consonant of the following syllable, to preserve /h/ from being deleted in that position (see 2.4.2.5).

(2.442)	nahrā	['næ.hə.rɔ]	river (III.8)
(2.443)	ehlu	[ˈɛ.hə.lu]	they have (IV.45)
(2.444)	qədāhlen	[qə.ˈdɒː.hə.lɛn]	they will be afraid (IV.98)
(2.445)	qāl ahbat	[qɔl ˈa.hə.ˌvɑt]	it made a noise (IV.139)

 $h > j / (C)V_C$: In one instance, (2.446), a loan word from Persian, the /h/ underwent a voicing process, becoming a voiced glide /j/ and forming a diphthong with the preceding vowel.

(2.446)	mehman-de	[m31.ˈmæn.də]	my guest (V.16)
---------	-----------	---------------	-----------------

2.7.2 Vowels in Allegro Speech

The vowels of Neo-Mandaic are also subject to allegro rules. These rules include centralization in closed accented syllables (§2.7.2.1), vowel harmony (§2.7.2.2), anaptyxis (§2.7.2.3), and reduction in word-final environment (§2.7.2.4).

2.7.2.1 Centralization

The four principal non-open phonemes, /i/, /u/, /e/, and /o/, are occasionally centralized in closed accented syllables.

/i/ > /i/: /i/ is infrequently centralized to the close central unrounded vowel /i/ in closed accented syllables.

(2.447) miṣrā ['mɨs.rb] boundary (IV.20)

/u/ > /u/: /u/ is occasionally centralized to the close central rounded vowel /u/ in closed accented syllables.

(2.448)	kuštā	[ˈkʉ∫.t [°] ɔ]	truth (IV.118)
(2.449)	Šuštar	[ˈʃʉʃ.tæɪ]	Shushtar (III.19)

2. Phonology

(2.450)	šu <u>b</u> bā	[ˈʃʉw.wɔ]	seven (III.18)
(2.451)	dukki	[ˈdʉk.ki]	his place (IV.140)

/e/ > /3/: /e/ is frequently centralized to the open mid central vowel /3/ in closed accented syllables.

(2.452)	əm <u>h</u> ade <u>t</u>	[m.ħa.'d3θ]	I will say (I.7)
(2.453)	qəmabğeşşən	[qə.mab.ˈʁɜs.sˤən]	I will stop (IV.108)
(2.454)	etber	[ˈst.fɛɹ]	it broke off (IX.12)
(2.455)	le <u>k</u> tan	['lɜχ.tan]	we are not (I.12)
(2.456)	pəherye	[pɛ.ˈhɜɪ.je]	it is flying (V.21)

/o/ > /o/: /o/ is occasionally centralized to the close-mid central rounded vowel /o/ in open unaccented syllables.

(2.457) <i>sotór-di</i> [se.'toɪ.di] his razor (VII	VIII.3)
---	---------

2.7.2.2 Umlaut

The marginal phoneme /ə/ is highly susceptible to influence from vowels in neighboring syllables.¹⁴⁷ Nearly all examples of this influence are restricted to /ə/ in an open pretonic syllable (with the single exception of *qamdəkon* [qam.'do. $\chi \tilde{o}$] 'to/for you' (X.7), which reflects the influence of the sentence intonation, §2.4.4.4). The result is a wide variety of realizations for /ə/, in addition to the more usual [ə] and [Ø]. Examples include:

 $\vartheta > \epsilon$ / _Ci: /ə/ is often fronted and raised to [1] or [ϵ] before an accented syllable containing /i/.

(2.458) <i>həwit</i>	[hɪ.'vit]	I was (II.16, III.5, V.21)
(2.459) <i>həwin</i>	<i>i</i> [hɪ. ['] viː.ni]	we were (II.22)
(2.460) <i>həziti</i>	[hɪ.ˈziː.ti]	I saw him (III.21)
(2.461) <i>bənili</i>	[bɛ.ˈniː.li]	he built it (IV.3)
(2.462) <i>məti</i>	[mɪ.ˈθiː]	lift! (IV.108)
(2.463) <i>yəmi</i>	[jɪ.ˈmiː]	swear! (IV.120)

 $\vartheta > \varepsilon / C\varepsilon$: / ϑ / is often fronted and raised to [I] or [ε] before an accented syllable containing /e/:

147 Malone ("Mandaic Phonology," 144) refers to this process as 'schwa coloring.'

(2.464) <i>əmer</i>	[ɛ.ˈmɛɹ]	I would say (I.3, II.1)
(2.465) nešemti	[nɪ.ˈʃɛm.ti]	a soul (IV.96)
(2.466) <i>pəherye</i>	[pɛ.ˈhɜɪ.je]	it is flying (V.21)
(2.467) gəhek	[gɛ.ˈhɛɣ]	he laughed (V.26, X.17)
(2.468) <i>yəhem</i>	[jɛ.ˈhɛm]	he sat (VI.2, VII.2)

 $\vartheta > o / Cu: /\vartheta/$ is occasionally backed and raised to [0] or [A] before an accented syllable containing /u/:

(2.469)	guš əbud	[gu∫ ʌ.ˈwud]	take a look (f.)! (IV.46)
---------	----------	--------------	---------------------------

 $\vartheta > o / Co: /\vartheta/$ is often backed and raised to [o] or [u] before an accented syllable containing /o/:

(2.470) <i>əm-ork</i>	[o.'mo.tk]	with (I.5)
(2.471) yəhomyon	[jo.'hom.jon]	you sit (I.5)
(2.472) qamdəkon	[qam.ˈdo.χõ]	to/for you (X.7)

 $\vartheta > a$ / _Ca: / ϑ / is often fronted and lowered to [a] or [æ] in the environment of /a/:

(2.473) <i>ə<u>k</u>al</i>	[a.ˈxɑl]	he ate (V.11)
(2.474) <i>təlab</i>	[t ^s a.'lab]	he asked (X.1)

 $\vartheta > \vartheta / C\vartheta$: / ϑ / is often backed and lowered to [ϑ] or [ϑ] in the environment of / ϑ /:

(2.475)	qəbānen	[qɔ.ˈbɔː.nɛn]	they build (IV.17, IV.42)
(2.476)	xabar məṭā	[xa.'bæ.ı mɔ.'t ^s ɔː]	the news arrived (IV.145)
(2.477)	qəbāyi	[qɔ.'bɔ:.ji]	he wants (V.2)

 $\vartheta > u / w$: Similarly, / ϑ / often raises and backs to [Λ] when it is followed by [w]:

(2.478)	həwā	[hʌ.'wɔː]	he was (II.3, VIII.1, X.6)
(2.479)	həwālu	[hʌ.ˈwɔː.lu]	they had (II.29)
(2.480)	rəwāhā	[rʌ.ˈwɔ.hɒ]	spirit (II.34)
(2.481)	šəbābāni	[∫ʌ.wɔ.ˈwɔː.ni]	his neighbors (X.1)
(2.482)	šəbaq	[∫ʌ.ˈwaq]	he left (X.13)

2.7.2.3 Anaptyxis

Anaptyctic /ə/ is often employed to break up clusters of /b/ and /r/, e.g. (2.483) and (2.484).

(2.483)	əmrānkon	[əm.bə.ˈrɒn.χon]	your sheep (IV.124)
(2.484)	obri	['o.bə.ri]	a mouse (V.8 and V.9)

An anaptyctic /a/ will often intervene in a consonant cluster composed of a consonant followed by the palatal approximant /j/, even across a syllable boundary, if the syllable containing /j/ is accented. In this environment, it is always realized as [I]:

(2.485)	qəšaryāte	[qə.∫a.rı.′jɔ:.te]	you will free me (IV.119)
(2.486)	qəmwadyātan	[qm. ₁ væ.dı. ¹ jɔ:.tã]	you will take us (IV.118)

On rare occasions this same anaptyctic /2/ will intervene even when the following syllable is not accented, e.g. (2.487) and (2.488).

(2.487) pəradyon	[pə.ˈra.dɪ.jon]	they fled (IV.105)
(2.488) <i>amtyu</i>	[ˈam.t ^s ɪ.ju]	he brought them (IV.144)

2.7.2.4 Word-Final Reduction

Lax vowels in word-final environment, such as the /a/ in (2.489) and the /e/ in (2.490), (2.491), and (2.492).are occasionally reduced to /ə/.

(2.489)	Abdalla	[Sab.'dał.tə]	Abdallah (II.8 and II.27)
(2.490)	тиууе	[ˈmʌj.jə]	what is it? (IV.106)
(2.491)	barnašāne	[ˌbaɪ.nɛ.ˈʃɔː.nə]	my people (IV.118)
(2.492)	mehman-de	[m31.ˈmæn.də]	my guest (V.16)

3. The Noun

The noun phrase (NP) consists of a noun (or pronoun; see Chapter 4) and any associated modifiers. Nouns¹⁴⁸ serve to identify a person, e.g. (3.1), place, e.g. (3.2), thing, e.g. (3.3), e.g. quality (3.4), or action, e.g. (3.5).

(3.1)	ganzibrā	ganzibra
(3.2)	Šuštar	Shushtar
(3.3)	mienā	water
(3.4)	hitruktā	happiness
(3.5)	əbādā	work

In Neo-Mandaic, nouns may be identified by the following attributes:

- They can be inflected for masculine or feminine gender (see §3.2), singular or plural number (§3.4), identifiability and referentiality (§3.6).
- They can be syntactically or morphologically marked as subjects (see §3.5.1), objects (§3.5.2) and other complements (§3.5.4) of verbs.
- They do not inflect for tense and aspect (§5.2), or mood (§5.3).
- They can be incorporated into the meaning of a verb (§3.5.2) and serve as the preverbal element in phrasal verbs (§5.1.1.4).
- They can be modified by other nouns in apposition (§3.7.1), adjectives (§3.7.2.1), and ordinal numbers (§3.7.3.1), all of which immediately follow the contextual form of the noun they modify. Cardinal numbers (§3.7.3.1) and most other quantifiers precede the noun.
- In an existential sentence, predicate nominals are indicated by the independent copula rather than the clitic form (§5.1.1.4).
- They constitute an open class, i.e. one that can be expanded through compounding, derivation, coining, borrowing, etc.
- 148 Unless otherwise indicated, the term 'noun' refers specifically to noun substantives, and 'adjective' refers to noun adjectives.

3.1 Structure of the Noun

Nouns may be inflected for gender, number and/or referentiality. The structure of the noun is as follows:

```
STEM \pm GENDER SUFFIX \pm NUMBER SUFFIX \pm INDEFINITE SUFFIX or \pm AUGMENT
```

The basic lexical form of most Neo-Mandaic nouns is that of the stem and the augment $-\bar{a}$, the presence or absence of which is determined largely by phonological factors (see §2.5.5.2). The feminine suffix (see §3.2 below) follows immediately after the stem. It is followed in turn by the plural morphemes (see §3.3 below), which precede the augment. The indefinite morpheme (see §3.6.3 below) takes the place of the augment. Non-nativized loan words possess the same structure as other nouns, but dispense with the augment.

Notwithstanding the influence of other languages upon the lexicon of Neo-Mandaic, the vast majority of noun stems may be organized according to both root and pattern, as in other Semitic languages. The former is the sequence of consonants (generally three in number, although two or four are not uncommon) that stay constant across a set of nouns and verbs that also share meanings within some semantic field. The latter consists of those characteristics which are independent of its root or any additional morphemes (such as those marking number and gender). These include the sequence of vowels within the stem, known as its "melody," and the arrangement of consonants and vowels, which is called a "template."¹⁴⁹ While the internal pattern system of the noun is not nearly as consistent or well-understood as that of the verb, it is nonetheless significant from a morphological perspective as an important means of derivation.

¹⁴⁹ Despite the synchronic nature of this description, I have adopted the terminology used by Joshua Fox in his diachronic analysis of the noun pattern across the Semitic languages, *Semitic Noun Patterns*, Harvard Semitic Series 52 (Winona Lake, IN: Eisenbrauns, 2003). This terminology is defined on pp. 37–46.

3.1.1 Unaugmented Patterns

Traditionally, the citation form for the various patterns is represented by a dummy root based upon three "strong" radicals (i.e. those radical consonants that remain unaffected by the inflection of the root), which is modified by the addition or subtraction of radicals to represent biradical or quadriradical roots. For the purposes of this description, the paradigmatic root will be *g*-*t*-*l* for triradical roots, *g*-*l* for biradical roots, and *g*-*t*-*l*-*d* for quadriradical roots. Note that the glottal stop /?/ (which is not transcribed when it appears in initial position) may also stand for the initial radical *g*. "Augmented patterns" (see §3.1.2 below) are those in which additional afformatives have been affixed to the original root. Far more numerous are the basic "unaugmented patterns," in which no consonants other than the radicals are found.

3.1.1.1 The CVC- Template

The monosyllabic biconsonantal template comprises a large number of nouns, many of which are of foreign origin (predominantly Persian and Arabic).

gal-

Most of the nouns belonging to this pattern are of foreign origin, predominantly from Persian.

(3.6)	gač	gypsum (Persian)
(3.7)	par	leaf (Persian)
(3.8)	pas	pass; (English)

gāl-

Many of the words belonging to this pattern are of Mandaic origin:

(3.9)	bābā	door	(3.12)	rāzā	secret
(3.10)	hālā	sand	(3.13)	tābā	good
(3.11)	qālā	sound			

There are also a substantial number of words of foreign origin:

(3.14)	bāz	hawk (Persian)
(3.15)	māh	month (Persian)

3. T	The Noun
------	----------

(3.16)	māl	property (Arabic)
(3.18)	sāl	year (Persian)
(3.19)	yād	memory (Persian)

gil-

Most of the words belonging to this pattern are of Mandaic origin:

(3.20)	diwā	demon	(3.24)	inā	eye
(3.21)	hilā	power	(3.25)	rihā	smell
(3.22)	ibā	cloud	(3.26)	rišā	head
(3.23)	idā	hand	(3.27)	zipā	lie

There are also a few words of foreign (mostly Persian) origin:

(3.28)	miwā	fruit (Persian)
(3.29)	tiz	sharp (Persian)

Within certain phonetic environments (see §2.5.3.5 for details), this pattern becomes *giel*-:

(3.30)	bie <u>tā</u>	house
(3.31)	pienā	evening
(3.32)	mienā	water
(3.33)	šiex	sheikh (Arabic)

gel-

Nearly all of the words belonging to this pattern are of foreign origin:

(3.34)	geš	all (possibly Arabic)
(3.35)	kef	good humor (Arabic kayf)
(3.36)	ler	spittle (Persian layr)
(3.37)	ser	numb (Persian sayr)
(3.38)	tel	hill (Arabic)
(3.39)	wel	detached (Persian)

Another example of this pattern is derived from a biradical root:

(3.40)	dəmā	blood (contextual <i>dem</i>)
(3.40)	uəma	biobu (contextual <i>deni</i>)

Two words of Persian origin appear to be examples of this pattern extended with the nominal augment $-\bar{a}$, but actually belong to a separate pattern, *gelā*, the final vowel of which is inseparable:

(3.41)	nedā	proclamation (Persian)
(3.42)	šenā	swimming (Persian)

gol-

This pattern includes words of both native and foreign origin:

(3.43)	kol	all
(3.44)	nodā	earthquake
(3.45)	ho <u>k</u> ā	scratch
(3.46)	hol	push (Persian)
(3.47)	pol	bridge (Persian)
(3.48)	gol	flower (Persian)

gul-

This pattern also includes words of diverse origin:

(3.49)	bu <u>t</u> ā	prayer
(3.50)	nurā	fire
(3.51)	țurā	mountain
(3.52)	yumā	day
(3.53)	guš	ear (Persian)
(3.54)	ruz	day (Persian)

3.1.1.2 The CVCC- Template

The monosyllabic triradical template is the most common template in the Mandaic language. Whenever the augment is deleted, resulting in a final consonant cluster, an epenthetic /a/ is generally inserted between the second and third radicals of the root (see §2.4.2.5 for more details).

gațl-

This pattern is one of the most common in the Mandaic language:

(3.55)	alpā	thousand	(3.64)	mațrā	rain
(3.56)	bazrā	seed	(3.65)	nahrā	river
(3.57)	gabrā	man	(3.66)	napšā	self
(3.58)	gațrā	smoke	(3.67)	palgā	split
(3.59)	kabdā	liver	(3.68)	radpā	persecution
(3.60)	karsā	belly	(3.69)	rațnā	dialect
(3.61)	kaspā	silver	(3.70)	šamnā	fat
(3.62)	mandā	knowledge	(3.71)	tafšā	stewpot
(3.63)	manzā	hair	(3.72)	talgā	snow

Also included in this pattern are numerous words of foreign origin, particularly Arabic loan words:

(3.73) <i>ahl</i> peop	le (Arabic)
(3.74) <i>alm</i> know	vledge (Arabic)
(3.75) 'aqəl mind	(Arabic)
(3.76) <i>čang</i> harp	(Persian)
(3.77) <i>jadrā</i> root	(Arabic)
(3.78) man'a proh	ibition (Arabic)
(3.79) <i>mand</i> mau	nd (Persian)
(3.80) <i>narmā</i> smoo	oth (Persian)
(3.81) qarn horn	(Arabic)
(3.82) rag'a patch	n (Arabic)
(3.83) <i>waxt</i> time	(Arabic via Persian)
(3.84) <i>xarj</i> expe	nditure (Arabic)

A subset of vocabulary based upon this pattern contains a semivowel as its second (*gawl-* and *gayl-*) or third (*gatw-* or *gaty-*) radical. In the latter case, the semivocalic radical becomes a full vowel when the augment is lost (see §2.4.2.5 for more details):

(3.85)	țawšā	dirt
(3.86)	<i>sayd</i>	hunt (Arabic)
(3.87)	kasyā	secret
(3.88)	mašwā	rope
(3.89)	patyā	wide

Another subset of this pattern, *gall*-, is characterized by a reduplicated final radical, which is simplified when the augment is lost (see §2.6.2.1 for further details):

(3.90)	ahhā	brother	(3.93)	hattā	new
(3.91)	babbā	grandfather	(3.94)	rabbā	great
(3.92)	gappā	wing	(3.95)	yammā	sea

gātl-

This is a small pattern comprising words of both native and foreign origin:

(3.96)	ālmā	world
(3.97)	rāst	true (Persian)
gețl-

Another common monosyllabic pattern, primarily comprising native vocabulary:

(3.98)	besrā	flesh	(3.106)	germā	bone
(3.99)	deqnā	beard	(3.107)	lebbā	heart
(3.100)	deštā	ground	(3.108)	mehlā	salt
(3.101)	ebrā	son	(3.109)	mendā	thing
(3.102)	emrā	sheep	(3.110)	meškā	skin
(3.103)	enšā	women	(3.111)	peršā	tomorrow
(3.104)	ešmā	name	(3.112)	serrā	moon
(3.105)	genzā	treasure	(3.113)	šeršā	religion

A few foreign words also belong to this pattern:

(3.114)	fekər	thought (Arabic)
(3.115)	perzā	few (Persian)
(3.116)	rezq	livelihood (Arabic)
(3.117)	šekəl	appearance (Arabic)

As in §3.1.2.1 above, a subset of vocabulary based upon this pattern contains a semivowel as its second or third radical. In the latter case, the semivocalic radical becomes a full vowel when the augment is lost (see §2.4.2.5 for more details):

Finally, another subset of this pattern, *gell*-, is characterized by a reduplicated final radical:

(3.120)	bellā	husband	(3.125)	heyyi	life
(3.121)	<i>beyyā</i>	egg	(3.126)	šennā	tooth
(3.122)	demm ā	tail	(3.127)	šerrā	war
(3.123)	emmā	mother	(3.128)	šettā	year
(3.124)	ettā	woman			

gițl-

The number of nouns belonging to this pattern is relatively small, and contains words of both foreign and native origin:

(3.129)	gird	round (Persian)
(3.130)	hitrā	rejoicing
(3.131)	jisrā	bridge (Arabic)
(3.132)	mimrā	literature

3. The Noun

(3.133)	mișrā	magical boundary
(3.134)	sidrā	book

Likewise, a subset of vocabulary based upon this pattern contains a semivowel as its second or third radical.

(3.135) *lilyā* night

goțl-

This is another small pattern primarily comprising words of native origin:

(3.136) gobrā	men	(3.140) <i>obrā</i>	mouse
(3.137) gomlā	camel	(3.141) <i>ohrā</i>	road
(3.138) <i>ko<u>k</u>bā</i>	star	(3.142) <i>soprā</i>	bird
(3.139) <i>logrā</i>	leg		

Another subset of this pattern, *goll*-, is characterized by a reduplicated final radical:

(3.143) onnā ear (3)	3.144)	pommā	mouth
----------------------	--------	-------	-------

guțl-

This small pattern comprises words both of native and foreign origin:

(3.145)	<i>'umər</i>	age (Arabic)
(3.146)	burkā	knee
(3.147)	dust	friend (Persian)
(3.148)	dukkā	place
(3.149)	ġubšā	twilight (Arabic)
(3.150)	ġușnā	branch (Arabic)
(3.151)	ḥusən	beauty (Arabic)
(3.152)	kušțā	truth
(3.153)	șun'a	benefit (Arabic)
(3.154)	turtā	cow
(3.155)	tulyā	worm
(3.156)	wușle	piece (Arabic)

3.1.1.3 The CVCVC- Template

The first of the bisyllabic triradical templates comprises a wide variety of patterns. The first syllable of the template most commonly contains the vowel /a/, although all other vowels are attested. Most of the words

belonging to this template are of foreign origin, as a result of the reduction of short vowels in the pretonic syllable (see \$2.6.2.2), which yields the *gtvl*- template (see \$3.1.1.4 below).

gațal-

This pattern is composed of loan words from Arabic and Persian:

(3.157)	nafár	individual (Persian)
(3.158)	safár	travel (Arabic via Persian)
(3.159)	xabár	news (Arabic via Persian)

gațāl-

This pattern is also composed of loan words for the most part, although there are a few words of native origin:

(3.160)	<i>ʿajā</i> j	fog (Arabic)
(3.161)	haṭāyā	sin
(3.162)	jawāb	answer (Arabic)
(3.163)	qanāyā	smith
(3.164)	qarār	decision (Arabic)
(3.165)	sawār	rider (Persian)
(3.166)	șabāķ	morning (Arabic)
(3.167)	tamām	perfection (Arabic)

gațil-

As with §3.1.3.2 above, this pattern also corresponds to loan words for the most part, although there are a few words of native origin, of which all are adjectives:

3. The Noun

Within certain phonetic environments (see §2.5.3.5 for details), this pattern becomes *gatiel*-:

(3.179)	rafieq	friend (Arabic)
(3.180)	hatieqā	old (inanimate objects)

gațol-

This pattern corresponds to a small number of nouns of native origin:

(3.181) hayotā	survival	(3.182) <i>yaroqā</i>	green
----------------	----------	-----------------------	-------

gațul-

This pattern corresponds to a small number of nouns of both foreign and native origin:

(3.183)	harubā	ruined
(3.184)	qabúl	acceptance (Arabic via Persian)
(3.185)	yalupā	scholar

gāțel-

This pattern is most commonly used to derive the G-stem active participle from verbal roots (for which, see §5.1.1.2). Apart from those nouns which are derived from the active participle, nearly all nouns belonging to this pattern are of foreign origin:

(3.186)	gāmeš	buffalo (Persian)
(3.187)	hāzér	present (Arabic via Persian)
(3.188)	<u> ḥā</u> kem	governor (Arabic)
(3.189)	kāfer	infidel (Arabic)
(3.190)	<u>șā</u> ḥeb	owner (Arabic)
(3.191)	šāmeš	sun
(3.192)	tājér	merchant (Arabic via Persian)

gāțil-

This pattern corresponds to a small number of nouns of foreign origin:

(3.193)	āxír	last (Arabic via Persian)
(3.194)	gāwi	strong (Arabic)
(3.195)	qābin	marriage (Middle Persian)

gețāl-

This pattern corresponds to nouns of foreign and native origin:

(3.196)	welāt	country (Arabic)
(3.197)	<u> hiṣān</u>	horse (Arabic)
(3.198)	<u>ḥizā</u> m	belt (Arabic)
(3.199)	lešānā	tongue

gețel-

This pattern is represented by two words, one of which is augmented by the *-t-* affix:

(3.200)	metelā́	stories
(3.201)	nešemtā	soul

gițāl-

This pattern is represented by one word, ultimately of Persian origin:

(3.202)	diwān	scroll (Persian)
---------	-------	------------------

gițel-

This pattern is represented by one word, ultimately of Arabic origin:

(3.203) *jihel* child (Arabic)

goțal-

This pattern corresponds primarily to nouns of foreign origin; note that the final vowel in (3.205) is inseparable, being part of the pattern, and so its pattern is technically *goțalā*, although it is synchronically indistinguishable from other plural nouns ending in the plural morpheme $-\dot{a}$:

(3.204) <i>komák</i>	help (Persian)
(3.205) <i>olamā́</i>	Islamic scholars (Arabic via Persian)
(3.206) <i>qolastā</i>	collection

goțil-

Both native and nativized words belong to this pattern:

3. The Noun

(3.207)	honinā	little
(3.208)	horinā	other
(3.209)	fonidā	fish (origin uncertain)

goțol-

This pattern corresponds to a small number of nouns of Persian origin:

(3.210)	dorós	correct (Persian)
(3.211)	sotór	razer (Persian)

goțul-

This pattern corresponds to a small number of nouns of native and foreign origin:

(3.212)	okumā	black
(3.213)	poquttā	neck (origin unknown)
(3.214)	zobún	shirt (Persian)

guțāl-

This pattern corresponds to a small number of nouns of Arabic origin:

(3.215)	ġubār	dust (Arabic)
(3.216)	muḥāl	absurd (Arabic)

3.1.1.4 The CCVC- Template

The initial consonant cluster at the beginning of this template is a result of the reduction of short vowels in the pretonic syllable (for which, see §2.6.2.2), affecting words originally belonging to the *gvtvl*-template (see §3.1.3 above). Consequently, nearly all of the words pertaining to this template are of native origin. As Neo-Mandaic does not tolerate most word-initial consonant clusters (see §2.4.2.5), an anaptyctic /ə/ is often inserted between the two consonants to break up the cluster.

gțāl-

This is a very common pattern corresponding almost entirely to substantives and adjectives of native origin:

(3.217)	bəțālā	bad	(3.223)	qəmāšā	clothing
(3.218)	əbādā	work	(3.224)	səmāqā	red
(3.219)	ənāšā	people	(3.225)	smālā	left
(3.220)	gəlālā	stone	(3.226)	sțānā	boy
(3.221)	gəțānā	short	(3.227)	šəbābā	neighbor
(3.222)	kədābā	book	(3.228)	šənārā	tom cat

A few examples of this template are built upon a root with a weak second or third radical; this radical behaves in precisely the same manner as the strong radicals:

(3.229)	həwārā	white	(3.232)	qəyānā	culture
(3.230)	həyānā	alive	(3.233)	rəwāhā	relief
(3.231)	kərāyā	leg			

When the second syllable becomes closed (for example, through the addition of the *-t-* affix), its vowel is unexpectedly reduced to /a/ despite the fact that the accent remains on the closed syllable (see §2.3.2):

(3.234) <i>gəlaltā</i> stone	(3.235) <i>šənartā</i>	queen cat
------------------------------	------------------------	-----------

Unlike most other noun patterns, this pattern is still "productive" in Neo-Mandaic, in the sense that terms for professions originally belonging to the Arabic fa " $\bar{a}l$ pattern are often "nativized" by fitting their root into this pattern:

(3.236) <i>hədādā</i> blacksmith $(3.$.237)	nəjārā	carpenter
--	-------	--------	-----------

gțil-

This pattern is generally associated with the *G*-stem passive participle (see §5.4.3.1), as well as many adjectives and a few nouns, often (but not necessarily) derived from the passive participle. Like *gāțel-* and *gțāl-* above, it continues to be productive in Neo-Mandaic. The word *səmikā* 'thick,' which does not appear in Classical Mandaic, appears to be derived from the root $\sqrt{s-m-k}$.

(3.238)	əlimā	fat (adj.)	(3.242)	nəhirā	nose
(3.239)	ərikā	long	(3.243)	səmikā	thick
(3.240)	ərqihā	heaven	(3.244)	səri	rotten
(3.241)	məsi	cold (adj.)	(3.245)	šlihā	apostle

A perfect example of the productivity of this pattern was illustrated for me when Nasser Sobbi and I were discussing some of the texts in Macuch's *Neumandäische Texte*. We came upon the Arabic word *tayyāra* 'airplane,' which is ultimately derived from the root $\sqrt{t-y-r}$ 'to fly.' Sobbi shook his head and said to me, "I don't understand why he used that word. We have a perfectly good Mandaic word – *pəhirā*." The Mandaic root $\sqrt{p-h-r}$ corresponds in meaning to the Arabic root $\sqrt{t-y-r}$. Clearly, *pəhirā* is a neologism, as neither the word nor its referent exist in Classical Mandaic.

gțul-

Few examples of this pattern are attested, and fewer still are based upon roots with three strong radical consonants. One of them, *səbu'a* 'week,' is based upon an Arabic root. The word *yəmutā* 'oath,' is not attested in Classical Mandaic, even though it is clearly derived from the root \sqrt{y} -*m*-^{*w*}/_{*y*} which means 'to swear;' the regular word for oath in Classical Mandaic is '**umamata**. This suggests that this pattern, though rare, may still be somewhat productive.

(3.246)	bəzuyā	hole
(3.247)	səbuʿa	week (Arabic)
(3.248)	yəmu <u>t</u> ā	oath

3.1.1.5 The CVCCVC- Template

This marginal template corresponds to a very small number of nouns, which are for the most part of foreign origin.

gațțal-

This pattern is here represented by one word of Arabic origin:

(3.249) *awwál* first (Arabic via Persian)

gațțāl-

This pattern is also represented by one word of Arabic origin; note terms referring to professions or occupations, which usually

correspond to this pattern in Arabic, generally become reformed along the analogy of the *gtāl*- pattern described in §3.1.4.1 above.

(3.250) *hammām* bathhouse (Arabic)

gațțel-

Another pattern represented by a single word, in this case of indeterminate origin. Macuch and Drower derive this word from the Classical Mandaic root **HUQ 1** 'to be distressed' (q.v.); the possibility also exists that it may somehow be derived from the Aramaic root $\sqrt{-y-q}$ (id.) or the Arabic word *dayyiq* 'tight.'

(3.251) *wayyeq* distress (origin unknown)

3.1.1.6 Quadriradical Patterns

Due to the comparative paucity of quadriradical roots within the original Aramaic and Mandaic vocabulary, most of the words that correspond to these patterns are predictably of foreign origin.

gatlad-

army (Arabic)
forest (Persian)
thanks (possibly Persian)
cooking pot (Persian)
dervish (Persian)
near (Persian)
guest (Persian)

3. The Noun

gtald-

(3.259) dəraxttree (Persian)(3.260) ešgandāacolyte

gițlid-

(3.261) *Ingliz* English

3.1.2 Augmented Patterns

As mentioned above (§3.1.1), "augmented patterns" are those in which afformatives have been affixed to the original root. Some afformatives, such as the sufformatives $-\bar{a}y\bar{a}$, -i, and $-ut\bar{a}$, are applied to existing lexemes in such a way as to change their meanings in a consistent and predictable manner. Other afformatives do not appear to impart any consistent meaning to the newly derived lexeme.

3.1.2.1 Preformative m-

A number of nouns, particularly (but not exclusively) those of Arabic origin, bear the m- preformative, which is characteristic of deverbal nouns as well as what Macuch calls nouns of time, place, and instrument.¹⁵⁰

(3.262)	maḥallát	neighborhood (Arabic via Persian)
(3.263)	mambuhā	ritual water
(3.264)	maxșuș	especially (Arabic)
(3.265)	mas'ul	responsible (Arabic)
(3.266)	mașbettā	baptism
(3.267)	mašknā	sanctuary

3.1.2.2 Preformative t-

The *t*- preformative is characteristic of Arabic vocabulary, particularly verbal nouns, but is also found on some native vocabulary.

(3.268)	tabliġí	prosylete (Arabic via Persian)
(3.269)	tarmidā	a Mandaean priest

150 Macuch, Handbook, 188.

3.1 Structure of the Noun

3.1.2.3 Sufformative -t-

This is one of the most common and productive afformatives in any Semitic language. It often marks the feminine gender on a noun, and regularly serves as a suffix marking the feminine on adjectives, but can also appear as part of an augmented pattern without any clear derivational role. Due to the fact that this sufformative appears not only on originally Mandaic words but also on Arabic vocabularly, including those Arabic words which have been borrowed by way of Persian, it is represented by five separate allomorphs: -t, -t, $-\acute{at}$, -a, and -e. The first two allomorphs are characteristic of native vocabulary, and the following three allomorphs indicate Arabic loan words, often borrowed via Persian. The allomorph $-\acute{at}$ in particular indicates that the word to which it is affixed was borrowed from Persian rather than directly from Arabic.

-t-

(3.270) bərattā (3.271) dəholtā (3.272) duktā (3.273) gəlaltā	daughter fear place stone	· /	1	baptism soul collection cow
- <u>t</u> - (3.278) <i>hā</i> <u>t</u> ā (3.279) qamie <u>t</u> ā	sister noon	(3.280)	qəmol <u>t</u> ā	louse

-át

(3.281) (3.282) (3.283)		deposit (Arabic via Persian) incantation (Arabic via Persian) harm (Arabic via Persian)
(3.285) (3.286)	quwwát	neighborhood (Arabic via Persian) misery (Arabic via Persian) police station (Arabic via Persian) strength (Arabic via Persian) hour (Arabic via Persian)

126		3. The Noun
<i>-a</i> (3.289)	benāya	building (Arabic)
<i>-e</i> (3.290) (3.291) (3.292)	^s amme qahwe xāle	paternal aunt (Arabic) coffeehouse (Arabic) maternal aunt (Arabic)

3.1.2.4 Sufformative - ānā

Formally identical to the plural morpheme, this sufformative does not serve any consistent derivational role.

(3.293)	kenyānā	family name	(3.295)	țawšānā	dirty
(3.294)	šehyānā	ruler	(3.296)	zipānā	liar

3.1.2.5 Sufformative -āyā

This sufformative was formerly used to derive adjectives from proper nouns, common nouns, and ordinal numbers. While it is fairly common within the native vocabulary, it is no longer productive, its function having been assumed by sufformative -i (§3.1.2.6).

(3.297)	Šuštrāyā	Shushtary	(3.299)	ərbiāyā	fourth
(3.298)	Yəhuwəțāyā	Jewish			

The sufformative $-\bar{a}y\bar{a}$ regularly becomes $-ay\bar{a}n\bar{a}$ in the plural (see §2.5.2), but the original plural form inherited from Classical Mandaic, $-\bar{a}yi$, survives in a few isolated words:

(3.300)	Mandāyā	Mandaean
(3.301)	Mandayānā	Mandaeans
(3.302)	mašknāyi	another classical term for Mandaeans
(3.303)	nașorāyi	Nazoreans

The feminine singular form, which is sparsely attested, is -etā.

(3.306)	əle <u>t</u> ā	exalted	(3.307)	gāwetā	inner
---------	----------------	---------	---------	--------	-------

3.1.2.6 Sufformative -í

This sufformative corresponds to several foreign morphemes, including the Arabic $y\bar{a}^2$ an-nisba. It fulfills the same role as the $-\bar{a}y\bar{a}$ sufformative (§3.1.2.5) and is most commonly found on words of foreign origin, although it is occasionally found on words of native origin, sometimes in conjunction with the sufformative $-\bar{a}y\bar{a}$ as demonstrated by the example of *Mandāyí* 'Mandaic' below:

(3.308)	amrikí	American	(3.311)	Mandāyí	Mandaic
(3.309)	Fārsí	Persian	(3.312)	masiķí	Christian
(3.310)	Inglizí	English			

3.1.2.7 Sufformative - utā / -uxtā

The Classical Mandaic abstraction morpheme -*utā* serves to denote an abstract concept, much like the English morphemes *-ship* or *-ness*. It survives mainly on inherited vocabulary from the classical language, particularly in religious terminology:

(3.313)	asutā	healing
(3.314)	hemanu <u>t</u> ā	faith
(3.315)	malkutā	kingship
(3.316)	nașiruțā	esoteric knowledge
(3.317)	<u>țabutā</u>	banquet

This form of the abstraction morpheme is no longer productive. The productive form of this morpheme, *-uxtā*, is an allomorph produced by an unusual sound change affecting the voiceless interdental fricative $/\theta$ / when geminated (see §2.4.4.2). It is still used to generate vocabulary for abstract concepts in Neo-Mandaic:

(3.318) *bațluktā* misfortune (3.319) *guțluktā* massacre

3.2 Nouns and Gender

Neo-Mandaic distinguishes between masculine and feminine gender, which generally correspond to natural gender in animate beings. Inanimate objects are also divided into (grammatically) masculine and feminine gender, with the former predominating. For most nouns, this grammatical distinction is indicated through agreement rather than any specific marking. Comparatively few feminine nouns take the morphological marker of the feminine gender. For this reason, and the fact that the grammatical gender of a given lexical item is generally arbitrary, it is often difficult to ascertain whether a given lexical item is masculine or feminine, if it appears without modifiers.

The bulk of feminine nouns refer to logically feminine animate beings; examples include the following:

(3.320)	bərattā	daughter	(3.323)	hātā	sister
(3.321)	emmā	mother	(3.324)	turtā	cow
(3.322)	e <u>tt</u> ā	woman			

Furthermore, all paired body parts are considered to be feminine:

(3.325)	burkā	knee	(3.328)	kərāyā	foot/leg
(3.326)	idā	hand	(3.329)	logrā	leg
(3.327)	inā	eye	(3.340)	onnā	ear

Several non-paired body parts take feminine gender as well, e.g. (3.341) - (3.344); the remainder take masculine gender, e.g. (3.345) - (3.347).

(3.341)	hassā	back (f.)	(3.345)	appā	face (m.)
(3.342)	karsā	belly (f.)	(3.346)	nəhirā	nose (m.)
(3.343)	pommā	mouth (f.)	(3.347)	rišā	head (m.)
(3.344)	šennā	tooth (f.)			

While the majority of inanimate objects are masculine by default, a surprising number of them take feminine gender, e.g. (3.348) - (3.352). Loan words from Arabic generally retain their original gender, e.g. (3.353); loan words from Persian, which lacks grammatical gender, usually assume the logical gender, as in (3.354).

(3.348)	-	house (f.)
(3.349)		ground (f.)
(3.350)	-	book (f.)
(3.351)	ohrā	road (f.)
(3.352)	nurā	fire (f.)
(3.353)	šerrā	war (f.; possibly Arabic <i>širra</i> 'angriness')
(3.354)	bāz	hawk (m.; Persian <i>bāz</i> id.)

3.3 Nouns and Number

Neo-Mandaic nouns have two numbers: singular, which is unmarked, and plural, which is marked, depending on the referentiality of the noun. In addition, mass nouns are obligatorily unmarked.

3.3.1 Inflectional Plurals

Nouns are marked as plural by adding the suffix $-\bar{a}n$ - after the stem of the noun but before the augment. This suffix is the default plural marking for all nouns, animate and inanimate, masculine and feminine.¹⁵¹

(3.355)	tarmid-ā	disciple (m.)	tarmid-ān-ā	disciples (m.)
(3.356)	tur-ā	bull (m.)	tur-ān-ā	bulls (m.)
(3.357)	kədāb-ā	book (m.)	kəda <u>b</u> -ān-ā	books (m.)
(3.358)	kərāy-ā	leg (f.)	kəray-ān-ā	leg (f.)

3.3.2 Lexicalized Plurals

There are a number of feminine nouns that take the feminine plural suffix $-{\binom{w}{y}}\bar{at}$. This suffix is obligatory in the plural for all nouns marked with the sufformative *-t*- (see §3.1.2.3 above)

(3.359)	nešem-t-ā	soul (f.)	nešm-āṯ-ā	souls (f.)
(3.360)	tur-t-ā	cow (f.)	tur-āt-ā	cows (f.)

Other feminine nouns are not explicitly marked as feminine, but take the feminine plural morpheme nonetheless.

(3.361)	biet-ā	house (f.)	biet-wāt-ā	houses (f.)
(3.362)	dešt-ā	ground (f.)	dešt-āṯ-ā	lands; wilderness (f.)
(3.363)	hewān-ā	animal (f.)	hewan-yāṯ-ā	animals; cattle (f.)

In addition to the feminine plurals, some nouns form their plurals upon different nominal stems. These include both masculine and feminine nouns.

(5.504) $b = 1a - u - a$ dauginer $(1.)$ $b = 11 - a - a$ dauginers $(1.)$	(3.364)) bəra-tt-ā	daughter (f.)	bən-āt-ā	daughters (f.)
--	---------	-------------	---------------	----------	----------------

151 This suffix is likely the Iranian plural morpheme *-ān*, which was more productive in Middle Persian than it is today. Macuch, *Handbook*, 224–25 derives this morpheme from the rare Classical Mandaic suffix *-ania*, which appears primarily on plurals of plants and related products.

(3.365) <i>ett-ā</i>	woman (f.)	enš-ā	women (f.)
(3.366) <i>gabr-ā</i>	man (m.)	gobr-ā	men (m.)
(3.367) <i>hāt-ā</i>	sister (f.)	ah-wāt-ā	sisters (f.)

A few of these heteroclite plurals are optionally doubly marked with the default plural morpheme.

(3.368) <i>gabr-ā</i>	man (m.)	gobr-ān-ā	men (m.)
(3.369) <i>ett-ā</i>	woman (f.)	enš-ān-ā	women (f.)
(3.370) <i>turt-ā</i>	cow (f.)	tur-āt-ān-ā	cows (f.)

The masculine plural suffixes in prior stages of the language were -i in the emphatic state and -e in the construct. These suffixes are preserved in a few archaic terms and frozen expressions.

(3.371)	b=ešm-e=hon	əd=Heyy-i	Rabb-i	
	in=name-PL.CS=3PL	of=life-PL.ES	great-PL.ES	
	In the name of the Great I	he name of the Great Life. (I.1)		

3.3.3 Number Marking of Loan Words

Most loan words which have not yet been assimilated into the lexicon are marked not with the default plural morpheme $-\bar{a}n$ - but with the plural morpheme $-h\bar{a}$ ($-\bar{a}$ after consonants; see §2.4.2.5), which was also originally borrowed from Persian. The use of the Persian plural morpheme is perhaps a reflection of the fact that Persian is the largest source of foreign vocabulary in Neo-Mandaic, including many words ultimately of Arabic origin. Nevertheless, this suffix is also applied to words that were not borrowed from Persian. As this suffix never appears on original Mandaic vocabulary, its use explicitly marks a noun as a loan word, regardless of the etymological origin of the noun.

(3.372) ča	ng harp	čang-ā́	harps	(Persian)
(3.373) da	<i>rwiš</i> dervish	darwiš-	<i>á</i> dervishes	(Persian)
(3.374) <i>jih</i>	el child	jihel-ā́	children	(Arabic)
(3.375) rat	<i>fieq</i> friend	rafieq-a	friends	(Arabic)
(3.376) rez	q livelihoo	od <i>rezq-ā</i> ́	livelihood	ls (Arabic)
(3.377) <i>șu</i>	<i>bbi</i> Sabian	subbi-h		(Arabic)
(3.378) wa	<i>zir</i> vizier	wazir-ā	viziers	(Arabic)

A much smaller set of loan words retain the plurals from their source languages.

(3.379)	deh	village	deh-āt	villages	(Persian)
(3.380)	Ingliz-í	Englishman	Ingliz	Englishmen	(Arabic)
(3.381)	waxt	time	awqāt	times	(Arabic)

As foreign loan words become assimilated to the Neo-Mandaic lexicon, they take the default plural suffix -an, though often without the augment.

(3.382) <i>ahl</i>	family	ahl-ān	families	(Arabic)
(3.383) <i>amrik-í</i>	American	amrik-ān	Americans	(Arabic)
(3.384) <i>pərā</i>	money	pərāh-ān	id.	(Ottoman)

Much like other lexicalized plurals, some loan words are doubly marked as plural, through the addition of the default plural ending.

(3.385) zāģ chicken zeh-un-ān chickens (Persian)

The word *olamá* 'Islamic scholars' (originally from Arabic '*ulamã*', plural of '*ālim* 'scholar') has entered the Neo-Mandaic lexicon as a collective noun via Persian *olama*.

3.3.4 Noun Compounds and Number

When a noun is compounded with another substantive (noun or adjective), then the plural suffix $-\bar{a}n$ attaches to the end of the compound (3.386). The compound thus resembles the construct phrase (3.387), but is distinguished from a noun modified by an adjective (3.388), in which the plural suffix attaches to the head noun in the chain.

(3.386)	<i>daqqen</i> beard	<i>həwar-ān-ā</i> white-PL-AUG	elders (IV.31)
(3.387)	<i>bie<u>t</u></i> house	<i>Manday-ān-ā</i> Mandaean-PL-AUG	Mandaean houses (III.22)
(3.388)	<i>biet-wāt</i> house-PL.F	<i>ba'id</i> distant	distant houses (III.10)

3.4 Nouns and State

In addition to gender and number, the noun was formerly characterized by a third inflectional category, conventionally known as state. In Classical Mandaic, nouns appeared in three states: emphatic, absolute, and construct. Nouns in the emphatic state were distinguished from those in the absolute and the construct by the presence of the augment. These latter two were further distinguished in most nouns by different allomorphs. Because the presence of the augment is determined largely by phonological considerations rather than morphological ones (see §2.5.5.2), and the unaugmented forms of the noun (formerly corresponding to the construct and the absolute states) have been leveled to a single form, the states have given way to a new system contrasting lexical and contextual forms. Nevertheless, some vestiges of this inflectional category survive in Neo-Mandaic.

3.4.1 The Emphatic State

The augment, which served as a definite article in earlier stages of Aramaic, had ceased to play this role by the time of Classical Mandaic and merely served to indicate the emphatic state of the noun. The lexical forms (those possessing the augment) correspond to the Classical Mandaic emphatic state.

3.4.2 The Construct State and the Ezāfe

The construct state is indicated by the absence of the augment. Nouns in the construct state always appear in their contextual forms. Two or more nouns may thus be linked in a genitive relationship with a variety of functions. This construction is known as the construct chain. The first element in the construct chain, known as the head noun, may be modified by a noun substantive (3.389), a noun adjective (3.390), or even a second construct chain (3.391). Independent pronouns cannot modify a noun in this manner.

(3.389)	<i>ḥākem</i> governor	<i>welāt</i> city	4	governor of the city (IV.83)
(3.390)	<i>rab</i> leader	<i>Manday-ān-ā</i> Mandaean-PL-AUG		leader of the Mandaeans (II.2)
(3.391)	<i>riš</i> head	<i>yum</i> day	<i>šerš-an</i> religion-1PL	The start of our religion (I.7)

The construct chain may consist of any number of nouns, so long as the chain is not broken by any attributive adjectives or demonstrative pronouns, in which case the relationship between the nouns will be indicated by means of a preposition, as in (3.68).

(3.392)	siddər	rapt	əl=Manday-ān	
	book	great	to=Mandaean-PL	The great book of the Mandaeans.

Note that the noun modified by the adjective appears in the contextual form. Nouns modified by an attributive adjective resemble those in construct.

Traditionally, the construct chain was used to form Mandaean patronymics, which take the form "son/daughter of PN":

(3.393)	<i>Hadiyye</i>	bərat	Kanjar	
	PN	daughter	PN	
	Hadiyye, dau	ughter of Kan	jar (II.10)	
(3.394)	Čečāwā	bar	Čečāwā	
	PN	son	PN	
	Chechawa the son of Chechawa (IV.			

More recently, Mandaeans have started following the naming conventions of the countries in which they reside. The patronymic has become largely obsolete, although most observant Mandaeans continue to be given a matronymic *malwāši* name which follows the same format.

The Persian *ezāfe* construction (from the Arabic word *idāfa* 'addition') occasionally intrudes into Neo-Mandaic speech. Like the Neo-Mandaic contextual forms (but unlike the construct state of many Semitic languages), this morpheme indicates a relationship between two nouns (substantive or adjective) corresponding to a variety of functions (generally attributive or genitive):

(3.395)	kol	dukkān=e	Amrik-ān	
	all	store=EZ	American-PL	all the American stores (III.4)

Macuch identifies the *ezāfe* as an integral part of the subdialect of Ahvāz.¹⁵² The *ezāfe* has a much more restricted application in the speech of my informants.

3.4.3 The Absolute State

In Neo-Mandaic, the absolute state of the noun has become indistinguishable from the contextual forms of the noun in the emphatic state. Traces of the absolute state survive in the participles used to form the imperfective (see \$5.1.1.2), the subjunctive mood (see \$5.1.1.2), and the analytic passive voice (see \$5.1.1.4 and \$5.4.3.1), as well as the plurals of the cardinal numbers and nouns following ordinals (see \$3.7.3.1 below).

3.5 Nouns and Case-Marking

The grammatical function of a noun within a clause is indicated by word order, verb agreement, and the use of prepositions. Neo-Mandaic is a nominative-accusative language. The subject of an intransitive verb and the agent of a transitive verb are unmarked, and the verb agrees with them in number and gender. The object (or patient) of a transitive verb generally follows the verb and may be marked by a preposition under some circumstances (see §3.5.2). The verb does not agree in number or gender with its object in the perfective, unlike most other Eastern Neo-Aramaic dialects.

3.5.1 Subject of a Transitive or Intransitive Verb

The subject of a clause is not introduced by any preposition. The verb typically agrees with the number and person of the subject by verb inflections. Even when the subject is a generic noun (and hence not marked for plural, see §3.3 above), the verb still reflects a plural subject, as illustrated in (3.74) below.

- (3.396) *kərāy=i ser tamm-at* leg=3SG.M numb become.PFV-3SG.F His leg went numb. (VII.3)
- (3.398) *barnāš-ā qə=bān-en ha<u>t</u>tā jisər* person-AUG IND=build-3PL anew bridge The people rebuilt the bridge. (IV.42)

3.5.2 The Object of a Transitive Verb

The position of the object within the verb phrase is governed primarily by the specificity of the referent and secondarily by its identifiability (see also §3.6 below). If the referent is generic, the object is incorporated into the meaning of the verb, appearing before it and any adverbs or the negative morpheme.

(3.399)	<i>darwiš-ā</i> dervish-PL Dervishes [] ma	<i>azimát</i> invocatio ade invoca		<i>q=ābd-en</i> IND=do.IPFV-31	PL	<i>šam'</i> candle
	<i>qə=masrik-en u</i> IND=light.C.IPFV-3PL and lit candles, and drew [magic] bo		and	<i>mișr-ā</i> boundary-AUG laries. (IV.20)	<i>q=ābd-e</i> IND=do	

If the object is indefinite and non-specific (i.e. explicitly marked with the indefinite morpheme *-i*), it will also generally appear before the verb.

(3.400)	əzg-it	mend-i	əmhassel-nā
	go.PFV-1SG	thing-INDF	obtain.SBJV-1SG
	[] I would ha	we gone, to pick s	something up. (III.16)

(3.401)	jisr-i	tum	əbad-yon
	bridge-INDF	again	do.PFV-3PL

If the referent of the object of a verb is identified by the speaker as a specific, particular entity (i.e. as referential), it is introduced by the referential object marker ∂I - and an anticipatory pronominal suffix on the verb (which is generally elided before the referential object marker, when the referential object marker directly follows it). This

object marker is identical with the preposition ∂l - 'to/for' (from which it is derived).¹⁵³

(3.402)	<i>qə=mahreb-āt=ø</i> IND=be.destroyed.C.IPFV-2SG=3SG.M You keep destroying the bridge! (IV.58)	<i>əl=pol</i> REF=bridge
(3.403)	an $q=mandi-n=\emptyset$ I IND=shake.IPFV-1SG=3SG.F I will cause the earth to quake. (IV.98)	<i>əl=dešt-ā</i> REF=ground-AUG

The referential object marker ∂l - is a phrasal clitic; it attaches to the entire noun phrase, as is clear from examples (3.80) and (3.81).

(3.404)	<i>abd-onn=ø</i> do.PFV-3PL=3SG.M They held this conference	<i>əl=ā</i> REF=this . (I.5)	<i>konfərens</i> conference
(3.405)	<i>abd-ø=ø</i> do.PFV-3SG.M=3SG.M He created all of the world		<i>alm-ān-ā</i> world-PL-AUG

In some cases (generally before proper nouns or noun phrases), the proclitic ∂l - is optionally replaced by the independent preposition $\partial l \bar{a} w$ to.

(3.406)	hadet-onn=0 speak.D.IMP-3PL=3SG.M Discuss the religion, languag $q = y \bar{a} n$ u $mimr-\bar{a}$ culture and literature culture, and literature of the l			<i>əlāw</i> REF	<i>šerš</i> religion	<i>lešān</i> language
					<i>d=Manday-a</i> of=Mandaea 5. (I.5)	

(3.407) $zarz-at=\emptyset$ $\partial l\bar{a}w$ $\partial n\bar{a}\check{s}$ $AR\bar{A}M$ inform.PFV-3SG.F=3PL REF people PN She informed the members of ARAM. (I.6)

The referential object marker *əl*- does not appear on nouns that have been marked with a possessive suffix, possibly because these are already explicitly marked as referential.

153 Macuch (*Neumandäische Texte*, 95) identifies this marker as the Arabic definate article *al*-. He fails to explain why this particle marks the noun as definite when it is the object of a verb but not elsewhere.

(3.408)	ba'aden	ləbaš-ni	əqmāš-ān-an
	after	wear.PFV-1PL	clothes-PL-1PL
	Afterwards,	we put on our clothes.	(III.10)

3.5.3 Nouns and Phrasal Verbs

After a phrasal verb (see \$5.1.1.4), the direct object of the verb is generally marked with the preposition *qam* to/for. The oblique complements of the verb may be marked with *qam* (as in 3.409) or with another preposition (as in 3.410).

(3.409)	remedy	<i>abd-etton</i> do.IPFV-2PL	for=0		<i>qan</i> to	n <i>tā</i> that	<i>jisər</i> bridge
	You will f	fix that bridge for	or me. (IV	/.39)			
(3.410)	marriage	<i>əbad-yon</i> do.PFV-3PL ried someone to		<i>yeki</i> someone n. (II.10)	e	<i>qəlāw</i> to	<i>e<u>tt</u>=i</i> woman=INDF

3.5.4 Nouns as Complements and Adjuncts

In addition to appearing as the subject or object of the verb, the noun may also serve as the complement or adjunct of a noun or verb phrase. In these capacities, it generally appears as a constituent of a prepositional phrase.

The adjunct contributes non-essential information. It generally follows the phrase it modifies, whether noun, as in (3.411), or verb, as in (3.412).

- (3.411) *jihel-ā gāw maḥallát* child-PL in neighborhood the children in the neighborhood
- (3.412) *darwiš-ā a<u>t</u>-on min ger welāt* dervish-PL come.PFV-3PL from outside country Dervishes came from outside the country. (IV.20)

The oblique complement of a transitive verb is generally introduced by a preposition. For example, the preposition *b*- 'with' indicates accompaniment in (3.413), *ba'ad* 'after' indicates time in (3.414), *bienā* 'among' indicates manner in (3.413), *om*- 'from' indicates source in (3.418), *gāw* 'in' expresses instrument in (3.415), *hemke* 'like' indicates manner in (3.416), *qam* 'for' indicates destination in

3. The Noun

(3.420) and recipient in (3.421), $q\bar{a}r$ 'at' indicates destination in (3.418) and location in (3.419), and *tum* 'until' indicates duration in (3.415).

(3.413)	əmḥaṯṯ-én	bienā=d=an	be=l=ənāš-ā		
	speak.IPFV-1PL	among=of=1PL	with=NEG=people-AUG		
	Let's talk among ourselves, without anybody else. (IV.81)				

- (3.414) *ba^cad* so ruz kammar-ø after three day return.PFV-3SG.M After three days, he came back. (X.2)
- (3.415) *q=əmbarak-ø* haw-én həyānā tum āxír IND=pray.tD.IPFV-1SG be.SBJV-1PL alive until last I pray that we will remain alive until the final

ālm-āgāwhilHeyyi Rabbiworld-AUGinpowerPNepoch of time, by the power of Heyyi Rabbi. (I.14)

- (3.416) $q = h\bar{a}zy \cdot \bar{a}$ hemke $i\bar{b} \cdot \bar{a}$ $a_{\bar{t}}\bar{a} \cdot \phi$ IND=see.IPFV-3SG.F like cloud-AUG come.PFV-3SG.M She saw something like a cloud arrive. (IV.49)
- (3.417) *həwā-ø* ork=e yeki be.PFV-3SG.M with=1SG someone There was someone with me. (III.8)
- (3.418) *at-at əm=Baġdād qār Šiex Abdalla* come-3SG.F from=PN to Sheikh PN She had come from Baghdad to Sheikh Abdallah's place. (II.27)

- (3.421) genz əbād abd-at qam Manday-ān-ā much work do.PFV-3SG.F for mandaean-PL-AUG She did much work for the Mandaeans. (I.6)

The complement of a verb of desire or communication (e.g. ask, desire, demand, suggest, request) is introduced by the preposition *min* 'from' or its enclitic form *əm-*.

3.6 Identifiability and Referentiality

- (3.422) *q=abi-n* min Heyyi Māre əmzahar-ø=lə=kon IND=want.PTC-1SG from PN protect.PTC-3SG.M=OBJ=2PL I want Heyyi Māre to protect you.¹⁵⁴
- (3.423) $q = m \check{s} i e l \emptyset$ $= m = j i h e l \bar{a}$ $= m h \underline{a} \underline{t} \underline{t} \emptyset$ Mandāyí IND=ask.D.PTC-3SG.F from=child-PL speak.SBJV-3SG.F Mandaic She asks the children to speak Mandaic. (II.15)
- (3.424) *təlab-ø* əm=šəbāb-ān=i qazġān-i request.PFV-3SG.M from=neighbor-PL=3SG.M cauldron-INDF He asked his neighbors for a cooking pot. (X.1)

3.6 Identifiability and Referentiality

Identifiability (also known as "definiteness") and referentiality are two pragmatic statuses that are marked in Neo-Mandaic. For the purposes of this discussion, the term "definite" with reference to the noun phrase implies that its referent is *identifiable* to both the speaker and listener. In English, definiteness is indicated appropriately enough by the definite article, as in (3.425) and (3.426).

(3.425) the man (3.426) the woman

By contrast, the term "indefinite" implies that the referent is *not identifiable* to the listener. This is indicated by the indefinite article 'a' / 'an' or the quantifier 'some.' Examples include (3.427) and (3.428).

(3.427) a man (3.428) some men

The referentiality of a noun phrase concerns whether its referent is a specific entity (and therefore considered referential) or a non-specific entity or an entire category of entities (in which case it is considered to be non-referential); this is known as objective referentiality. If the referent has already appeared in the discourse, it must also be considered referential whenever it is subsequently mentioned, regardless of its initial pragmatic status; this is known as discourse referentiality.¹⁵⁵ In English, the entity indicated as indefinite may be

¹⁵⁴ Cf. de Morgan, Mission Scientifique V, 282, In. 8-9

¹⁵⁵ Thomas E. Payne, *Describing Morphosyntax: A Guide for Field Linguists* (Cambridge: Cambridge, 1997), 263–66.

unique or merely a member or members of a category. In the former case, the referent is said to be specific; in the latter, it is said to be non-specific. The definite article, by contrast, may indicate that the referent is specific, or that it is generic, representing an entire category. In count nouns (unlike mass nouns such as "water," "rice," "money," etc.), generic referents may be indicated either by the plural of the noun without the definite article or the singular with the definite article, e.g. (3.429) and (3.430).

(3.429) Croissants were invented in 1689

(3.430) The croissant was invented in 1689

Both examples refer to the entire category of croissants rather than any particular croissant. Note that identifiability is similar to, but not identical with referentiality; while definite (or identifiable) entities are commonly referential, and indefinite (or unidentifiable) entities are often non-referential, they need not be, as the example of the crossaint suggests.

3.6.1 Nonmorphological Marking of Referentiality

Most unmarked noun phrases are ambiguous as to referentiality and may refer to specific entities, as in (3.431) and (3.432), or generic categories, as in (3.433).

(3.431)	<i>ganzibr-ā</i> ganzibra-AUG The ganzibra is in		<i>Moḥam.</i> PN nshahr.	rā	<i>e<u>k</u>t=i</i> COP=3S	SG.M
(3.432)	<i>hab-t=ell=i</i> give.PFV-1SG=0 I gave it to the ju		.M	<i>qam</i> to	<i>ḥākem</i> judge	
(3.433)	<i>barnāš-ā</i> person-AUG People came from	<i>at॒-on</i> come.Pf n outside		<i>min</i> from try.	<i>ģer</i> outside	<i>welāt</i> country

There are, however, a number of words in Neo-Mandaic, such as proper names, demonstrative pronouns, and personal pronouns, that necessarily refer to an objectively referential entity. Independent of its position within the discourse, a noun phrase must be interpreted as objectively referential:

• if it is modified by a demonstrative adjective (3.434), a possessive pronoun (3.435), or an ordinal number (3.436);

(3.434)	<i>ā</i> this	<i>ett-ā</i> woman-AUG	this woman (II.24)
(3.435)	<i>bie<u>t</u>=e</i> house=1	SG	my house (IV.118)
(3.436)	<i>awwál</i> first	<i>barnāš</i> person	the first people (I.11)

• if it is marked as plural with the plural morpheme -ān;

(3.437)	manday-ān-ā		
	mandaean-PL-AU	Ĵ	the Mandaeans (I.3)

- if it is the head noun of a construct chain (see §3.4.2);
- if it is followed by a restrictive relative clause (see §4.3.3.2).

3.6.2 Morphological Marking of Referentiality

The enclitic preposition ∂l -, introduced above in §3.5.2, marks the referent of the object of a verb as specific. Generally, it also indicates that the referent is identifiable, as well; its primary function, however, is to identify the referentiality of the object, as the following examples will demonstrate. In (3.458), the relative pronoun *man* 'who' is used to represent a particular individual (the one whom). In this example, the referent is both specific and definite, and therefore it takes the referential object marker ∂l -.

(3.438) $g = t - \phi = \phi$ a = man $q = rahem - \phi = l = i$ kill.PFV-3SG.M=3SG.M REF=who IND=love.IPFV-3SG.M=OBJ=3SG.M He killed the one whom he loves.

In (3.439), the direct object is the indefinite pronoun *hemman* 'whomever.' The referent of this pronoun is indefinite and generic, and therefore does not take the referential object marker.

(3.439)	gəțal-ø	hemman	qə=rahem-ø=l=i
	kill.PFV-3SG.M	whomever	IND=love.IPFV-3SG.M=OBJ=3SG.M
	He killed whomeve	er he loves.	

3. The Noun

In (3.440) the direct object is the indefinite pronoun *yeki* 'someone.' In this example, the referent is indefinite and non-specific, and therefore does not take the referential object marker.

(3.440)	gəțal-ø	yeki	qə=rahem-ø=l=i
	kill.PFV-3SG.M	someone	IND=love.IPFV-3SG.M=OBJ=3SG.M
	He killed somebody	y he loves.	

As noted above in §3.6, once an entity has been introduced into the discourse, all subsequent referents to the same entity must be considered referential, regardless of whether the entity was marked as objectively referential at the moment it was introduced. Note that a specific individual who has already been identified to the listener may be marked both with the enclitic preposition ∂I - (or its independent form $\partial I a \bar{w}$) and the indefinite morpheme -i, as in (3.441).

(3.441)	merchant	<i>əmar-ø</i> say.PFV-3SG.M nt said, "I saw a l		<i>həz-it</i> see.PFV-1SG	<i>ya</i> a	<i>bāz-i</i> hawk-INDF
		•	on-INDF	<i>pəher=ye</i> fly.PASS=COP	.3SG.	М
	<i>amānatdār</i> trustee The trustee s	<i>sāyaḥ</i> scream screamed.	<i>məhā-ø</i> hit.PFV	-3SG.M		
		<i>āmr-et</i> D=say.IPFV-2SC ou saying, 'a hav		INDF		
	<i>əq=dar-ø=ø</i> IND=take.II took a boy ¹⁵	PFV-3SG.M=3S0		<i>w <u>ș</u>țān-i</i> F boy-INDI	7	
	and INI	<i>=pāher-ø</i> D=fly.IPFV-3SG ?"" (V.21-25)	.M			

In this example from Text V, the trustee is searching for his son, who was kidnapped by the merchant. The merchant lies to the trustee,

156 Namely, the aforementioned one, explicitly referential. Cf. Persian *pesar-i-rā* 'a (particular) boy (obj.)'

telling him that he saw a hawk fly off with a boy. At first, the object of the verb is not marked with the referential object marker, because the merchant is offering it as an non-referential and presumably unidentifiable entity. Having thus been introduced into the discourse, the boy is subsequently marked as referential by the object marker, despite the fact that the two parties in the conversation continue to refer to him as unidentifable.

(3.442) *Gāw* welāt-i k^{\prime} eh=l=i obr-ā in country-RES REL COP=to=3SG.M mouse-AUG In the country that has mice Ø sad hadid qə-ākel-ø mand REL. 100 maund iron IND=eat.IPFV-3SG.M who can eat 100 maunds of iron. muiur bāz-i lá=a=dar-ø=ø hawk-INDF NEG=IND=take.IPFV-3SG.M=3SG.M what.wav "How could a hawk not take a boy əl=ebr-i ke dah eh=l=i mand **REF=son-RES** REL 10 maund COP=to=3SG.M who weighs only 10 maunds?" (V.26)

In this example, taken from the same conversation, the boy is marked with both the referential object marker ∂I - and the morpheme -i, which performs a demonstrative function analogous to the Persian $y\bar{a}$ -ye *esārat* when it is followed by the restrictive relative pronoun ke (for which, see §4.3.2.2).¹⁵⁷

Even generic nouns can be marked with the referential object marker once they have been introduced into the discourse, as in example (3.443). In addition to nouns that have already been introduced into the discourse, certain types of noun phrases that inherently refer to specific entities will therefore always take the referential object marker ∂I . These include proper names (3.444), nouns marked as plural (3.445), nouns modified by a demonstrative pronoun (3.446) or a possessive pronoun (3.447), and nouns in construct with another noun (3.448).

¹⁵⁷ See Gernot L. Windfuhr, *Persian Grammar: History and State of its Study* (The Hague: de Gruyter, 1979), 37.

3. The Noun

- (3.443) *obr-ā* genz qə=rahem-ø=l=ø əl=ḥadid mouse-AUG very IND=love.IPFV-3SG.M=OBJ=3SG.M REF=iron The mouse really loves iron. (V.12)
- (3.444) *əmal-@=l=@ əl=Joḥa mu=yye āhā* say.PFV-3SG.M=OBJ=3SG.M REF=PN what-3SG.M this He said to Joha, "What is this?" (X.4)
- (3.445) $d \partial r \cdot \phi = \phi$ $\partial l = p \partial r \bar{a} h \cdot \bar{a} n$ $\partial m = b \bar{a} \bar{b}$ $hamm \bar{a} m$ take.PFV-3SG.M=3SG.M REF=money-PL from=father hammam He took the money from the owner of the hammam. (VIII.8)
- (3.446) $abd-onn=\emptyset$ $\partial l=\bar{a}$ konforms do.PFV-3PL=3SG.M REF=this conference They held this conference. (I.5)
- (3.447) $b \partial dq \phi = \phi$ $\partial l = k \partial r \bar{a} y = i$ $tu \underline{t} = i$ put.PFV-3SG.M=3SG.F REF=leg=3SG.M under-3SG.M He stuck his leg beneath him. (VII.2)

3.6.3 Marking of Identifiability

As noted above in §3.6, indefiniteness is a pragmatic status, indicating that the referent is offered to the listener as unidentifiable. It can be marked with ya 'a / one' as in (3.450), the indefinite morpheme -i, or both ya and -i. Both indicate that the referent of the noun is unidentifiable, but are ambiguous as to whether it is referential or non-referential (that is, whether it indicates a specific or non-specific entity).

(3.449)	<i>ya</i> a	<i>kədā<u>b</u></i> book	<i>mandāyí</i> Mandaic	a Mandaic book (III.16)
(3.450)	<i>tarmid-i</i> tarmida			a tarmida (II.8)
(3.451)	<i>ya</i> a	<i>tājér-i</i> merchant-	INDF	a merchant (V.1)

When added to the plural endings, the indefinite morpheme -i indicates a restrictive (but still unidentifiable) selection out of a

generic unit (3.452) or a plurality (3.453); in the latter case, the noun it modifies is occasionally preceded by the preposition *min* 'from.'

(3.452)	<i>mien-i</i> water-II	NDF	(some) water
(3.453)	<i>min</i> from	<i>barnaš-ān-i</i> person-PL-INDF	some people (I.13)

In these examples, the noun modified by the morpheme -i contrasts with the unmodified noun stem (with or without the augment), which often has a generic sense, e.g. *mienā* 'water,' *barnāšā* 'people;' see example (3.476) below under §3.6.4. In Persian grammar, this morpheme is known as the *yā-ye vahdat* or '*-i* of unity.'¹⁵⁸ This function of the morpheme is clearly evident in the phrasal verbs, as demonstrated by the following examples:

(3.474)	<i>əbād</i> deed	<i>q=ābed-nā</i> IND=do.PTC-1SG	I am working (lit. deed-doing).
(3.475)	<i>əbād-i</i> deed-IN	<i>q=ābed-nā</i> DF do.PTC-1SG	I am doing something.

3.6.4 Genericness

Generic reference denotes an entire category of referents rather than a single entity. Unless marked as referential or identifiable by any of the morphemes introduced in §3.6.2 and §3.6.3 above, such as the plural morpheme, the indefinite morpheme, pronominal clitics, or the object markers, any common noun can be construed as having a generic referent.

(3.456)	<i>atti-ø</i> come.D.PFV-3SG.M	2	<i>barnāš-ā</i> person-AUG
	<i>qə=bān-en</i> IND=build.IPFV-3PL	<i>ha<u>tt</u>ā</i> new	
	The ruler brought peopl	le who built [i	t] anew. (IV.17)

3.7 Modifiers

Within the noun phrase, the noun may be modified by one or more additional elements, including a second noun in an appositive relationship (§3.7.1), an attributive, predicate, or comparative adjective (§3.7.2), a numeral or non-numeral quantifier (§3.7.3), and a possessor (§3.7.4).

3.7.1 Appositive Nouns

A noun may also be modified by other nouns in an appositive relationship. This applies particularly to geographic expressions.

(3.457)	Šuštar	welāt	
	PN	city	the city Shushtar

3.7.2 Adjectives

Adjectives fall into three categories in Neo-Mandaic: attributive adjectives (§3.7.2.1), predicate adjectives (§3.7.2.2), and comparative adjectives (§3.7.2.3). All three categories express some property of the head noun that they modify, such as age, dimensions, value, or color. The first two categories of adjective may be employed to introduce the properties of indefinite and definite referents (i.e. those that have not yet been introduced onto the discourse stage as well as those that have already been introduced), whereas the third category can only apply to definite referents (i.e. those that have already been introduced onto the discourse stage and are identifiable to the speaker and his audience).

3.7.2.1 Attributive Adjectives

Attributive adjectives directly follow the noun which they modify, e.g. (3.458) and (3.459). As the examples below demonstrate, the head noun may be inflected to reflect gender and number, e.g. (3.460); most adjectives do not agree with the head noun in gender and number, although a few do, cf. (3.461) with (3.462). The nominal augment on the head noun is deleted whenever it is modified by an attributive adjective, as when the head noun is followed by a second noun in construct. Unlike the construct chain (see §3.7.4.2 below), a noun

3.7 Modifiers

followed by an attributive adjective may be construed as either definite or indefinite; in the latter case, the indefinite morpheme is appended at the end of the noun phrase rather than on the constituent elements (3.458).

(3.458)	<i>qazġān</i> cauldron	<i>honin-i</i> small-INDF	a small cooking pot (X.2)
(3.459)	<i>kədā<u>b</u></i> book	<i>Mandāyí</i> Mandaic	a Mandaic book (III.16)
(3.460)	<i>bie<u>t</u>-wā<u>t</u> house-PL.F</i>	<i>ba'id</i> distant	far-off houses (III.10)
(3.461)	<i>barnāš</i> person	<i>horin-ā</i> other-AUG	another person
(3.462)	<i>šer</i> war	<i>horet-t-ā</i> other-F-AUG	Second World War (III.2)

3.7.2.2 Predicative Adjectives

Predicative adjectives are always accompanied by the copular verb (3.463) or the enclitic copula (3.464); for further details on the copula, see (§5.1.1.4).

(3.463)	ya yeki	həwā-ø	mariz-i			
	a someone	be.PFV-3	3SG.M ill-INDF			
	There once was a	s a man who was ill. (VI.1)				
(3.464)	həwā-ø	ya yeki	i pahli=d=i			
	be.PFV-3SG.M	a some	beside=of=3SG.M			
	There was someone who was sitting					
	yəhem=ye	sā	āt-ø=i			
	sit.PASS=COP.3	SG.M lis	listen.PFV-3SG.M=3SG.M			
	beside him, who heard him. (VI.3)					

3.7.2.3 Absolute and Relative Comparison

A noun may be modified by an adjective to express two forms of comparison: relative comparison, in which the noun is explicitly compared with a second noun, or absolute comparison, in which it is compared against all other members of its category. In Neo-Mandaic, as in most other Semitic languages, relative comparison is expressed 3. The Noun

by means of a predicative adjective (describing the property being compared) modified by a complement with the preposition *min* or its enclitic form ∂m -, see (3.465).

(3.465) *mienā* beh əm=bira water-AUG good from=beer Water is better than beer.

Absolute comparison, by contrast, is always expressed with the Persian suffix *-tar*. Adjectives modified by this suffix can be used predicatively (3.466) or attributively (3.467). They may also be used adverbially (3.468).

(3.466) awwáli rap-tar u=xorda=xorda honin-tar $q = h\bar{a}w$ -en first big-COMP and=little=little small-COMP IND=be.IPFV-3PL The first was bigger, but little by little they became smaller.

(3.467)	<i>q=āṯ-en</i> IND=come.IPI They came in a	FV-3PL i	n :	<i>mentekār</i> motor.car	<i>rap̄-tar-i</i> big-COMP-INDF
(3.468)	<i>šo<u>b</u>b-ā</i> seven-AUG Seven years ag	<i>šett-ā</i> year-AUG o. (III.18)	<i>eštā</i> now	<i>qamā-tar</i> before-CO	MP

The superlative is expressed with the absolute comparative and the word *geš* 'all.'

(3.469)	beh-tar	əm=geš	
	good-COMP	from=all	the best

3.7.3 Quantifiers

Quantifiers are a subset of noun modifiers that indicate how many or how much of the referent is intended. There are two basic types of quantifiers in Neo-Mandaic: numerals such as cardinal numbers (§3.7.3.1) which indicate a definite quantity of the referent, and nonnumeral quantifiers (§3.7.3.2), which indicate an indefinite quantity. Ordinal numbers (§3.7.3.1), by contrast, indicate the order of a referent with respect to other members of its class, rather than the quantity of the referent. These are treated exactly like attributive

¹⁵⁹ Examples (3.466) and (3.467) are adapted from Macuch, *Neumandäische Texte*, 118, ln. 196 and 198 ln. 1062, respectively.

3.7 Modifiers

adjectives in Neo-Mandaic (§3.7.2.1) but have been included here with the other number words.

3.7.3.1 Numerals

Numerals fall into two broad categories in Neo-Mandaic: cardinal numbers, which indicate the quantity of the referent, and ordinal numbers, which indicate the order of the referent with respect to other members of its class. The former precede the noun that they modify, and the latter follow it.

Cardinal Numbers

The cardinal numbers most commonly used in Neo-Mandaic are borrowed from colloquial Persian, although the original Neo-Mandaic numbers survive alongside them. Regardless of their origin, cardinal numbers appear before the noun, and are invariable; they do not agree in gender with the noun. Likewise, the noun modified by the number always appears in the singular.

Traditional	Modern	Gloss	Traditional	Modern	Gloss
ehdā	yek	one	asrā	dah	ten
tren	do	two	esrin	bis	twenty
klātā	SO	three	klātin	si	thirty
arbā	čahār	four	ārbin	čehél	forty
hamšā	panj	five	hamšin	panjāh	fifty
šettā	šeš	six	šettin	šast	sixty
šubbā	haft	seven	šubbin	haftād	seventy
təmānā	hašt	eight	təmānin	haštād	eighty
eččā	noh	nine	eččin	navád	ninety

Table 13: Cardinal Numbers

The number *esrin* 'twenty' is a reflex of the dual of the number *asrā* 'ten;' the other decades are formed on the basis of the numbers 3-9 with the old absolute plural morpheme appended (see §3.4.3). Despite the fact that the addition of this morpheme causes the accent to shift, allomorphic variation (such as the pretonic reduction of the preceding syllable or the simplification of the geminate consonant) does not

result. The fact that they do not behave in same manner as the rest of the Neo-Mandaic lexicon underscores their learned status.

All numbers apart from the units and the decades are formed from compounds. Numbers 11-19 are formed from the unit in juncture with the contextual form of *asrā* 'ten;' my informant had difficulty producing these, as in (3.470). In other compound numbers, the unit and the decade are combined by the conjunction u 'and,' e.g. (3.471). The centuries are formed from the units in juncture with the word for one hundred, *emmā*, e.g. *čemmā* 'nine-hundred.' The word for one thousand is *alpā*. In counting, centuries always precede decades, and millennia always precede centuries, e.g. (3.472).

(3.470)	<i>čin=assər</i> 90=10	<i>šett-ā</i> year-AUG	ninete	en years (III.14) ¹⁶⁰	
(3.471)	<i>čin u=hamš-ā</i> 90 and=five-AUG		ninety-five (III.18) ¹⁶¹		
(3.472)	<i>alp̄-ā</i> 1000-AUG nineteen ninety-f	<i>u=čemmā</i> and=900 five (III.18)	<i>u=čin</i> and=90	<i>u=hamš-ā</i> and=five-AUG	

Ordinal Numbers

Ordinal numbers generally behave more like attributive adjectives than the other numerals and non-numeral quantifiers. They appear infrequently in the text collection. The Classical Mandaic ordinal *qadmāyā* 'first' appears in the first text, (3.473), in a frozen expression, alongside the more common *awwál*, (3.474), which is a loan word; the adjectives *horinā* and *horettā*, introduced above (see \$3.7.2.1), are used in the place of the ordinal 'second;' *klatāyā* 'third' appears once, (3.474), and the loan word *āxír* 'last' appears frequently, e.g. (3.475), instead of the classical **batraia**. Note that, with the exception of the loan words *awwál* and *āxír*, ordinals follow the noun that they modify.

¹⁶⁰ Macuch provides the form *eččassar* for this number in the dialect of Ahvāz; cf. CM '**tšasar** 'nineteen.' Macuch, *Handbook*, 231.

¹⁶¹ Macuch indicates that the unit precedes the decade in the dialect of Ahvāz, in the Persian manner; e.g. *ārba u srin*, 'twenty-four.' Ibid., 231.
3.7 Modifiers

(3.473)	<i>Ādam</i> PN	<i>gobr-ā</i> men-AUG	<i>qadmāy-i</i> first-ES.P		m, the First Man (I.8)
(3.474)	<i>ani</i> we We Ma	<i>Manday-ān-ā</i> mandaean-PL-A ndaeans were the f	UG firs	. I	be.PFV-1PL
(3.475)	<i>səbuʿa</i> week	<i>kla<u>t</u>āy-ä</i> third-Al		the	hird week (VIII.10)
(3.476)	<i>āxír</i> last	<i>ālm-ā</i> world-A	UG	the	ast world (I.14)

3.7.3.2 Non-Numeral Quantifiers

Non-numeral quantifiers are terms which express an indefinite quantity. Like cardinal numbers, they precede the noun that they modify.

(3.477) genzā	much, many	genz ənāš	many people
(3.478) <i>perzā</i>	few	perz pərāhi	a little money
(3.479) <i>baʿaẓ</i>	some	baʿaẓ awqāṯ	sometimes
(3.480) <i>kol</i>	all	kol barnašānā	everybody

3.7.4 Nouns and Possession

Possession may be expressed in five ways: the use of objective / possessive pronominal clitics ($\S3.7.4.1$), the construct chain (see $\S3.4.2$ above and $\S3.7.4.2$), the use of the prepositions $q\bar{a}r$ 'at' and al-'to' ($\S3.7.4.3$), the use of the relative pronoun *d*- ($\S3.7.4.4$), and the use of the noun *māl* 'property' ($\S3.7.4.5$). The first four constructions may express either alienable or inalienable possession; *māl* is used to indicate alienable possession only.

3.7.4.1 Pronominal Clitics

The pronominal clitics which indicate the object, complement, or adjunct of a verb may also serve to indicate possession of a noun; see §4.1.2 for further details.

3.7.4.2 The Construct Chain

The Construct Chain, introduced in §3.4.2 above, also serves to indicate possession:

POSSESSED + POSSESSOR

The possessor can be any noun or noun phrase. The possessed must always appear in its contextual form, but otherwise neither the possessor nor the possessed is morphologically marked to indicate possession.

(3.481)	<i>ebbər</i> son	<i>ebr=i</i> son=3SG.M	his grandson, lit. his son's son (I.10)
(3.482)	<i>qazġān</i> cauldron	<i>šəbāb-ā</i> neighbor-AUG	the neighbor's cooking pot
(3.483)	<i>bā</i> b father	<i>ḥammām</i> bathhouse	the owner of the bathhouse (VIII.4)

3.7.4.3 Possession Indicated By Preposition

The preposition ∂I - is used for general possession, most often with people and other animate beings; see (3.504). It is also employed in situations where the genitive construct chain would be inappropriate (for example, when the head noun is indefinite), as in (3.505). This same preposition can be applied to the interrogative pronoun *man* 'who,' as in (3.506).

(3.484)	rišāmm-ā	əl=Manday-ān-ā	
	rishama-AUG	to=Mandaean-PL-AUG	
	The leader of the Mandaeans (IV.35)		

(3.485)	ebr-i	əl=gabr-ā	qə=lakt-ø-i
	son-INDF.SG	to=man-ES.SG	IND=seize.PTC-3SG.M-3SG.M
	He seized one of	the man's sons. (V.)	18)

(3.486)	ahni	əl=man=non	
	those	to=who=3PL	Whose are they? (III.11)

The preposition $q\bar{a}r$, by contrast, is employed to specifically indicate that the object possessed is on the possessor's person or that the possessor otherwise has immediate access to it; see (3.507).

3.7 Modifiers

(3.487)	rāz-ā	qār=e	
	secret-AUG	at=1SG	my secret (IV.92)

3.7.4.4 The Relative Pronoun d-

In Classical Mandaic, possession is generally expressed by the relative pronoun d- (which is usually written with a separate letter of the alphabet, transcribed <u>d</u>-).¹⁶² The relative pronoun d- can either follow the noun in the emphatic state, as in (3.488) or the noun modified by an anticipatory pronominal suffix, as in (3.489). In addition to these frozen expressions, this construction continues to be used in some epithets and in archaizing expressions, e.g. (3.490) and (3.491); note the obligatory appearance of the augment on the Arabic word <u>hākem</u> 'governor' in (3.491), despite the fact that the augment would not otherwise appear on a non-nativized loan word.

(3.488)	•	<i>=Hi<u>b</u>el Zi</i> f=PN		om of Hibel Ziwā (IV.1)
(3.489)	<i>b=ešm-e=hon</i> in=name-PL.CS= In the name of the	3PL	<i>əd=Heyy-i</i> of=life-PL.ES ife (I.1)	<i>Rabb-i</i> great-PL.ES
(3.491)	<i>Mand-ā</i> knowledge-AUG	<i>d=Hey</i> of=life	-	owledge of Life (I.9)
(3.492)	<i>ḥākim-ā</i> governor-AUG	<i>d=Šuši</i> of=PN		governor of Shushtar (IV.65)

3.7.4.5 Māl Construction

The noun $m\bar{a}l$ 'property' is used, albeit rarely, to express alienable possession. This construction is likely derived from one of the colloquial Iraqi dialects of Arabic in which $m\bar{a}l$ is the usual "genitive" particle. (3.493) is the single example of the $m\bar{a}l$ construction elicited from the texts from Khorramshahr, although it is also attested in the Iraqi dialects of Neo-Mandaic, as example (3.494), collected by Drower, demonstrates.

162 See Häberl, "The Relative Pronoun $\underline{\mathbf{d}}$ -," for further details on the classical relative pronoun and its reflexes in Neo-Mandaic.

3. The Noun

(3.493)		<i>māl</i> property	<i>șobbi-hā́</i> Sabian-PL	the S	abians' houses (III.12)
(3.494)	zidq-a	bri	i <u>kh</u> -a	māl	gmā <u>sh</u> -i

oblation-AUG blessed-AUG property clothes-PL The blessed oblation of the clothes.¹⁶³

163 Drower, Mandaeans, 215.

4. Pronouns

Pronouns may be substituted for any noun phrase. There are five types of pronouns in Neo-Mandaic: personal pronouns (§4.1), demonstrative pronouns (§4.2), relativizers (§4.3), indefinite pronouns (§4.4), and interrogative pronouns (§4.5). The first three refer endophorically to an antecedent, such as a noun phrase already introduced within the context of the discourse; the fourth category indicates non-specific referents, and the final category is used in interrogative clauses to indicate the type of information which is being requested.

4.1 Personal Pronouns

Personal pronouns may represent the subject of a transitive or intransitive verb, the object of a transitive verb, and any nominal or verbal complement or adjunct that has already appeared in the context of the discourse. There are four types of personal pronouns in complementary distribution: independent personal pronouns (§4.1.1), enclitic personal pronouns (§4.1.2), reflexive personal pronouns (§4.1.3), and reciprocal personal pronouns (§4.1.4).

4.1.1 Independent Personal Pronouns

The independent personal pronouns in Neo-Mandaic are composed of one or two syllables; they distinguish between genders only in the third person singular. All pronouns are lightly accented on the first syllable. The first person pronouns, singular and plural, fall together in their contextual forms:

	Singular			Plural	
Lexical	Contextual	Gloss	Lexical	Contextual	Gloss
anā	an	Ι	ani	an	we
āt	-	you (sg.)	atton	-	you (pl.)
huwi	huw	he	honni	-	they
hidā	hid	she			

Table 14: Independent Personal Pronouns

Although the second person is generally expressed without reference to gender, gender can be indicated by the word "man" or "woman," which follows the independent pronoun.

(4.1)	āt e <u>tt</u> ā	you (f. sg.)
(4.2)	āt gabrā	you (m. sg.)

The Neo-Mandaic independent personal pronouns are not often used. They are employed exclusively to represent the subject of a transitive or intransitive verb. Their presence in this role is purely optional, and they effect no change in the meaning of the utterance when omitted. Whenever they are employed, they appear before the verb in their contextual forms; see (§4.3). The lexical form of the independent personal pronouns may be employed in a contrastive or emphatic function, as in (§4.4).

(4.3)	<i>(an)</i> (1SG) I sat bes	<i>yəhm-it</i> sit.PFV-1SG ide you.	<i>pahli=d=a<u>k</u></i> beside=of=2SG.M
(4.4)	<i>anā</i>	<i>yəhm-it</i>	<i>pahli=d=ak</i>
	1SG	sit.PFV-1SG	beside=of=2SG.M
	<i>I</i> (was t	he one who) sat be	eside you.

4.1.2 Enclitic Personal Pronouns

The enclitic personal pronouns are in complementary distribution with the independent personal pronouns; they may represent the object of a transitive verb (§4.1.2.1), a nominal or verbal complement or adjunct (§4.1.2.2), or indicate possession on the noun (§4.1.2.3). They cannot, however, ever be substituted for the enclitic copula or the independent personal pronouns.

Person	Masculine Singular		Feminine Singular	Plural
1 st	-	-е	-	-an
2^{nd}	-a <u>k</u>		-e <u>k</u>	- <u>k</u> on
3 rd	- <i>i</i>		-a	- <i>u</i>

4.1.2.1 Pronominal Object Suffixes

The enclitic pronominal suffix may serve as the object of a transitive verb; in this function, they are attached to the inflected form of the verb either directly, as in (4.5) or via an enclitic preposition, *-l*-, as in (4.6). The third masculine singular suffix can be dropped whenever it is pragmatically inferable; see, for example, (4.28) below.

(4.5)	<i>tum</i> then	<i>sat-onn=i</i> hear.PFV-3PL=3SG.M		Then they heard h	im. (VII.7)
(4.6)		<i>= =j</i> /-3SG.M=OBJ=3SG.M to him, "Yikes! Let's ma	yikes		<i>pard-én</i> flee.SBJV-1PL

4.1.2.2 Pronominal Adjuncts and Complements

These enclitic pronominal suffixes may also appear on prepositions in the place of nominal adjuncts or complements (see §3.5.4 above).

(4.7)	<i>qam-yon</i> stand.PFV-3PL	<i>barābar</i> opposite		They stood before them. (IV.76)
(4.8)	<i>a<u>t</u>-it=i</i> go.D.PFV-1SG=3	3SG.M	<i>qam=də=kon</i> for=of=2PL	I brought it for you. (X.7)

4.1.2.3 Pronouns and Possession

One of the most common roles of the enclitic pronominal suffixes is to indicate possession. On originally Mandaic vocabulary and nativized loan words, the suffixes attach directly to the noun or the noun phrase.

(4.9)	babb=e	
	grandfather=1SG	my grandfather (II.2)

4. Pronouns

(4.10)	<i>bā<u>b</u></i> father	emm=e mother-1SG	my mother's father (II.2)
(4.11)	<i>nap̃š=e</i> self-1SC	3	myself

Nouns borrowed from other languages take pronominal clitics after an enclitic morpheme *-d-*, which appears after any other inflectional morphemes on the noun:

(4.12)	<i>'umər=d=e</i> age=of=1SG (Arabic)	my age (III.14)
(4.13)	<i>jamiʿa=də=kon</i> entirety=of=you (Arabic)	all of you (IV.124)
(4.14)	<i>rezq-ấ=d=an</i> livelihood-PL=of=1PL	our jobs (IV.132) (Arabic)

Whenever the enclitic copula (see \$5.1.1.4) is attached to a noun modified by the first person singular pronominal suffix and the third person pronominal suffixes, an epenthetic *-h*- (see \$2.5.3.4) intervenes between the pronominal suffix and the copula.

(4.15)	mujur	bāb=ah=ye	
	what.way	father=3SG.F=COP.3SG.M	What's her father like? (IX.14)

4.1.2.4 Personal Pronouns on Quantifiers

The plural enclitic pronominal suffixes also modify quantifiers such as dota 'both' and ges 'all.' The indefinite pronoun *ehdi* 'one' is also included to complete the paradigm, even though the enclitic pronouns are attached to the independent preposition *min* 'from' rather than the enclitic morpheme *-d-*.

4.1 Personal Pronouns

Table 16: Quantifiers with Enclitic Pronominal Suffixes

One		Two^{164}		Three or More	
ehdi minn-an	one of us	do <u>t</u> ā-d-an	both of us	geš-d-an	all of us
ehdi min- <u>k</u> on	one of you	do <u>tā-t-k</u> on	both of you	geš-də- <u>k</u> on	all of you
ehdi minn-u	one of them	do <u>t</u> ā-d-u	both of them	geš-d-u	all of them

Unlike other the quantifiers introduced in (§3.7.3), the quantifiers dota'both' and *geš* 'all' cannot govern a noun in a construct chain, e.g. (4.16), and are obliged to take the pronominal clitic. They can, however, be governed by a noun, much like an adjective, in which case they follow it and take a resumptive clitic, e.g. (4.17).

(4.16)	kol lilyā	every night
(4.17)	lili geš-di	the entire night

These quantifiers with the enclitic pronominal suffixes are frequently used to modify the plural independent personal pronouns, intervening between the independent pronoun and the verb.

(4.18)	atton	dotā=t=kon	hab-tonn=i	qamdi=d=i
	2PL	both=of=2PL	give.PFV-2PL=3SG.M	to=of=3SG.M
	Did bo	th of you give it to	o him?	

4.1.3 Reflexive Pronouns

The reflexive personal pronoun is employed within the verb clause to refer to an antecedent within the same clause. It is formed from the noun $na\bar{p}s\bar{a}$ 'soul, personality, self' with the enclitic pronominal suffixes.

Table 17: Inflected Forms of the Reflexive Pronoun

Person	Masculine	Feminine	Plural
	Singular	Singular	
1^{st}	nap	napš-an	
2^{nd}	napš-a <u>k</u>	napš-ek	napš-kon
3 rd	napš-i	napīš-a	napīš-u

164 This quantifier is composed of the Persian *do* 'two' and the Persian classifier $t\bar{a}$. Note that the voiced dental stop /d/ becomes fricativized before the voiceless velar stop / χ /; see §2.5.4.1.

4.1.3.1 Reflexive Pronouns as Anaphoric Objects

The reflexive pronoun most commonly appears in object position, functioning as an object, and referring back to the subject of the sentence, though the subject may be dropped as in other sentence constructions.

In the classical language, the reflexive pronoun supplemented the so-called reflexive stems of the verb (for which, see §5.1.1.3), although they have since become the sole means of forming the reflexive in Neo-Mandaic, in conjunction with the referential object marker *əl*-, as in (4.58).

(4.19) $bašqir-\emptyset$ $\partial l=naps=ak!$ know.IMP-2SG REF=self=2SG.M Know yourself!

4.1.3.2 Emphatic Uses of the Reflexive Pronoun

As in the classical language, the reflexive pronoun is often employed to emphasize one of the constituents of the verb phrase rather than serve as an anaphoric object; compare (4.20), in which the reflexive pronoun is not expressed, with (4.21).

(4.20)	<i>yəhom-ø</i> sit.IMP.M-SG	<i>demme</i> down		Sit down!
(4.21)	<i>ayhim-ø-a<u>k</u></i> sit.C.IMP-SG-2S Sit yourself down		<i>napš-ak</i> self-2SG.M ke yourself sit you	<i>demme</i> down urself down!).

4.1.3.3 Reflexive Pronouns and Possession

Governed by another noun, the word $na\bar{p}s\bar{a}$ 'self' suffixed with the enclitic pronominal suffixes can also express a reflexive possessive.

(4.22)	Nāșer	bənā-ø	biet	napš=i
	PN	build.PFV-3SG.M	house	self=3SG.M
	Nasser	built his own house.		

4.1.4 Reciprocal Pronouns

Neo-Mandaic has two reciprocal pronouns, *ham* and *hədādā*. The first, *ham* 'each other,' is a Persian loan word. In Neo-Mandaic, it is used to

modify verbs directly as an oblique complement. In Persian, however, it usually appears as part of constructions such as *be-ham-digar* 'to one another.' The second pronoun, $h \partial d\bar{a} d\bar{a}$, can either be employed as an object or governed by a preposition as an adjunct or complement, e.g. *mork hodādā* 'with one another,' *be-hdādā* 'to one another.' These examples show reciprocals functioning as the object of a verb in (4.23), an object complement in (4.24) and (4.25), as an oblique complement in (4.26), and as an adjunct to a verbal phrase in (4.27).

hədādā (4.23)kol=waxt *qə=haz-én* ani IND=see.PTC-1PL we always one another We always see one another. (4.24)genz həter=nān qə=hazi-n=kon anā rejoice.PTC.PASS=1SG IND=see.IPFV-1SG=2PL Ι verv I'm very happy to see you all geš mork hədādā gāw Harvard University all with one.another in PN together at Harvard University. (I.4) (4.25)atton kol=waxt q=haze-l=kon ork hədādā 2PL IND=see.PTC-3PL=2PL with always one another You were all seen together. (4.26)qənav-ān-ā inš-ān-ā и tamm-on əras-von womin-PL-AUG become.PFV-3PL marry.PFV-3PL smith-PL-AUG and The jewelers and women became married be=hdādā hattā to=one.another anew again to one another. (IV.148) (4.27)geš=d=an əhab-ni əhāb-ā ham give.PFV-1PL all=of=1PL each other gift-AUG

4.2 Demonstrative Pronouns

All of us gave each other gifts.

Neo-Mandaic distinguishes between near-deixis and far-deixis in the singular demonstrative pronouns, but not in the plural. It also makes no distinction in gender. The original far-deictic plural demonstrative pronoun *ahni* 'those' has assumed the function of a general plural

4. Pronouns

demonstrative pronoun.¹⁶⁵ This same demonstrative pronoun is often employed in the place of the third plural personal pronoun, as in (4.28).

(4.28) *ahni saṭ-yon əm=lu=Manday-ān-ā* those hear.PFV-3PL from=non=Mandaean-PL-AUG They heard [it] from non-Mandaeans. (I.13)

The singular forms of the demonstrative pronouns possess lexical and contextual allomorphs.

Table 18: Forms of the Demonstrative Pronouns

	Near-Deixis			Far-Deixis	
Lexical	Contextual	Gloss	Lexical	Contextual	Gloss
āhā	ā	this	a <u>k</u> u	a <u>k</u>	that
ahni	-	these	ahni	-	those

The demonstrative pronouns usually precede the noun they modify.¹⁶⁶ In this position, they assume their contextual forms.

(4.29)	ā bie <u>t</u> ā	this house
(4.30)	a <u>k</u> enšā	that man

Note that there is no contextual form for the plural demonstrative pronoun; because contextual forms lack any overt expression of number, the contextual forms of the singular pronouns are also used for plural nouns, e.g. (4.31). In some cases, a demonstrative pronoun and a noun function together with an adverbial significance, e.g. (4.32) and (4.33).

(4.31)	ā šeršānā	these religions (never **ahni šeršānā)
(1.00)		

- $(4.32) \quad \bar{a} y u m \bar{a} \qquad \text{today} (\text{lit. "this day"})$
- (4.33) *ak waxt* then (lit. "that time")
- 165 NM *ahni* 'these; those' is derived from CM **hania** 'those.' Macuch (*Neumandäische Texte*, 56) cites a far-deictic plural demonstrative pronoun *ānnex* or *hānnex* 'dasselbe,' which he describes as having been supplanted by Persian *hamu* in the dialect of Ahvāz; neither form is employed by my informant.
- 166 In the text collection there is a single exception: *deštā āhā* 'this land' (IV.105).

When the demonstrative pronouns are not employed to modify a noun, they appear in their lexical forms.

(4.34)	<i>mu=yye</i> what=3SG.M	<i>āhā</i> this	what's this?
(4.35)	<i>mu=yye</i> what=3SG.M	<i>a<u>k</u>u</i> that	what's that?

4.2.1 Locative Demonstrative Pronouns

Neo-Mandaic has two locative demonstrative pronouns, $h = n\bar{a} / ehn\bar{a}$ 'here' and $ekk\bar{a}k$ 'there.' These typically follow the verb, but may also appear at the beginning of the phrase, as in (4.38) and (4.39).

(4.36)	<i>man</i> who Who wa	<i>huw</i> 3SG.M as sitting	<i>yəhem-ø</i> sit.PFV-3SC here earlier?	<i>hənā</i> 6.M here	<i>qamā-tar</i> before-COMP
(4.37)	<i>7-11</i> PN They ha	ıd a 7-11	<i>həwā-ø=l=u</i> be.PFV-3SG.M there. (III.6)	A=to=3PL	<i>ekka<u>k</u></i> there
(4.38)	<i>hənā</i> here Here, ev	<i>ḥatta</i> even ven the ch	<i>jihel-á</i> child-PL hildren speak it!	1	
(4.39)	<i>ekka<u>k</u> there</i> There w	<i>ya</i> a vas a hole	<i>bəzuy-i</i> hole-INDF there. (V.8)	<i>həwā-ø</i> be.PFV-3SG	ð.M

4.2.2 Anomalous Forms of the Demonstrative Pronoun

In Text IV, Hirmiz bar Anhar's account of the legend of the Bridge of Shushtar, as retold by Nasser Sobbi, the far-deictic demonstrative pronoun $t\bar{a}$ appears in two places (IV.33, IV.39); while this pronoun is attested in other Neo-Mandaic dialects, it does not appear elsewhere in the texts collected, and for this reason I did not include it in Table 18. Likewise, in VIII.16, the form $h\bar{a}$ appears after the preposition $g\bar{a}w$; while one might be tempted to identify this form with Classical Mandaic **ha** 'this,' it is also possible that this form represents $\bar{a}h\bar{a}$ after the initial vowel has been elided.

4.3 Subordinating Conjunctions

Subordinating conjunctions introduce a relative clause. The type of conjunction employed depends upon the presence or absence of a head noun, and whether the relative clause is referential (§4.3.1) or non-referential (§4.3.2). An indefinite pronoun (§4.3.3) may be employed to introduce a headless relative clause, and a relativizer (§4.3.4) is used to introduce a relative clause that refers back to an entire clause or verb phrase rather than a nominal antecedent.

If the relative clause depends upon a head noun, the type of relative pronoun employed is further determined by whether the antecedent is definite or indefinite, and whether the clause is restrictive or non-restrictive. As mentioned above (§3.6.3), nouns may be morphologically marked as indefinite (in opposition to definite nouns, which are unmarked), and syntactically or morphologically marked as non-specific (as opposed to specific or generic, which are unmarked). If the antecedent is the object of the verb of the subordinate clause (direct, indirect, or oblique), it must be marked by a resumptive pronoun on the verb.

4.3.1 Indefinite Antecedents

Relative pronouns are completely absent from relative clauses headed by indefinite antecedents. In examples (4.40) and (4.41), the gap left by the absence of the relative pronoun is indicated by the null sign \emptyset . This indicates that the speaker assumes that his or her audience cannot identify the referent of the head noun; note also that none of the relative pronouns may introduce a headless relative clause, this function being relegated to the indefinite pronouns *kol mā* 'whatever' and *kol man* 'whoever' (see §4.3.3 below).

(4.40)	<i>gāw</i> in In Ahvi	<i>Aḥwāz</i> PN āz, there was a	<i>həwā-ø</i> be.PFV-3SG.M a tarmida,	<i>tarmid-i</i> tarmida-INDF	
	Ø REL whose	<i>ešm-i</i> name-3SG.M name was She	<i>Šiex</i> M sheikh ikh Abdallah. (II.8)	<i>'Abdalla</i> PN	

4.3 Subordinating Conjunctions

(4.41) $q=\bar{a}mr-en$ ya $t\bar{a}j\acute{e}r-i$ $h \ni w\bar{a}-\phi$ IND=say.IPFV-3PL a merchant-INDF be.PFV-3SG.M It is said (lit. 'they say') that there was once a merchant,

Ø	həwā-ø=l=i	perz	pərāh-i
REL	be.PFV-3SG.M=to=3SG.M	little	money-INDF
who ha	ad [only] a little cash. (V.1)		

4.3.2 Definite Antecedents

Whenever the head noun of the clause is presumed to be identifiable to the audience, Neo-Mandaic employs one of two relative pronouns, which have been borrowed from the languages with which it has come into contact, Arabic and Persian:

(4.42)	illi	which (Colloquial Arabic illi)
(4.43)	ke	that (Persian ke)

The primary distinction between the two relative pronouns lies in the type of relative clauses which they govern. The relative pronoun *illi* is employed to introduce non-restrictive relative clauses (\$4.3.2.1), namely those which contribute supplemental information to the main clause. The second relative pronoun, *ke*, introduces restrictive clauses (\$4.3.2.2), namely those which contribute information essential to the understanding of the main clause.

4.3.2.1 Non-Restrictive Relative Clauses

The relative clause introduced by the relative pronoun *illi* contributes supplemental information about the head noun to which it refers.

(4.44) *q=abi-n əmer-ø genz farwāh Profesər Buckley* IND=want.PTC-1SG say.SBJV-1SG many thanks PN I want to give many thanks to Professor Buckley,

illi genz əbād abd-at qam Manday-ān-ā REL much work do.PFV-3SG.F for mandaean-PL-AUG who has done so much work for the Mandaeans. (I.6)

4.3.2.2 Restrictive Relative Clauses

Relative clauses introduced by *ke* contribute information essential to the understanding of the main clause, as in (4.45). When the preceding

noun is marked with -i, it must always be interpreted as demonstrative in meaning, as in (4.46) and (4.47). Note that the antecedent is the object of the verb of the subordinate clause in these two examples, and that the verb or copula takes a resumptive pronoun.

(4.45)		.IMP-PL e people	p	<i>arnaš-ān-ā</i> erson-PL-A	AUG	ke REL		
			sı	<i>ābš-el=l=u</i> ıbdue.IPFV (IV.14)		BJ=3PL		
(4.46)	<i>gāw</i> in in a (spe	<i>welāt-i</i> city-RE ecific) cit		<i>k'</i> REL as mice	<i>eh=l=i</i> COP=to	=3SG.M		<i>obr-ā</i> mouse-AUG
	<i>sad</i> 100 that can	<i>mand</i> maund eat 100 r	naunds	<i>ḥadid</i> iron of iron (V	.26)	<i>q=ākel-e</i> IND=ea	⌀ t.IPFV-3	SG.M
(4.47)	<i>ebr-i</i> son-IND a boy wi	-	<i>ke</i> REL s only	<i>dah</i> 10 10 maunds	<i>mand</i> maund (V.26)		<i>eh=l=i</i> COP=to	=3SG.M

The morpheme -i, which generally indicates the noun as indefinite in other contexts, here imparts a demonstrative meaning upon the antecedent, in a manner analogous to the Persian $y\bar{a}$ -ye eš \bar{a} rat (see §3.6.2). Once again, the resumptive object pronoun on the verb in (4.48) refers back to the antecedent.

(4.48) *dukk-ān-i ke həz-it=u awwál* place-PL-RES REL see.PFV-1SG=3PL before The places which I saw before (III.19)

4.3.3 The Relativizer

Ke can also serve as a relativizer after verbs of perception and verbs that introduce direct speech.

(4.49)	<i>nedā</i> herald Throughout the c	<i>əbad-yon</i> do.PFV-3PL ity it was proclaimed,	<i>gāw</i> in	<i>welāt</i> city	ke REL
	<i>ebr-i</i> son-INDF "a boy has gone :	<i>əzgā-ø</i> go.PFV-3SG.M from us." (V.20)	<i>minn=a</i> from=11		

In this example, it is more accurate to identify ke as a relativizer rather than a relative pronoun, as it does not represent the head noun within the relative clause. The information introduced by these verbs always assumes the form of direct speech, whether it is introduced by ke or not:

(4.50) yəmi-ø qamd=in q=əmwadi-n=kon Əmšuni Kušţā swear.IMP.M-SG to=me IND=take.IPFV-1SG=2PL PN Swear to me, "I will take you to Əmšuni Kušţā." (IV.120)

4.3.4 Headless Relative Clauses

The indefinite pronouns *kol man* and *kol mā* (\$4.4.3) are used to introduce headless relative clauses, e.g. (4.51).

(4.51)	kol mā	q=āzi-ø	q=āzi-ø
	whatever	IND=go.PTC-3SG.M	IND=go.PTC-3SG.M
	Whatever happe	ens, happens (cf. Italian che	sarà, sarà). (VIII.12)

4.4 Indefinite Pronouns

The following nouns (\$4.4.1) and adjectives (\$4.4.2), modified by the indefinite morpheme *-i*, are employed as pronouns to indicate non-specific or indefinite referents. As these indefinite pronouns cannot be modified by an adjective or govern another word in a construct chain, they never occur in contextual forms. A third category of indefinite pronouns (\$4.4.3) consists of indefinite nouns or pronouns governed by a quantifier.

4.4.1 Indefinite Pronouns Derived from Nouns

The indefinite morpheme -i can be used to derive an indefinite pronoun from a noun. These indefinite pronouns may refer to an unspecified animate or inanimate referent.

(4.52)	enši	someone, anyone (lit. a person)
(4.53)	yeki	someone, anyone (Persian yaki id.)
(4.54)	mendi	something, anything (lit. a thing)
(4.55)	əbādi	something, anything (lit. a deed)

The indefinite pronouns *enši* and *yeki* are used for nonspecific human beings and other animate referents, such as demons. The indefinite pronoun *mendi* is employed to indicate nonspecific inanimate referents. For nonspecific actions, the indefinite pronoun $\partial \underline{b} \bar{a} di$ is used.

4. Pronouns

(4.56)	əbād-i	əbod-ø	qamd=in
	deed-INDF	do.IMP.M-SG	for=1SG
	Do something	for me.	

4.4.2 Indefinite Pronouns Derived from Quantifiers

There are a few quantifiers which may take the indefinite morpheme to form indefinite pronouns.

(4.57)	ehdi	one (lit. a one)
(4.58)	perzi	a little, a few
(4.59)	genzi	a lot, many

The indefinite pronoun *ehdi* is derived from the numeral *ehdā* 'one' modified by the indefinite morpheme *-i*. The other two are derived from the non-numeral quantifiers *perzā* 'few/little' and *genzā* 'much/many.' All three can indicate either animate or inanimate referents. These three indefinite take nominal adjuncts with the partitive preposition *pm-*, as in (4.60), or pronominal adjuncts with *men*, as in (4.61).

(4.60)	ehd-i	$m=\bar{a}$	šerš-ān
	one-INDF	from=this	religion-PL
	one of these re	eligions (I.13)	
(4.61)	ehd-i	minn=an	
	one-INDF	from=1PL	one of us

Much like the indefinite pronouns derived from nouns above, these two take ∂m -/min to imply a portion of the noun modified:

(4.62)	perz-i	minn=an	
	few-INDF	from=1PL	a few of us

4.4 Indefinite Pronouns

(4.63)	perzi	əm=mien-ā		
	few-INDF	from=water-AUG	a little of the water ¹⁶⁷	

4.4.3 Compound Indefinite Pronouns

In addition to indefinite pronouns formed from nouns and adjectives with the indefinite morpheme, there are a number of compound indefinite pronouns composed of a quantifier or an interrogative preposition and a second nominal or pronominal element.

(4.64)	kol enši	everybody
(4.65)	kol mendā	everything
(4.66)	kol dukkā	everywhere
(4.67)	kol waxt	every time
(4.68)	kol man	whoever
(4.69)	kol mā	whatever
(4.70)	hemman	whoever

Unlike most indefinite pronouns, *kol man* and *kol mā* can also introduce headless relative clauses (see §4.3.3 above).

In the verb phrase, most indefinite pronouns are negated by the compound indefinite pronoun with $l\dot{a}$, as in (4.72).

(4.71)	-	<i>ašqer-en</i> now.PTC	-	e <i>nši</i> omeon	e	They know somebody.
(4.72)	NEG=1	<i>mbašqer-</i> ND=knov now nobc	w.PTC-3	PL	<i>kol enši</i> everybody i't know eve	erybody).
(4.73)	<i>āhā</i> this	<i>qam</i> to	<i>enši</i> anybody	у	<i>lá=q=m</i> NEG=I	<i>allep̄-ø</i> ND=teach.IPFV-3SG.M

This cannot be taught to anyone (IV.84a-b).

In the compound indefinite pronouns, examples (4.64) - (4.69) above, the original Semitic quantifier *kol* 'all' is replaced by the Persian quantifier *hič* 'any/no' whenever these pronouns serve as the argument of a negative verb.

167 Compare with *perz mienā* 'a little water.' Nouns which follow *perz* are inevitably in the singular, much like Persian *čand*; e.g. *perz beyyi* 'a few eggs.'

4. Pronouns

(4.74)	<i>hič</i> no	<i>waxt</i> time	never
(4.75)	<i>hič</i> no	<i>mend-i</i> thing-INDF	nothing

4.5 Interrogative Clauses

Neo-Mandaic has two different ways of indicating that a given utterance is a request for information rather than a declarative speech act. Requests for a simple yes/no answer are indicated by a rising intonation; e.g. (4.76).

(4.76)	āt qallet?	
	[ət 'qállɛt]	are you going?

Questions that expect a more elaborate response employ a number of interrogative pronouns or question words to elicit specific information, e.g. (4.77) - (4.86)

(4.77)	man	who
(4.78)	mu	what (contextual form <i>mu-</i>)
(4.79)	elyā	where (contextual form <i>elli</i>)
(4.80)	hemdā	when
(4.81)	qamu	why
(4.82)	kammā	how (contextual form kam)
(4.83)	hem	which
(4.84)	mujur	how; in what way
(4.85)	čand	how much
(4.86)	ka <u>t</u> kammā	how much/many (contextual form <i>katkam</i>)

4.5.1 Interrogative Pronouns

The words *man* and *mu* are the only interrogative pronouns in Neo-Mandaic that may substitute for either the subject or the object of a verb. They generally appear at the beginning of the clause, regardless of the function they serve within the phrase. In either case, the interrogative pronoun must be followed by a verb.

4.5.1.1 Interrogative Pronoun as Subject

In example (4.88), the interrogative pronoun *man* 'who' takes the place of the pronoun *huwi* 'he,' which is the subject of the sentence. It remains at the beginning of the clause.

(4.87)		<i>əmal-ø=l=a<u>k</u></i> say.PFV-3SG.M=OBJ=2SG.M	he told you
(4.88)	<i>man</i> who	<i>əmal-ø=l=a<u>k</u> say.</i> PFV-3SG.M=OBJ=2SG.M	who told you?

4.5.1.2 Interrogative Pronoun as Object

In the example (4.90), the interrogative pronoun *mu* what replaces the object $r\bar{a}zi$ 'a secret.' The interrogative pronoun is then fronted to the beginning of the clause, leaving a gap behind, which is here indicated by the null sign \emptyset .

(4.89)	3SG.M	<i>əmal-ø=l=ak</i> say.PFV-3SG.M=OBJ=2SG.M you a secret.	<i>rāzi</i> secret-INDF
(4.90)	what	<i>əmal-ø=l=ak</i> say.PFV-3SG.M=OBJ=2SG.M d he tell you?	Ø

When the interrogative pronoun anticipates the object of the verb rather than the subject, it may be followed by the copula (for which see \$5.1.1.4), as in (4.91). More frequently, however, the copula is omitted in this position, as in (4.92).

(4.91)	<i>mu=yye</i> what=3SG.M What did you giv	<i>hab-t=ell=i</i> give.PFV-2SG=OBJ=3SG.M we him?
(4.92)	<i>mu</i> what What did you giv	<i>hab-t=ell=i</i> give.PFV-2SG=OBJ=3SG.M ve him?

The interrogative pronouns may also introduce a dependent clause as a relative pronoun. In this function, they may appear after the verb, particularly when their referent is specific and definite, as in (4.93).

4. Pronouns

Otherwise, they will appear at the beginning of the sentence, as in (4.94).

- (4.93) $ga! l = \emptyset$ al = man $qa = rahem = \emptyset = l = i$ kill.PFV-3SG.M=3SG.M REF=who IND=love.PTC-3SG.M=OBJ=3SG.M He killed the one whom he loves.

4.5.2 Other Question Words

Other question words in Neo-Mandaic include *elyā* 'where,' *hemdā* 'when,' *qamu* 'why,' *kammā* 'how,' *hem* 'which' (in compounds), *mujur* 'how, in what way,' and *katkammā* 'how much/many.' The interrogatives *elyā*, *hemdā*, *qamu*, *kammā*, and *mujur* are primarily adverbial.

(4.95)	<i>əmal-ø=l=a</i> say.PFV-3SG.M=OBJ=3SG.F He said to her, "Where did he go?" (U	, V-3SG.М
(4.96)	hemdāqə=mwady-āt=anwhenIND=take.IPFV-2"When will you take us?" (IV.122)		
(4.97)	<i>qamu</i> həre₫-ø why be.destroyed.PFV Why was the bridge destroyed? (IV.		<i>pol</i> bridge
(4.98)	<i>pešimān tamm-on kam.</i> penitent become.PFV-3PL how They regretted how they had not sta	prisoner	0.2
(4.99)	<i>mu ā əm=hem</i> what this with=which Which way did they go?	<i>ohhər honni</i> road 3PL	<i>ezg-on</i> go.PFV-3PL
(4.100)	qabul $l\dot{a}=q=\bar{a}\underline{b}ed-\phi$ acceptanceNEG=IND=do.IPFV-3He would not believe how it could here	3SG.M what.w	<i>metø</i> vay die.PFV-3SG.M
(4.101)	katkamm-āhəzi-nihow.much-AUGsee.PFV-We have seen so much misfortune. (<i>t-ā</i> tune-AUG

The interrogatives *hem*, *hemdā*, *mujur*, and *katkammā* are fused compounds;¹⁶⁸ compounds of question words and prepositions are not uncommon in Neo-Mandaic.

(4.102)	-	ī here-AU	G	whence
(4.103)	<i>əm=hen</i> with=wl		<i>ohhər</i> road	whither
(4.104)	<i>min</i> with	<i>mujur</i> what.kii	nd	in comparison with, like

Question words may not introduce relative clauses, this function having been assumed by the interrogative pronouns *man* and *mu* or relativizers such as *ke* and *illi*.

168 These were originally compound pronouns; Noeldeke, Mandäische Grammatik, 94, suggests the first came from a fossilized prepositional phrase such as *hai minhun 'which of them,' and Macuch, Handbook, 160 fn. 30, suggests that hemdā derives from *hai mn 'dan 'which of the time.' Mujur derives from mu 'what' and the Persian word jur 'kind,' and katkammā is likely derived from the classical relative pronoun kd 'like, as' and the question word kma 'how.'

5. The Verb

The verb phrase functions primarily as the predicate of a clause. It consists of a verb, its auxiliaries, its complements, modifiers such as adverbs, and any other adjuncts; it is normally structured in this order when the object is definite and specific:

[SUBJECT \pm] ADVERB \pm VERB \pm OBJECT

If the object is non-specific or generic, it becomes incorporated into the meaning of the verb, and it appears before the verb rather than after it, as described in §3.5.2. The adverb may still intervene between the verb and any noun that has become incorporated into its meaning.

 $[SUBJECT \pm] OBJECT \pm ADVERB \pm VERB$

5.1 The Structure of the Verb

The Neo-Mandaic verb may appear in two aspects: perfective and imperfective; three moods: indicative, subjunctive, and imperative; and three voices: active, middle, and passive. Its inflected forms are built upon two basic stems: the *G*-stem or basic verbal stem for most triradical roots, and the *Q*-stem, which is reserved for verbs built upon four root consonants. The *G*-stem may yield as many five derived stems: the *D*-stem or transitivizing verbal stem, the *C*-stem or causative verbal stem, and the so-called reflexive forms, which were originally derived from the preceding stems with the insertion of an infixed *t*- before the first root consonant, which has since been deleted in most verbs (see \$2.5.4.2 for other examples of this sound change). Each stem normally yields three principle parts: a perfective, an imperfective, and a participle, upon which the derivational and inflectional morphemes are affixed.

The unmodified base forms of the perfective and the imperative are identical with the third masculine singular of the former and the singular of the latter. The imperfective is built upon the active participle of the verb, in the absolute state (§3.4.2), and is inflected in the indicative and subjunctive moods with personal morphemes that agree with its subject in number and gender. The passive voice may either be formed periphrastically from the passive participle (§5.4.3.1) or with an impersonal construction (§5.4.3.2). While all stems can theoretically yield an active participle, only transitive stems yield a passive participle.

Verbs are cited according to their perfective, imperative, and imperfective base forms. The paradigmatic "strong" verb (that is, one which possesses no fewer or more than three consonants that remain stable throughout all inflectional paradigms) is $gatal \sim gatol(gatel)$ 'to kill.'

5.1.1 Verb Classes

Neo-Mandaic verbal roots can be divided into classes according to the vowel patterns in their perfective and imperative base forms (\$5.1.1.1), the composition of their root consonants (\$5.1.1.2), the stems in which they appear (\$5.1.1.3), and the role they play within the syntax of the sentence (\$5.1.1.4).

5.1.1.1 Thematic Vowel Classes

The unmodified *G*-stem perfective and imperfective both comprise two syllables. The vowel of the initial syllable of the root generally succumbs to pretonic reduction (§2.6.2.2), reducing to $/\partial/$, whereas the vowel of the second syllable is preserved. Three vowels are attested in the perfective: /a/, /e/, and /o/. In the imperative, the vowel of this syllable regularly becomes /o/ (or /u/ in the feminine singular imperative). The combination of the thematic vowel of the perfective with the thematic vowel of the imperative yields three thematic vowel classes.

g = a = a = a = b. This is the largest class. Members of this class are predominantly active and often transitive.

Transitive		Intransitive	
bədaq ~ bədoq	to put	hədar ~ hədor	to turn
dəkar ~ dəkor	to remember	nəpāl ~ nəpol	to fall
gəțal ~ gəțol	to kill	nəpāq ~ nəpoq	to exit
həpār ~ həpor	to dig	pərad ~ pərod	to flee
kəbaš ~ kəboš	to subdue/bend	šəbaq ~ šəboq	to leave
kəda <u>b</u> ~ kədo <u>b</u>	to write		
kətap ~ kətop	to fetter		
ləbaš ~ ləboš	to wear		
ləkat ~ ləkot	to seize		
nəkat ~ nəkot	to bite		
zəban ~ zəbon	to buy		
zəraz ~ zəroz	to inform		

A number of transitive verbs, which formerly belonged to other thematic vowel classes, have become associated with this class in the Neo-Mandaic dialect of Khorramshahr.

bəgās ~ bəgos	to stand (CM bgis ~ bgus 'to stay, wait')
gəha <u>k</u> ~ gəho <u>k</u>	to laugh (CM ghik ~ ghuk 'to laugh')

Triradical roots borrowed from other languages such as Arabic are also generally incorporated into this thematic vowel class:

təlab ~	təlob	to request	iəmad ~	jəmod	to freeze

 $d \partial hel \sim d \partial hol (e \sim o)$: This class is much smaller than the former class. Many of the verbs that appear as $(e \sim o)$ verbs in earlier stages of the language and in other dialects have become associated with the $(a \sim o)$ class. It consists mostly of intransitive and stative verbs.

Table 20: $(e \sim o)$ Thematic Vowel Class

Intran	sitive	Stative		
yəhem ~ yəhom	to sit down	dəhel ~ dəhol	to be afraid	
		həre <u>b</u> ~ həro <u>b</u>	to be destroyed	

The third masculine singular prefective of verbs in this class is likely to be confused with the passive participle of the *G*-stem.

kədor ~ *kədor* (o ~ o): The smallest of the three thematic vowel classes, this class consists mainly of statives and intransitives. Macuch identified two verbs of this class in the dialect of Ahvāz, *kədor* ~ *kədor* (*kāder*) 'to be a burden' and *šəkob* ~ *šəkob* (*šākeb*) 'to lie down.' Of these two, only *kədor* ~ *kədor* (*kāder*) is attested in the dialect of Khorramshahr, as the verb *gənā* ~ *gəni* (*gāni*) is normally used in the place of *šəkob* ~ *šəkob* (*šākeb*).

Table 21: $(o \sim o)$ Thematic Vowel Class

Intra	nsitive	Stative		
šəkob ~ šəkob	to lie down	kədor ~ kədor	to be a burden	

5.1.1.2 Root Consonant Classes

Verbs may also be categorized according to the composition of their root consonants. Most verbal roots are composed of three root consonants. In many stems, one or more of these root consonants is susceptible to assimilatory changes or loss in certain environments. These consonants are called "weak consonants." The position of the root consonant is identified by one of three Roman numerals (I, II, and III) corresponding to each of the three root consonants. Three basic root consonant classes exist: the strong verb, roots with weak initial root consonants (I-weak verbs), roots with weak final consonants (IIIweak verbs), and a composite class derived from roots that originally had weak middle consonants (II-weak verbs) or identical second middle and final root consonants (Geminate verbs), the two of which have fallen together in Neo-Mandaic. There are also four verbs defy easy categorization according to root consonants: *abā* (*bāvi*) 'to want' (no imperative); $ezg\bar{a} \sim ezgi (\bar{a}zi)$ 'to go;' tammā 'to become' (no imperative or imperfective); and *yad* 'to know' (a Persian loan word).¹⁶⁹

169 Macuch, *Handbook*, 161 fn. 15 originally derived the root of this pseudo-verb from the participle of the verb $y \partial d\bar{a} \sim y \partial di (y\bar{a}di)$ 'to know,' but it does not behave like any of the other III-weak verbs; if it were derived from this participle rather than the Persian word, the stem would likely take the form $y\bar{a}di$ - before consonant-initial suffixes such as -kon. See §5.1.1.2 below.

Strong Verbs

The largest single root consonant class of verbs is composed of those whose three root consonants are not subject to deletion or assimilation in any position. In Neo-Mandaic, this class also includes verbs beginning with the consonants /n/ or /j/, which are subject to deletion or assimilation in other Aramaic dialects, including Classical Mandaic.

Strong Perfective and Imperative

Both the perfective and imperative stems of the verb may have as many as three allomorphs, the appearance of which depends upon the prefixes and suffixes that are appended to it. Without any suffixes and before suffixes beginning with a consonant, the perfective assumes the form C₂CVC, e.g. (5.1), but before suffixes beginning with a vowel, it assumes the form C₂CC-, e.g. (5.2).¹⁷⁰ After the negative morpheme $l\dot{a}$ - it assumes the form -CCVC-, e.g. (5.3).

(5.1)	gəțalton	you killed
(5.2)	gəțlat	she killed
(5.3)	lágțal	he did not kill

The imperative CəCoC assumes the form CoCC- before object suffixes beginning with vowels, such as (5.4) and the form CəCoCelsewhere, e.g. the plural imperative form (5.6). The feminine singular of the imperative is CəCuC, e.g. (5.5), although this too becomes CoCC- before object suffixes beginning with vowels. Intransitive and stative verbs do not yield the latter form as they do not take object suffixes. The imperative form never receives the negative morpheme, as prohibitions are derived from the subjunctive rather than the imperative.

170 When suffixes beginning with a vowel are added to the perfective, its second syllable becomes open, as the final consonant of the stem becomes the onset of the following syllable. The accent retreats to the preceding closed syllable, necessitating the insertion of /ə/, which becomes raised to $[\varepsilon]$ in the closed syllable. The thematic vowel in the second syllable, which has become open and unaccented, is deleted.

(5.4)	goțli	kill (sg.) him!
(5.5)	gəțul	kill (f. sg.)!
(5.6)	gətolyon	kill (pl.)!

Note also that the presence of a syllable-closing /r/ results in the coloring of the vowel /e/ to /a/ in stressed syllables (see §2.5.1.3). More commonly, however, the vowels which appear are those typical for each allomorph, as in (5.7).

(5.7) <i>porki</i>	rub (sg.) it!
--------------------	---------------

Strong Indicative and Subjunctive

With the exception of the perfective, which is by its nature indicative, and the imperative, which belongs to the broader category of directive modalities, the imperfective is the basis for both the indicative and the subjunctive forms of the verb.

The base of the imperfective is CāCeC, regardless of the thematic vowel class of the corresponding verbal root. In the indicative, the morpheme q- is frequently (but not always) prefixed to this form of the verb, and an anaptyctic vowel intervenes between the morpheme and the initial consonant of the participle, e.g. (5.8). This vowel disappears when the negative morpheme is prefixed, e.g. (5.9). Neither morpheme affects the shape of the active participle.

(5.8) qənāpeq he comes out
(5.9) láqnāpeq he doesn't come out

Before a suffixed morpheme beginning with a consonant, the stress of the active participle is pulled forward to the second syllable and the vowel of the initial syllable is reduced to /a/, e.g. (5.10). Before a suffix beginning with a vowel, the participle assumes the form CāCC-after the syncope of the vowel of the second syllable. This is the form assumed before the feminine morpheme $-\bar{a}$, e.g. (5.11), and the plural morpheme -en, e.g. (5.12). The first and second singular forms of the verb are based upon the masculine active participle, regardless of gender, and the plural inflected forms of the verb are based upon the plural active participle.

(5.10)	gațelnā	I will kill
(5.11)	gāhķā	the laughing (woman)
(5.12)	gāhķen	the laughing (people)

The subjunctive is also based upon the imperfective. Subjunctive verbs generally follow another verb in the contextual form and never take the indicative morpheme. Their forms are largely identical to the contextual forms of the indicative imperfective, with the notable exception that the vowel of its initial syllable, which is open, reduces to /a/ in pretonic position, cf. (5.13) with (5.14). Note that the stress in the subjunctive form remains in the second syllable even though the first person singular suffix has been deleted.

5. The Verb

(5.13) qənapeqnā I will go out
(5.14) qabin nəpeq I want to go out

Personal Suffixes on the Strong Verb

Table 22 illustrates the personal suffixes found on the perfective, the imperative, and the imperfective. Because the syllable structure of the root changes as inflectional morphemes are added to the stem, the syllable structure is indicated to the left of each column. In this column, the syllables are separated by periods (.). In this column, personal suffixes and object suffixes are separated from the stem by dashes (-).

Table 22: Personal Suffixes on the Strong Verb

Person	Perfective	Imperative	Imperfective
3 sg m	$C_1 \mathfrak{a}. C_2 V C_3$	-	$q \partial$ -C ₁ ā.C ₂ eC ₃
3 sg f	$C_1 \Rightarrow C_2$. C_3 -at	-	$q \partial$ -C ₁ āC ₂ .C ₃ - \bar{a}
2 sg	$C_1 a. C_2 a C_3 - t$	$C_1 a. C_2 a C_3$	$q \partial$ -C ₁ \bar{a} C ₂ .C ₃ - et
1 sg	$C_1 \Rightarrow C_2 \cdot C_3 - it$	-	$q \partial$ -C ₁ a.C ₂ eC ₃ - $n \bar{a}$
3 pl	$C_1 \Rightarrow . C_2 V C_3 - yon$	-	$q \partial$ -C ₁ āC ₂ .C ₃ -en
2 pl	$C_1 \Rightarrow . C_2 V C_3 - ton$	$C_1 a. C_2 oC_3$ -yon	$q \ge C_1 a C_2 . C_3 - etton$
1 pl	$C_1 a. C_2 V C_3 - ni$	-	$q $ - $C_1 a C_2 . C_3 - enni$

The third person singular forms of the imperfective derive from the masculine and feminine active participles, respectively, and the plural forms of the imperfective are built upon the plural participle. As

mentioned above, the feminine singular imperative is built on the pattern $C_1 \Rightarrow C_2 u C_3$, e.g. (5.5).

Some of the personal suffixes also have contextual forms, which are used most commonly before complements. The contextual form of the third plural perfective personal suffix and the imperative plural personal suffix is *-on* rather than *-yon*. The first singular form of the imperfective loses its personal suffix entirely and resembles the third masculine singular imperfective, and the first plural form of the imperfective reduces from *-énni* to *-én*, retaining the accent.

The Strong Verb with Object Suffixes

The addition of object suffixes (and occasionally the enclitic object marker -*l*-) results in substantial changes to the form of the preceding personal suffixes and verbal root. Table 23 illustrates the paradigmatic verb $g \partial f a = g \partial f a (g \bar{a} f e l)$ 'to kill' in the perfective with the third masculine singular and second plural object suffixes.

Table 23: The Strong Perfective with Object Suffixes

Person	Verb + V-		Verb + C-	
3 sg m	$C_1 a C_2 C_3$ -	gəțl-i	$C_1 a. C_2 V C_3 -$	gəțal- <u>k</u> on
3 sg f	$C_1 a. C_2 V C_3 - t$ -	gəțal-t-i	$C_1 \Rightarrow C_2 \cdot C_3 - at$ -	gəțl-at- <u>k</u> on
2 sg	$C_1 a. C_2 V C_3 - t$ -	gəțal-t-i	-	-
1 sg	$C_1 a. C_2 V C_3 - t$ -	gəțal-t-i	$C_1 \Rightarrow C_2 \cdot C_3 - it$ -	gəțl-it- <u>k</u> on
3 pl	$C_1 = C_2 \cdot C_3 - on.n$ -	gəțl-onn-i	$C_1 \Rightarrow C_2 \cdot C_3 - on$ -	gəțl-on- <u>k</u> on
2 pl	$C_1 a. C_2 V C_3$ -ton.n-	gəțal-tonn-i	-	-
1 pl	$C_1 \Rightarrow . C_2 V C_3 - n -$	gəțal-n-i	$C_1 a. C_2 V C_3 - ni$ -	gəțal-ni- <u>k</u> on

The second plural object suffix *-kon* is the only object suffix that begins with a consonant. This suffix cannot appear on second person forms of the verb. Note that the first plural perfective with the third masculine singular object suffix is identical to the first plural perfective without any object suffix.

Table 24 illustrates the paradigmatic verb $g = ial \sim g = ial (g = ial \sim g = ial (g = ial \sim g = ial \circ g =$

Table 24: The Strong Imperative with Object Suffixes

Person	Verb + V-	-
2 sg	$C_1 O C_2 . C_3 -$	goțl-i
2 pl	C ₁ oC ₂ .C ₃ -on.n-	goțl-onn-i

Note that the masculine and feminine singular forms of the imperative merge before object suffixes beginning with a vowel. As the only object suffix beginning with a consonant is the third plural, and the imperative cannot take this suffix, these are the only forms given.

Table 25 illustrates the paradigmatic verb $g \partial f a l \sim g \partial f o l (g \bar{a} f e l)$ 'to kill' in the imperfective with the third masculine singular and second plural object suffixes.

Table 25: The Strong Imperfective with Object Suffixes

Person	Verb + V	-	Verb +	C-
3 sg m	<i>qə</i> -C ₁ a.C ₂ eC ₃ - <i>1</i> -	qə-gațel-l-i	$q \rightarrow C_1 a. C_2 eC_3 - l \rightarrow -$	qə-gațel-lə-kon
3 sg f	$q \partial$ -C ₁ aC ₂ .C ₃ - \bar{a} - l -	qə-gațl-ā-l-i	$q \partial$ -C ₁ aC ₂ .C ₃ - \bar{a} - l -	qə-gațl-ā-l- <u>k</u> on
2 sg	$q \partial$ -C ₁ a.C ₂ C ₃ - $\bar{a}t$ -	qə-gațl-āt-i	-	-
1 sg	$q \rightarrow C_1 a. C_2 eC_3 - n - $	qə-gațel-n-i	$q \partial$ -C ₁ a.C ₂ eC ₃ - $n\bar{a}$ -	qə-gațel-nā-kon
3 pl	$q \partial$ -C ₁ aC ₂ .C ₃ - ∂ <i>l</i> - <i>l</i> -	qə-gațl-əl-l-i	$q \partial$ -C ₁ aC ₂ .C ₃ - ∂ l-	qə-gațl-əl- <u>k</u> on
2 pl	qə-C ₁ a.C ₂ eC ₃ -ton.n-	qə-gațel-tonn-i	-	-
1 pl	q -C ₁ a.C ₂ eC ₃ - <i>nann</i> -	qə-gatel-nann-i	$q $ - $C_1 a. C_2 eC_3$ - nan -	qə-gațel-nan- <u>k</u> on

The second plural and singular forms of the imperfective are rare; the imperative is preferred. Note that all third person imperfective forms take the enclitic object marker -*I*- before the object suffix. The final consonant of the third plural personal suffix *-en* regularly assimilates to the object marker (§2.5.1.1).

I-Weak Verbs

This class includes roots that historically began with the phonemes /?/ and /?/, which had merged already in Classical Mandaic, as well as $\partial hab \sim \partial hob (\bar{a}heb)$ 'to give,' the root of which began with /j/ in Classical Mandaic but has since assimilated to this class.

I-Weak Perfective and Imperative

The allomorphs of the I-weak perfective and imperative generally follow the same patterns as those of the strong verbal, with the obvious absence of an initial radical. The bare perfective stem assumes the form $\partial C_2 VC_3$; this is also the form assumed before suffixes beginning with a consonant, e.g. (5.15), but before suffixes beginning with a vowel, it assumes the form aC_2C_3 -, e.g. (5.16).

(5.15) *əbadyon* they did (5.16) *abdat* she did

The vowel which emerges in the initial syllable before suffixes beginning with a vowel is /a/ rather than the /e/ attested in the paradigm of the strong verb. After the negative morpheme $l\dot{a}$ - both allomorphs preserve their form. Before the bare perfective stem $\partial C_2 V C_3$, the vowel of the negative morpheme is generally preserved, rather than elided, e.g. (5.17), as vowels are not elided before /ə/ (see 2.5.5.3).

(5.17) *lá-əbad* he did not do

The imperative assumes the form oC_2C_3 - before object suffixes beginning with vowels, such as (5.18) and the form ∂C_2oC_3 elsewhere, e.g. the plural imperative form (5.19). The feminine singular of the imperative is ∂C_2uC_3 , e.g. (5.20), although this too becomes oC_2C_3 before object suffixes beginning with vowels.

(5.18)	o <u>b</u> di	do (sg.) it!
(5.19)	əbodyon	do (pl.)!
(5.20)	əbud	do (f. sg.)!

I-Weak Indicative and Subjunctive

The conjugation of I-weak participles is not appreciably different from that of the strong verb. In the indicative, the prefix q- is appended directly to the participle, e.g. (5.21). Otherwise, few differences distinguish the conjugation of the I-weak verbs from that of the strong verb.

(5.21) *qābed* he does

Personal Suffixes on the I-Weak Verb

The personal suffixes on I-weak verbs are identical to those found on the strong verb. They are illustrated in Table 26.

Table 26: Personal Suffixes on the I-Weak Verb

Person	Perfective	Imperative	Imperfective
3 sg m	$a.C_2VC_3$	-	q-ā.C ₂ eC ₃
3 sg f	$aC_2.C_3$ -at	-	q -āC ₂ .C ₃ - \bar{a}
2 sg	$a.C_2aC_3-t$	$\partial C_2 O C_3$	q - $\bar{a}C_2.C_3$ - et
1 sg	$aC_2.C_3-it$	-	q-a.C ₂ eC ₃ - <i>nā</i>
3 pl	ə.C ₂ VC ₃ -yon	-	q - $\bar{a}C_2.C_3$ - en
2 pl	$a.C_2VC_3$ -ton	əC20 C3C-yon	q-aC ₂ .C ₃ -etton
1 pl	ə.C ₂ VC ₃ - <i>ni</i>	-	q-aC ₂ .C ₃ -enni

The I-Weak Verb with Object Suffixes

The addition of object suffixes (and occasionally the enclitic object marker -*I*-) results in substantial changes to the form of the preceding personal suffixes and verbal root, much as it does in the paradigm of the strong verb. Table 27 illustrates the perfective with the third masculine singular and second plural object suffixes by means of the verb *əbad* ~ *əbod* ($\bar{a}bed$) 'to do.'

Table 27: The I-Weak Perfective with Object Suffixes

Person	Verb + V-		Verb + C-	
3 sg m	$aC_2.C_3$ -	abd-i	ə.C ₂ VC ₃ -	əbad-kon
3 sg f	ə.C ₂ VC ₃ - <i>t</i> -	ə <u>b</u> ad-t-i	aC ₂ .C ₃ -at-	abd-at-kon
2 sg	ə.C ₂ VC ₃ - <i>t</i> -	ə <u>b</u> ad-t-i	-	-
1 sg	ə.C ₂ VC ₃ - <i>t</i> -	əbad-t-i	aC ₂ .C ₃ - <i>it</i> -	abd-it-kon
3 pl	aC ₂ .C ₃ -on.n-	abd-onn-i	$aC_2.C_3-on-$	abd-on-kon
2 pl	ə.C ₂ VC ₃ -ton.n-	əbad-tonn-i	-	-
1 pl	ə.C ₂ VC ₃ - <i>n</i> -	ə <u>b</u> ad-n-i	ə.C ₂ VC ₃ - <i>ni</i> -	əbad-ni-kon

The paradigm of the imperative of the I-weak verb, as illustrated in Table 28, offers no surprises.

Table 28: The Imperative of the I-Weak with Object Suffixes

Person	Verb + V	7_
2 sg	oC ₂ .C ₃ -	o <u>b</u> d-i
2 pl	oC ₂ .C ₃ -on.n-	obd-onn-i

The paradigms of the verbs *omar* ~ *omor* (*āmer*) 'to say' and *ohab* ~ *ohob* (*āheb*) 'to give' present a few irregularities not found in other I-weak verbs. Both of these verbs take the enclitic *-l*- before object suffixes in all singular forms of the paradigm as well as the third plural forms, and lose their initial syllables in propretonic position. The final consonant of the stem of *omar* ~ *omor* (*āmer*) assimilates to the following sonorant with which it comes into contact (§2.5.1.1). Furthermore, the third and second plural forms are derived from the singular forms, with the addition of the plural morpheme *-on* after the enclitic *-l*- but before the object suffix. Table 29 illustrates the perfective with the third masculine singular and second plural object suffixes.

Table 29: Irregular I-Weak Perfectives with Object Suffixes

Person	Ver	b	Verb + -V	/ -VC	Verb +	CVC
3 sg m	əmar	əha <u>b</u>	əmal-l-i	əha <u>b</u> -l-i	əmal-lə- <u>k</u> on	əha <u>b</u> -lə-kon
3 sg f	amr-at	ah <u>b</u> -at	mār-el-l-i	hāb-el-l-i	mār-əl-kon	hāb-əl-kon
2 sg	əmar-t	əha <u>b</u> -t	mar-t-ell-i	ha <u>b</u> -t-ell-i	-	-
1 sg	amr-it	ah <u>b</u> -it	mar-t-ell-i	ha <u>b</u> -t-ell-i	mar-t-əl- <u>k</u> on	hab-t-əl-kon
3 pl	əmar-yon	əha <u>b</u> -yon	mal-l-onn-i	ha <u>b</u> -l-onn-i	mal-l-on- <u>k</u> on	hab-l-on-kon
2 pl	əmar-ton	əha <u>b</u> -ton	mar-t-onn-i	ha <u>b</u> -t-onn-i	-	-
1 pl	əman-ni	əha <u>b</u> -ni	man-nann-i	ha <u>b</u> -nann-i	əman-na- <u>k</u> on	əha <u>b</u> -na- <u>k</u> on

The forms of the imperative of these two verbs with suffixes, as illustrated in Table 30, are even more irregular.

Table 30: Irregular I-Weak Imperatives with Object Suffixes

Person	Verb		Verb + V-	
2 sg m	əmor	əho <u>b</u>	mol-1-i	bal-l-i
2 sg f	əmur	əhu <u>b</u>	mol-1-i	bal-l-i
2 pl	əmor-yon	əho <u>b</u> -yon	mol-l-onn-i	bal-l-onn-i

Table 31 illustrates the paradigm of the imperfective with the third masculine singular and second plural object suffixes on the verb $\partial bad \sim \partial bod (\bar{a}bed)$ 'to do.' The verbs $\partial mar \sim \partial mor (\bar{a}mer)$ 'to say' and $\partial hab \sim \partial hob (\bar{a}heb)$ 'to give' do not deviate from this paradigm in any significant way, although the final consonant of the stem of $\partial mar \sim \partial mor (\bar{a}mer)$ continues to assimilate when it comes into contact with a following sonorant.

Person	Verb + -V / -VC		Verb + CVC	
3 sg m	<i>q</i> -a.C ₂ eC ₃ - <i>l</i> -	q-abed-1-i	<i>q</i> -a.C ₂ eC ₃ - <i>l</i> ə-	q-abed-lə-kon
3 sg f	<i>q</i> -aC ₂ .C ₃ - <i>ā</i> 1-	q-abd-ā-1-i	<i>q</i> -aC ₂ .C ₃ - <i>āl</i> -	q-abd-ā-l-kon
2 sg	<i>q</i> -a.C ₂ .C ₃ - <i>āt</i> -	q-abd-āt-i	-	-
1 sg	q-a.C ₂ eC ₃ - n -	q-a <u>b</u> ed-n-i	<i>q</i> -a.C ₂ eC ₃ - <i>nā</i> -	q-abed-nā-kon
3 pl	<i>q</i> -aC ₂ .C ₃ - <i>əl.l</i> -	q-a <u>b</u> d-əl-l-i	q-aC₂.C₃-əl-	q-abd-el-kon
2 pl	q-a.C ₂ eC ₃ -ton.n-	q-a <u>b</u> ed-tonn-i	-	-
1 pl	<i>q</i> -a.C ₂ eC ₃ - <i>nann</i> -	q-abed-nann-i	q-a.C ₂ eC ₃ -nan-	q-abed-nan-kon

II-Weak and Geminate Verbs

In contrast to the two classes of roots above, the roots of this class are monosyllabic throughout most of their paradigms, and comprise a large number of verbs from a wide variety of sources. These include roots with /j/ and /w/ as their middle radicals, e.g. $qam \sim qom (q\bar{a}yem)$ 'to stand,' and $met \sim mot (m\bar{a}yet)$ 'to die'; roots originally containing /?/ or /\frac{1}{2}, such as $\bar{s}ayyel \sim \bar{s}ayyel$ ($am\bar{s}iyel$) 'to ask,' which is attested only in the *D*-stem; and roots with geminated second root consonants, also known as geminate roots, such as $ham \sim hom (h\bar{a}yem)$ 'to be hot.' The Classical Mandaic verb sliq ~ saq (saliq) 'to go up' (NM saq ~ soq (sayeq) id.) was also reformed on the analogy of other verbs in this class.

II-Weak and Geminate Perfective and Imperative

The root remains relatively stable throughout the paradigm, with few allomorphs, although the final radical of the root is often geminated to preserve the syllable structure whenever the syllable would otherwise become open (see §2.4.4.2). The verb $met \sim mot (mayet)$ 'to die' is a
unique case. In environments such as those in which the interdental fricative θ would be subject to gemination, the consonant cluster χt is instead produced, e.g. (5.21) instead of (5.22). See §2.4.4.2 for further details on this phenomenon.

(5.21)	me <u>k</u> tat	she died
(5.22)	**me <u>tt</u> at	she died

II-Weak and Geminate Indicative and Subjunctive

Much like the perfective and imperative forms, the forms derived from the participle are fairly stable throughout its conjugation. In the indicative, the morpheme q- is often prefixed to the imperfective form of the verb, and an anaptyctic vowel intervenes between the morpheme and the initial consonant of the imperfective, e.g. (5.23).¹⁷¹ In open accented syllables, the expected diphthong $*\bar{a}y$ collapses to /i/, e.g. (5.24); see §2.3.2.1 for further details about the collapse of this dipthong. Among a subset of verbs within this class, the vowel /i/ breaks in stressed syllables after segments originating in area of the lips or the region between the alveolar ridge and the soft palate, becoming /iĕ/, e.g. (5.25); see §2.3.2.1 for more information on breaking. The subjunctive is also derived from the imperfective, with the obligatory absence of the indicative morpheme.

(5.23)	qəşāyet	he listens
(5.24)	qəşitet	you (sg.) listen
(5.25)	mie <u>tnā</u>	I will die

Personal Suffixes on II-Weak and Geminate Verbs

The personal suffixes attached to the monosyllabic verb roots and participles are identical to those found on the strong verb. In the perfective and imperative, the stem remains monosyllabic throughout the paradigms. In the imperfective, the stem behaves much like the strong roots in the third and first singular persons, but becomes monosyllabic in other persons. In Table 32, the paradigm of this root category with personal suffixes is illustrated.

171 Note that the verb $qam \sim qom (q\bar{a}yem)$ 'to stand' does not take the indicative morpheme.

Person	Perfective	Imperative	Imperfective
3 sg m	C_1VC_3	-	$q \partial$ -C ₁ ā.yeC ₃
3 sg f	$C_1V.C_3$ -at	-	qə-C ₁ i.C ₃ -ā
2 sg	C_1VC_3-t	$C_1 o C_3$	qə-C ₁ i.C ₃ -et
1 sg	$C_1V.C_3$ - <i>it</i>	-	qə-C ₁ i.yeC ₃ -nā
3 pl	C ₁ VC ₃ -yon	-	qə-C ₁ i.C ₃ -en
2 pl	C_1VC_3 -ton	C ₁ oC ₃ -yon	$q $ - $C_1 i.C_3$ -etton
1 pl	C ₁ VC ₃ - <i>ni</i>	-	qə-C ₁ i.C ₃ -enni

Table 32: Personal Suffixes on II-Weak and Geminate Verbs

The addition of the personal suffixes beginning with vowels opens the preceding syllable, lengthening the stem vowel. In the imperfective, the shifting of the accent causes the triphthong $-\bar{aye}$ - to collapse to a monophthong in most forms of the verb.

The feminine singular form of the imperative is $C_1 u C_3$.

 $(5.26) \quad duš \qquad \text{enter (f. sg.)!}$

II-Weak and Geminate Roots with Object Suffixes

As with the other verbal root forms, the addition of object suffixes (and occasionally the enclitic object marker -*I*-) results in substantial changes to the form of the preceding personal suffixes and verbal root. Table 33 illustrates the paradigm of the verb *sat* ~ *sot* (*sāyet*) 'to listen' with the third masculine singular and second plural object suffixes. Normally this verb takes its object with the preposition *min*; when the direct object is appended directly to the verb, it means 'to watch s.o. or s.t.' rather than 'to listen to s.o. or s.t.'

Table 33: II-Weak & Geminate Perfectives w/ Object Suffixes

Person	Verb +	- V-	Verb	+ C-
3 sg m	$C_1V.C_3$ -	<u>ṣāt-i</u>	$C_1V.C_3$ -	șat- <u>k</u> on
3 sg f	C ₁ VC ₃ - <i>t</i> -	sat-t-i	C ₁ V.C ₃ -at-	ṣāt-at-kon
2 sg	C ₁ VC ₃ - <i>t</i> -	sat-t-i	-	-
1 sg	C_1VC_3 -t-	sat-t-i	C ₁ V.C ₃ - <i>it</i> -	ṣāt-it-kon

Table 33 (Cont.):

3 pl	C ₁ V.C ₃ -on.n-	sat-onn-i	$C_1V.C_3$ -on-	sat-on-kon
2 pl	C ₁ VC ₃ -ton.n-	șat-tonn-i	-	-
1 pl	C ₁ VC ₃ - <i>n</i> -	ṣat-n-i	$C_1V.C_3$ -ni-	ṣat-ni- <u>k</u> on

Table 34 illustrates the paradigm of the verb $sat \sim sot (sayet)$ 'to listen' in the imperative with the third masculine singular. Note that the masculine and feminine singular forms of the imperative merge before object suffixes beginning with a vowel.

Table 34: II-Weak & Geminate Imperatives w/ Object Suffixes

Person	Verb + V-		
2 sg	C ₁ o.C ₃ -	șot-i	
2 pl	C ₁ o.C ₃ -on.n-	șot-onn-i	

Table 35 illustrates the paradigm of the verb $sat \sim sot (sayet)$ 'to listen' in the imperfective with the third masculine singular and second plural object suffixes. Note that all third person imperfective forms take the enclitic object marker -*I*- before the object suffix; the final consonant of the third plural personal suffix -*en* assimilates to the enclitic object marker.

Table 35: II-Weak & Geminate Imperfectives w/ Object Suffixes

Person	Verb + -V	/ -VC	Verb +	CVC
3 sg m	qə-C ₁ i.yeC ₃ -1-	qə-siyet-l-i	qə-C ₁ i.yeC ₃ - <i>l</i> ə-	qə-siyet-lə-kon
3 sg f	qə-C ₁ i.C ₃ -ā-1-	qə-șit-ā-l-i	qə-C ₁ i.C ₃ -āl-	qə-șit-ā-l- <u>k</u> on
2 sg	qə-C ₁ i.C ₃ -āt-	qə-șit-āt-i	-	-
1 sg	qə-C ₁ i.yeC ₃ -n-	qə-siyet-n-i	qə-C ₁ i.yeC ₃ -nā-	qə-siyet-nā-kon
3 pl	<i>qə-</i> C ₁ i.C ₃ <i>-əl.l-</i>	qə-șit-əl-l-i	qə-C ₁ i.C ₃ -el-	qə-sit-əl- <u>k</u> on
2 pl	<i>qə</i> -C ₁ iC ₃ - <i>ton.n</i> -	qə-sit-tonn-i	-	-
1 pl	<i>qə</i> -C ₁ iC ₃ - <i>nann</i> -	qə-sit-nann-i	<i>qə</i> -C ₁ i.C ₃ - <i>nan</i> -	qə- <u>ş</u> it-nan- <u>k</u> on

III-Weak Verbs

Much like the II-weak and geminate roots, III-weak roots derive from a variety of sources. These include roots with the semivowels /j/ and /w/ as their third radicals (e.g. $h \partial z \bar{a} \sim h \partial z i$ ($h \bar{a} z i$) 'to see'); roots originally terminating in /?/ and /%/ (e.g. $s \partial b \bar{a} \sim s \partial b i$ ($s \bar{a} b i$) 'to

baptize'); and roots originally ending in /h/ and /ħ/, which merged to /h/ and were eventually lost in this position.¹⁷² The verb $ganab \sim ganob (ganeb)$ 'to steal' also occasionally behaves like a member of this class (in which case it is conjugated exactly like $gana \sim gani$ (gāni) 'to sleep').

III-Weak Perfective and Imperative

The root likewise remains relatively stable throughout the paradigm, with few allomorphs. The basic form of the perfective is $C_1
ildes C_2 \overline{a}$, e.g. (5.27). The masculine singular form of the imperative is $C_1
ildes C_2 i$, e.g. (5.28), and the feminine singular form of the imperative is $C_1
ildes C_2 e$, e.g. (5.29). Before personal suffixes beginning with a vowel, both the perfective and the imperative become $C_1
ildes C_2 -$, e.g. (5.30). Before personal suffixes beginning with a consonant, they become $C_1
ildes C_2 i$, e.g. (5.31). In contrast with the other root consonant classes, the third plural personal suffix of the perfective and the plural personal suffix of the perfective and plural personal suffix of the perfective perfective and the plural personal suffix of the perfective perfective perfective perfective perfective perfective per

(5.27)	həzā	he saw
(5.28)	həzi	see (m.sg.)!
(5.29)	həze	see (m.sg.)!
(5.30)	həzit	I saw
(5.31)	həziton	you (pl.) saw
(5.32)	həzon	they see or see (pl.)!

When preceded by the negative morpheme, the stem assumes the form $-C_1C_2$ -, e.g. (5.33), or $-C_1C_2$ i-, e.g. (5.34).

(5.33)	láhzat	she didn't see
(5.34)	láhziton	you (pl.) didn't see

As noted above, the stem of a III-weak verb always assumes the form $C_1 \Rightarrow C_2$ - before personal suffixes beginning with a vowel. Before the object suffixes that begin with a vowel, the root assumes the form $C_1 \Rightarrow C_2 i$ -, e.g. (5.35). Consult the section on object suffixes for more details.

172 This group is represented by the passive participles *pati* 'empty' (CM **ptia** 'open, wide, broad') and *amšabbā* 'praised' (CM **mšaba** id.)

5.1 The Structure of the Verb

(5.35) *səbyi* he baptized him

This category also includes a few verbs that have another weak radical in addition to the weak final radical, such as the verbs $a\underline{t}\overline{a} \sim do\underline{t}i (\underline{a}\underline{t}i)$ 'to come' and $h \partial w \overline{a} \sim h \partial w i (h \overline{a} w i)$ 'to be.' The latter root behaves like a III-weak verb, as the middle radical is treated as a strong radical in the conjugation of the verb. The former root manifests some of the features of the I-weak roots as well as the III-weak root in the perfective, as illustrated in Table 36.

Table 36: The Perfective and Imperative of atā ~ ati (āti)

	Independent	Verb + V or VC	Verb + CV or CVC
Perfective	atā	a <u>t</u> -	a <u>t</u> i-
Neg. Perfective	látā	la <u>t</u> -	lá <u>t</u> i-
Imperative	do <u>t</u> i	do <u>t</u> -	do <u>t</u> i-

The prefix *d*- found on the imperative of $a\underline{t}\overline{a} \sim do\underline{t}i$ ($\overline{a}\underline{t}i$) is anomalous and not attested in Classical Mandaic. Macuch suggests that it is derived from a "modern Syriac" adverb $d\overline{u}$ 'so, then.'¹⁷³ Another potential candidate is the colloquial Iraqi Arabic particle d(i), which intensifies the meaning of the ordinary form of the imperative.¹⁷⁴ This particle also appears in the Neo-Aramaic dialect of Jilu.¹⁷⁵

III-Weak Indicative and Subjunctive

Much like the perfective and imperative, the imperfective is fairly stable throughout its conjugation. In the indicative, the morpheme q-is optionally prefixed to this form of the verb, and an anaptyctic vowel intervenes between the morpheme and the initial consonant of the participle, as in the strong verb, e.g. (5.36).

(5.36) qədāri he takes

- 173 Macuch, Handbook, 306.
- Salih L. Altoma, *The Problem of Diglossia in Arabic: A Comparative Study of Classical and Iraqi Arabic* (Cambridge, MA: Harvard University Press, 1969),
 See Haim Blanc, *Communal Dialects in Baghdad* (Cambridge, MA: Harvard University Press, 1964), 117 for this particle in the dialect of Baghdad.
- 175 Samuel Ethan Fox, *The Neo-Aramaic Dialect of Jilu* (Wiesbaden: Harrassowitz, 1997), 100, fn. 16.

The stem of the III-weak imperfective, $C_1\bar{a}C_2i$, takes the form $C_1\bar{a}C_2$ before personal suffixes beginning with vowels, e.g. (5.37), and the form $C_1\bar{a}C_2i$ - before object suffixes beginning with vowels, e.g. (5.38) and (5.39), with the exception of the third masculine singular suffix *-i* (see Table 38 below). Note that the first and second plural personal suffixes are disyllabic, and therefore draw the accent away from the initial syllable of the participle, causing the vowel to become reduced to its lax equivalent, /a/, resulting in the pattern C_1aC_2 -, e.g. (5.40).

(5.37)	qəmāhen	they hit
(5.38)	qəmāhya	he hits her
(5.39)	qəhawinā	I will be
(5.40)	qəşabenni	we will baptize

In the imperfective of $a\underline{t}\overline{a} \sim do\underline{t}i$ ($\overline{a}\underline{t}i$) 'to come,' the morpheme q- is usually prefixed directly to the participles that begin with a vowel, e.g. (5.41). The vowel of the initial syllable does not succumb to umlaut, due to the fact that it is regularly uvularized after the imperfective suffix q-. It will, however, be reduced to its lax equivalent /a/ in pretonic position, e.g. (5.42).

(5.41)	qāti	he comes
(5.42)	qatina	I come

Personal Suffixes on the III-Weak Verb

Table 37 illustrates the personal suffixes found on the perfective, imperfective, and the imperative.

Table 37: Personal Suffixes on III-Weak Verbs

Person	Perfective	Imperative	Imperfective
3 sg m	$C_1 aable . C_2 \bar{a}$	-	<i>qə</i> −C₁ā.C₂i
3 sg f	$C_1 a. C_2 - at$	-	<i>qə</i> −C₁āC₂y− <i>ā</i>
2 sg	$C_1 a. C_2 e-t$	C ₁ ə.C ₂ i	qə-C ₁ ā.C ₂ e- t
1 sg	$C_1 a. C_2 - it$	-	<i>qə</i> -C ₁ a.C ₂ i- <i>nā</i>
3 pl	$C_1 \Rightarrow . C_2 - on$	-	$q \partial$ -C ₁ ā.C ₂ - <i>en</i>
2 pl	C ₁ ə.C ₂ i-ton	$C_1 a. C_2 - on$	q ə-C ₁ a.C ₂ - <i>etton</i>
1 pl	C ₁ ə.C ₂ i- <i>ni</i>	-	<i>qə-</i> C ₁ a.C ₂ - <i>enni</i>

5.1 The Structure of the Verb

Much like the personal suffixes found on the strong verb, these personal suffixes also have contextual forms, which are used most commonly before complements. Unlike the strong verb, the third plural form of the perfective does not have a contextual form. The contextual form of the first singular personal suffix of the imperfective is -n rather than $-n\bar{a}$, and the contextual form of the first plural morpheme of the imperfective is $-\acute{en}$ rather than -enni (note that the accent is retained on the first syllable of the morpheme).

The III-Weak Verb with Object Suffixes

The addition of object suffixes (and occasionally the enclitic object marker -*I*-) results in substantial changes to the form of the stem and any personal suffixes that may be attached to it. Table 38 illustrates the paradigm of the verb $h \partial z \bar{a} \sim h \partial z i (h \bar{a} z i)$ 'to see' in the perfective with the third masculine singular and second plural object suffixes.

Table 38: The III-Weak Perfective with	Object Suffixes
--	-----------------

Person	Verb + V-		Verb + C-	
3 sg m	$C_1 a. C_2$ -	həz-i	$C_1 a. C_2 i$ -	həzi- <u>k</u> on
3 sg f	C ₁ ə.C ₂ - <i>āt</i> -	həz-āt-i	C ₁ ə.C ₂ - <i>āt</i> -	həz-āt-kon
2 sg	C ₁ ə.C ₂ - <i>āt</i> -	həz-āt-i	-	-
1 sg	C ₁ ə.C ₂ - <i>it</i> -	həz-it-i	C ₁ ə.C ₂ - <i>it</i> -	həz-it- <u>k</u> on
3 pl	C ₁ ə.C ₂ - <i>on.n</i> -	həz-onn-i	C ₁ ə.C ₂ - <i>on</i> -	həz-on- <u>k</u> on
2 pl	C ₁ ə.C ₂ i- <i>ton.n</i> -	həzi-tonn-i	-	-
l pl	C ₁ ə.C ₂ i- <i>n</i> -	həzi-n-i	C ₁ ə.C ₂ i- <i>n</i> -	həzi-ni- <u>k</u> on

Note that the final vowel of the third masculine singular form of the verb is elided before the third masculine singular object suffix. This also occurs to the first plural form of the verb before the same object suffix, with the result that the form with this object suffix, i.e. "we saw him," is identical to the form with any object suffix, i.e. "we saw." Table 39 illustrates the paradigm of the verb $h \partial z \bar{a} \sim h \partial z i (h \bar{a} z i)$ 'to see' in the imperative with the third masculine singular and second plural object suffixes.

Table 39: The III-Weak Imperative with Object Suffixes

Person	Verb + V-			
2 sg	C ₁ o.C ₂	hoz-i	or	hozy-i
2 pl	C ₁ ə.C ₂ -on.n-	həz-onn-i		

Note that the masculine and feminine singular forms of the imperative merge before the object suffixes. Note also that the final vowel of the third masculine singular form of the verb is elided before the third masculine singular object suffix, as in the perfective, although it occasionally reappears.

Table 40 illustrates the paradigm of the verb $h \partial z \bar{a} \sim h \partial z i (h \bar{a} z i)$ 'to see' in the imperfective with the third masculine singular and second plural object suffixes.

Table 40: The III-Weak Imperfective with Object Suffixes

Person	Verb +	- V-	Verb	+ C-
3 sg m	<i>qə-</i> C ₁ a.C ₂ i <i>-1-</i>	qə-hazi-l-i	<i>qə-</i> C ₁ a.C ₂ i <i>-1-</i>	qə-hazi-l- <u>k</u> on
3 sg f	<i>qə</i> -C₁aC₂.y <i>-āl-</i>	qə-hazy-ā-l-i	<i>qә</i> -C₁aC₂.y <i>-āl-</i>	qə-hazy-ā-l- <u>k</u> on
2 sg	$q \partial$ -C ₁ aC ₂ .y- $\bar{a}t$ -	qə-hazy-āt-i	-	-
1 sg	<i>qə</i> -C ₁ a.C ₂ i- <i>n</i> -	qə-hazi-n-i	<i>qə-</i> C ₁ a.C ₂ i- <i>n</i> -	qə-haz-in- <u>k</u> on
3 pl	<i>qə</i> -C ₁ a.C ₂ - <i>el.1</i> -	qə-haz-el-l-i	<i>qə</i> -C ₁ a.C ₂ - <i>el</i> -	qə-haz-el- <u>k</u> on
2 pl	$q \ge C_1 a C_2 - ton.n$ -	qə-haz-tonn-i	-	-
1 pl	$q \partial$ -C ₁ aC ₂ - <i>nann</i> -	qə-haz-nann-i	<i>qə-</i> C ₁ aC ₂ - <i>nan</i> -	qə-haz-nan- <u>k</u> on

Note that all third person imperfective forms take the enclitic object marker -*I*- before the object suffix; the final consonant of the third plural personal suffix *-en* assimilates to the enclitic object marker. In this particular verb, the final consonant of the stem assimilates in voice to the initial consonant of the personal suffix in the second plural form: *qəhaztonni* [qəhæs¹tonni] 'you see him.'

Irregular Verbs

A number of important verbs defy classification, either because they are deficient in one or more of their principle parts or make use of suppletive stems. These include the verbs $ab\bar{a} (b\bar{a}yi)$ 'to want' (no imperative); $ezg\bar{a} \sim ezgi (\bar{a}zi)$ 'to go;' $tamm\bar{a}$ 'to become' (no

imperative or imperfective); and the pseudo-verbs *yād* 'to know' (a Persian loan word) and *qomb*-.

The Verb abā (bāyi)

The verb $ab\bar{a} (b\bar{a}yi)$ 'to want' generally behaves like $a\underline{t}\bar{a} \sim do\underline{t}i (\bar{a}\underline{t}i)$ 'to come'in the perfective; Table 41 illustrates its forms with the personal suffixes.

Table 41: The Perfective of *abā (bāyi)*

	Singular	Plural
3 m	abā	ab-on
3 f	ab-at	ab-on
2	abe-t	abi-ton
1	ab-it	abi-ni

The verb $ab\bar{a}$ ($b\bar{a}yi$) has no imperative form. When the negative morpheme $l\dot{a}$ - is prefixed to the verbal root, its vowel is elided, e.g. (5.42).

(5.42) *labit* I didn't want

The imperfective of the verb, illustrated in Table 42, is formed with the suppletive stem $b - \frac{W}{V_{v}} - \frac{W}{V_{v}}$.

Table 42: The Imperfective of *abā* (*bāyi*)

	Singular	Plural
3 m	qə-bāyi	qə-beyy-en
3 f	qə-beyyā	qə-beyy-en
2	qə-bā-t	q-ab-etton
1	q-abi-nā	q-ab-enni

The third singular and plural forms of the paradigm appear to be derived from the root $b^{-w}/_{y}^{-w}/_{y}$ with II- and III-weak radical consonants, as is the second singular form. In these forms, an anaptyctic vowel intervenes between the morpheme and the initial consonant. By contrast, all of the other forms appear to be derived from the root ${}^{2}b^{-w}/_{y}$ with I- and III- weak radical consonants like $a\underline{t}\overline{a} \sim do\underline{t}i$ ($\underline{a}\underline{t}i$). In these forms, the indicative morpheme is attached directly

to the participle. No subjunctive or passive forms are associated with this verb.

The Verb ezgā ~ ezgi (āzi)

The conjugation of the verb $ezg\bar{a} \sim ezgi(\bar{a}zi)$ 'to go' resembles that of the III-weak verbs in the perfective and the imperative; Table 43 illustrates the forms of its perfective and imperative forms with the personal suffixes.

Table 43: Perfective and Imperative of ezgā ~ ezgi (āzi)

	Perfective Singular	Perfective Plural	Imperative Singular	Imperative Plural
3 m	ezgā	ezg-on	-	
3 f	ezg-at	ezg-on		
2 m	ezg-et	ezgi-ton	ezgi	ezg-on
2 f	ezg-et	ezgi-ton	ezge	
1	ezg-it	ezgi-ni		

When the negative morpheme $l\dot{a}$ - is prefixed to the verbal root, the vowel of the negative morpheme is elided, e.g. (5.43).

(5.43) *lezgini* we didn't go

The conjugation of the imperfective of the verb, illustrated in Table 44, is characterized by a pair of suppletive forms, much like $ab\bar{a}$ ($b\bar{a}yi$). Unlike $ab\bar{a}$ ($b\bar{a}yi$), however, the second of the two suppletive forms of the imperfective is derived from a separate root, CM **al** ~ '**ul** (**aiil**) 'to enter.' Furthermore, either one of these two forms may be employed for each inflected form of the root, although the forms derived from CM **al** ~ '**ul** (**aiil**) are more common (with the exception of the third masculine singular form, which is not attested).

Table 44: Imperfective of ezgā ~ ezgi (āzi)

	Singular	Plural	Singular	Plural
3 m	q-āzi	q-āz-en	-	q-all-en
3 f	q-āzyā	q-āz-en	q-all-ā	q-all-en
2	q-āze-t	q-az-etton	q-alle-t	q-all-etton
1	q-azi-nā	q-az-enni	q-ali-nā	q-all-enni

The first of the two participles appears to be derived from the root $ezg\bar{a} \sim ezgi(\bar{a}zi)$. The radical consonant /g/ has been deleted, with the result that the participle more closely conforms to the pattern of $a\underline{t}\bar{a} \sim do\underline{t}i(\bar{a}\underline{t}i)$ and other III-weak verbs. Of the inflected forms of these participles, however, only the third masculine singular is regularly used; note the similarity to the paradigm of the verb $h \partial z\bar{a} \sim h \partial zi(h\bar{a}zi)$ above, particularly after the syncope of the initial /h/ (see §2.5.5.1).

By contrast, the suppletive root associated with this verb never appears in the third masculine singular. In fact, no perfective or imperative is associated with this root; it has no independent existence or function apart serving as the imperfective form of this verb. The inflected forms of the imperfective resemble those of the monosyllabic verb, with the exception of the first singular, which conforms to the paradigm of the III-weak verbs. These inflected forms are much more frequently employed in the imperfective of this verb than the forms derived from $ezg\bar{a} \sim ezgi(\bar{azi})$.

The subjunctive forms are derived from the imperfective, without the indicative marker q-. This verb also has a passive participle, *ezgi* 'gone,' e.g. (5.44).

The Verb tammā

The verb *tammā* 'to become' behaves like the III-weak verbs in the perfective; Table 45 illustrates its forms with the personal suffixes.

Table 45: Perfective of tammā

	Singular	Plural
3 m	tammā	tamm-on
3 f	tamm-at	tamm-on
2	tamme-t	tami-ton
1	tamm-it	tami-ni

The contextual form of the third masculine singular is *tam*. As this verb is often followed by its object, this is the form most frequently

encountered. This verb is also frequently found in light verbal constructions (see §5.1.1.4 below).

The verb *tammā* has no imperative or imperfective forms. The functions of the latter are always rendered by the imperfective of $h \partial w \bar{a} \sim h \partial w i (h \bar{a} w i)$ 'to be,' e.g. (5.45), which becomes (5.46) in the imperfective.

(5.45)	<i>tamām</i> finished	<i>tammā-ø</i> become.PFV-3SG.M	it was finished (IV.25)
(5.46)	<i>tamām</i> finished	<i>qə=hāwi-ø</i> IND=be.IPFV-3SG.M	it will be finished

The Pseudo-Verbs yad and qomb

The Persian loan word $y\bar{a}d$ 'memory' is employed in a pseudo-verbal construction, which has seemingly the same semantic range as the verb *bašqer* ~ *bašqer* (*ambašqer*) 'to know,' but is the less common of the two in the dialect of Khorramshahr. All aspects, moods, and voices of this pseudo-verbal construction, save for the imperfective, are formed analytically with the verb $h \partial w\bar{a} \sim h \partial wi (h \bar{a} wi)$ 'to be.' In its form as well as its etymological origins, it is parallel to Persian expressions such as (5.47).

(5.47)	yād-am	bud-ø	
	memory-1SG	is.PFV-3SG	I remembered (my memory was)

In the imperfective, it is generally preceded by the indicative morpheme $q\partial$ -, but takes the enclitic pronominal suffixes (see §4.1.2) rather than the personal suffixes.

Table 46: Imperfective of yād

	Singular	Plural
3 m	qə-yād-i	qə-yād-u
3 f	qə-yād-a	qə-yād-u
2 m	qə-yād-ak	qə-yād-kon
2 f	qə-yād-e <u>k</u>	qə-yād-kon
1	qә-yād-e	qə-yād-an

Frequently the anaptyctic $\langle \vartheta \rangle$ is deleted between the indicative morpheme and the root. Table 47 illustrates the inflected forms of $y\bar{a}d$ in the perfective.

Table 47: Perfective of yād

	Singular	Plural
3 m	qə-yād-i həwā	qə-yād-u həw-on
3 f	qə-yād-a həw-at	qə-yād-u həw-on
2 m	qə-yād-ak həwe-t	qə-yād-kon həwi-ton
2 f	qə-yād-ek həwe-t	qə-yād-kon həwi-ton
1	qə-yād-e həw-it	qə-yād-an həwi-ni

In Persian, the grammatical subject of this construction is $y\bar{a}d$ 'memory,' as in (5.47); note that the copula does not inflect to agree with the logical subject. In Neo-Mandaic, the logical subject of this construction is also the grammatical subject, and the copula inflects to agree with it, as illustrated in Table 47 above. This verb does not appear in the imperative. The pseudoverb *qomb*- 'to be able,' which is treated below with the other modals (introduced in §5.1.1.4), is conjugated in an identical fashion.

5.1.1.3 Verbal Stem Classes

Neo-Mandaic verbs may appear in any one or more of seven verbal stems, which are derived from the root of the verb through various inflectional processes, such as vowel apophony, the gemination of root consonants, and the affixation of derivational morphemes, as well as any combination of the preceding. These seven stems include the two basic stems, the *G*-stem or triradical ground stem, and the *Q*-stem or quadriradical verbal stem, and the five derived stems, the *D*-stem or double stem, characterized by a geminated middle radical, the *C*-stem or causative verbal stem, and the so-called reflexives, the *tG*-, *tD*-, and *tC*-stems, characterized by a prefixed derivational morpheme *t*-. These last three stems express the middle voice of their counterparts in the *G*-, *D*-, and *C*- stems. Most Neo-Mandaic roots appear as *G*-stem verbs, although some are attested only as *D*- or *Q*- stem verbs.

The G-Stem

The *G*-stem or "ground stem," also known as the *Peel* in grammars of Syriac and other Aramaic languages, is the basic stem for most Neo-Mandaic verbal roots. Unless otherwise noted, all the examples cited in the preceding sections have been culled from the paradigm of the *G*-stem.

The D-Stem

The *D*-stem or "double stem," also known as the *Pael*, comprises a large range of verbal roots, including those that also occur as *G*-stem verbs and those that do not occur in any other stem. The perfective and imperative forms of verbs that appear only within the *D*-stem include transitive and intransitive verbs, as illustrated in Table 48.

Table 48: D-Stem Verbs

Transitive		Intransitive	
allep ~ allep (mallep)	to teach	hawwer ~ hawwer	
arreq ~ arreq (marreq)	to spill	(əmhawwer)	to wash up
kammer ~ kammer (əmkammer)	to return	țallel ~ țallel (əmțallel)	to play
šadder ~ šadder (əmšadder)	to organize		
šayyel ~ šayyel (əmšiyel)	to ask		
zahher ~ zahher (əmzahher)	to protect		

Many of the verbal roots loaned from Arabic appear only as *D*-stem verbs. All of the examples given below are transitive.

ḥaddeṯ ~ ḥaddeṯ (əmḥaddeṯ)	to speak
ḥaṣṣel ~ ḥaṣṣel (əmḥaṣṣel)	to obtain
tawweh ~ tawweh (əmtawweh)	to throw
waddi ~ waddi (əmwaddi)	to move

The primary morphological characteristics of the *D*-stem are the gemination of the middle radical and the vowel pattern $C_1aC_2C_2eC_3$ in both the perfective and imperative. The imperfective is formed on the pattern $\Im C_1aC_2C_2eC_3$. With the addition of personal or object suffixes to this stem, the accent is moved from the second syllable of the stem, resulting in the simplification of the geminated second radical to $\Im C_1aC_2eC_3$ - (see §2.6.2.1 for more information on

geminate simplification). Note also that the vowel /e/ is colored to /a/ in stressed syllables whenever it is followed by a syllable-closing /r/ or a pharyngeal consonant (see §2.5.1.3). Consequently, the roots *kammer* ~ *kammer* (*amkammer*), *šadder* ~ *šadder* (*amšadder*), *zahher* ~ *zahher* (*amzahher*) and *tawweh* ~ *tawweh* (*amtawweh*) yield the forms *kamar*-, *šadar*-, *zahar*-, and *tawah*- when the accent is drawn to the second syllable of the root.

G-Stem and D-Stem Verbs

Verbal roots that appear in the D-stem as well as the G-stem are distinguished from their G-stem equivalents semantically, as illustrated in Table 49.

Table 49: G-Stem Roots with D-Stem Equivalents

<i>G</i> -Stem		D-Stem	D-Stem		
yəlep ~ yəlop (yālep)	to learn	allep ~ allep (mallep)	to teach		
qərā ~ qəri (qāri)	to read	qarri ~ qarri (əmqarri)	to call		
zəban ~ zəbon (zāben)	to buy	zabben ~ zabben (əmzabben)	to sell		

While it is difficult to draw any conclusions on the basis of such a small sample size, it would seem that the primary effect of the *D*-stem upon such roots is to increase their transitivity, either by making intransitive verbs transitive, e.g. examples given in (5.48) and (5.49), or by increasing the semantic valence of the verbal root to encompass greater numbers of arguments, e.g. examples (5.50) and (5.51).¹⁷⁶

(5.48)	anā	yəlp-it	lešānā	d=șoprā
	Ι	learn.PFV-1SG	tongue	
	I have	e learned the language of the birds. ¹⁷⁷		

(5.49)	āhā	qam	enši	lá=q=mallep̄-ø
	this	to	anybody	NEG=IND=teach.IPFV-1SG
	I won'	t teach th	is to anyone.	(I.13)

176 Payne, Describing Morphosyntax, 169-75.

177 Macuch, Neumandäische Chrestomathie, 190.

- (5.50) *ani qar-én əlāw šəbih-ān-ā* we read.IPFV-1PL OBJ evil spirit-PL-AUG We will enchant the evil spirits. (IV.16)
- (5.51) q=mqar-el=l=an ehdi $m=\bar{a}$ $\check{s}er\check{s}-\bar{a}n$ ekt-anIND=call.D.IPFV-3PL=OBJ=1PL one from=this religion-PL COP-1PL They label us one of these religions. (I.13)

It is also not apparent from the small sample whether these distinctions between the G-stem and the D-stem represent a productive inflection of the verb or frozen forms restricted purely to the lexicon.

Inflected Forms of the Strong D-Stem

As noted above, the past and present stems of the *D*-stem verb are identical, and the participles are formed with the sufformative *m*-(here ∂m -). Whenever the initial syllable loses the accent, the geminated middle radical is simplified (see §2.6.2.1). Before suffixes beginning with a vowel, the vowel of the second syllable has a tendency to be deleted and the gemination of the middle radical is simplified to accommodate the loss of the second syllable of the stem. Table 50 illustrates the personal suffixes found on the perfective, the imperative, and the imperfective.

Table 50: Personal Suffixes on the Strong D-Stem Verb

Person	Perfective	Imperative	Imperfective
3 sg m	$C_1aC_2.C_2eC_3$	-	q-əm.C ₁ aC ₂ .C ₂ eC ₃
3 sg f	$C_1aC_2.C_3-at$	-	q -əm.C ₁ aC ₂ .C ₃ - \bar{a}
2 sg	$C_1aC_2.C_3-et$	$C_1aC_2.C_2eC_3$	q-əm.C ₁ aC ₂ .C ₃ - et
1 sg	$C_1aC_2.C_3$ - <i>it</i>	-	q-əm.C ₁ a.C ₂ eC ₃ -nā
3 pl	C ₁ a.C ₂ eC ₃ -yon	-	q-əm.C ₁ aC ₂ .C ₃ - en
2 pl	$C_1a.C_2eC_3$ -ton	C ₁ a.C ₂ eC ₃ -yon	q-əm.C ₁ aC ₂ .C ₃ - <i>etton</i>
1 pl	C ₁ a.C ₂ eC ₃ - <i>ni</i>	-	q-əm.C ₁ aC ₂ .C ₃ - <i>enni</i>

Note that the *D*-stem does not distinguish between masculine and feminine forms of the singular imperative, unlike the *G*-stem. Also note that the final radical of this root, /n/, does not assimilate to the following consonant as it does elsewhere (see §2.5.1.1 for further details).

Many of the personal suffixes have contextual forms, which are used most commonly before complements. The contextual form of the first singular form of the imperfective is identical to the third masculine singular of the imperfective, and the contextual form of the first plural form of the imperfective is identical to the third plural form of the imperfective, but remains accented. The contextual form of the imperative plural personal suffix is *-on*. The subjunctive is derived from the imperfective without the indicative marker.

Table 51 illustrates the object suffixes on the strong *D*-stem perfective on the verb *zabben* ~ *zabben* (*amzabben*) 'to sell.' The forms marked by an asterisk are included to illustrate the paradigm, but the use of the second plural personal suffix with this verb was rejected by the informant.

Person	Verb + V	-	Verb	+ C-
3 sg m	$C_1aC_2.C_3$ -	zabn-i	$C_1a.C_2eC_3-$	*zaben- <u>k</u> on
3 sg f	$C_1a.C_2eC_3-t-$	zaben-t-i	$C_1a.C_2eC_3-t-$	*zaben-t- <u>k</u> on
2 sg	$C_1a.C_2eC_3-t-$	zaben-t-i	-	-
1 sg	$C_1a.C_2eC_3-t-$	zaben-t-i	$C_1a.C_2eC_3$ -t-	*zaben-t- <u>k</u> on
3 pl	$C_1aC_2.C_3$ -on.n-	zabn-onn-i	$C_1aC_2.C_3$ -on-	*zabn-on- <u>k</u> on
2 pl	C ₁ a.C ₂ eC ₃ -ton.n-	zaben-tonn-i	-	-
1 pl	$C_1a.C_2eC_3-n-$	zaben-n-i	C ₁ a.C ₂ eC ₃ - <i>ni</i> -	*zaben-ni- <u>k</u> on

The paradigm of the *D*-stem perfective with object suffixes resembles that of the *G*-stem perfective with object suffixes; while the root takes the form $C_1aC_2eC_3$ - or $C_1aC_2C_3$ - depending upon whether it is immediately followed by a suffix with a vowel or a consonant (thus opening or closing the preceding syllable), the form of the personal suffixes and object suffixes remains consistent from paradigm to paradigm. Table 52 illustrates the paradigm of the *D*-stem imperative with object suffixes.

Table 52: The Strong D-Stem Imperative with Object Suffixes

Person	Verb + V-		
2 sg	$C_1aC_2.C_3$ -	zabn-i	
2 pl	C ₁ aC ₂ .C ₃ -on.n-	zabn-onn-i	

Note that the only distinction between the paradigm of the D-stem imperative with object suffixes and that of the G-stem imperative with object suffixes is the form of the root. Table 53 illustrates the paradigm of the D-stem imperfective with object suffixes.

Table 53: The Strong D-Stem Imperfective with Object Suffixes

Person 3 sg m	Verb + V- <i>q</i> -əm.C ₁ a.C ₂ eC ₃ - <i>l</i> - <i>q</i> -əmzaben- <i>l</i> - <i>i</i>	Verb + C- q-əm.C ₁ a.C ₂ eC ₃ - <i>lə-</i> *q-əmzaben-lə- <u>k</u> on
3 sg f	q-əm.C ₁ aC ₂ .C ₃ -āl- q-əmzabn-ā-l-i	<i>q-</i> əm.C ₁ aC ₂ .C ₃ -āl- *q-əmzabn-ā-l- <u>k</u> on
2 sg	<i>q-</i> əm.C ₁ aC ₂ .C ₃ - <i>āt-</i> <i>q-əmzabn-āt-i</i>	-
1 sg	<i>q-</i> əm.C ₁ a.C ₂ eC ₃ <i>-n-</i> <i>q-əmzaben-n-i</i>	<i>q-</i> əm.C ₁ a.C ₂ eC ₃ <i>-nā-</i> * <i>q-əmzaben-nā-<u>k</u>on</i>
3 pl	q-əm.C ₁ aC ₂ .C ₃ -əl.l- q-əmzabn-əl-l-i	<i>q-</i> əm.C ₁ aC ₂ .C ₃ -əl- * <i>q-əmzabn-əl-<u>k</u>on</i>
2 pl	<i>q-</i> əm.C ₁ a.C ₂ eC ₃ <i>-ton.n-</i> <i>q-əmzaben-tonn-i</i>	-
1 pl	<i>q-</i> əm.C ₁ a.C ₂ eC ₃ <i>-nann-</i> <i>q-əmzaben-nann-i</i>	<i>q-</i> əm.C ₁ a.C ₂ eC ₃ <i>-nan-</i> * <i>q-əmzaben-nan-<u>k</u>on</i>

Once again, the forms marked by an asterisk are included to illustrate the paradigm, but the use of the second plural personal suffix with this verb was rejected by the informant. Note that the final radical of this root, /n/, does not assimilate to the following consonant as it does elsewhere (see §2.5.1.1 for further details).

I-Weak Roots in the D-Stem

The paradigm of I-weak verbal roots is identical in most respects to that of the strong verb, with the difference that the sufformative found on the participles assumes the form *m*-, e.g. *allep* ~ *allep* (*mallep*) 'to teach.' The paradigm of the perfective, imperative, and imperfective is illustrated in Table 54.

Table 54: Personal Suffixes on the I-Weak D-Stem Verb

Person	Perfective	Imperative	Imperfective
3 sg m	$aC_2.C_2eC_3$	-	$q \partial$ -maC ₂ .C ₂ eC ₃
3 sg f	$aC_2.C_3$ -at	-	q ə-maC ₂ .C ₃ - \bar{a}
2 sg	$aC_2.C_2eC_3-t$	$aC_2.C_2eC_3$	q a-m aC ₂ .C ₂ eC ₃ - t
1 sg	aC ₂ .C ₃ - <i>it</i>	-	<i>qə-</i> ma.C ₂ eC ₃ - <i>nā</i>
3 pl	a.C ₂ eC ₃ -yon	-	$q \partial$ -maC ₂ .C ₃ -en
2 pl	a.C ₂ eC ₃ -ton	a.C ₂ eC ₃ -yon	q a-maC ₂ .C ₃ -etton
1 pl	a.C ₂ eC ₃ - <i>ni</i>	-	<i>qə</i> -maC ₂ .C ₃ -enni

Table 55 illustrates the objective suffixes on the I-weak *D*-stem perfective, as they appear on the verb $alle\bar{p} \sim alle\bar{p} (malle\bar{p})$ 'to teach.'

Table 55: The I-Weak D-Stem Perfective with Object Suffixes

Person	Verb + V	<i>′</i> _	Verb +	- C-
3 sg m	$aC_2.C_3$ -	alp-i	$a.C_2eC_3-$	alep-kon
3 sg f	$a.C_2eC_3-t-$	alep-t-i	a.C ₂ eC ₃ - <i>t</i> -	alep-t-kon
2 sg	a.C ₂ eC ₃ - t -	alep-t-i	-	-
1 sg	a.C ₂ eC ₃ - t -	alep-t-i	a.C ₂ eC ₃ - <i>t</i> -	alep-t- <u>k</u> on
3 pl	aC ₂ .C ₃ -on.n-	alp-onn-i	$aC_2.C_3$ -on-	alp-on-kon
2 pl	a.C ₂ eC ₃ -ton.n-	alep-tonn-i	-	-
1 pl	a.C ₂ eC ₃ - <i>n</i> -	alep-n-i	a.C ₂ eC ₃ - <i>ni</i> -	alep-ni-kon

The paradigm of the *D*-stem perfective with object suffixes resembles that of the *G*-stem perfective with object suffixes; while the root takes the form aC_2eC_3 - or aC_2C_3 - depending upon whether it is immediately followed by a suffix with a vowel or a consonant (thus opening or closing the preceding syllable), the form of the personal suffixes and object suffixes remains consistent from paradigm to paradigm. Table 56 illustrates the paradigm of the *D*-stem imperative with object suffixes.

Table 56: The I-Weak D-Stem Imperative with Object Suffixes

Person	Verb + V-	
2 sg	aC ₂ .C ₃ -	alp-i
2 pl	aC ₂ .C ₃ on.n-	alp-onn-i

Once again, the only distinction between the paradigm of the D-stem imperative with object suffixes and that of the G-stem imperative with object suffixes is the form of the root. The D-stem Table 57 illustrates the paradigm of the D-stem imperfective with object suffixes.

Table 57: The I-Weak D-Stem Imperfective with Object Suffixes

Person 3 sg m	Verb + V- qə-ma.C ₂ eC ₃ - <i>l</i> - qə-malep-l-i	Verb + C- <i>qə-</i> ma.C ₂ eC ₃ - <i>lə-</i> <i>qə-malep̄-lə-kon</i>
3 sg f	qə-maC ₂ .C ₃ -āl- qə-malp̄-ā-l-i	qə-maC ₂ .C ₃ -ā-l- qə-malp̄-ā-l-k̯on
2 sg	qə-ma.C ₂ C ₃ -āt- qə-malp̄-āt-i	-
1 sg	<i>qə-</i> ma.C ₂ eC ₃ - <i>n</i> - <i>qə-malep̄-n-i</i>	qə-ma.C ₂ eC ₃ -nā- qə-malep̄-nā-kon
3 pl	qə-maC ₂ .C ₃ -əl.l- qə-malp̄-əl-l-i	<i>qə-</i> maC ₂ .C ₃ -əl- qə-malp-əl- <u>k</u> on
2 pl	<i>qə-</i> ma.C ₂ eC ₃ <i>-ton.n-</i> <i>qə-malep̄- tonn-i</i>	-
1 pl	<i>qə-</i> ma.C ₂ eC ₃ <i>-nann-</i> <i>qə-malep̄-nann-i</i>	<i>qə-</i> ma.C ₂ eC ₃ <i>-nan-</i> <i>qə-malep-nan-kon</i>

Inflected Forms of the II-Weak and Geminate D-Stem Verb

Those verbal roots with semivowels as their middle radical, which become monosyllabic throughout most of their *G*-stem paradigms, are conjugated exactly like strong verbs in the perfective and imperative forms of the *D*-stem, e.g. *tawweh* ~ *tawweh* (*omtawweh*) 'he threw.' The chief exception to this is the verb *šayyel* ~ *šayyel* (*omšiyel*) 'to ask,' which is reproduced in Table 58 in its entirety.

Person	Perfective	Imperative	Imperfective
3 sg m	šayyel	-	q-əmšiyel
3 sg f	šiel-at	-	q-əmšiel-ā
2 sg	šayyel-t	šayyel	q-əmšiyel-t
1 sg	šiel-it	-	q-əmšiyel-nā
3 pl	šiyel-yon	-	q-əmšiyel-en
2 pl	šiyel-ton	šiyel-yon	q-əmšiel-etton
1 pl	šiyel-ni	-	q-əmšiel-enni

Table 58: The Paradigm of šayyel ~ šayyel (amšiyel)

The irregularities of the paradigm are due to the collapse of the diphthong /aɪ/. In the forms collected from Sobbi, the original form of the stem with a geminated middle radical is preserved in the perfective and imperfective (cf. CM šaiil 'he asked'), but collapses with the addition of the personal suffixes. With the addition of a suffix beginning with a vowel, the second syllable becomes open and its vowel is syncopated (see §2.6.2.4), resulting in a diphthong (CM šailit 'I asked'). In Neo-Mandaic, this diphthong collapses to /i/ (see §2.3.2.1) and the resulting vowel breaks (see §2.5.3.5). Whenever the addition of a suffix would cause the second syllable to remain closed but receive the accent, the diphthong in the preceding syllable collapses to /i/. In the imperfective, the diphthong has collapsed in every form. No examples of originally geminate roots are attested in the *D*-stem.

Inflected Forms of the III-Weak D-Stem Verb

III-weak verbal roots behave much as they do in the *G*-stem; the vowel of the second syllable, /i/, is preserved before suffixes that begin with a consonant but elided before suffixes beginning with a vowel. Table 59 illustrates the paradigm of the personal suffixes on the III-weak *D*-stem verb.

Person	Perfective	Imperative	Imperfective
3 sg m	$C_1aC_2.C_2i$	-	<i>q-</i> əm.C ₁ aC ₂ .C ₂ i
3 sg f	$C_1aC_2.C_2$ -at	-	q -əm.C ₁ aC ₂ .y- \bar{a}
2 sg	$C_1aC_2.C_2e-t$	$C_1aC_2.C_2eC_3$	q-əm.C ₁ aC ₂ .C ₂ e- t
1 sg	$C_1aC_2.C_2$ - <i>it</i>	-	<i>q-</i> əm.C ₁ a.C ₂ i- <i>nā</i>
3 pl	$C_1a.C_2$ -yon	-	q-əm.C ₁ aC ₂ .C ₂ - en
2 pl	C ₁ a.C ₂ i-ton	C ₁ a.C ₂ eC ₃ -yon	q-əm.C ₁ a.C ₂ - <i>etton</i>
l pl	C ₁ a.C ₂ i- <i>ni</i>	-	<i>q</i> -əm.C ₁ a.C ₂ - <i>enni</i>

Table 59: Personal Suffixes on the III-Weak D-Stem Verb

Note that the geminated second radical is simplified whenever the accent shifts from the first syllable of the stem, e.g. (5.52), or the stem is followed by *-yon*, the third plural personal suffix in the perfective and the plural personal suffix in the imperative. In the former forms, the simplification is accompanied by the umlaut of the vowel of the initial syllable (see §2.5.1.2). The latter forms become $C_1aC_2C_2$ -on in context; the other contextual forms resemble those of the III-weak stem, e.g. (5.53). The feminine singular imperative is built upon the pattern $C_1aC_2C_2e$, e.g. (5.54).

(5.52)	qəmwadin	I will take (contextual)
(5.53)	qəmwadén	we will take (contextual)
(5.54)	qarre	read (f.sg.)!

Table 60 illustrates the objective suffixes on the III-weak *D*-stem perfective, using the verb *waddi* ~ *waddi* (*amwaddi*).

Table 60: The III-Weak D-Stem Perfective with Object Suffixes

Person	Verb +	V-	Verb +	C-
3 sg m	$C_1aC_2.i$ -	wadi-i	C1a.C2i -	wadi- <u>k</u> on
3 sg f	C ₁ a.C ₂ i- <i>t</i> -	wadi-t-i	$C_1a.C_2i$ - <i>t</i> -	wadi-t- <u>k</u> on
2 sg	C ₁ a.C ₂ i- <i>t</i> -	wadi-t-i	-	-
1 sg	C ₁ a.C ₂ i- <i>t</i> -	wadi-t-i	C ₁ a.C ₂ - <i>it</i> -	wad-it- <u>k</u> on
3 pl	C ₁ a.C ₂ -on.n-	wad-onn-i	$C_1a.C_2$ -on-	wad-on- <u>k</u> on
2 pl	C ₁ a.C ₂ i-ton.n-	wadi-tonn-i	-	-
1 pl	$C_1a.C_2i$ - <i>n</i> -	wadi-n-i	$C_1a.C_2i$ - <i>n</i> -	wadi-n- <u>k</u> on

The paradigm of the III-weak *D*-stem perfective with object suffixes resembles that of the III-weak *G*-stem perfective, with the exception of the vowel in the initial syllable. The root assumes the form

 $C_1aC_2C_2$ - or C_1aC_2i - depending upon whether it is immediately followed by a suffix with a vowel or a consonant (thus opening or closing the preceding syllable). The form of the personal suffixes and object suffixes remains consistent from paradigm to paradigm. Table 61 illustrates the paradigm of the *D*-stem imperative with object suffixes.

Table 61: The III-Weak D-Stem Imperative with Object Suffixes

Person	Verb + V	7_
2 sg	C_1aC_2 -	wad-i
2 pl	C ₁ a.C ₂ - <i>on.n</i> -	wad-onn-i

Note that the masculine and feminine forms of the imperative merge before object suffixes. Note also that the vowel of the root is subjected to umlaut in the forms in which the geminated radical is simplified and followed by the vowel /i/. Table 62 illustrates the paradigm of the *D*-stem imperfective with object suffixes.

Table 62: The III-Weak D-Stem Imperfective w/ Object Suffixes

Person 3 sg m	Verb + V- q-əmC ₁ a.C ₂ i- <i>I</i> - q-əmwadi- <i>I</i> -i	Verb + C- <i>q</i> -əmC ₁ a.C ₂ i- <i>l</i> - <i>q</i> -əmwadi- <i>l</i> - <u>k</u> on
3 sg f	q-əmC₁aC₂.y- <i>āl-</i> q-əmwady-ā-l-i	q-əmC ₁ aC ₂ .y <i>-ā-l-</i> q-əmwady-ā-l-kon
2 sg	q-əmC₁a.C₂-āt- q-əmwady-āt-i	-
1 sg	<i>q-</i> əmC ₁ a.C ₂ i <i>-n-</i> <i>q-əmwadi-n-i</i>	<i>q-</i> əmC ₁ a.C ₂ i <i>-nā</i> <i>q-əwadin-<u>k</u>on</i>
3 pl	q-əmC₁a.C₂e <i>-l.l-</i> q-əmwade-l-l-i	<i>q-</i> əmC ₁ aC ₂ .C ₂ e- <i>l-</i> <i>q-əmwade-1-<u>k</u>on</i>
2 pl	<i>q-</i> əmC ₁ aC ₂ - <i>ton.n-</i> <i>q-əmwad-tonn-i</i>	-
1 pl	<i>q-</i> əmC ₁ aC ₂ - <i>nann-</i> <i>q-əmwad-nann-i</i>	<i>q-</i> əmC ₁ aC ₂ - <i>nan-</i> <i>q-əmwad-nan-<u>k</u>on</i>

The Q-Stem

The quadriradical verbs attested in Neo-Mandaic are primarily built upon roots derived from other languages. This category includes a number of fairly common verbs, such as *andeš* ~ *andeš* (*mandeš*) 'to think, worry' (Persian) and *pardas* ~ *pardas* (*əmpardas*) 'to be finished' (Persian). There are also a small number of originally Mandaic verbs, such as *bašqer* ~ *bašqer* (*əmbašqer*) 'to know;' this verb originally meant 'to examine' in Classical Mandaic, and appears to be related to the root *b-q-r* 'to enter into; examine.'¹⁷⁸ The vowel pattern of the *Q*-stem is identical to that of the *D*-stem in all tenses, aspects, and moods, but unlike that of the *D*-stem, the vowel of the second syllable of the *Q*-stem never reduces before vocalic suffixes.

Perfective and Imperative Forms of the Q-Stem Verb

The *Q*-stem of the perfective and imperfective remains fairly stable throughout the conjugation of the verb. Without suffixes or before suffixes which begin with consonants, the stem assumes the pattern $C_1aC_2C_3eC_4$, or, in the case of *pardas* ~ *pardas* (*ampardas*), $C_1aC_2C_3aC_4$. The lax vowel in the second syllable of the root is generally lengthened to its tense equivalent whenever the syllable becomes open and stressed, yielding the pattern $C_1aC_2C_3iC_4$ -, e.g. (5.55). The root *pardas* ~ *pardas* (*ampardas*) has succumbed to paradigmatic pressure, yielding the form *pardis*- rather than the expected ***pardās*-, e.g. (5.56).

(5.55)	andišit	I thought, was concerned
(5.56)	pardisit	I was finished

When the negative morpheme *lá*- is prefixed to the root *andeš*, the vowel is elided, just as it is elided before the I-weak verbal roots (see \$2.5.5.3). Note also that the presence of a syllable-closing /r/ results in the opening of the vowel /e/ to /a/ in stressed syllables, as outlined in \$2.5.1.3, e.g. (5.57).

178 The new root is possibly derived from a metanalysis of the *šaph'el* form of the stem after the metathesis of the causative morpheme and the initial consonant of the root. For the *šaph'el* in Classical Mandaic, see Macuch, *Handbook*, 248.

(5.57) *bašqarni* we knew

Q-Stem Indicative and Subjunctive

Participles derived from the quadriradical verbs assume the form $\Im CaCCeC$. The indicative morpheme *q*- is generally prefixed to the participle, e.g. (5.58). I-weak quadriradical verbs such as *andeš* ~ *andeš* (*mandeš*) 'to be worried' formally resembles the *C*-stem of a strong triradical root (see §5.1.1.3).

(5.58)	qəmbašqarnā	I know
(5.59)	mandeš	worried

The subjunctive is identical to the contextual of the imperfective, without the indicative morpheme. The absence of open syllables precludes any of the vowels from being reduced, as they are in the subjunctive of the strong triradical roots.

Table 63 illustrates the forms of the personal suffixes on the perfective, imperfective, and imperative of the quadriradical verbs.

Table 63: Personal Suffixes on Quadriradical Verbs

Person	Perfective	Imperative	Imperfective
3 sg m	$C_1aC_2.C_3eC_4$	-	q-əm.C ₁ aC ₂ .C ₃ eC ₄
3 sg f	C ₁ aC ₂ .C ₃ i.C ₄ -at	-	q -əm.C ₁ aC ₂ .C ₃ i.C ₄ - \bar{a}
2 sg	$C_1aC_2.C_3eC_4-t$	$C_1aC_2.C_3eC_4$	q-əm.C ₁ aC ₂ .C ₃ i.C ₄ - et
1 sg	C ₁ aC ₂ .C ₃ i.C ₄ - <i>it</i>	-	q -əm.C ₁ aC ₂ .C ₃ eC ₄ - $n\bar{a}$
3 pl	$C_1aC_2.C_3eC_4$ -yon	-	q-əm.C ₁ aC ₂ .C ₃ i.C ₄ -en
2 pl	$C_1aC_2.C_3eC_4$ -ton	C ₁ aC ₂ .C ₃ eC ₄ -yon	q-əm.C ₁ aC ₂ .C ₃ əC ₄ -etton
1 pl	$C_1aC_2.C_3eC_4$ -ni	-	q-əm.C ₁ aC ₂ .C ₃ əC ₄ -enni

Much like the personal suffixes found on the strong verb and the IIIweak verbs, these personal suffixes also have contextual forms, which are used most commonly before complements. The contexual form of the third plural perfective and the third plural imperative personal suffixes is *-on*, e.g. (5.60). The contextual form of the first person singular of the imperfective dispenses with the suffix entirely, e.g. (5.62), and the contextual form of the first person plural suffix of the imperfective *-én*, e.g. (5.64). Note that these forms are easily distinguished from the analogous third person forms, in the paradigm of the verb *bašqer* ~ *bašqer* (*\partialmbašqer*) 'to know,' e.g. (5.61) and (5.63).

(5.60)	bašqiron	they knew; know (pl.)!
(5.61)	qəmbašqer	he knows
(5.62)	qəmbašqar	I know (contextual)
(5.63)	qəmbašqiren	they know
(5.64)	qəmbašqərén	we know (contextual)

The Quadriradical Verb with Object Suffixes

The quadriradical verb behaves much like the strong verb before object suffixes. The stem of the verb remains stable throughout the paradigm, the only variation occurring in the second syllable of the stem, the vowel of which becomes tense whenever that syllable becomes open and stressed.

The C-Stem

The *C*-stem or causative stem, also known as the *Aphel*, comprises a large number of verbal roots, including those that also occur as *G*-stem verbs and those that do not occur in any other stem. The verbal roots attested within the *C*-stem are transitive, as illustrated in Table 64.

Table 64: C-Stem Verbs

abges ~ abges	(mabgēes)	to stop (trans.)
aḥbi ~ aḥbi	(maḥbi)	to keep
ahre <u>b</u> ~ ahre <u>b</u>	(mahre <u>b</u>)	to destroy
amți ~ amți	(<i>mamți</i>)	to bring
andi ~ andi	(<i>mandi</i>)	to shake (trans.)
anhe <u>t</u> ~ anhe <u>t</u>	(<i>manhe<u>t</u></i>)	to bring down
asre <u>k</u> ~ asre <u>k</u>	(masre <u>k</u>)	to light
a <u>tt</u> i ~ a <u>tt</u> i	(<i>ma<u>tt</u>i</i>)	to bring
ayhem ~ ayhem	(mayhem)	to seat
aymi ~ aymi	(maymi)	to promise

The primary morphological characteristics of the *C*-stem are the gemination of the middle radical and the vowel pattern $aC_1C_2eC_3$ in both the perfective and imperative; note that these two forms are

identical, e.g. the *D*- and the *Q*-stems. The imperfective is formed on the pattern $maC_1C_2eC_3$.

G-Stem and C-Stem Verbs

Verbal roots that appear in the C-stem as well as the G-stem are distinguished from their G-stem equivalents semantically, as illustrated in Table 65.

Table 65: G-Stem Roots with C-Stem Equivalents

G-Stem		C-Stem		
atā ~ doti (āti)	to come	a <u>tt</u> i ~ a <u>tt</u> i (ma <u>tt</u> i)	to bring	
bəğeş ~ bəğoş (bāğeş)	to stay	abges ~ abges (mabges)	to make stay	
həreb ~ hərob (hāreb)	to be destroyed	ahreb ~ ahreb (mahreb)	to destroy	
məṭā ~ məṭi (māṭi)	to arrive	amți ~ amți (mamți)	to bring	
nədā ~ nədi (nādi)	to shake (off)	andi ~ andi (mandi)	to make shake	
qam ~ qom (qāyem)	to stand	aqqem ~ aqqem (maqqem)	to raise	
yəhem ~ yəhom (yāhem)	to sit	ayhem ~ ayhem (mayhem)	to seat	
yəmā ~ yəmi (yāmi)	to swear	aymi ~ aymi (maymi)	to promise	

In each case, with the exception of the last, the *C*-stem is the causative equivalent of the *G*-stem, as illustrated in examples (5.65) to (5.68). The final verb, *aymi* ~ *aymi* (*maymi*) 'to promise,' illustrated in examples (5.69) and (5.70), is also found in the *C*-stem with the meaning 'to adjure' in Classical Mandaic, although the initial consonant of the root becomes /w/ in the *C*-stem, not /j/.¹⁷⁹

- (5.65) *darwiš-ā* <u>at-on</u> <u>min</u> <u>ger</u> <u>welāt</u> dervish-PL come.PFV-3PL from outside country Dervishes came from outside the country. (IV.20)
- (5.66) *atti-ø* šehyān-ā barnāš-ā come.C.PFV-3SG.M ruler-AUG person-AUG The ruler brought people

qə=bān-en ha<u>t</u>tā IND=build.IPFV-3PL new who built [it] anew. (IV.17)

179 E.g. maumina 'I adjure;' see Nöldeke, Mandäische Grammatik, 246.

- (5.67) *mət-on qār ahl-ān=d-u* arrive.PFV-3PL at family-PL=of=3PL They arrived at their families' homes. (IV.113)
- (5.68) *amty-ø=u qār yəhem Əmšuni Kuštā* arrive.C.PFV-3SG.M=3PL to sit.PASS.PTC PN He brought them to a settling place in Əmšuni Kuštā. (IV.144)
- (5.69) *yəmā-ø* Čečāwā bə=rāz-ā d=kasy-ā swear.PFV-3SG.M PN on=secret-AUG of=private-AUG Čečāwā swore upon the private secret. (IV.121)
- (5.70) qə=maymi-n=ak an at-it
 IND=swear.C.IPFV-1SG=2SG.M 1SG come.PFV-1SG
 I promise you, I came
 ekkā gāw ḥammām
 thus in hammam
 like this to the hammam. (VIII.18)

The form of the I-Y participles in the *C*-stem, which are not directly inherited from Classical Mandaic but rather reformed on the analogy of the strong verb, suggest that this stem continues to be productive in Neo-Mandaic, rather than a purely lexical phenomenon.

Inflected Forms of the Strong C-Stem

As noted above, the past and present stems of the *C*-stem verb are identical, and the participles are formed with the derivational morpheme m-. Table 66 illustrates the personal suffixes found on the perfective, the imperative, and the imperfective.

Table 66: Personal Suffixes on the Strong C-Stem Verb

Person	Perfective	Imperative	Imperfective
3 sg m	$aC_1.C_2eC_3$	-	q a-maC ₁ .C ₂ eC ₃
3 sg f	$aC_1.C_2e.C_3-at$	-	$q a$ -maC ₁ .C ₂ i.C ₃ - \bar{a}
2 sg	$aC_1.C_2eC_3-t$	$aC_1.C_2eC_3$	q a-maC ₁ .C ₂ eC ₃ - t
1 sg	$aC_1.C_2e.C_3-it$	-	$q a$ -maC ₁ .C ₂ eC ₃ - $n \bar{a}$
3 pl	aC ₁ .C ₂ e.C ₃ -yon	-	q a-m aC ₁ .C ₂ i.C ₃ - <i>en</i>
2 pl	$aC_1.C_2e.C_3$ -ton	aC ₁ .C ₂ eC ₃ -yon	q a-maC ₁ .C ₂ a .C ₃ - <i>etton</i>
1 pl	$aC_1.C_2eC_3-ni$	-	q a-maC ₁ .C ₂ a .C ₃ - <i>enni</i>

The stem remains fairly stable throughout the paradigm; the feminine participle, upon which the third feminine singular imperfective is based, assumes the form maCCiCā; the plural participle, which is the basis for the plural form, maCCiCen, becomes maCCəCen- when the accent moves from the second syllable of the stem. The *C*-stem of strong roots does not distinguish between masculine and feminine forms of the singular imperative, like the *D*-stem of strong roots but unlike the *G*-stem of strong roots. In this particular verb, the final radical of the root assimilates in voice to the the consonant that immediately follows it, e.g. *ahrebton* [ah'refton] 'you destroyed.'

As with these other stems, some of the personal suffixes have contextual forms. The contextual form of the third plural perfective and the plural imperative personal suffixes is *-on*, e.g. (5.71). Unlike the *G*-stem, the contextual forms of the first person singular (5.73) and plural (5.75) of the imperfective do not fall together with the third person singular (5.72) and plural (5.74).

(5.71)	ahre <u>b</u> on	they destroyed; destroy (pl.)!
(5.72)	qəmahre <u>b</u>	he destroys
(5.73)	qəmahrem	I destroy (contextual) ¹⁸⁰
(5.74)	qəmahriben	they destroy
(5.75)	qəmahrə <u>b</u> én	we destroy (contextual)

Table 67 illustrates the object suffixes on the strong *C*-stem perfective using the verb *ahreb* ~ *ahreb* (*mahreb*) 'to destroy.'

Table 67: The Strong C-Stem Perfective with Object Suffixes

Person	Verb + -V / -	VC	Verb + C	VC
3 sg m	aC1.C2iC3-	ahri <u>b</u> -i	$aC_1.C_2eC_3-$	ahre <u>b-k</u> on
3 sg f	$aC_1.C_2eC_3-t-$	ahre <u>b</u> -t-i	$aC_1.C_2eC_3-t-$	ahre <u>b</u> -t- <u>k</u> on
2 sg	$aC_1.C_2eC_3-t-$	ahre <u>b</u> -t-i	-	-
1 sg	$aC_1.C_2eC_3-t-$	ahre <u>b</u> -t-i	$aC_1.C_2eC_3-t-$	ahre <u>b</u> -t- <u>k</u> on
3 pl	aC1.C2e.C3-on.n-	ahre <u>b</u> -onn-i	$aC_1.C_2 a.C_3$ -on-	ahrə <u>b</u> -on- <u>k</u> on
2 pl	aC ₁ .C ₂ eC ₃ -ton.n-	ahre <u>b</u> -tonn-i	-	-
l pl	$aC_1.C_2eC_3-n-$	ahre <u>b</u> -n-i	aC ₁ .C ₂ eC ₃ -ni-	ahre <u>b</u> -ni- <u>k</u> on

180 The /n/ of the first person suffix is usually preserved in the *C*-stem contextual forms; e.g. *qəmabğeşşən* 'I stopped.'

The paradigm of the *C*-stem perfective with object suffixes resembles that of the *G*-stem perfective with object suffixes; while the stem takes the form $aC_1C_2iC_3$ - whenever the second syllable is open and accented, the form of the personal suffixes and object suffixes remains consistent from paradigm to paradigm. Table 68 illustrates the paradigm of the *C*-stem imperative with object suffixes.

Table 68: The Strong C-Stem Imperative with Object Suffixes

Person	Verb + V-	-
sg.	$aC_1.C_2i.C_3$ -	ahri <u>b</u> -i
pl.	aC ₁ .C ₂ ə.C ₃ -on.n-	ahrə <u>b</u> -onn-i

Note that there is no distinction between the masculine and feminine forms of the imperative in the *C*-stem. Table 69 illustrates the paradigm of the *C*-stem imperfective with object suffixes.

Table 69: The Strong C-Stem Imperfective with Object Suffixes

Person 3 sg m	Verb + -V / -VC qə-maC ₁ .C ₂ eC ₃ - <i>l</i> - qə-mahre <u>b</u> - <i>l</i> - <i>i</i>	Verb + CVC qə-maC ₁ .C ₂ eC ₃ -lə-n qə-mahreb-lə-kon
3 sg f	qə-maC ₁ .C ₂ ə.C ₃ -ā-l- qə-mahrə <u>b</u> -ā-l-i	<i>qə-</i> maC ₁ .C ₂ ə.C ₃ -ā-l- qə-mahrə <u>b</u> -ā-l- <u>k</u> on
2 sg	qə-maC ₁ .C ₂ ə.C ₃ -āt- qə-mahrə <u>b</u> -āt-i	-
1 sg	qə-maC ₁ .C ₂ eC ₃ -n- qə-mahre <u>b</u> -n-i	<i>qə-</i> maC ₁ .C ₂ eC ₃ - <i>nā</i> - <i>qə-mahreb-nā-kon</i>
3 pl	<i>qə-</i> maC ₁ .C ₂ ə.C ₃ -əl-l- qə-mahrə <u>b</u> -əl-l-i	<i>qə-</i> maC ₁ .C ₂ ə.C ₃ -əl- qə-mahrə <u>b</u> -el- <u>k</u> on
2 pl	<i>qə-</i> maC ₁ .C ₂ eC ₃ <i>-ton.n-</i> <i>qə-mahre<u>b</u>-tonn-i</i>	-
1 pl	<i>qə-</i> maC ₁ .C ₂ eC ₃ -nann- qə-mahre <u>b</u> -nann-i	<i>qə-</i> maC ₁ .C ₂ eC ₃ - <i>nan-</i> <i>qə-mahreb-nan-kon</i>

Inflected Forms of the I-Weak C-Stem Verb

The paradigm of I-weak verbal roots in the *C*-stem has fallen together with that of the *D*-stem. Whenever the accent falls on a syllable containing a lax vowel, like the initial syllable of the strong verbal roots in the *C*-stem, the onset of the following syllable is geminated (which corresponds to the second root consonant) and the accented syllable becomes closed (see 2.4.4.2). As the accent shifts, the gemination is lost (see 2.6.2.1), just as it is in the *D*-stem.

The only example of a I-weak verb in the C-stem attested in the texts, $\underline{atti} \sim \underline{atti} (\underline{matti})$ 'to bring,' is doubly weak, and will be discussed below with the III-weak verbs.

Inflected Forms of the II-Weak and Geminate C-Stem Verb

Those verbal roots with semivowels as their middle radical, which become monosyllabic throughout most of their *G*-stem paradigms, resemble the I-weak verbs in their conjugation, e.g. $aqqem \sim aqqem$ (maqqem) 'to raise,' from $qam \sim qom$ ($q\bar{a}yem$) 'to stand.' When the stress falls upon the initial syllable, which contains a lax vowel, the onset of the following syllable (here corresponding to the first root consonant rather than the second root consonant) is geminated and the accented syllable becomes closed. When stress moves away from the initial syllable of the root, the gemination is lost. No examples of geminate verbs were attested in the *C*-stem.

Inflected Forms of the III-Weak C-Stem Verb

III-weak verbal roots behave much as they do in the *G*-stem; the vowel of the second syllable, /i/, is preserved before suffixes that begin with a consonant but elided before suffixes beginning with a vowel. Table 70 illustrates the paradigm of the personal suffixes on the III-weak *C*-stem verb.

Person	Perfective	Imperative	Imperfective
3 sg m	aC1.C2i	-	<i>qə-</i> maC ₁ .C ₂ i
3 sg f	$aC_1.C_2$ - <i>at</i>	-	q ə-maC ₁ .C ₂ - \bar{a}
2 sg	$aC_1.C_2e-t$	aC1.C2i	$q \partial$ -maC ₁ .C ₂ - et
1 sg	$aC_1.C_2-it$	-	<i>qə-</i> maC ₁ .C ₂ i- <i>nā</i>
3 pl	aC ₁ .eC ₂ -yon	-	$q \partial$ -maC ₁ .C ₂ - en
2 pl	aC ₁ .eC ₂ -ton	aCeC-yon	q a-maC ₁ .C ₂ -etton
l pl	aC ₁ .eC ₂ - <i>ni</i>	-	<i>qə-</i> maC ₁ .C ₂ - <i>enni</i>

Table 70: Personal Suffixes on the III-Weak C-Stem Verb

Note the two chief allomorphs of the stem, aCeC- before personal suffixes beginning with consonants, and aCC- before those beginning with vowels. Like III-weak verbs in the *D*-stem, III-weak verbs in the *C*-stem distinguish between masculine and feminine forms of the imperative; the feminine imperative has the form aCCe, e.g. *ahbe* put away (f.sg.)! The contextual forms of the third plural perfective and the plural imperative are identical to their lexical forms. The contextual form of the first singular imperfective (5.76) is distinguished from the third singular imperfective (5.76), but the third plural imperfective (5.78) and the contextual form of the first plural imperfective (5.79) are distinguished by accent alone, as in the *G*-stem.

(5.76)	qəmahbi	he keeps
(5.77)	qəma <u>h</u> bin	I keep (contextual)
(5.78)	qəma <u>h</u> ben	they keep
(5.79)	qəma <u>h</u> bén	we keep (contextual)

Table 71 illustrates the objective suffixes on the perfective of the IIIweak *C*-stem verb *ahbi* ~ *ahbi* (*mahbi*) 'to keep.'

Table 71: The III-Weak C-Stem Perfective with Object Suffixes

Person	Verb + -V / -VC		Verb $+ l$ - or CVC	
3 sg m	$aC_1.C_2$ -	a <u>ḥb</u> -i	$aC_1.C_2i$ -	aḥbi-kon
3 sg f	a.C ₁ eC ₂ - t -	aḥeḇ-t-i	a.C ₁ eC ₂ - t -	aḥeḇ-t-k̯on
2 sg	a.C ₁ eC ₂ - t -	aḥeḇ-t-i	-	-
1 sg	a.C ₁ eC ₂ - t -	aḥeḇ-t-i	a.C ₁ eC ₂ - t -	aḥeḇ-t-k̯on
3 pl	aC ₁ .C ₂ - <i>on.n</i> -	aḥb-onn-i	$aC_1.C_2-on-$	aḥb-on-kon
2 pl	a.C ₁ eC ₂ - <i>ton.n</i> -	aḥeḇ-tonn-i	-	-
1 pl	a.C ₁ eC ₂ - <i>n</i> -	aḥeḇ-n-i	a.C ₁ eC ₂ - ni -	aḥeḇ-ni-k̯on

The III-weak *C*-stem perfective is distinguished by several traits. The first is the elision of the final vowel of the third masculine singular form before the third masculine singular object suffix,¹⁸¹ with the result that it resembles the bare stem of the third masculine singular form. The second is the insertion of an anaptyctic vowel between the last two root consonants before suffixes beginning with a consonant, thus preventing a cluster from forming, and giving much of the paradigm (with the exception of the third masculine singular form) the appearance of a I-weak *D*-stem perfective.

The C-stem verb $\underline{atti} \sim \underline{atti} (\underline{matti})$ 'to bring,' which also has a weak first radical, also resembles the D-stem in the perfective. It is illustrated with the object suffixes in Table 72.

Table 72: Perfective of atti ~ atti (matti) with Object Suffixes

Person	Verb + -V / -VC		Verb + l - or CVC	
3 sg m	a.C ₂ i-	a <u>t</u> -i	a.C ₂ i-	ati-kon
3 sg f	a.C ₂ i- <i>t</i> -	a <u>t</u> i-t-i	a.C ₂ i- <i>t</i> -	a <u>t</u> i-t- <u>k</u> on
2 sg	a.C ₂ i <i>-t</i> -	a <u>t</u> i-t-i	-	-
1 sg	a.C ₂ i <i>-t</i> -	a <u>t</u> i-t-i	a.C ₂ i- <i>t</i> -	a <u>t</u> i-t- <u>k</u> on
3 pl	a.C ₂ - <i>on.n</i> -	a <u>t</u> -onn-i	a.C ₂ - <i>on</i> -	a <u>t</u> -on- <u>k</u> on
2 pl	a.C ₂ i- <i>ton.n</i> -	a <u>t</u> i-tonn-i	-	-
1 pl	a.C ₂ i- <i>n</i> -	a <u>t</u> i-n-i	a.C ₂ i- <i>n</i> -	ati-n-kon

As a result of the simplification of the geminated consonant, the paradigms of the *C*-stem verb $\underline{atti} \sim \underline{atti} (\underline{matti})$ 'to bring' resembles that of the *G*-stem verb $\underline{ata} \sim \underline{ati} (\overline{ati})$ 'to come' before the object suffixes; fortunately the latter does not take object suffixes, and therefore no semantic confusion between the two is possible. Table 73 illustrates the paradigm of the *C*-stem imperative with object suffixes.

Table 73: The III-Weak C-Stem Imperative with Object Suffixes

Person	Verb + V-		
2 sg	$aC_1.C_2$ -	aḥb-i	
2 pl	aC ₁ .C ₂ - <i>on.n</i> -	a <u>ḥb</u> -onn-i	

181 But not, apparently, before the third masculine plural suffix; e.g. *amtyu* 'he brought them' (IV.144).

5. The Verb

Note that the feminine and the masculine forms of the singular imperative merge before the object suffixes. The paradigm of the imperative of the doubly weak verb $a\underline{tti} \sim a\underline{tti} (ma\underline{tti})$ with object suffixes is given in Table 74.

Table 74: Imperative of atti ~ atti (matti) with Object Suffixes

Person	Verb + V-	
2 sg	a.C ₂ -	a <u>t</u> -i
2 pl	a.C ₂ -on.n-	a <u>t</u> -onn-i

Table 75 illustrates the paradigm of the *C*-stem imperfective with object suffixes.

Table 75: The III-Weak C-Stem Imperfective w/ Object Suffixes

Person 3 sg m	Verb + -V / -VC <i>qə-</i> ma.C ₁ eC ₂ - <i>l-</i> <i>qə-maḥeb॒-l-i</i>	Verb + CVC qə-ma.C ₁ eC ₂ - <i>lə-</i> qə-maḥeb-lə-kon
3 sg f	<i>qə-</i> maC₁.C₂- <i>ā-l-</i> <i>qə-maḥḇ-ā-l-i</i>	<i>qə-</i> maC₁.C₂- <i>ā-l-</i> <i>qə-maḥḇ-ā-l-k॒on</i>
2 sg	<i>qə-</i> maC₁.C₂- <i>āt-</i> <i>qə-maḥḇ-āt-i</i>	-
1 sg	<i>qə-</i> ma.C ₁ e.C ₂ - <i>n</i> - <i>qə-maḥe<u>b</u>-n-i</i>	<i>qə-</i> ma.C ₁ e.C ₂ - <i>nā-</i> <i>qə-maḥeb॒-nā-k॒on</i>
3 pl	<i>qə-</i> maC ₁ .C ₂ - <i>el-l-</i> <i>qə-maḥ<u>b</u>-el-l-i</i>	<i>qə-</i> maC ₁ .C ₂ - <i>el-</i> <i>qə-maḥ<u>b</u>-el-<u>k</u>on</i>
2 pl	<i>qə</i> ma.C ₁ eC ₂ - <i>ton.n-</i> <i>qə-maḥe<u>b</u>-tonn-i</i>	-
1 pl	<i>qə-</i> ma.C ₁ eC ₂ - <i>nann-</i> <i>qə-maḥe<u>b</u>-nann-i</i>	<i>qə</i> -ma.C ₁ eC ₂ - <i>nan-</i> <i>qə-maḥeb॒-nan-kon</i>

Once again, the final vowel is elided before personal suffixes, and an anaptyctic vowel is inserted between the first and the second radical, giving the verb the appearance of a I-weak *D*-stem verb.

Table 76 illustrates the *C*-stem imperfective paradigm of $a\underline{t}\underline{t}i \sim a\underline{t}\underline{t}i$ (*matti*) with object suffixes. Note again the resemblance to the paradigm of the *D*-stem imperfective.

Person	Verb + -V /	-VC	Verb + <i>l</i> -	or CVC
3 sg m	<i>qə-</i> ma.C ₂ i <i>-1-</i>	qə-ma <u>t</u> i-l-i	<i>qə-</i> ma.C ₂ i- <i>l-</i>	qə-mati-l-kon
3 sg f	<i>qə-</i> ma.C ₂ i <i>-ā-l-</i>	qə-mati-ā-l-i	<i>qə-</i> ma.C₂i- <i>āl-</i>	qə-mati-ā-l-kon
2 sg	<i>qə-</i> ma.C ₂ <i>-āt-</i>	qə-ma <u>t</u> -āt-i	-	-
1 sg	<i>qə-</i> ma.C ₂ i- <i>n</i> -	qə-ma <u>t</u> i-n-i	<i>qə-</i> ma.C ₂ i- <i>nā</i> -	qə-ma <u>t</u> i-n- <u>k</u> on
3 pl	<i>qə-</i> ma.C ₂ - <i>el-l-</i>	qə-ma <u>t</u> -el-l-i	<i>qə-</i> ma.C ₂ - <i>el-</i>	qə-ma <u>t</u> -el- <u>k</u> on
2 pl	<i>qə-</i> ma.C ₂ i- <i>ton.n-</i>	qə-ma <u>t</u> i-tonn-i	-	-
1 pl	<i>qə-</i> ma.C ₂ i <i>-n-</i>	qə-ma <u>t</u> i-n-i	<i>qә-</i> ma.C₂i <i>-n-</i>	qə-ma <u>t</u> i-n- <u>k</u> on

Table 76: Imperfective of atti ~ atti (matti) with Object Suffixes

The Reflexive Stems

The so-called "reflexive" stems were originally formed with a derivational morpheme *t*-, which was prefixed to the root of the verb. This derivational morpheme was metathesized with the following radical whenever the root began with a sibilant, e.g. *estəbā* ~ *estəbi* (*mestəbi*) 'to be baptized.' In all other roots, the derivational morpheme remained in its position before the root and subsequently succumbed to oral depletion (see §2.5.4.2).

Table 77: tG-Stem Verbs

ehter ~ ehter	(mehter)	to be pleased (from <i>h-<u>t</u>-r</i> 'to rejoice')
epseq ~ epseq	(mepseq)	to become extinct, lit. to be cut off
eșțəbā ~ eșțəbi	(meșțəbi)	to be baptized
etber ~ etber	(met <u>b</u> er)	to break (intrans.)

The *tG*-stem verbs resemble those in the *C*-stem, save for the forms in which the derivational morpheme has been preserved. For the majority of verbs (i.e. those in which the morpheme has been lost), the vowel pattern is $eC_1C_2eC_3$ in both the perfective and the imperative of the *tG*-stem. The participle derived from this root is built on the pattern meC_1C_2eC_3.

The *tD*-stem survives in two verbs, *kammar* ~ *kammar* (*mekammar*) 'to turn back' and *barrak* ~ *barrak* (*mebarrak*) 'to pray.'¹⁸² The vowel of the final syllable and the form of the

¹⁸² Macuch (*Neumandäische Texte*, 65) also sites this verb, in addition to one other, *eštallem* ~ *eštallem* (*meštallem*) 'to be welcome,' as examples of *tD*-

derivational morpheme *me*- set this verb apart from the regular *D*stem paradigm. The *tC*-stem is also represented, albeit only by a single unambiguous example, *etaqlab* ~ *etaqlab* (*metaqlab*) 'to turn around' (originally from Arabic).¹⁸³ In addition, the verb *ettar* ~ *ettar* (*mettar*) 'to wake up' may also belong to the *tC*-stem. Macuch classifies this verb as a *tC*-stem verb due to the pattern of its vowels and cognate forms in other Aramaic languages, e.g. Syriac '*ichd*'*c*''' 'to wake up.'¹⁸⁴ Unfortunately, this root is not attested in Classical Mandaic, but a secondary root (presumably derived from the original *tC*-stem) is attested in the *G*-stem, '**tar** ~ '**tar** (*mettar*) 'to wake up.' Consequently, it is not clear whether *ettar* ~ *ettar* (*mettar*) 'to wake up' is a *tG*-stem form derived from the secondary root or a *tC*stem form derived from the original root, which is unattested in Classical Mandaic.

The lack of other examples of tD- and tC-stem verbs can just as easily be attributed to the small size of the material collected as the obsolescence of these particular stems. Pending further research, their paradigms will not be reproduced.

G-Stem and tG-Stem Verbs

Verbal roots that appear in the tG-stem as well as the G-stem are distinguished from their G-stem equivalents semantically, as illustrated in Table 78.

Table 78: G-Stem Roots with tG-Stem Equivalents

G-Stem		<i>tG</i> -Stem	
pəsaq ~ pəsoq (pāseq)	to cut	epseq ~ epseq (mepseq)	to die out
ṣəbā ~ ṣəbi (ṣābi)	to baptize (trans.)	eșțəbā ~ eșțəbi (meșțəbi)	to baptize (intr.)
təbar ~ təbor(tāber)	to break (trans.)	etber ~ etber (metber)	to break (intr.)

stem verbs in the Neo-Mandaic dialect of Ahwāz. The latter verb was not known to my chief informant.

¹⁸³ Macuch, *Neumandäische Texte*, 65–66. Macuch also sites this verb and an aditional verb, *etarraq* ~ *etarraq* (*metarraq*) 'to be poured,' as examples of the tC-stem. The latter was not recognized by my chief informant.

¹⁸⁴ Macuch, Neumandäische Texte, 76.
Contrary to Macuch's observations,¹⁸⁵ the *tG*-stem primarily expresses the middle voice, not the reflexive, of the *G*-stem, as examples (5.80) and (5.81) demonstrate. Note that the grammatical subject of the verb in (5.81) is both the agent of the action described by the verb (the baptizers) as well as the patient (the baptized), thus eliminating the distinction between these two. As a result, translations of verbs in this stem have ranged from the reflexive (e.g. they baptized themselves) to the passive (e.g. they were baptized). Neither truly captures the sense of this stem; note that the impersonal passive (§5.4.3.2), which is the most productive form of the passive in Neo-Mandaic, implies an agent, even though the role of the agent is downplayed, e.g. (5.82).

(5.80)	<i>Hibel Ziwā</i> PN Hibel Ziwā bapt	<i>ṣəbi-@=ø</i> baptize.PFV-3SG.M=3 ized Ādam. (I.9)	<i>əl=Ādam</i> SSG.M REF=PN
(5.81)	<i>esțəb-yon</i> baptize.tG.PFV- They baptized ac	<i>gāw napš</i> 3PL in same ccording to the same bap	e baptism-AUG
	<i>illi əštā</i> REL now with which we b	<i>qə=meṣṭəb-én</i> IND=baptize.tG.IPFV- aptize today. (I.10)	<i>gāww-a</i> -1PL in-3SG.F
(5.82)	səbonni	he was	baptized (lit. 'they baptized him')

Text IX also provides a textbook example of a verb in the middle voice:

(5.83)	etber-ø	minn=i	wusle
	break.tC.PFV-3SG.M	from=3SG.M	piece
	A piece broke off of it. (IX.12)		

185 Macuch, Handbook, 247-48.

Inflected Forms of the Strong Reflexive Stem Verbs

Due to the fact that only a limited number of examples of the other socalled "reflexive" stems are thus far attested in the Neo-Mandaic dialect of Khorramshahr, the forms given in this section will represent the tG-stem alone. Table 79 illustrates the personal suffixes found on the perfective, the imperative, and the imperfective.

Person	Perfective	Imperative	Imperfective
3 sg m	$eC_1.C_2eC_3$	-	$q \partial$ -meC ₁ .C ₂ eC ₃
3 sg f	$e.C_1 = C_2.C_3$ -at	-	$q \partial$ -mC ₁ eC ₂ .C ₃ -ā
2 sg	$eC_1.C_2eC_3-t$	$eC_1.C_2eC_3$	q p-mC ₁ eC ₂ .C ₃ -et
1 sg	$e.C_1 \Rightarrow C_2.C_3$ -it	-	q∂-meC ₁ .C ₂ eC ₃ -nā
3 pl	eC ₁ .C ₂ eC ₃ -yon	-	q p-mC ₁ eC ₂ .C ₃ -en
2 pl	eC ₁ .C ₂ eC ₃ -ton	eC ₁ .C ₂ eC ₃ -yon	q p-mC ₁ eC ₂ .C ₃ -etton
1 pl	eC ₁ .C ₂ eC ₃ - <i>ni</i>	-	q a-mC ₁ eC ₂ .C ₃ -enni

Table 79: Personal Suffixes on the Strong tG-Stem Verb

The perfective and imperative of the tG-stem verb are identical, and the participles are formed with the derivational morpheme m-. Note that the accent is usually retained on the initial syllable of the stem, with the result that the vowel of the second syllable is not lengthened as it is in the Q-stem. The tG-stem of strong roots does not distinguish between masculine and feminine forms of the singular imperative, like the other derived stems of strong roots.

As in the other stems of the verb, some of the personal suffixes have contextual forms, which are used most commonly before complements. The contextual form of the third plural perfective personal suffix and the plural imperative personal suffix is *-on*. The contextual form of the first singular form of the imperfective is distinguished from that of the third masculine singular only by accent, as are the contextual forms of the first plural and the third plural forms of the imperfective. The subjunctive is derived from the imperfective without the indicative marker.

As the *tG*-stem represents the middle voice, verbs in this stem do not appear with object suffixes.

The Verb 'To Baptize'

The only verb attested in the reflexive stems with a weak radical also happens to be one of a small number of verbs to preserve the derivational morpheme *t*- (another, observed by Macuch but not attested in the Neo-Mandaic dialect of Khorramshahr, is the verb *eštallam* ~ *eštallam* (*meštallam*) 'to be welcome'). Table 80 illustrates the personal suffixes found on the perfective, the imperative, the imperfective of the verb *eṣtəbā* ~ *eṣtəbi* (*meṣtəbi*) 'to baptize (intr.).'

Table 80: Personal Suffixes on the Verb "To Baptize"

Person	Perfective	Imperative	Imperfective
3 sg m	eșțəbā	-	qə-mestəbi
3 sg f	estəb-at	-	qə-mestəb-ā
2 sg	estəb-et	estəbi	qə-mestəb-et
1 sg	estəb-it	-	qə-mestəb-nā
3 pl	estəb-yon	-	qə-mestəb-en
2 pl	estəb-ton	estəb-yon	qə-mestəb-etton
1 pl	eșțəb-ni	-	qə-me <u>ş</u> tə <u>b</u> -enni

Naturally, several of the forms given in this paradigm also appear in contextual forms. The contextual form of the third plural perfective personal suffix and the plural imperative personal suffix is *-on*. The contextual form of the first singular form of the imperfective is distinguished from that of the third masculine singular only by accent, as are the contextual forms of the first plural and the third plural forms of the imperfective. The subjunctive is derived from the imperfective without the indicative marker.

The verb $estaba \sim estabi (mestabi)$ 'to baptize (intr.)' does not take object suffixes.

5.1.1.4 Verb Category Classes

Neo-Mandaic verbs can be classified as lexical verbs, phrasal verbs, copular verbs, auxiliary verbs, and modal verbs according to their meaning and syntactic function.

Lexical Verbs

Lexical verbs (or main verbs) compose the largest category of verbs in the Neo-Mandaic lexicon. Unlike phrasal verbs, the meaning of the verb is derived entirely from the verb itself, which constitutes the minimal predicate of the clause. The morphology of the lexical verb is described in sections §5.1.1.1 through §5.1.1.3. Further examples of the lexical verb are given in (5.84) and (5.85).

(5.84)	•		əl=kərāy=i REF=leg=3SG.M nim. (VII.2)	<i>tut=i</i> under=3SG.M
(5.85)	<i>hafte</i> week Joha ca	<i>horet-t-ā</i> other-F-AUG me the following v	<i>aṯā-ø</i> come.PFV-3SG.M week. (X.9)	<i>Joḥa</i> PN

Phrasal Verbs

Phrasal verbs, which compose a very large and productive class of verbs in Neo-Mandaic, consist of a verbal element and a non-verbal element, which form a single semantic and syntactic unit. The non-verbal element is most often a noun such as $\partial \underline{b} \bar{a} d\bar{a}$ 'deed,' e.g. the compound $\partial \underline{b} \bar{a} d$ $\partial \underline{b} a d \sim \partial \underline{b} o d$ ($\bar{a} \underline{b} e d$) 'to work or to do something,' or an adjective such as $h \partial y \bar{a} n \bar{a}$ 'alive,' e.g. the compound $h \partial y \bar{a} n \bar{a}$ tammā 'to survive,' although prepositions such as $q \bar{a} r$ 'at,' e.g. the compound $q \bar{a} r tamm \bar{a}$ 'to be born to s.o.,' are attested.

Although phrasal verbs are attested in Classical Mandaic, most Neo-Mandaic phrasal verbs are calqued upon Persian phrasal verbs, and most non-verbal elements are Persian or Arabic loan words. Even those phrasal verbs attested in Classical Mandaic have not escaped the influence of the Persian compounds. Compare the Classical Mandaic phrasal verb in (5.86) with the Neo-Mandaic phrasal verb in (5.87), which has probably been influenced by the Persian phrasal verb in (5.88).

(5.86)	zipa abid	he lies (lit. he lie-does)
(5.87)	zip qāheb	he lies (lit. he lie-gives)
(5.88)	ferib midehad	he lies (lit. he lie-gives)

5.1 The Structure of the Verb

In many phrasal verbs, the verbal element is a "light" verb, which serves only to indicate verbal inflections such as person, tense, mood, and aspect; the meaning of these compounds is primarily derived from the non-verbal element. The most common light verbs are $\frac{\partial b}{\partial a} \sim \frac{\partial b}{\partial b}$ ($\frac{\bar{a}bed}$) 'to do,' $\frac{\partial hab}{\partial a} \sim \frac{\partial hob}{\partial b}$ ($\frac{\bar{a}heb}{\partial a}$) 'to give,' $m \partial h\bar{a} \sim m \partial hi$ ($m\bar{a}hi$) 'to hit,' and tammā 'to become,' which closely correspond to the Persian light verbs kardan, dādan, zadan, and šodan. Other phrasal verbs are formed when a generic or non-specific noun becomes incorporated into meaning of the verb and ceases to be treated as its grammatical patient. The non-verbal element always precedes the verbal element.

Table 81: Some Examples of Phrasal Verbs

Main or Light Verb	Gloss	Phrasal Verb	Gloss
əbad ~ əbod (ābed)	to do	hitrā əbad ~ əbod (ābed)	to celebrate
əhab ~ əhob (āheb)	to give	subāḥ ṭāḇ əhaḇ ~ əhoḇ (āheḇ)	to say good morning
əmar ~ əmor (āmer)	to say	farwāh əmar ~ əmor (āmer)	to thank
atā ~ doti (āti)	to come	wayyeq aṯā ~ doṯi (āṯi)	to become distressed
bəğeş ~ bəğoş (bāğeş)	to stand	bandí bəğeş ~ bəğoş (bāğeş)	to confine oneself
dərā ~ dəri (dāri)	to take	ḥammām dərā ~ dəri (dāri)	to bathe
ezgā ~ ezgi (āzi)	to go	ohrā ezgā ~ ezgi (āzi)	to walk
həzā ~ həzi (hāzi)	to see	radpā həzā ~ həzi (hāzi)	to be persecuted
məhā ~ məhi (māhi)	to hit	safár məhā ~ məhi (māhi)	to start a journey
mətā ~ məti (māti)	to lift	id məṯā ~ məṯi (māṯi)	to give up, desist
ša <u>t</u> ~ ši <u>t</u> (šāye <u>t</u>)	to drink	<u> ḥašiš šaṯ ~ šiṯ (šāyeṯ)</u>	to smoke hashish
tammā	to become	tamām tammā	to be finished

Stress in Phrasal Verbs

Primary stress in the Phrasal verb (indicated by an acute accent) falls on the last syllable of the non-verbal element; the verbal element also receives secondary stress (indicated by a grave accent) on the expected syllable (see §2.4.4.3).

(5.89)	<i>at-on</i> come.PFV-3PL They came from	U	<i>əbàd-yo</i> do.PFV- old a wedding. (-3PL	<i>m=Aḥwāz</i> from=PN
(5.90)	5 5	U	.PFV-3SG.M	0	<i>q=ā̀mr-et</i> IND=speak.IPFV-2SG

	5.	The	Verb
--	----	-----	------

(5.91) *ḥammā́m dərā̀-ø tum nəp̃aq-ø* bath take.PFV-3SG.M then exit.PFV-3SG.M He took a bath and then came out. (VIII.7)

Negation in Phrasal Verbs

The negative morpheme *lá*- is prefixed to the verbal element of the phrasal verb to negate it. Because the negative morpheme is always stressed, the primary stress of the phrasal verb moves to the negative morpheme, leaving a secondary stress on the pre-verbal element.

(5.92)	<i>ā</i> this	<i>Čečāwā</i> PN	<i>zìp̄</i> deceit	<i>lá=q=āhe</i> NEG=INI	<u>¢</u> -ø D=give.IPF	V-3SG.M
	This Čečāwā doesn't lie. (IV.134)					
(5.93)	penitent	<i>tàmm-on</i> become.l gretted that th	PFV-3PL	how	prisoner	<i>lá=bīgeṣ-yon</i> NEG=stop.PFV-3PL

Generic Objects in Phrasal Verbs

Generic and non-specific objects do not follow the verb, but become incorporated into the meaning of the verb, and appear before it (see §3.5.3). As a result, they are not marked with the referential object marker.

(5.94)	<u> hašiš</u>	qə=ši <u>t</u> −en
	hashish	IND=drink.IPFV-3PL
	They smoked has	shish (lit. they hashish-drank). (VII.1)

(5.95) *šam*[•] *qə=masrik-en* candle IND=light.C.IPFV-3PL They lit candles (lit. they candle-lit). (IV.20)

Object Agreement in Compound Verbs

Definite nouns may be identified as the object of the verb by the referential object marker ∂l - (see §3.5.2) or form oblique complements (see §3.5.4) by means of prepositions:

(5.96) q=beyy-en komák abdel=l=ø əl=Rusya IND=want.IPFV-3PL help do.SBJV-3PL=OBJ=3SG.F REF=PN They wanted to help Russia. (III.2)

(5.97)	qābin	ə <u>b</u> ad-yon	qam	yeki	qəlāw	e <u>tt</u> =i
	marriage	woman=INDF				
	They married someone to a woman. (II.10)					

The object pronominal suffixes are generally attached to the nonverbal element of the compound verb rather than the verbal element.

(5.98)	əzgā-ø	rag`a=d=a	ābed-ø
	go.PFV-3SG.M	patch=of=3SG.F	do.SBJV-3SG.M
	He has gone to patch her up (lit. to do her patching). (IX.12)		

Many non-verbal elements can also take a possessive suffix. In the event that the non-verbal element already has a possessive suffix attached, the object pronominal suffix appears on the verbal element.

(5.99)	genz	jawāb=d=ø	ah <u>b</u> -it=a
	much	answer=of=1SG	give.PFV-1SG=3SG.F
	I answered her m	any times (lit. I gave her m	any of my answers). (IV.20)
(5.100)	rag`a=d=ak	ābed-ø=l=u	geš=d=u
	patch=of=2SG.M	do.IPFV-3SG.M=OB	J=3PL all=of=3PL
	He will fix all of	those things for you. (VI.6)

Copular Verbs and Predicate Nominals

In existential declarative or interrogative statements, the subject of the statement may simply be juxtaposed with its predicate, or linked to it by means of an existential particle with long and short forms. In the latter construction, subject agreement is indicated by means of suffixes on the existential particle. This existential particle does not inflect for tense, aspect, or mood. In other tenses, the verb $h \partial w \bar{a} \sim h \partial w i (h \bar{a} w i)$ 'to be' is employed as a copular verb to link the subject and the predicate.

Long and Short Forms of the Copula

In addition to indicating that a question word anticipates the object of a verb, the copula is also used to join the subject of an existential declarative or interrogative statement with a predicate nominal, adjective, or locative. For this purpose, Neo-Mandaic employs long (independent) and short (enclitic) forms of an existential particle.

Table 82: The Short Copula

Person 3	Singular <i>šəbir-ve</i>	Gloss he is good	Plural <i>šəbir-non</i>	Gloss they are good
2	šəbir-yāt	you are good	-	they are good
1	šəbir-nā	I am good	-	

Note that the copula does not distinguish between masculine and feminine gender, and that there are no short forms for the second and first persons plural.¹⁸⁶ The short copula is also employed as an auxiliary verb in the analytic passive construction, with the notable exception of the plural form, which appears only on predicate nominals, e.g. (5.101); the analytic passive plural employs the long copula instead, e.g. (5.102).

(5.101)	<i>šəbir=non</i> good=3PL		They are well.
(5.102)	<i>yəhem</i> sit.PASS	<i>e<u>k</u>t=u</i> COP=3PL	They are seated.

The enclitic pronominal copula is most commonly used with question words, e.g. (5.103), prediate adjectives, e.g. (5.104), and predicate locatives, e.g. (5.105).

(5.103)	<i>man=ye?</i> who=3SG.	.М		Who is he?
(5.104)	<i>šəbir=nā</i> good=1SG	ŕ		I am well.
(5.105)	they	<i>barra</i> outside outside the city	<i>əm=belā<u>t</u>=non</i> from=city=3PL y. (III.11)	

The independent and enclitic forms are generally interchangeable. The enclitic forms are negated with the prefix lu-, which appears on the noun modified by the copula. The existential particle, which takes the enclitic personal pronouns introduced in §4.1.2, is negated with the negative morpheme la- not:

186 The long form of the copula is employed in their place; see Table 83.

Copula	Gloss	<i>lá-</i> + Copula	Gloss
ekt-i	he is	l-ekt-i	he is not
e <u>k</u> t-a	she is	l-ekt-a	she is not
e <u>k</u> t-a <u>k</u>	you (m.) are	l-e <u>k</u> t-a <u>k</u>	you (m.) are not
e <u>k</u> t-e <u>k</u>	you (f.) are	l-e <u>k</u> t-e <u>k</u>	you (f.) are not
e <u>k</u> t-e	I am	l-e <u>k</u> t-e	I am not
e <u>k</u> t-u	they are	l-e <u>k</u> t-u	they are not
e <u>k</u> t-ə <u>k</u> on	you are	l-ekt-əkon	you are not
e <u>k</u> t-an	we are	l-e <u>k</u> t-an	we are not

Table 83: The Long Copula (and its Negation)

The existential particle always follows the predicate or question word, e.g. (5.106). Predicate nominals, e.g. (5.107), and negative predicates, e.g. (5.108), are indicated by the existential particle or the copular verb $h \partial w \bar{a} \sim h \partial w i (h \bar{a} w i)$ exclusively.¹⁸⁷

(5.106)	<i>man</i> who	<i>e<u>k</u>t=a<u>k</u> COP=2SG.M</i>		Who are you? (IV.51)
(5.107)	1SG	<i>qanāy-ā</i> smith-AUG ilver smith.	<i>d=kasp-ā</i> of=silver-AUG	<i>ekt=e</i> COP=1SG
(5.108)	-	<i>l=ekt=e</i> NEG= COP=1	SG	I am not well.

When followed by the proclitic preposition l- 'to, for,' the copula is used for possessive clauses. The long copula is more commonly used for possessive clauses, e.g. (5.109), although the short copula may also be employed, e.g. (5.110).

(5.109)	eh=l=i		rāz-ā	d=Šāmeš
	COP=to=3		secret-AUG	of=PN
	He has the	secret of	Šāmeš. (IV.135	5)
(5.110)	ahni	əl=man=	non	
	these	to=who=	3PL	Whose are these? (III.11)

187 Compare the distribution of the Spanish copulas *ser* and *estar*, e.g. *Ofelia está enferma* 'Ofelia is sick,' *Ofelia es enferma* 'Ofelia is a sick woman,' and *Ofelia es profesora* 'Ofelia is a teacher,' but never ***Ofelia está profesora* (examples from Payne, *Describing Morphosyntax*, 120). Note the unexpected allomorph of the long copula before the proclitic *I*-. The discrepancy between the phonetic realizations of the two allomorphs, *ekt*- and *eh*-, is attributable to regular sound changes; see §2.4.4.2 and §2.5.4.2. The forms of the long copula before proclitic *I*- are given in Table 84.

Copula + <i>l</i> -	Gloss	lá- + Copula + l-	Gloss
eh-l-i	he has	l-eh-l-i	he doesn't have
eh-l-a	she has	l-eh-l-a	she doesn't have
eh-l-a <u>k</u>	you (m.) have	l-eh-l-a <u>k</u>	you (m.) don't have
eh-l-e <u>k</u>	you (f.) have	l-eh-l-e <u>k</u>	you (f.) don't have
eh-l-e	I have	l-eh-l-e	I don't have
eh-l-u	they have	l-eh-l-u	they don't have
eh-l-əkon	you (pl.) have	l-eh-l-ə <u>k</u> on	you (pl.) don't have
eh-l-an	we have	l-eh-l-an	we don't have

Table 84: The Long	Copula witl	n Proclitic <i>l</i> -	(and its Negation)
--------------------	-------------	------------------------	--------------------

This construction always stands at the head of possessive clauses, e.g. (5.111), in contrast with other declarative clauses, in which the predicate nominal always precedes the copula.

(5.111)	eh=l=e	pas-i	
	COP=to=1SG	pass-INDF	I have a pass. (III.4)

The perfective of the copula is formed with the verb $h \partial w \bar{a} \sim h \partial w i$ ($h \bar{a} w i$) 'to be,' the forms of which are given in Table 85.

Table 85: The Verb "To Be" with Proclitic *l*- (and its Negation)

Copula + <i>l</i> -	Gloss	<i>lá-</i> + Copula + <i>l</i> -	Gloss
həwā-l-i	he had	l-ehwā-l-i	he didn't have
həwā-l-a	she had	l-ehwā-l-a	she didn't have
həwā-l-a <u>k</u>	you (m.) had	l-ehwā-l-a <u>k</u>	you (m.) didn't have
həwā-l-e <u>k</u>	you (f.) had	l-ehwā-l-eķ	you (f.) didn't have
həwā-l-e	I had	l-ehwā-l-e	I didn't have
həwā-l-u	they have	l-ehwā-l-u	they didn't have
həwā-l-kon	you (pl.) have	l-ehwā-l-kon	you (pl.) didn't have
həwā-l-an	we had	l-ehwā-l-an	we didn't have

Auxiliary Verbs

There are two verbs in Neo-Mandaic that are regularly employed as auxiliaries. The first, the copula, is used to form the passive voice, as in English. The second, $qam \sim qom (q\bar{a}yem)$ 'to stand,' is employed as an inchoative auxiliary verb.

The Copula and the Analytic Passive

Table 86: The Copula in the Analytic passive

Person 3 sg m	Short Copula <i>gəțel-ye</i>	Long Copula <i>gəțel e<u>k</u>t-i</i>	Perfective gəțel həwā	Imperfective gəțel qə-hāwi
3 sg f	gəțil-i	gəțil ekt-a	gəțil həw-at	gəțil qə-hāwyā
2 sg m	gəțel-yāt	gəțel ekt-ak	gəțel həw-et	gətel qə-hāwet
2 sg f	gəțel-yāt	gəțil ekt-ek	gəțil həw-et	gəțil qə-hāwet
1 sg	gəțel-nān	gəțel e <u>k</u> t-e	gətel həw-it	gətel qə-hawi-nā
3 pl	-	gəțilen e <u>k</u> t-u	gəțilen həw-on	gəțilen qə-hāw-en
2 pl	-	gəțilen ekt-əkon	gəțilen həwi-ton	gəțilen qə-haw-etton
1 pl	-	gəțilen e <u>k</u> t-an	gəțilen həwi-ni	gəțilen qə-haw-enni

Note that there are no plural forms of the short copula. As the existential particle is used for all existential and simple present statements, e.g. (5.112), the imperfective is reserved for statements in the progressive present or the future tense, e.g. (5.113). Likewise, the subjunctive of $h \partial w \bar{a} \sim h \partial w i (h \bar{a} w i)$ 'to be,' which is regularly derived from the imperfective after the removal of the indicative marker and pretonic reduction, is used to indicate suggestions, wishes, and hopes, e.g. (5.114).

(5.112)	<i>yəhim-en</i> sit.PASS-PL	<i>e<u>k</u>t=an</i> COP=1PL	We are seated.
(5.113)	<i>yəhim-en</i> sit.PASS-PL	<i>qə=haw-enni</i> IND=be.IPFV-1PL	We are being / will be seated.

(5.114) <i>yəhim-en</i>	haw-én		
sit.PASS-PL	be.SBJV-1PL	Let us be seated.	

Even though the analytic passive is well represented in Classical Mandaic,¹⁸⁸ the impersonal passive construction (§5.4.3.2) is much more common in Neo-Mandaic.

The Inchoative Auxiliary Verb

When followed by a verb in the imperfective, $qam \sim qom (q\bar{a}yem)$ 'to stand' serves to indicate the beginning of an action, state, or event.¹⁸⁹ In this capacity, it appears only in the perfective, and inflects to reflect the subject. The subject may precede the auxiliary verb, e.g. (5.116), or intervene before the auxiliary verb and main or compound verb that it modifies, e.g. (5.117) and (5.118).

(5.115)	<i>qam-yon</i> stand.PFV-3PL They started talk	IND	=talk.D.IPF		e	t=u cof=3PL
(5.116)	<i>darwiš-ā́</i> dervish-PL The dervishes be	stan	n <i>-yon</i> d.PFV-3PL chanting. (IV.		<i>qār-en</i> read.IPI	FV-3PL
(5.117)	<i>qam-ø</i> stand.PFV-3SG.I Čečāwā began di		PN	<i>qə=hāp</i> o IND=di	<i>er-ø</i> g.IPFV-3	SG.M
(5.118)	<i>qam-ø</i> stand.PFV-3SG.I The ruler proceed	М		banquet	-AUG	<i>q=ābed-ø</i> IND=do.IPFV-3SG.M
In addition to serving as an auxiliary verb, the verb $qam \sim qom$ $(q\bar{a}yem)$ 'to stand' is also an independent verb in its own right.						

(5.119)	qam-yon	barābar=d=u	
	stand.PFV-3PL	opposite=of=3PL	They stood before them. (IV.76)

188 Macuch, Handbook, 433–34.

189 It also serves in this role in the Neo-Aramaic dialect of Aradhin; see George Krotkoff, *A Neo-Aramaic Dialect of Kurdistan* (New Haven, CN: American Oriental Society, 1982), 56. Crosslinguistically, this is not uncommon; cf. Italian *il sole sta per tramontar* "the sun is about to set."

Modal Verbs

There are two modal verbs in Neo-Mandaic: $ab\bar{a} (b\bar{a}yi)$ 'to want' and $\bar{s}abaq \sim \bar{s}aboq (\bar{s}\bar{a}beq)$ 'to permit.' These regularly take a complement verb in the subjunctive to express a notion of modality. All three are independent verbs in their own right. In addition, the pseudoverb *qomb*- 'to be able' and the modal particle $b\bar{a}yad$ 'must' are used in modal constructions. Macuch claims that the Mandaeans have borrowed the Persian particle $\bar{s}\bar{a}yad$ 'might' in addition to $b\bar{a}yad$, but aside from one isolated example of $b\bar{a}yad$ in his Handbook,¹⁹⁰ neither of these two particles appears in any of the texts he has published.¹⁹¹ Additionally, there is a pseudoverb in Text IV, *lābod* 'must,' which takes a complement subjunctive verb.

The Modal Verb 'To Want'

The first of the modal verbs, $ab\bar{a} (b\bar{a}yi)$ 'to want,' is used to express optative modality (§5.3.2.2), and is illustrated with a complementary subjunctive verb, $\partial mar \sim \partial mor(\bar{a}mer)$ 'to say,' in Table 87.

Table 87: The Modal Verb "To Want" with the Verb "To Say"

Person	Form	Gloss
3 sg m	qə-bāyi āmer	he wants to say
3 sg f	qə-beyy-ā āmr-ā	she wants to say
2 sg	qə-bā-t āmr-et	you want to say
1 sg	q-abi-n əmer	I want to say
3 pl	qə-beyy-en āmr-en	they want to say
2 pl	q-ab-etton əmer-ton	you want to say
1 pl	q-ab-én amr-en	we want to say

Examples of the modal verb *abā* (*bāyi*) 'to want' in the texts include:

(5.120)	q=abi-n	əmer-ø	farwāh= <u>k</u> on	geš
	IND=want.IPFV-1SG	say.SBJV-1SG	thanks=2PL	all
	I want to thank all of you. (I.3)			

- 190 Macuch, Handbook, 486: bāyæd bālax botqāti 'you have to be smart.'
- 191 Both *bāyad* and *šāyad* are conspicuously absent from his *Neumandäische Chrestomathie* and *Neumandäische Texte im Dialekt von Ahwāz*.

(5.121) *qə=beyy-en min=ṣubāḥ* IND=want.IPFV-3PL from=morning The next day, they wanted

ohr-ā	all-en	əlāww=i
way-AUG	go.SBJV-3PL	to-3SG=M
to walk on it.	(IV.6)	

Abā (bāyi) 'to want,' can also appear in independent usage.

(5.122) *mu qə=bā-t* what IND=want.IPFV-2SG What do you want? (IX.8)

The Modal Verb 'To Permit'

The modal verb $š \partial p a q \sim \tilde{s} \partial p o q$ ($\tilde{s} \partial p e q p$) 'to permit' is conjugated in Table 88 with the third masculine singular subjunctive of $haddet \sim haddet$ ($\partial m haddet$) 'to speak.'

Table 88: The Modal Verb 'To Permit' with the Verb 'To Speak'

Person	Form	Gloss
3 sg m	qə-šabeq-l-i əmhadde <u>t</u>	he lets him speak
3 sg f	qə-šabq-ā-l-i əmḥaddeṯ	she lets him speak
2 sg	qə-šabq-āt-i əmḥaddeṯ	you let him speak
1 sg	qə-šabeq-n-i əmḥaddeṯ	I let him speak
3 pl	qə-šabq-el-l-i əmḥaddeṯ	they let him speak
2 pl	qə-šābeq-tonn-i əmḥaddeṯ	you let him speak
1 pl	qə-šabeq-nānn-i əmḥaddeṯ	we let him speak

Note that the subject of the subjunctive verb, which is the object of the modal verb, is anticipated by an object pronominal suffix on the modal verb. Note also that the third masculine singular form of the modal verb complemented by a subjunctive verb in the same number and gender, e.g. (5.123), would not be construed as a reflexive. The reflexive requires the reflexive pronoun *nabš*-, e.g. (5.124).

(5.123)	qə=šābeq-ø=l=i	əmḥaddeṯ-ø
	IND=let.IPFV-3SG.M=OBJ=3SG.M	say.SBJV-3SG.M
	He lets him say.	

(5.124) $q = \bar{s}\bar{a}\bar{b}eq - \phi = l = \phi$ IND=leave.IPFV-3SG.M=OBJ=3SG.M REF=self=3SG.M say.SBJV-3SG.M He lets himself say.

Examples of the modal verb $\tilde{s} \partial p a q \sim \tilde{s} \partial p o q$ ($\tilde{s} \partial p a p q$) in the texts include:

- (5.125) *šobq-ø əl=dešt-ā bāgṣ-ā* leave.IMP.M-SG REF=ground-AUG stop.IPFV-3SG.F Let the ground stop [shaking]. (IV.109)
- (5.126) *lá=šabq-onn=an all-enni ekkak* NEG=leave.PFV=3PL=1PL go.SBJV-1PL there They wouldn't let us go there. (III.13)

This verb also appears independently:

(5.127)	zipān-ā	šəbaq-ø	и	pərad-ø
	liar-AUG	leave.PFV-3SG.M	and	flee.PFV-3SG.M
	The liar left and a	can off. (IX.13)		

The Pseudoverb 'To Be Able'

The pseudoverb *qomb*- 'to be able' is used to express possibility. The present is illustrated here with the verb *haddet* ~ *haddet* (*amhaddet*) 'to speak.'

Table 89: The Modal Verb 'To Be Able' w/ the Verb 'To Speak'

Person	Form	Gloss
3 sg m	qomb-i əmḥadde <u>t</u>	he can speak
3 sg f	qomb-a əm <u>ḥattā</u>	she can speak
2 sg	qomb-ak əmḥattet	you can speak
2 sg	qomb-e <u>k</u> əmḥa <u>tt</u> et	you can speak
1 sg	qomb-e əmḥade <u>t</u>	I can speak
3 pl	qomb-u əmḥaṯṯen	they can speak
2 pl	qomb-ə <u>k</u> on əm <u>hatt</u> etton	you can speak
1 pl	qomb-an əmḥa <u>tt</u> én	we can speak

Note that the subject of *qomb*- is indicated by a possessive suffix on the pseudoverb, exactly like the pseudoverb $y\bar{a}d$ 'to know' above (see §5.1.1.2). Likewise, the other tenses and aspects are indicated by the copula $h \partial w\bar{a} \sim h \partial wi (h \bar{a} wi)$ 'to be.' In the example below, taken from

the dialect of Salem Choheili, this pseudoverb takes the form *yomb*rather than *qomb*- as in the dialect of Nasser Sobbi and Nasser Saburi (for which, see section §1.7 on the modern dialects of Neo-Mandaic).

(5.128)	<i>yomb=u</i> be.able=3PL (So that) they ma	<i>hāvi-ø</i> be.SBJV-3SG.M y be able to read and	<i>šibír</i> good	
	<i>qār-en</i> read.SBJV-3PL write well in Mai	<i>u=kād<u>b</u>-en</i> and=write.SBJV-3PL ndaic. ¹⁹²	<i>gū</i> in	<i>mand(y)öyi̇́</i> Mandaic

The Modal Particles bayad and labod

The final modal is the particle $b\bar{a}yad$ 'must,' which has been borrowed from Persian. Unlike the pseudoverbs $y\bar{a}d$ 'to know' and *qomb*- 'to be able,' $b\bar{a}yad$ does not inflect for tense, mood, aspect, or person. The paradigm of this particle with the verb <u>haddet</u> ~ <u>haddet</u> (*amhaddet*) 'to speak' can be found below.

Person	Form	Gloss
3 sg m	bāyad əmḥaddeṯ	he must speak
3 sg f	bāyad əmḥaṯṯā	she must speak
2 sg	bāyad əmḥaṯṯet	you must speak
1 sg	bāyad əmḥadeṯ	I must speak
3 pl	bāyad əmḥaṯṯen	they must speak
2 pl	bāyad əmḥaṯṯetton	you must speak
1 pl	bāyad əmḥa <u>tt</u> én	we must speak

Another pseudoverb, *lābod* 'must,' appears in the legend of the Bridge of Shushtar (Text IV) in the place of the modal verb *bāyad*. In this text, it is complemented by a verb in the subjunctive and is invariable like *bāyad*. The form, derived from Arabic *lā budda* 'inevitably,' lit. 'no escape,' expresses a wide range of deontic modalities (§5.3.2.1). Examples of *lābod* in the texts include (5.129).

(5.129)	lābod	malp-āt=e	
	certainly	teach.SBJV-2SG=1SG	
	You must teach me. (IV.86)		

Independently, *lābod* means 'certainly' or 'inevitably,' as in Arabic and Persian.

(5.130)	lābod	kef	ābd-en	ənāš-ā	
	certainly	pleasure	do.SBJV-3PL	people-AUG	
	The people will certainly celebrate! (IV.72)				

5.2 Tense and Aspect

Tense and aspect are closely related in Neo-Mandaic, although aspect is the more salient of the two within the verbal system. At first glance, the Neo-Mandaic verbal system appears to be based on the distinction of tenses. The perfective is the general form used in simple past narratives. In opposition to the imperfective (which Macuch indeed calls the "participial present-future tense"), it appears to correspond to the past tense. Nonetheless, aspect is as basic to the Neo-Mandaic verbal system as tense; the inflected forms derived from the participle are imperfective, and as such indicate habitual actions, progressive or inchoative actions, and actions in the future from a past or present perspective. The perfective forms are not only preterite but also resultative-stative, which is most apparent from the verbs relating to a change of state, e.g. $met \sim mot(mayet)$ 'to die.' For these reasons, it is more accurate to discuss the Neo-Mandaic verbal system in aspectual terms rather than simply treating it as a system of tenses.

5.2.1 The Perfective

The perfective has simple past (§5.2.1.1), past perfect (§5.2.1.2), and resultative-stative (§5.2.1.3) meanings.

5.2.1.1 The Simple Past

The most common use of the perfective is to render the simple past.

(5.131)	Mār=e	abd-ø	əl=kol	alm-ān-ā	
	lord=1SG	do.PFV-3SG.M	REF=all	world-PL-AUG	
	God (lit. my Lord) created all the worlds. (I.11)				

(5.132) $b\bar{a}b$ <u>hammām</u> əhab- ϕ =l=i pərā father hammam give.PFV-3SG.M=OBJ=3SG.M money The owner of the bathhouse gave him some money. (VIII.5)

5.2.1.2 The Past Perfect

The perfective may also be used to indicate an event that has occurred before another action in the past.

(5.133)	həz-it-i		jisər	șun	i′a=ye
	see.PFV-1S	G-3SG.M	bridge	goo	od=3SG.M
	I saw that th	ne bridge w	vas repaired (lit. I saw "it i	s a good bridge"),
	<i>u</i> and and another		<i>i</i> ge-INDF 1 been built. (<i>tum</i> again III.21)	<i>ə<u>b</u>ad-yon</i> do.PFV-3PL
(5.134)		hree	<i>ruz</i> day ume back,	<i>kammar-ø</i> return.PFV-3	3SG.M
	<i>bədaq-ø</i> put.PFV-3S having put a	G.M	<i>kars=i</i> belly=3SG.M inside it. (X.		<i>honin-i</i> n small-INDF

5.2.1.3 The Resultative-Stative

With inherently stative verbs or stative compounds (usually formed with *tammā* 'to become'), the perfective indicates a state or situation resulting from a completed event. When verbs of knowledge and opinion (such as *bašqer* ~ *bašqer* (*ambašqer*) 'to know,' *dakar* ~ *dakor* (*dāker*) 'to remember,' *fekr abad* ~ *abod* (*ābed*) 'to think,' *qabul abad* ~ *abod* (*ābed*) 'to believe') appear in the perfective, they have a resultative-stative meaning.

- (5.135) *mekt-at* əštā die.PFV-3SG.F now Now she is dead. (II.34)
- (5.136) kərāy=e wel tamm-at leg=1SG independent become.PFV-3SG.F [He said]: "My leg has come off!" (VII.5)

5.2 Tense and Aspect

(5.137)	we	<i>Manday-ān-ā</i> mandaean-PL-AU andaeans were the f	JG :	<i>awwal</i> first eople	<i>barr</i> pers		<i>həw-</i> be.P	<i>-ini</i> FV-1PL
	<i>ke</i> REL	<i>bašqar-ni</i> know.PFV-1PL		<i>lār=e</i> =lord=1SC		<i>ehd-i</i> one-IN		<i>e<u>k</u>t=i</i> COP=3SG.M

who recognized that God (lit. my Lord) is a single being. (I.11)

5.2.2 The Imperfective

The imperfective is formed from the active participle of the verb. It can represent the simple present (§5.2.2.1), habitual present (§5.2.2.2), habitual past (§5.2.2.3), narrative present (§5.2.2.4), and future tenses (§5.2.2.5).

5.2.2.1 The Simple Present

The simple present is the most common tense represented by the imperfective.

(5.138)	<i>q=abi-n</i> IND=want.IPFV-1SG I want to thank all of ye	•	<i>farwāh=kon</i> SG thanks=2PL	<i>geš</i> all
(5.139)	<i>qə=haz-én</i> IND=see.IPFV-1PL We see houses in the di	<i>biet-wāt</i> house-PL.F stance, (III.10)	<i>baʻid</i> distant	

5.2.2.2 The Progressive Present

The progressive aspect implies an action or condition which is currently in process (as opposed to the resultative-stative aspect, which indicates that the event which resulted in the current state has already been completed).

(5.140)	qə=mšaddar-t=ø	əl=ā	konfərens	
	IND=organize.D.IPFV-2SG=3SG.M	REF=this	conference	
	You are organizing this conference. (I.	organizing this conference. (I.6)		

(5.141) *qamu aziyāt q=ābd-et* why harm IND=do.IPFV-2SG Why are you causing such trouble? (IV.57)

5.2.2.3 The Habitual Present

The habitual present denotes an action or situation that occurs regularly or continuously.

(5.142) *kol* qə=ban-ann=i šett-ā IND=build.IPFV-1PL=3SG.M everv vear-AUG Every year we build it, *qə=hāreb-ø* и IND=be.destroved.IPFV-3SG.M and and it is destroyed! (IV.34) (5.143) *a=all-én* ərabt-ā kol ekkak IND=go.IPFV-1PL Friday-AUG there everv We go there every Friday.

5.2.2.4 The Habitual Past

The imperfective is also used for actions that occurred regularly or continuously in the past; these are often introduced by another past tense verb to contextualize them. Verbs of knowledge and opinion (such as *bašqer ~ bašqer (əmbašqer)* 'to know,' $d \partial \underline{k} ar \sim d \partial \underline{k} or (d \overline{a} \underline{k} er)$ 'to remember,' *fekr əbad ~ əbod (ābed)* 'to think,' *qabul əbad ~ əbod (ābed)* 'to believe') generally appear in the imperfective with a habitual present or habitual past tense meaning; when they do appear in the perfective, they have a resultative-stative meaning (§5.2.1.3).

(5.144)	<i>q=all-én</i> IND=go.IPFV-11 We used to go to		Dezful PN
	7-11 həwā-ø PN be.PFV (they had a 7-11)	-3SG.M=to=3PL	<i>ekka<u>k</u></i> there
(5.145)	<i>mu q=əmba</i> what IND=kr What did you kn	now.IPFV-2SG	<i>a<u>k</u>=waxt</i> that=time
(5.146)	<i>gobr-ā</i> men-AUG The men there di	<i>ekkak</i> there dn't know	<i>la=q=əmbašqer-en</i> NEG=IND=know.IPFV-3PL

əmḥaṯt-øMandāyíspeak.SBJV-3PLMandaichow to speak Mandaic. (II.25)

5.2.2.5 The Narrative Present

The Narrative or Historical Present tense is extremely common in the text collection. Generally, clauses in the Narrative Present are first introduced by a perfective verb, which places the situation or action described in a past tense context.

(5.147)	<i>at-it</i> come.PFV-1SG I came up to her -	<i>qamāy=a</i> before=3SG	.F	
	<i>q=abi-n</i> IND=want.IPFV-1S I want only to speak	J	<i>məḥadeṯ-ø</i> speak.SBJV-1SG 7)	<i>ork=a</i> with=3SG.F
(5.148)	<i>əzgā-ø</i> go.PFV-3SG.M He went up a hill,	<i>qəlāw</i> to	<i>tel</i> hill	
	<i>w=ebr-i</i> and=son-INDF and seizes one of th	<i>əl=gab॒r-ā</i> to=man-AUG e man's sons. ((V-3SG.M=3SG.M

5.2.2.6 The Future

The imperfective is also used to describe events in the near and distant future.

(5.149)	from	<i>hənā</i> here ays from	<i>tum</i> until now,	<i>ārbin</i> 40	<i>yum-ā</i> day-AUG	
	IND=co	<i>ti-n=<u>k</u>on</i> ome.C.IP ring you.				
(5.150)		= <i>l=i</i>	ODIA	60 M	perš	<i>q=ati-nā</i>

(3.130) amar-o-1-1 pers q-an-ma say.PFV-3SG.M=OBJ=3SG.M tomorrow IND=come.IPFV-1SG He said to him, "I will come tomorrow." (V.17)

5.3 Moods and Modalities

Speakers of Neo-Mandaic employ the indicative, the subjunctive, and the imperative moods to indicate their position on a given situation, including their purpose in stating the situation and their belief in its reality and likelihood.

5.3.1 The Indicative Mood

The indicative mood is used to make assertions or declarations about situations which the speaker holds to have happened (or, conversely, which have not happened), or positions which he maintains to be true. It is also the mood used for questions and other interogative statements. The perfective, by its very nature, refers to situations that the speaker holds to have happened or not to have happened, and thus pertains to the indicative, except in explicitly counterfactual statements (see §5.3.4.6).

The imperfective, on the other hand, is used to describe situations which have yet to happen, or about which there exists some uncertainty or doubt; when accompanied by the indicative marker, it is used to express assertions and questions, just like the perfective. When it is not explicitly marked as indicative, it indicates subjunctive mood (see section §5.3.2) and may correspond to any one of a wide variety of modalities.

5.3.1.1 Declarative Statements

All assertions and declarative statements are, by their nature, made in the indicative mood.

(5.151)	an	a <u>t</u> -it	ekkā	gāw	<u>ḥammām</u>
	Ι	come.PFV-1SG	thus	in	hammam
	I came	II.18)			

(5.152) *obr-ā* genz $q = rahem - \phi = l = \phi$ $\partial l = hadid$ mouse-AUG very IND=love.IPFV-3SG.M=OBJ=3SG.M REF=iron Mice really love iron. (V.12)

5.3.1.2 Interrogative Statements

Interrogative statements are also made in the indicative mood.

5.3 Moods and Modalities

- (5.153) *əmar-ø=l=u mu q=amr-etton əlāw tā pol* say.PFV-3SG.M=OBJ=3PL what IND=say.IPFV-2PL for that bridge He said to them, "What do you suggest for that bridge?" (IV.33)
- (5.154) $man qabúl q=\bar{a}\underline{b}ed-\phi$ who acceptance IND=do.IPFV-3SG.M "Who believes (=would believe) $qaz \underline{g} \overline{a} n mu \underline{j} ur q \overline{\phi} = m \overline{a} y \underline{e} \underline{f} - \phi$ cauldron what.way IND=die.IPFV-3SG.M that a cooking pot dies (=would die)?" (X.16)

5.3.2 The Subjunctive Mood

Neo-Mandaic forms the subjunctive mood from the imperfective. In the *G*-stem of the verb (with the exception of the II-weak and geminate root classes), the subjunctive is also marked by the pretonic reduction of the vowel in the initial syllable to /a/; in all other root classes and stems, the subjunctive is identical to the indicative without the indicative morpheme *q*-. The subjunctive mood is used to express three distinct modalities: deontic (§5.3.2.1), optative (§5.3.2.2), and potential (§5.3.2.3).

5.3.2.1 The Deontic Modality

The deontic modality indicates the subject's duty or obligation to perform an irrealis act, which is expressed by a verb in the subjunctive mood. Deontic modalities may be either volitive, e.g. (5.155) and (5.156), commissive, e.g. (5.157), or directive, e.g. (5.158). The latter two modalities are introduced by the modal particles $b\bar{a}yad$ 'must' (or $l\bar{a}bod$ 'inevitably' in the legend from Iraq).

(5.155)	lā	hāreb-ø	pol	Šuštar
	NEG	be.destroyed.SBJV-3SG.M	bridge	Šuštar
	May the	e bridge of Shushtar not be rui	ined! (IV.24)	
(5.156)	ani	nešemt-i la	a=gāțl-én	
	1PL	soul-INDF N	EG=kill.SBJ	V-1PL
	Let's no	ot kill any living being. (IV.96	5)	

5. The Verb

- (5.157) *lābod kef ābd-en ənāš-ā* certainly pleasure do.SBJV-3PL people-AUG Surely the people will celebrate! (IV.72)
- (5.158) *lābod malp-āt=e* certainly teach.SBJV-2SG=1SG You must teach me. (IV.86)

The imperative (§5.3.3) also belongs to the class of directive modalities, but has a separate form.

5.3.2.2 The Optative Modality

The subjunctive is used after verbs of desire to indicate the object of a wish.

(5.159)	ani	q-ab-én	safar	mah-én
	1PL	IND=want.IPFV-1PL	trip	hit.SBJV-1PL
	We wan	t to travel. (IV.132)		

(5.160) q=*mbarak*- \emptyset haw-én həyānā tum āxir ālmā IND=pray.tD.IPFV-1SG be.SUBJ-1PL alive until last world I pray that we will survive until the end of time. (I.14)

5.3.2.3 The Potential Modality

The subjunctive may also be used for potential situations or the possible results of an action.

(5.161) *gobr-ā ekkak la=q=əmbašqer-en* men-AUG there NEG=IND=know.IPFV-3PL The men there didn't know

> *əmḥaṯt-ø* Mandāyí speak.SBJV.3PL Mandaic how to speak Mandaic. (II.25)

5.3.3 The Imperative

The imperative, in contrast to the indicative, does not assert that a situation has occurred, but orders that it come about. The imperative

and the prohibitive belong syntactically to the broader class of directive modalities. All of these modalities, including the prohibitive, are expressed using the subjunctive, with the single exception of the imperative.

5.3.3.1 Commands

The imperative is the mood in which commands are issued.

(5.163)	<i>yəmi-ø</i> swear.IMP.M-SG	1	<i>qə=mwedi-n=<u>k</u>on</i> IND=take.IPFV-1SG-2PL	<i>Əmšuni Kušțā</i> PN
	Swear to me, "I will	l take you t	o Əmšuni Kušțā." (IV.120)	
(5, 164)	id moti-a		minn=e	

(5.164) *id* $m = ti = \emptyset$ minn = ehand lift.IMP.M-SG from=1SG Release me (lit. lift your hand from me)! (IV.108)

5.3.3.2 Negative Commands and Prohibitions

The imperative cannot be directly negated in Neo-Mandaic, as in other Neo-Aramaic dialects and in other Semitic languages more generally.¹⁹³ As in these other languages, the subjunctive is used in the place of the imperative for negative commands and prohibitions.

(5.165)	<i>l=all-etton</i>	ġer	welāt
	NEG=go.SBJV-2PL	outside	city
	Do not go outside the ci		

5.3.4 Moods and Adverbial Clauses

Mandaic uses verbs in the subjunctive mood, and to a lesser extent the indicative mood, to indicate adverbial clauses. These adverbial clauses contribute information about the time, location, manner, purpose, and reason of the action or situation in the main phrase, as well as indicating situations or actions that are simultaneous to the main clause. Simple, hypothetical, and counterfactual conditional clauses also constitute a category of adverbial clauses.¹⁹⁴

¹⁹³ Geoffrey Khan, A Grammar of Neo-Aramaic: The Dialect of the Jews of Arbel (Leiden: Brill, 1999), 282.

¹⁹⁴ Payne, Describing Morphosyntax, 316–20.

5.3.4.1 Time Clauses

Clauses that indicate the time when a particular action or situation will occur take verbs in the subjunctive rather than the indicative.

- (5.166) *qəmāy malp̄-āt=i ana gaṭel-nā=kon geš* before teach.SBJV-2SG=3SG.M 1SG kill.SBJV-1SG=2PL all Before you teach him, I will kill you all! (IV.93)
- (5.167) an q=əmbašqir-it qamāy ābed-ø=l=a
 1SG IND=know.IPFV-1SG before do.SBJV-3SG.M=OBJ=3SG.F
 I knew before he did it.

5.3.4.2 Circumstantial Clauses

Verbs found in adverbial clauses which provide supplemental information about the circumstances in which the main clause is situated usually appear in the indicative rather than the subjunctive.

- (5.168) *at-on om=ohhər Moḥamrā qə=mwadd-en ekkak* come.PFV-3PL from=road PN IND=take.IPFV-3PL there They came along Khorramshahr Road [where] they were sent. (III.3)
- (5.169) *nəpāq-yon barra ani qə=mṭall-én* exit.PFV-3PL outside we IND=play.D.IPFV-1PL They came outside [where] we were playing. (II.13)

5.3.4.3 Purpose Clauses

Clauses which indicate the purpose for the main clause take verbs in the subjunctive.

(5.170)	əzg-it	ya	kədā <u>b</u>	mandāyí	əmḥaṣel-nā
	go.PFV-1SG	а	book	Mandaean	obtain.SBJV-1SG
	I would have gone to pick up a Mandaic book. (III.16)				

(5.171)	q=all-én	əmzabn-én	qəlāww=u
	IND=go.IPFV-1PL	sell.D.SBJV-1PL	to=3PL
	We used to go to sell to		

5.3.4.4 Reason Clauses

Verbs in clauses which indicate the reason for which the action or situation in the main clause occur typically appear in the indicative.

5.3 Moods and Modalities

(5.172) *at-on gāw Irān qə=beyy-en* come.PFV-3PL in PN IND=want.IPFV-3PL They came to Iran [because] they wanted

komak	abdəl=l=ø	əl=Rusya
help	do.SBJV-3PL=OBJ=3SG.F	REF=PN
to help Russia. (I	II.2)	

5.3.4.5 Simultaneous Clauses

The verb of a clause in which the action or situation occurs simultaneously with that in the main clause will appear in the subjunctive.

(5.173) $\partial mar \cdot \phi$ \bar{a} $t\bar{a}j\acute{e}r$ narm $tamm\bar{a} \cdot \phi$ say.PFV-3SG.M this merchant soft become.PFV-3SG.M He thought to himself, "This merchant has gone soft in the head,"

āmer-ørāstq=amen-nāsay.SBJV-3SG.MrightIND=say.IPFV-1SGwhile saying, "I'm telling the truth." (V.14-15)

 $\begin{array}{cccc} (5.174) & \bar{a}\underline{t}i \cdot \vartheta & belw \bar{a} = d = an & \bar{a}\underline{b}ed \cdot \vartheta \\ & \text{come.SBJV-3SG.M} & \text{disturbance=of=1PL} & \text{do.SBJV-3SG.M} \\ & \text{He may come to make trouble for us,} \end{array}$

āmer-ø	gənab-tonn=ø	əl=kərāy=e
say.SBJV-3SG.M	steal.PFV-2PL=3SG.F	REF=leg=1SG
saying, "You stole m	y leg!" (VII.9)	

5.3.4.6 Conditional Clauses

Neo-Mandaic recognizes three basic types of conditional clauses: simple conditionals, hypotheticals, and counterfactuals.

Simple Conditional Clauses

The indicative is used in the protasis to form simple conditional clauses. The copula may also appear in the protasis, e.g. (5.175). The verb of the apodosis of the simple conditional clause is either found in the indicative or the imperative, e.g. (5.176).

- pərāh-ā eh=l=e qə=zaben-nā (5.175) agar turt-i COP=to=1SG cow-INDF IND=buy.IPFV-1SG if money-AUG When I have money, I will buy a cow.¹⁹⁵ gən-it (5.176) anā agar əbud-ø āt guš
- I if sleep.PFV-1SG you ear do.IMP.F-SG If I fall asleep, keep a look out. (IV.46)

Hypothetical Conditional Clauses

Hypothetical conditional clauses employ the subjunctive in the protasis to posit hypothetical or imaginary situations in the past, and the indicative to posit such situations in the present. These differ from conterfactual conditional clauses (see below) in that the apodosis appears in the imperfective, rather than the perfective, indicating that the situation, while hypothetical, is not outside the realm of possibility.

- (5.178) an agar gāw ā 'aqəl hawi-nā əzg-it I if in this mind be.SBJV-1SG go.PFV-1SG If I knew then what I know today, 196 I would have gone. (III.16)
- (5.179) $\bar{a}h\bar{a}$ qam enši $la=q=malle\bar{p}-\phi$ w=agar this to anybody NEG=IND=teach.IPFV-1SG and=if I will not teach this to anyone, for

 $q = malle\bar{p} = \emptyset$ $miet = \emptyset$ IND=teach.IPFV-1SGdie.SBJV-1SGif I did teach it, I could die. (IV.84)

- 195 Example adapted from Macuch, *Handbook*, 432. Compare with example (5.181) below.
- 196 Lit. 'if I were in this mind.'

Counterfactual Conditional Clauses

In counterfactual conditional clauses, the perfective is used to indicate that the information contained in the apodosis is explicitly counterfactual.

(5.180) agar lá=hw-it an 1SG NEG=be.PFV-1SG if If I had not been there. lá=att-at=ø əl=vana-ā NEG=come.D.PFV-3SG.F=3SG.M REF=baby-AUG she would not have given birth to the baby. 197 (5.181) agar həwā=l=e pərāh-ā turt-i zabn-it be.PFV=to=1SG cow-INDF buy.PFV-1SG if money-AUG

If I had had money, I would have bought a cow. 198

5.4 Voice

In Neo-Mandaic, the relationship of the action or state described by the verb to its arguments can be described by one of three voices: the active voice (§5.4.1), the middle voice (§5.4.2), and the passive voice (§5.4.3).

5.4.1 The Active Voice

When the action described by the verb is initiated by its grammatical subject, the verb is described as being in the active voice, and the grammatical subject is described as its agent. The forms of the active voice are outlined in §5.1 above.

5.4.2 The Middle Voice

The *t*-stem verbs described in §5.1.1.3 represent the middle voice in Neo-Mandaic. The agents of the actions described by verbs in these stems, which are syntactically active and intransitive, experience the results of these actions as if they were also the patient; in many cases, the action of the verb appears to occur on its own, e.g. (5.182)

- 197 Example adapted from Macuch, Handbook, 429.
- 198 Example adapted from Macuch, Handbook, 432.

and (5.183). As a result, verbs in these stems are often translated as if they were agentless passives, e.g. (5.184), or reflexive actions that the subject takes on its own behalf, e.g. (5.185).

(5.182)	<i>amānatdār</i> trustee The trustee was pleased	-	ø ed.tG.PFV-:	3SG.M
(5.183)	<i>bāb-ā</i> father-AUG The father woke up. (IV		ø en.tC-3SG.l	М
(5.184)	<i>etber-ø</i> break.tC.PFV-3SG.M A piece broke off of / w		3SG.M	<i>wusle</i> piece X.12)
(5.185)	<i>estəb-yon</i> baptize.tG.PFV-3PL They baptized / were ba		<i>napš</i> same cording to t	<i>mașbett-ā</i> baptism-AUG the same baptism. (I.10)

5.4.3 The Passive Voice

In the passive voice, the grammatical subject of the verb is the recipient of the action described by it, namely the patient. There are two ways of forming the passive voice in Neo-Mandaic: the analytic passive (§5.4.3.1) and the much more common impersonal passive (§5.4.3.2). The agent of the action is expressed neither by the inflection of the verb nor by an oblique complement in these two constructions; the agent of the action can only be expressed in the active or middle voice.

5.4.3.1 The Analytic Passive

As described in §5.1.1.4, the analytic passive is formed from the passive participle of the verb and the long or short forms of the copula.

Not all verbal roots yield a passive participle. Furthermore, many of the intransitive verbs yield passive participles that are passive in form but active in meaning, as in Syriac. The basic forms of the passive participles for the paradigmatic verbs found in §5.1 are given in Table 91.

5.4 Voice

Table 91: The G-Stem Passiv	e Participles
-----------------------------	---------------

Root Class	Mas	с.	Fen	n.	Plura	al
Strong	$C_1 a C_2 e C_3$	gəțel	$C_1 a C_2 i C_3 \bar{a}$	gəțilā	C ₁ əC ₂ iCen	gəțilen
I-Weak	$\partial C_2 e C_3$	əbed	əC₂iC₃ā	əbidā	∂C2iC3en	əbiden
II-Weak	C ₁ iC ₃	diš	C1iC3ā	dišā	C ₁ iC ₃ en	dišen
III-Weak	$C_1 a C_2 i$	həzi	C_1 ə C_2 yā	həzyā	C ₁ əC ₂ eyyen	həzeyyen

The basic form of the passive participle of the strong verb is $C_1
arrow C_2 e C_3$. Note that this form resembles the forms of the subjunctive with pretonic lenition, e.g. (5.186). The form $C_1
arrow C_2 e C_3$ is retained before suffixes beginning with a consonant, e.g. (5.187) but becomes $C_1
arrow C_2 i C_3$ - before the feminine morpheme $-\overline{a}$ and other suffixes beginning with a vowel, e.g. (5.189) and (5.190).

(5.186) qabin əmer I want to say (subjunctive, not passive)
(5.187) yəhemye he is seated
(5.188) yəhimā the seated woman
(5.189) yəhimen the seated ones

The third personal forms of the short copula are obligatory on the passive participle only when it is used predicatively. The third feminine singular personal suffix takes the form $-\dot{r}$; this unusual form (which is identical to the third *masculine* singular object and possessive suffix) presumably arose after the collapse of the diphthong produced by the feminine morpheme $-\bar{a}$ and the short copula, e.g. (5.190). The final consonant of the first singular personal suffix *-nān* is often subject to deletion, e.g. (5.191).

```
(5.190) yəhimi she is seated (from*yəhimā-ye)
(5.191) yəhemnā I am seated
```

For the derived stems, it is not possible to reproduce a full paradigm of passive participles. The *D*-stem is represented by one passive participle, $\partial m \check{s} a b b \bar{a}$ 'praised,' which belongs to the III-weak root consonant class. The *C*-stem is also represented by one III-weak passive participle, $mah b \check{a}$ 'kept.'

5. The Verb

(5.192) əmšabbā-ø Mār=e praise.D.PASS-SG lord=1SG Praised be my Lord!
(5.193) kol waxt rāz-i maḥb=i every time secret-INDF keep.C.PASS=3SG.F It has always been kept a secret. (I.13)

5.4.3.2 The Impersonal Passive

The most common and productive form of the passive is the impersonal passive construction, in which an impersonal 'they' occupies the role of the agent. Whereas most personal pronouns correspond to a noun phrase that has already been introduced into the discourse, the impersonal 'they' is immediately distinguished by the fact that it has no referent within the context of the discourse.

(5.194)	jisr-i	tum	əbad-yon	
	bridge-INDF	again	do.PFV-3	PL
	Another brid	ge had been bui	lt. (III.21)	
(5.195)		əbad-yon	gāw	welāt
	herald	do.PFV-3PL	in	city

It was proclaimed in the city. (V.20).

The patient of this impersonal passive is expressed either by the enclitic object suffixes or by the noun in object position.

(5.196)	la=ša <u>b</u> q-onn=an	all-enni	ekka <u>k</u>
	NEG=leave.PFV-3PL=1PL	go.SBJV-1PL	there
	We weren't allowed to go the	re. (III.13)	
(5.197)	gațl-onn=u	geš=d=u	
	kill.IPFV-3PL=3PL	all=of=3PL	
	They were all killed. (III.23)		

Frequently, the impersonal passive appears in topic-comment or resumptive sentence form, the patient being expressed as the topic and the passive verb (with a resumptive enclitic object suffix) following as its comment. Note that the patient, once topicalized, does not take the enclitic object marker; cf. (5.198) with (5.199).

(5.198)	ā	xuțb	šadr-onn=i	qamd=in
	this	letter	send.PFV-3PL=1SG	to=1SG
	This letter was sent to me.			

~ 4	T T	
5 /	Voi	ICA
J.T	101	ιυυ

(5.199)	<i>šadr-onn=</i> ø	əl=ā	xuțb	qamd=in
	sent.PFV-3PL=3SG.M	REF=this	letter	to=1SG
	This letter was sent to me			

6. Conclusion

6.1 Classical Mandaic and the Modern Dialects

In light of the data outlined in the preceding chapters, it is possible to make a few general comments about the relationship between Classical Mandaic and the Neo-Mandaic dialects of Ahvāz and Khorramshahr, as well as the relationship between the two contemporary dialects. My remarks here will necessarily be brief and far from comprehensive, as a full description of the historical grammar of the modern dialects falls beyond the scope of this grammar. They should, however, serve to give a general idea of the differences between these three dialects.

6.1.1 Phonology

Classical Mandaic is written with a rudimentary alphabet rather than one of the abjads employed by other Aramaic dialects (and, indeed, most other Semitic languages).¹⁹⁹ The advantages of the Mandaic script against these other scripts are two-fold: it employs distinct vowel signs rather than *matres lectionis*, which serve both as consonants and as vowels, and its orthography represents the contemporary pronunciation of the language to a more consistent degree than that of the other Aramaic dialects, the orthographies of which are generally a composite of vernacular forms and historical spellings preserved by the literary tradition. As a result, it is possible to divine a great deal about the phonology of Classical Mandaic.

Unfortunately, there are a number of phonemic distinctions that are not represented by the Mandaic script. These include some

¹⁹⁹ An abjad is a writing system in which each grapheme represents a consonant. Vowels may be represented by vowel markings on the basic graphemes. Alphabets, by contrast, have basic graphemes for both vowels and consonants. For more on this distinction and its relevance to the Mandaic script, see Häberl, "Iranian Scripts for Aramaic Languages."

qualitative and all quantitative distinctions in the vowels, in addition to consonant gemination, stress, and the fricativization of the /b/, /g/, /d/, /k/, /p/, and /t/ series of stops. Consequently, a full understanding of Classical Mandaic phonology is possible only with the aid of historical reconstruction (of which the first and only exhaustive attempt was Nöldeke's *Mandäische Grammatik*, published in 1875), the traditional pronunciation of the texts (documented by Macuch in his *Handbook*, published in 1965), and the modern dialects (documented in Macuch's *Handbook*, his two text collections from 1989 and 1993, and this grammar). A full account of Classical Mandaic phonology in light of this evidence remains to be done and is a basic prerequisite for any discussion of the historical phonology of the modern dialects. Nonetheless, it is possible to draw a few conclusions about the development of Mandaic phonology, from the classical language to the modern dialects.

The first (and most obvious) innovation is the introduction of new phonemes not found in Classical Mandaic, such as $\check{c}/tJ/, j/dz/, c/S/$ and $\dot{h}/\hbar/$ (see §2.1). Later stages of the language, such as Post-Classical Mandaic and Neo-Mandaic, use digraphs and diacritics to indicate these new phonemes in writing, as they are not indicated in the Classical Mandaic script. Additionally, segments that were formerly allophones of Classical Mandaic phonemes, such as the spirants /v/, / $\kappa/$, / $\chi/$, /f/, and / θ /, which were allophones of /b/, /g/, /d/, /k/, /p/, and /t/ in post-vocalic position, have become phonemic in Neo-Mandaic and have a slightly different distribution, occuring in word-initial position among other places.

Another innovation is the collapse of the Classical Mandaic diphthongs /ai/ and /au/ to /i/ and /u/. In Classical Mandaic, these diphthongs had already collapsed in closed accented syllables. In the dialects of Ahvāz and Khorramshahr, these diphthongs have also collapsed in open accented syllables. See \$2.3.2.1 and \$2.3.2.2 for further details. There is some evidence that the collapse of diphthongs in open accented syllables had not yet occurred in the Iraqi dialect documented by Drower (see \$2.3.2.1). Other diphthongs (listed in \$2.4.3.3), including those in words borrowed from other languages or resulting from inflectional processes, are retained in the dialect of

Khorramshahr, but in Ahvāz other diphthongs appear to be succumbing to this process; compare Khorramshahr $g\bar{a}w/g_{00}/$ 'in' with Ahvāz gu/gu:/ id.

Closely tied to the collapse of the diphthong /aɪ/ in open accented syllables is the breaking of its outcome, /iː/, in the same environment; see §2.5.3.5 for further details. This sound change is typical of both Ahvāz and Khorramshahr but is not present in either the text from Iraq collected by Drower or the earlier texts collected by Macuch in Ahvāz. The later texts from Ahvāz also show some evidence of palatalization not found in Khorramshahr.²⁰⁰ The final conclusion must be that the phonology of the dialect of Khorramshahr is slightly more conservative than that of Ahvāz, although both dialects have experienced substantial phonological changes in recent years.

6.1.2 Morphology

Both of the documented modern dialects retain much of the morphology of the classical dialect. While the morphology of the noun has been heavily influenced by Persian, the morphology of the verb remains fairly intact and free of contact influence. Much more about the morphology of the verb can be divined from the dialect of Ahvāz than that of Khorramshahr, although this is primarily due to the larger corpus of texts collected from that dialect. Despite all the evidence for the morphology of both dialects that has thus far been collected, much more work remains to be done.

6.1.2.1 The Noun

The nominal morphology of Neo-Mandaic has been greatly affected by contact with Persian. The classical system of states (see §3.4) has become obsolete, and only vestiges of it survive in some frozen forms and grammatical constructions. The primary distinctions made by the morphology of the noun reflect referentiality and identifiability (see §6.1.3 below). As a result, the most common inflectional morphemes associated with the states have been replaced by morphemes borrowed

²⁰⁰ See Table 4 in this grammar and §2.3.1 in Macuch, *Neumandäische Texte*, 38 for examples.
from Persian, such as the plural morphemes ($\S3.3.1$) and (\$3.3.3), the indefinite morpheme (\$3.6.3), and the *ezāfe* (\$3.4.2). This last morpheme, which appears in both the dialects of Ahvāz and Khorramshahr, is far more characteristic of the speech of the former rather than the latter.

Despite the collapse of the system of states, much of the morphology of the noun has been preserved, and the influence of Persian has not been as pervasive as it initially might seem. Grammatical gender (which is not found in Persian) and its morphological markers have been preserved ($\S3.2$), and many of the original heteroclite plurals have also been retained (\$3.3.2), even though the most common plural morpheme -**ia** has been replaced by one borrowed from Persian (\$3.3.1). It is also worth noting that Neo-Mandaic has innovated a system to keep loan words distinct from the native vocabulary by morphologically marking them in the plural and before possessive suffixes. This system is unique among the living Semitic languages (most of which aggressively assimilate loan words to their own phonology and morphology) and perhaps among the languages of the world.

One of the major innovations of both dialects of Neo-Mandaic visà-vis Classical Mandaic is the new means of marking referentiality and identifiability on the noun (see §3.6). These means include morphological indicators such as the referential object marker (§3.6.2) and the indefinite morpheme (§3.6.3), as well as various syntactic ways of indicating referentiality (§3.6.1). This is an areal phenomenon affecting languages straddling the border between the Semitic world, Iran, and Anatolia, including Neo-Mandaic and various other Neo-Aramaic dialects. The first attestations of the indefinite morpheme, originally borrowed from the Iranian languages, are found already in the Classical Mandaic texts.²⁰¹

Related to the general category of referentiality is the phenomenon of noun incorporation (§3.5.2). Whenever the object of the verb refers to an entire class of referents rather than one specific referent, it appears before the verb rather than after it and is incorporated into its meaning. Once incorporated in this fashion, the noun may take the

201 Macuch, Handbook, 207.

indefinite morpheme, indicating a restrictive selection out of the range implied by the generic noun.

6.1.2.2 The Verb

As mentioned above, the Neo-Mandaic dialects preserve much of the morphology of the Classical Mandaic verb. All of the forms employed in Neo-Mandaic are also attested in Classical Mandaic, and Neo-Mandaic preserves (to varying extents) all Classical Mandaic forms with the exception of the prefix conjugation, described by Macuch as the imperfect.²⁰² The derived stems, including the *D*-stem, the *C*-stem, and the so-called "reflexive" stems, are also attested in both dialects of Neo-Mandaic, although more examples are attested in the dialect of Ahvāz than that of Khorramshahr. This is likely due to the greater volume of texts collected from the former. The reflexive stems are quite uncommon in Classical Mandaic, and varnishingly rare in both modern dialects. The passive forms of the other derived stems are fairly well attested in the modern dialect, even though they appear to be giving way to the impersonal passive construction (see §6.1.3 below).

Both dialects preserve three of the six thematic vowel classes of Classical Mandaic (see §5.1.1.1), although the thematic vowel of the perfective of many of the verbs originally from the $(e \sim o)$ and the $(o \sim o)$ thematic vowel classes have leveled towards that of the $(a \sim o)$ vowel class, which is the largest and most productive of the three in Neo-Mandaic. This process, which had begun already in the classical dialect, appears to have advanced further in the Neo-Mandaic dialect of Khorramshahr than it has in that of Ahvāz, which preserves more verbs in the former two classes than the dialect of Khorramshahr does.

With regard to the different root classes of the verb (see \$5.1.1.2), both modern dialects show important innovations in comparison with the classical dialect. While the paradigm of the strong verb remains much the same, the paradigm of the I-*n* and I-*y* verbs have been reformed on the analogy of the strong verbs, and the collapse of the

²⁰² All functions of the classical imperfect have been assumed by the Neo-Mandaic imperfective (which Macuch calls the "participial present-future tense"). Participles inflected as verbs are found already in Classical Mandaic.

original Classical Mandaic diphthongs /aɪ/ and /au/ has wreaked havoc upon the II-weak verbs and affected the paradigm of the III-weak verbs. Many geminate verbs had already been reformed on the analogy of the middle-weak verbs in Classical Mandaic. The reduction in the number of root classes found in Neo-Mandaic is thus the latest stage of a process which had already begun in Classical Mandaic.

The modern dialects are further distinguished from the classical dialect by the forms of the personal morphemes on the verb. The full third person plural suffix on the Classical Mandaic perfective is -**u**. The form used in the modern dialects is *-yon*, which is attested in the classical as **-un** or **-iun**, albeit less frequently. Conversely, in Neo-Mandaic, the final *-n* is consistently dropped from the first person plural suffix on the perfective, **-nin**, Neo-Mandaic *-ni*. Neo-Mandaic preserves the classical personal suffixes on the imperfective (Macuch's "participial present-future"), although once again the final *-n* is consistently dropped from the first person plural suffix, *-inin*, Neo-Mandaic *-enni*. The form of this suffix before object suffixes, *-nan-*, retains the final *-n*, but the vowel of the modern suffix differs from that of the classical dialect.

One of the more obvious differences between the two modern dialects is the absence of the third and second feminine plural forms in the verbal paradigms of the dialect of Khorramshahr. Furthermore, there is a second feminine singular form in the dialect of Ahvāz, formally identical to the first common singular, that is attested in neither the dialect of Khorramshahr nor in Classical Mandaic. It appears to be a complete innovation in the dialect of Ahvāz.

In his *Handbook*, Macuch claims that the Neo-Mandaic third plural forms are "rare in the classical, but they are used with absolute consistency in modern Mandaic," and that the second feminine plural, though often conflated with the masculine in Classical Mandaic, is "still used with relative consistency" in Neo-Mandaic.²⁰³ In the paradigms offered in his two text collections, however, the feminine forms are enclosed within brackets, suggesting that he may have changed his mind about the consistency of their appearance. The brevity of the sketch grammars attached to both text collections

203 Macuch, Handbook, 262.

6. Conclusion

prevents us from drawing any concrete conclusions about the distribution of these forms. In any case, more research is necessary to determine whether the use of the feminine forms in Ahvāz is an innovation in that dialect, an unconscious (or perhaps deliberate) archaism on the part of his informants (both of whom were literate in the classical dialect), or a genuine survival of the classical forms, albeit with a different distribution in the modern language.

Another major innovation is the impersonal passive (§5.4.3.2), which is the most productive form of the passive in Neo-Mandaic and coexists alongside the analytic passive construction inherited from earlier stages of the language.

6.1.3 Syntax

The syntax of Classical Mandaic has received little attention since the days of Theodor Nöldeke. As Macuch himself notes,²⁰⁴ syntax was the aspect of Mandaic grammar least addressed by him in his *Handbook* and the two text collections. As a proper examination of the comparative syntax of the various Mandaic dialects would fall beyond the scope of this grammar, I will restrict myself to a few general remarks about this subject. A full study of the syntax of Neo-Mandaic must remain a desideratum for the immediate future.

It would appear that Neo-Mandaic preserves the SVO word order of Classical Mandaic, despite its longstanding contact with Persian (which follows SOV word order). Topic-fronting, which tends to obscure the word order, is typical of all three languages. Simple sentences consist of a subject, which may be implied in the verb, and a predicate, which is headed by a verb or the copula (see §5.1.1.4). The independent forms of the copula introduce predicate nominal and predicate locative constructions, and the enclitic forms introduce predicate adjectives. Much like other Semitic languages, Neo-Mandaic employs a predicate locative construction to express the notion of possession. In the simple present tense, this construction uses the independent form of the existential particle *et and the preposition I 'to/for,' which takes the enclitic suffixes introduced in

204 Macuch, Handbook, 383.

Table 15. Before *I*, the existential particle assumes the form *eh*-, yielding the forms *ehli* 'he has' (lit. 'there is for him'), *ehla* 'she has,' and so forth. In tenses, aspects, and moods other than the simple present, the copular verb $h \partial w \bar{a} \sim h \partial w i$ ($h \bar{a} w i$) is used in the place of the existential particle, e.g. *agar pərāhā həwāle, turti zabnit* 'if I had money, I would have bought a cow.'

Compound sentences combine two or more simple sentences with coordinating conjunctions such as u 'and,' ammā 'but,' lo 'or,' and the correlative conjunction -lo ... -lo 'either ... or.' Complex sentences consist of a main clause and one or more dependent clauses introduced by a relative pronoun, provided that the referent of the antecedent of the clause is definite-if it is indefinite, no relative pronoun is used. The Classical Mandaic relative pronoun d- has not survived, having been replaced by *illi*, an Arabic loan that introduces non-restrictive relative clauses (§4.3.2.1), and ke, a Persian loan that introduces restrictive relative clauses (§4.3.2.2), both of which appear immediately following the antecedent of the clause. It is worth nothing that the distinction between restrictive relative clauses and non-restrictive relative clauses is not found in Classical Mandaic, and therefore constitutes an important innovation distinguishing the contemporary language from the classical. The antecedents of restrictive relative clauses are marked with the restrictive morpheme -*i*, which resembles the indefinite morpheme in form alone, e.g. *ezgit* dukkāni ke həzitu awwál 'I went to the places which I saw before (III.19).' If the antecedent is the object of the relative clause, it will be represented within the relative clause by a resumptive relative pronoun, as in the example above (*həzitu* 'I saw them').

6.1.4 Lexicon

As indicated in the introduction (§1.6.4), the Neo-Mandaic dialect of Khorramshahr preserves much more of the vocabulary of Classical Mandaic than scholars such as Macuch and Drower have previously indicated. As much of 85% of the "basic" vocabulary of Neo-Mandaic (here represented by the 207 term Swadesh list) is Aramaic in origin; the remaining 15% derives from Arabic, Persian, and the other languages with which it has come into contact. While there has been

some semantic development (e.g. *barrak* ~ *barrak* (*mebarrak*) 'to pray' is derived from the classical verb meaning 'to kneel' or 'to bless' in the *D*-stem), the majority of this vocabulary also appears to preserve its original meaning. There are even a few words that appear to be classical in form (e.g. *yəmutā* 'oath', cf. CM '**umamata** id.) but are not attested in any of the classical texts.

Within the modern dialects, certain differences in vocabulary distinguish speakers of the two surviving dialects from one another, e.g. Ahvāz *mork* 'with' vs. Khorramshahr *orke* 'with.' Unfortunately, due to the small size of the remaining communities of speakers and the absence of a written standard, the total vocabulary of Neo-Mandaic (as distinct from that of Mandaic in general, including the written language) is restricted almost entirely to that of the individual speakers, and vocabulary outside of the domain of "basic" vocabulary is often freely borrowed from other languages. Nevertheless, the vocabulary collected herein and in Macuch's publications comprises several thousand terms. If these terms were combined with the vocabulary collected from other sources and other informants, they could serve as the basis for a very extensive dictionary of this Neo-Aramaic language, which would be extremely useful for all scholars of Aramaic.

Certain semantic areas (e.g. cultural terms, modern technology, numerals) have become thoroughly saturated with loans. Even so, most recent loan words continue to be marked with a separate set of morphemes in the plural (§3.3.3) and before possessive pronouns (§4.1.2.3). These morphemes are not found in the Classical language but are typical of the modern dialects. It is my impression²⁰⁵ that the dialect of Khorramshahr preserves much more of the original vocabulary, and that the lexicon of Ahvāz has been more heavily influenced by Persian. Without a dictionary of the modern language, however, this impression must remain unsubstantiated. The situation

²⁰⁵ This impression is derived primarily from my sessions of reading Macuch's *Neumandäische Texte* with Sobbi who, during the course of these sessions, would occasionally remark, "I don't know why he used that word. We have a perfectly good word..."

is complicated by the fact that my chief informant has an unusual gift for coining neologisms; this was confirmed for me when he took issue with Choheili's use of the word *tayyāra* 'airplane' (a word found in Persian and many colloquial Arabic dialects, originally coined by the Ottomans), claiming that the word *pəhirā* was the proper Mandaic term for airplane.

6.2 Conclusions and Desiderata

The Neo-Mandaic dialects are characterized by a remarkable uniformity, as few differences separate the dialect of Khorramshahr from that of Ahvāz or the Iraqi dialect documented by Drower. The two surviving dialects that have thus far been documented are mutually intelligible to the point that speakers of either dialect will deny that there are any differences. A possible explanation for the uniformity between these dialects is given in the introduction (see §1.7). There are, however, sufficient differences to justify examining them as separate dialects in their own right, owing to their different origins and the ways in which they have diverged from one another in terms of their phonology, morphology, and lexicon.

The ensemble of the features described above suggest that the grammar of Neo-Mandaic is remarkably conservative in comparison with that of Classical Mandaic, and that most of the features that distinguish the former from the latter are the result of developments already attested in Classical and Postclassical Mandaic. Though it was not the chief focus of this grammar or either of Macuch's text collections, a thorough historical grammar of Mandaic could potentially be derived from the collected material on the modern dialects and the existing grammars of the classical dialect. Given the position of Neo-Mandaic within the Aramaic family (being a distinct branch of the Eastern Aramaic sub-group and the only documented form of Neo-Aramaic derived directly from a literary dialect), such a grammar would naturally be of great interest not only to Semitists but also to linguists in general.

6. Conclusion

As "the most massive and inclusive art we know, a mountainous and unconscious work of anonymous generations,"²⁰⁶ language necessarily defies any attempt at a fully comprehensive treatment. Furthermore, the exigencies of documenting endangered languages often make comprehensive treatments impossible. For these reasons, grammars such as this one can only be the latest word on the subject, never the final one. If I had extended the same treatment to Neo-Mandaic syntax that I have already extended to phonology and morphology, I could easily have filled another volume. Likewise, I have avoided analyzing the Neo-Mandaic dialect of Khorramshahr from a diachronic perspective, an analysis which is also worthy of its own separate treatment. Those volumes, and many others, remain to be written.

²⁰⁶ Edward Sapir, *Language: An Introduction to the Study of Speech* (New York: Harcourt Brace, 1921), 220.

7. Text Collection

I am fortunate to have found in Nasser Sobbi a consummate storyteller and a fount of information about the Mandaeans and their religion. Nasser Sobbi is equally at home discussing religion, politics, history, and nearly any other subject related to the Middle East. He is able to draw from a vast store of knowledge accumulated during a lifetime of experiences and from his personal library, much of which was unfortunately destroyed during the Battle for Khorramshahr (1980).

Consequently, it was not difficult to elicit texts from him on a wide variety of subjects that were both interesting and illustrative of the finer points of Neo-Mandaic grammar. I have selected ten of these texts for inclusion in this grammar. All of these texts were recorded in the basement of the Sobbi household, in which Nasser Sobbi's office and personal library are found, as well as a pantry, a laundry room, and a guestroom. The elicitation sessions all took place in the early afternoon, between the hours of 3-5 PM. Unless otherwise noted, the only people immediately present were Nasser Sobbi and myself, although his wife Shukrieh was always within earshot, and would often wander into the basement as she performed her household chores, occasionally commenting on the session or contributing some information.

It seemed only natural to begin the text collection with the speech that he gave at the ARAM Thirteenth International Conference, "The Mandaeans" (Harvard University 13-15 June 1999), on the opening day of the conference. I recorded the text on a second occasion on June 13, 2003, in the company of my friend Cliff Shin from nearby Great Neck, NY.

The following two texts are autobiographical anecdotes, both recorded on December 13, 2003. The first details his encounter with Lady Drower in the year 1932, when he was only eight years old. The second text describes two trips he made to the cities of Shushtar and

Dezful in northern Khuzestān, the chief cities of the Mandaeans in Iran during the 19th century. The first trip took place during the Second World War, in 1943, and the second trip occurred in 1995.

The fourth text, collected on March 27, 2004, is the largest of the ten, and the most interesting from a literary perspective. Jorunn J. Buckley first drew my attention to this legend, which is a Neo-Mandaic version of the most famous Mandaean folktale, the Bridge of Shushtar. The basic outline of the plot is as follows: the ruler of Shushtar seeks to build a bridge over the river Karun. The first time the bridge is built, it collapses overnight. The collapse of the bridge is attributed to supernatural causes. The second time it is built, the ruler enlists dervishes and wise men to aid the builders while they are constructing the bridge. Once again, the bridge collapses overnight. The third time the bridge is built, the elders of the city advise the ruler to seek the help of the chief priest of the Mandaeans, who are widely regarded for their skills in fighting demons. The chief priest is summoned and instructs the ruler to rebuild the bridge a third time, and that he and his daughter will stand watch over the bridge once it is completed. Surely enough, the demon arrives and attempts to destroy the bridge, but before he can do so, he is tricked by the daughter into revealing his name. Armed with the knowledge of the demon's name, the chief priest enslaves him and commands him to stop destroying the bridge.

When the chief priest informs the ruler that he has succeeded in trapping the demon, the ruler demands that he relinquish control over the demon to him. The demon refuses to allow the Mandaeans to teach anyone else his secret (swearing that he will kill them first), and so the chief priest instructs him to cause the earth around Shushtar to quake, in the hopes of compelling the ruler to give up his demands. While the earth is quaking, the ruler approaches the chief priest for help. The chief priest agrees, provided that the ruler stops demanding the secret from him. Once the ruler agrees, the chief priest summons the demon and asks him to stop shaking the earth.

He then asks the demon to move the entire Mandaean quarter of Shushtar out of the city, and the demon agrees to do so in exchange for his freedom. He instructs the Mandaeans to stay within their quarter for forty days while he digs beneath it. The jewelers, the carpenters, and the blacksmiths disobey these commands and leave the city before the demon lifts it into the air and carries it away.

The first half of this folktale is ultimately based upon the nearuniversal motif of the foundation sacrifice, despite the fact that the sacrifice itself has been thwarted in this version. Similar legends, involving the construction of a bridge or some other building that is repeatedly destroyed by a supernatural agency, circulate throughout the Balkans, Anatolia, Kurdistan, and even India.²⁰⁷ In nearly all of these tales, the mason responsible for building the bridge (or fortress) must sacrifice a female relation (either a wife or a daughter) in order to placate the supernatural agency responsible for destroying the bridge, usually by immuring her in its foundation. The Mandaeans have placed a uniquely Mandaean spin upon this legend—by having the daughter trick the demon into revealing his name, they reaffirm their age-old reputation as master magicians and trappers of demons. The Mandaean contribution to this folktale is apparent only once all of its variants have been compared and the principle of the selection of each variant has been determined.²⁰⁸

The second half of this folktale, which is specific to the Mandaeans and not found in the other versions, is clearly etiological; it explains why the Mandaeans of Khuzestān traditionally work in these three professions, and also explains the presence of a large pit outside the city of Shushtar, known to the residents of that city as the *Naṣṣat-e Sobba* 'Pit of the Sabians.' Versions of this folktale appear in translation in Petermann's *Reisen im Orient*²⁰⁹ and Drower's *The*

- 207 Alan Dundes has written on the distribution of this particular folktale and the folly of trying to attribute it to a particular region or people; see Alan Dundes, "How Indic Parallels to the Ballad of the 'Walled-Up Wife' Reveal the Pitfalls of Parochial Nationalistic Folkloristics," *Journal of American Folklore* 108, no. 427 (Winter, 1995), 38–53. For a variant in the Neo-Aramaic dialect of the Jews of Zakho, not mentioned by Dundes, see Donna Shai, "A Kurdish Jewish Variant of 'The Bridge of Arta," *Association for Jewish Studies Review* 1 (1976): 303–310.
- 208 This principle was established by Vladimir S. Propp in his *Morphology of the Folktale*, translated by Laurence Scott (Austin: University of Texas Press, 1968), 9.
- 209 Petermann, Reisen im Orient, 100.

Mandaeans of Iraq and Iran,²¹⁰ as well as a partial version in the Neo-Mandaic dialect of Ahvāz collected by Macuch.²¹¹ This particular version was collected by Drower herself in Neo-Mandaic using an *ad hoc* transcription system, but never published. Even though this version differs from the one given in her book and she did not record the name of her informant, the source for this story can be none other than her old friend Hirmiz bar Anhar, a silversmith from Baghdad. Bar Anhar, whom Sobbi met during the 1940s, was also descended from the Mandaeans of Shushtar, and given the many similarities between the dialect of this version of the story and Sobbi's own dialect, it is clear that the tale originated with him, like so many others found in Drower's book. With the help of Sobbi, I was able to make sense out of Drower's text, and recorded his rendition of it for this collection.

The fifth text, recorded on November 7, 2004, is a second folktale, which has an even more convoluted history than that of the Bridge of Shushtar. It is a selection from the *Kalila-o Demne-ye Bahrāmshāhi*, a 12th century Persian translation of Ibn al-Muqaffa^c's 8th century Arabic *Kalīla wa-Dimna* made by Abolma'āli Nasrollāh ebn-e Mohammad. The Arabic itself was allegedly translated from a lost Pahlavi translation of the Sanskrit Bidpai fables. In this tale, a merchant leaves a consignment of iron with a friend before departing on a long journey. While the merchant is gone, his friend sells the entire consignment and pockets the money. Upon his return, the merchant demands his iron back. His friend tells him an improbable story about the iron having been eaten by mice.

To his astonishment, the merchant appears to accept this improbable story and departs. As he is leaving the house, the merchant spots his friend's young son and kidnaps him. When he returns the next day, his friend tells him that his son has gone missing, and the merchant tells him an equally improbable story, claiming that his son has been snatched away by a hawk. The friend declares that the merchant's story is ludicrous, but the merchant retorts, "in the city

²¹⁰ Drower, The Mandaeans, 289-92.

²¹¹ Macuch, "The bridge of Shushtar," 153–172.

7. Text Collection

where mice can eat iron, shouldn't a hawk be able to carry off a young boy?" Realizing that the merchant is much shrewder than he previously thought, he exchanges the money from the iron for his son.

Four of the last five short texts are all traditional jokes, originally told in colloquial Arabic and retold in Neo-Mandaic. All five were recorded on November 27, 2004. Sobbi assures me that all of these jokes are very familiar to him and were of the sort that he and his friends would tell one another when he was a young man. The last one is taken from the cycle of stories surrounding the character Joha, also known as Mollah Nasreddin (in Persian) or Nasrettin Hoca (in Turkish). I have seen versions of this particular tale both in Classical Arabic and colloquial Levantine Arabic, as well as Turkish and even Ladino (Djudezmo) versions.

In a previous version²¹² of the present work, the texts were presented in interlinear format (much like the examples in the preceding chapters). In the interest of preserving space, the texts are here presented without interlinearization, followed by a free translation on the facing page or immediately below in the case of the shorter texts (VI-X).

²¹² Charles G. Häberl, *The Neo-Mandaic Dialect of Khorramshahr*. Ph.D. diss. Harvard University, 2006. Ann Arbor: ProQuest/UMI, Nov. 2006. AAT 3217746

Text I: Speech, June 13, 1999

6-13- 79

occed occur - purcostape Vacues oggeo Mysicand and 19M4 - Sid Marcho of abe Menorals Toto Nara the and Maland Maran Maran Maran der representation for the the march of a cheer and an is abd more do man here . No caland - theor ata and Acer or Andre or share and a man and a man a e acesters vore vogel aperap do voeram vogele Vocalos 11 دمد مدحم Balo africe damagener and Mrs and Mrsa Vacyando merano en en en alo apar la miceros Vocalasi en en אבוב גר אבם. אאבמוצייט שנ ניישאביויאמע אף, וביטא אר אב אר alter make or orale gores to be service to the to be all المعدة عد والمد الحليمة معم (عمر المحليم) - السم عمد arel occur = 1980 - 10 - 10 - 10 cor and a stand - and appendig בומבו . גבוצה ביצ מו בות ביצי מי און הגרוש השומבו , גמבומבו 40. all Var master at 900 - lo at 4 mast 9999 - S Voce alma + cy 60 ano cosasto ques qose as que apquesas - wo vocas ده د مارجم دور الجرب عد . مر الجرب (المحسورة دلم مراحد ما مردام The Aodesdo to an on persent of cable of the هسه سهايمه والحلا العليمة، ومهدوها على العم سامه مردامي العدوال · pay / ve marage so are many and are and like are and las Wenner . come occur Icu it asho in the star of a mu house 14mm - 44mg

Original text of speech given by Nasser Sobbi at Harvard University on June 13, 1999, in Sobbi's own handwriting.

Text I: Speech, June 13, 1999

(1) B-ešmehon ed-Heyyi Rabbi!

(2) Asuţā nehwilkon, inšā u gobrānā. (3) Qabin əmer farwāhkon geš illi aţiton qam kānfərāns šerš Mandayānā. (4) Anā genz həternān qəhazinkon geš mork hədādā gāw Harvard University atiton əm-kol dukkān ālmā məḥadtetton qam hayot Mandayānā. (5) Qabin əmer farwāh ənāš ARĀM illi abdonn² əl-ā kānfərāns, əmarlonnan, "Doton, yəhomyon əm-ork yəlupānā u ḥadetonn² əl-ā kānfərāns, qəyān, u mimrā d-Mandayān." (6) Qabin əmer genz farwāh Profesər Buckley, illi genz əbād abdat qam Mandayānā, u zarzat əlāw ənāš ARĀM qəmsadart² əl-ā kānfərāns qəmbašqart Mandayānā.

(7) Qabin əmhadet mendi, əm-riš yum šeršan, əm-Ādam, No, w-ebri Sām, ebbər ebri Rām, tum ā-yumā. (8) Ani Mandayānā gāw hemanutan šeršan m-awwál ālm həwat, w-Ādam, gobrā qadmāyi bāb kol barnašānā. (9) Kədebye gāw kədāb moqaddas-dan, Genzā Rabbā, Hibel Ziwā, šlihā d-kuštā d-Māre, şeby' əl-Ādam gāw eššəm Heyyi Māre u Mandā d-Heyyi. (10) Nuh, u ebri Sām, ebbər ebri Rām, eştebyon gāw napš maşbettā illi eštā qəmeştebén gāwwa. (11) Ani Mandayānā awwál barnāš həwini ke bašqarni əl-Māre ehdi ekti, u abd' əl-kol almānā, u kol mendi gāwwu.

(12) Ani lektan Zardoští wə-le-Yhuwəțāyā we-l-Masihí we-l-Māni. (13) Ani šeršan tabliģi lekta, u kol waxt rāzi mahbi, u min barnašāni ke qəmqarellan ehdi m-ā šeršān ektan, ahni şatyon əm-lu Mandayānā. (14) Ani katkammā həzini batluktā u mehnāt, u genz əhabni gutluktā, u radpā həzini, əm-šəbābānan, u-genz ənāš andešyon epseqni geš-dan, ammā ani tamini həyānā tum eštā, we-qəmbarrek hawén həyānā tum āxír ālmā, gāw hil Heyyi Rabbi.

(15) Farwahkon geš.

(1) In the name of the Great Life! (2) Greetings, ladies and gentlemen.

(3) I want to thank all of you who came to the conference on the religion of the Mandaeans. (4) I'm very happy that I am seeing you all together at Harvard University, having come from all over of the world to discuss the survival of the Mandaeans. (5) I want to thank the members of ARAM who held this conference, and who said to us, "Come, sit with the scholars, and talk about the religion, language, culture, and literature of the Mandaeans. (6) I want to give many thanks to Professor Buckley, who has done so much work for the Mandaeans, and informed the members of ARAM that you are organizing this conference, and that you know the Mandaeans.

(7) I want to say a few things, from the beginning of our religion, from Ådam, No, and his son Sām, his grandson Rām, until today. (8) As for us Mandaeans, according to our beliefs, our religion has existed from the first epoch of time, and Adam, the Primal Man, is the father of all people. (9) It is written in our sacred book, the *Genzā Rabbā*, that Hibel Ziwā, the apostle of God's truth, baptized Adam in the name of Life, God (lit. my Lord), and Knowledge of Life (three divine epithets). (10) Noah, his son Shem, and his grandson Aram baptized with the same baptism ceremony with which we baptize today. (11) We Mandaeans were the first people who recognized that God (lit. my Lord) is a single being, and that he created all the worlds, and everything within them.

(12) We are neither Zoroastrian, nor Jewish, nor Christian, nor Manichaean. (13) Our religion is non-proselytizing, and has always been kept a secret, and some people, those who labeled us one of these religions, those people heard [it] from non-Mandaeans.²¹³ (14) We have experienced so much misfortune and hardship, and we experienced many massacres, and experienced persecution from our neighbors, and many people were concerned that all of us had been exterminated, but we have survived until now, and I pray that we will remain alive until the final epoch of time, by the power of Heyyi Rabbi.

(15) Thank you all.

213 The third masculine singular suffix is occasionally omitted, as noted in §4.1.2.1.

Text II: Meeting Lady Drower

(1) Qabin əmer mendi eqdakarnā, ke həwit jihel, waxti ke həwit jihel. (2) Ak waxt babbe həwā—bāb emme yaʿani—həwā rab Mandayānā. (3) Lu ganzibriye, həwā rabbā ekkak.

(4) Gāw šet šubbā, təmānā—láqdakar şun'a—šettā šubbā, təmānā, həwini qəmṭallén gāw maḥallat jihelā geš-dan. (5) Aton əm-Aḥwāz. (6) Qəbeyyen ābden qābin gāw Moḥamrā. (7) Həwālan ganzibrā gāw Moḥambrā, ešmi Šiex Masbub. (8) Gāw Aḥwāz həwā tarmidi, ešmi Šiex ʿAbdalla. (9) Láhwālu ganzibrā, ammā eštā ehlu. (10) Aton qābin əbadyon m-Aḥwāz qam yeki, ešmi Yusef, qəlāw etti, ešma Ḥadiyye bərat Kanjar. (11) Həwā Yusef—Šabut qamrelli, ammā ešmi Yusef.

(12) Tum pardason. (13) Nəpaqyon barra, ani qəmtallén. (14) Qəhazén orku et inglizí həwat! (15) Nepqat barra qəmšiel əm-jihela qəmhatt Mandayí. (16) An həwit jihel. (17) Atit qamaya qabin bas əmhadet orka. (18) Inglizí? Láqəmbasqar Inglizí!
(19) Hattat orke Mandayí, jawab-d' ahbita. (20) Genz hattat ke genz jawab-d' ahbita.
(21) Ak waxt amrat, "Gāw Erāq, tarmidi u ganzibrā láqəmhatten Mandayí; həna hatta jihela qəmhatten!" (22) Ani həwini ečča eččin u bistar geš-dan qəmhatten Mandayí.
(23) Ehlan geš Mandayí həwa gāw Mohamra, faqat ke həna qaten m-Eraq, Azazi u Giyalin; láqəmhatten Mandayí.

(24) Qəyādkon man həwat ā ettā? (25) An láqyāde muyye Erāq, láqyāde tarmidi u ganzibrā u gobrā ekkak láqəmbašqeren əmhatt Mandāyí. (26) Ā ettā hidā amrat Ms. Drower! (27) Atat əm-Baġdād qār Šiex 'Abdalla. (28) Gāw Aḥwāz həwālu qābin.
(29) Qəbeyyen āten Moḥamrā, ya ani Xorramšahr sətā, u aton qār Šiex Masbub. (30) Qābin əbadyon. (31) Hid nepqat qəmšiel əmḥattā qə-jihelā. (32) Ya ani, Moḥamrā geš məḥatten Mandāyí həwā. (33) W-eqdakarnā tum eštā šekəl-da u napša. (34) Rəwāhā nehwila, mektat eštā. (35) Mār' əmzaharla! (1) I want to say something that I remember from when I was a child, the time when I was a child. (2) At that time, my grandfather, that is to say, my mother's father, was the leader of the Mandaeans. (3) He wasn't a *ganzibrā*, ²¹⁴ he was the leader there.

(4) When I was seven or eight, I don't remember well, the year was seven or eight, all of us children were playing in the neighborhood. (5) (Some people) came from Ahvāz. (6) They wanted to perform a marriage in Mohammerah. (7) We had a *ganzibrā* in Mohammerah, whose name was Sheikh Masboob. (8) In Ahvāz, there was a *tarmidā*,²¹⁵ whose name was Sheikh Abdallah. (9) They didn't used to have *ganzibrās*, but now they have one. (10) They came from Ahvāz to marry someone named Yusef to a woman whose name was Hadiyye, the daughter of Kanjar. (11) He was Yusef - he was called Šabut, but his name was Yusef.

(12) Then they finished. (13) They came outside where we were playing. (14) We saw that there was an English woman with them. (15) She came outside to ask the children to speak Mandaic. (16) I was a child. (17) I came up to her - I only want to speak to her. (18) But English? I don't know English! (19) She spoke Mandaic with me, I answered her. (20) She talked a lot because I gave her a lot of answers. (21) Then she said, "In Iraq, the *tarmidās* and the *ganzibrās* don't speak Mandaic; here, even the children speak (it)!" (22) Ninety-nine percent or more of us spoke Mandaic. (23) We all had Mandaic in Mohammerah, except for those (families) who came here from Iraq, the 'Azāzi and the Gilāni (families), who didn't speak Mandaic.

(24) Do you know who this woman was? (25) I didn't know what Iraq was, I didn't know that the *tarmidās* and the *ganzibrās* and the men there didn't know how to speak Mandaic. (26) This woman said (that she was) Ms. Drower! (27) She came from Baghdad to Sheikh Abdallah. (28) In Ahvāz, they had a wedding. (29) They wanted to come (to) Mohammerah, that is, Khorramshahr now, and they came to Sheikh Masboob. (30) They had a wedding. (31) She came out, asking to speak to the children. (32) That's because, (in) Mohammerah, everybody used to speak Mandaic.

(33) I still remember her appearance and her personality. (34) May she rest in peace. She's dead now. (35) May God (lit. my Lord) protect her.

214 A type of priest; see §1.2.2.

215 Another type of priest, of lower rank than the ganzibra; see §1.2.2.

Text III: Two Trips to Shushtar

(1) Qabin əmer mendi eqdakernā, gāw šet alfā u čemmā, ārbin u klātā. (2) Həwat šer horettā, u geš aton, maxṣuṣ Amrikān, Ingliz, aton gāw Irān, qəbeyyen komák abdell' əl-Rusya. (3) Aton əm-ohhər Moḥamrā, qəmwadden ekkak.

(4) An həwit əm-Provost Marshal (qəmbašqarle) ehle pāsi kol dukkān-e Amrikān qəmzabnen əlāwwu. (5) Həwit qanāyā d-kaspā (silversmith) ak waxt həwit.
(6) Ak waxt ezgini qam Haft Tappé, u qallén qam Bandar-e Ma'šur, qallén qam Dezful (7-11 həwālu ekkak), qallén əmzabnén qəlāwwu. (7) Ezgit əm-Şolḥabād (həwini gāw Şolḥabād) u qallenni, qam Dezful həzini ax waxt.

(8) An həwā orke yeki (rəwāhā nehwili) Yabber, met Yabber, ezgini ekkak, həzin' əl-nahrā. (9) Šənā əbadni ekkak, šamni u barrakni. (10) Ba'aden ləbašni eqmāšānan, šiyelni, "Qəhāzen bietwāt ba'id. (11) Ahni əl-mannon, lek barra əmbelātnon?"

(12) Əmaryon "ahni," gāw Fārsi, "xāne kāferān," ya'ani bietwāt ke illi (láqəmbašqer) šerš lehlu, biet māl şobbihā, xāne kāferān, şubbiye, şābi'e, ya'ni biet Mandayān.

(13) An həzit bietwāt illi mandi həwālu, əlāw nahrā, 'ād lasāqit, lášabqonnan allenni ekkak. (14) Həwit ak waxt jihel həwit, ya'ani bis sāl, bis sālu 'omər-de, činassər šettā, 'ād esrin šettā. (15) 'Ād lásāqit ak waxt; kamarni. (16) An agar gāw ā 'aqəl hawinā, əzgit, ya kədāb Mandāyí, mendi əmḥaşelnā. (17) Lezgini.

(18) Gāw šet eštā, ya ani šubbā šettā eštā qamātar, ya ani alpā u čemmā u čin u hamšā, yeah, ezgit ekkak yektā tum. (19) Ezgit Dezful, ezgit Šuštar, dokkāni ke həzitu awwál, (20) qəmšiyel, jisər ke həwā ekkak, gobrā ke gāw metelā Mandāyí abdonni, həwā pəseq ak waxt. (21) Eštā həziti jisər şun aye, u jisri tum əbadyon. (22) Ellinon biet Mandayānā, lekkā - geš dəronnu w-abdonnu, xiyabānhā abdonnu, street abdonnu, bənon bənon benāyāt - u mendi lekkā. (23) Ya ani, Mandayānā geš pəradyon, gațlonnu geš-du, šabqon əm-Šuštar u Dezful u Šawāli. (24) Eštā lek enši, bietwāt ham dəronnu. (25) Ā metelā Mušmanān orke Mandayānā. (1) I want to say something that I remember, [that happened] in 1943. (2) It was the Second World War, and everybody came, especially the Americans and the British, they came to Iran, because they wanted to help Russia. (3) They came along Khorramshahr Road, they sent (them) there.

(4) I had from the Provost Marshal (who knew me) a pass for all the American stores to which we would sell. (5) I was a silversmith at that time, I was. (6) At that time we went to Haft Tappe, and we used to go to Bandar-e Ma'sur, we used to go to Dezful (they had a 7-11 there), we used to go to sell to them. (7) I went from Solhabad (we were in Solhabad) and then we went to see Dezful.

(8) There was someone with me, Yabber, God bless him [lit. may he have relief], Yabber is dead, we went there and we saw the river. (9) We swam there, performed our ablutions, and prayed. (10) Afterwards, we put on our clothes, and we asked, "We see distant houses. (11) To whom do they belong, the ones that are outside of the city?" (12) They said, in Persian, "Those are the houses of unbelievers," meaning the houses of those who (he didn't know) don't have a religion, the Sabians' houses, the houses of unbelievers; they are the Ṣubba, the Sabians, that is, Mandaean houses.

(13) I saw houses which had a Mandaean temple by the river, but I didn't go up, we weren't allowed to go there.²¹⁶ (14) I was just a child then, I was, that is to say, (in Persian:) 20 years, that was my age, (in Mandaic:) 19 years, no - wait - 20 years. (15) No, I didn't go up then. We went back. (16) If I knew then what I know today (lit. I were in this mind), I would have gone, to pick up a Mandaic book or something. (17) We didn't go.

(18) In this year, that is to say, seven years ago, that is, 1995, I went there once again. (19) I went to Dezful, I went to Shushtar, the places which I saw before. (20) I asked about the bridge that was there, the men who, according to the stories, were Mandaeans who did it; it was broken then. (21) Now I saw that it had been repaired [lit. I saw "the bridge is good"], and another bridge had been built. (22) Wherever there were Mandaean houses, (now) there weren't. They were taken away and avenues and streets were built, building after building, and there was nothing left. (23) That is, the Mandaeans had all fled, or they were all killed; they left Shushtar and Dezful and Shāh Wāli. (24) Now there isn't anyone left, and even the houses have been taken away. (25) These are the stories of the Muslims with the Mandaeans.

216 Impersonal passive; see §5.4.3.2.

Text IV: The Bridge of Shushtar

(1) Hakyātā gāw Mandāyí, ratnā d-Hibel Ziwā, anhit' ad-Ādam.

(2) Gāw Šuštar ekkā šehyāni qəbāy jisri ābed. (3) Bənili əm-riš šettā tum riš šettā. (4) Bānen gāwi, geš gāw gəlālā u gač. (5) Tum tamām abdelli. (6) Qəbeyyen min şubāh ohrā allen əlāwwi. (7) Qəhazelli geš harub, geš gāw mienā qāzi. (8) Wayyeq qāti gāwwu! (9) Šehyānā u barnašānā wayyeq qāti gāwwu! (10) Láqyādu muyye mahrebli.

(11) Šadder əlāw wazirā w-olamā. (12) Aton qāri u qəmšielu muyye ekkā qəhāreb jisrā. (13) "Əbād šibiāhā," qāmren, "mahrebli." (14) Əmallu šehyānā, "Atton barnašānā ke šibihānā kabšellu!"

(15) Aton jami'a ke qəyādu gāw alm šibihānā. (16) Əmallonni əl-šehyānā, "Atton barnāšā u bonyi w-ani qarén əlāw šibihānā u man'a qābednannu." (17) Atti šehyānā barnāšā qəbānen hattā w-ahni olamā qāren subāh u qamietā wə-ʿaṣreyā. (18) So waxt qāren u benyān qəbānen. (19) Inšānā aton u qāren əlāw mienā u qəmṭāḥen əlāw pol, əlāw jisrā. (20) Darwišā aton min ģer welāt u qamyon qāren azimát qābden, šam qəmasriken u miṣrā qābden, u dawāzdáh māh qəbānen gāwwi. (21) Tum pardas.

(22) Qam šehyānā tabutā qābed qam ənāšā u qə-āklen. (23) Qəmbarken, "yā Elāhā! (24) Lā harub pol Šuštar." (25) Tum tamām tammā.

(26) So ruz qāllen min subāḥ. (27) Geš həreb! (28) Ālmā u barnašānā, olamā u darwišā wayyeq tam gāwwu. (29) Qamu həreb pol?

(30) *Hākem Šuštar wayyeq tam gāwwi.* (31) *Šadder əlāw daqqen həwarānā.* (32) *Aton qāri.* (33) *Əmarlu, "Mu qamretton əlāw tā pol?* (34) *Kol šettā qəbannanni u qəhāreb!"* (35) *Əmallonni, "Šadder əlāw rišāmmā əl-Mandayānā, u huwi qəmbašqer ā mendānā."*

(36) Šehyānā šadder əlāw rišāmmā. (37) Atā qāri rišāmmā u so tarmid' orki. (38) Aton qār šehyānā u subāh tāb hablonni. (1) A story in Mandaic, the idiom of Hibel Ziwā, which he sent down to Adam.

(2) In Shushtar, there is a ruler who wants to build a bridge. (3) He built it from the beginning of one year to the beginning of the next. (4) They build [it] solidly,²¹⁷ entirely out of stone and gypsum. (5) Then they finish it. (6) The next day, they want to walk to it. (7) They see that it is completely ruined, (and that) everything is going into the water. (8) They become distressed [lit. distress comes into them]! (9) The ruler and the people become distressed! (10) They don't know what is destroying it.

(11) He sent for the ministers and the scholars. (12) They came to him and he asks them, "What is destroying the bridge like this?" (13) They say, "It is the work of the planets²¹⁸ that is destroying it." (14) The ruler said to them, "Bring the people who will subdue the planets!"

(15) All who are versed in the knowledge of evil spirits came. (16) They said to the ruler, "Bring people, and [re-]build it, and we will enchant the evil spirits and ward them off." (17) The ruler brought people who [re-]built [it] anew,²¹⁹ and those scholars read morning, noon, and night. (18) They read three times, while the others are building. (19) Women came and enchanted the water, and cast [sc. incantations] to the bridge. (20) Dervishes came from outside the country and began to chant, make incantations, light candles, and draw [magic] boundaries, and for twelve months they build within it. (21) Then it was finished.

(22) The ruler proceeded to prepare a banquet for the people, and they eat. (23) They pray, "O God, (24) may the bridge of Shushtar not be ruined!" (25) Then it was done.

(26) Three days later, they go in the morning. (27) Everything is destroyed! (28) Everybody [lit. the world]—the people, the scholars, and the dervishes—became distressed [lit. distress became within them]! (29) Why was the bridge destroyed?

(30) The governor of Shushtar was distressed. (31) He sent for the elders [lit. white beards]. (32) They came to him. (33) He said to them, "What do you suggest for that bridge? (34) Every year we build it, and it is destroyed!" (35) They said to him, "Send for the *rišāmmā* of the Mandaeans,²²⁰ and he will know these things."

(36) The ruler sent for the $ris\bar{a}mm\bar{a}$. (37) The $ris\bar{a}mm\bar{a}$ came to him along with three *tarmidās*. (38) They came to the ruler and wished him a good morning.

- 217 The third masculine singular suffix is occasionally omitted, as noted in §4.1.2.1. The imperfective here appears without the indicative marker, but remains indicative in meaning; see §5.1.1.2.
- 218 The archons are the rulers of the material world, and identified with the seven visible planets.
- 219 As noted in fn. 217 above, the third masculine singular suffix is occasionally omitted.
- 220 The highest grade of priest in the Mandaean priesthood; see §1.2.2.

(39) Əmallu, "Qabin minkon čāre abdetton qamdin qam tā jisər. (40) Kol šettā qəbannanni u qəhāreb, u čāre qārak, yā rišāmmā?"

(41) Əmalli rišāmmā, "Bonyi hattā u an qatinā qamāy mahrebli."

(42) Qām šehyānā abdelli hattā. (43) Qəmatti barnāšā qəbānen hattā jisər. (44) Ehdā šettā bānen gāwi. (45) Qamāy əmpardas so ruz, rišāmmā atā, huwi u bəratti, alm ehlu qəhāzen diwā u šibihānā.

(46) *Əmall' əl-bəratti, "Anā agar gənit, āt guš əbud."* (47) *Bābā gənā, u qəhāzya bərattā.*

(48) Gubšā! (49) Qəhāzya hemke ibā atā qāl ehli. (50) Barābar bərattā bəgeş ibā.
(51) Mārelli, "Man ektak?"

(52) Əmalla, "Ana Čečāwā bar Čečāwā. (53) Ana ekkā husən lahzit!"

(54) Bāb ettar. (55) Nur asrek rišāmmā u qam qāri butā d-Šāmeš, "Kotp? əldiwā." (56) Əmall? əl-ganzibrā, "Ana daxil-dak!" (57) Ganzibrā əmall? əl-diwā, "Qamu aziyát qābdet? (58) Qəmahrebāt əl-pol! (59) Kol šettā banelli u qəmahrebāti! (60) Aziyát qābdet!"

(61) *Əmall' əl-rišāmmā, "Tum láqmahrem, u ana daxil-dəkon."*

(62) Əmalli ganzibrā, "Əbādi əbod qamdin!"

(63) Diw əmalli, "Mu qəbāt, ana qabed qamdak."

(64) Rišāmmā əmall' əl-diwā, "An qalin bietā, u āt doti qāre qam šehyānā."

(65) Hākimā d-Šuštar, huwi u barnašāni, sewār hisān tammon. (66) Aton guš qābden. (67) Qəhāzen pol láhəreb. (68) Jisrā d-Šuštar! (69) Kef əbadyon! (70) Barnašānā u wazirā mallonni, "M-ehnā tum hafté hazén!"

(71) Əmallu hākem, "Lājur láhribi ā-yumā, tum láqhāreb."

(72) Əmallu hākem, "Lābud kef ābden ənāšā!"

(73) Mallonni wazirā, "Šadder əlāw rišāmmā qāți qāraķ."

(74) Šadder əlāw rišāmmā āti qāri. (75) Atā rišāmmā qāri huwi u so tarmidā qār hākem. (39) He said to them, "I want you to fix that bridge for me. (40) Every year we build it and it is destroyed, so do you have a solution, *rišāmmā*?"

(41) The *rišāmmā* said to him, "Build it again and I will come before it is destroyed [lit. before he destroys it]."

(42) The ruler arose, they rebuild it, he brings some people to rebuild the bridge. (43) For one year they build $[it]^{221}$ strong. (44) Three days before it is finished, the *rišāmmā* and his daughter came. (45) They have the knowledge to see demons and evil spirits.

(46) He said to his daughter, "If I fall asleep, keep a look out." (47) The father fell asleep, and the daughter watches.

(48) Twilight! (49) She sees that something like a cloud has arrived, which has a voice. (50) The cloud stopped in front of the daughter. (51) She said to him, "Who are you?"

(52) He said to her, "I am Čečāwā, son of Čečāwā. (53) I have never seen such beauty!"

(54) The father awoke. (55) The *rišāmmā* lit a fire and began reading a prayer of Šāmeš,²²² "Demon-Trapper" [lit. "Seize the demon," the name of the prayer]. (56) He said to the *ganzibrā*, "I am at your mercy!" (57) The *ganzibrā* said to the demon, "Why are you causing such trouble? (58) You keep destroying the bridge. (59) Every year, they build it, and you destroy it! (60) You're doing [a lot of] damage!"

(61) He said to the *rišāmmā*, "I won't destroy (it)²²³ any longer, since I am at your mercy."

(62) The ganzibra said to him, "Do something for me."

(63) The demon said to him, "Whatever you want, I will do for you."

(64) The $ris\bar{a}mm\bar{a}$ said to the demon, "I am going home, so you come with me to the ruler."

(65) The governor of Shushtar, along with a few other people, rode on horseback [lit. became horse-riders]. (66) They came to take a look. (67) They see that the bridge had not been destroyed! (68) The Shushtar bridge! (69) They celebrated!

(70) The people and the ministers said to him, "Let's see a week from now!"

(71) The ruler told them, "Since he didn't destroy it today, he will no longer destroy it."

(72) The governor told them, "The people must celebrate!"

(73) The advisors told him, "Send for the *rišāmmā* to come before you."

(74) He sent for the *rišāmmā* to come before him. (75) The *rišāmmā* came to the governor along with three *tarmidās*.

221 Once again, the third masculine singular suffix has been omitted; see §4.1.2.1.

- 222 Šāmeš is the sun, one of the seven archons who govern the fates of mankind. He is traditionally identified with the god of the Jews (CM adunai).
- 223 Note again the omission of the third masculine singular suffix; see §4.1.2.1.

7. Text Collection

(76) Qamyon barābar-du, u hākem ləkt? əl-ganzibrā. (77) Ayhimi qāri, u əmall? əl-ganzibrā, "Mu qāmret? (78) Jisrā qəhāreb kandi?"

(79) Ganzibrā əmall' əl hākem, "Tum láqhāreb."

(80) Kef əbad šehyānā, geš-du kef əbadyon! (81) Čangā məhon, hiţrā əbadyon, inšā u gobrā, u šehyānā əmall' əl-ganzibrā, "Doţi qāre, āt u tarmidānā, əmḥaṯtén bienā-dan bel-ənāšā."

(82) Ezgon qāri so tarmidā u rišāmmā. (83) Hākem welāt əmallu, hākem əmall⁹ əl-ganzibrā, "Qabin minnak malpāte butā d-Šāmeš ke ləkt⁹ əl-diwā."

(84) Ganzibrā əmalli, "Āhā qam enši láqmallep, w-agar qəmallep, met. (85) Āhā rāzā d-kasyā."

(86) *Hākem wayyeq əbad gāw ganzibrā, "Lābud malpāte!"* (87) *Ganzibrā əlhākem əmalli, "Əm-ehnā tum so māh, anā xabár qahebnak."*

(88) Atā ganzibrā əl-bieti, huwi u so tarmidā. (89) Qamyon qəmhatten bienā-du.
(90) Mallonni tarmidānā əl-ganzibrā, "Šadder əlāw Čečāwā bar Čečāwā!" (91) Qam ganzibrā nur asreķ, buti eqrā, hāzér tammā Čečāwā bar Čečāwa qār ganzibrā. (92) Ganzibrā əmalli əl-Čečāwā, "Hākem welāt qəbāy' əmalpenanni²²⁴ rāzā qāre."

(93) Čečāwā əmalli, "Qamāy malpāti, ana gațelnākon geš!"

(94) Əmalli ganzibrā əl-Čečāwā, "Ya čāre həzi qamdan!"

(95) Čečāwā əmall' əl-ganzibrā, "Qəbāt alin gațeln' əl-hākem?"

(96) Ganzibrā əmalli, "Ani nešemti lágatlèn, (97) hatāyā qāran láqhāwi."

(98) Čečāwā əmalli əl-ganzibrā, "An qəmandin' əl-deštā, u ḥākem u jami'a rafieqā-di qədāhlen u qəmšadder əlāwwak."

(99) Čečāwā ezgā gāw Šuštar welāt. (100) Gənā tul deštā. (101) Deštā qāmat. (102) Qallā u qātyā u qənidā, u qāl ṣāyaḥ qəmāhyā deštā. (103) Ḥākem u ʿaskar pəradyon min welāt gāw deštātā.

(104) Hākem ler əkal, huwi u wazirā qār ganzibrā ezgon. (105) Əmalli, "Deštā āhā Šuštar qənidā genzā, w-ahl Šuštar geš pəradyon, inšā u gobrā. (106) Lābod əmbašqiret muyye āhā!"

(107) Ganzibrā əmalli əl-ḥākem, (108) "Id məti minne, w-anā qmabgeṣṣən əldeštā." (109) Ḥākem u wazirā́ mallonni əl-ganzibrā, "Láqbèn malpātan, šobq' əldeštā bāgṣā!" (76) They stood before them, and the governor took the *ganzibrā*. (77) He sat him beside him, and said to the *ganzibrā*, "What do you say? (78) Will he destroy the bridge any more?"

(79) The ganzibra said to the ruler, "He won't destroy it any longer."

(80) The ruler celebrated, all of them celebrated! (81) They played harps, everybody²²⁵ rejoiced, and the ruler said to the *ganzibrā*, "Come with me, along with the *tarmidās*, let's talk among ourselves, without anybody else."

(82) The three *tarmidās* and the *rišāmmā* went with him. (83) The governor of the city said to them, the governor told the *ganzibrā*, "I want you to teach me the prayer of Šāmeš that captured the demon."

(84) The *ganzibra* told him, "I will not teach this to anyone, for if I did teach it, I would die." (85) This is a private secret."

(86) The ruler threatened [lit. made distress for] the *ganzibrā*, "You must teach me." (87) The *ganzibrā* said to the ruler, "I will tell you in three months."

(88) The *ganzibrā* and the three *tarmidās* came to his house. (89) They started talking among themselves. (90) The *tarmidās* said to the *ganzibrā*, "Send for Čečāwā bar Čečāwā!" (91) The *ganzibrā* rose, kindled a fire, recited a prayer, and Čečāwā bar Čečāwā appeared before the *ganzibrā*. (92) The *ganzibrā* said to Čečāwā, "The ruler of the country wants us to teach him the secret I have."

(93) Čečāwā said to him, "Before you teach him, I will kill you all!"

(94) The ganzibra said to Čečāwā, "Find a solution for us!"

(95) Čečāwā said to the ganzibrā, "Do you want me to go kill the governor?"

(96) The *ganzibrā* said to him, "Let's not kill anyone [lit. a living being], (97) lest the sin be upon us."

(98) Čečāwā said to the *ganzibrā*, "I will cause the earth to quake, and the governor and all of his friends will become afraid, and he will send for you."

(99) Čečāwā went into the city of Shushtar. (100) He lay down beneath the ground. (101) The ground rose. (102) It goes back and forth, heaves, and makes a noise [lit. screamed a noise]. (103) The governor and the army fled from the city into the wilderness.

(104) The governor and the ministers turned around [lit. ate spittle] and went to the *ganzibrā*. (105) He said to him, "This land, Shushtar, is shaking very much, and the people of Shushtar—every last one of them [lit. women and men)—have all fled. (106) You surely know what is behind this (lit. what this is)."

(107) The *ganzibrā* told the ruler, "Lay off me, (108) and I will make the earth stop." (109) The governor and the ministers said to the *ganzibrā*, "We don't want you to teach us, just let the earth stop."

225 Lit. 'women and men.' This is a merism, a conventional phrase representing a totality by listing several of its more conspicuous parts.

7. Text Collection

(110) Ganzibrā əmalli əl-ḥākem, əmallu, "Atton ezgon, u ana 'aṣreyā qəmabgeṣna."

(111) Ezgon min ganzibrā šehyānā u wazirā . (112) Ezgon qamietā. (113) Məṭon qār ahlān-du u deštā qənidā. (114) Tum ʿaṣreyā ganzibrā nurā asrek, buti eqrā, atā qāri Čečāwā. (115) Deštā baġṣat əm-nōdā.

(116) Šehyānā u ənāšā kef əbadyon! (117) Umər hatti atā qamdi-du!

(118) Ganzibrā əmalli əl-Čečāwā, "Lājur ekkā mendānā quwwát ehlak, qabin minnak biete u barnašāne əmwadyātan gāw Əmšuni Kušţā." (119) Čečāwā əmalli əlganzibrā, "Ana qəmwadinkon, qəyāmet qəšaryāte."

(120) Ganzibrā əmalli əl-Čečāwā, "Anā qəšarinak bi-yəmutā, wa-laken āt yəmi qamdin, 'Qəmwadinkon Əmšuni Kuštā'."

(121) Yəmā Čečāwā eb-rāzā d-kasyā qam ganzibrā, "Qəmwadinkon Əmšuni Kušţā."

(122) Ganzibrā əmalli əl-Čečāwā, "Hemdā qəmwadyātan?"

(123) Əmallu Čečāwā, "Ārbin yumā yəhomyon bietwātkon, u 'ād m-ehnā tum ārbin yumā yəhomyon. (124) Lálletton ģer welāt, u ana qəmwadinkon, inšā, u gobrā, u torātā, u gāmeš, u jihelá, u əmrānkon, u gomlānkon, u zehunānkon - jami'a-dəkon!
(125) Ya dukki yəhomyon u anā qəhaper deštā qəmatinkon. (126) Min hənā tum ārbin yumā qəmatinkon."

(127) Qam Čečāwā qəhāper. (128) Ənāšā u qənayānā ārbin yumā əmaryon, "Mu qəmatilan əl-deštā?" (129) Ārbin yumā bandir tammon. (130) Qənayānā u nəjarānā u hədadānā ezgon qār ganzibrā.

(131) Qamellonni,²²⁶ "Mu qəmatil' əl-deštā? (132) Dahel zip qāheb u ani qabén safár mahén əlāw rezqā-dan."

(133) Ganzibrā əmallu, "Bandir bəğoşyon. (134) Ā Čečāwā zip láqāheb. (135) Ehli rāzā d-Šāmeš." (136) Qənayānā ezgon.

(137) Ba'ad čehél ruz bəğaşyon deštā u deštā lámətyat, u qənayānā ezgon min şubāh. (138) 'Aşreyā sā'át d-ahtā qəhāzen deštā nādat. (139) Qāl ahbat, u mətyat deštā barnašānā u inšānā u jihelā. (140) Ganzibrā əmallu, "Hem man yəhom dukki!"

(141) Deštā pehrat gāw bušme, halpāt əlāw Baġdād. (142) Ənāš Baġdād həzonna, u torātā, u gāmeš, u zehunān. (143) Ṣāyaḥ qəmāhen!

(144) Amţyu qār yəhem Əmšuni Kušţā, u qənayān tammon gāw safár. (145) Xabár məţā qāru, "Ahlān-dəkon u ganzibrā, deštā sāqat gāw ərqihā u ezgon!" (146) Pešimān tammon kammā bandir lábgeṣyon. (147) "Ganzibrā əmallan, 'Čečāwā zip láqāheb!'"

226 This appears to be a hybrid form. The personal morpheme is regularly found on the perfective stem of the verb, not the imperfective. (110) The *ganzibrā* told the governor, he said to them, "Leave, and I will make it stop this afternoon."

(111) The ruler and the ministers went from the *ganzibrā*. (112) They went at noon. (113) They arrived at their families' homes, and the earth continued to shake. (114) Then, in the afternoon, the *ganzibrā* kindled a fire, read a prayer, (and) Čečāwā came to him.

(115) The earth stopped quaking. (116) The ruler and the people celebrated! (117) A new age had dawned upon them!

(118) The *ganzibrā* told Čečāwā, "Since you have power (to do) such things, I would like you to take my house and people away to Əmšuni Kušțā (lit. the Sublimated of Truth)." (119) Čečāwā said to the *ganzibrā*, "I will take you, (if) you swear to release me."

(120) The *ganzibrā* told Čečāwā, "I will release you from the oath, but swear to me, 'I will take you to Əmšuni Kuštā.""

(121) Čečāwā swore to the *ganzibrā* upon the private secret, "I will take you to Əmšuni Kušţā."

(122) The ganzibrā said to Čečāwā, "When will you take us?"

(123) Čečāwā told them, "For forty days, remain in your homes, and stay put for the next forty days. (124) Do not go outside the city, and I will take you, the women and the men, the cows and the buffalos, and children, as well as your sheep, camels, and chickens - all of you! (125) Stay put in one place, and I will dig up the earth, and bring you. (126) Forty days from now, I will bring you."

(127) Čečāwā began to excavate. (128) For forty days, the people and the jewelers said for forty days, "What will bring the land for us?" (129) For forty days, they were confined [to their homes].

(130) The jewelers and the carpenters and the blacksmiths went to the *ganzibrā*. (131) They said to him, "What will bring the land? (132) I fear that he is lying, and we want to hit the road for the sake of our jobs."

(133) The *ganzibrā* said to them, "Stay confined (within your homes). (134) This Čečāwā doesn't lie. (135) He has the secret of Šāmeš." (136) The jewelers went.

(137) After they had stood on the [patch of] land for forty days and the earth hadn't risen, the jewelers left in the morning. (138) At one o'clock in the afternoon, they see the ground shaking. (139) There was a noise, and the [patch of] land rose, [as well as] the people, the women, and children! (140) The *ganzibrā* said to them, "Everyone, stay in place!"

(141) The [patch of] land flew through the air, passing over Baghdad. (142) The people of Baghdad saw it and the cows and buffalos and chickens. (143) They scream out loud!

(144) He brought them to settle in Əmšuni Kušṭā, while the jewelers were still on the road. (145) Word reached them that "your families, the *ganzibrā*, and the ground rose up into the sky and they went." (146) They regretted that they had not stayed put. (147) "The *ganzibrā* told us, 'Čečāwā doesn't lie!""

7. Text Collection

(148) Qənayānā u inšānā tammon ərasyon be-hdādā hattā, inšānā u jihelā-du ezgon orke ganzibrā. (149) Aton guš əbadyon deštā mətyat. (150) Mušmanānā aton guš əbadyon, be-hdādā qəmhatten, "Naşşat-e Şubba!"

Text V: The City of Mice who Eat Iron

(1) Qāmren ya tājéri həwā, həwāli perz pərāhi. (2) Qəbāyi safár ābed. (3) Həwāli emmā man hadid. (4) Bedqu amānátu əlāw ya biet dusti ehli u ezgā. (5) Bas tājér ezgā, amānatdār dərilu, zabnu geš, u pərā xarj-di əbad geš-di.

(6) Tājér yumi atā qam hadid-di.

(7) Gabər əmalli, "Hadidānak əm-duktā qarār biete āhebtu, (8) ekkak ya bəzuyi həwā obri qənāpeq menni. (9) Ekkak obri həwā ke waxti ke an atit hezinu əbād abdit geš-du aklu! (10) Obrā ākel geš hadidānā."

(11) Ya'ani, obri sad man hadid əkal!

(12) Tājér jawāb əhab, "Rāst qāmret, obrā genz qəraheml' əl-ḥadid, u šenāni genz hilānnon āklell' əl-ḥadid."

(13) Amānatdār ehter. (14) Əmar, "Ā tājér narm tammā," (15) āmer, "Rāst qamennā!" (16) Əmalli, "Ā pienā doti qār biete, mehman-de həwi!" (17) Əmalli, "Perš qatinā."

(18) Ezgā qəlāw tel, w-ebri əl-gabrā qəlakti orki ekkak u ahbi. (19) Peršā ezgon qār qumát. (20) Nedā əbadyon gāw welāt ke "Ebri ezgā mennan," ekkā ekkā. (21) Tājér əmar, "An həzit ya bāzi dərā ya ebri pəherye."

(22) Amānatdār sāyaḥ məhā. (23) Əmar, "Āt zip̄ qahbet! (24) Ā muḥālye! Mujur?!? (25) Mu qāmret, bāzi eqder' əlāw ṣṭāni u qəpāher?

(26) Tājér gəhe<u>k</u> u əmalli, "Gāw welāti k' ehli obrā sad mand hadid qə-ākel, mujur bāzi láqder' əl-ebri ke dah mand ehli?

(27) Amānatdār ģor əbad. (28) Əmalli, "Obrā lá-aklu. (29) Att? əl-ebrā, qahebnak hadidānak." (148) The jewelers and some women became married again to one another, (their) women and children having gone with the *ganzibrā*. (149) They came to see that the [patch of] land had risen. (150) The Muslims [also] came to look, and they said to one another, "[this is] the Pit of the Sabians."

Text V: The City of Mice who Eat Iron

(1) It is said that there once was a merchant, who had [only] a little cash. (2) He wanted to go on a trip. (3) He had 100 maunds of iron. (4) He placed them as a deposit [for a loan] in a house of a friend of his and went. (5) As soon as the merchant left, the trustee took them, sold them all, and spent all of the money.

(6) One day, the merchant came back one day for his iron. (7) The man said, "In the place in my house where I put away your iron for safekeeping, (8) there was a hole, out of which comes a mouse. (9) There was a mouse there which, no sooner had I arrived, I saw that he had eaten everything I had done. (10) The mouse is eating all the iron."

(11) That is, a mouse ate 100 maunds of iron!

(12) The merchant responded, "You are correct – the mouse really loves iron, and his teeth are very strong so that they can eat the iron."

(13) The trustee was pleased. (14) He said to himself, "This merchant has gone soft in the head," (15) while saying, "I'm telling the truth." (16) He said to him, "This evening, come to my house, be my guest."

(17) He said to him, "I will come tomorrow." (18) He went up a hill and seized one of the man's sons with him there and put him away.

(19) The next day they went to the prefecture. (20) Throughout the city it was proclaimed²²⁷ that "a boy had gone from us," and so forth.

(21) The merchant said, "I saw a hawk who had seized a boy flying off."

(22) The trustee screamed. (23) He said, "You're lying! (24) This is absurd. How?!? (25) What are you saying, a hawk took a boy and flew off?

(26) The merchant laughed and said to him, "In a city that has a mouse that can eat 100 maunds of iron, how could a hawk not be able to take a boy who weighs only 10 maunds?"

(27) The trustee understood. (28) He said to him, "The mouse did not eat it. (29) Bring back the boy, I'll give you back your iron."

227 Impersonal passive; see §5.4.3.2.

Texts VI-X: Short Texts

VI (1) Ya yeki həwā marizi. (2) Yəhem, qəṭāleb əm-māre, "Obde ṣunʿa ine, u onne, u karse, u ide, u kərāye." (3) Həwā ya yeki pahlidi yəhemye, ṣāti. (4) Tum pardas. (5) Əmalli, "Māre! (6) Ewaz mu ragʿa-dak ābedlu geš-du, ābed yeki haṯti qamdi."

(1) There once was a man who was ill. (2) He sat down and he asked God [lit. my Lord], "Heal my eye, and my ear, and my belly, and my arm, and my leg." (3) There was someone who was sitting beside him, who listened to him. (4) Then he finished.
(5) He said to him, "My Lord! (6) Instead of patching up all those things for you, He could make somebody new for him!

VII (1) Ya qahwi həwat hašiš qəšiten. (2) Həwā ya yeki, yəhem, bedq' əl-kərāyi tuti.
(3) Kərāyi ser tammat. (4) Láqyādi pesqat kərāyi. (5) (Əmar:) "Waylak, yā ənāšā, kərāye wel tammat!"

(6) Həwā pahli-di do nafár yəhimen, ham qəšiten hašiš. (7) Tum şatonni. (8) Yeki minnu əmall' əl-horinā, əmalli, "Willā! Doti pardén! (9) Āti belwā-dan ābed, āmer, 'Gənabtonn' əl-kərāye!'"

(1) There was a coffee shop in which hashish was smoked.²²⁸ (2) There once was someone, who sat down and stuck his leg beneath him. (3) His leg went numb. (4) He didn't know his leg was cut off.²²⁹ (5) He said, "Oh my God, hey people, my leg has come off (lit. become detached)!"

(6) There were two sitting beside him who were also smoking hashish. (7) Then they heard him. (8) One of them said to the other, he said, "Yikes! Let's make a run for it! (9) He may come to make trouble for us, saying, "You stole my leg!"

VIII (1) Ya yeki həwā qāzi ḥammām. (2) Ezgā qəbāy ləbeš eqmāšāni. (3) Həzā sotórdi gənib. (4) Ezgā əmall' əl-bāb ḥammām. (5) Bāb ḥammām əhabli pərā.

(6) Hafté horettā atā, daš. (7) Hammām dərā, tum nəpaq. (8) Həz' əl-zobuni ezgi.
(9) Dər' əl-pərāhān əm-bāb ḥammām.

(10) Səbu'a klatāyā atā. (11) Əmalli, "Āt laqšabeqnā allet. (12) Láqhāwi ekkā, kol hafté ahebnak pərā." (13) Əmalli, "Babbā, āt mas'ul lektak. (14) Kol mā qāzi, qāzi."

228 Impersonal passive; see §5.4.3.2.

229 The verb is in the *G*-stem but behaves like a passive or middle verb.

 (15) Daš gāwwa, hawwar, tum nəpaq. (16) Həz' əl-qəmāšāni geš ezgeyyen, bas *hizām bədeqye.* (17) Qam lebš' əl-hizām w-atā qam bāb hammām. (18) Əmalli, "Qəmayminak, an atit ekkā gāw hammām bas gāw hā!"

(1) There once was a man who went to a hammam. (2) When he came out, he wanted to put on his clothes. (3) He saw that his razor had been stolen. (4) He went to talk to the owner of the hammam. (5) The owner gave him some money.

(6) He came a second week and went in. (7) He took a bath and then came out.(8) He saw that his undershirt was gone. (9) He took the money from the owner.

(10) The third week he came, [and the owner] said, (11) "I can't let you go [in]. (12) It can't be like this, with me giving you money every week." (13) He said to him, "Sir [lit. grandfather], you won't be responsible. (14) Whatever happens, happens."

(15) He went inside, he washed, then came out. (16) He saw that his clothes were all gone, and that only his belt was left. (17) He got up and put on the belt, and came to the owner. (18) He told him, "I swear to you, I came like this to the hammam, in this alone!"

IX (1) Həwā ya yeki genz zip qāheb. (2) Ya yumi şat yeki ke tum genz zip qāheb. (3) Əmar, "qabin hezin mu zipāni min mujurnā." (4) Tum ezgā bieţi. (5) Məh' əl-bābā. (6) Láhzi gāw bieţā. (7) Həz' əl-bəratti. (8) Bəratti šiyelti, "Mu qəbāt?"

(9) Əmalla, "Qabin bābek."

(10) Mārelli, "Bābe luhnāye."

(11) Əmalla, "Elli ezgā?"

(12) Mārelli, "Ərqi; etber minni wuşle, ezgā rag'a-da ābed."

(13) Zipānā šəbaq u pərad. (14) Əmar, "Bərat ekkā zip qāheb, mujur bābahye?"

(1) There once was a man who lied very much. (2) One day, he heard about someone who also lied very much. (3) He said, "I want to see what a liar he is, in comparison with me." (4) Then he went to his house. (5) He knocked on the door. (6) He didn't see him in the house. (7) He did, however, see his daughter. (8) His daughter asked him, "What do you want?"

(9) He said to her, "I want (to see) your father."

(10) She said to him, "My father is elsewhere."

(11) He said to her, "Where did he go?"

(12) She told him, "Heaven; a piece broke off of it, and he has gone to patch it up."

(13) The liar left and ran off. (14) He said, "If the daughter lies like that, then what's her father like?"

7. Text Collection

X (1) Ya yumi ezgā Joḥa ṭəlab əm-šəbābāni qazġāni tāpi gāwwi. (2) Baʿad so ruz kammar, bədaq karsi qazġān honini. (3) Šəbābāni həzonn' əl-qazġān u əkadi minjodi. (4) Mallonn' əl-Joḥa, "Muyye āhā?"

(5) Joha əmallu, "Āhā qazġān-kon. (6) Həwā qāre, tam qāri jihel. (7) Atiti qamdəkon."

(8) Ṣāḥeb bietā əmalli, "Muḥālye," dəri u aḥbi.

(9) Hafté horettā atā Joḥa u təlab minni ya qazġāni. (10) Hablonni ya tafšā rabbi.
(11) Dər' əl-rabbā. (12) Lákamri qamdi-du. (13) Ba'ad esbu'a, ezgā šəbābā qam Joḥa.

(14) Əmalli, "Qazġān-de elliye?"

(15) Əmalli, "Qazġān me<u>t</u>."

(16) Əmall' əl-Joha, "Man qabul qābed qazġān mujur qəmāyet?"

(17) Joḥa gəhek, əmalli, "Man qabul qābed qazġān tam qāri ṣṭānā, ʿād qabul láqābed mujur met?"

(1) One day, Joha went to ask his neighbors for a cooking pot. (2) Three days later, he came back, having put a small pot inside it. (3) His neighbors saw the pot, and he took it out of it. (4) He said to Joha, "What is this?"

(5) Joha told them, "This is your cooking pot. (6) While it was at my place, it gave birth to a child (lit. the child became at it). (7) I brought it for you."

(8) The owner of the house said to him, "That's absurd," took it and kept it.

(9) The second week Joha came and asked him for a cooking pot. (10) They gave him a large stew pot. (11) He took the large one. (12) He didn't return it to them. (13) A week later, the neighbor went to Joha. (14) He said to him, "Where is my cooking pot?"

(15) He said, "The cooking pot has died."

(16) He told Joha, "Who would believe that a cooking pot could die?"

(17) Joha laughed, [and] said to him, "Who would believe that a cauldron has given birth to a boy, but not believe that it has died?"

8. Lexicon

Organization of the Lexicon

The nouns in this lexicon are organized according to their lexical form (elicited in isolation), followed by contextual forms and inflected forms. Each entry is followed by its attestations in the texts; in some cases, the lexical form was not independently elicited, in which case it is not followed by these attestations. Verbs are organized according to their triradical or quadriradical root, followed by their inflected forms. In the interests of preserving space and presenting the data in as concise a manner as possible, a minimum of data has been provided about each form, in the following format:

entry		[IPA transcription] part of speech gloss; language of
		origin original form 'gloss;' attestations within the texts
	inflected form	[IPA transcription] inflection gloss; attestations

The gender of a given noun should be assumed to be masculine unless otherwise indicated. Likewise, it should be assumed that the forms given are lexical or isolated forms, unless it is specifically noted that they are contextual forms. For example, the noun $qan\bar{a}y\bar{a}$ 'smith' appears in the lexicon as:

qanāyā	[qa'nɔːjɔ] n. smith; CM qanaia id.; 111.5
qənayānā	[qəna ['] jɔ:nɔ] <i>pl.</i> smiths; IV.144, IV.136, IV.128, IV.130, IV.137,
	IV.148

The entry indicates that the lexical form, $qan\bar{a}y\bar{a}$, pronounced [qa'nɔ:jɔ], is a noun meaning 'smith,' derives from Classical Mandaic **qanai**, which also means 'smith,' and occurs once in Text III at line 5. Note that the classical forms are transcribed according to the system developed by Drower and Macuch for their *A Mandaic Dictionary*. The plural of this word, *qənayānā*, is pronounced [qəna'jɔ:nɔ], means 'smiths,' and occurs six times in Text IV.

8. Lexicon

Phrasal verbs are organized according to the non-verbal element (as the meaning of the verb is primarily derived from the non-verbal element). The phrasal verb $kef \partial bad \sim \partial bod (\bar{a}bed)$ 'to celebrate, have fun' appears in the lexicon as follows:

kef '-b-d	<i>v.i.</i> to celebrate, have fun; Arabic <i>kayf</i> 'good humor;' cf.
	Persian kayf kardan 'to enjoy oneself, have fun'
kef ə <u>b</u> adyon	['kef \mathfrak{d}_1 vadjon] <i>G perf.</i> 3^{rd} <i>pl.</i> they celebrated; IV.69, IV.72,
	IV.80 (x2)
kef ā <u>b</u> den	['kef pvdɛn] G subjv. 3^{rd} pl. they celebrate; IV.72

This entry indicates that it is an intransitive verb meaning 'to celebrate' and 'to have fun,' and that it is calqued upon Persian *kayf kardan*, which means the same thing. The inflected forms that appear within the text collection include *kef abadyon*, the third person plural of the perfective of the verb, in the *G*-stem, which means 'they celebrated.' This form occurs four times in Text IV. For phrasal verbs, secondary stress is always indicated; elsewhere it is only indicated if it is not otherwise predictable according to the normal rules governing stress (see 2.4.4.3).

Most of the loan words in Neo-Mandaic have Arabic origins, either indirectly borrowed via Persian or directly borrowed from the colloquial dialect of the city of Khorramshahr, which closely resembles the *gilit* dialects of southern Iraq, in particular the dialects around Basra.²³⁰ Nevertheless, whenever reference is made to Arabic, it should be assumed that the form cited reflects Modern Standard Arabic rather than the colloquial dialect, unless otherwise noted. The citation form for verbs borrowed from Arabic and other Semitic languages is the triradical root. The citation form for verbs borrowed from the Iranian languages (including Middle Persian, Classical Persian, and modern Persian) is the infinitive (e.g. *kardan* 'to do').

The corpus of written material in Neo-Mandaic currently available to scholarship is extremely limited, comprising only the texts collected by Rudolf Macuch and Jaques de Morgan in addition to the present collection. Nearly all of this material has been published in transcription, and only five texts—those collected by de Morgan—

²³⁰ Kees Verteegh, *The Arabic Language* (Edinburgh: Edinburgh University Press, 2001), 158.
have ever been published in their original script. For this reason, and because of the fact that knowledge of the Mandaic script is far from commonplace, the entries in the lexicon are organized not according to the order of the original script (as in Macuch's publications), but according to the following order:

[^] ` A B B C Č D D E Ə F G Ġ Ġ H Ḥ I J K K L M N O P P Q R S Ṣ Š T Ţ Ț U W X Y Z

)

'- <u>b</u> -d		v.t. to do, to make; CM ABD; see also azimát ² -b-d,
-		aziyát '-b-d, belwā '-b-d, čāre '-b-d, əbād '-b-d, fekər '-b-
		d, ġor '-b-d, hiţrā '-b-d, kef '-b-d, komák '-b-d, man 'a '-b-
		d, mișrā '-b-d, nedā '-b-d, qābin '-b-d, qabúl '-b-d, rag'a
		·b-d, safár ·b-d, šenā ·b-d, tamām ·b-d, tabutā ·b-d,
		wayyeq '-b-d, xarj '-b-d
	abd	[avd] G perf. 3^{rd} m.sg. (contextual) he did; I.11
	abdat	['avdat] $G perf. 3^{rd} f.sg.$ she did; 1.6
	əbadyon	[a'vadi] G perf. 3rd pl. they made; III.21
	abdon	[av'don] G perf. 3^{rd} pl. (contextual) they did; I.5
	-	$[av^{l}doni] G perf. 3^{rd} pl. (contextual) and y did, 1.5[av^{l}doni] G perf. 3^{rd} pl. (contextual) + 3^{rd} m.sg. they did$
	a <u>b</u> donni	
	1.1	it; III.20 $[-1] C = C 2^{rd} + 2^{rd} + 4$
	a <u>b</u> donnu	[av'donnu] <i>G perf.</i> $3^{rd} pl. + 3^{rd} pl.$ they made them; III.22,
		III.22 (x^2)
	əbadni	$[\varepsilon^{\text{vadni}}] G perf. I^{st} pl.$ we did; III.9
	ābed	['o:ved] G subjv. 3^{rd} m. sg. he does; IV.2, VI.6
	qabed	[qa'ved] G imperf. 1^{st} sg. (contextual) I will do; IV.63
	ābden	['ovdɛn] G subjv. $3^{rd} pl$. they do; II.6
	a <u>b</u> delli	[æv ['] dɛlli] <i>G</i> imperf. 3^{rd} pl. + 3^{rd} m. sg. they do it; IV.5,
		IV.42
	obde	['ovde] G impv. sg. + 1^{st} sg. make me; VI.2
^{>} -b- ^w / _y		See $b - \frac{w}{y} - \frac{w}{y}$.
[,] -h-b		v.t. to give; CM AHB id.; see also guțluktā '-h-b, jawāb
-		'-h-b, qāl '-h-b, qarār '-h-b, şubāḥ ṭāb '-h-b, xabár '-h-b,
		zip ⁻ -h-b
	əhabli	['hauli] G perf. 3^{rd} m.sg. + 3^{rd} m.sg. he gave him; VIII.5
	əhabyon	$[\exists$ 'havjon] G perf. 3^{rd} pl. they gave
	hablonni	[hav'lonni] <i>G perf.</i> 3^{rd} <i>pl.</i> + 3^{rd} <i>m.sg.</i> they gave him; X.10
	əhabni	[a'havni] G perf. 1^{st} pl. we gave; 1.14
	qahebnak	$[qa'hevna\chi]$ <i>G imperf.</i> 1 st sg. + 2 nd m.sg. I will give you;
	quilogna <u>k</u>	V.29
	ahe <u>b</u> na <u>k</u>	[a'hevna χ] <i>G imperf.</i> 1^{st} sg. + 2^{nd} m.sg. I give you; VIII.10
'- <u>k</u> -d		<i>v.t.</i> to take, seize; CM AKD id.
-	ə <u>k</u> adi	[' χ a:di] <i>G perf.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>m.sg.</i> he took it; X.3

'- <u>k</u> -1		<i>v.t.</i> to eat; CM AKL id.
	əkal	$[a'\chi a]$ G perf. 3^{rd} m.sg. he ate; V.11
	aklu	['axlu] G perf. 3^{rd} m.sg. + 3^{rd} pl. he ate them; V.9
	lá-a <u>k</u> lu	['la?a χ lu] G neg. perf. 3^{rd} m.sg. + 3^{rd} pl. he didn't eat
		them; V.28
	qākel	['qp:xɛl] G imperf. 3 rd m.sg. he eats; V.26
	ākel	[' $p:\chi el$] G subjv. 3^{rd} m.sg. he eats; V.10
	- qə-āklen	$[q \exists 2^{\prime} p \chi l \epsilon n]$ G imperf. $3^{\prime \prime d} p l$. they eat; IV.22
	aklelli	$[a\chi' lelli]$ G subjv. $3^{rd} pl. + 3^{rd} m.sg.$ they eat it; 2.340
	aklell	[a χ 'lelli] <i>G</i> subjv. 3^{rd} pl. + 3^{rd} m.sg. they eat it; 2.340 ['a χ lel] <i>G</i> subjv. 3^{rd} pl. + 3^{rd} m.sg. (contextual) they eat
	-	it; V.12
[,] -1-1		<i>v.i.</i> to go (only imperf.); CM ALL id.; see also $^{2}-z-g^{-w}/_{y}$
		and ohra '-z-g-"/y
	qallā	['qallo] G imperf. 3 rd f.sg. she goes; IV.102
	qallət	['qallet] G imperf. 2 nd sg. you go; VIII.10
	qalin	[qa'lin] G imperf. 1 st sg. (contextual) I am going; IV.64
	alin	[a'lin] G subjv. 1 st sg. I go; IV.95
	allen	['allen] G subjv. 3 rd m.sg. they go; IV.6
	lalletton	['lalletton] <i>G neg. subjv.</i> 2 nd pl. do not go; IV.124
	qallenni	[qa'lɛnni] <i>G imperf. 1st pl.</i> we go; III.7
	qallén	[qa'lɛn] <i>G</i> imperf. 1^{st} pl. (contextual) we go; III.6 (x3)
	allenni	[a'lɛnni] G subjv. $1^{st} pl$. we go; III.13
'- <i>l-p</i> ¯		v.i. G to learn; D to teach; CM YLP id. see also yalupā
	qəmallep	[qəˈmallɛf] D subjv. 3 rd m. sg. (if) I were to teach; IV.84
	láqmallep	['laq ₁ mallef] <i>D neg. imperf.</i> 3 rd m.sg. (contextual) I will not teach; IV.84
	malpāti	[mæl ¹ fɔ:ti] D subjv. 2^{nd} sg. + 3^{rd} m.sg. you could teach
	maipau	him; IV.93
	malpāte	[mæl ¹ fɔ:te] D subjv. 2^{nd} sg. + 1^{st} sg. you teach me; IV.83,
	maipute	IV.86
	malpātan	[mæl ¹ fɔtæn] D subjv. 2^{nd} sg. + 1^{st} pl. you teach us; IV.109
	əmalpenanni	[emælfɛ'nɑnni] D subjv. 1^{st} $pl. + 3^{rd}$ m.sg. we will teach
	omuponumi	him; IV.92
[,] -m-r		v.t. to tell, say; CM AMR id.; Tab. 1.5; see also farwah
		² -m-r, rāst ² -m-r
	əmar	[əˈmɑ] G perf. 3 rd m.sg. he said; V.14, V.21, V.23, IX.3, VII.5,
		IX.14
	əmall'	$[\mathfrak{d}^{Imal}]$ <i>G perf.</i> 3^{rd} <i>m. sg.</i> $+ 3^{rd}$ <i>m.sg.</i> (contextual) he said
		to s.o.; IV.41, IV.46, IV.56, IV.57, IV.61, IV.62, IV.63, IV.64, IV.77,

		IV.78, IV.79, IV.81, IV.83, IV.84, IV.93, IV.94, IV.95, IV.96, IV.105,
		IV.107, IV.110, IV.119, IV.120, IV.122, V.7, V.16, V.17, V.26, V.28,
		VII.8 (x2), VIII.4, VIII.10, VIII.16, X.4, X.14, X.16, X.17
	əmalli	$[\exists$ 'malli] G perf. 3^{rd} m. sg. $+ 3^{rd}$ m.sg. he said to him;
		IV.87, IV.92, IV.98, IV.118, VI.5, VIII.11, X.8, X.15
	əmalla	$[\exists$ 'malla] <i>G perf.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>f.sg.</i> he said to her; IV.52,
		IX.9, IX.11
	əmallu	$[\exists$ 'mallu] <i>G perf.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>pl.</i> he said to them; IV.14,
		IV.33, IV.39, IV.71, IV.72, IV.83, IV.110, IV.123, IV.133, IV.140, X.5
	əmallan	$[\exists$ 'mallæn] G perf. 3^{rd} m. sg. + 1^{st} pl. he said to us; IV.147
	amrat	['amrat] G perf. 3 rd f.sg. she said; II.21, II.26
	mārelli	[mɔ'rɛlli] G perf. 3^{rd} f.sg. + 3^{rd} m.sg. she said to him;
		IV.51, IX.10, IX.12
	əmaryon	[ə ['] maijon] G perf. 3 rd pl. they said; III.12, IV.128
	mallonni	[ə'maɪjon] <i>G perf.</i> 3^{rd} <i>pl.</i> they said; III.12, IV.128 [ma'lonni] <i>G perf.</i> 3^{rd} <i>pl.</i> $+ 3^{rd}$ <i>m.sg.</i> they said to him,
		IV.16, IV.35, IV.70, IV.73, IV.90, IV.109
	əmarlonnan	[əma. ^l lonnæn] G perf. 3^{rd} pl. + 1 pl. they told us; 1.5
	āmer	['o:meI] G subjv. 3^{rd} m.sg. he said; V.15, VII.9
	qāmrət	['qpmret] G imperf. 2 nd sg. you say; IV.77, V.25
	əmer	$[\exists' m \in I]$ G subjv. I^{st} sg. (contextual) I say; I.3, I.5, I.6, II.1,
		III.1
	qāmren	['qpmren] G imperf. 3 rd pl. they say; IV.13, V.1
	qamrelli	[gam'relli] G imperf. 3^{rd} pl. + 3^{rd} m.sg. they call him; II.11
	qamellonni	[qam'rɛlli] G imperf. 3^{rd} $pl. + 3^{rd}$ $m.sg.$ they call him; II.11 [,qamɛ'lonni] G imperf. 3^{rd} $pl. + 3^{rd}$ $m.sg.$ they said to
	1	him; IV.131
	qamretton	[qam ['] retton] G imperf. 2 nd pl. you say; IV.33
	1	
'-n-d-š		v.i. Q to think, consider, be worried; Middle Persian
		<i>handēšīdan</i> id. via CM ANDŠ id.
	andešyon	[an'dɛʃjon] Q perf. 3^{rd} pl. they were worried; I.14
	mandeš	['mandeʃ] Qact. ptc. worried
<i>'-r-h</i>		<i>v.t.</i> to smell; CM C of RHA id.
' - r-q		<i>v.t.</i> D to empty; to spit; CM ARQ 2 to empty, to pour out
'- <i>r</i> -s		<i>v.t.</i> to marry; CM ARS id.
	ərasyon	$[a^{t} asjon]$ <i>G perf.</i> 3^{rd} <i>m. sg.</i> they married; IV.148
°-t−r		<i>v.i.</i> G to be awake; CM ATR id. secondary root from the
		<i>tC</i> -stem of ^{cw} / _y -r, cf. Syriac ناله ' <i>ettə 'ir</i> 'to wake up.'
	ettar	['stta1] tG perf. 3^{rd} m.sg. he awoke; IV.54

$^{2}-t-^{W}/_{V}$		v.i. G to come, D and C to bring; CM ATA id.; see also
- ,		wayyeq $\frac{-t}{v}$
	atā	$[a'\theta \sigma:] G perf. 3^{rd} m.sg.$ he came; IV.37, IV.44, IV.49, IV.74,
		IV.75, IV.88, IV.114, IV.117, V.6, VIII.6, VIII.9, VIII.15, X.9
	a <u>t</u> at	$[a^{l}\theta at] G perf. 3^{rd} f. sg.$ she came; II.27
	a <u>t</u> it	$[\mathfrak{a}^{\dagger} \theta i \theta]$ <i>G perf.</i> 1^{st} sg. I came; II.17, V.9, VIII.16
	a <u>t</u> on	$[a^{\dagger}\theta on] G perf. 3^{rd} pl.$ they came; II.5, II.10, II.29, III.2, III.2,
		III.3, IV.12, IV.15, IV.19, IV.20, IV.32, IV.38, IV.66, IV.149, IV.150
	atiton	$[a'\theta:ton]$ G perf. 2^{nd} pl. you came; I.3, I.4
	qāti	['qɔ:θi] G imperf. 3^{rd} m.sg. he comes; IV.8; IV.73
	āti	['ɔ:θi] <i>G subjv.</i> 3 rd m.sg. he comes; VII.9
	<i>qā<u>t</u>yā</i>	['qp: θ jo] <i>G</i> imperf. 3^{rd} f.sg. she comes; IV.102
	qatinā	[qa' θ i:no] G imperf. 1 st sg. I will come; IV.41, V.17
	qāten	[qo: θ en] <i>G</i> imperf. 3^{rd} pl. they come; II.23
	āten	[' \mathfrak{den}] <i>G</i> imperf. \mathfrak{Z}^{rd} <i>pl</i> . they come; II.29
	doti	['do:θi] G impv. m. sg. come; IV.64, IV.81, V.16, VII.8
	doton	[do'θon] <i>G impv. pl.</i> come; I.5
	att'	$[a\theta] C impv. sg.$ (contextual) bring; V.29
	atiti	$[\mathfrak{a}^{l}\mathfrak{h}\mathfrak{i}\mathfrak{t}\mathfrak{i}]$ <i>C perf.</i> 1^{st} sg. + 3^{rd} m.sg. I brought it; X.7
	atti	$[^{l}a\theta\thetai]$ C perf. 3^{rd} m. sg. he brought; IV.17
	atton	['aθθon] <i>C impv. m. pl.</i> bring; IV.14, IV.16
	qəma <u>tti</u>	$[q \exists ma \theta \theta i] C imperf. 3^{rd} m. sg. he brought; IV.42$
	qəma <u>t</u> ila	[qmæ' θ i:la] <i>C imperf.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>f.sg.</i> he brings her;
		IV.131
	qəma <u>t</u> ilan	[qmæ ^l θ i:læn] <i>C imperf.</i> 3^{rd} <i>m.sg.</i> + 1^{st} <i>pl.</i> he will bring for
	annatinkan	us; IV.128 [qmæˈ θ inxon] <i>C</i> imperf. 1 st sg. + 2 nd pl. I will bring you;
	qəmatinkon	
		IV.125, IV.126
' -z-g- [₩] /	/	v.i. to go; CM SGA id.; the modern form is derived from
-z-g- /	У	the <i>C</i> -stem rather than the <i>G</i> -stem, resulting in the
		assimilation of the first radical to the second and the
		initial vowel. See also '-/-/, which complements this root
		in the imperfective.
	ezgā	[$\epsilon z'g_{3}$:] $Q perf. 3^{rd} m.sg.$ he went; IV.99, V.4, V.5, V.18, V.20,
	0250	VIII.2, VIII.4, IX.4, IX.11, IX.12, X.1, X.13
	ezgat	[ez'gat] Q perf. 3rd f. sg. she went; IV.144
	ezgit	$[ez[gat] \ Q \ perf. \ 1^{st} sg. \ I \ went; III.7, III.16, III.18, III.19 (x2)$
	ezgon	$[z^{2}gon] Q perf. 3^{rd} pl.$ they went; IV.82, IV.104, IV.110,
		IV.111, IV.112, IV.130, IV.136, IV.137, IV.138, IV.145, IV.148, V.19
	ezgini	$[\exists z'g:ni] Q perf. 1^{st} pl.$ we went; III.6, III.8
	lezgini	[lz_gini] Q neg. perf. 1^{st} pl. we didn't go; III.17

	qāzi qāzen ezgi ezgeyyen	['qp:zi] Q imperf. 3^{rd} m.sg. he goes; IV.7, VIII.1, VIII.12 (x2) ['qp:zen] Q imperf. 3^{rd} pl. they go; IV.26 [$\epsilon z'$ gi:] Q pass ptc. gone; VIII.7 [$\epsilon z'$ gɛijɛn] Q pass ptc. + 3^{rd} pl. they were gone; VIII.14
		c
Abdalla		[Sab'dałłə] <i>prop. n.</i> Abdallah, the <i>tarmidā</i> from Ahvāz who accompanied Lady Drower to Khorramshahr ca. 1932; II.8, II.27
ʿād		[fod] <i>adv.</i> yet, then; colloquial Arabic <i>ʿād</i> 'then;' III.13, III.14, III.15, X.17
<i>ʿajāj</i>		[ʕaˈʤɒʤ] n. fog; Arabic ʿajāj ʿdust storm'
^c amme	bərat ʿamme ebbər ʿamme	['ʕamme] n . (paternal) aunt; Arabic <i>ʿamma</i> id. [bə,rat 'ʕamme] n . cousin [ˌɛbbr 'ʕamme] n . cousin
^c ammu	bərat ʿammu ebbər ʿammu	['ʕammu] n. (paternal) uncle; Arabic <i>ʿamm</i> id. [bə _i rɑt 'ʕammu] n. cousin [ˌɛbbr 'ʕammu] n. cousin
'aqəl		['Saql] n. mind; Arabic 'aql 'mind, intellect;' III.16
<i>ʿaskar</i>		['Suskaı] <i>n.</i> army; Arabic <i>'askar</i> id.; IV.103
ʿaṣreyā		[ʕasˤrɛɪjɔ:] <i>adv.</i> in the afternoon; Arabic <i>ʿaṣr</i> 'afternoon;' IV.17, IV.110, IV.114, IV.138
'Azāzí		[ʕazɔ'zi:] <i>prop. n.</i> The ʿAzāzi, a Mandaean family originally from Sūq esh-Shuyūkh that settled in Khorramshahr; II.23
<i>'umər</i>	ʿumər-de	[' Ω ,mr] <i>n</i> . age; Arabic <i>'umr</i> 'number; age;' IV.117 [' Ω ,mrde] + 1 st sg. my age; III.14

300

Α

ā		See <i>āhā</i>
ad		[ad] <i>prep.</i> until, as far as CM ad id.; IV.1
Ādam		['b:dæm] prop. n. Adam; CM adam id.; 1.7, 1.8, 1.9, IV.1
agar		['agæı] <i>conj.</i> if; Persian <i>ágar</i> 'if;' III.16, IV.46, IV.84
āhā		['o:ho] <i>dem. pron.</i> this; CM aha id.; IV.84, IV.85, IV.106, X.4, X.5
	ā	[9 :] this; I.5 (x2), I.6, I.13, II.24, II.26, III.16, III.25, IV.33, IV.35,
	hā	IV.40, IV.128, IV.134, V.14, V.16, V.24 [hɔ:] this; VIII.16
ahhā	ah bellā ah e <u>tt</u> ā	[ˈæhhɔ] n. brother; CM aha id. [ˌæh ˈbɛllɔ] n. brother-in-law [ˌæh ˈεθθɔ] n. brother-in-law
ahl	ahlān-dəkon ahlān-du	[æl] <i>n</i> . people; Arabic <i>ahl</i> 'people;' IV.105 (x2) [æh'lɒndə _i χon] $pl. + 2^{nd} pl.$ your families; IV.145 [æh'lɒndu] $pl. + 3^{rd} pl.$ their families; IV.113
ahni		[æh ^l niː] <i>dem. pron.</i> those/these; they; CM hania 'those;' I.13, III.11, III.12, IV.17
ahtā		['æhtə] quant. one (f.); CM ahta id.; IV.138
<u>Aḥwāz</u>		[aħ ^l wdz] <i>prop. n.</i> Ahvāz, a city of SW Iran, north- northeast of Basr, Iraq. Ahvāz is the main center of Mandaean life in Iran today; II.5, II.8, II.10, II.28
a <u>k</u> u	a <u>k</u>	[αχu] <i>dem. pron.</i> that; CM hak id., cf. Iraqi Arabic <i>aku</i> id. id. [αχ] that (contextual); II.2, II.21, III.5, III.6, III.7, III.14, III.15, III.20
alm		[alm] <i>n</i> . knowledge; cf. Arabic <i>'ilm</i> id.; IV.15, IV.45

302		8. Lexicon
ālmā	almānā	['plmɔ] <i>n</i> . world; epoch of time; people, beings; CM alma id.; 1.4, 1.8, 1.14, IV.28 [al'mɔ:nɔ] <i>pl</i> . worlds, epochs of time; 1.11
alpā		['alfo] quant. thousand; CM alpa id.; III.1
amānát		[ampˈnɑt] n. deposit; Arabic <i>amāna</i> id. via Persian <i>amānat</i> id.; V.4
amānato	lār	[amp ₁ nɑt ¹ dɒJ] <i>n</i> . trustee; Persian <i>amānatdār</i> id.; V.5, V.13, V.27, V.22
ammā		[ˈæmmə] <i>conj.</i> but; colloquial Arabic <i>amma</i> id.; I.14, II.9, II.11
amrikí		[amriˈkiː] <i>adj.</i> American; Arabic <i>amrīkī</i> id. via Persian <i>amriki</i> id.
	amrikān	[amri ¹ kon] <i>pl</i> . Americans; III.2, III.4
anā		['æ:nɔ] <i>pers. pron.</i> I; CM ana; I.4, IV.46, IV.87, IV.108, IV.120, IV.125
	ana	[¹ æ:na] <i>pers. pron.</i> I; IV.53 (x2), IV.56, IV.61, IV.63, IV.93, IV.110, IV.119, IV.124
	an	[æn] <i>pers. pron.</i> (contextual) I; II.16, II.25, III.4, III.8, III.13, III.16, IV.41, IV.64, IV.98, V.9, V.21, VIII.16
ani	an	[¹ æ:ni] <i>pers. pron.</i> we; CM anin id.; I.8, I.11, I.12, I.13, I.14, I.14, II.13, IV.16, IV.96, IV.132 [æn] (contextual) we; II.22
ARĀM		[a'rɔm] <i>prop. n.</i> ARAM Society for Syro-Mesopotamian Studies; 1.5, 1.6
arbā		['a.bɔ] <i>quant.</i> four; CM arba id.
ārbin		[ɔ.ɪ'bin] <i>quant.</i> forty; CM arbin id; III.1, IV.123, IV.123, IV.126, IV.128, IV.129
asu <u>t</u> ā		[a'su:θo] <i>n</i> . healthiness; CM asuta id.; 1.2

asuṯā h-w- ^w /y l- asutā nehwil <u>k</u> on	<i>v.i.</i> to be welcome, lit. to have health $[a'su:\theta \circ neh_ivil\chi \circ n] G impf. 3^{rd} m.sg. + 2^{nd} pl.$ you are welcome; lit. may health be for you; 1.2
āt	[ot] pers. pron. you (sg.); CM anat id.; V.23, VIII.9, VIII.11
atton	['ætton] pers. pron. you (pl.); CM anatun id.; IV.110
awwál	[av ¹ væl] <i>adj.</i> first; Arabic <i>awwal</i> id. via Persian <i>avval</i> id.; I.8, I.11, III.19
āxír	[b'χiɪ] <i>adj.</i> last, final; Arabic <i>āxir</i> id. via Persian <i>āxir</i> id.; I.14
azimát '- <u>b</u> -d	<i>v.i.</i> to make an incantation; Arabic <i>'azīma</i> 'incantation' via Persian <i>azimat</i> id.
azimát qābden	[azi'ma θ , qpvdɛn] <i>G</i> imperf. 3^{rd} pl. they invoked; IV.20
aziyát '-b-d	<i>v.i.</i> to do wrong; Arabic <i>adīya</i> 'harm, injury' via Persian <i>aziyat</i> id.
aziyát qā <u>b</u> det	[azi'jat 'qpvdɛt] <i>G imperf.</i> 2 nd m.sg. you have done wrong; IV.57, IV.60

В

<i>b</i> -		[εb] <i>prep.</i> in, with, by; CM b id.; I.1, IV.81, IV.121, IV.148, IV.150
	be-l-	[bel] <i>prep.</i> without; IV.81
b-d-q		v.t. to put, place; CM BDQ id.
	bədaq	[bə'daq] G perf. 3^{rd} m.sg. (pausal) he put; X.2
	bədeqye	[bə'dɛqje] G pass. ptc. + 3^{rd} m.sg. cop. it was put; VIII.14
	bədq'	[betq] G perf. 3^{rd} m.sg. (contextual) he put; VII.2
	bədqu	['petqu] G perf. 3^{rd} m.sg. + 3^{rd} pl. he put them; V.4
b- <u>g</u> -ș		v.i. G to stand, stay; cease; C to stop (trans.); CM BGS
		id. See also $bandi(r) b - \bar{g} - \bar{s}$
	bəges	[bəˈ $\kappa \epsilon s^{s}$] G perf. 3^{rd} m.sg. he stayed; IV.50
	bagṣat	['bars ^s at] G perf. $3^{rd} f$. sg. she stayed; IV.115
	bəgasyon	[bəˈʁɑsjon] <i>G perf.</i> 3 rd pl. they stayed; IV.137 ['bɑʁs [°] ɒ] <i>G imperf.</i> 3 rd f.sg. she ceases; IV.109
	bāgṣā	['bass'p] G imperf. 3 ^{ra} f.sg. she ceases; IV.109
	qəmabgesna	[qmab ¹ κ 3sna] <i>C</i> imperf. 1 st sg. + 3 rd f.sg. I will stop her;
		IV.110
	qəmabğeşşən	[qəmabˈʁɜss [°] n] <i>C imperf. 1st sg.</i> (contextual) I will stop; IV.108
$b-n-W/_{y}$		<i>v.t.</i> to build; CM BNA id.; see also <i>benyān b-n-</i> ^{$w/_y$}
·y	bənili	[bɛ'ni:li] G perf. 3^{rd} m. sg. + 3^{rd} m. sg. he built it; IV.3
	bənon	[bə'non] G perf. 3^{rd} pl. they built; III.22
	qəbānen	[qə'bənɛn] G imperf. 3^{rd} pl. they build; IV.17, IV.20, IV.42
	bānen	['bo:nen] G imperf. 3^{rd} pl. they build; IV.4, IV.42, IV.43
	banelli	['bɔ:nɛn] <i>G imperf.</i> 3^{rd} <i>pl.</i> they build; IV.4, IV.42, IV.43 [ba'nɛlli] <i>G imperf.</i> 3^{rd} <i>pl.</i> + 3^{rd} <i>m.sg.</i> they build it; IV.59
	qəbannanni	[qəbæn'nanni] G imperf. $1^{st} pl. + 3^{ra} m. sg.$ we build it;
	honvi	IV.34, IV.40 ['bonji] G imp. m.sg. + 3^{rd} m.sg. build it; IV.16, IV.41
	bonyi	$\begin{bmatrix} \text{bonj1} \end{bmatrix} G \text{ imp. m.sg. } + 5 \text{ m.sg. build it; } 1V.16, 1V.41$
<i>b-r-<u>k</u></i>		<i>v.i.</i> to bless; to kneel; <i>tD</i> to pray; CM BRK id.
-	barakni	[bar'raxni] D perf. 1 st pl. we prayed; III.9
	qəmbarak	[qmbar'rax] D imperf. 1 st sg. I pray; 1.14
	qəmbarken	[qm/bar χ en] <i>D imperf.</i> 3^{rd} <i>pl.</i> they pray; IV.23
b-š-q-r		v.t. to know, understand; CM BŠQR 'to recognize,
-		examine;' possibly a secondary root from CM BQR 'to
		cleave; examine' in the Š-stem, produced by the

	bašqarni qəmbašqer qəmbašqarle	metathesis of the derivational morpheme and the initial radical of the stem. $[ba \int^{t} qa.mi] Q perf. 1^{st} pl. +3^{rd} m.sg.$ we recognized; I.11 $[qm^{t}ba \int qa.le] Q imperf. 3^{rd} m. sg.$ he knows; IV.35 $[qmba \int^{t} qa.le] Q imperf. 3^{rd} m.sg. + 1^{st} sg.$ he knows me; III.4
	láqəmbašqer	['laqmbaʃqɛJ] Q neg. imperf. 3^{rd} m. sg. he doesn't know; III.12
	qəmbašqirət qəmbašqart láqəmbašqar	[qmbaʃ'qirrt] Q imperf. 2^{nd} sg. you know; IV.106 [qmbaʃ'qaɪt] Q imperf. 2^{nd} sg. (contextual) you know; I.6 ['laqmbaʃqaɪ] Q neg. imperf. 1^{st} sg. (contextual) I don't know; II.18
	láqəmbašqeren	['laqmbaʃqɛrɛn] Q neg. imperf. 3^{rd} pl. they didn't know; II.25
	bašqir əl-nabšak	[baj'qi:ren 'nævjax] Q impv. m.sg. + 2^{nd} m.sg. know yourself
<i>b</i> - ^{<i>w</i>} / _{<i>y</i>} - ^{<i>w</i>} / _{<i>y</i>}	qəbāyi	<i>v.t.</i> to want (+ <i>men</i> s.o. + <i>subjv.</i> to do s.t.) CM BAA id. [qə'bɔ:ji] <i>G imperf.</i> 3^{rd} <i>m.sg.</i> he wants; V.2
	qəbāy'	[qə'bɔi] <i>G imperf.</i> 3 rd m.sg. (contextual) he wants; IV.2, IV.92, VIII.2
	qəbāt qabin	[qə'bət] <i>G imperf.</i> 2 nd sg. you want; IV.63, IV.95, IX.8 [qa'bin] <i>G imperf.</i> 1 st sg. I want; I.3, I.5, I.6, I.7, II.1, II.17, III.1, IV.39, IV.83, IV.118, IX.3, IX.9, IV.6
	qabeyyen qabén láqbèn	[qa'bɛıjɛn] <i>G imperf.</i> 3^{rd} <i>pl.</i> they want; II.6, II.29, III.2 [qa'bɛn] <i>G imperf.</i> 1^{st} <i>pl.</i> (contextual) we want; IV.132 ['lɑʁbɛn] <i>G neg. imperf.</i> 1^{st} <i>pl.</i> (contextual) we don't want; IV.109
ba'ad		['basad] prep. after; Arabic basdid.; X.2, X.13
ba ^c aden		[basa'den] <i>adv.</i> afterwards; Arabic <i>ba'dayn</i> id.; III.10
ba'aẓ		['basazs] <i>quant.</i> some; Arabic <i>ba</i> d id.
baʻid		[ba'sid] <i>adj.</i> far, distant; Arabic <i>ba'id</i> id.; III.10
bābā	bāb ḥammām	['bɔwɔ] <i>n.</i> father, proprietor; CM baba 2 id.; Tab. 1.1, Tab. 1.3, Tab. 1.5, I.8d, II.2b, IV.47, IV.54 ['bɔu ħam'mɒm] the proprietor of a hammam; VIII.4, VIII.5, VIII.8, VIII.15

	bābahye	$[bb'wahje] + 3^{rd} f.sg. + 3^{rd} m.sg. cop.$ he is her father; IX 14
	bābe <u>k</u> bābe	$[b:we\chi] + 2^{nd} f.sg.$ your father; IX.9 $[b:we] + 1^{st} sg.$ my father; IX.10
babbā	babbe	['bæbbɔ] <i>n</i> . grandfather; VIII.11 ['bæbbe] + I^{st} sg. my grandfather; II.2
Baġdād		[bus'dɒd] <i>prop. n.</i> Baghdad, the capital of Iraq; II.27, IV.141, IV.142
balwā '-	- <u>b</u> -d	<i>v.t.</i> to make trouble, cause a disturbance (+ -d- for s.o.); cf. Persian <i>balvā kardan</i> id.
	balwā-dan ābed	[bælwæ'dan ,ɔ:wɛd] <i>G</i> imperf. 3^{rd} m.sg. + 1^{st} pl. he will make trouble for us; VII.9
Bandar-	e Ma'šur	[bæn ₁ dare ma?'ʃuɪ] <i>prop. n.</i> Bandar-e Ma'šur, a port city in southwest Iran; III.6
bandi b	-ġ-ș	<i>v.i.</i> to stay put; see §2.7.1.5 for forms with final /r/; Persian <i>bandi</i> 'prisoner;' see also <i>bandi t-m-m</i>
	bandi bəğoşyon bandir lábğeşyon	[bæn'di: bə'ko:s [°] jon] <i>G impv. pl.</i> stay put; IV.133 <i>a</i> [bæn'di: 'lab \varkappa 3s [°] jon] <i>G neg. perf.</i> 3 rd <i>pl.</i> they did not stay put; IV.146
bandi t-	<i>m-m</i>	<i>v.i.</i> to be confined; see §2.7.1.5 for forms with final /r/; Persian <i>bandi</i> 'prisoner;' see also <i>bandi b-g-ṣ</i>
	bandir tammon	[bæn ¹ di1 ₁ tæmmon] G perf. 3^{rd} pl. they were confined; IV.129, IV.146
barābar	barābar-du	[bærb'bæ1] prep. opposite; Persian <i>barābar</i> id.; IV.50 [bærb'bæ1du] + 3^{rd} pl. suff. opposite them; IV.76
barnāšā	barnāš	[baɪˈnɔ:ʃɔ] <i>n</i> . person; CM br ʿnaša id; IV.16, IV.17, IV.42 [baɪˈnɔʃ] person (contextual); I.11
	barnašānā barnašāni barnašāne	[$baine^{1}$ 5:n5] <i>pl.</i> people; I.8, IV.9, IV.14, IV.28, IV.70, IV.139 [$baine^{1}$ 5:ni] <i>pl. indef.</i> (some) people; I.13, IV.65 [$baine^{1}$ 5:ne] <i>pl.</i> + 1 st sg. my people; IV.118
barra		['barra] adv. outside; Arabic barra id.; II.13, II.15, III.11

306

bas		[bæs] <i>adv.</i> only, just; colloquial Arabic <i>bas</i> id.; II.17, V.5, VIII.14, VIII.16
bātar batlu <u>k</u> tā	ī	['bɔ:θaɪ] <i>adv</i> . behind; CM abatar id. [bɑt ^{s'} lʌɣtɔ] <i>n</i> . misfortune; cf. CM baṭla 'vain;' 1.14
bāz	bāzi	<i>n.</i> hawk; Persian <i>bāz</i> id. ['bɒ:zi] indef. a hawk; V.21, V.25, V.26
bazrā		['bæzro] n. seed; CM bazira / bazra id.
behtar	behtar	[bɛh] <i>adj.</i> good; Persian <i>beh</i> id. [bɛh'tæɪ] <i>adv.</i> better; Persian <i>behtar</i> id.
bellā		['bɛllɔ] n. husband; CM bila id.
benāya	benāyāt	<i>n.</i> building; Arabic <i>bināya</i> id. [bɪnəˈjət] <i>pl</i> . buildings; III.22
benyān	b-n- ^w / _y benyān qəbānen	<i>v.i.</i> to build; cf. Persian <i>benā kardan</i> id. [bɛn'jɔn qə'bɔ:nɛn] <i>G imperf. 3rd pl.</i> they build; IV.17
besrā		['bɛsrɔ] n. meat, flesh; CM bisra id.
beyyā		['bɛɪjɔ] <i>n</i> . egg; CM baia id.
bərattā	bərat bəratti bərat ahhā bərat bərattā bərat ebrā bərat hāṯā bərat xāle bərat xālu bənāṯā	[bə'rattɔ] <i>n</i> . (pausal) daughter; CM brata id.; IV.47, IV.50 [bə'rat] (contextual) daughter; II.10, IX.14 [bə'ratti] + 3^{rd} <i>m.sg.</i> his daughter; IV.44, IX.7, IX.8 [bə _r rat 'æhhɔ] <i>n</i> . niece [bə _r rat bə'rattɔ] <i>n</i> . granddaughter [bə _r rat 'b:bɔ] <i>n</i> . granddaughter [bə _r rat 'h:b:b] <i>n</i> . niece [bə'rat 'yp:le] <i>n</i> . cousin [bə _r rat 'yp:lu] <i>n</i> . cousin [bə _r rat 'yp:lu] <i>n</i> . cousin
bəțālā		[bəˈt ^s ɔːlɔ] <i>adj.</i> bad; cf. CM baṭla 'vain, useless'
bəzuyā	bəzuyi	<i>n.</i> hole; CM <i>bzuia</i> id. [bɪ'zuːji] <i>indef.</i> a hole; V.8

308		8. Lexicon
bi		[bi] <i>prep.</i> without; Persian <i>bi</i> id.; IV.120
bibi		[¹ bi:bi] <i>n</i> . grandmother
bienā	bienā-dan bienā-du	['biĕnɔ] <i>prep.</i> between, among; Arabic <i>bayna</i> id. ['biĕnɔdan] <i>prep.</i> + 1 st pl. among ourselves; IV.81 ['biĕnɔdu] <i>prep.</i> + 3 rd pl. among themselves; IV.89
bie <u>t</u> ā	bie <u>t</u> bie <u>t</u> wā <u>t</u> bie <u>t</u> i bie <u>t</u> e bie <u>t</u> wā <u>tk</u> on	['biě θ_0] <i>n</i> . house (f.); CM baita (pl. baitauata) id.; IV.64, IX.6, X.8 ['biě θ] (contextual) house; III.11, III.12, III.22 V.4 [biě θ 'w 0θ] <i>pl.</i> houses; III.13, III.10, III.12, III.24 ['biě θ i] <i>indef.</i> a house; IV.88, IX.4 ['biě θ e] + 1 st sg. my house; IV.118, V.16 [biě θ 'w 0θ \chion] + 2 nd pl. your houses; IV.123
bis		[bis] <i>quant.</i> twenty; Persian <i>bist</i> id.; III.14 (x2)
bištar		['bista1] adj. more; Persian bištar id.; II.22
burkā		['bʌ.ɪkɔ] <i>n</i> . knee
bušme		['bʉ∫me] <i>n</i> . sky, air; IV.141
bu <u>t</u> ā	buțā d-Šāmeš buți	[ˈbu:θɔ] <i>n</i> . prayer; CM buta id. [ˌbu:θɑd'ʃɒ:mɛʃ] n. prayer of Šāmeš, the sun; IV.55, IV.83 [ˈbu:θi] <i>indef</i> . a prayer; IV.91, IV.114
bu <u>t</u> i q-r	^w / _y bu <u>t</u> i eqrā	ν.i. to recite a prayer ['bu:θi jɛqrɒ] <i>G perf. 3rd m.sg.</i> he prayed; IV.114

₿

Č

['wo:wo] n. door, gate; CM baba 1 id; IX.5

bābā

čand [tfænd] adv. how much; Persian čand id. $\check{c}ang m-h-{}^{w}/_{y}$ v.i. to play a harp; cf. Persian čang zadan id. [tfæn'gp: mə hon] G perf. 3^{rd} pl. they played harps; IV.81 čangā́ məhon čāre ['tfp:re] n. remedy; Persian čāre id.; IV.40 čāre '-b-d v.t. to remedy (+ qam obj. + qam for s.o.); cf. Persian čāre kardan id. ['tfp:re av.detton] G imperf. 2nd pl. you will fix; IV.39 čāre abdetton Čečāwā [tʃɛ'tʃɔ:wɔ] prop. n. Čečāwā bar Čečāwā, the name of a cloud-shaped demon that is capable of moving the earth; IV.53 (x2), IV.92, IV.93, IV.94, IV.95, IV.98, IV.99, IV.114, IV.119, IV.120, IV.121, IV.122, IV.123, IV.127, IV.134, IV.147 Čečāwā bar Čečāwā [tfe'tfo:wo bas tfe'tfo:wo] prop. n. Čečāwā, son of Čečāwā; IV.90, IV.91 čehél quant. forty (used in counting); Persian čehel forty čehél ruz ['tferruz] quant. forty days; IV.137 ['tf3mm3] quant. nine hundred; CM tšima id.; III.1 čemmā čin ['tfin] quant. ninety; CM tšin id.; III.1 činassər [tf1'nessr] quant. nineteen (probably the result of metanalysis; the expected form is eččassor; cf. CM tšasar id.); III.14

D

<i>d</i> -		[d] <i>rel. pron.</i> (archaism) of; CM <u>d</u> -; I.1, I.9 (x2), IV.55, IV.65, IV.68, IV.83, IV.85, IV.121, IV.135, IV.138
d-h-l	dahel qədāhlen	<i>v.i.</i> to fear, be afraid; CM DHL id. [da'hɛl] <i>G imperf.</i> 1 st sg. (contextual) I am afraid; IV.132 [qə'dɒhəlɛn] <i>G imperf.</i> 3 rd pl. they will be afraid; IV.98
d- <u>k</u> -r	láqda <u>k</u> ar	<i>v.t.</i> to remember; CM DKR ['laqda, xaı] <i>G neg. imperf.</i> 1 st sg. (contextual) I don't remember well; II.4
	qədakarnā	[qəda ¹ xa.ino] G imperf. 1 st sg. I remember; II.1, II.33, III.1
d - r - $^{w}/_{y}$	dər'	<i>v.t.</i> to take; CM DRA id.; see also $hammam d-r^w/_y$ [del] <i>G perf.</i> $3^{rd} m.sg. + obj.$ (contextual) he took; VIII.8, X.11
	dərā dəri dərilu	[dɛ'rɔ:] <i>G perf.</i> 3^{rd} <i>m.sg.</i> he took; V.21 [dɛ'ri:] <i>G perf.</i> 3^{rd} <i>m.sg.</i> (contextual) he took; X.8 [dɛ'ri:lu] <i>G perf.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>pl.</i> he took them; V.5
	dəronnu	[də'ronnu] <i>G perf.</i> $3^{rd} pl. + 3^{rd} pl.$ they took them; III.22, III.24
	qədāri eqdar'	[qə'dɔ:ri] <i>G</i> imperf. 3^{rd} m.sg. he will take [ϵ q'd α I] <i>G</i> imperf. 3^{rd} m.sg. + obj. (contextual) he takes it; V.25
	láqdar'	['laʁdæɪ] <i>G neg. imperf.</i> 3^{rd} <i>m.sg.</i> + <i>obj.</i> (contextual) he doesn't take it; V.26
	qədarinā dəri	[qədæ'ri:nɔ] G imperf. 1^{st} sg. I will take [də'ri:] G impv. m.sg. take!
<i>d</i> - ^{<i>w</i>} / _{<i>y</i>} - <i>š</i>	daš	<i>v.t.</i> to enter; CM DUŠ and DIŠ 'to trample down' $[dæ \int] G$ perf. 3^{rd} m.sg. he entered; VIII.6, VIII.13
dah		[dæh] quant. ten (used in counting); Persian dah id.; V.26
dahbā		['dæh.wə] n. gold; CM dahba id.
darwiš	darwišā́	[dæɪ'vi∫] <i>n</i> . dervish; Persian <i>darviš</i> id. [dæɪˌvi'∫ɔ:] <i>pl</i> . dervishes; IV.20, IV.28

dawāzdáh	[dævoz'dæh] <i>quant.</i> twelve (used in counting); Persian <i>davāzdah</i> id.; IV.20
daxil daxil-da <u>k</u> daxil-də <u>k</u> on	$[dae'\chi il]$ <i>n</i> . charge, ward; Arabic <i>daxīl</i> id. $[dae'\chi ilda\chi] + 2^{nd}$ <i>m.sg.</i> your ward, under your control; IV.56 $[dae'\chi ilda_1\chi on] + 2^{nd}$ <i>pl.</i> your ward, under your control; IV.61
demmā	[ˈdɛmmə] n. tail; CM dinba id.
deqnā daqqen həwarānā	['dɛqnɔ] <i>n</i> . beard; CM ziqna and diqna (st. abs. daqin) id. ['daqqɛn huwɛ'rɔnnɔ] elders, lit. "white beards"; IV.31
deštā deštā <u>t</u> ā	[¹ dɛ∫tɔ] <i>n.</i> (patch of) land, ground, earth (f.); CM dišta id.; IV.98, IV.100, IV.101, IV.102, IV.105, IV.108, IV.109, IV.115, IV.125, IV.128, IV.131, IV.137, IV.137, IV.138, IV.139, IV.141, IV.145, IV.149 [dɛʃ'tɔ:θɔ] <i>pl.</i> wilderness; CM dištata id.; IV.103
Dezful	[dɛz'ful] <i>prop. n.</i> Dezful, a city in northern Khuzestān, Iran; III.6, III.7, III.19
dəholtā	[dəˈholtə] n. fear; CM dhulta id.;
dəmā	[dəˈmɒ:] <i>n</i> . blood; CM dma id.
dəraxt	[dəˈræɣt] n. tree; Persian deraxt id.
dəros	[də'ros] <i>adj.</i> correct; Persian <i>dorost</i> id.
diwā diw	[¹ di:vɔ] <i>n.</i> demon; Middle Persian <i>dēw</i> id. via CM daiua id.; IV.45, IV.55, IV.57, IV.64, IV.83 [div] demon (contextual); IV.63
Diwān Diwān A <u>b</u> ā <u>t</u> ar	[di ¹ von] <i>n</i> . a type of scroll, usually illustrated; CM diuan id., cf. Persian <i>divān</i> a small book (usually of poetry) [di ₁ von a ¹ wo:θa1] <i>prop. n</i> . an illustrated description of the Lightworld

312		8. Lexicon
Diwā	n Malkuṯā Əleṯā	[di,von mal,ku:θο ə'le:θο] <i>prop. n.</i> the "Scroll of Exalted Kingship," an esoteric commentary on the initiation of the <i>tarmidā</i>
do	do <u>t</u> a- <u>t</u> -kon	[do] <i>quant.</i> two (used in counting); Persian <i>do</i> id.; VII.6 [do ¹ $\theta a \theta \chi on$] + 2 ^{<i>nd</i>} <i>pl</i> . two of you, both of you
Drāši	Drāši d-Yahya	['drɒʃi] <i>n</i> . teachings; CM drašia id. [ˌdrɒʃi 'dɪhjæ] <i>prop. n</i> . the Teachings of John, one of the primary Mandaean religious texts
dukkā	kol dukkā dukki dukki dukkāni	['dʌkkɔ] <i>n</i> . place; CM duka id.; I.4; see also $du\underline{k}t\overline{a}$ [kol 'dʌkkɔ] every place, every part ['dʌkki] <i>indef</i> . a place; IV.125 ['døkki] + 3 rd m. sg. his place; IV.140 [dʌk'kɔni] pl. + rest. the places (which) III.19
dukkān		[dʌkˈkɔn] <i>n</i> . store; Arabic <i>dukkān</i> id. via Persian <i>dokkān</i> id.; III.4
du <u>k</u> tā		['daxto] n. place; CM duka id.; see also dukkā; V.7
dust	dusti	['dust] <i>n</i> . friend; Persian <i>dust</i> id. ['dusti] <i>indef</i> . a friend; V.4

Ε

ebrā		['ɛbrɔ] n. son; CM abra and 'bra id.; V.29
	ebri	$[{}^{\prime}\epsilon bri] + 3^{rd} m. sg.$ his son; I.7, I.10, V.18, V.20, V.21, V.26
	bar	[ba.] son (only in patronymic or matronymic); IV.53
	ebbər ahhā	[ˌɛbbr ˈæhhɔ] n. nephew
	ebbər bərattā	[ɛbbr ˈbəˈrattə] n . grandson
	ebbər ebrā	[ɛbbr 'ɛbrɔ] n. grandson; CM br bra id.
	ebbər ebri	$[\varepsilon bbr \ \varepsilon bri] + 3^{rd} m.sg.$ his grandson; I.7, I.10
	ebbər hātā	$[\epsilon bbr ho: \thetao]$ <i>n</i> . nephew
	ebbər xāle	['ebbr 'xp:le] n. cousin
	ebbər xālu	['ebbr 'xp:lu] n. cousin
eččā		['ɛtʃtʃɔ] quant. nine; CM tša id.; II.22
	eččin	[ɛtʃ ^t tʃin] <i>pl</i> . ninety; CM tšin id.; II.22
	eččā eččin	[¹ ɛtʃtʃɔ ɛtʃ ¹ tʃin] ninety-nine; II.22
ehdā		['ehdo] quant. one; CM hda id.; cf. Arabic ihdā 'one (f.);'
		IV.43
	ehdi	['ɛhdi] <i>indef</i> . (a particular) one; I.11, I.13
ehl-		<i>pseudoverb</i> to have (lit. there is s.t. for s.o.); CM 't l- id.;
ehl-		see also <i>e<u>k</u>t-, asutā ehl-</i> , and <i>rəwāhā ehl-</i>
ehl-	ehli	see also e_kt -, $asut\bar{a}$ ehl -, and $r = w\bar{a}h\bar{a}$ ehl - [' ϵ hli] + 3^{rd} m. sg. he has; IV.49, IV.135, V.4, V.26, V.26
ehl-	ehli ehla <u>k</u>	see also <i>e<u>k</u>t-, asutā ehl-</i> , and <i>rəwāhā ehl-</i>
ehl-		see also <i>ekt-</i> , <i>asutā ehl-</i> , and <i>rəwāhā ehl-</i> ['ɛhli] + 3^{rd} <i>m. sg.</i> he has; IV.49, IV.135, V.4, V.26, V.26 ['ɛhəlak] + 2^{nd} <i>m.sg.</i> you have; IV.118 ['ɛhle] + 1^{st} <i>sg.</i> I have: III.4
ehl-	ehla <u>k</u>	see also <i>ekt-</i> , <i>asutā ehl-</i> , and <i>rəwāhā ehl-</i> ['ɛhli] + 3^{rd} <i>m. sg.</i> he has; IV.49, IV.135, V.4, V.26, V.26 ['ɛhəlak] + 2^{nd} <i>m.sg.</i> you have; IV.118 ['ɛhle] + 1^{st} <i>sg.</i> I have: III.4
ehl-	ehla <u>k</u> ehle	see also ekt -, $asut\bar{a} ehl$ -, and $r \partial w \bar{a} h \bar{a} ehl$ - [' ϵ hli] + $3^{rd} m$. sg . he has; IV.49, IV.135, V.4, V.26, V.26 [' ϵ helak] + $2^{nd} m$. sg . you have; IV.118 [' ϵ hle] + $1^{st} sg$. I have; III.4 [' ϵ hlu] + $3^{rd} pl$. they have; IV.45 [' $l\epsilon$ hlu] neg . + $3^{rd} pl$. they don't have; III.12
ehl-	ehla <u>k</u> ehle ehlu	see also ekt -, $asut\bar{a} ehl$ -, and $r \partial w \bar{a} h \bar{a} ehl$ - [' ϵ hli] + $3^{rd} m$. sg. he has; IV.49, IV.135, V.4, V.26, V.26 [' ϵ helak] + $2^{nd} m$.sg. you have; IV.118 [' ϵ hle] + 1^{st} sg. I have; III.4 [' ϵ hlu] + $3^{rd} pl$. they have; IV.45 [' $l\epsilon$ hlu] neg . + $3^{rd} pl$. they don't have; III.12 [' ϵ hlan] + 1^{st} pl. we have; II.23
ehl-	ehla <u>k</u> ehle ehlu lehlu	see also <i>ekt-</i> , <i>asutā ehl-</i> , and <i>rəwāhā ehl-</i> ['ehli] + 3^{rd} m. sg. he has; IV.49, IV.135, V.4, V.26, V.26 ['ehəlak] + 2^{nd} m.sg. you have; IV.118 ['ehle] + 1^{st} sg. I have; III.4 ['ehlu] + 3^{rd} pl. they have; IV.45 ['lehlu] <i>neg.</i> + 3^{rd} pl. they don't have; III.12 ['ehlan] + 1^{st} pl. we have; II.23 [hə'wɔ:li] G perf. 3^{rd} m.sg. + 3^{rd} m.sg. he had, it was for
ehl-	ehla <u>k</u> ehle ehlu lehlu ehlan həwāli	see also <i>ekt-</i> , <i>asutā ehl-</i> , and <i>rəwāhā ehl-</i> ['ehli] + 3^{rd} m. sg. he has; IV.49, IV.135, V.4, V.26, V.26 ['ehəlak] + 2^{nd} m.sg. you have; IV.118 ['ehle] + 1^{st} sg. I have; III.4 ['ehlu] + 3^{rd} pl. they have; IV.45 ['lehlu] <i>neg.</i> + 3^{rd} pl. they don't have; III.12 ['ehlan] + 1^{st} pl. we have; II.23 [hə'wɔ:li] G perf. 3^{rd} m.sg. + 3^{rd} m.sg. he had, it was for him; V.1, V.3
ehl-	ehla <u>k</u> ehle ehlu lehlu ehlan həwāli həwālu	see also <i>ekt-</i> , <i>asutā ehl-</i> , and <i>rəwāhā ehl-</i> ['ehli] + 3^{rd} <i>m. sg.</i> he has; IV.49, IV.135, V.4, V.26, V.26 ['ehlak] + 2^{nd} <i>m.sg.</i> you have; IV.118 ['ehle] + 1^{st} <i>sg.</i> I have; III.4 ['ehlu] + 3^{rd} <i>pl.</i> they have; IV.45 ['lehlu] <i>neg.</i> + 3^{rd} <i>pl.</i> they don't have; III.12 ['ehlan] + 1^{st} pl. we have; II.23 [hə'wɔ:li] <i>G perf.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>m.sg.</i> he had, it was for him; V.1, V.3 [hu'wɔ:lu] <i>G perf.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>pl.</i> they had; II.29, III.13
ehl-	ehla <u>k</u> ehle ehlu lehlu ehlan həwāli	see also <i>ekt-</i> , <i>asutā ehl-</i> , and <i>rəwāhā ehl-</i> ['ɛhli] + 3^{rd} m. sg. he has; IV.49, IV.135, V.4, V.26, V.26 ['ɛhlak] + 2^{nd} m.sg. you have; IV.118 ['ɛhle] + 1^{st} sg. I have; III.4 ['ɛhlu] + 3^{rd} pl. they have; IV.45 ['lɛhlu] <i>neg.</i> + 3^{rd} pl. they don't have; III.12 ['ɛhlan] + 1^{st} pl. we have; II.23 [hə'wɔ:li] G perf. 3^{rd} m.sg. + 3^{rd} m.sg. he had, it was for him; V.1, V.3 [hu'wɔ:lu] G perf. 3^{rd} m.sg. + 3^{rd} pl. they had; II.29, III.13 ['lɛhwɔlu] G neg. perf. 3^{rd} m.sg. + 3^{rd} pl. they didn't
ehl-	ehla <u>k</u> ehle ehlu lehlu ehlan həwāli həwālu lehwālu	see also <i>ekt-</i> , <i>asutā ehl-</i> , and <i>rəwāhā ehl-</i> ['ɛhli] + 3^{rd} <i>m. sg.</i> he has; IV.49, IV.135, V.4, V.26, V.26 ['ɛhlak] + 2^{nd} <i>m.sg.</i> you have; IV.118 ['ɛhle] + 1^{st} <i>sg.</i> I have; III.4 ['ɛhlu] + 3^{rd} <i>pl.</i> they have; IV.45 ['Iɛhlu] <i>neg.</i> + 3^{rd} <i>pl.</i> they don't have; III.12 ['ɛhlan] + 1^{st} pl. we have; II.23 [hə'wɔ:li] <i>G perf.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>m.sg.</i> he had, it was for him; V.1, V.3 [hu'wɔ:lu] <i>G perf.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>pl.</i> they had; II.29, III.13 ['Iɛhwɔlu] <i>G neg. perf.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>pl.</i> they didn't have; II.9
ehl-	ehla <u>k</u> ehle ehlu lehlu ehlan həwāli həwālu	see also <i>ekt-</i> , <i>asutā ehl-</i> , and <i>rəwāhā ehl-</i> ['ɛhli] + 3^{rd} m. sg. he has; IV.49, IV.135, V.4, V.26, V.26 ['ɛhlak] + 2^{nd} m.sg. you have; IV.118 ['ɛhle] + 1^{st} sg. I have; III.4 ['ɛhlu] + 3^{rd} pl. they have; IV.45 ['lɛhlu] <i>neg.</i> + 3^{rd} pl. they don't have; III.12 ['ɛhlan] + 1^{st} pl. we have; II.23 [hə'wɔ:li] G perf. 3^{rd} m.sg. + 3^{rd} m.sg. he had, it was for him; V.1, V.3 [hu'wɔ:lu] G perf. 3^{rd} m.sg. + 3^{rd} pl. they had; II.29, III.13 ['lɛhwɔlu] G neg. perf. 3^{rd} m.sg. + 3^{rd} pl. they didn't
ehl- ehnā	ehla <u>k</u> ehle ehlu lehlu ehlan həwāli həwālu lehwālu	see also <i>ekt-</i> , <i>asutā ehl-</i> , and <i>rəwāhā ehl-</i> ['ɛhli] + 3^{rd} <i>m. sg.</i> he has; IV.49, IV.135, V.4, V.26, V.26 ['ɛhlak] + 2^{nd} <i>m.sg.</i> you have; IV.118 ['ɛhle] + 1^{st} <i>sg.</i> I have; III.4 ['ɛhlu] + 3^{rd} <i>pl.</i> they have; IV.45 ['Iɛhlu] <i>neg.</i> + 3^{rd} <i>pl.</i> they don't have; III.12 ['ɛhlan] + 1^{st} pl. we have; II.23 [hə'wɔ:li] <i>G perf.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>m.sg.</i> he had, it was for him; V.1, V.3 [hu'wɔ:lu] <i>G perf.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>pl.</i> they had; II.29, III.13 ['Iɛhwɔlu] <i>G neg. perf.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>pl.</i> they didn't have; II.9
	ehla <u>k</u> ehle ehlu lehlu ehlan həwāli həwālu lehwālu	see also <i>ekt-</i> , <i>asutā ehl-</i> , and <i>rəwāhā ehl-</i> ['ehli] + 3^{rd} <i>m. sg.</i> he has; IV.49, IV.135, V.4, V.26, V.26 ['ehlak] + 2^{nd} <i>m.sg.</i> you have; IV.118 ['ehle] + 1^{st} <i>sg.</i> I have; III.4 ['ehlu] + 3^{rd} <i>pl.</i> they have; IV.45 ['lehlu] <i>neg.</i> + 3^{rd} <i>pl.</i> they don't have; III.12 ['ehlan] + 1^{st} pl. we have; II.23 [hə'wɔ:li] <i>G perf.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>m.sg.</i> he had, it was for him; V.1, V.3 [hu'wɔ:lu] <i>G perf.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>pl.</i> they had; II.29, III.13 ['lehwolu] <i>G neg. perf.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>pl.</i> they didn't have; II.9 ['hwo:lan] <i>G neg. perf.</i> 3^{rd} <i>m.sg.</i> + 1^{st} <i>pl.</i> we had; II. 7

0 1	r •
8.	Lexicon

	lekkā lek enši	['l3kkɔ] <i>neg</i> . there isn't (anything); III.22 [l3k 'mʃi] <i>neg</i> . (contextual) there isn't anyone; III.24
ekkā 2	ekkā ekkā	['ɛkkɔ] <i>adj.</i> such; CM 'ka id.; IV.53, IV.118 ['ɛkkɔ 'ɛkkɔ] such and such, etc. etc.; V.20
ekkā 3		['ɛkkɔ] <i>adv</i> . thus; CM 'ka id.; IV.12
ekka <u>k</u>		[ɛk'kɑɣ] <i>dem. pron.</i> there; possibly derived from CM 'ka 'there' + CM hak 'that;' II.3, II.25, III.3, III.6, III.8, III.9, III.13, III.18, III.20, V.8, V.9, V.18
e <u>k</u> t-	e <u>k</u> ti le <u>k</u> ta le <u>k</u> tan e <u>k</u> ta <u>k</u> le <u>k</u> ta <u>k</u>	cop. copula; CM 'it id.; see also <i>ehl</i> - [${}^{1}\epsilon\chi$ ti] + 3 ^{<i>rd</i>} <i>m.sg.</i> he is; I.11c [${}^{1}l3\chi$ ta] <i>neg.</i> + 3 ^{<i>rd</i>} <i>f.sg.</i> she is not; I.13a [${}^{1}l3\chi$ tan] + 1 ^{<i>st</i>} <i>pl.</i> we are not; I.12a [${}^{1}\epsilon\chi$ ta χ] + 2 ^{<i>nd</i>} <i>m.sg.</i> you are; IV.51 [${}^{1}l\xi\chi$ ta χ] <i>neg.</i> + 2 ^{<i>nd</i>} <i>m.sg.</i> you are; VIII.11b
elyā	elli elliye ellinon	[¹ ɛljɔ] <i>interrog. and indef. pron.</i> where; CM 'lia id. [¹ ɛlli] where (contextual); IX.11 [¹ ɛllije] where is it; X.14 [¹ ɛllinon] + 3^{rd} pl. where they are; III.22
<i>emmā</i> 1	emme	[1 ɛmmɔ] <i>n</i> . mother; CM ' ma id. [1 ɛmme] + 1^{st} sg. my mother; II.2
emmā 2	2	['ɛmmɔ] <i>quant.</i> 100; CM ma id.; V.3
emrā	emrān <u>k</u> on	['ɛmbrɔ] <i>n</i> . sheep; CM ' mbra id. [ɛmbə'rɒn χ on] + 2^{nd} pl. your sheep; IV.124
enšā		See ettā
enši		['ɛnʃi] <i>indef. pron.</i> anyone; cf. CM 'niš id.; III.24, IV.84
Enyāni		[ɛnˈjɔːni] <i>n</i> . antiphonal responses, part of the Canonical Prayerbook of the Mandaeans; CM 'niania id.
esbu ^c a	səbu ^c a	[ɛsˈbuːʕa] <i>n</i> . week; colloquial Arabic <i>isbū</i> ^c id.; X.13 [səˈbuːwɔ] week; VIII.9

esrin		[ɛs'rin] quant. twenty; CM 'srin id.; III.14
ešgandā		[ɛʃ'gændɔ] <i>n.</i> acolyte; CM šganda id.
<i>ešmā</i>	eššəm šom ešmi ešma ešmehon	[${}^{l}\epsilon$]mo] <i>n</i> . name; CM šum 1 and šuma id. [ϵ]jm] name (contextual); I.9 [jom] name (contextual); CM šum id. [${}^{l}\epsilon$]mi] +3 rd m.sg. his name; II.7, II.8, II.10, II.11 [${}^{l}\epsilon$]ma] + 3 rd f.sg. her name; II.10 [ϵ]me:hon] pl. + 3 rd pl. their names; I.1
eštā	šətā	[Iʃ ¹ tɔ:] <i>adv.</i> now; CM hašta id.; I.10c, I.14e, II.33a (x2, II.34, III. 18a; III.18b III.21, III.24a [ʃ1 ¹ tɔ:] now; II.29b
e <u>tt</u> ā	et etti inšā inšānā	<i>n</i> . woman, wife (f.); CM 'nta id.; II.24, II.26 [εθ] woman (contextual); II.14 ['εθθi] <i>indef.</i> a woman; II.10 ['Inʃɔ] <i>coll.</i> women; CM 'nšia id.; I.2, IV.81, IV.105, IV.124 [In¹ʃɔ:nɔ] <i>pl.</i> women; IV.19, IV.139, IV.148, IV.148
ewaz		[əˈvɑz] <i>prep.</i> instead of; Arabic <i>'iwaḍa</i> id. via Persian <i>(dar) evaz</i> id.; VI.6

Ð

əbādā		['wo:do] n. work; CM 'bada id.; IV.137
	əbād	[əˈwɔd] (contextual) work; IV.13
	əbādu	$[\exists w du] + 3^{rd} pl$. their work; IV.138
əbādā '-	b-d	v.i. to do something, work; cf. Persian kār kardan id.
	əbād abdat	$[\mathfrak{d}'w\mathfrak{d} \mathfrak{a}v\mathfrak{d}\mathfrak{a}t]$ <i>G perf.</i> $\mathfrak{Z}^{rd}f$. sg. she has worked, done work; 1.6
	əbād abdit	[əˈwəd avˌdit] G perf. 1 st sg. I did something; V.9
	əbādi əbod	[ə'wədi ə,wod] G impv. m.sg. do something; IV.62
əhā <u>b</u> ā		[ə'hə:wə] n. gift; CM ahaba 'presentation, giving'
əl-		[11] <i>prep.</i> to, for; referential object marker (clitic form); CM I id.; I.5 (x2), I.6, I.9, I.11 (x2), III.2, III.8, III.11, IV.35, IV.55, IV.56, IV.57, IV.58, IV.61, IV.64, IV.76, IV.77, IV.79, IV.81, IV.83 (x2), IV.87, IV.88, IV.90, IV.92, IV.94 (x2), IV.95, IV.98, IV.107, IV.109 (x2), IV.119, IV.110, IV.120, IV.122, IV.131, V.12 (x2), V.18, V.25, V.26, V.29, VII.2, VII.8, VII.9, VIII.4, VIII.7, VIII.8, VIII.15, IX.5, IX.7, X.3, X.4, X.11, X.16
əlāw	əlāwwi əlāwwa <u>k</u> əlāwwu	$[\exists^{l} \exists \upsilon]$ <i>prep.</i> to, for; CM [au- id.; III.13, IV.11, IV.19, IV.19, IV.19, IV.19, IV.33, IV.35, IV.36, IV.73, IV.74, IV.132, IV.138, IV.141, V.4 [$\epsilon^{l} \exists \upsilon \upsilon \upsilon i] + 3^{rd} m.sg.$ to it; IV.6 [$^{l} \exists \upsilon \upsilon \upsilon \alpha \chi$] + 2 nd m.sg. for you; IV.98 [$\vartheta^{l} \exists \upsilon \upsilon \upsilon u$] + 3 rd pl. to them; III.4
əlimā		[əˈliːmɔ] <i>adj.</i> thick; CM 'lima youthful; strong, sturdy
əm-		See men
əmšabba	ā	[m¦∫abbɔ] <i>adj.</i> praised; CM mšaba id.
Əmšuni	Kušțā	[m,ʃu:nikʉʃ't [°] ɔ:] <i>prop. n.</i> the "Sublimated of Truth," i.e. the Realm of Ideals; CM mšunia kušţa id.; IV.118, IV.120, IV.121, IV.144
ənāšā	ənāš	$[\vartheta^{1}n\vartheta;\vartheta]$ <i>n</i> . people (collective; takes plural referents) CM `naša id.; IV.22, IV.72, IV.81, IV.116, IV.128, VII.5 $[\vartheta^{1}n\vartheta]$ people (contextual); I.5, I.6c, I.14d, IV.142a

ərbiāhā	[rbɪˈjɔːhɔ] <i>adj.</i> fourth; cf. Syriac אפרבא <i>י rəbīʿāyā</i> id.
ərbiāyā	[rbɪˈjɔːjɔ] <i>adj.</i> fourth; cf. Syriac ،تحدك <i>rəbِīʿāyā</i> id.
ərikā	[əˈriːɣɔ] <i>adj.</i> long; CM ʿ rika id.
ərqihā ərqi	[r ^l qi:hɔ] <i>n</i> . heaven; CM rqiha sky, firmament; IV.145 [r ^l qi:] heaven (contextual); IX.12
	F
faqat ke	[¹ faqat [§]] <i>prep.</i> except; Arabic <i>faqaț</i> 'only' via Persian <i>faqat-ke</i> 'except;' II.23
Fārsí	[fɒ.1'si] adj. Persian; Persian Fārsi id.; III.12
farq farqi	[faɪq] <i>n</i> . difference; Arabic <i>farq</i> id. ['faɪqi] <i>indef</i> . a difference
farwāh	[faɪ'woh] <i>n</i> . thanks; cf. CM pruk , possibly Persian <i>farrox</i> 'fortunate' or Arabic <i>faraḥ</i> 'joy;' 1.5, 1.6; see also <i>farwāh</i> '- <i>m</i> - <i>r</i>
farwāh <u>k</u> on farwāh <u>k</u> on geš	$[fa. wo: \chi on] + 2^{nd} pl.$ thank you; I.3
farwāh '-m-r əmer farwāhkon	<i>v.t.</i> to thank, say thanks [ε 'mɛı faı'wo: χ on] <i>G subjv.</i> 1 st sg. + 2 nd pl. I thank you; I.3
fekər '-b-d fekər əbodyon	<i>v.t.</i> to think; Arabic <i>fikr</i> 'thought;' cf. Persian <i>fekr kardan</i> 'to think, reflect;' Tab. 1.3, Tab. 1.5 [fɛkrɛ ^l vodjon] <i>G impv. pl.</i> think; alternate reading for II.24
fonidā	[fo'ni:do] <i>n</i> . fish

G

g-h- <u>k</u>		v.i. to laugh; CM GHK id.
	gəhe <u>k</u>	[gə'h $\epsilon\chi$] <i>G perf.</i> 3^{rd} <i>m.sg.</i> he laughed; V.26, X.17
g-h- ^w / _y		<i>v.i.</i> to vomit; CM GSA to vomit, perhaps conflated with CM GHA to sob
g-n- <u>b</u>	gəna <u>b</u> ton gəni <u>b</u>	<i>v.t.</i> to steal; CM GNB id. [ge ['] natton] <i>G perf.</i> 2^{nd} <i>pl.</i> you stole; VII.9 [ge ['] ni:] <i>G pass. ptc.</i> stolen; VIII.3
g-n- ^w / _y	gənā gənit	v.i. to sleep, lie down; CM GNA id. [ge'no:] G perf. 3^{rd} m.sg. he slept; IV.47, IV.100 [gə'nit] G perf. 1^{st} sg. I slept; IV.46
g- <u></u> t-1	gațlonnu qəgāțel qəgāțelnā gațeln² gațelnākon lágāțlèn gəțol	<i>v.t.</i> to kill; CM GȚL id. [gat ^{\S1} lonnu] <i>G perf.</i> 3 ^{<i>rd</i>} <i>pl.</i> + 3 ^{<i>rd</i>} <i>pl.</i> they killed them; III.23 [qə ¹ go:t ^{\S} el] <i>G imperf.</i> 3 ^{<i>rd</i>} <i>sg.</i> he will kill [qəga ¹ t ^{\S} elno] <i>G imperf.</i> 1 ^{<i>st</i>} <i>sg.</i> I will kill [ga ¹ t ^{\S} elno] <i>G subjv.</i> 1 ^{<i>st</i>} <i>sg.</i> + <i>obj.</i> I should kill; IV.95 [ga ¹ t ^{\S} elno ₁ χon] <i>G imperf.</i> 1 ^{<i>st</i>} <i>sg.</i> + 2 ^{<i>nd</i>} <i>pl.</i> I'll kill you; IV.93 [¹ lagot ^{\S} , len] <i>G neg. subjv.</i> 1 ^{<i>st</i>} <i>pl.</i> let's not kill; IV.96 [gə ¹ t ^{\S} ol] <i>G impv. m.sg.</i> kill!
gabrā	gabbər gobrā gobrānā gobrā qadmāyi	['gævrɔ] <i>n.</i> man; CM gabra id.; V.18 ['govvṛ] man (contextual); V.7 ['govrɔ] <i>pl.</i> men CM gubria id.; II.25, III.20, IV.81, IV.105, IV.124 [govr'rɔ:nɔ] men; I.2 [govrbqad'mɔ:ji] <i>prop. n.</i> the Primal Man; CM gabra qadmaia but pronounced in NM as if plural, i.e. gubria qadmaiia; I.8
gač		[gætʃ] <i>n</i> . gypsum; Persian <i>gač</i> id.; IV.4
gāmeš		[ˈɡɔːmɛʃ] <i>n.</i> buffalo; Middle Persian <i>gāwmēš</i> via PCM gamiš id.; IV.124, IV.142

ganzibi	ā ganzi <u>b</u> ri	[gan ¹ zɛʊrɔ], [gan ¹ zɪvrɔ] <i>n.</i> ganzibra, the highest rank currently occupied by Mandaean priests today (below <i>riš</i> <i>āmmā</i> , q.v.); CM ganzibra id.; II.21, II.25, IV.56, IV.57, IV.62, IV.76, IV.77, IV.79, IV.81, IV.83, IV.84, IV.87 (x2), IV.88, IV.90, IV.91, IV.92 (x2), IV.94, IV.95, IV.96, IV.104, IV.107, IV.109, IV.110, IV.112, IV.118, IV.119, IV.120, IV.121, IV.122, IV.130, IV.133, IV.140, IV.145, IV.147, IV.148 [gæn ¹ zɪvri] <i>indef.</i> a ganzibra; II.3
gappā		['gæppɔ] n. feather; wing; CM kanpa / ganpa id.
gāw	gāwwi gāwwa gāwwu	[gou] <i>prep.</i> in(to); about; through; according to; (made) from; NM $g\bar{a}w$ assumes many of the functions of CM b -; CM gaua 'inside,' cs. gu 'in;' I.4, I.8, I.9 (x2), I.10, I.14, II.4, II.4, II.6, II.7, II.8, II.21, II.23, II.28, III.1, III.2, III.7, III.12, III.16, III.18, III.20, IV.1, IV.2, IV.4, IV.7, IV.15, IV.86, IV.99, IV.103, IV.118, IV.141, IV.144, IV.145, V.20, V.26, VIII.16 (x2), IX.6 ['gouvi] <i>prep.</i> + 3 m. sg. in him; IV.8, IV.20, IV.30, X.1 ['gouwa] + 3 rd f.sg. in her; I.10, VIII.13 ['gouwu] + 3 rd pl. in them; I.11, IV.9, IV.28
gāwi		['gɔ:vi] adj. strong; colloquial Arabic gawī id.; IV.4, IV.43
genzā 1	Genzā Rabbā Genzā Yaminā	 <i>n.</i> treasure; CM ginza 1 id. ['gɛnza 'rabbɔ] <i>prop. n.</i> the Great Treasure; CM ginza rba id.; I.9 ['gɛnza jɛ'mi:nɔ] <i>prop. n.</i> the Right Ginz, the first portion of the <i>Genzā Rabbā</i>; CM ginza iamina id.
genzā 2	genz	['gɛnzɔ] <i>adv</i> . very; many; much; CM ginza 2 id.; IV.105 [gɛnz] very, much (contextual); I.14, I.4, I.6, II.20, II.20, V.12, V.12, IX.1, IX.2
germā		['gɛɪmɔ] <i>n</i> . bone; CM girma id.
geš	geš-di geš-du geš-dan	[gIʃ] <i>n</i> . entirety, all; possibly from \sqrt{q} - <i>š</i> - <i>š</i> 'to collect, gather,' see Häberl, "Relative Pronoun;" I.3, I.4, I.15, II.32; IV.4, IV.7 (x2), IV.27, IV.93, IV.105, V.5, V.10, VIII.14 ['gɛʃdi] + 3 rd m. sg. all of it; V.5 ['gɛʃdu] + 3 rd pl. all of them; III.23, IV.80, VI.6, V.9 ['gɛʃdan] + 1 st pl. all of us; I.14, II.4, II.22

320	8. Lexicon
gəlālā	[gə'lɔ:lɔ] n. stone; CM glala id.; IV.4
gəlaltā	[gə'laltə] n. stone; CM glalta id.
gəțānā	[gəˈt [°] ɔ:nə] <i>adj.</i> short; CM gṭana id.
gird	[gi.id] <i>n</i> . round; Persian <i>gird</i> id.
Giyālin	[gijp ^l lin] <i>prop. n.</i> the Gilānis, a Mandaean family from Basra that settled in Khorramshahr; II.23
gobrā qadmāyi	See gabrā
gomlā gomlān <u>k</u> on	['gomlo] <i>n</i> . camel; CM gumla id. [gom'lon χ on] <i>pl</i> . + 2 ^{<i>nd</i>} <i>pl</i> . your camels; IV.124
guš '-b-d	<i>v.i.</i> take a look, keep an eye out; Persian <i>guš</i> 'ear,' cf. Persian <i>guš kardan</i> to listen
guš ə <u>b</u> adyon	[$^{1}gus \varepsilon_{1}vadjon$] <i>G perf.</i> 3^{rd} <i>pl.</i> they took a look; IV.149, IV.150
guš əbud guš qābden	[gu $\int \vartheta'$ wud] <i>G impv. f.sg.</i> you keep an eye out; IV.46 [gu $\int '$ qpvdɛn] <i>G imperf. 3rd pl.</i> they are taking a look; IV.66
guțluktā '-h-b əhabni guțluktā	<i>v.i.</i> to be massacred [a'hovni gAt'lA χ tɔ] <i>G perf.</i> 1^{st} <i>pl.</i> we were massacred; 1.14
guțrā	['gʌt ^s rɔ] <i>n</i> . smoke; CM guṭra 1 id.

Ġ

ģer	[KeI] <i>adv.</i> outside; Arabic <i>ġayr</i> 'unlike, different;' IV.20, IV.124
ġor '-ḇ-d ġor əḇad	v.t. to understand (+ -d- s.o.); Persian $\dot{g}awr$ 'bottom, depth,' cf. Persian $\dot{g}awr kardan$ 'to meditate, reflect' ['BOJ ε_1 vad] <i>G perf.</i> 3^{rd} <i>m.sg.</i> he understood; V.27
ģubār	[кл'bɔ1] <i>n</i> . dust; Arabic <i>ġubār</i> id.
ģubšā	['ʁʌbʃɔ] <i>n.</i> twilight; Arabic <i>ġubša</i> 'twilight' (before dawn); IV.48
ġușnā	['ʁʌsʿnɔ] n. root; Arabic <i>ġuṣn</i> id.

Η

h-d-r		v.i. to turn; CM HDR 1 id.
h-l-p	halpat	<i>v.t.</i> to pass over; CM HLP id. ['halfat] <i>G perf.</i> 3 ^{<i>rd</i>} <i>f.sg.</i> it passed over; IV.141
h-p-r		v.t. to dig; CM HPR id.; Tab. 1.5
	qəhāper	['qp:fɛ1] <i>G imperf.</i> 3 rd m.sg. he digs; IV.127
	qəhaper	[qa ^l fɛ1] <i>G imperf.</i> 1 st sg. I will dig; IV.125
h-r- <u>b</u>		v.s. G to be ruined, destroyed, C to ruin, destroy; CM
		HRB id.; see also haruba
	həre <u>b</u>	[hə'ro:] G perf. 3^{rd} m.sg. it was destroyed; IV.27, IV.29
	láhəre <u>b</u>	['lahero] G perf. 3 rd m.sg. it was not destroyed; IV.67
	qəhāreb	[qə'hɔ:ro] <i>G imperf.</i> 3^{rd} <i>m.sg.</i> it is destroyed; IV.12, IV.34, IV.40, IV.78
	láqhāreb	['laqhəro] <i>G neg. imperf.</i> 3 rd m.sg. it will not be
	1 -	destroyed; IV.79, IV.71
	lā hāreb	[lo 'horro] G neg. subjv. 3 rd m.sg. may it not be destroyed;
	-	IV.24
	lahribi	['læh ₁ rivi] C neg. perf. 3^{rd} m.sg. + 3^{rd} m.sg. he has not
	_ -	destroyed it; IV.71
	mahrebli	[mah'revli] C imperf. 3^{rd} m.sg. + 3^{rd} m. sg. suff. he
	_	destroys it; IV.10, IV.13, IV.41
	qəmahrebāt	[qəmahrɛ'wət] C imperf. 2^{nd} m.sg. + 3^{rd} m.sg. you
	<i>q</i> • • • • • • • • • • • •	destroy; IV.58
	qəmahrəbāti	[qəmaı'wə:ti] C imperf. 2^{nd} m. sg. + 3^{rd} m. sg. you destroy
	quintantegan	it; IV.59
	láqmahrem	['laq'mahrɛm] <i>C neg. imperf.</i> 1 st sg. I won't destroy; IV.61
h- <u>t</u> -r		<i>v.i.</i> to rejoice, show pride, be glad; CM HTR id.; see also
-		hitrā '-b-d
	həternān	[hɛ' θ ɛınon] G pass. ptc. + 1^{st} sg. I am glad; I.4
	ehter	[' $\epsilon h \theta \epsilon I$] tG perf. 3 rd m.sg. he was delighted; V.13
	-	
$h-w-W/_{v}$		v.t. to be; CM HWA id.; see also ehl- 'to have,' həyānā
·····y		$h-w^{-w}/_{v}$
	həwā	[hə'wo:] G perf. 3^{rd} m.sg. he was; II.2 (x2), II.3, II.8, II.11,
		II.23, II.32, III.8, III.20, V.1, V.8, V.9, VI.1, VI.3, VII.2, VII.6, VIII.1,
		IX.1, X.6

	həwat həwit	[hə ¹ vat] <i>G perf.</i> 3^{rd} <i>f.sg.</i> she was; 1.8, II.14, II.24, III.2, VII.1 [hə ¹ vit] <i>G perf.</i> 1^{st} sg. I was; II.1 (x2), II.16, III.5 (x2), III.4,
	nəwn	$[13 VH]$ G perf. 1 sg. 1 was, $11.1 (x_2)$, 11.16 , $11.5 (x_2)$, 11.4 , III.14 (x2)
	həwini	$[ho^{1}vi:ni]$ <i>G perf.</i> 1^{st} <i>pl.</i> we were; 1.11, II.4, II.22, III.7
	láqhāwi	['laq,hovi] G neg. imperf. 3^{rd} m.sg. let it not be; IV.97,
	Iuqnum	VIII.12
	hawinā	[hɛ ^l vi:nɔ] <i>G imperf.</i> 1 st sg. I am; III.16
	hawén	[ha ^l vɛn] G subjv. 1^{st} pl. (contextual) may we be; I.14
	həwi	$[ha^{\dagger}vi:]$ <i>G impv. m.sg.</i> be; V.16
$h - \frac{W}{y} - r$		<i>v.t.</i> to wash, lit. to make white; CM HWR id.
·y -	hawwer	['hawwei] D perf. 3 rd m.sg. he washed (himself); VIII.13
h- ^w / _y -ț		v.t. to sew; CM HUŢ id.
$h-z-^{w}/_{y}$		<i>v.t.</i> to see, to seek; CM HZA id.; see also <i>radpā</i> $h-z^{-w}/_{y}$
	həzā	[həˈzɔ:] <i>G perf.</i> 3 rd m.sg. he saw; VIII.3
	həz'	[hɛz] G perf. 3 rd m.sg. (contextual) he saw; VIII.7, VIII.14,
		IX.7
	lahzi	['lɛhzi] <i>G perf.</i> \mathcal{J}^{rd} <i>m.sg.</i> $+ \mathcal{J}^{rd}$ <i>m.sg.</i> he didn't see him; IX.6
	həzit	[həˈzit] G perf. 1 st sg. I saw; V.21
	həzitu	[hə'zi:tu] G perf. 1^{st} sg. + 3^{rd} pl. I saw them; III.19
	lahzit	['lɛhzit] G neg. perf. 1 st sg. I have not seen; IV.53
	həzon	[hɛ'zon] G perf. 3^{rd} pl. (contextual) they saw; X.3
	həzonna	[hə'zonna] <i>G perf.</i> $3^{rd} pl. + 3^{rd} f.sg.$ they saw her; IV.142
	heztonne	[hes'tonne] G perf. $2^{nd} pl. + 1^{st} sg$ you saw me
	həzini	[həˈzi:ni] <i>G perf. 1st pl.</i> we saw; I.14, III.7
	həzin	[həˈzin] G perf. 1 st pl. (contextual) we saw; III.8
	qəhāzen	[qə'hp:zen] G imperf. 3^{rd} pl. they see; IV.45, IV.138, IV.67
	qəhāzyā	[qə'hɒzjɔ] G perf. 3 rd f.sg. she sees; IV.47, IV.49
	hazin	[hɛ'zin] G subjv. 1^{st} sg. I see; IX.3
	hazinu	[hɛ'zi:nu] G imperf. 1^{st} sg. + 3^{rd} pl. I see them; V.9
	qahzin <u>k</u> on	[qah'zin χ on] G imperf. I^{st} sg. $+2^{nd}$ pl. I see you; I.4
	qəhazelli	[qa'zɛlli] G imperf. 3^{rd} pl. + 3^{rd} m.sg. they see him; IV.7
	qəhazén	[qəha'zɛn] G imperf. 1^{st} pl. (contextual) we see; II.14, III.10
	hazén	[ha ¹ zɛn] G imperf. 1^{st} pl. (contextual) let's see; IV.70
	həzi	[$h \Rightarrow 'zi$:] G impv. m.sg. seek; IV.94
hā		See <i>āhā</i>
1-66		$[1, C, C] \rightarrow (1, 1) \rightarrow (C, C, 1)$

323

hafif

['hæfif] *n*. thin; Arabic *xafif* id.

324		8. Lexicon
Haft Taj	ppé	[₁ hæftæp'pe] prop. n. Haft Tepe, the "Seven Hills," is an archaeological site in Khuzestān, located midway between Susa and Chogha Zambil; III.6
hafté		[hæf ⁴ te:] <i>n</i> . week (f.); Persian <i>hafte</i> id.; IV.70, VIII.6, VIII.10, X.9
hālā		['hɔ:lɔ] n. sand; CM hala 1 id.
ham		[hæm] adv. also, even; Persian ham id.; III.24, VII.6
hām		[hom] <i>n</i> . father-in-law
hamā <u>t</u> ā		[hæ ¹ mɔ:θɔ] <i>n</i> . mother-in-law
hamim		[hæˈmim] <i>adj.</i> warm; Arabic <i>ḥamim</i> id. via Persian <i>hamim</i> id.
hamšā		[^t hæm∫ɔ] <i>quant.</i> five; CM hamša id.; III.18
Harrān (Gāwe <u>t</u> ā	[,haraŋ'g ^w e:θɔ] <i>prop. n.</i> a legend detailing the events surrounding the Mandaeans' settlement in the Arsacid empire; CM haran gauaita id.
haru <u>b</u> ā	haru <u>b</u>	<i>adj.</i> destroyed; CM haruba id. [haˈruʊ] it was destroyed (contextual); IV.7
Harvard	University	[haɪ'vəɪd junə'və sıti] <i>prop. n.</i> .Harvard University in Cambridge, MA; 1.4
hā <u>t</u> ā	ahwāṯā hāṯ bellā hāṯ eṯṯā	['hɔ:θɔ] n . sister [æh'wɔ:θɔ] pl . sisters [,hɔθ 'bɛllɔ] n . brother-in-law [,hɔθ 'ɛθθɔ] n . sister-in-law
ha <u>tt</u> ā	ha <u>tt</u> i	['hæθθɔ] <i>adv.</i> anew; CM hadta (pl. hatia) id.; IV.17, IV.41, IV.42, IV.42, IV.148 ['hæθθi] <i>indef.</i> a new (thing); IV.117, VI.6
hatieqā haṭāyā		[hæˈtiĕqɔ] <i>adj.</i> old (inanimate objects); CM hatiqa id. [haˈtˤɔːjɔ] <i>n.</i> sin; CM haṭaiia id.; 1V.97

hayo <u>t</u> ā	hayo <u>t</u>	[haˈjo:θɔ] <i>n</i> . survival; CM haiuta 3 'life' [haˈjoθ] survival (contextual); 1.4
hāzér t-	m-m hāzér tammā	<i>v.i.</i> to appear before, be present; Arabic <i>hādir</i> 'present' via Persian <i>hāzér</i> id., cf. Persian <i>hāzér šodan</i> 'to appear before, be present.' [hb'zɛ1 tam'mɔ:] <i>G perf.</i> 3 rd m.sg. he appeared, became present; IV.91
hemanı	ı <u>t</u> ā hemanu <u>t</u> an	[hema'nu: θ o] <i>n</i> . faith, belief; CM haimanuta id. [hema'nu: θ æn] + 1^{st} pl. our faith; 1.8
hem		[hem] <i>pron.</i> which
hemdā		['hɛmdɒ] <i>pron.</i> (interrogative) when; CM *hai mn 'dan 'which of time;' IV.122
hemke		['hɛmke] <i>prep.</i> like; PCM hamkia or hamka id.; IV.49
hemma	n	['hɛmmæn] <i>indef. pron.</i> whoever; IV.140
hewyā		['hɛvjɔ] <i>n</i> . snake; CM hiuia id.
Heyyi l	Māre	['hɛɪji 'mɔ:re] <i>prop. n.</i> epithet of God, lit. life, my lord; CM hiia marai id.; 1.9
Heyyi l	Rabbi	['hɛɪji 'ræbbi] <i>prop. n.</i> Great Life, one of the epithets of God; nb. "life" is plural; CM hiia rbia id.; I.1, I.14
hədādā	be-hdādā	[hɛˈdɔːdɔ] <i>pron.</i> one another; CM hda had id.; I.4 [bɛhˈdɔːdɔ] <i>pron.</i> to one another; IV.148, IV.150
hənā	hənāye luhnāye	[hə ['] nɔ:] <i>dem. pron.</i> here; Arabic <i>hunā</i> id.; II.21, IV.126 [hə ['] nɔ:je] + 3^{rd} <i>m.sg.</i> it is here [luh ['] nɔ:je] <i>neg.</i> + 3^{rd} <i>m.sg.</i> it is not here; IX.10
həwārā	həwarānā	['h ^w ɔ:rɔ] <i>adj.</i> white; CM huara id. [huwɛ'rɔ:nɔ] <i>pl.</i> white; IV.31
həyānā	h-w- ^w /y hawén həyānā	<i>v.i.</i> to survive, lit. to be alive; see also <i>həyānā t-m-m</i> [hæ'vɛn 'hʲɔ:nɔ] <i>G subjv. 1st pl.</i> (contextual) we will survive; 1.14

həyānā		<i>v.i.</i> to survive, lit. to stay alive; see also $h = \sqrt{ana} h - w - \sqrt[w]{y}$ ['tæmmini 'h ^j o:no] <i>G perf.</i> 1 st pl. we survived, I.14
hič		[hitʃ] <i>adv</i> . any, no; Persian <i>hič</i> id.
hi <u>t</u> rā		['hıθra] <i>n</i> . rejoicing, merrymaking, pomp; CM hitra id.
hiţrā '-b	-d hi <u>t</u> rā ə <u>b</u> adyon	<i>v.i.</i> to rejoice, celebrate; cf. CM hitra 'rejoicing, merrymaking, pomp' ['hɪθrɑ 'vɑdjon] <i>G perf.</i> 3 rd pl. they rejoiced; IV.81
hitruktā		[hιθ'rʌχtɔ] <i>n</i> . happiness; cf. CM hitra 'rejoicing, merrymaking, pomp'
Hibel		['hi:bɛl] prop. n. Abel; CM hibil id.; 1.9
hidā	hid	['hi:dɔ] <i>pron.</i> she (f.); CM h ^c id.; II.26 [hid] she (contextual); II.31
hilā	hil hilānnon	[^{$hh:l_{0}$] <i>n</i>. power; CM hila id. [hil] power (contextual); I.14 [hi^{l}lpnnon] <i>pl</i>. + 3^{<i>rd</i>} <i>pl</i>. <i>cop</i>. they are powerful; V.12}
hoķā		['hoːɣɔ] <i>n</i> . scratch; CM hauka 'irritation, scab, itching' from CM HKK / HUK 'to itch, irritate'
hol '- <u>b</u> -o	d	['hol] n. push; Persian hol id.
hol '- <u>b</u> -c	d	<i>v.t.</i> to push; cf. Persian <i>hol dādan</i> id.
<i>honinā</i>	honini	[hoˈniːnɔ] <i>adj.</i> small, little; CM hunina id. [hoˈniːni] <i>indef.</i> a small (thing); X.2
honni		['honni] <i>pron.</i> they; CM hinun id.
horinā	horettā	[ho ['] ri:nɔ] <i>adj</i> . other; second; next, following; CM hurina id.; VII.8 [ho ['] rɛttɔ] <i>f</i> . other; second; next, following; CM hurintia id.; III.2, VIII.6, X.9
huwi		['hu:wi] <i>pron.</i> he; CM hu id.; IV.35, IV.44, IV.65, IV.75, IV.88, IV.104

Ĥ

<u></u> <i>ḥ-<u></u><u></u>_y-<u></u>^w/_y</i>		<i>v.t.</i> C to keep, put away; Arabic $h-b^{-2}$ II 'to conceal' or $h-f^{-w}/_{v}$ IV id., cf. CM HBA id.;
	ahbi	['aħvi] C perf. 3^{rd} m.sg. he put away; V.18, X.8
	maḥbi	['mɑħvi] C imperf. 3^{rd} m.sg. he conceals; I.13
<u> </u> <i>ḥ-d-<u>t</u></i>		v.t. D to speak, discuss; Arabic h-d-t II id.
	<u>hatt</u> at	[^t ha θ bat] D perf. 3^{rd} f.sg. she spoke; II.19, II.20
	qəmḥadde <u>t</u>	$[qm^{h}add\epsilon\theta] D imperf. 3^{rd} m.sg$ he speaks
	qəm <u>hatta</u>	[qm ha $\theta \theta \mathfrak{d}$] <i>D imperf.</i> 3^{rd} <i>f.sg.</i> she speaks; II.31
	qəmḥa <u>tt</u> '	$[qm^{\dagger}ha\theta] D$ imperf. 3^{rd} f.sg. (contextual) she speaks; II.15
	qəmḥadeṯnā	[qmħaˈdɛθnɔ] D imperf. 1 st sg. I speak
	əmḥade <u>t</u>	$[m\hbar a'd3\theta] D subjv. 1^{st} sg.$ (contextual) I speak; I.7, II.17
	qəm <u>hatt</u> en	[qm/ha $\theta \theta en$] D subjv. T sg. (contextual) Tspeak, 1.7, 11.17 [qm/ha $\theta \theta en$] D imperf. 3^{rd} pl. they speak; II.21, IV.89, IV.150 [mə/ha $\theta \theta en$] D subjv. 3^{rd} pl. they speak; II.32
	məḥatten	[məˈħaθθɛn] <i>D subjv. 3rd pl.</i> they speak; II.32
	láqəmḥa <u>tt</u> en	[¹ laqm,haθθεn] <i>D neg. imperf.</i> 3 rd pl. they don't speak;
		II.21, II.23
	əmḥa <u>t</u> t'	$[m^{h}a\theta] D subjv. 3^{rd} pl.$ they speak; II.25
	əmḥattetton	$[m^{h}a\theta^{\theta}etton]$ D subjv. 2^{nd} pl. you are discussing
	əmḥadṯetton	[mħad' θ ɛtton] <i>D</i> subjv. 2 nd pl. you are discussing; I.4
	qəm <u>hatt</u> én	$[qm\hbar a\theta'\theta \epsilon n] D imperf. 1st pl. (contextual) we speak; II.22$
	əm <u>hatt</u> én	$[m\hbar a\theta'\theta \epsilon n] D subjv. 1^{rd} pl.$ (contextual) let's talk; IV.81
	<i>ḥade<u>t</u>on</i>	[ħaˈdɛθon] <i>D impv. pl.</i> discuss; 1.5
ḥ-ṣ-l		v.t. D to obtain, get; Arabic <i>h-s-1</i> II id.
	əmḥaṣṣelnā	[m hass s elno] <i>D imperf.</i> 1^{st} sg. I would get; III.16
<u>ḥadid</u>		[ħa ^l did] <i>n.</i> iron (bullion); Arabic <i>ḥadīd</i> 'iron;' V.3, V.11,
		V.12, V.12, V.26
	ḥadid-di	$[\hbar a' diddi] + 3^{rd} m.sg.$ his iron (bullion); V.6
	<u> ḥadidānā</u>	[ħadi ¹ dɔ:nɔ] <i>pl</i> . iron bullion; V.10
	<u> ḥadidānak</u>	[hadi'də:na χ] <i>pl.</i> + 2 ^{<i>nd</i>} <i>m.sg.</i> your iron bullion; V.7, V.29
<i>Ḥadiyye</i>	,	[ħa'di:je] prop. n. Hadiyye, the bride at the wedding in
		Khorramshahr; II.10
<u> h</u> ākem		['hp:kem] n. governor; Arabic hākim 'judge, governor,'
		cf. PCM hikim id.; IV.30, IV.71, IV.72, IV.75, IV.79, IV.83,
		IV.86, IV.92, IV.98, IV.103, IV.103, IV.104, IV.107, IV.109, IV.110,
		IV.83, IV.87

ḥakimā d-Šuštar ['ħɒ:ki,mɔ ɛd'ʃustaɪ] the governor of Shushtar; IV.65

<u>ḥakyāṯā</u>	[ħak'jɔ:θɔ] n. story, tale; cf. Arabic <i>ḥikāya</i> id.; IV.1
<u>ḥammām</u>	[ħam ¹ mɒm] <i>n.</i> bath(house); Arabic <i>ḥammām</i> id.; VIII.1, VIII.4, VIII.5, VIII.8, VIII.15, VIII.16
ḥammām d-r- ^w / _y ḥammām dərā	<i>v.i.</i> to bathe, take a bath [ham'mpm dɛ'rɔ:] <i>G perf.</i> 3^{rd} <i>m.sg.</i> he bathed; VIII.7
<u>ḥašiš</u>	[hæ¹∫iʃ] n. hashish; Persian <i>hašiš</i> id. via Arabic <i>ḥašīš</i> id.; VII.1, VII.6
<u>ḥatta</u>	['ħatta] adv. even; Arabic <i>ḥatta</i> 'until, even;' II.21
ḥiṣān	[ħr's ^s ɒn] <i>n</i> . horse; Arabic <i>ḥiṣān</i> id.; IV.65
<i>ḥizām</i>	[ħı'zɒm] n. belt, girdle; Arabic <u>hizām</u> id.; VIII.14, VIII.15
ḥədādā ḥədadānā	[ħəˈdɔ:dɔ] <i>n</i> . blacksmith; cf. Arabic <i>ḥaddād</i> id. [ħədɛˈdɔ:nɔ] <i>pl.</i> blacksmiths; IV.130
ḥusən	[^h ʌsn <code>] n. beauty; Arabic <i>husn</i> id.; IV.53</code>

Ι

ibā		['i:vɔ] <i>n</i> . cloud, mist; CM aiba id.; IV.49, IV.50
idā	ide	['i:də] <i>n.</i> hand (f.); CM 'da id. ['i:de] + 1^{st} sg. my hand; VI.2
idā m- <u>t</u> -	^{w/} y id mə <u>t</u> i menne	<i>v.t.</i> to release, desist (obj. + <i>men</i>); CM 'da 'hand' + $m-\underline{t}^{-w}/\underline{y}$ 'to lift;' cf. Persian <i>dast bar dāštan</i> id. [id mɪ'θi: 'mɪnne] <i>G impv. m.sg.</i> lay off me, leave me alone; IV.108
illi		['Illi] <i>rel. pron.</i> that, which; who; used to indicate a non-restrictive clause dependent upon a grammatically definite noun; colloquial Arabic <i>illi</i> id.; I.3, I.5, I.6, I.10, III.12, III.13
inā	ine	['i:nɔ] <i>n</i> . eye (f.); CM ain, st. abs. and cs. 'in id. ['i:ne] + $I^{st} sg$. my eye; VI.2
Inglizí	Ingliz	[ɪŋglɪ'zi:] <i>adj.</i> English; II.14, II.18 (x2) [ɪŋ'gliz] <i>pl.</i> (collective) British; III.2
Irān		[i'ron] <i>prop. n.</i> Iran; III.2
Irāq		[I'rɒq] <i>prop. n.</i> Iraq; II.21, II.23, II.25

J

j-m-d		<i>v.i.</i> to freeze; Arabic <i>j-m-d</i> id.
jadrā		[¹ dʒaðrə] <i>n</i> . root; Arabic <i>jadr</i> id.
jami ^c a	jami ^c a-də <u>k</u> on	[$\mathfrak{G}\mathfrak{E}^{I}mi$?a] n . entirety; total, all; Arabic <i>jamī</i> ^c id.; IV.15, IV.98 [$\mathfrak{G}\mathfrak{E}\mathfrak{m}i^{I}jad\mathfrak{d}_{I}\chion$] + 2 nd pl. all of you; IV.124
jangal		[cʃæŋˈgɑl] n. forest; Persian jangal id.
jawāb	jawāb-de	[&æ'wob] n. an answer; Arabic <i>jawāb</i> id. via Persian <i>javāb</i> id. [&æ'wobde] + 1 st sg. my answer
jawāb -	<i>p-h-<u>b</u></i>	<i>v.t.</i> to answer; Arabic <i>jawāb</i> 'answer' via Persian <i>javāb</i>
	jawāb-d' ahbita	id.; cf. Persian <i>javāb dādan</i> id. [d_{xe}^{w} wpbdæh ₁ vi:tə] <i>G perf.</i> 1 st sg. + 3 rd f. sg. I answered
	jawāb əhab	her; II.19, II.20 [&a'wob hav] <i>G perf.</i> 3 rd m.sg. he responded; V.12
jedde		['dsedde] n. grandmother; Arabic jadda id.
jeddu	jedduye	['&ɛddu] n. grandfather; Arabic <i>jadd</i> id. [&ɛd'du:je]+ 1 st sg. my grandfather
jihel	jihelā jihelā-du	[¹ dʒi:hɛl] <i>n.</i> boy, child; Arabic <i>jāhil</i> 'ignorant,' here in the sense of 'ingénu;' II.1, II.1, II.16, III.14, X.6 [₁ dʒihɛ'lɔ:] <i>pl.</i> children; II.4, II.15, II.31, IV.124, IV.139 [₁ dʒihɛ'lɔ:du] <i>pl.</i> + 3^{rd} <i>pl.</i> their children; IV.148
jisrā	jisər jisrā d-Šuštar jisri	[¹ \\$\dotsr5] <i>n.</i> bridge; Arabic <i>jisr</i> id.; IV.12, IV.19, IV.78 [¹ \\$\dotsrf] <i>n.</i> bridge (contextual); III.20, IV.39, IV.42 [¹ \\$\dotsrad,\\$\u00edustaI] <i>prop. n.</i> Shushtar bridge; IV.68 [¹ \\$\dotsrif] <i>indef.</i> a bridge; III.21, IV.2
Joḥa		[¹ �oħa] <i>prop. n.</i> Joh, aka Nasrettin Hoca or Mullah Nasreddin; X.1, X.4, X.5, X.9, X.13, X.16, X.17
Κ

k- <u>b</u> -š	kabšellu	<i>v.t.</i> to subdue; CM KBŠ id. [kav ¹ ʃɛllu] <i>G imperf.</i> $3^{rd} pl. + 3^{rd} pl.$ they subdue them; IV.14
k-d- <u>b</u>	<i>kəde<u>by</u>e</i>	<i>v.t.</i> to write; CM KDB id.; see also $k \partial d\bar{a} b \bar{a}$ [kəˈdɛvje] <i>G pass. ptc.</i> + 3^{rd} <i>m.sg.</i> it is written; 1.9
k-d-r		<i>v.s.</i> to be heavy, be a burden; CM KDR id.
k-m-r	lákamri kammar kamarni	<i>v.t. D</i> to turn, return (trans.); <i>tD</i> return (intrans.); CM KMR id.; ['lakambri] <i>D neg. perf.</i> 3^{rd} <i>m.sg.</i> he didn't return it; X.12 ['kammaJ] <i>tD perf.</i> 3^{rd} <i>m.sg.</i> he returned; X.2 [ka'maIni] <i>D perf.</i> 1^{st} <i>pl.</i> we returned, we went back; III.15
k-t-p	kotp [°]	<i>v.t.</i> to trap; cf. Arabic <i>k-t-f</i> I 'to fetter, shackle' [kʌtf] <i>G impv. m.sg.</i> trap; IV.55
kabdā		[ˈkæbdɔ] n. liver; CM kabda id.
kadirā		[kɛˈdiːrə] <i>adj.</i> heavy; CM kadira id.
kāfér	kāferān	[kɒˈfɛr] n. infidel, unbeliever; Arabic <i>kāfīr</i> via Persian <i>kāfer</i> id. [kɒfɛˈrɒn] <i>pl.</i> unbelievers; III.12
kalbā		[ˈkælbə] <i>n</i> . dog; CM kalba id.
kammā		['kæmmo] inter. pron. how; CM kma id.; IV.146
kandi		['kændi] adv. yet, still, again; CM kandia id.; IV.78
Kanjar		[ˈkænʤaɪ] <i>prop. n.</i> father of the bride at Khorramshahr; II.10
karsā	karsi karse	['kaisə] <i>n</i> . stomach, belly; CM karsa id. ['kaisi] + 3^{rd} <i>m.sg</i> . his stomach; X.2 ['kaise] + 1^{st} <i>sg</i> . my stomach; VI.2

332		8. Lexicon
kaspā		['kæspo] n. silver; CM kaspa; III.5
kasyā		['kɛsjɔ] adj. hidden, secret; CM kasia id.; IV.85, IV.121
ka <u>t</u> kami	mā	[kæθ ₁ kæmmɔ] <i>adj.</i> so much; CM k <u>d</u> kma id.; 1.14
<i>ke</i> 1		[ke] <i>rel. pron.</i> who; Persian <i>ke</i> id.; 1.11, 1.13, III.12, III.20, IV.14, IV.15, V.9, V.26, IX.2
ke 2		[ke] <i>rel. pron.</i> that (indicating a restrictive clause; preceded by <i>-i</i>); Persian <i>ke</i> id.; III.19, IV.83, V.20, V.26
<i>ke</i> 3		[ke] <i>conj.</i> when; Persian <i>kay</i> id.; II.1 (x2), II.2
ke 4		[ke] <i>conj.</i> because; II.20
kef		[kef] <i>n</i> . good humor; Arabic <i>kayf</i> id.
kef '-b-d kef əbadyon		<i>v.i.</i> to celebrate, have fun; Arabic <i>kayf</i> 'good humor;' cf. Persian <i>kayf kardan</i> 'to enjoy oneself, have fun' [kef \exists 'vadjon] <i>G perf.</i> 3^{rd} <i>pl.</i> they celebrated; IV.69, IV.72,
	kef ābden	IV.80 [kef 'pvdɛn] <i>G subjv. 3rd pl.</i> they celebrate; IV.72
kədā <u>b</u> ā	kədā <u>b</u>	[kəˈdɔːwɔ] <i>n.</i> book (f.); CM kdaba id. [kəˈdɔʊ] book (contextual); 1.9, III.16
kərāyā	kərāyi kərāye	[kə'rɔ:jɔ] <i>n</i> . leg (f.); CM kraia 'foot, leg, paw' [kə'rɔ:ji] + 3^{rd} m.sg. his leg; VII.2b, VII.3, VII.4 [kə'rɔ:je] + 1^{st} sg. my leg; VI.2, VII.5, VII.9
kief		[kiĕf] <i>n</i> . good humor; Arabic <i>kayf</i> id.
klā <u>t</u> ā	kla <u>t</u> āyā	[ˈkl̥ɔ:θɔ] <i>quant.</i> three; CM tlata id.; III.1 [kl̥ɛˈθɔ:jɔ] <i>adj.</i> third; CM tlitaia id.; VIII.9
ko <u>k</u> bā		['koχwə] <i>n</i> . star; CM kukba id.; 1.9
kol	kol waxt kol mā	[kol] <i>quant.</i> each, every; (+ <i>pl.</i>) all; CM kul id.; I.4, I.8, I.11, I.11, III.4, IV.34, IV.40, IV.59, VIII.10 ['kolwaχt] <i>adv.</i> always; I.13 ['kolmɒ] <i>indef. pron.</i> whatever; VIII.12

komák '-b-d komák abdell'	<i>v.t.</i> to help, give aid; Persian <i>komak</i> 'aid,' cf. Persian <i>komak kardan</i> 'to help' [ko'mak av ₁ dɛlli] <i>G subjv.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>f.sg.</i> they help (Russia); III.2
konfərens	[kõfəˈrɛ̃s] <i>n</i> . conference; French <i>conférence</i> 'conference;' I.3, I.5
kušțā	[kʉʃ'tɔ:] <i>n.</i> truth; CM kušța; 1.9

L

l- <u>b</u> -š	lə <u>b</u> š lə <u>b</u> eš lə <u>b</u> ašni	v.t. to wear, put on (a garment); CM LBŠ id. [lɛfʃ] <i>G perf.</i> 3^{rd} <i>m.sg.</i> (contextual); VIII.15 [lɛ'vɛʃ] <i>G subjv.</i> 3^{rd} <i>m.sg.</i> he put on; VIII.2 ['lʰɑʃni] <i>G perf.</i> 1^{st} <i>pl.</i> we put on (our clothes); III.10
l- <u>k</u> -ț	lə <u>k</u> t² qəla <u>k</u> ti	<i>v.t.</i> to seize, take; hold; CM LGT id. $[ls\chi t^{S}] G perf. 3^{rd} m.sg. + obj.$ he seized; IV.76, IV.83 $[q\varepsilon^{l} \varepsilon \chi t^{S} i] G imperf. 3^{rd} m.sg. + 3^{rd} m.sg.$ he takes him; V.18
lābod		['lɔ:bod] <i>adv.</i> necessarily, certainly; Arabic <i>lā budda</i> 'no escape' via Persian <i>lābod</i> 'necessarily, certainly;' IV.72, IV.86, IV.106
lājur		[ləˈdʒuɪ] <i>conj.</i> since; IV.71, IV.118
lebbā		['lɛbbɔ] <i>n</i> . heart; CM liba id.
lek		[lɛk] adv. thither; CM lka id.; III.11
ler		[le1] n. spittle; Persian layr id.
ler '- <u>k</u> -l	ler ə <u>k</u> al	<i>v.i.</i> and <i>v.t.</i> to turn around; cf. Persian <i>layr</i> 'spittle' [le1 ε ' χ al] <i>G perf.</i> 3^{rd} <i>m. sg.</i> he turned around; IV.104
lešānā	lešān	[lɛˈʃɔːnɔ] <i>n.</i> tongue, language; CM lišana id. [lɛˈʃɔn] (contextual) language; I.5
lilyā		['lɪljə] <i>n</i> . night; CM lilia 1 id.
lo	-lolo	[lo] <i>conj.</i> or[lo] <i>correlative conj.</i> either or
loğrā		['loɣərɔ] <i>n</i> . leg; CM ligra id.
lu-		[lu] <i>adv.</i> non-; used to negate the enclitic copula; CM la- 1 id.; I.13, II.3, IX.10

Μ

<i>m</i> -		See men
<i>m-h-^w/_y</i>	məh'	<i>v.t.</i> to strike, hit, beat, stab; CM MHA id.; see also <i>čang</i> $m-h^{-w}_{/y}$, <i>safár</i> $m-h^{-w}_{/y}$, <i>şāyaḥ</i> $m-h^{-w}_{/y}$ [mɛh] <i>G perf.</i> \mathcal{J}^{rd} <i>m.sg.</i> (contextual) he hit; IX.5
<i>m-1-^w/_y</i>		v.t. and v.i. to fill, be full; CM MLA id.
<i>m-n-^w/_y</i>		v.t. to count; CM MNA id.
<i>m-ț-^w/_y</i>	məțon amțyu	<i>v.i.</i> G to arrive, C to bring; CM MTA id.; see also <i>xabár</i> $m-t^{-w}/_{y}$ [mə ^t t [°] on] <i>G perf.</i> 3^{rd} <i>pl.</i> they arrived; IV.113 ['amt [°] tju] <i>C perf.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>pl.</i> he brought them (caused them to arrive); IV.144
<i>m-<u>t</u>-^w/_y</i>	mə <u>t</u> yat lámə <u>t</u> yat	<i>v.t.</i> to lift up; CM MTA 'to stretch, to straighten;' see also <i>idā</i> $m_{\underline{f}}$ - $w/_y$ ['mɛθjat] <i>G perf.</i> 3^{rd} <i>f.sg.</i> it lifted (itself) up; IV.139, IV.149 ['la ₁ mɛθjat] <i>G neg. perf.</i> 3^{rd} <i>f.sg.</i> it hadn't lifted (itself) up; IV.137
<i>m-^w/_y-<u>t</u></i>	me <u>t</u> me <u>k</u> tat qəmāye <u>t</u> mie <u>t</u>	<i>v.i.</i> to die; CM MUT id. [m ϵ 0] <i>G perf.</i> 3^{rd} <i>m. sg.</i> he is dead; III.8b, X.15, X.17 ['m ϵ xtat] <i>G perf.</i> 3^{rd} <i>f. sg.</i> she is dead; II.34 [q φ 'm σ :j ϵ 0] <i>G imperf.</i> 3^{rd} <i>m.sg.</i> he will die; X.16 [mi ϵ 0] <i>G subjv.</i> 1^{st} sg. I will die (contextual); IV.84
Madday		[mɑd'daɪ] <i>prop. n.</i> Media; CM madai id.
māh		[mb] <i>n</i> . month; Persian <i>māh</i> id.; IV.20, IV.87, X.16, X.17
maḥallá	t	[maħal'lɑt [¬]] n. neighborhood, quarter (of a city); Arabic <i>maḥalla</i> id. via Persian <i>maḥallát</i> id.; II.4
māl		[mpl] <i>n</i> . property, possession; Arabic <i>māl</i> 'wealth, possessions,' cf. also Persian <i>māl</i> id.; II.12

336	8. Lexicon
māmā	['mɔ:mɔ] <i>n</i> . grandmother; likely a contraction of <i>em emmā</i> "mother's mother."
mambugā	[mam'bu:во] <i>n</i> . consecrated water used in rituals; CM mambuga id.
mambuhā	[mam ¹ bu:hɔ] <i>n</i> . consecrated water used in rituals; CM mambuga id.
man el-mannon	[mæn] <i>inter. pron.</i> who; CM man id.; II.24, IV.51 [ϵ l'mænnon] + 3^{rd} pl. whose are they? III.11
manʻa '- <u>b</u> -d	<i>v.t.</i> to prevent, prohibit; Arabic <i>man</i> "prohibition," cf.
man'a qa <u>b</u> ednannu	Persian <i>man</i> ^c <i>kardan</i> id. ['man ^c a qaved'nannu] <i>G imperf.</i> $1^{st} pl. + 3^{rd} pl.$ we will prevent them; IV.16
mand	[mæn] <i>n</i> . maund, a unit of measure (here ca. 14 lbs); Persian <i>man</i> id.; V.3, V.11, V.26, V.26
Mandā	['mændɔ] <i>n.</i> knowledge; CM manda; cf. Biblical Aramaic מַנְדַע id.
Mandā d-Heyyi	[¹ mændad ₁ hɛɪji] <i>prop. n.</i> epithet of God, CM manda <u>d</u> - hiia id.
Mandāyí	[mændəˈji:] <i>adj.</i> Mandaean, Mandaic (language); CM mandaia id.; II.15, II.19, II.21, II.22, II.23 (x2), II.25, II.32, III.16, IV.1
Mandayānā	[mænde'jo:no] pl. mandaean(s); I.3, I.5, I.6 (x2), I.8, I.11, I.13,
mandayān	III.25, IV.35 [mænde'jɔn] <i>pl</i> . mandaeans (contextual); III.12
mandi	[mɛnˈdi] <i>n</i> . a cultic hut; CM bimandia id.; III.13
Māni	[¹ mp:ni] <i>adj.</i> Manichaean; CM mania 'Mani;' 1.12
manzā	['mænzɔ] <i>n</i> . hair; CM manza id.
mārā māre	['mɔ:rɔ] <i>n.</i> lord; CM mara id. ['mɔ:re] + 1^{st} sg. God (lit. my lord); I.9 (x2), I.11, II.35 VI.2, VI.5

mariz marizi	[mæˈriz] <i>adj.</i> sick, ill; Arabic <i>marīḍ</i> id. via Persian <i>mariz</i> id.; [mæˈriːzi] <i>indef.</i> a sick person; VI.1
mas'ul	[mas ¹ ⁷ ul] <i>adj.</i> responsible; Arabic <i>mas²ū</i> /id. via Persian <i>mas²u</i> /id.; VIII.11b
Masbub	[mas'bub] <i>prop. n.</i> Sheikh Masbub, the <i>ganzibrā</i> in Khorramshahr during the 1930s; II.7, II.29
masiḥí	[mɑsiˈħi:] <i>adj.</i> Christian; Arabic <i>masīḥī</i> id. via Persian <i>masihi</i> id.; 1.12
mașbettā	[mas ^{§1} b3tt3] <i>n</i> . baptism; CM masbuta id.; 1.10
mašknā mašknāyí	[ˈmæʃknɔ] <i>n</i> . (religious) sanctuary; CM maškna id. [mæʃknɔˈjiː] <i>n</i> . those of the <i>mašknā</i> , an early term for the Mandaeans; CM mašknaiia id.
mašwā	['mæ∫wɔ] <i>n</i> . rope; CM mašua id.
mațrā	['mat [°] rɔ] <i>n</i> . rain; CM mițra 1 id.
maxșuș	[mɑɣˈs ^s us ^s] <i>adv.</i> especially; Arabic <i>maxṣūṣ</i> 'special,' and Persian <i>maxsus</i> id.; III.2
mehlā	['mɛhlɔ] n. salt; CM mihla id.
mehmān mehman-de	[meh ^l mon] <i>n</i> . guest; Persian <i>mehmān</i> id. [m \mathfrak{s} rlmænd \mathfrak{d}] + 1^{st} sg. my guest; V.16
meḥnát	[mɛħ'nɑt] <i>n</i> . hardship, misery, misfortune; cf. CM mihna id., probably Arabic <i>miḥna</i> id. via Persian <i>mehnat</i> id.; 1.14
mendā mendi mendānā	<i>n.</i> thing; CM mindam id. ['mɛndi] <i>indef. pron.</i> something; I.7, I.11, II.1, III.16, III.22 [mɛn'dɔ:nɔ] <i>pl.</i> things; IV.35, IV.118
mentekār	[mente'kor] n. car; English motor car id.
meškā mešk šijrā	[ˈmɛʃkɔ] <i>n.</i> skin; CM miška id. [ˈmɛʃk ˈʃɪʤrɔ] bark

metel	metelā	[mɛ'tɛl] <i>n.</i> story; cf. CM mitla id. [mɛtɛ'lɔ] <i>pl.</i> stories; III.20, III.25
məsi		[məˈsi:] <i>adj.</i> cold; CM msia 'solidified'
məşamı	mar	[məˈs ^s ɑmmæ1] <i>adj.</i> wet; swollen
məțā		[mə't [°] ɔ:] <i>n</i> . place
mienā		['miĕnɔ] <i>n</i> . water; CM mia id.; IV.7, IV.19
miez		['miĕz] <i>n</i> . table; Persian <i>miz</i> id.
mimrā		['mımro] n. literature; CM mimra id.; 1.5
min	minni minnak minne minnu minkon minjoo minjo-di əm- m- m-ehnā	[min] <i>prep.</i> from; (+ <i>pl.</i>) some; at (a particular time); CM mn from, with; I.13, IV.6, IV.20, IV.26, IV.103, IV.112, IV.126, IV.137, IX.3 ['minni] + 3 rd m.sg. from him; V.8, IX.12, X.9 ['minnax] + 2 nd m. sg. from you; IV.83, IV.118 ['minne] + 1 st sg. from me; IV.108 ['minnu] <i>prep.</i> + 3 rd <i>pl.</i> from them; some of them; VII.8 ['minxon] + 2 nd <i>pl.</i> from you; some of you; IV.39 ['minnæn] + 1 st <i>pl.</i> from us; V.20 <i>prep.</i> from within; cf. Arabic <i>juwwa</i> 'in it, within; inside' [min'dʒo:di] out of it, from within it; X.3 [m] <i>prep.</i> from; CM 'm 'with;' I.4, I.7, I.7, I.13, I.14, II.5, II.23, II.26, III.3, IV.3, IV.115, V.7, VI.2, VIII.8, X.1 [m] from (before vowel); I.8, I.13, II.10 ['mɛhnɔ] <i>adv.</i> henceforth; CM 'm with + Arabic <i>hunā</i> 'here;' IV.70, IV.87, IV.123
minjo		See min
mișrā '-	bd mișrā qābden	<i>v.i.</i> to draw a (magical) boundary; cf. CM miṣra 'boundary' ['mɪs [°] .ɪɒ 'qɒvdɛn] <i>G imperf. 3rd pl.</i> they draw (magical) boundaries; IV.20
miwā		['mi:wo] <i>n</i> . fruit; Persian <i>mive</i> id.

moqaddas moqaddas-dan		[moqad'das] <i>adj.</i> sacred, holy; Arabic <i>muqaddas</i> id. via Persian <i>moqaddas</i> id. [moqad'dasdæn] + 1 st pl. our holy (book); 1.9
m-ork		See orke
moḥāl	moḥālye	[mo ^l ħbl] <i>adj.</i> impossible, absurd; Arabic <i>muḥāl</i> id., cf. Persian <i>mohal</i> id. [mo ^l ħblje] + 3 rd m.sg. it is impossible; V.24, X.8
Moḥamrā		[mo ^l ħambrɔ] <i>prop.</i> Mohammerah (today Khorramshahr); Arabic <i>Muḥammara</i> ; II.6, II.7, II.23, II.29, II.32, III.3
Ms. Drower		[₁ mɪsə ¹ drɑuwə ⁻] <i>prop. n.</i> Ms. Drower, i.e. Lady Ethel Stefana Drower; 11.26
mu	muyye muyya	[mu] <i>inter. pron.</i> (independent form) what; CM ma id.; IV.33, IV.77, IV.128, IV.131, V.25, VI.6, IX.3, IX.8; see also <i>mujur</i> [${}^{l}m\Lambda jje$] + 3^{rd} m.sg. what is it; IV.10, IV.12b, X.4 [${}^{l}m\Lambda jja$] + 3^{rd} f.sg. what is it; IV.106
mujur	mujurnā	[¹ mA d AI] <i>inter. pron.</i> how, in what way; CM ma 'what' + Persian <i>jur</i> 'sort, kind;' V.24, V.26, IX.14, X.16, X.17 [mA ¹ d u.no] + 1^{st} sg. how I am; IX.3
Mušmānā Mušmanānā Mušmanān		<i>n.</i> Muslim; PCM mušmana id. [mʌ∫mɛˈnɔːnɔ] <i>pl.</i> Muslims; IV.150 [mʌ∫maˈnɔn] <i>pl.</i> (contextual) Muslims; III.25

Ν

<i>n-h-<u>t</u></i>	anhi <u>t</u> '	<i>v.i.</i> to come down, descend; CM NHT id. [$\epsilon n'hi\theta$] <i>C perf.</i> 3^{rd} <i>m. sg.</i> + 3^{rd} <i>f.sg.</i> he brought it down; IV.1
<u>n-k</u> -t		<i>v.t.</i> bite; CM NKT id.
n-p-l		<i>v.i.</i> to fall; CM NPL id.
n-p̄-q	nəpaq nəpqat nəpaqyon qənāpeq	<i>v.i.</i> to exit, come out; CM NPQ id. [nə ¹ faq] <i>G perf.</i> 3^{rd} <i>m.sg.</i> he exited; VIII.7, VIII.13 ['nɔfqat] <i>G perf.</i> 3^{rd} <i>f.sg.</i> she exited; II.15, II.31 [nə ¹ faqjon] <i>G perf.</i> 3^{rd} <i>pl.</i> they exited; II.13 [qə ¹ nɒ:fɛq] <i>G imperf.</i> 3^{rd} m.sg. he exits; V.8
<i>n-^w/_y-d</i>	nādat qənidā qəmandin'	<i>v.i.</i> to shake; CM NUD and NDA; see also <i>nodā</i> ['nɔ:dat] <i>G perf.</i> 3 rd <i>f.sg.</i> it shook; IV.138 [qə'ni:dɔ] <i>G imperf.</i> 3 rd <i>f.sg.</i> it shook; IV.102, IV.105 [qəmæn'din] <i>C imperf.</i> 1 st sg. I will make (s.t.) shake (contextual); IV.98
nafár		[na ^l fa1] <i>n</i> . individual (used in counting); Arabic <i>nafar</i> id. via Persian <i>nafar</i> id.; VII.6
nahrā		['næhro] n. river; CM nahra id.; III.8, III.13
napšā	napš mașbetta napša	['næfʃɔ] <i>n</i> . soul, personality, self; same; CM napša id. ['næfʃ mɑs ^{{1} bətta] (contextual) the same baptism; I.10 ['næfʃa] + $3^{rd} f. sg.$ her personality; II.33
narmā		['naımo] adj. smooth; Persian narm 'soft, smooth'
narm t-r	n-m narm tammā	<i>v.i.</i> to go soft (in the head); Persian <i>narm</i> 'soft, smooth' ['naım 'tæmmo] <i>G perf.</i> 3 rd <i>m.sg.</i> he went soft (in the head); V.14
nașiruțā		[nɑs [°] i ^l ru:θɔ] <i>prop. n.</i> esoteric knowledge; CM naşiruta id.

nașorayi		[nɑs ^s orɔ ^l ji:] <i>prop. n.</i> Nasoreans, a generic term for the initiated priesthood; CM naṣuraiia id.
Nașșat-e Șubba		[nɑs ^ɛ 's ^ɛ ate 's ^ɛ Abba] <i>prop. n.</i> a large pit outside of Shushtar (q.v.), said to be the location of the Mandaean quarter of town before Čečāwā bar Čečāwā (q.v.) carried it off to Əmšuni Kušṭā (q.v.); IV.150
nazdik		[næz ¹ dik] <i>prep.</i> near; Persian <i>nazdik</i> id.
nedā		[nɛ ^l dɒ:] <i>n</i> . proclamation; Persian <i>nedā</i> id.
nedā '- <u>l</u>	b-d	<i>v.i.</i> to proclaim, announce; Persian <i>nedā</i> 'proclamation,' cf. <i>nedā kardan</i> id.
	nedā ə <u>b</u> adyon	[nɛ'dɒ: 'vadjon] <i>G perf.</i> 3^{rd} <i>pl.</i> they proclaimed; V.20
nešemta	ā nešemti	[nɛˈʃɛmti] <i>n</i> . soul, living being (f.); CM nišimta id. [nɛ¹ʃɛmti] <i>indef</i> . a soul, living being; IV.96
nəhirā		[nəˈhi:rə] n. nose; CM nhira id.
nəjārā		[nəˈʤɔ:rɔ] <i>n</i> . carpenter; PCM nagara id. from Arabic <i>najjār</i> id.
	nəjāri	[nəˈdʒɔ:ri] <i>indef.</i> a carpenter
	nəjarānā	[nəʤɛˈrɔ:nɔ] pl. carpenters; IV.130
	nəjārí	$[n \partial_1 d \beta \partial' ri!]$ <i>n</i> . carpentry
No		[no:] <i>prop. n.</i> Noah; CM nu id.; 1.7
nodā		['no:də] <i>n</i> . earthquake, turmoil; CM nauda id.; IV.115
Nuḥ		[nʌħ] prop. n. Noah; Arabic <i>Nuḥ</i> id.; 1.10
nurā	nur	[¹ nu:rɔ] <i>n</i> . fire; CM nura id.; IV.114 [nu.1] (contextual) fire; IV.55, IV.91

0

obrā	obri	['obro] <i>n.</i> mouse; CM aubr, 'ubra id.; V.10, V.12, V.26, V.28 ['obri] <i>indef.</i> a mouse; V.8, V.9, V.11
ohrā	ohhər	['ohrə] <i>n</i> . road, path (f.); CM 'uhra id. ['ohhə ⁻] road (contextual); III.3
ohrā '-z	-g- ^w / _y ohrā allen	<i>v.i.</i> to walk; cf. Persian <i>rāh raftan</i> 'to go, walk' ['ohro 'allɛn] <i>G imperf.</i> 3 rd pl. they walk; IV.6
okumā		[ŏ ¹ ku:mɔ] <i>adj.</i> black; CM 'kuma id.
olamā́		[olæ'mp:] <i>n</i> . wise men, scholars (collective); Arabic <i>'ulamā'</i> 'scholars' (pl. of <i>'ālim</i> 'scholar') via Persian <i>olamā</i> id.; IV.11, IV.17, IV.28
onnā	onne	['onno] <i>n</i> . ear; CM ' udna id. ['onne] + 1^{st} sg. my ear; VI.2
orke	orki orka orke orku m-ork	[¹ o.ike] <i>prep.</i> with; PCM 'urkia id.; III.25, IV.148 [¹ o.iki] + 3^{rd} m.sg. with him; IV.37, V.18 [¹ o.ika] 3^{rd} f. sg. with her; II.17 [¹ o.ike] + 1^{st} sg. with me; II.19, III.8 [¹ o.iku] + 3^{rd} pl. with them; II.14 [mo.ik] prep. with (Ahvāzi); CM 'm id.; I.4, I.5

Р

p-h-r	pahrat	<i>v.i.</i> to fly, fly off; CM PRA 2; PHR id. ['pahrat] <i>G perf.</i> 3 rd <i>f.sg.</i> it flew; IV.141 [qə'pɒ:hɛɪ] <i>G imperf.</i> 3 rd <i>m.sg.</i> it flies; V.25
	qəpāher pəherye	[qə'pɒ:hɛɹ] <i>G imperf.</i> 3^{rd} <i>m.sg.</i> it flies; V.25 [pə'hɛɹje] <i>G pass. ptc.</i> + 3^{rd} m.sg. it is flying; V.21
p-r-d	pərad pəradyon pardén	<i>v.i.</i> to flee; CM PRD id. [pə ['] rad] <i>G perf.</i> 3^{rd} <i>m.sg.</i> he fled; IX.13 [pə ['] radjon] <i>G perf.</i> 3^{rd} <i>pl.</i> they fled; III.23, IV.103, IV.105 [paɪ'dɛn] <i>G imperf.</i> 1^{st} <i>pl.</i> (contextual) let's free; VII.8
<i>p-r-<u>k</u></i>		<i>v.t.</i> to rub; CM PRK II to crush, bind together
p-r-d-s	pardas pardason əmpardas	<i>v.i.</i> to be finished, to be done; CM PRDS id.; Middle Persian <i>pardaxtan</i> 'to be done with, be free of,' commonly supplanted in modern Persian with the causative, <i>pardāxtan</i> id.; see H.S. Nyberg, <i>A Manual of</i> <i>Pahlavi II</i> (Wiesbaden: Harrassowitz, 1974), 151-52. [paɪ'dɑs] Q perf. 3^{rd} m.sg. he was finished; IV.21, VI.4 [paɪ'dɑson] Q perf. 3^{rd} pl. they were finished; II.12 [m'paɪdɑs] Q imperf. 3^{rd} m.sg. it was done; IV.44
p-r-q	parqi	<i>v.t.</i> to free; CM PRQ id. ['paɪqi] <i>G perf.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>m.sg.</i> he freed him
p-s-q	pəsqat pəseq epseqni	<i>v.t.</i> to cut (off); CM PSQ id. ['pɛsqat] <i>G perf.</i> 3 rd <i>f.sg.</i> she cut off; VII.4 [pə'sɛq] <i>G pass. ptc.</i> cut off; III.20 [ɛp'sɜqni] <i>G perf.</i> 1 st <i>pl.</i> we were exterminated (cut off); 1.14
pahli	pahli-di	[pæh'li:] prep. beside; Persian <i>pahlu</i> 'side' [pæh'li:di] + 3^{rd} m.sg. beside him; VI.3, VII.6
palgā		['palγo] <i>n</i> . split; CM palga 1 share, portion
par		[pe1] n. leaf; Persian par id.
pas	pasi	[pɑs] <i>n.</i> pass; English <i>pass</i> id. ['pɑ:si] <i>indef.</i> a pass; III.4

344		8. Lexicon
pa <u>t</u> yā		['pæθjɔ] adj. wide; CM ptia 'open, wide, broad'
pelti		[pelti] adj. few, little (Dezful)
peršā	perš	['pɛɪʃɔ] <i>n</i> . tomorrow; CM pirša 1 'dawn;' V.19 ['pɛɪʃ] (contextual) tomorrow; V.17
perzā	perzi perz	['pɛɹzɔ] <i>adj.</i> few, little; cf. Persian <i>porze</i> 'a scrap or piece' ['pɛɹzi] (indef.) a little [pɛɹz] (contextual) little; V.1
pešimāi	n t-m-m pešimān tammon	<i>v.i.</i> to repent, be sorry; Persian <i>pašimān šodan</i> id. p[pe ji'mon tæmmon] G perf. 3rd pl. they were sorry; IV.146
pəhirā		[pəhi:rɔ] <i>n</i> . airplane; neologism from $\sqrt{p-h-r}$ 'to fly'
pərā	pərāhi pərāhā	[pə'rɔ:] <i>n</i> . money; Ottoman Turkish <i>pār</i> , a coin first introduced during the reign of Murat IV (1623 to 1640); cf. (Modern) Turkish <i>para</i> 'money;' V.5, VIII.5, VIII.10 [pə'rɔ:hi] <i>indef</i> . some money; V.1 [pə _r ɔ'hɔ:] <i>pl</i> . money; VIII.8
pə <u>t</u> i		[pə'bi:] adj. empty; cf. CM ptia 'open, wide, broad'
pienā		['piĕnɔ] <i>n</i> . evening; CM paina id.; V.16
pol		[pol] <i>n</i> . bridge; Persian <i>pul</i> id.; IV.19, IV.24, IV.29, IV.33, IV.58, IV.67
pommā		['pommo] n. mouth; CM puma id.
Profesa	r Buckley	['profɛsə [,] 'bʌkl̥i] <i>prop. n.</i> Jorunn Jacobsen Buckley, historian of religions and Mandaean advocate; I.6
Provost	[•] Marshal	['provos'ma1sel] <i>n</i> . the chief of the military police; English <i>provost marshal</i> ; III.4

Ē

poquttā		[fo'qAtto] <i>n</i> . neck, throat; CM pquta id.
		Q
<i>q</i> -		[qə] <i>prep.</i> to, for (enclitic form); CM qa- id.; II.31
q - l - $w/_y$		v.i. to burn; CM QLA id.
<i>q-r-^w/_y</i>	qərā	<i>v.t.</i> G to read, to recite; $(+\partial l\bar{a}w)$ to enchant; D to call; CM QRA id.; see also <i>butā q</i> - <i>r</i> - ^{<i>w</i>} / _{<i>y</i>} [qə ⁱ rɔ:] <i>G perf.</i> 3^{rd} <i>m.sg.</i> he read; IV.91 ['qɔ:ri] <i>G ptc. pres</i> 3^{rd} <i>m.sg.</i> he reads; IV.55
	qāri qāren qarén	['qɔ:rɛn] <i>G imperf.</i> 3^{rd} <i>pl.</i> they read; IV.17, IV.18, IV.19, IV.20 [qa'rɛn] <i>G imperf.</i> 3^{rd} <i>m.sg.</i> (contextual) we will read; IV.16
	qəmqarellan	[qmqa'rɛllæn] D imperf. 3^{rd} $pl. + 1^{st}$ pl . they call us; 1.13
<i>q-^w/_y-m</i>		<i>v.i.</i> to stand, rise; <i>v.t.</i> (as an auxiliary) to begin; CM QUM id.
	qam	[qam] <i>G perf.</i> 3 rd m.sg. he stood; he started; IV.22 (x2), IV.42, IV.55 (x2), IV.91, IV.127, VIII.17
	qāmat qamyon	['q \mathfrak{c} :mat] <i>G perf.</i> \mathfrak{I}^{rd} <i>f. sg.</i> she arose; IV.101 ['q \mathfrak{c} mjon] <i>G perf.</i> \mathfrak{I}^{rd} <i>pl.</i> they began; IV.20, IV.76, IV.89
qābin		[qp'vin] <i>n</i> . marriage; likely Middle Persian <i>kābēn</i> 'dowry' via CM qabin 'marriage;' II.6, II.28
qābin '-	þ-d qābin əbadyon	<i>v.i.</i> to have a wedding; cf. Persian <i>arus kardan</i> id. [qp ¹ vin ə ¹ vadjon] <i>G perf.</i> 3 rd pl. they had a wedding; II.10, II.30
qabúl '- <u>i</u>	<u>b</u> -d	<i>v.t.</i> to accept, believe; Arabic <i>qabūl</i> 'acceptance' via Persian <i>qabul</i> id., cf. Persian <i>qabul kardan</i> id.
	qabúl qābed qabúl láqābed	[qa'bul 'qp:ved] G imperf. 3^{rd} m.sg. he believes; X.16, X.17 [qa'bul 'la ₁ qpved] G neg. imperf. 3^{rd} m.sg. he doesn't believe; X.17
qahwe	qahwi	['qahve] <i>n.</i> coffeehouse; colloquial Arabic <i>qahwa</i> id. ['qahvi] <i>indef.</i> a coffeehouse; VII.1

346		8. Lexicon
qālā	qāl	[¹ qɔ:lɔ] <i>n</i> . sound, noise; CM qala id. [qəl] (contextual) sound, noise; IV.49, IV.102
qālā '-h	-b qāl ahbat	<i>v.i.</i> to make a noise; cf. Persian <i>sadā dādan</i> id. [qɔl 'ɑhvɑt] <i>G perf.</i> 3 rd <i>f. sg.</i> it made a noise; IV.139
qam	qamdi qamda <u>k</u> qamdin qamdi-du qamdə <u>k</u> on qamdan	[qam] <i>prep.</i> to or for; CM qam id.; I.3, I.4, I.6, II.10, III.6 (x3), III.7, IV.39, IV.64, IV.84, IV.121, V.6, VIII.15, X.13 ['qamdi] + 3^{rd} m.sg. for him; VI.6 ['qamda\chi] + 2^{nd} m.sg. for you; IV.63 [qam'din] + 1^{st} sg. for me; IV.39, IV.62, IV.120 [qam'di:du] + 3^{rd} pl. for them; IV.117, X.12 ['qamdə\chi] + 2^{nd} pl. for you; X.7 ['qamdæn] + 1^{st} pl. for us; IV.94
qamāy		[qaˈmɔɪ] <i>prep.</i> before; CM qadmaia 'first;' IV.41, IV.44, IV.93
	qamāya qamātar	$[qa'mo;ja] + 3^{rd} f. sg.$ before her; II.17 [qamo'ta1] <i>adv.</i> (comparative) prior; III.18
qamie <u>t</u> ā	Ī	[qaˈmiĕθɔ] <i>n</i> . noon and <i>adv</i> . at noon; CM qamaita noontime devotions; IV.17, IV.112
qamu		['qɑ:mu] <i>inter. pron.</i> why; cf. CM I mahu / almahu / amu id.; IV.29, IV.57
qanāyā	qənayānā	[qa'nɔːjɔ] <i>n.</i> smith; CM qanaia id.; III.5 [qəna'jɔːnɔ] <i>pl.</i> smiths; IV.144, IV.136, IV.128, IV.130, IV.137, IV.148
qār		[qo1] <i>prep.</i> at, before, in the presence of, to, by, with; semantically similar to French <i>chez</i> and Arabic <i>'inda</i> ; CM qar id.; II.27, II.29, IV.38, IV.75, IV.91, IV.104, IV.130, IV.144, V.16, V.19
	qāri	$['q_{2}:r_{1}] + 3^{rd} m. sg.$ to him; IV.12, IV.32, IV.37, IV.73, IV.74, IV.75, IV.77, IV.82, IV.114
	qāra <u>k</u> qāre	$[^{1}q\Im ra\chi] + 2^{nd} m.sg.$ with you; IV.40, IV.73 $[^{1}q\Im re] + 1^{st} sg.$ to me; with me = I have; IV.64, IV.81, IV.92, X.6
	qāru qāran	$['q\mathfrak{z}:ru] + 3^{rd} pl.$ with them; IV.145 $['q\mathfrak{z}:ran] + 1^{st} pl.$ with us = we have; IV.97
	941411	$[Y^{on uni}] + 1 p_0$, with us - we have, $1^{v.37}$

qarār '-h-b qarār biete əhabtu		<i>v.t.</i> to place, fix, set up; Arabic and Persian <i>qarār</i> 'decision,' cf. Persian <i>qarār dādan</i> id. [qa'rɑɪ 'biĕθe ə'hæftu] <i>G perf.</i> $I^{st} sg. + 3^{rd} pl.$ you placed them in my house (for safekeeping); V.7
		them in my nouse (for safekeeping), v./
qarn		[qarn] <i>n</i> . horn; Arabic <i>qarn</i> id.
qazġān	qazģāni qazģān-de qazģān-ķon	$[qaz' ext{ \wp n}]$ <i>n.</i> cooking pot; Persian <i>qazġān</i> id., cf. Arabic <i>qazān</i> id. and Turkish <i>kazan</i> id.; X.2, X.3, X.15, X.16, X.17 $[qaz' ext{ $\varsis n}]$ <i>indef.</i> a cooking pot; X.1, X.9 $[qaz' ext{ $\varsis n}] + 1^{st}$ sg. my cooking pot; X.14 $[qaz' ext{ $\varsis n}] + 2^{nd}$ pl. your (pl.) cooking pot; X.5
qəlāw-		[qə'ləu] <i>prep.</i> to, for; CM qa- + 'lau-; II.10, V.18
qəmol <u>t</u> a	ī	$[q \vartheta' mol \theta \vartheta] \boldsymbol{n}.$ louse
<i>qəmāšā</i>	qəmāšāni qəmāšānan	<i>n</i> . (article of) clothing $[qmo' \int cmi] pl. + 3^{rd} m.sg.$ his clothes; VIII.2, VIII.14 $[qmo' \int cmcan] pl. + 1^{st} pl.$ our clothes; III.10
qəyānā	qəyān	<i>n</i> . culture [qɪ'jən] (contextual) culture; 1.5
Qolastā		[qo ^l læstɔ] <i>n</i> . collection; the Canonical Prayerbook of the Mandaeans; CM qulasta id.
qumát		[qu'mot] <i>n</i> . prefecture; possibly Arabic <i>qawamat</i> (pl. of <i>qā'em</i> 'administrator, prefect'); V.19
quwwát	f	[quv'vat] <i>n</i> . strength, force, authority; Arabic <i>quwwát</i> id. via Persian <i>quvvat</i> id.; IV.118

R

r-h-m	qəraheml'	<i>v.t.</i> to love; CM RHM id. [qəra'hɛml] <i>G imperf.</i> 3^{rd} <i>m.sg.</i> $+ 3^{rd}$ <i>m.sg.</i> (<i>elided</i>) he loves (s.t.); V.12
rabbā	raptā rabbi raptar	['ræbbɔ] <i>adj.</i> great, big; CM rba id.; X.11 ['ræftɔ] f. great; CM rabtia id. ['ræbbi] <i>indef.</i> a large (thing); X.10 ['ræfta] <i>comp.</i> greater
rabbā	rab	['ræbbɔ] <i>n.</i> leader; II.3 ['ræb] (contextual) leader; II.2
radpā		['rædfə] n. persecution; CM radpa id.; 1.14
radpā h-	2	<i>v.i.</i> to be persecuted; CM radpa 'persecution,' cf. Persian <i>ranj didan</i> id.
	radpā həzini	['rædfɔ hə ₁ zi:ni] G perf. 1^{st} pl. we were persecuted; I.14
rafieq	rafieqā́-di	[ræ'fiĕq] <i>n</i> . friend, companion; Arabic & Persian <i>rafīq</i> id. [ræfiɛ̃'qɔ:di] <i>pl.</i> + $\mathcal{3}^{rd}$ <i>m. sg.</i> his friends; IV.98
ragʻa '- <u>b</u>	p-d	<i>v.t.</i> to fix, patch up (s.o. or s.t <i>d</i> -); Arabic <i>ruq'a</i> 'a patch or a scrap,' cf. Persian <i>ruq'a doxtan / ruq'a zadan</i> 'to patch up'
	ragʿa-da āḇed	[rag ¹ ?ada ₁ o:vɛd] <i>G</i> imperf. 3^{rd} m.sg. + 3^{rd} f.sg. he patches it (the sky) up; IX.12
	ragʻa-da <u>k</u> ā <u>b</u> edu	[rag ¹ ?ada χ_1 , sizedu] <i>G</i> imperf. 3^{rd} m.sg. + 3^{rd} pl. + 2^{nd} m.sg. he fixes them for you; VI.6
Rām		[rbm] <i>prop. n.</i> Aram, the son of Shem and grandson of Noah; CM ram id.; 1.7, 1.10
rāst		[rost] or [ros] adj. true; straight; Persian rāst id.
rāst '-m-	-1	v.i. to speak the truth; Persian <i>rāst</i> 'true,' cf. Persian <i>rāst</i> goftan id.
	rāst qāmrət rāst qamennā	[rost 'qpmret] <i>G imperf.</i> 2^{nd} sg. you speak the truth; V.12 [rost qa'menno] <i>G imperf.</i> 1^{st} sg. I speak the truth; V.15

rațnā rațnā d-Hibel Ziwā		['ræt ^s nɔ] <i>n</i> . dialect (espec. the colloquial dialect of Mandaic against the written form, <i>genzā</i>); CM raṭna id., cf. Arabic G <i>r-ṭ-n</i> 'to speak gibberish' ['ræt ^s nɑd 'hi:vɛl 'zi:wɔ] <i>prop. n.</i> the idiom of Hibel Ziwā, ie. Mandaic; IV.1
rāzā	rāz' ed-kasyā rāz' ed-Šāmeš rāzā qāre rāzi	['rD:ZO] <i>n</i> . secret, mystery; CM raza id. ['rD:Zed 'kæsjo] closely guarded secret; CM raza kasia id.; IV.85, IV.121 ['rD:Zed,∫D:me∫] secret of Šāmeš, the archon ruling the sun and one of the <i>šəbihānā</i> (q.v.); IV.135 ['rD:Za 'qD:re] the secret that I have; IV.92 ['rD:Zi] <i>indef.</i> a secret; I.13
rehwā	rehhu rehhu '- <u>t</u> - ^w /y	['rɛhwə] <i>n</i> . wind; CM rihua id. [,rɛhhu] (contextual) wind [,rɛhhu ɛ' θ ə:] <i>G perf.</i> 3^{rd} <i>m.sg.</i> the wind blew
rezq	rezqā́-dan	[rɛs'q] <i>n</i> . livelihood; Arabic <i>rizq</i> id. via Persian <i>rezq</i> id.; cf. CM rziqa id. [rɛs'qɔ:dæn] pl . + $I^{st} pl$. our livelihoods; IV.132
rəwāhā		[rəˈwɔ:hɒ] <i>n</i> . relief, solace; CM ruaha id.; II.34, III.8
rəwāhā e	ehl-	<i>v.i.</i> to rest in peace (lit. to have relief; said of those who are dead)
	rəwāhā nehwili	[rə ^l wo:ha nɛh ^l vi:li] <i>G impf.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>m.sg.</i> may he rest in peace (lit. may he have relief); III.8
	rəwāhā nehwila	[rə ¹ wo:ha nɛh ₁ vi:la] G impf. 3^{rd} m.sg. $+ 3^{rd}$ f. sg. may she rest in peace (lit. may she have relief); II.34
rihā		['ri:hə] n. smell; CM riha id.
rišā	riš šettā	['ri:∫ɔ] <i>n</i> . head, beginning; CM riša id. [ri∫ '∫ɛttɔ] the beginning of the year, i.e. New Year's Day; IV.3 (x2)
	rišāmmā	[ri ¹ ∫ɔmmɔ] <i>n.</i> "head of the people," the highest rank of the Mandaean priesthood; CM rišama id.; IV.35, IV.36, IV.37, IV.40, IV.41, IV.44, IV.55, IV.61, IV.64, IV.73, IV.74, IV.75, IV.82
	riš yum	$[ri\int jum]$ the first day, the start (in time); I.7

350		8. Lexicon
Rusya		['rusja] prop. n. Russia; III.2
ruz		[ruz] <i>n</i> . day (used in counting); Persian <i>ruz</i> id.; IV.26, IV.44, IV.137, X.2
		S
s-d-r	qəmsadart	<i>v.t.</i> D to organize; CM SDR id. [qmsæ ^l dæ.t] <i>D imperf.</i> 2 nd sg. (contextual) you are organizing; 1.6
s-l-q	sāqat Iásāqit	<i>v.i.</i> to go up; CM SLQ id. ['sɔ:qat] <i>G perf.</i> $3^{rd} f. sg.$ it (the ground) went up; IV.145 ['la _i sɔqıt] <i>G neg. perf.</i> $1^{st} sg.$ I didn't go up; III.13, III.15
<i>s-r-<u>k</u></i>	nurā asre <u>k</u> nur asre <u>k</u> šam qəmasri <u>k</u> en	<i>v.t.</i> to kindle or light; CM SRK 'to kindle' ['nu:rɔ 'æsrɛɣ] <i>C perf.</i> 3^{rd} <i>m. sg.</i> he made light; IV.114 [nuɪ 'æsrɛɣ] <i>C perf.</i> 3^{rd} <i>m. sg.</i> he made light; IV.55, IV.91 [ʃɑm qəmas'ri:ɣɛn] <i>C imperf.</i> 3^{rd} <i>pl.</i> they light candles; IV.20
sāʿát	sāʿát d-ahtā	<i>n.</i> hour; Arabic <i>sāʿa</i> id. via Persian <i>sāʿát</i> id. [sɔˈl͡sat ˈdæhtə] one o' clock; IV.138
sad		[sæd] quant. 100; Persian sad id.; V.11, V.26
safár		[sæˈfæɪ] <i>n.</i> travel, journey; Arabic <i>safar</i> id. via Persian <i>safar</i> id.; IV.144
safár '- <u>b</u>	-d	<i>v.i.</i> to travel; Persian <i>safar kardan</i> id.; see also <i>safár m</i> -
	safár ābed	$h^{-w'_{y}}$ [sæ'fæi 'ɔ:vɛd] <i>G subjv.</i> 3^{rd} <i>m.sg.</i> he travels; V.2
safár m-	h- ^w / _y safár mahén	 v.i. to go traveling, "hit the road;" cf. Persian safar oftādan id.; see also safár '-b-d [sæ'fæı ma'hɛn] G subjv. 1st pl. we travel; IV.132
sāl	bis sālu ʿumər-de	[spl] <i>n</i> . year (used in counting); Persian $s\bar{a}/id$. [bis 'sp:lu ' Ω mrde] + 3^{rd} <i>pl</i> . my age is twenty years (lit. twenty years are my age); III.14

Sām		[som] <i>prop. n.</i> Shem, son of Noah; CM šum id.; I.7, I.10
sawār t-m-m sawār ḥisān tammon		<i>v.t.</i> to ride s.t., lit. to become a rider of s.t.; Persian <i>savār</i> 'rider,' cf. Persian <i>savār-e s.t. šodan</i> id. [sɛ'woュ ħr'sɒn tæm'mon] <i>G perf.</i> 3^{rd} <i>pl.</i> they rode horseback, lit. they became a horserider; IV.65
ser t-m-m ser ta	mmat	<i>v.i.</i> to become numb, insensate; Persian <i>sayr šodan</i> id. [seɪ 'tammat] <i>G perf.</i> 3^{rd} <i>f.sg.</i> it (the leg) became numb; VII.3
serrā		['serro] n. moon; CM sira id.
səbuʿa		See <i>esbu'a</i>
səmāqā		[səˈmɔːqɔ] <i>adj.</i> red; CM sumaqa id.
səmi <u>k</u> ā		[səˈmiːχɔ] adj. thick; cf. CM sumka 'thickness'
səri		[səˈriː] <i>adj.</i> rotten; cf. CM saria 3 stinking, putrid, act. ptc. from SRA I to stink, be putrid
sidrā Sidrā	d-Nešmā <u>t</u> ā	['sɪdrɔ] <i>n</i> . book; CM sidra id. [_ı sɪdrɒdnɛʃ'mɔ:θɔ] <i>prop. n.</i> the "Book of Souls," a collection of baptismal hymns from the Canonical Prayerbook of the Mandaeans; CM sidra <u>d</u> -nišmata id.
SO		[s0] <i>quant.</i> three (used in counting); Persian <i>se</i> id.; IV.18, IV.26, IV.37, IV.44, IV.75, IV.82, IV.87, IV.88, X.2
sotór sotór	-di	$[s\Theta^{\dagger}to_{I}]$ <i>n</i> . razor; Persian <i>sotor</i> id. $[s\Theta^{\dagger}to_{I}di] + 3^{rd} m.sg.$ his razor; VIII.3
street		[strit] <i>n. sg.</i> street; English <i>street</i> id.; III.22

Ş

<i>ṣ-<u></u>b-^w/_y</i>	șəby' cșțəbyon qəmeșțəbén	v.t. G to baptize (trans.); tG to baptize (intrans.); CM ŞBA id. [s^{ς} εvji] <i>G perf.</i> 3^{rd} <i>m. sg.</i> + 3^{rd} <i>m.sg.</i> he baptized him; 1.9 [$\varepsilon s^{\varsigma} t^{\varsigma} s^{j}$ svjon] <i>tG perf.</i> 3^{rd} <i>pl.</i> they baptized (intrans.); 1.10 [$q = m s s^{\varsigma} t^{\varsigma} s^{j}$ ven] <i>tG imperf.</i> 1^{st} <i>pl.</i> we baptized (intrans.); 1.10
ș-m-r		<i>v.i.</i> to swell, become wet; CM SMR 1 to inflame, become red
<i>ṣ-r-^w/_y</i>		v.t. C to squeeze; CM ŞRR 2 to surround, tie up
Ş- ^w / _y -t	șat șāti șatyon șatonni	<i>v.t.</i> to hear (+ s.o.) to watch, (+ <i>min</i>) to listen; CM ŞUT id. [s [°] at] <i>G perf.</i> 3^{rd} <i>m.sg.</i> he heard; IX.2 ['s [°] o:ti] <i>G perf.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>m.sg.</i> he watched him; VI.3 ['s [°] atjon] <i>G perf.</i> 3^{rd} <i>pl.</i> they heard; I.13 [s [°] a'tonni] <i>G perf.</i> 3^{rd} <i>pl.</i> + 3^{rd} <i>m.sg.</i> they watched him; VI.7
șābi'e	șābi'ūn	['s [°] ɑ:bije] <i>coll.</i> Sabians; Arabic <i>ṣābi'a</i> id.; III.12 ['s [°] ɑ:bi?un] <i>pl.</i> Sabians
ṣāḥeb		['s ^s ɒ:ħɛb] <i>n</i> . owner; Arabic <i>ṣāḥib</i> id.; X.8
șāyaḥ m	<i>-h-^w/y</i>	<i>v.i.</i> and <i>v.t.</i> to scream; Arabic <i>sayh</i> 'scream,' cf. Persian <i>sayhe zadan</i> id.
	ṣāyaḥ məhā ṣāyaḥ qəmāhyā	['s [°] b:jaħ mə ^l hɔ:] G perf. 3^{rd} m.sg. he screamed; V.22 ['s [°] b:jaħ qə ^l mbhjɔ] <i>G imperf.</i> 3^{rd} <i>f. sg.</i> she screamed; IV.102
	ṣāyaḥ qəmāhen	['s [°] D:jaħ qə'mD:hɛn] G imperf. I^{st} pl. they screamed; IV.143
şayd		[s ^s aɪd] <i>n</i> . hunt; Arabic <i>şayd</i> id.
<i>şəbā</i> ḥ		[s [°] ə'bɒħ] <i>n.</i> morning; Arabic <i>ṣabāḥ</i> id., Persian <i>sobh</i> id; IV.6, IV.17, IV.26, IV.137, IV.138

șəbāḥ ṭāḇ '-h-ḇ ṣəbāḥ ṭāḇ haḇlonni		<i>v.t.</i> to wish someone a good morning $[s^{s} \partial b b \hbar t^{s} \partial u h a u^{l} lonni] G perf. 3^{rd} pl. + 3^{rd} m.sg. they wished him a good morning; IV.38$
smālā		['smo:lo] adj. left; CM smala id.
șoprā		[¹ s [°] oprə] <i>n</i> . bird; CM şipra 1 id.
șoprā		[¹ s [°] ofro] <i>adj.</i> yellow; cf. CM sipra 2 dawn
șubbi	șubba șubbihấ	['s ^s ʌbbi] n. Sabian; Arabic <i>şubbi</i> id. ['s ^s ʌbba] <i>coll.</i> Sabians; Arabic <i>şubba</i> id. [s ^s ʌbbi'hɔ:] <i>pl.</i> Sabians; III.12
Şulḥabād		[s [°] Alħa'bɒd] <i>prop. n.</i> a city in Iran; III.7
șun ^c a	sun'aye	['s ^s Ansa] adv. well; Arabic <i>sun</i> ^c 'benefit, favor;' VI.2 ['s ^s Ansajɛ] + 3^{rd} m. sg. suff. it was fixed; II.4, III.21
<u>șț</u> ānā	șțāni	['s [°] t [°] ɔ:no] n. boy; CM sṭana id.; X.17 ['s [°] t [°] ɔ:ni] <i>indef.</i> a boy; V.25

Š

<u>š-</u> <u>b</u> -q	šə <u>b</u> aq šabqon láša <u>b</u> qonan	<i>v.t.</i> to leave; to permit; CM ŠBQ id. $[\int \mathfrak{d}^{l} waq] G perf. 3^{rd} m.sg.$ he left; IX.13 $[\int \mathfrak{a} u^{l} qon] G perf. 3^{rd} pl.$ (contextual) they left; III.23 $['la_{j}\mathfrak{a} uqon \mathfrak{e} n] G neg. perf. 3^{rd} pl. + 1^{st} pl.$ they didn't let us; III.13
	láqšabeqnā šobq	['laqja ₁ vɛqnɔ] G neg. imperf. 1^{st} sg. I can't permit; VIII.9 [jouq] G impv. m.sg. (contextual) free, permit, allow; IV.109
š-d-r	šadder	<i>v.t.</i> to send (+ <i>əlāw</i> for s.o.); CM ŠDR id. ['ʃaddεɪ] <i>D perf. 3rd m.sg.</i> he sent; IV.11, IV.31, IV.35, IV.36, IV.73, IV.74, IV.90
	qəmšadder	[qml $fadd\epsilon I$] <i>D imperf.</i> 3^{rd} <i>m.sg.</i> he will send; IV.98
š- <u>k</u> - <u>b</u>		v.i. to lie down (Ahwaz); CM ŠKB id.
<i>š-</i> г- ^w / _y	qəšaryāte	<i>v.t.</i> to loosen, untie, free; open (a bottle); CM ŠRA id. $[q_{9}, j_{a1}]_{j_{o2}}$ (a <i>inperf.</i> 2^{nd} sg. + 1^{st} sg. you will release me; IV.119
	qəšarina <u>k</u>	[$q \Rightarrow [\varepsilon' rima \chi]$ <i>G imperf.</i> $1^{st} sg. + 2^{nd} sg.$ I will release you; IV.120
š- [₩] / _y -1	šiyelti šiyelni qəmšiyellu	<i>v.t.</i> D to ask; CM ŠAL id. [$\int I'$ jɛlti] D perf. 1 st sg. + 3 rd m.sg. I asked him; IX.8 [$\int I'$ jɛlni] D perf. 1 st pl. we asked; III.10 [qm $\int I'$ jɛllu] D imperf. 3 rd m. sg. + 3 rd pl. he asks them; IV.12
	qəmšielā qəmšiel'	[qm/ \mathfrak{field}] <i>D imperf.</i> 3^{rd} <i>f. sg.</i> she asks [qm/ \mathfrak{field}] <i>D imperf.</i> 3^{rd} <i>f. sg.</i> (contextual) she asks; II.31, II.15
	qəmšiyel	$[qm \int I' j \epsilon I] D$ imperf. I^{st} sg. (contextual) I ask; III.20
š- ^w / _y -т		<i>v.i.</i> to perform ablutions (not attested in the classical language)
	šamni	[' \int amni] <i>G perf.</i> 1 st pl. we performed our ablutions; III.9
<i>š-^w/_y-<u>t</u></i>		<i>v.t.</i> to drink, suck; to smoke; CM ŠTA id. cf. Arabic <i>š-r-b</i> 'to drink; to smoke (cigarettes, etc.)' and Turkish <i>içmek</i> id.

	qəši <u>t</u> en	$[{}^{l}\widehat{qJi}:\theta \in n]$ <i>G imperf.</i> 3^{rd} <i>pl.</i> they are smoking (hashish); VII.1, VII.6
<i>Šabut</i>		[$\int a'bu\theta$] <i>prop. n.</i> name of Yusef (q.v.); II.11
šam'		[∫am] <i>n</i> . candle; Arabic <i>šam</i> ^c id. via Persian <i>šam</i> id.; IV.20
šamnā		[∫æmnɔ] <i>n</i> . fat; CM šamina id.
Šāmeš		['ʃɒmɛʃ] <i>n</i> . the sun and <i>prop. n</i> . the archon who governs the sun, one of the <i>šəbihānā</i> (q.v.); identified with adunai , the god of the Jews; CM šamiš id.; IV.55, IV.83, IV.135
Šawāli		[ʃɑwɒ'li:] <i>prop. n.</i> Shāh Vāli, a town in northern Khuzestan, Iran, near Shushtar, and formerly the home of a large Mandaean population, the majority of which settled in Ahwaz; III.23
šehyānā	ā šehyāni	[ʃɛh'jɔ:nɔ] or [ʃɪh'jɔ:nɔ] <i>n</i> . ruler; king; prince; PCM šihiana id.; IV.9, IV.14, IV.16, IV.17, IV.22, IV.36, IV.38, IV.42, IV.64, IV.80, IV.81, IV.116 [ʃɛ'hjɔ:ni] <i>indef</i> . ruler; IV.2
šekəl	šekəl-da	[' \mathfrak{fek}] <i>n</i> . appearance; Arabic <i>šikl</i> id. and Persian <i>šekl</i> id. [' \mathfrak{fek}]da] + 3^{rd} f.sg. her appearance, II.33
šenā		[∫ɛ'nɔ:] <i>n</i> . swimming; Persian <i>šenā</i> id.
šenā '-b॒-d šenā əb॒adni		<i>v.i.</i> to swim; Persian <i>šenā</i> 'swimming,' cf. Persian <i>šenā kardan</i> id. [∫ε'nɔ: ə'wɑdni] <i>G perf. 1st pl.</i> we swam; III.9
šennā	šenāni	[' \mathfrak{fenn}] <i>n</i> . tooth; CM šina 1 id. [\mathfrak{fe} 'no:ni] <i>pl.</i> + \mathfrak{I}^{rd} <i>m.sg.</i> his teeth; V.12
šerrā	šer horettā	['∫ɛrrɔ] n. war (f.); possibly Arabic <i>šiıra</i> 'disaster, calamity' [∫ɛı ho'rɛttɔ] Second World War; III.2

356		8. Lexicon
šerrā '- <u>b</u> -d		<i>vi.</i> to wage war; possibly Arabic <i>širra</i> 'disaster, calamity,' cf. Persian <i>jang kardan</i> id.
šeršā	šerš šeršan šeršān	[¹ \mathfrak{fel} 5] <i>n</i> . religion; CM širša 'tribe; community; religion' [\mathfrak{fel}] (contextual) religion; I.3, III.12 [¹ \mathfrak{fel} æn] + 1 st pl. our religion; I.7, I.8, I.13 [\mathfrak{fel} fon] pl. (contextual) religions; I.13
šettā	šet	['∫ɛttɔ] <i>n</i> . year; CM šidta id.; II.4, IV.34, IV.40, IV.43, IV.59 [∫ɛt] (contextual) year; II.4, III.1, III.18
<i>šə<u>b</u>ābā</i>	šə <u>b</u> ā <u>b</u> āni šə <u>b</u> ā <u>b</u> ānan	[$\int \vartheta' w \vartheta: w \vartheta$] <i>n</i> . neighbor; CM *šababa id.; X.13 [$\int \breve{u} w \vartheta' w \vartheta: n i$] + 3 rd m.sg. his neighbors; X.1, X.3 [$\int \vartheta w \vartheta' w \vartheta: n \mathring{e} n$] + 1 st pl. our neighbors; I.14
šə <u>b</u> iāhā		[∫əvɪˈjɔ:hɔ] <i>n</i> . evil spirit; CM šibiahia 'planets; archons;' IV.13, IV.45
	šəbihānā	[ʃəviˈhɔ:nɔ] <i>pl</i> . evil spirits; IV.14, IV.15
šənārā	šənartā	[ʃəˈnɔːrɔ] <i>n</i> . (tom) cat; CM šunara id. [ʃəˈnaɹtɔ] <i>f</i> . (queen) cat; CM šinarta id.
šətā		See <i>eštā</i>
šiex		[$\int i \tilde{z} \chi$] <i>n</i> . sheikh; Arabic $\tilde{s}ayx$ 'old man;' II.7, II.8, II.27, II.29
šlihā		[¹∫li:hɔ] <i>n</i> . messenger, apostle; CM šliha id.; 1.9
šom		See ešmā
Šom Yāwar		[∫om 'jɒwaɪ] <i>prop. n.</i> Name of Yāwar; CM šum iauar id.
Šom Heyyi		[∫om ˈhɛɪji] <i>prop. n.</i> Name of Life; CM šum hiia id.
šu <u>b</u> bā		[¹ Juwwo] <i>quant.</i> seven; CM šuba id.; III.18
<i>Šuštar</i>		[¹ ʃuʃtæ1] <i>prop. n.</i> Shushtar, a city in northern Khuzestan, formerly home to a large Mandaean population, the greater part of which settled in Khorramshahr; III.19, III.23, IV.2, IV.24, IV.30, IV.65, IV.68, IV.99, IV.103, IV.105, IV.105
	Šuštrāyā	[ʃuʃ'trɔ:jɔ] adj. pertaining to the city of Shushtar

Т

t- <u>b</u> -r	etber	<i>v.t.</i> to break (down, spoil, destroy; CM TBR id. ['stfɛɹ] <i>tG perf.</i> 3 rd <i>m.sg.</i> it broke (middle); IX.12
<i>t-1-</i> ^w / _y		<i>v.t.</i> to pull; CM TLA id.
<i>t-m-m</i>	tam tam qāri tammon	<i>v.i.</i> to become; $(+ q\bar{a}r)$ to give birth; CM TMM id.; see also <i>həyānā t-m-m, hāzér t-m-m, narm t-m-m, pešimān</i> <i>t-m-m, ser t-m-m, sewār t-m-m, tamām t-m-m, wayyeq</i> <i>t-m-m,</i> and <i>wel t-m-m</i> [tæm] <i>G perf.</i> 3^{rd} <i>m.sg.</i> he became; IV.30 [tæm 'qɔ:ri] <i>G perf.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>m.sg.</i> he gave birth to it; X.6, X.17 ['tæmmon] <i>G perf.</i> 3^{rd} <i>pl.</i> they remained; IV.144, IV.148
	tamini	[tæ ^l mi:ni] G perf. 1^{st} pl. we became; 1.14
$t-\bar{p}-w/_y$	tāpi	<i>v.t.</i> to cook; CM TPA 4 (from APA 1) id. [t to:fi] <i>G subjv.</i> 3^{rd} <i>m.sg.</i> he cooks; X.1
tā		[to] demonstr. pron. that; PCM ta id.; IV.33, IV.39
tabliģí		[tæbliˈʁiː] <i>adj.</i> proselyte; Arabic <i>tablīģī</i> id. via Persian <i>tabliģi</i> id.; 1.13
tafšā		['tæfʃɔ] <i>n</i> . stewpot; Arabic <i>tafša</i> id.; X.10
tājér	tājéri	[tɒ ¹ dʒɛ1] <i>n</i> . merchant; Arabic $t\bar{a}jir$ id. via Persian $t\bar{a}j\acute{e}r$ id.; V.5, V.6, V.12, V.14, V.21, V.26 [tɒ ¹ dʒɛri] <i>indef</i> . a merchant; V.1
talgā		['tælgo] n. snow, ice; CM talga id.
talg n-h	- <u>t</u>	<i>v.i.</i> to snow; CM talga 'snow,' cf. Persian <i>barf āmadan</i> id.
tamām ⁻	2- <u>b</u> -d	<i>v.t.</i> to complete; Arabic <i>tamām</i> 'perfection, completeness,' via Persian <i>tamām</i> id., cf. Persian <i>tamām</i> <i>kardan</i> 'to complete;' see also <i>tamām t-m-m</i>
	tamām abdelli	[tæ'mom av _i delli] <i>G perf.</i> 3^{rd} <i>pl.</i> + 3^{rd} <i>m.sg.</i> they completed it; IV.5

358 8. Lexicon v.t. to be complete, finished; Arabic tamām 'perfection, tamām t-m-m completeness,' via Persian tamām id., cf. Persian tamām tamām šodan 'to be complete;' other tenses take h-w-y; see also tamām '-b-d [tæ'mpm 'tæmmɔ] G perf. 3^{rd} m. sg. it was finished; IV.25 tamām tammā [tæ'mpm qə'hɔ:vi] G imperf. 3rd m. sg. it will be finished tamām gəhāwi tarmidā [taɪˈmi:dɔ] *n*. the first grade of the Mandaean priesthood; CM tarmida id.: IV.75. IV.88 tarmid [ta1'mid] (contextual) priest; IV.37, IV.82 tarmidi [ta1'mi:di] indef. a priest; II.8, II.21, II.25 tarmidānā [ta.mi'do:no] pl. priests; IV.81, IV.90 tel [tel] n. hill; Arabic tall id. via Persian tel id.; V.18 təmānā [tə'mɔ:nɔ] quant. eight; CM tmana id.; II.4 (x2) tiz [tiz] *n*. sharp; Persian *tiz* id. [tren] quant. two; CM trin id. tren tu See *tum* 3 tul See tut tulyā ['tʌljɔ] *n*. worm; CM tulita id. tum 1 [tum] *conj.* then, after that; (+ neg.) any longer; CM tum 1 id.; II.12, IV.5, IV.21, IV.24, IV.61, IV.71, IV.79, VII.7, VI.4, VIII.7, VIII.13, IX.4 tum 2 [tum] *prep.* until, for (the duration of a period of time); CM tum 1 'then;' I.7, I.14 (x2), II.33, IV.3, IV.70, IV.87, IV.123, IV.126 tum 3 [tum] adv. again (follows noun), also; III.18, IX.2 [tu] adv. again (follows noun), also; BTA tū id.; III.21 turā ['tu:ro] n. bull; CM taura id. [tu'rexto] n. cow (f.); CM turita id. turektā turtā ['tA.tto] n. cow (f.); CM turta id.

	turātā	[tʌ'rɔ:θɔ] <i>pl.</i> cows; IV.124, IV.142
tu <u>t</u>	tul tu <u>t</u> i	<i>prep.</i> under; CM atutia id. [tul] under; IV.100 ['tu: θ i] + 3 rd m.sg. under it; VII.2
		Ţ
<u>ț</u> -1-b	țəlab qəțāleb	v.t. to ask; (+ <i>men</i>) to request s.o.; Arabic <i>t-l-b</i> id. $[t^{s} \vartheta^{l} lab] G perf. 3^{rd} m.sg.$ he requested; X.1, X.9 $[q \vartheta^{l} t^{s} \vartheta^{l} lab] G imperf. 3^{rd} m.sg.$ he requested; VI.2
<u>ț-l-l</u>	qəmțallen qəmțallenni qəmțallén	<i>v.i.</i> D to play; CM ȚLL id.; Tab. 1.5 [qml ^t ^s allɛn] <i>G imperf.</i> 3 rd pl. they play [qmt ^s all ^l ɛnni] G <i>imperf.</i> 1 st pl. (lexical) we play [qmt ^s all ^l ɛn] G <i>imperf.</i> 1 st pl. (contextual) we play; II.4, II.13
<i>ṭ-[₩]/_y-ḥ</i>	qəmṭāḥen	v.t. D to toss; Arabic <i>t-w-h</i> id. [qm ^t t ^s p:ħɛn] D imperf. 3^{rd} pl. they tossed; IV.19
<u>țā</u> bā		[^t t ^s o:wo] <i>adj.</i> good CM taba id.; IV.38
țabutā '-	bd tabutā qābed	<i>v.i.</i> to hold a banquet; CM tabu(ta) 'ritual food; a banquet' [$t^{S}a^{I}wu:\theta_{2}$,qp:ved] <i>G imperf.</i> 3^{rd} <i>m.sg.</i> he holds a banquet; IV.22
țawšā		['t ^s ɑʊʃə] <i>n</i> . dirt, ashes; CM ṭupša id.
țawšānā		[t ^s au'ʃə:nə] <i>adj.</i> dirty; CM tupšana id.
țurā		['t ^s u:ro] <i>n</i> . mountain; CM țura id.
Țurā d-Madday		['t [°] u:rɔd mad'daɪ] <i>prop. n.</i> the "Mountain of Media" (or "Median mountain country"); CM tura <u>d</u>-madai id.

U

и	[u] <i>conj.</i> and (independent form); CM u id.; I.3, I.5 (x2), I.6,
	I.9, I.10, I.11 (x2), I.13 (x2) , I.14 (x5), II.21, II.22, II.25 (x2), II.29,
	II.33, III.1 (x2), III.2, III.6, III.7, III.9, III.18 (x3), III.21, III.22, III.23
	(x2), IV.16 (x2), IV.17, IV.18, IV.19, IV.20 (x2), IV.22, IV.28 (x2),
	IV.34, IV.35, IV.37, IV.40 (x2), IV.41, IV.44, IV.45, IV.47, IV.55,
	IV.59, IV.61, IV.64, IV.65, IV.70, IV.75, IV.76, IV.81 (x3), IV.82,
	IV.88, IV.98 (x3), IV.102 (x2), IV.103, IV.105, IV.109, IV.110,
	IV.116, IV.118 (x2), IV.123, IV.124 (x4), IV.128, IV.132, IV.137,
	IV.139 (x3), IV.142 (x3), IV.145, IV.148 (x2), IV.124, IV.137, IV.104,
	V.4, V.5, V.12, V.18, V.25, V.26, VI.2 (x4), IX.13, X.3, X.8, X.9
WƏ-	[wə] and (proclitic); IV.17
<i>W</i> -	[w] and (proclitic, before vowels); I.7, 1.8, III.22, IV.11, IV.16,
	IV.17, IV.84, IV.105, IV.108, V.18, VIII.15
WC-	[wɛ] and (proclitic, before clusters); I.12 (x2), I.14, II.33

W

W-

See *u*

w - d - $w/_y$		<i>v.t.</i> D to take; Arabic <i>w-d-</i> ^{\circ} II 'to see off, bid farewell' or Arabic <i>w-d-</i> ^{\circ} IV 'to entrust'
	qəmwadin <u>k</u> on	[qmvɛ'din χ on] <i>D</i> imperf. 1^{st} sg. + 2^{nd} pl. suff. I will take
	qəmwadden qəmwadyātan	you; IV.119, IV.120, IV.121, IV.124 [qmˈwaddɛn] D imperf. 3^{rd} pl. they take; III.3 [ˌqmˈwadˈjɔ:tæn] D imperf. 2^{nd} sg. + 1^{st} pl. you take us;
	əmwadyātan	IV.122 [mwad'jo:tæn] D subjv. 2^{nd} sg. + 1^{st} pl. you take us; IV.118
w- ^w / _y -q		<i>v.t.</i> D to tie; cf. CM HUQ 1 'to be distressed,' cf. Aramaic '- <i>y</i> - <i>q</i> id. and Arabic <i>dayyiq</i> 'tight'
wa-lakin		[wa'lækın] <i>conj.</i> but; Arabic <i>wa-lākin</i> id.; IV.120
wa <u></u> hši		['wæħʃi] n. time; Arabic waḥšī "wild"
waxt	awqā <u>t</u>	$[væ\chi t]$ <i>n</i> . time; Arabic <i>waqt</i> id. via Persian <i>vaxt</i> id.; IV.18 [au'qa θ] <i>pl</i> . times

	waxti ke a <u>k</u> waxt	[¹ væxtike] <i>conj.</i> when (lit. the time that); II.1, V.9 [¹ axwaxt] then, that time; II.2, II.21, III.5, III.6, III.7, III.14, III.15, III.20
	kol waxt	[ˈkolwɑɣt] always; I.13
wāy	wāyla <u>k</u>	<i>interject.</i> woe; Arabic $w\bar{a}y$ id., Persian $v\bar{a}y$ id. ['waıla χ] + 2 nd m.sg. woe to you; VII.5a
wayyeq		[¹ vɑjjɛq] <i>n</i> . distress; CM HUQ 1 'to be distressed,' cf. Aramaic '- <i>y</i> - <i>q</i> id. and Arabic <i>dayyiq</i> 'tight;' IV.8, IV.9, IV.28, IV.30, IV.86
wayyeq '-b-d wayyeq əbad gāwwi		<i>v.t.</i> to distress, threaten $(+ g\bar{a}w$ s.o.) ['vajjɛq ɛ ₁ vad 'gɔʊvi] <i>G perf.</i> 3^{rd} <i>m.sg.</i> he threatened him (lit. he made distress in him); IV.86
wayyeq ' <u>-t</u> - ^w / _y wayyeq qā <u>t</u> i gāwwi		v.i. $(+ g\bar{a}w)$ to become upset, lit. distress came into s.o. ['vajjɛq 'qɔ: θ i 'gɔ: v i] <i>G imperf.</i> 3^{rd} <i>m.sg.</i> he became upset (lit. distress came in him); IV.8, IV.9
wayyeq t-m-m wayyeq tam gāwwu		<i>v.i.</i> to be distressed, lit. distress became in s.o. ['vajjɛq tam 'gɔuwu] <i>G imperf.</i> 3 rd pl. they were distressed (lit. distress became in them); IV.28, IV.30
wazir	wazirā	[vɛ'zir] <i>n</i> . vizier; Arabic <i>wazīr</i> id. via Persian <i>vazir</i> id. [vɛzi'rɔ:] pl. ministers; IV.11, IV.70, IV.73, IV.104, IV.109
wel t-m	-m	[vel] <i>n</i> . loose, detached; Persian <i>vel</i> id.; 3.39
wel t-m	-m wel tammat	<i>v.i.</i> to become loose, detached; Persian <i>vel šodan</i> id. [vɛl 'tæmmat] <i>G perf.</i> 3^{rd} <i>f.sg.</i> it (the leg) became loose; VII.5
welāt	welāti	[vε'lɔt] <i>n.</i> city, country; Arabic <i>wilāyah</i> 'province'; cf. Kurdish <i>wilat</i> , III.11, IV.20, IV.83, IV.92, IV.99, IV.103, IV.124, V.20 [vε'lɔ:ti] <i>rest.</i> the country (that); V.26
		[.e.tota] rest the country (that), v.20
willā		[wıl'lɔ:] <i>interject.</i> oh! (surprise); Arabic <i>wa-llāhi</i> id.; VII.8
wusle		[wAs ^{SI} le:] <i>n.</i> piece; Arabic <i>wuṣla</i> 'connecting piece, junction,' via Persian <i>vosle</i> id.; IX.12

Х

xabár	xabár məṭā qāru	[$\chi \alpha^{l}bar$] <i>n</i> . news, word; Arabic <i>xabar</i> id. via Persian <i>xabar</i> id.; IV.145 [$\chi \alpha^{l}bar$ mə't [§] o 'qo:ru] <i>G perf.</i> 3 rd m.sg. word reached them; cf. Persian <i>xabar rasid ke</i> 'word arrived that ;' IV.145
xabár '		v.t. to inform; Persian <i>xabar dādan</i> id. [$\chi a'baı qa'hevna\chi$] <i>G imperf.</i> $1^{st} sg. + 2^{nd} sg.$ I will let you know; IV.87
xāle		[' $\chi p:le$] <i>n</i> . (maternal) aunt; Arabic $x\bar{a}la$ id.
xālu		['xɒ:lu] n. (maternal) uncle; Arabic xāl id.
xāne		['zp:ne] <i>n</i> . house; Persian xāne id.; III.12 (x2)
xarj ⁻ -b-o	d xarj-di ə <u>b</u> ad	<i>v.t.</i> to spend (- <i>d</i> - s.t.); Persian <i>xarj kardan</i> id. [' χ aıʒdi ₁ wad] <i>G perf.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>m.sg.</i> he spent it; V.5
xiyābān	xiyābānhấ	[χijɒ'bɔn] <i>n.</i> avenue; Persian <i>xiyābān</i> id. [χijɒbɔn'hɒː] <i>pl.</i> avenues; III.22
Xorrams	šahr	[χorrɑmˈʃæɪ] <i>prop. n.</i> Khorramshahr, Persian name for Mohammerah after 1924; II.29
xorde	xorde-xorde	[χοι'de:] <i>adj.</i> little; Persian <i>xorde</i> id. [χοι'de: χοι'de:] <i>adj.</i> little by little
xuțb		[$\chi \Lambda t^{s}$] <i>n</i> . letter; cf. Arabic <i>xuțba</i> id.

Y

y-h-m	yəhem yəhom yəhomyon yəhem yəhemye yəhimen ayhimi ayhimu	v.i. to sit; CM YTB id. [jə ¹ hem] <i>G perf.</i> 3^{rd} <i>m.sg.</i> he sat; VI.2 [jə ¹ hom] <i>G impv. m.sg.</i> sit; IV.140 [jə ¹ homjon] <i>G impv. pl.</i> sit; I.5, IV.123, IV.123, IV.125 [jə ¹ hem] <i>G pass. ptc.</i> sitting; IV.144, VII.2 [jə ¹ hemje] <i>G pass. ptc.</i> + 3^{rd} <i>m.sg. cop.</i> he is sitting; VI.3 [jə ¹ hi:men] <i>G pass. ptc. pl.</i> sitting; VII.6 [ar ¹ hi:mi] <i>C perf.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>m.sg.</i> he made him sit; IV.77 [ar ¹ hi:mu] <i>C perf.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>pl.</i> he made them sit; IV.76
y-1-p		Ahwaz variant of '- <i>l-p</i> (q.v.)
<i>y-m-^w/_y</i>	yəmā qəyāmet yəmi qəmaymina <u>k</u>	<i>v.t.</i> G to swear, C to promise; CM YMA id. [jə'mə:] or [i'mə:] <i>G perf.</i> 3^{rd} <i>m. sg.</i> he swore; IV.121 [qə'jə:met] <i>G imperf.</i> 2^{nd} <i>sg.</i> you swear; IV.119 [jə'mi:] <i>G impv. m.sg.</i> swear; IV.120 [qəmat'mi:nɑx] <i>C imperf.</i> 1^{st} <i>pl.</i> + 2^{nd} <i>m.sg.</i> I promise you; VIII.16
ya		[jæ] <i>quant.</i> one, a (single); Persian <i>yak</i> id. and <i>ya</i> id.; III.16, IV.94, IV.125, V.1, V.4, V.8, V.21 (x2), VI.1, VI.3, VII.1, VII.2, VIII.1, IX.1, IX.2, X.1, X.9, X.10
	yeki	[¹ jɛki] <i>indef. pron.</i> someone; Persian <i>yaki</i> id.; II.10, III.8, VI.1, VI.3, VI.6, VII.2, VII.8, VIII.1, IX.1, IX.2
	yektā	['jɛktɒ] <i>adv.</i> once; Persian <i>yak tā</i> 'simple, single;' III.18
yā		[jp:] <i>interject.</i> oh (vocative); Arabic <i>yā</i> id.; IV.40, IV.23, VII.5
ya'ani		['ja?ani] or ['ja:ni] <i>interject.</i> meaning, that is to say, etc.; Arabic <i>ya'ni</i> 'it means;' II.2, II.29, II.32, III.14, III.18, III.18, III.23, V.11
Yabber		['jæbbṛ] <i>prop. n.</i> Yabber, a friend of Nasser Sobbi, who accompanied him to Shushtar; III.8 (x2)
yabušā		[jæˈbu:∫ɔ] <i>adj</i> . dry

364	8. Lexicon
yā	[jɔ] <i>conj.</i> or; Persian yā id.
yād láqyādi láqyāde	[jɔd] <i>n</i> . memory; Persian $y\bar{a}d$ id. ['laq,jɔ:di] <i>G neg. imperf.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>m.sg.</i> he doesn't know; VII.4 ['laq,jɔ:de] <i>G neg. imperf.</i> 3^{rd} <i>m.sg.</i> + 1^{st} <i>sg.</i> I don't know; II.25 (x2)
qəyādu láqyādu	[qə'jɔ:du] <i>G imperf.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>pl.</i> they know; IV.15 ['laq,jɔ:du] <i>G imperf.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>pl.</i> they don't know; IV.10
yalupā yəlupānā	[jæˈluːfɔ] <i>n.</i> scholar; CM ialupa orig. 'disciple,' hence 'literate person, one who can read and write Mandaic' [jəluˈfɔːnɔ] <i>pl.</i> scholars; 1.5
yaminā	[jɛˈmiːnɔ] <i>adj.</i> right; CM iamina id.
yammā	['jæmmɔ] <i>n</i> . sea, lake; CM iama id.
yanqā	['jænqɔ] n. infant; CM ianqa id.
yaroqā	[jæ'ro:qo] adj. green; cf. CM iaruqa 'herbs, green-stuff'
yeki	See ya
yektā	See ya
yəhuwəțāyā	[jəhuwəˈt [°] ɔ:jɔ] <i>adj.</i> Jewish; CM iahuṭaiia id.; 1.12
yəmu <u>t</u> ā	[jə ['] mu:θɔ] <i>n</i> . oath; cf. CM 'umamata id.; IV.120
yumā ā yumā yum yumi Yusef	[¹ ju:mɔ] <i>n.</i> day; CM iuma id.; IV.123 (x2), IV.126, IV.128, IV.129 [ɔ ¹ ju:mɔ] <i>adv.</i> today; cf. CM iumai id.; I.7, IV.71 [jum] (contextual) day; I.7 [¹ ju:mi] <i>indef.</i> a day; V.6, IX.2, X.1 [¹ ju:sɛf] <i>prop. n.</i> Yusef, the bridegroom at the wedding in
	Khorramshahr, also known as Šabu <u>t</u> (q.v.); II.10, II.11

Ζ

<i>z-<u>b</u>-n</i>	za <u>b</u> nu zabnu qəmzabnen əmzabnen	<i>v.t.</i> G to buy, D to sell; CM ZBN id. ['zavnu] <i>G perf.</i> 3^{rd} <i>m.sg.</i> $+ 3^{rd}$ <i>pl.</i> he bought them ['zabnu] <i>D perf.</i> 3^{rd} <i>m.sg.</i> $+ 3^{rd}$ <i>pl.</i> he sold them; v.5 [qm'zabnen] <i>D imperf.</i> 1^{st} <i>pl.</i> we sell; III.4 [m'zabnen] <i>D subjv.</i> 1^{st} <i>pl.</i> we sell; III.6
z-h-r	əmzaharla	<i>v.t.</i> D to protect; CM ZHR id. [mza'hæıla] <i>D subjv.</i> 3^{rd} <i>m.sg.</i> + 3^{rd} <i>f.sg.</i> may he protect her; II.35
z-m-r		<i>v.t.</i> and <i>v.i.</i> to sing; CM ZMR id.
Z-1 ⁻ -Z	zarzat	<i>v.t.</i> to inform; CM ZRZ 'to equip, be prepared' ['zæ.zɑt] <i>G perf.</i> 3 rd <i>f.sg.</i> she informed; I.6
zāġ	zehunān zehunānķon	[ZDB] n. chicken; Arabic $z\bar{a}g$ 'crow' via Persian $z\bar{a}g$ id. [Zɛhu'nõ] pl . chickens; IV.142 [Zɛhu'nɒnχon] pl . + 2^{nd} pl . your chickens; IV.124
zah <u>b</u> ā		['zæhwo] n. gold; CM zahba id.
zalilā		[zæ'li:lɔ] <i>adj.</i> narrow; CM zalila light, fine, slender; Tab. 1.5
zammā		['zæmmə] n. blood; CM zma id.
zardoští		[zæ.tdoʃ'ti:] adj. Zoroastrian; Persian zardošti id.; I.12
zipā	zip	[¹ zi:fɔ] <i>n</i> . lie; CM zipa id.; cf. Arabic <i>zayf</i> 'falseness' [zif] (contextual) lie; IV.132, IV.134, IV.147, V.23, IX.1, IX.2, IX.14
zipā '-h-	b zip qāheb	<i>v.i.</i> to lie, deceive; cf. Persian <i>farib dādan</i> id. [zif 'qɔ:hɛv] <i>G imperf. 3rd m.sg.</i> he lies; IV.132, IX.1, IX.2, IX.14
	zip láqāheb	[zif 'la ₁ qo:hev] <i>G neg. imperf.</i> 3^{rd} <i>m. sg.</i> he doesn't lie; IV.134, IV.147
	zip̄ qahḇət	[zif 'qahvet] G imperf. 2^{nd} sg. you are lying; V.23

366		8. Lexicon
zipānā	zipāni	[zi ¹ fɔ:nɔ] <i>n</i> . liar; CM zipana 'false; cheat, deceiver;' IX.13 [zi ¹ fɔ:ni] <i>indef</i> . a liar; IX.3
ziwā		['zi:wə] n. splendor; CM ziua id.; 1.9
zobún	zobúni	[zo'bun] <i>n</i> . (under-)shirt; Persian <i>zobun</i> id. [zo'bu:ni] + 3^{rd} <i>m.sg</i> . his undershirt; VIII.7
- Altoma, Salih L.: The Problem of Diglossia in Arabic: A Comparative Study of Classical and Iraqi Arabic. Cambridge, MA: Harvard University Press, 1969.
- Arnold, Werner: Review of Neumandäische Texte im Dialekt von Ahwāz, by Rudolf Macuch. Zeitschrift der deutschen morgenländischen Gesellschaft 146 (1996): 199–203.
- Blanc, Haim: *Communal Dialects in Baghdad*. Cambridge, MA: Harvard University Press, 1964.
- Borghero, Roberta: "A 17th Century Glossary of Mandaic." *ARAM* 11–12 (1999–2000): 311–19.
- Brandt, A.J.H. Wilhelm: *Mandäische Schriften. Aus der grossen Sammlung heiliger Bücher Genza, oder Sidra Rabba, übersetzt und erläutert.* Göttingen: Vanderhoeck and Ruprecht, 1893.
- Buckley, Jorunn J.: "A study of the two liturgical collections in J. de Morgan's *Textes Mandaïtes.*" Le Muséon 104 (1991): 191–203.
- Buckley, Jorunn J.: The Scroll of Exalted Kingship (Diwan malkuta 'laita), Mandaean Manuscript No. 34 in the Drower Collection, Bodleian Library, Oxford. New Haven, CT: American Oriental Society, 1993.
- Buckley, Jorunn J.: "Glimpses of a Life: Yahia Bihram, Mandaean Priest." *History* of *Religions* 39.1 (1999): 32–49.
- Buckley, Jorunn J.: "Foreword" in Ethel S. Drower: *The Mandaeans of Iraq and Iran* by , i*–v* (2nd edition reprint). Piscataway, NJ: Gorgias Press, 2002.
- Buckley, Jorunn J.: *The Mandaeans: Ancient Texts and Modern People*. Oxford: Oxford University Press, 2002.
- Budge, E.A. Wallis: Amulets and Superstitions, 1st edition reprint. New York: Dover, 1978.
- Clevenot, Michel (ed.): L'état des religions dans le monde. Paris: Editions La Découverte, 1987.
- Croft, W.: *Typology and Universals*, 2nd ed. Cambridge: Cambridge University Press, 2003.
- Curzon of Kedleston, George Nathaniel Gordon, Marquess: *Persia and the Persian Question*, vol. II. London: Longman and Co., 1892.
- Drower, Ethel S.: "A Mandaean Book of Black Magic." *Journal of the Royal Asiatic Society* (1943): 149–81.
- Drower, Ethel S.: *The Haran Gawaita and the Baptism of Hibil-Ziwa*. Vatican City: Biblioteca Apostolica Vaticana, 1953.
- Drower, Ethel S.: Canonical Prayerbook of the Mandaeans. Leiden: E.J. Brill, 1959.

- Drower, Ethel S.: *The Secret Adam: A Study of Nașorean Gnosis.* Oxford: Clarendon, 1960.
- Drower, Ethel S.: *The Mandaeans of Iraq and Iran* (2nd edition reprint). Piscataway, NJ: Gorgias, 2002.
- Drower, Ethel S. and Rudolf Macuch: A Mandaic Dictionary. Oxford: Clarendon, 1963.
- Dundes, Alan: "How Indic Parallels to the Ballad of the 'Walled-Up Wife' Reveal the Pitfalls of Parochial Nationalistic Folkloristics" in *Journal of American Folklore* 108, no. 427 (Winter, 1995), 38–53.
- Ebadirad, Julie A.J.: "Information on Mandaeans in Iran in Regards to Human Rights" in *ASUTA* (Jan. 2003).
- Euting, Julius: Qolasta oder Gesänge und Lehren von der Taufe und dem Ausgang der Seele als mandäischer Text mit sämtlichen Varianten, nach Pariser und Londoner Manuscripten, mit Unterstützung der deutschen morgenländischen Gesellschaft in Leipzig, autographiert und hg. von J. Euting. Stuttgart: Karl J. Trübner, 1867.
- Euting, Julius and B. Poertner: *Mandaeischer Diwan nach photographischer Aufnahme*. Strassburg: Karl J. Trübner, 1904.
- Fox, Joshua: *Semitic Noun Patterns* (Harvard Semitic Series 52). Winona Lake, IN: Eisenbrauns, 2003.
- Fox, Samuel Ethan: *The Neo-Aramaic Dialect of Jilu*. Wiesbaden: Harrassowitz, 1997.
- di Gesù, Ignazio: Narratio originis rituum et errorum Christianorum S. Joannis, cui adiungitur discursus per modum dialogi in quo confutantur XXXIIII errores eiusdem nationis. Rome: Sacra congregatio de propaganda fide, 1652.
- Gordon, Cyrus H.: Review of Rudolf Macuch: *Handbook of Classical and Modern Mandaic* in *Journal of Near Eastern Studies* 26.2 (1967):133–35.
- Greenfield, Jonas C.: Review of Rudolf Macuch: *Handbook of Classical and Modern Mandaic* in *Oriens* 21 (1968): 391–94.
- Häberl, Charles G.: "Iranian Scripts for Aramaic Languages: The Origin of the Mandaic Script" in *Bulletin of the American Schools of Oriental Research* 341 (2006): 53–62.
- Häberl, Charles G.: "The Relative Pronoun <u>d</u>- and the Pronominal Suffixes in Mandaic" in *Journal of Semitic Studies* 52.1 (2007): 71–78.
- Häberl, Charles G.: Review of J.B. Segal and E.C.D. Hunter: *Catalogue of the Aramaic and Mandaic Incantation Bowls in The British Museum* in *Orientalistische Literaturzeitung* (forthcoming).
- Heinrichs, Wolfhart: "Introduction" in Wolfhart Heinrichs (ed.): Studies in Neo-Aramaic, ix-xvii. Atlanta, GA: Scholars, 1990.
- Houtum-Schindler, A.: "Notes on the Sabians" in *Proceedings of the Royal Geographic Society* (Nov. 1891): 665.
- Hunter, Erica C.D.: "Two Mandaic Incantation Bowls from Nippur" in *Baghdader Mitteilungen* 25 (1994): 605–18.

- Ingham, Bruce: "Languages of the Persian Gulf" in Alvin J. Cottrell (ed.): *The Persian Gulf States: A General Survey*, 323–24. Baltimore: Johns Hopkins University Press, 1980.
- Jastrow, Otto: Review of Rudolf Macuch: Neumandäische Chrestomathie mit grammatischer Skizze, kommentierter Übersetzung und Glossar. Unter Mitwirkung von Klaus Boekels in Bulletin of the School of Oriental and African Studies 55.3 (1992): 544–46.
- Khan, Geoffrey: A Grammar of Neo-Aramaic: The Dialect of the Jews of Arbel. Leiden: Brill, 1999.
- King, Karen L.: What is Gnosticism? Cambridge: Belknap, 2003.
- Krotkoff, George: A Neo-Aramaic Dialect of Kurdistan. New Haven, CN: American Oriental Society, 1982.
- Lehmann, C.: "Directions for Interlinear Morphemic Translations" in *Folia Linguistica* 16 (1983): 193–224.
- Lidzbarski, Mark: "Ein mandäisches Amulett" in G.C.C. Maspero (ed.): Florilegium ou recueil de travaux d'érudition dédiés à M. le Marquis Melchior de Vogüé à l'occasion du quatre-vingtième anniversaire de sa naissance, 349–73. Paris: Geuthner, 1909.
- Lidzbarski, Mark: Das Johannesbuch der Mandäer. Giessen: Töpelmann, 1915.
- Lidzbarski, Mark: Mandäische Liturgien, mitgeteilt, übersetzt und erklärt. Berlin: Weidmann, 1920.
- Lidzbarski, Mark: *Ginza: Der Schatz; oder, Das grosse Buch der Mandäer*. Göttingen: Vandenhoeck and Ruprecht, 1925.
- Lupieri, Edmondo: *The Mandaeans: The Last Gnostics*. Translated by Charles Hindley. Grand Rapids, MI: Eerdmans, 2002.
- MacKenzie, D.N.: A Concise Pahlavi Dictionary. London: Oxford University Press, 1971.
- Macuch, Rudolf: "The bridge of Shushtar. A Legend in Vernacular Mandaic with Introduction, Translation and Notes" in Stanislav Segert (ed.): *Studia Semitica Philologica necnon Philosophica Ioanni Bakoš Dedicata*, 153–72. Bratislava: Slovenskej Akademie Ved, 1965.
- Macuch, Rudolf: *Handbook of Classical and Modern Mandaic*. Berlin: De Gruyter, 1965.
- Macuch, Rudolf: Zur Sprache und Literatur der Mandäer. Berlin: De Gruyter, 1976.
- Macuch, Rudolf: Neumandäische Chrestomathie mit grammatischer Skizze, kommentierte Übersetzung und Glossar. Wiesbaden: Harrassowitz, 1989.
- Macuch, Rudolf: Neumandäische Texte im Dialekt von Ahwāz. Wiesbaden: Harrassowitz, 1993.
- Malone, Joseph L.: Review of Rudolf Macuch: *Handbook of Classical and Modern Mandaic* in *Language* 45.1 (1968): 191–203.
- Malone, Joseph L.: "Notes" in Language 45.4 (1969): 951.
- Malone, Joseph L.: "Rules of Synchronic Analogy: A Proposal Based on Evidence from Three Semitic Languages" in *Foundations of Language* 5 (1969): 534–59.

- Malone, Joseph L.: "Systematic Metathesis in Mandaic" in *Language* 47.2 (1971): 394–415.
- Malone, Joseph L.: "Classical Mandaic Radical Metathesis, Radical Assimilation and the Devil's Advocate" in *General Linguistics* 25 (1981): 92–121.
- Malone, Joseph L.: "Modern and Classical Mandaic Phonology" in Alan S. Kaye (ed.): *Phonologies of Asia and Africa (Including the Caucasus)*, 141–59. Winona Lake, IN: Eisenbrauns, 1997.
- Malone, Joseph L.: "Suffix Metastasis and Phonological Change in Classical Mandaic" in *General Linguistics* 36.1–2 (1998): 1–16.
- Malone, Joseph L.: "Metathesis and Antiwedging in Classical Mandaic" in *General Linguistics* 36.4 (1999): 227–55.
- Montgomery, James A.: Aramaic Incantation Texts from Nippur. Philadelphia: The University Museum Press, 1913.
- de Morgan, Jacques: Mission scientifique en Perse, tome V (études linguistiques), deuxième partie: textes mandaïtes. Paris: Imprimerie Nationale, 1904.
- Müller-Kessler, Christa: "Aramaic 'k', *lyk'* and Iraqi Arabic 'aku, māku: The Mesopotamian Particles of Existence" in *Journal of the American Oriental Society* 123.3 (2003): 641–46.
- Najmolmolk, Abdolgaffar: Safarname-ye Khuzestan. Tehran: Elmi, 1962.
- Nöldeke, Theodor: "Ueber die Mundart der Mandäer" in Abhandlungen der historisch-philologischen Classe der königlichen Gesellschaft der Wissenschaften zu Göttingen 10 (1862): 81–160.
- Nöldeke, Theodor: Mandäische Grammatik. Halle: Waisenhaus, 1875.
- Nöldeke, Theodor: Review of Henri Pognon: Inscriptions mandaïtes des coupes de Khouabir in Wiener Zeitschrift für die Kunde des Morgenlandes 12 (1899): 141–47; 353–61.
- Norberg, Matthias: Codex Nasaraeus: Liber Adami appellatus, Syriace transscriptus, loco vocalium, ubi vicem literarum gutturalium praestiterint, his substitutis Latineque redditus a Matth. Norberg. London: Berlingianus, 1815– 16.
- Norberg, Matthias: Lexidion Codicis Nasaraei: cui Liber Adami nomen. London: Berlingianus, 1816.
- Norberg, Matthias: Onomasticon Codicis Nasaraei: cui Liber Adami nomen. London: Berlingianus, 1817.
- Nyberg, Henrik Samuel: A Manual of Pahlavi II: Ideograms, Glossary, Abbreviations, Index, Grammatical Survey, Corrigenda to Part I. Wiesbaden: Harrassowitz, 1974.
- Payne, Thomas E.: *Describing Morphosyntax: A Guide for Field Linguists*. Cambridge: Cambridge University Press, 1997.
- Petermann, J. Heinrich: Reisen im Orient (2 vols). Leipzig: Von Veit, 1860 and 1861.
- Petermann, J. Heinrich: *Thesaurus sive Liber Magnus vulgo Liber Adami appellatus,* opus Mandaeorum summi ponderis (2 vols). Leipzig: J.O. Weigel, 1867.

- Pognon, Henri: "Une incantation contre les génies malfaisants en mandaïte" in *Mémoires de la Société linguistique de Paris* 8 (1894): 193–234.
- Pognon, Henri: Inscriptions mandaïtes des coupes de Khouabir. Paris: Imprimerie Nationale, 1898.
- Propp, Vladimir S.: *Morphology of the Folktale* (translated by Laurence Scott). Austin: University of Texas Press, 1968.
- Prym, Eugene and Albert Socin: *Der neuaramäische Dialekt des Tur Abdin*. Göttingen:Vanderhoech, 1881.
- Rudolph, Kurt: Mandaeism. Leiden: E.J. Brill, 1978.
- Sapir, Edward: *Language: An Introduction to the Study of Speech*. New York: Harcourt Brace, 1921.
- Segal, J.B.: "Obituary: Ethel Stefana, Lady Drower (1 December 1879–27 January 1972)" in *Bulletin of the School of Oriental and African Studies* 35.3 (1972): 621–22.
- Segal J.B. and E.C.D. Hunter: *Catalogue of the Aramaic and Mandaic Incantation Bowls in the British Museum*. London: British Museum Press, 2000.
- Shai, Donna: "A Kurdish Jewish Variant of 'The Bridge of Arta" in Association for Jewish Studies Review 1 (1976): 303–310.
- Siouffi, Nicholas: Études sur la religion des Soubbas ou Sabéens: leurs dogmes, leurs moeurs. Paris: Imprimerie Nationale, 1880.
- Socin, Albert: Die neu-aramäischen Dialekte von Urmia bis Mosul: Texte und Übersetzung herausgegeben (2 vols). Tübingen: Laupp, 1882.
- Tahvildar, Abbas: *Ṣābe'in-e Irān Zamin* (English title: *Baptists of Iran*, photos by Abbas Tahvildar, text by Massoud Fourouzandeh and Alain Brunet). Tehran: Key Press, 2001.
- United Nations High Commissioner for Refugees: Report: Iran: Information on the Mandean Religion and Whether or not the Children of Mandeans are taken from their Homes and Forcibly Converted to Islam by Iranian Authorities, August 26, 1997. http://www.unhcr.org/ (accessed Sept. 1, 2007).
- Vaux, Bert and Justin Cooper: Introduction to Linguistic Field Methods. Munich: LINCOM Europa, 1999.
- Verteegh, Kees: *The Arabic Language*. Edinburgh: Edinburgh University Press, 2001.
- Windfuhr, Gernot L.: *Persian Grammar: History and State of its Study* (Trends in Linguistics State of the Art Reports 12). The Hague: de Gruyter, 1979.
- Yamauchi, Edwin: "The Present Status of Mandaean Studies" in *Journal of Near Eastern Studies* 25.2 (1966): 88–96.
- Yamauchi, Edwin: Gnostic Ethics and Mandaean Origins. Cambridge: Harvard University Press, 1970.
- Yamauchi, Edwin: Review of Rudolf Macuch: Zur Sprache und Literatur der Mandäer in Journal of the American Oriental Society 100.1 (1980): 79–81.
- Yassin, Mahmoud Aziz F.: "Bi-polar Terms of Address in Kuwaiti Arabic" in Bulletin of the School of Oriental and African Studies 40.2 (1977): 297–301.

Abadan (Iran), 29 abstraction, 80, 93, 94, 127 Adam Yuhana, 16, 19 adjectives, XXXIII, 89, 109, 131, 132, 133, 139, 141, 146, 147, 159, 167, 226, 229 comparative, XXI, XXVIII, 89, 123, 146, 148, 262, 346 absolute comparison, 147 relative comparison, 147 superlative, 148 adverbs, XXXIII, 174, 191 Ahwaz, 7, 354, 355, 363 allegro speech, 46, 84, 91, 92, 98, 99, 105 Amara (Iraq), 7 anaptyxis, 50, 63, 67, 68, 69, 70, 72, 85, 86, 93, 104, 105, 108, 120, 179, 187, 191, 195, 199, 219, 220 Anhar, Hirmiz bar, 9, 24, 36, 163, 270apocope, 79, 91-92, 92 apposition, 109 Aramaic, XX, XXI, XXV, XXVI, 1, 2, 3, 11, 12, 13, 20, 22, 23, 26, 80, 82, 88, 89, 96, 102, 103, 114, 123, 132, 134, 178, 191, 200, 222, 234, 247, 256, 259, 263, 264, 265, 269, 336, 360, 361 Samaritan, 11 Tūrōyo, XXI, 13, 20 Baghdad (Iraq), 7, 12, 18, 30, 36, 88, 138, 191, 270, 277, 287, 306 baptism, XX, 1, 19, 52, 54, 124, 125, 190, 191, 192, 221, 222, 223, 225, 252, 275, 337, 340, 352

Basra (Iraq), 7, 9, 12, 14, 16, 30, 294, 320 Buckley, Jorunn J., XX, 3, 5, 8, 11, 17, 19, 21, 24, 25, 32, 33, 165, 268.344Carmelite order, 12, 14, 15 Choheili, Salem, 28, 37, 38, 39, 238, 265 Christians Assyrian, XX, 11, 13, 33 Chaldean, XX, 12, 13, 19, 33 conjunctions, XXXIII, 67, 70, 92, 127, 150, 160, 164, 263 consonants aspiration, 50, 53, 54, 56 excrescence, 85 fricativization, 38, 57, 71, 88, 89, 99, 102, 103, 104, 159, 257 gemination, 67, 70, 75, 76, 90, 94, 95, 114, 115, 116, 127, 186, 199, 200, 202, 207, 208, 209, 212, 217, 219, 257 oral depletion, 90, 221 paragoge, 99 pharyngealization, 47, 53, 54 Damascus (Syria), 7, 11, 13 de Morgan, Jacques, XXVIII, 9, 18, 21, 28, 36, 139, 294 demographics, 9, 10, 355, 356 derived stems, XXV, 50, 79, 174, 199, 224, 253, 260 Dezful (Iran), 8, 9, 37, 39, 99, 100, 242, 268, 278, 279, 311, 344 dialects, 5, 7, 12, 26, 46, 61, 80, 87, 88, 90, 96, 150, 162, 191, 222, 234, 269 Diwān Abātar, 21, 311

Diwaniya (Iraq), 7 Drāši d-Yahya, 21, 312 Drower, Ethel S., XXVI, XXVIII, XXIX, 2, 3, 4, 5, 6, 7, 9, 11, 15, 18, 21, 23, 24, 25, 34, 36, 38, 77, 123, 153, 154, 257, 258, 263, 265, 267, 269, 270, 276, 277, 293, 300, 339 elision, 92, 93, 135, 163, 183, 193, 194, 195, 196, 207, 210, 217, 219, 220.348 English, XXX, XXXI, 11, 16, 31, 34, 35, 37, 39, 40, 41, 42, 43, 44, 54, 111, 124, 127, 139, 233, 277, 329, 337, 343, 344, 351 Envāni, 21, 314 epenthesis, 67, 85, 113, 158 epigraphy, XIX, XXII etymology, 1, 2, 86, 111, 112, 115, 116, 117, 118, 119, 120, 127 Euphrates River, 7 Euting, Julius, 18, 21 ezāfe, XXXIV, 133, 134, 259 gapping, 164, 171 Gesù, Ignazio di, 14, 15, 16, 21 Ginza, 1, 4, 14, 16, 17, 18, 21, 32, 168, 274, 275, 319 Glossarium, 15, 25 Harran Gāwetā, 2 historical grammar, XXI, 256, 265 Hormuz (Iran), 14 Hunter, Erica C.D., 22, 23 Hussein, Saddam, 7, 32 infinitive, XXX, 294 interjections, XXXIII Iran, XIX, XX, XXI, 1, 2, 5, 7, 8, 9, 10, 11, 22, 23, 24, 26, 29, 30, 31, 32, 33, 36, 129, 249, 256, 259, 268, 270, 279, 294, 301, 306, 311, 329, 353, 355 Iraq, XIX, XX, XXX, 1, 2, 5, 7, 8, 9, 10, 11, 12, 13, 20, 23, 24, 30, 36, 38, 61, 88, 102, 153, 191, 245,

257, 258, 265, 270, 277, 301, 306, 329 Islam, 3, 4, 10, 32, 65, 279, 289, 339 Jastrow, Otto, 26, 28, 38, 85 jewelers, 7, 161, 269, 287, 289 Judaism, 3, 4, 11, 13, 20, 22, 32, 33, 80, 126, 247, 269, 275, 283, 355, 364 Karun River, 7, 8, 268 Khamenei, Ali, 33 Khaz'al Khan, 29 Khorramshahr (Iran), XXX, 7, 8, 9, 23, 25, 28, 29, 32, 37, 39, 71, 81, 140, 153, 176, 177, 198, 224, 225, 248, 256, 257, 258, 259, 260, 261, 263, 264, 265, 267, 271, 277, 279, 294, 300, 320, 327, 331, 337, 339, 356, 362, 364 Khuzestan (Iran), 7, 8, 9, 23, 25, 29, 30, 36, 38, 88, 268, 269, 311, 324, 339, 355, 356 kinship terminology, 6, 7 Kirkuk (Iraq), 7 Kurt Rudolph, 10, 36 Kut (Iraq), 7 Kuwait, 30, 31 Lebanon, XX Leipzig Rules, XXX lexical and contextual forms, 21, 35, 59, 64, 65, 66, 69, 70, 74, 75, 77, 79, 87, 88, 91–92, 93, 94, 95, 96, 97, 99, 109, 110, 112, 128, 132, 133, 134, 131-34, 150, 152, 155, 156, 162, 163, 167, 170, 180, 181, 193, 197, 203, 208, 211, 212, 214, 215, 218, 224, 225, 226, 293, 296, 297, 298, 299, 301, 302, 304, 305, 306, 307, 308, 310, 311, 314, 315, 316, 317, 318, 319, 323, 324, 325, 326, 327, 330, 332, 334, 335, 336, 339, 340, 341, 342, 343, 344, 345, 346, 347, 348, 349, 350, 354, 356, 358, 359, 364, 365 Lidzbarski, Mark, 17, 18, 21, 22

loan words, XXIX, XXX, 5, 33, 34, 35, 36, 45, 49, 54, 56, 58, 62, 64, 66, 69, 73, 75, 79, 80, 83, 88, 91, 93, 94, 98, 104, 105, 110, 114, 117, 125, 128, 130, 131, 150, 153, 157, 160, 177, 195, 198, 226, 259, 264.294 etymology, 34, 35, 45, 62, 67, 111, 112, 113, 114, 116, 117, 118, 119, 120, 122, 123, 127 Arabic, 9, 48, 54, 112, 119, 120 Persian, XXIX, 45, 58, 119, 120, 122, 124, 130, 294 possessive suffixes, 80, 158 Lupieri, Edmondo, 2, 12, 13, 14, 16, 19 Macuch, Rudolf, XXVI, XXVII, XXVIII, XXIX, 2, 10, 13, 15, 21, 24, 25, 26, 27, 28, 29, 31, 32, 34, 36, 37, 38, 39, 74, 77, 84, 85, 87, 88, 89, 90, 103, 122, 123, 124, 129, 134, 136, 148, 150, 162, 173, 177, 191, 201, 210, 221, 222, 223, 225, 234, 235, 238, 239, 250, 251, 257, 258, 259, 260, 261, 262, 263, 264, 265, 270, 293, 294 magic, XIX, 1, 116, 135, 281, 338 amulets, XIX, 20, 22 incantation bowls, 22, 23 Malone, Joseph L., 27, 63, 71, 89, 106 Mandā d-Heyyi, 2, 274, 336 Mandaean clergy ešgandā, 5, 124, 315 ganzibrā, 5, 109, 276, 277, 282, 283, 284, 285, 286, 287, 288, 289, 319, 337 rišāmmā, 5, 280, 282, 284, 349 tarmidā, 5, 124, 277, 282, 284, 300, 312, 358 Mandaic archaisms in Neo-Mandaic, 62, 130

Classical, XXV, XXIX, XXXIII, 11, 19, 27, 35, 36, 37, 39, 58, 77, 89, 121, 122, 123, 126, 127, 129, 131, 132, 150, 153, 163, 178, 182, 186, 191, 210, 213, 214, 222, 226, 234, 256-65, 293, 370 Neo-Mandaic subdialects, 256-65 origin of name, 1, 2, 80, 144, 248 Postclassical Mandaic, 265 mandi, 33, 136, 212, 213, 278, 336 Manichaeism, XXII, 275, 336 Marsh Arabs, 7 matronymic names, 133, 313 Mosul (Iraq), XXX, 7, 18, 20, 371 Nabat al-Irāq, 12 Najmolmolk, Abdolgaffar, 8, 36 Nāșeroddin Shāh, 8 Nasoriya (Iraq), 7 Neo-Assyrian empire, 11 Neo-Babylonian empire, 11 Nöldeke, Theodor, 12, 15, 18, 19, 20, 21, 22, 23, 24, 25, 29, 89, 213, 257, 262 nominal augments, 35, 61, 64, 86, 87, 91, 93, 94, 110, 112, 113, 114, 115, 129, 131, 132, 145, 146, 153 Norberg, Matthias, 16, 18 nouns animacy, XXXIII, 90, 118, 127, 128, 129, 152, 167, 168, 324 collective nouns, XXXIII, 131, 316, 329, 342 derivation, 109, 110, 125, 126 gender, 109, 110, 125, 127, 128, 131, 134, 146, 149, 156, 161, 175, 179, 230, 236, 259, 293 identifiability and objective referentiality, XXX, 47, 89, 91, 127, 132, 134, 135, 136, 139-45, 146, 147, 148, 155, 164, 166, 167, 170, 171, 174, 227, 228, 259, 263, 269, 329, 341

noun patterns melody, 110 productive patterns, 89, 97, 102, 121, 122, 125, 126, 127, 129, 202, 214, 223, 226, 254, 260, 262 template, 110, 111, 113, 116, 120, 121, 122 object marker, 82, 92, 135, 136, 135-37.141.142.143.145. 160, 181, 182, 184, 188, 189, 193, 194, 228, 254, 259, 316 possession. 33, 45, 87, 132–34. 136, 141, 143, 151, 152, 153, 151-54, 156, 157-58, 160, 229, 231, 232, 237, 253, 259, 262, 264, 335 proper names, XXX, 22, 140, 143 states, 131-34 absolute, 132, 134, 175 construct, 61, 132, 133, 141, 146, 151, 152, 159, 167 ezāfe, XXXIV, 133, 134, 259 emphatic, 61, 130, 132, 134, 153 numbers cardinal, 134, 148, 149, 151 ordinal, 109, 126, 141, 149 Pahlavi, XXII, 2, 25, 29, 270, 343, 369, 370 Pahlavi, Reza, 29 patronymic names, 87, 133, 313 Petermann, J. Heinrich, 17, 18, 19, 20, 21, 24, 37, 269 Poertner, B., 21 Pognon, Henri, 22, 23 Portuguese, 12, 14 prefix conjugation, XXV, 260 prepositions, XXXIII, 37, 86, 89, 92, 100, 133, 134, 136, 137, 138, 141, 142, 145, 148, 152, 157, 158, 161, 163, 168, 169, 188, 231, 262

pronouns anomalous forms, 79 demonstrative, XXXIII, 133, 140, 141, 143, 155, 161, 162, 163, 166 enclitic forms, 145, 151, 158 indefinite, XXXIII, 141, 142, 155, 158, 164, 167, 168, 169 independent forms, XXXIII, 64, 155.156.159 interrogative, 152, 155, 169, 170, 171, 173, 229, 325 quantifiers, 91, 109, 146, 148, 150, 151, 158, 159, 167, 168, 169 reciprocal, 155, 160 reflexive, 160, 236 relative, XXXIII, 38, 141, 143, 151, 153, 164, 165, 167, 171, 173,263 relativizers, 155, 164, 166, 167, 173 used for emphasis, 156, 160 prothesis, 68, 73 Prym, Eugen, 20 Oal'at Saleh (Iraq), 7 Oolastā, 18, 21, 347 question words, 170, 172, 173, 230 Ourna (Iraq), 7 Ram Zihrun, 19 relativizers, 164, 166, 167 Sabians, 1, 9, 91, 130, 154, 269, 279, 289, 352, 353 Saburi, Nasser, 26, 36, 37, 38, 39, 238 Šāmeš, 231, 283, 285, 287 San Giuseppe, Matteo di, 15 Segal, J.B., 22, 23, sentences complex sentences direct speech, 166, 167 non-restrictive relative clauses, 38, 164, 165, 263, 329

restrictive relative clauses, 38, 141, 143, 144, 164, 165, 260, 263, 329, 332 declarative sentences, 170, 229, 232, 244 interrogative sentences, 152, 155, 169, 170, 171, 173, 229, 325 simple sentences, 263 Shah Wali (Iran), 8, 37, 279 Shatt el-Arab, 8, 9 Shooshtary, Dakhil, XXVII, 31 Shushtar (Iran), 8, 9, 18, 19, 24, 36, 37. 100. 106. 109. 146. 153. 245. 267, 268, 269, 270, 278, 279, 281, 283, 285, 328, 330, 341, 355, 356, 363, 369 Siouffi, Nicholas, XXVIII, 18, 19, 20, 38, 371 Sobbi, Nasser, XX, XXI, XXVII, XXX, 1, 7, 9, 23, 29, 32, 36, 37, 82, 122, 163, 238, 267, 273, 363 Sobbi, Shukrieh, XXI, XXVII, 31, 267 Socin, Albert, 18, 20, 24 sociolinguistics, 46 stress and accent, 46, 59, 60, 65, 75, 76, 77, 78, 79, 80, 81, 77-81, 83, 85, 93, 94, 95, 96, 121, 149, 178, 179, 180, 181, 188, 192, 193, 200, 202, 207, 208, 215, 217, 218, 224, 225, 227, 228, 257, 294 suffix conjugation, XXV Sug ash-Shuyukh (Iraq), 7 Swadesh list, XXVIII, 34, 39, 47, 263 syllabification, 50, 51, 54, 55, 56, 65, 67, 70, 72, 73, 74, 92, 93 syncope, 95, 96, 97, 179, 197, 207 Syriac, 12, 16, 19, 74, 77, 84, 96, 191, 200, 222, 252, 298, 317 Talmud, 12, 13 Taylor, J.E., 16, 17, 18, 19, 37 Tigris River, 7, 9 transcription system, XXIX, XXX, 16, 18, 19, 24, 26, 28, 34, 39, 46,

53, 54, 55, 84, 111, 153, 270, 293, 294 Urdu, 31 Urūmīya (Iran), 20 Vecchietti, Gerolamo, 12, 36 verbs aspect imperfective, XXXIII, 70, 73, 79, 82, 83, 94, 97, 134, 174, 175, 177-225, 233, 234. 239, 241, 242, 243, 244. 245, 250, 260, 261, 281, 286.299 perfective, XXX, XXXIII, 69, 96, 134, 174, 175, 177-225, 232, 234, 239, 240, 242, 243, 244, 250, 251, 260, 261, 286, 294 auxiliary verbs, 11, 25, 225, 230, 233, 234, 345 inchoative *qām*, 233, 239 modal verbs, 233, 239 complements, 92, 109, 137, 157, 174, 181, 193, 203, 211, 224, 228, 299 copula, 77, 87, 89, 90, 109, 147, 156, 158, 166, 171, 199, 229-34, 237, 250, 252, 253, 262, 314, 334 desire and communication, 138, 246 historical present tense, 241 knowledge and opinion, 240, 242 lexical verbs, 225 mood imperative, XXV, XXX, XXXIII, 73, 174, 175, 177-225, 244, 246, 247, 250 indicative, 70, 92, 174, 175, 179, 180, 183, 187, 191, 195, 197, 198, 199, 203, 211, 224, 225, 233, 244, 245, 247, 248, 249, 250, 281

subjunctive, XXXIII, 82, 92, 94, 134, 174, 175, 178, 179, 180, 187, 196, 197, 203, 211, 224, 225, 233, 235, 236, 238, 244, 245, 246, 247, 248, 249, 250, 253 negation, 70, 80, 83, 92, 97, 135, 178, 179, 183, 190, 195, 196, 210, 228, 230 oblique complements, 137, 161, 228, 252 participles, XXX, XXXIII, 118, 121, 174, 175, 176, 177, 179, 180, 183, 187, 191, 192, 196, 197, 211, 215, 221, 233, 239, 241, 252, 253 phrasal verbs, 34, 35, 79, 109, 137, 145, 225, 226, 227, 228, 229, 234, 294 pseudoverbs, 199, 235, 237, 238, 313 roots geminates, 217, 261 quadriradical verbs, 97, 111, 123, 199, 210, 211, 212, 293 strong roots, 206, 260 stems C-stem, 174, 199, 210, 211, 212, 213, 214, 215, 216, 217, 218, 219, 220, 221, 253, 260, 299, 343 D-stem, 75, 174, 186, 199, 200, 201, 202, 203, 204, 205, 206, 207, 208, 209, 210, 215, 217, 218, 219, 220, 222, 253, 260, 264 G-stem, 94, 118, 121, 174, 175, 176, 199, 200, 201, 202, 203, 204, 205, 206, 207, 208, 212, 213, 215, 216, 217, 218, 219, 222, 223, 245, 290, 294, 299

Q-stem, 174, 199, 210, 213, 2.24transitivity, XXXIII, 134, 137, 155, 156, 157, 175, 176, 178, 200, 201, 212, 239, 240, 241, 242, 251, 252, 294 verbs of perception, 166 voice active, XXX, 118, 174, 175, 179, 180, 241, 251, 252 middle, 160, 199, 223, 224, 225, 251, 252, 260 passive, XXXIII, 121, 134. 174, 175, 176, 190, 196, 197, 223, 230, 233, 234, 251, 252, 253, 254, 260, 262, 279, 289, 290 analytic passive, 230, 234, 252 impersonal passive, 223, 234, 252, 254, 260, 262 vowels breaking, 38, 74, 88, 187, 258 centralization, 19, 38, 105, 106 diphthongs, 51, 56, 59, 61, 62, 66, 73, 74, 76, 77, 83, 92, 105, 187, 207, 253, 257, 258, 261 hiatus, 13, 16, 28, 69, 74, 75 lax, XXIX, 38, 59, 63, 67, 75, 76, 77, 78, 83, 95, 96, 97, 192, 210, 217 length, 75 principal vowels, 72, 92 tense, XXIX, 59, 78, 79, 83, 94, 95, 96, 97, 109, 210, 212, 227, 229, 233, 238, 239, 241, 242, 243, 260, 262 umlaut, 74, 83, 192, 208, 209 Yahia Bihram, 16, 17, 18, 19, 367 Yamauchi, Edwin, 3, 10, 371 yā-ye ešārat, 143, 166 vā-ve vahdat, 145 Zoroastrianism, XXII, 4, 32, 275, 365